18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 1 of 92

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

In re:

CENVEO, INC., et al.,¹

Chapter 11

Case No. 18-22178 (RDD)

Debtors.

(Jointly Administered)

INTERIM ORDER PURSUANT TO 11 U.S.C. §§ 105, 361, 362, 363, 364, AND 507 (I) AUTHORIZING THE DEBTORS TO OBTAIN SENIOR SECURED PRIMING SUPERPRIORITY POSTPETITION FINANCING, (II) GRANTING LIENS AND SUPERPRIORITY ADMINISTRATIVE EXPENSE CLAIMS, (III) AUTHORIZING USE OF CASH COLLATERAL, (IV) GRANTING ADEQUATE PROTECTION, (V) MODIFYING THE AUTOMATIC STAY; (VI) SCHEDULING A FINAL HEARING, AND (VII) GRANTING RELATED RELIEF

Upon the motion (the "<u>Motion</u>")² of the above-captioned debtors and debtors-inpossession (collectively, the "<u>Debtors</u>") pursuant to sections 105, 361, 362, 363, 364(c)(1), 364(c)(2) and 364(c)(3), 364(d), 364(e), and 507 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>"), Rules 2002, 4001, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (as amended, the "<u>Bankruptcy Rules</u>"), and Rules 2002-1, 4001-2, 9006-1, 9013-1, 9014-1 and 9014-2 of the Local Bankruptcy Rules (the "<u>Local Rules</u>") for the United States Bankruptcy Court for the Southern District of New York (this "<u>Court</u>"), requesting, among other things:

¹ The last four digits of Cenveo, Inc.'s tax identification number are 0533. Due to the large number of Debtor entities in these chapter 11 cases, for which the Debtors have requested joint administration, a complete list of the Debtor entities and the last four digits of their federal tax identification numbers is not provided herein. A complete list of such information may be obtained on the website of the Debtors' proposed claims and noticing agent at https://cases.primeclerk.com/cenveo. The location of the Debtors' service address for purposes of these chapter 11 cases is: 777 Westchester Avenue, Suite 111, White Plains, New York 10604.

² Capitalized terms used but not defined herein shall have the meanings ascribed to them in the DIP Loan Documents (as defined below) or <u>Appendix 1</u>.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 2 of 92

(1) authorization for Cenveo Corporation, as debtor and debtor-in-possession (the

"Borrower") to obtain, and for each of the other Debtors to guarantee, unconditionally, on a joint

and several basis, postpetition financing in the form of:

(a) a revolving credit and letter of credit facility in accordance with the terms and conditions set forth in the Prepetition ABL Credit Agreement (as defined in **Appendix 1**), as ratified and amended by that certain Ratification and Amendment Agreement, dated as of February 5, 2018 (the "<u>Ratification Agreement</u>"), a copy of which is attached hereto as <u>Exhibit 1</u> (as amended, supplemented or otherwise modified from time to time in accordance with the terms and conditions set forth herein, the "<u>ABL Credit Agreement</u>"), by and among the Borrower, the Guarantors, Bank of America, N.A., as administrative agent and collateral agent (in such capacities, "<u>ABL Agent</u>"), and the lenders party thereto (the "<u>ABL Lenders</u>", and together with the ABL Agent, the "<u>ABL Lender Parties</u>"), and the other Loan Documents (as defined in the ABL Credit Agreement, the "<u>ABL Loan Documents</u>"), in a maximum aggregate principal amount of \$190,000,000 (the "<u>ABL DIP Facility</u>"); and

(b) a term loan facility as set forth in that certain Senior Secured Super-Priority Debtor in Possession Term Loan Credit Agreement, substantially in the final form attached hereto as Exhibit 2 (as amended, supplemented or otherwise modified from time to time, the "DIP Term Loan Agreement", and together with ABL Credit Agreement, the "DIP Loan Agreements"), by and among the Borrower, Guarantors, Wilmington Savings Fund Society, FSB, as administrative agent and collateral agent (in such capacities, the "DIP Term Agent", and together with the ABL Agent, the "DIP Agents"), and the lenders party thereto (the "DIP Term Lenders", and together with the ABL Lenders, the "DIP Lenders", and the DIP Term Lenders, together with the DIP Term Agent, the "DIP Term Parties", and together with the DIP ABL Lender Parties, the "DIP Secured Parties) and all other agreements, documents and instruments executed and delivered, to or in favor of the DIP Term Agent and DIP Term Lenders in connection therewith (the "DIP Term Loan Documents" and together with the ABL Loan Documents, the "DIP Loan Documents"), in a maximum principal amount of \$100,000,000 (the "DIP Term Facility", together with the ABL DIP Facility, the "DIP Facilities");

(2) authorization for the Debtors to execute, deliver and enter into the DIP Loan

Agreements and the other DIP Loan Documents and to perform all of their respective obligations

thereunder and such other and further acts as may be required in connection with the DIP Loan

Documents;

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 3 of 92

(3) authorization for the Debtors to pay (a) all amounts, obligations and liabilities owing or payable to the DIP Term Parties under the DIP Term Facilities pursuant to the respective DIP Loan Documents, including, without limitation, all principal, interest, fees, commitment fees, administrative agent fees, audit fees, closing fees, servicing fees, facility fees, any other fees, costs, expenses, charges and disbursements of the respective DIP Term Parties (including the reasonable and documented fees and expenses of each of the DIP Secured Parties' attorneys, advisors, accountants and other consultants) any obligations in respect of indemnity claims, whether contingent or absolute, including, without limitation, all "Obligations" as defined in the DIP Term Loan Agreement (the "<u>DIP Term Obligations</u>") and (b) all Revolving Loans, Swingline Loans, Letters of Credit, Credit Extensions and to incur indebtedness and other DIP Obligations (as defined in the Ratification Agreement and hereinafter referred to as the "<u>Postpetition ABL Obligations</u>" and, together with the DIP Term Obligations, the "<u>DIP Obligations</u>"), in each case, as expressly set forth in the applicable DIP Loan Documents;

(4) authorization for the Debtors to remit all proceeds of ABL Priority Collateral to the ABL Agent and ABL Lenders for application, or deemed application, first to all Prepetition ABL Obligations (as defined below), until such obligations are fully repaid in accordance with the ABL Credit Agreement, and then to the repayment of all Postpetition ABL Obligations (as defined below) in accordance with the ABL Credit Agreement and the other ABL Loan Documents, and to deem all Prepetition ABL Obligations with respect to Letters of Credit and Bank Products to have been issued or provided, as applicable, under the ABL Credit Agreement and the other ABL Loan Documents;

(5) authorization for the Debtors, immediately upon entry of this interim order (this "<u>Interim Order</u>"), to use proceeds of the DIP Term Facility in an amount up to the Interim Term

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 4 of 92

Financing Amount (as defined below) during the Interim Financing Period (as defined below) and Cash Collateral (as defined below) to be applied to the repayment of the Prepetition ABL Obligations until paid in full, including interest and fees payable under the Prepetition ABL Credit Agreement through the date of repayment, which repayment shall be final and indefeasible, subject only to the ABL Challenge Period (as defined below);

(6) the grant and approval of superpriority administrative claim status, pursuant to section 364(c)(1) of the Bankruptcy Code, to the DIP Agents, for the benefit of themselves and the other DIP Lenders, in respect of all DIP Obligations (as defined below), subject to the Carve Out;

(7) approval of the grant to DIP Agents and DIP Lenders of valid, enforceable, nonavoidable, automatically and fully perfected DIP Liens (as defined below) in all DIP Collateral (as defined below), including, without limitation, all property constituting Cash Collateral (as defined below), to secure the DIP Obligations (defined below), which DIP Liens shall be subject to the relative rankings and priorities and other terms set forth herein;

(8) authorization for the Debtors to use Cash Collateral and all other Prepetition Collateral in which any of the Prepetition Secured Parties (as defined below) have an interest in accordance with this Interim Order and providing adequate protection to the Prepetition Secured Parties to the extent set forth herein;

(9) the modification of the automatic stay imposed by section 362 of the BankruptcyCode to the extent necessary to implement and effectuate the terms and provisions of this InterimOrder to the extent hereinafter set forth;

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 5 of 92

(10) the scheduling of a final hearing on the Motion (the "<u>Final Hearing</u>") to consider entry of a final order (the "<u>Final Order</u>") granting the relief requested in the Motion on a final basis, and approving the form of notice with respect to the Final Hearing; and

(11) related relief.

The initial hearing on the Motion having been held by this Court on February 2, 2018 (the "Interim Hearing"); and upon the record of, and representations made at, such hearing, including the record made by the Debtors in, among other things, the Motion, the Declaration of Ayman Zameli, Executive Vice President of Cenveo, Inc. (I) in Support of Chapter 11 Petitions and First Day Pleadings and (II) Pursuant to Local Bankruptcy Rule 1007-2 (the "First Day Declaration") and any exhibits thereto, the Declaration of Neil Augustine in Support of Motion of the Debtors for Entry of Interim and Final Orders (I) Authorizing the Debtors to Obtain Senior Secured Priming Superpriority Postpetition Financing, (II) Granting Liens and Superpriority Administrative Expense Claims, and (III) Authorizing Use of Cash Collateral, (IV) Granting Adequate Protection, (V) Scheduling a Final Hearing, and (VI) Granting Related Relief (the "Augustine Declaration,"), and the filings and pleadings in the above-captioned chapter 11 cases (collectively, the "Cases"); and upon the limited objection to the Motion; and, after due deliberation, the Court having found and concluded that the interim relief requested in the Motion and granted herein is fair and reasonable and is in the best interests of Debtors, their estates, their stakeholders and other parties in interest, and represents a sound exercise of the Debtors' business judgment and is essential for the continued operation of the Debtors' businesses; and it appearing to the Court that granting the interim relief requested in the Motion is necessary to avoid immediate and irreparable harm to the Debtors and their estates pending the Final Hearing; and due and appropriate notice of the Motion, the interim relief requested therein,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 6 of 92

and the Interim Hearing (the "Notice") being adequate under the circumstances, the Notice having been served by the Debtors in accordance with Bankruptcy Rules 4001 and 9014 and the Local Bankrkuptcy Rules on (a) the United States Trustee for the Southern District of New York (the "U.S. Trustee"); (b) the holders of the 50 largest unsecured claims against Cenveo (on a consolidated basis) ("50 Largest Unsecured Creditors"); (c) counsel to the ABL Agent and the ABL Agent; (d) counsel to the DIP Term Agent; (e) counsel to the ad hoc group of First Lien Noteholders; (f) counsel to Brigade Capital Management, LP, in its capacity as a First Lien Noteholder and Second Lien Noteholder; (g) the First Lien Notes Trustee; (h) the Second Lien Notes Trustee; (i) the FILO Notes Trustee; (j) the indenture trustee of the 6.00% Senior Unsecured Notes due 2024; (k) counsel to the members of the ad hoc committee of certain First Lien Noteholders; (1) the Pension Benefit Guaranty Corporation; (m) the United States Attorney's Office for the Southern District of New York; (n) the Internal Revenue Service; (o) the United States Securities and Exchange Commission; (p) the Environmental Protection Agency and all similar state environmental agencies; (q) the attorneys general in the states where Cenveo conducts its business operations; (r) Debtors' landlords, (s) all parties which, to the best of the Debtors' knowledge, information, and belief, have asserted or may assert a lien in the Debtors' assets; and (t) any party that has requested notice pursuant to Bankruptcy Rule 2002. (collectively, the "Notice Parties"); and the opportunity for a hearing on the Motion was appropriate, and no other notice being required except as set forth herein with respect to the Final Hearing; and all objections to the relief requested in the Motion having been withdrawn, resolved or overruled by the Court for the reasons stated by it at the Interim Hearing; and after due deliberation and sufficient cause appearing therefor;

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 7 of 92

THE COURT HEREBY MAKES THE FOLLOWING FINDINGS OF FACT AND CONCLUSIONS OF LAW:

A. <u>Petition</u>. On February 2, 2018 (the "<u>Petition Date</u>"), each Debtor filed a voluntary petition (each, a "<u>Petition</u>") under chapter 11 of the Bankruptcy Code. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee or examiner has been appointed in any of the Cases.

B. <u>Jurisdiction and Venue</u>. This Court has jurisdiction over this case and the Motion pursuant to 28 U.S.C. §§ 157(a)-(b) and 1334(b). This is a core proceeding pursuant to 28 U.S.C. § 157(b). Venue is proper before this Court pursuant to 28 U.S.C. §§ 1408 and 1409. The statutory and legal predicates for the relief sought herein are sections 105, 361, 362, 363, 364 and 507 of the Bankruptcy Code and Bankruptcy Rules 2002, 4001, 9013 and 9014 and Local Rules 2002-1, 4001-2, 9006-1, 9013-1, 9014-1 and 9014-2.

C. <u>Committee Formation</u>. As of the date hereof, no official committee of unsecured creditors under section 1102 of the Bankruptcy Code (the "<u>Committee</u>") or any other statutory committee has been appointed in the Cases.

D. <u>Notice</u>. The Notice was given in the manner described in the Motion. Under the circumstances, the Notice given by the Debtors of the Motion, the Interim Hearing and the relief granted under this Interim Order constitutes due and sufficient notice thereof and complies with Bankruptcy Rule 4001 and the Local Rules.

E. <u>Debtors' Acknowledgments and Agreements</u>. In requesting the DIP Facilities, and in exchange for and as a material inducement to the DIP Lenders to agree to provide the DIP Facilities, and as a condition to providing financing under the DIP Facilities, subject in all

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 8 of 92

respects to the rights of the Committee and other parties in interest set forth in section 4.1 of this Interim Order, the Debtors admit, stipulate, acknowledge and agree with respect to the Prepetition Secured Parties, and the liens, claims, and rights of the Prepetition Secured Parties, as follows.

(i) Prepetition ABL Loan Documents. The Debtors admit, stipulate, acknowledge and agree that, prior to the Petition Date, the ABL Agent and the ABL Lenders made loans, advances and provided other financial accommodations to Debtors pursuant to the terms and conditions set forth in (1) the Prepetition ABL Credit Agreement; and (2) all other agreements, documents and instruments executed and/or delivered with, to, or in favor of ABL Agent or any ABL Lender in connection with the Prepetition ABL Credit Agreement, including, without limitation, all security agreements, notes, guarantees, mortgages, Uniform Commercial Code financing statements and all other related agreements, documents and instruments executed and/or delivered in connection therewith or related thereto (all of the foregoing, together with the Prepetition ABL Credit Agreement, as all of the same have heretofore been amended, supplemented, modified, extended, renewed, restated and/or replaced at any time prior to the Petition Date, collectively, the "Prepetition ABL Loan Documents"). Copies of the operative Prepetition ABL Loan Documents are available upon written request, by mail or e-mail, to the Debtors' proposed counsel.

(ii) <u>Prepetition ABL Obligations</u>. The Debtors admit, stipulate, acknowledge and agree that, as of the Petition Date, without giving effect to the Reaffirmation (as defined below), the Debtors were indebted to the ABL Agent and the ABL Lenders under the Prepetition ABL Loan Documents in respect of Revolving Loans, Letters of Credit, Bank Products, Credit Extensions and all other Obligations (each as defined in the Prepetition ABL Credit Agreement)

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 9 of 92

in an aggregate outstanding principal amount of not less than approximately \$142,183,541.19, plus interest accrued and accruing thereon, together with all costs, fees, expenses and other charges accrued, accruing or chargeable with respect thereto in accordance with the Prepetition ABL Loan Documents (collectively, the "Prepetition ABL Obligations", and together with the Postpetition ABL Obligations, the "<u>ABL Obligations</u>"). Subject to the rights of the Committee and other parties in interest set forth in Section 4.1(a) hereof, the Prepetition ABL Obligations constitute allowed, legal, valid, binding, enforceable and non-avoidable obligations of Debtors, and are not subject to any offset, defense, counterclaim, avoidance, recharacterization or subordination pursuant to the Bankruptcy Code or any other applicable law, and the Debtors do not possess, shall not assert, hereby forever release, and are forever barred from bringing any claim, counterclaim, setoff or defense of any kind, nature or description which would in any way affect the validity, enforceability and non-avoidability of any of the Prepetition ABL Obligations or the ABL Liens in the Prepetition ABL Collateral.

(iii) <u>Prepetition ABL Collateral</u>. The Debtors admit, stipulate, acknowledge and agree that, as of the Petition Date, the Prepetition ABL Obligations were fully secured pursuant to the Prepetition ABL Loan Documents by valid, perfected, enforceable and nonavoidable liens upon and security interests in all the "Collateral" (as defined in the Prepetition ABL Credit Agreement and hereinafter referred to as the "<u>Prepetition ABL Collateral</u>"), including without limitation, (x) first-priority liens on and security interests in all of the "ABL Collateral" (as defined in that certain First Lien Intercreditor Agreement (as defined below)) owned by each Debtor (all such ABL Collateral in existence on the Petition Date, or the identifiable proceeds thereof, the "<u>Prepetition ABL Priority Collateral</u>") and (y) second priority liens on and security interests in all Fixed Asset Priority Collateral (as defined in the First Lien

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 10 of 92

Intercreditor Agreement) owned by each such Debtor (all such Fixed Asset Priority Collateral in existence on the Petition Date, or the identifiable proceeds thereof, the "Prepetition Term Loan Priority Collateral"), subject only to (A) the liens specifically permitted under Section 7.01 of the Prepetition ABL Credit Agreement to the extent that such security interests, liens or encumbrances are (1) valid, perfected and non-avoidable security interests, liens or encumbrances existing as of the Petition Date, and (2) senior to and have not been or are not subject to being subordinated to ABL Agent's and ABL Lenders' liens on and security interests in the Prepetition ABL Collateral under the Prepetition ABL Loan Documents or otherwise avoided, and, in each instance, only for so long as and to the extent that such encumbrances are and remain senior and outstanding (hereinafter referred to as the "Permitted ABL Liens"), and (B) the terms and conditions of the Prepetition Intercreditor Agreements (as defined below). The Debtors do not possess and will not assert any claim, counterclaim, setoff or defense of any kind, nature or description which would in any way affect the validity, enforceability and nonavoidability of any of the ABL Agent's and the ABL Lenders' liens, claims or security interests in the Prepetition ABL Collateral.

(iv) [Reserved.]

(v) <u>First Lien Notes</u>. The Debtors admit, stipulate, acknowledge and agree that pursuant to that certain Indenture dated as of June 16, 2014 (as amended, supplemented, restated or otherwise modified prior to the Petition Date, the "<u>First Lien Notes Indenture</u>"), by and among The Bank of New York Mellon, as indenture trustee and collateral agent (together with its permitted successors and assigns, the "<u>First Lien Notes Trustee</u>"), and together with the ABL Agent, the "<u>Prepetition Agents</u>"), Cenveo Corporation, as issuer, and the guarantors party thereto, and all the other agreements, documents, instruments and certificates executed or

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 11 of 92

delivered in connection therewith, including, without limitation, the "Security Documents" (as defined in the First Lien Notes Indenture) (collectively, together with the First Lien Notes Indenture, the "<u>First Lien Notes Documents</u>"; and together with the Prepetition ABL Loan Documents, the "<u>Prepetition Loan Documents</u>"), Cenveo Corporation issued the 6.000% Senior Priority Secured Notes due 2019 (the "<u>First Lien Notes</u>"; the holders of such First Lien Notes, collectively, the "<u>First Lien Noteholders</u>", and together with the First Lien Notes Trustee, the "<u>First Lien Secured Parties</u>"; the First Lien Noteholders, and the ABL Lenders, collectively, the "<u>Prepetition Lenders</u>"; the First Lien Secured Parties and the ABL Secured Parties, collectively, the "<u>Prepetition Secured Parties</u>") and the guarantors provided the "Note Guarantee" as defined in the First Lien Notes Indenture.

(vi) <u>First Lien Notes Obligations</u>. The Debtors admit, stipulate, acknowledge and agree that, as of the Petition Date, the First Lien Notes Parties were indebted to the First Lien Secured Parties, without defense, counterclaim or offset of any kind, in the aggregate outstanding principal amount under the First Lien Notes Documents of not less than \$540,000,000 under the First Lien Notes Documents, *plus* accrued but unpaid interest, *plus* any other amounts incurred or accrued but unpaid prior to the Petition Date in accordance with the First Lien Notes Documents, including, without limitation, principal, accrued and unpaid interest at the default rate, premiums, any reimbursement obligations (contingent or otherwise), any fees, costs, charges, expenses and disbursements (including, without limitation, attorneys' fees, financial advisors' fees, related expenses and disbursements), indemnification obligations, any other charges, amounts and costs of whatever nature owing, whether or not contingent, whenever arising, accrued, accruing, due, owing or chargeable in respect thereof, including all "Notes Obligations" as defined in the First Lien Notes Indenture, in each case, to the extent provided in the First Lien Notes Documents

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 12 of 92

(collectively, the "<u>First Lien Notes Obligations</u>", and together with the Prepetition ABL Obligations, the "<u>Prepetition Secured Obligations</u>").

First Lien Notes Liens and First Lien Notes Collateral. (vii) The Debtors admit, stipulate, acknowledge and agree that to secure the First Lien Notes Obligations, the First Lien Notes Parties granted to the First Lien Notes Trustee, for its benefit and the benefit of the First Lien Noteholders, properly perfected continuing liens, mortgages and security interests (the "First Lien Notes Liens", and together with the ABL Liens on the Prepetition ABL Collateral securing the Prepetition ABL Obligations, the "Prepetition Liens") in all "Collateral" under and as defined in the First Lien Notes Indenture (the "First Lien Notes Collateral"), including without limitation, (x) first-priority liens on and security interests in all Prepetition Term Loan Priority Collateral and (y) second priority liens on and security interests in all Prepetition ABL Priority Collateral, subject only to (A) the liens specifically permitted under Section 4.10 of the First Lien Notes Indenture to the extent that such security interests, liens or encumbrances are (1) valid, perfected and non-avoidable security interests, liens or encumbrances existing as of the Petition Date, and (2) senior to and have not been or are not subject to being subordinated to First Lien Notes Trustee's and First Lien Noteholders' liens on and security interests in the First Lien Collateral under the First Lien Loan Documents or otherwise avoided, and, in each instance, only for so long as and to the extent that such encumbrances are and remain senior and outstanding (the "Permitted Term Liens" and together with the Permitted ABL Liens, the "Permitted Liens"), subject in all cases to the terms and conditions set forth in the Prepetition Intercreditor Agreements. The Debtors do not possess and will not assert any claim, counterclaim, setoff or defense of any kind, nature or description which would in any way affect

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 13 of 92

the validity, enforceability and non-avoidability of any of First Lien Notes Trustee's and First Lien Noteholders' liens, claims or security interests in the First Lien Notes Collateral.

(viii) Validity of Prepetition Liens and Prepetition Secured Obligations. The Debtors admit, stipulate, acknowledge and agree that: (1) the Prepetition Secured Obligations constitute legal, valid, enforceable, non-avoidable and binding obligations of each of the Debtors; (2) no offsets, challenges, objections, defenses, claims or counterclaims of any kind or nature whatsoever to the Prepetition Secured Obligations or Prepetition Liens exist, and no portion of the Prepetition Secured Obligations or Prepetition Liens is subject to any offsets, challenges, objections, defenses, claims or counterclaims of any kind or nature whatsoever, including, without limitation, avoidance, disallowance, reduction, surcharge, disgorgement, recharacterization or subordination, whether pursuant to the Bankruptcy Code or applicable nonbankruptcy law; (3) the Prepetition Loan Documents are valid and enforceable by the Prepetition Agents and Prepetition Lenders against each of the Debtors (other than in respect of the stay of enforcement arising under section 362 of the Bankruptcy Code); (4) the Prepetition Liens were fully perfected as of the Petition Date and constitute legal, valid, binding, enforceable and perfected liens in and to the Prepetition Collateral and are not subject to avoidance, disallowance, reduction, surcharge, disgorgement, recharacterization or subordination, whether pursuant to the Bankruptcy Code or applicable non-bankruptcy law; and (5) the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the Debtors and their estates hereby forever, unconditionally and irrevocably release, discharge and acquit each of the Prepetition Agents and Prepetition Lenders and each of their respective successors, assigns, affiliates, parents, subsidiaries, partners, controlling persons, representatives,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 14 of 92

agents, attorneys, advisors, financial advisors, consultants, professionals, officers, directors, members, managers, shareholders, and employees, past, present and future, and their respective heirs, predecessors, successors and assigns (collectively, the "Released Parties") of and from any and all claims, controversies, disputes, liabilities, obligations, demands, damages, expenses (including, without limitation, attorneys' fees), debts, liens, actions and causes of action of any and every nature whatsoever, whether arising in law or otherwise, and whether known or unknown, matured or contingent, arising under, in connection with or relating to the Prepetition Secured Obligations, the Prepetition Loan Documents, the DIP Facilities or the DIP Loan Documents, including, without limitation, (A) any so-called "lender liability" or equitable subordination claims or defenses, (B) any and all claims (as defined in the Bankruptcy Code) and causes of action arising under the Bankruptcy Code, and (C) any and all offsets, defenses, claims, counterclaims, set off rights, objections, challenges, causes of action and/or choses in action of any kind or nature whatsoever, whether arising at law or in equity, including any recharacterization, subordination, avoidance, or other claim arising under or pursuant to section 105 or chapter 5 of the Bankruptcy Code or under any other similar provisions of applicable state or federal law, including, without limitation, any right to assert any disgorgement or recovery, in each case, with respect to the extent, amount, validity, enforceability, priority, security and perfection of any of the Prepetition Secured Obligations, the Prepetition Loan Documents or the Prepetition Liens, and the Debtors further waive and release any defense, right of counterclaim, right of setoff or deduction to the payment of the Prepetition Secured Obligations that the Debtors now have or may claim to have against the Released Parties, arising under, in connection with, based upon or released to any and all acts, omissions or events occurring prior to entry of this Interim Order. For the avoidance of doubt, the release contemplated by the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 15 of 92

preceding sentence is not being approved pursuant to this Interim Order and the rights of the Committee and all other parties in interest regarding such release are fully reserved and preserved pending the Final Hearing on the Final Order.

(ix) <u>First Lien Intercreditor Agreement</u>. The Debtors admit, stipulate, acknowledge and agree that the ABL Agent, the FILO Notes Trustee, the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), and the Debtors entered into that certain Intercreditor Agreement, dated as of June 26, 2014, (as amended, supplemented or otherwise modified from time to time, the "<u>First Lien Intercreditor Agreement</u>"), which (x) sets forth the relative priority of the security interests of ABL Agent, FILO Notes Trustee, and First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent) in the assets and properties of the Debtors, and (y) provides certain other rights and obligations between ABL Agent, on behalf of the ABL Lenders, FILO Notes Trustee, on behalf of FILO Noteholders, on the one hand, and the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), on behalf of the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), on behalf of the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), on behalf of the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), on behalf of the First Lien Notes Trustee (in its capacity as Fixed Asset Collateral Agent), on behalf of the First Lien Noteholders, on the other hand, relating to the Prepetition Collateral.

(x) <u>Second Lien Intercreditor Agreement</u>. The Debtors admit, stipulate, acknowledge and agree that the ABL Agent, the First Lien Notes Trustee, the Second Lien Notes Trustee, and the Debtors entered into that certain Intercreditor Agreement, dated as of June 26, 2014, (as amended, supplemented or otherwise modified from time to time, the "<u>Second Lien Intercreditor Agreement</u>"), which (x) sets forth the relative priority of the security interests of ABL Agent, on behalf of the ABL Lenders, on the one hand, and the First Lien Notes Trustee, on behalf of the First Lien Noteholders, and Second Lien Notes Trustee, on behalf of the Second Lien Noteholders, in the assets and properties of the Debtors, and (y) provides certain other rights and obligations between the ABL Agent, on behalf of the ABL Lenders, the First Lien Noteholders, the First Lien Noteholders, the First Lien Noteholders, the First Lien Noteholders, the Second Lien Noteholders, the First Lien Noteholders, the First Lien Noteholders, the First Lien Noteholders, the First Lien Noteholders, the Second Lien Noteholders, the First Lien Noteholders, the Second Lien Noteholders, the First Lien Noteholders

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 16 of 92

Notes Trustee, on behalf of the First Lien Noteholders and the Second Lien Notes Trustee, on behalf of the Second Lien Noteholders, relating to the Prepetition Collateral.

(xi) <u>FILO Notes Intercreditor Agreement</u>. The Debtors admit, stipulate, acknowledge and agree that the ABL Agent, the FILO Notes Trustee, and he Debtors entered into that certain Intercreditor Agreement, dated as of June 10, 2016, (as amended, supplemented or otherwise modified from time to time, the "<u>FILO Notes Intercreditor Agreement</u>", and together with the First Lien Intercreditor Agreement and the Second Lien Intercreditor Agreement, collectively, the "<u>Prepetition Intercreditor Agreements</u>"), which (x) sets forth the relative priority of the security interests of the ABL Agent and the FILO Notes Trustee in the assets and properties of the Debtors, and (y) provides certain other rights and obligations between the ABL Agent, on behalf of the ABL Lenders, on the one hand, and the FILO Notes Trustee and FILO Noteholders, on the other hand, relating to the Prepetition Collateral.

(xii) <u>Cash Collateral</u>. The Debtors admit, stipulate, acknowledge and agree that all of the Debtors' cash, wherever located, including any cash in deposit accounts of the Debtors, whether as Prepetition Collateral or proceeds of other Prepetition Collateral, whether existing on the Petition Date or thereafter, constitutes "cash collateral" of the Prepetition Secured Parties within the meaning of Section 363(a) of the Bankruptcy Code ("<u>Cash Collateral</u>").

F. Findings Regarding the Postpetition Financing.

(i) <u>Postpetition Financing</u>. The Debtors have requested from each of the DIP Agents and the DIP Lenders, and the DIP Agents and DIP Lenders are willing, subject to the terms of this Interim Order, to extend, certain loans, advances and other financial accommodations on the terms and conditions set forth in this Interim Order, the DIP Loan Agreements and the other DIP Loan Documents, respectively.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 17 of 92

Need for Postpetition Financing. The Debtors do not have sufficient (ii) liquidity, including cash collateral, to operate their businesses in the ordinary course of business without the financing requested in the Motion. The Debtors' ability to maintain business relationships with their vendors, suppliers and customers, to pay their employees, pay certain fees and expenses as set forth herein and in the DIP Loan Documents, and to otherwise fund their operations is essential to the Debtors' continued viability as the Debtors seek to maximize the value of the assets of the Estates (as defined below) for the benefit of all creditors of the Debtors. The ability of the Debtors to obtain sufficient working capital and liquidity through the proposed postpetition financing arrangements with the DIP Agents and DIP Lenders as set forth in this Interim Order, the DIP Loan Agreements, and other DIP Loan Documents, as applicable, is vital to the preservation and maintenance of the going concern value of each Debtor. Accordingly, the Debtors have an immediate need to obtain the postpetition financing to, among other things, permit the orderly continuation of the operation of their businesses, minimize the disruption of their business operations, and preserve and maximize the value of the assets of the Debtors' bankruptcy estates (as defined under section 541 of the Bankruptcy Code, the "Estates") to maximize the recovery to all creditors of the Estates.

(iii) <u>Priming of Prepetition Liens on Term Loan Priority Collateral</u>. The Debtors believe that the priming of the liens granted in favor of each of the Prepetition Secured Parties in the Prepetition Term Priority Collateral under Section 364(d)(1) of the Bankruptcy Code, by the DIP Term Facility, as contemplated by the DIP Loan Documents and this Interim Order, will enable the Debtors to continue to operate their businesses for the benefit of their Estates and stakeholders. The First Lien Notes Trustee, on behalf of the First Lien Noteholders, has not objected or shall be deemed to not object to the priming of the First Lien

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 18 of 92

Notes Liens in Prepetition Collateral on the terms set forth herein, and along with each of the Prepetition Secured Parties with an interest in the Term Loan Priority Collateral, is entitled to receive adequate protection as and to the extent set forth herein.

(iv) <u>No Credit Available on More Favorable Terms</u>. The Debtors are unable to procure financing in the form of unsecured credit allowable as an administrative expense under sections 364(a), 364(b), or 503(b)(1) of the Bankruptcy Code or in exchange for the grant of a superpriority administrative expense, junior liens on encumbered property of the Estates, or liens on property of the Estates not subject to a lien pursuant to § 364(c)(1), 364(c)(2) or 364(c)(3) of the Bankruptcy Code. The Debtors assert in the Motion and demonstrated at the Interim Hearing that they have been unable to procure the necessary financing on terms more favorable, taken as a whole, than the financing offered by each of the DIP Agents and DIP Lenders pursuant to the DIP Loan Documents. After considering all alternatives, the Debtors have properly concluded, in the exercise of their sound business judgment, that the DIP Facilities represent the best financing available to them at this time, and are in the best interests of all of their stakeholders.

(v) <u>Budget</u>. The Debtors have prepared and delivered to DIP Agents and DIP Lenders an initial budget (the "<u>Initial Budget</u>"), a copy of which is attached hereto as <u>Exhibit 3</u>. The Initial Budget reflects the Debtors' anticipated cash receipts and anticipated disbursements for each calendar week during the period from the Petition Date through and including the end of the thirteenth (13th) calendar week following the Petition Date (the Initial Budget and each subsequent Approved Budget (an "<u>Approved Budget</u>"). The Debtors believe that the Initial Budget is reasonable under the facts and circumstances. The DIP Agents and DIP Lenders are relying upon the Debtors' agreement to comply with the Approved Budget, the other DIP Loan Documents, and this Interim Order in determining to enter into the postpetition

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 19 of 92

financing arrangements provided for herein. For the avoidance of doubt, all references to the Initial Budget or Approved Budget shall be subject in all respects to the Permitted Variance (as defined in the DIP Loan Documents).

Business Judgment and Good Faith Pursuant to section 364(e). Based (vi) upon the pleadings and proceedings of record in the Cases, (a) the terms of the DIP Loan Documents and this Interim Order are fair, just and reasonable under the circumstances, are ordinary and appropriate for secured financing to debtors-in-possession, reflect the Debtors' exercise of their prudent business judgment consistent with their fiduciary duties, and are supported by reasonably equivalent value and fair consideration; (b) the terms and conditions of the DIP Loan Documents and this Interim Order have been negotiated in good faith and at arms' length by and among the Debtors and DIP Agents, with all parties being represented by counsel. Any credit extended, loans made and other financial accommodations extended to the Debtors by the DIP Secured Parties, including, without limitation, pursuant to this Interim Order, have been extended, issued or made, as the case may be, in "good faith" within the meaning of section 364(e) of the Bankruptcy Code and in express reliance upon the protections offered by Bankruptcy Code section 364(e), and the DIP Facilities, the DIP Liens and the DIP Superpriority Claims shall be entitled to the full protection of Bankruptcy Code section 364(e) in the event that this Interim Order or any provision hereof is vacated, reversed or modified, on appeal or otherwise.

(vii) <u>Sections 506(c) and 552(b)</u>. The Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that as a material inducement to the DIP Secured Parties to agree to provide the DIP Facilities, and in

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 20 of 92

exchange for (a) the DIP Agents' and the DIP Lenders' willingness to provide the DIP Facilities and their agreement to subordinate their liens and superpriority claims to the Carve Out to the extent set forth herein, (b) the ABL Secured Parties' agreement to subordinate their ABL Liens in the DIP Collateral to the Carve Out to the extent set forth herein and (c) the First Lien Secured Parties' agreement to subordinate their First Lien Notes Liens to the DIP Term Liens and the Carve Out to the extent set forth herein, and (d) the consensual use of Cash Collateral in accordance with and subject to the Approved Budget and the terms of this Interim Order, each of the DIP Agents, the DIP Lenders and the Prepetition Secured Parties are entitled to receive, (1) a waiver of any equities of the case exceptions or claims under section 552(b) of the Bankruptcy Code and a waiver of unjust enrichment and similar equitable relief as set forth below, and (2) a waiver of the provisions of section 506(c) of the Bankruptcy Code subject to the terms hereof (which waiver shall be without prejudice to the contention of the DIP Agents and the DIP Lenders that section 506(c) of the Bankruptcy Code does not apply to claims arising under the DIP Facilities).

(viii) <u>Good Cause</u>. The relief requested in the Motion is necessary, essential and appropriate, and is in the best interest of and will benefit the Debtors, their creditors and their Estates, as its implementation will, among other things, provide the Debtors with the necessary liquidity to (1) minimize disruption to the Debtors' businesses and on-going operations, (2) preserve and maximize the value of the Debtors' Estates for the benefit of all the Debtors' creditors, and (3) avoid immediate and irreparable harm to the Debtors, their creditors, their businesses, their employees, and their assets.

(ix) <u>Adequate Protection</u>. The Prepetition Secured Parties are entitled pursuant to sections 361, 362, 363 and 364 of the Bankruptcy Code, to receive adequate protection against

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 21 of 92

any diminution in value of their respective interests in the Prepetition Collateral (including Cash Collateral), to the extent set forth in this Interim Order.

(x) <u>Immediate Entry</u>. Sufficient cause exists for immediate entry of this Interim Order pursuant to Bankruptcy Rule 4001(c)(2). No party appearing in the Cases has filed or made an objection to the relief sought in the Motion or the entry of this Interim Order, or any objections that were made (to the extent such objections have not been withdrawn, waived, resolved, or settled) are hereby overruled on the merits.

Based upon the foregoing, and upon the record made before the Court at the Interim Hearing, and after due consideration and good cause appearing therefor;

IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

Section 1. <u>Authorization and Conditions to Financing.</u>

1.1 <u>Motion Granted</u>. The Motion is granted to the extent provided in this Interim Order. Any objection to the entry of this Interim Order that has not been withdrawn, waived, resolved or settled, and all reservations of rights included therein, is hereby denied and overruled on the merits.

1.2 <u>Authorization to Borrow, Guaranty, and Use Loan Proceeds</u>. The Debtors are hereby authorized and empowered, effective as of the close of the Interim Hearing, to immediately borrow, incur and guarantee (as applicable) on a joint and several basis, (a) the Postpetition ABL Obligations, and the Guarantors are hereby authorized to guaranty such ABL Obligations including without limitation all ABL Obligations, up to an aggregate amount equal to the sum of \$190,000,000 (inclusive of the Prepetition ABL Obligations) for the Interim Financing Period (as defined below), including (i) approximately \$16,565,586 with respect to all Obligations (as defined in the Ratification Agreement) relating to Letters of Credit (inclusive of all Prepetition ABL Obligations with respect to Letters of Credit, which are deemed to have been

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 22 of 92

issued or provided, as applicable, under the ABL Credit Agreement and the other ABL Loan Documents), plus (ii) all interest, Bank Products, costs, fees, and expenses accrued or accruing under the ABL Credit Agreement and other ABL Loan Documents, all pursuant to the terms and conditions of this Interim Order, the ABL Credit Agreement and the other ABL Loan Documents, and (b) the DIP Term Loans in the principal amount of \$100,000,000 (after taking into account any original issue discount), of which \$50,000,000 (after taking into account any original issue discount) (the "Interim Term Financing Amount") may be borrowed by Debtors during the Interim Financing Period, plus all interest, costs, and fees, accrued or accruing under the DIP Term Loan Agreement and other DIP Term Loan Documents, all pursuant to the terms and conditions of this Interim Order, the DIP Term Loan Agreement and the other DIP Term Loan Documents, in each case during the period commencing on the date of this Interim Order through and including the entry of the Final Order (the "Interim Financing Period"). Subject to the terms and conditions contained in this Interim Order and the DIP Loan Documents, the Debtors shall use the proceeds of the DIP Facilities, including without limitation all Revolving Loans, Letters of Credit, DIP Term Loans, and other credit and financial accommodations provided by DIP Agents and DIP Lenders under the DIP Loan Agreements and the other DIP Loan Documents, solely for payment of amounts set forth in the Approved Budget and amounts owing to DIP Agents and DIP Lenders in accordance with the terms and conditions of the DIP Loan Documents and this Interim Order.

(a) Notwithstanding anything to the contrary set forth in the Ratification Agreement, for the period commencing the Ratification Effective Date through the entry date of the Final Order, the Borrower shall pay to the Administrative Agent, for the ratable benefit of the Lenders (other than any Defaulting Lender), a fee equal to the Unused Line Fee

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 23 of 92

Rate multiplied by the amount by which \$132,853,000.00 exceeds the average daily balance of outstanding Revolving Loans (other than Swingline Loans) and stated amount of outstanding Letters of Credit during any fiscal month, which fee shall be paid in lieu of the Unused Line Fee set forth in Section 2.05(a) of Exhibit A to the Ratification Agreement. Such fee shall accrue commencing on the Ratification Effective Date, and will be payable in arrears, on the first day of each fiscal month, commencing March 1, 2018; provided, however, this Section 1.2(a) shall not prejudice the ABL Agent or the ABL Lenders from seeking approval from the Court at the Final Hearing, for the full amount of the Unused Line Fee, as originally contemplated.³

1.3 <u>Financing Documents</u>

(a) <u>Authorization</u>. The Debtors are hereby authorized to enter into, execute, deliver, comply with all of the terms, conditions and covenants of, and perform any obligations under the DIP Loan Documents. Subject to the rights of the Committee and all other parties in interest under Section 4.1(a) hereof, no obligation, payment, transfer or grant of security hereunder or under the DIP Loan Documents shall be stayed, restrained, voidable, avoidable, or recoverable under the Bankruptcy Code or under any applicable law (including, without limitation, under chapter 5 of the Bankruptcy Code or under any applicable state Uniform Fraudulent Transfer Act, Uniform Fraudulent Conveyance Act, or similar statute or common law), or be subject to any defense, reduction, setoff, counterclaim, recoupment, offset, recharacterization, subordination (whether equitable, contractual or otherwise), cross-claims, or any other challenge under the Bankruptcy Code or any applicable law, rule or regulation by any person or entity.

³ All capitalized terms used in this Section 1.2(a) and not otherwise defined herein shall have their meanings set forth in the ABL Credit Agreement.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 24 of 92

Approval; Evidence of Borrowing Arrangements. (b) All terms. conditions and covenants set forth in the DIP Loan Documents (including, without limitation, each of the DIP Loan Agreements) are approved. All such terms, conditions and covenants shall be sufficient and conclusive evidence of (i) the borrowing arrangements by and among the Debtors, the DIP Agents and the DIP Lenders, and (ii) each Debtor's assumption and adoption of, and agreement to comply with, all the terms, conditions, and covenants of each DIP Loan Agreement and the other DIP Loan Documents for all purposes, including, without limitation, to the extent applicable, the payment of all DIP Obligations arising thereunder, including, without limitation, all principal, interest, fees and other expenses, including, without limitation, all of each DIP Agent's and DIP Lenders' closing, arranger and administrative fees, consultant fees, professional fees, attorney fees and legal expenses, as more fully set forth in the DIP Loan Documents. Upon execution thereof, the DIP Loan Documents shall evidence the DIP Obligations, which DIP Loan Documents and DIP Obligations shall be valid, binding and enforceable against the Debtors, their Estates and any successors thereto, including, without limitation, any trustee appointed in any of the Chapter 11 Cases or any case under chapter 7 of the Bankruptcy Code upon the conversion of any of the Chapter 11 Cases (collectively, the "Successor Cases"), and their creditors and other parties-in-interest, in each case, in accordance with the terms of this Interim Order and the DIP Loan Documents. The Debtors shall be jointly and severally liable for the DIP Obligations.

(c) <u>Payment of DIP Premiums and Other Expenses</u>. The Lender Payments, the Administrative Agent Fees, and the other fees arising in accordance with the DIP Term Loan Documents (including, for the avoidance of any doubt, any original issue discount (each, as defined in the DIP Term Documents, together the "<u>DIP Term Fees</u>")) and all

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 25 of 92

fees arising in accordance with the ABL Loan Documents (the "<u>ABL Fees</u>") are hereby approved, and the Debtors are authorized to and shall pay the DIP Term Fees and the ABL Fees as set forth in the applicable DIP Loan Documents; <u>provided</u> that any "unused line payment" originally proposed under the DIP Term Agreement is not approved. The Debtors are also authorized and directed to pay, in cash and on a current basis, all reasonable and documented out-of-pocket costs, disbursements, and expenses of the DIP Agents and the DIP Lenders incurred at any time, as provided by the DIP Loan Documents and this Interim Order in accordance with Section 5.16 hereof. Subject to the rights of the Committee and other parties in interest pursuant to Section 4.1(a) hereof with respect to a Prepetition ABL Challenge, such fees and expenses shall not be subject to any offset, defense, claim, counterclaim or diminution of any type, kind or nature whatsoever.

(d) <u>Non-Material Amendments to DIP Loan Documents</u>. Subject to the terms and conditions of the applicable DIP Loan Documents, the Debtors and the applicable DIP Agents and DIP Lenders may make ministerial or non-material amendments, modifications, or supplements to any DIP Loan Document, and the DIP Agents and the DIP Lenders may waive any provisions in the DIP Loan Documents, without further approval of the Bankruptcy Court; *provided* that Debtors shall file a notice on the Court's docket of any such ministerial or nonmaterial amendment, modification, or supplement, or waiver, not more than three (3) business days after the effective date thereof. Any amendments, modifications or supplements to any DIP Loan Documents, other than as set forth in the preceding sentence (collectively, "<u>Material</u> <u>DIP Amendments</u>"), shall be filed with the Court, and the Debtors shall provide prior written notice of the Material DIP Amendment to the Notice Parties. If no party in interest files an objection to the Material DIP Amendment with the Court within three (3) business days from the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 26 of 92

date the notice of the Material DIP Amendment is filed with the Court in accordance with this Section 1.3, the Debtors and the applicable DIP Agent are authorized and empowered to implement, in accordance with the terms of the applicable DIP Loan Documents, such Material DIP Amendment, without further notice, hearing or approval of this Court. Any Material DIP Amendment subject to a timely and unresolved objection must be approved by the Court to be effective.

1.4 <u>Payment of Prepetition Debt</u>. The Debtors are authorized during the Interim Financing Period to use the Interim Term Financing Amount to fully repay, cash collateralize or have deemed issued or incurred (in the case of Letters of Credit, Bank Products, and any other Prepetition ABL Obligations to the extent provided for in the Prepetition ABL Loan Documents), as applicable, any existing Prepetition ABL Obligations, together with all accrued and unpaid interest, fees, costs, and other changes accrued and accruing thereon (the "<u>Reaffirmation</u>) in accordance with the ABL Loan Documents and this Interim Order, including, without limitation, Sections 1.4, 1.6, and 4.1(a) of this Interim Order. For the avoidance of any doubt, the rights of the Committee and other parties in interest with respect to the validity, extent, and priority of the Prepetition ABL Obligations and the liens securing the Prepetition ABL Obligations and any Roll-Up thereof are fully reserved and preserved to the extent provided in section 4.1(a) of this Interim Order.

1.5 <u>Payments and Application of Payments</u>. All payments and transfers of Estate property to the DIP Agents as provided for, permitted and/or required under the DIP Loan Documents, which payments and transfers shall not be avoidable or recoverable from the DIP Agents or any DIP Lender under sections 547, 548, 550, 553 or any other section of the Bankruptcy Code, or by reason of any other claim, charge, assessment, or other liability, whether

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 27 of 92

by application of the Bankruptcy Code, other law or otherwise. All proceeds of the ABL Priority Collateral (as defined herein) received by the ABL Agent, any ABL Lender, or by any of the Debtors, and any other amounts or payments received by the ABL Agent, any ABL Lender or any of the Debtors in respect of the Postpetition ABL Obligations, may be applied or deemed to be applied on a dollar-for dollar basis by the ABL Agent, in its discretion, first to the repayment in full of the Prepetition ABL Obligations, and then in repayment of the Postpetition ABL Obligations, all in accordance with the ABL Credit Agreement, the other ABL Loan Documents and this Interim Order. The Debtors are further authorized to immediately use proceeds of the ABL DIP Facility, DIP Term Facility, and Cash Collateral (as defined below) to repay all of the Prepetition ABL Obligations in full, or any portion thereof, including without limitation all interest, fees, expenses, Bank Products, and costs outstanding under the Prepetition ABL Credit Agreement, through the date of repayment, which repayment shall be final and indefeasible, provided that any repayment of the Prepetition ABL Obligations shall be subject to, the rights of the Committee and other parties in interest set forth in Section 4.1(a) hereof, with respect to the validity, extent, and priority of the Prepetition ABL Obligations and the liens securing the Prepetition ABL Obligations and any Roll-Up thereof, and all such rights are fully reserved and preserved to the extent provided in section 4.1(a) of this Interim Order.

1.6 <u>Continuation of Prepetition Procedures</u>. Except to the extent expressly set forth in the Prepetition ABL Loan Documents and DIP Loan Documents, all prepetition practices and procedures for the payment and collection of proceeds of the Prepetition ABL Collateral, the turnover of cash, the delivery of property to the ABL Agent and the ABL Lenders, including the Deposit Account Control Agreements (as such term is defined in the ABL Credit Agreement)

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 28 of 92

and any other similar lockbox or blocked depository bank account arrangements, are hereby approved and shall continue without interruption.

1.7 Rights and Benefits Under Prepetition Intercreditor Agreements. Except to the extent (if any) that such terms and conditions are inconsistent with the express terms of this Interim Order, the terms and conditions of each of the Prepetition Intercreditor Agreements (including with respect to the relative rights of the ABL Agent, on behalf of the ABL Lenders, the First Lien Notes Trustee, on behalf of the Second Lien Noteholders, and the FILO Notes Trustee, on behalf of the Second Lien Noteholders, and the FILO Notes Trustee, on behalf of the Interview of this Interim Order, entry into the DIP Loan Documents, or the incurrence of the DIP Obligations. The ABL Agent, on behalf of the ABL Lenders, the First Lien Notes Trustee, on behalf of the Second Lien Notes Trustee, on behalf of the First Lien Notes Trustee, on behalf of the Second Lien Notes Trustee, on behalf of the First Lien Notes Trustee, on behalf of the PIP Obligations. The ABL Agent, on behalf of the ABL Lenders, the First Lien Notes Trustee, on behalf of the Second Lien Notes Trustee, on behalf of the First Lien Notes Trustee, on behalf of the FILO Notes Trustee having not objected to but having expressed reserved the right to object at the Final Hearing as to, among other things, the granting of adequate protection.

1.8 <u>Indemnification</u>. The Debtors are authorized to jointly and severally indemnify and hold harmless the DIP Agents, each DIP Lender and, solely in their capacities as such, each of their respective successors, assigns, affiliates, parents, subsidiaries, partners, controlling persons, representatives, agents, attorneys, advisors, financial advisors, consultants, professionals, officers, directors, members, managers, shareholders and employees, past, present and future, and their respective heirs, predecessors, successors and assigns (each, an

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 29 of 92

"<u>Indemnified Party</u>"), in accordance with the DIP Loan Agreements, which indemnification is hereby authorized and approved.

Section 2. <u>Postpetition Lien; Superpriority Administrative Claim Status.</u>

2.1 Postpetition Lien.

(a) <u>Postpetition ABL Lien Granting</u>. To secure the prompt payment and performance of any and all Postpetition ABL Obligations of the Debtors to the ABL Agent and the ABL Lenders of whatever kind, nature or description, absolute or contingent, now existing or hereafter arising, the ABL Agent, for the benefit of itself and the other the ABL Lenders, shall have and is hereby granted, effective as of the Petition Date, continuing, valid, binding, enforceable, non-avoidable, and automatically and properly perfected security interests in and liens upon all DIP Collateral (defined below), subject to the priority set forth in Section 2.1(b), below; <u>provided</u> that the foregoing shall not be deemed to prejudice, modify, or impair the rights of the Committee or any party in interest pursuant to section 4.1(a) of this Interim Order with respect to any Prepetition ABL Challenge, including as to any Roll-Up.

(b) <u>ABL Lien Priority in DIP Collateral</u>. The ABL Liens of the ABL Agent for itself and the benefit of the other ABL Lenders on the DIP Collateral ("<u>ABL Liens</u>") securing all ABL Obligations shall be first and senior in priority to all other interests and liens of every kind, nature and description, whether created consensually, by an order of the Court or otherwise, including, without limitation, liens or interests granted in favor of third parties in conjunction with sections 363, 364 or any other section of the Bankruptcy Code or other applicable law; <u>provided</u>, <u>however</u>, that the ABL Liens on the (i) ABL Priority Collateral shall be subject only to (A) Permitted ABL Liens, and (B) the Carve Out, and (ii) Term Priority Collateral, shall be subject to (A) the DIP Term Liens, (B) the First Lien Notes Adequate Protection Liens, (C) the First Lien Notes Prepetition Liens and (D) the Carve Out.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 30 of 92

(c) <u>Postpetition Term Lien Grant</u>. To secure the prompt and complete performance and payment when due (whether at the stated maturity, by acceleration or otherwise) of any and all DIP Term Obligations of Debtors to the DIP Term Agent and DIP Term Lenders of whatever kind, nature or description, absolute or contingent, now existing or hereafter arising, DIP Term Agent, for the benefit of itself and the other DIP Term Lenders, shall have and is hereby granted, effective as of the Petition Date, continuing, valid, binding, enforceable, non-avoidable, and automatically and properly perfected security interests in and liens (collectively, the "<u>DIP Term Liens</u>", and together with the ABL Liens on the DIP Collateral, the "<u>DIP Liens</u>") in and upon all DIP Collateral, subject to the priority set forth in Section 2.1(d), below.

(d) <u>Term Lien Priority in Collateral</u>. The DIP Term Liens securing all DIP Term Obligations shall be first and senior in priority to all other interests and liens of every kind, nature and description, whether created consensually, by an order of the Court or otherwise, including, without limitation, liens or interests granted in favor of third parties in conjunction with sections 363, 364 or any other section of the Bankruptcy Code or other applicable law; <u>provided</u>, <u>however</u>, that the DIP Term Liens on the (i) Term Priority Collateral shall be subject only to (A) Permitted Term Liens, and (B) the Carve Out, and (ii) ABL Priority Collateral, shall be subject to (A) all ABL Liens securing the ABL Obligations, (B) the ABL Replacement Lien, (C) the FILO Noteholder Replacement Lien, (D) the ABL Prepetition Liens, (E) the FILO Prepetition Liens, and (F) the Carve Out.

(e) <u>Postpetition Lien Perfection.</u> This Interim Order shall be sufficient and conclusive evidence of the priority, perfection and validity of the DIP Liens and security interests granted herein, effective as of the Petition Date, without any further act and without

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 31 of 92

regard to any other federal, state or local requirements or law requiring notice, filing, registration, recording or possession of the DIP Collateral, or other act to validate or perfect such security interest or lien, including without limitation, control agreements with any financial institution(s) party to a Deposit Account Control Agreement or holding a Blocked Account (as defined in the ABL Credit Agreement) or other depository account consisting of DIP Collateral (a "Perfection Act"). Notwithstanding the foregoing, if either DIP Agent, shall, in its sole discretion, elect for any reason to file, record or otherwise effectuate any Perfection Act, then such DIP Agent is authorized to perform such act, and the Debtors are authorized to perform such act to the extent necessary or required by such DIP Agent, which act or acts shall be deemed to have been accomplished as of the date and time of entry of this Interim Order notwithstanding the date and time actually accomplished, and in such event, the subject filing or recording office is authorized to accept, file or record any document in regard to such act in accordance with applicable law. The DIP Agents may choose to file, record or present a certified copy of this Interim Order in the same manner as a Perfection Act, which shall be tantamount to a Perfection Act, and, in such event, the subject filing or recording office is authorized to accept, file or record such certified copy of this Interim Order in accordance with applicable law. Should either DIP Agent so choose and attempt to file, record or perform a Perfection Act, no defect or failure in connection with such attempt shall in any way limit, waive or alter the validity, enforceability, attachment, or perfection of the postpetition liens and security interests granted herein by virtue of the entry of this Interim Order.

(f) [Reserved.]

(g) Except as expressly set forth herein, the DIP Liens and the DIP Superpriority Claims (i) shall not be made subject to or *pari passu* with (A) any lien, security

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 32 of 92

interest or claim heretofore or hereinafter granted in any of the Chapter 11 Cases or any Successor Cases and shall be valid and enforceable against the Debtors, their estates, any trustee or any other estate representative appointed or elected in the Chapter 11 Cases or any Successor Cases and/or upon the dismissal of any of the Chapter 11 Cases or any Successor Cases, (B) any lien that is avoided and preserved for the benefit of the Debtors and their estates under section 551 of the Bankruptcy Code or otherwise, and (C) any intercompany or affiliate lien or claim; and (ii) shall not be subject to sections 510, 549, 550 or 551 of the Bankruptcy Code. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the DIP Liens and DIP Superpriority Claims shall not be subject to section 506(c) of the Bankruptcy Code.

2.2 Superpriority Administrative Expenses.

(a) <u>ABL DIP Loans</u>. For all Postpetition ABL Obligations now existing or hereafter arising pursuant to this Interim Order, the ABL Loan Documents, or otherwise, the ABL Agent, for the benefit of itself and the other ABL Lenders, is granted an allowed superpriority administrative claim pursuant to section 364(c)(1) of the Bankruptcy Code, having priority in right of payment over any and all other obligations, liabilities and indebtedness of Debtors (other than the Carve Out), whether now in existence or hereafter incurred by the Debtors, and over any and all administrative expenses or priority claims of the kind specified in, or ordered pursuant to, inter alia, sections 105, 326, 328, 330, 331, 364(c)(1), 503(b), 507(a), 507(b), 546(c), 1113 or 1114 of the Bankruptcy Code (other than the Carve Out), whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, which allowed superpriority administrative claim shall be payable from and

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 33 of 92

have recourse to all prepetition and postpetition property of the Debtors and all proceeds thereof (other than the proceeds of avoidance actions under chapter 5 of the Bankruptcy Code) (the "<u>DIP</u> <u>ABL Superpriority Claim</u>"), subject to Section 4.1(a) hereof.

DIP Term Loans. For all DIP Term Obligations now existing or (b) hereafter arising pursuant to this Interim Order, the DIP Term Loan Documents or otherwise, DIP Term Agent, for the benefit of itself and the other DIP Term Lenders, is granted an allowed superpriority administrative claim pursuant to section 364(c)(1) of the Bankruptcy Code, having priority in right of payment over any and all other obligations, liabilities and indebtedness of Debtors (other than the Carve Out), whether now in existence or hereafter incurred by Debtors, and over any and all administrative expenses or priority claims of the kind specified in, or ordered pursuant to, inter alia, sections 105, 326, 328, 330, 331, 364(c)(1), 503(b), 507(a), 507(b), 546(c), 1113 or 1114 of the Bankruptcy Code (other than the Carve Out), whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, which allowed superpriority administrative claim shall be payable from and have recourse to all propertition and postpetition property of the Debtors and all proceeds thereof (other than the proceeds of avoidance actions under chapter 5 of the Bankruptcy Code) (the "DIP Term Superpriority Claim" and together with the DIP ABL Superpriority Claim, the "DIP Superpriority Claims").

(c) Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the DIP Superpriority Claims (1) shall have priority over any claims under 506(c) of the Bankruptcy

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 34 of 92

Code and (2) shall be payable from and have recourse to the proceeds of avoidance actions under chapter 5 of the Bankruptcy Code.

2.3 <u>Carve Out</u>.

As used in this Interim Order, the "Carve Out" means (i) all fees (a) required to be paid to (A) the Clerk of the Court and (B) the Office of the United States Trustee under section 1930(a) of title 28 of the United States Code, plus interest at the statutory rate; plus the sum of (ii) all reasonable fees and expenses up to \$50,000 incurred by a trustee under section 726(b) of the Bankruptcy Code (the "Chapter 7 Trustee Carve Out"); (iii) to the extent allowed at any time, whether by interim order, procedural order, or otherwise, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by persons or firms retained by the Debtors pursuant to section 327, 328, or 363 of the Bankruptcy Code (the "Debtor Professionals"), and the Committee (if appointed) pursuant to section 328 or 1103 of the Bankruptcy Code (the "Committee Professionals" and, together with the Debtor Professionals, the "Professional Persons") at any time before or on the first business day following delivery by the ABL Agent or DIP Term Agent of a Carve Out Trigger Notice (as defined herein), whether allowed prior to or after delivery of a Carve Out Trigger Notice (solely with respect to the ABL Priority Collateral, in an aggregate amount not to exceed the ABL Professional Fee Carve Out Cap (defined below)); and (iv) Allowed Professional Fees of Professional Persons in an aggregate amount not to exceed \$3,000,000 incurred after the first business day following delivery by the ABL Agent or the DIP Term Agent of the Carve Out Trigger Notice, to the extent allowed at any time, whether by interim order, procedural order, or otherwise (the amounts set forth in this clause (iv) being the "Post-Carve Out Trigger Notice Cap").⁴ For purposes of the foregoing, "Carve Out

⁴ Notwithstanding the foregoing, up to \$500,000 of the Post-Carve Out Trigger Notice Cap may be used to pay Allowed Professional Fees of Professional Persons incurred prior to the delivery of a Carve Out Trigger Notice

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 35 of 92

<u>Trigger Notice</u>" shall mean a written notice delivered by email (or other electronic means) by the ABL Agent or DIP Term Agent (at the direction of the Required DIP Lenders under the DIP Term Loan Agreement (the "<u>Required DIP Term Lenders</u>") to the Debtors, their lead restructuring counsel (Kirkland & Ellis LLP), the U.S. Trustee, the ABL Agent or DIP Term Agent (as applicable), and counsel to the Committee (if appointed), which notice may be delivered following the occurrence and during the continuation of a DIP Term Obligations under the ABL DIP Facility and the DIP Term Obligations under the DIP Term Facility, respectively, stating that the Post-Carve Out Trigger Notice Cap has been invoked.

(b) <u>Delivery of Weekly Fee Statements</u>. Not later than 7:00 p.m. (prevailing Eastern Time) on the fourth Business Day of each week starting with the second week following the Petition Date, each Professional Person shall deliver to the Debtors, the ABL Agent, and the DIP Term Agent, a statement setting forth the amount of the fees and expenses, collectively, "<u>Professional Fees</u>") incurred during the preceding week (through Saturday of such week, the "<u>Calculation Date</u>") by such Professional Person, along with a cumulative total and a statement of the amount of such Professional Fees which have been paid to date by the Debtors (each such statement, a "<u>Weekly Statement</u>"); provided that within one business day of the occurrence of the Termination Declaration Date (as defined herein), each Professional Person shall deliver an additional Weekly Statement (the "<u>Final Statement</u>") setting forth the amount and a description of the fees and expenses incurred during the period commencing on the calendar day after the prior Calculation Date and concluding on the Termination Declaration Date. If any Professional Person fails to deliver a Weekly Statement within three (3) calendar

to the extent such Allowed Professional Fees exceed the ABL Professional Fee Carve Out Cap (as defined herein).

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 36 of 92

days after such Weekly Statement is due hereunder and such failure continues unremedied for a period of two (2) days, then a DIP Termination Event shall be deemed to occur hereunder and, in the event that such Professional Person fails to cure any such DIP Termination Event before the funding of any of the Carve Out Reserves (as defined herein), such Professional Person's entitlement to any funds in the Carve Out Reserves with respect to the aggregate unpaid amount of Allowed Professional Fees for the applicable period(s) for which such Professional Person failed to deliver a Weekly Statement covering such period shall be limited to any aggregate unpaid amount of Allowed Professional Fees included in the Budget for such period to the extent allocated in the Approved Budget to such Professional Person; provided that, to the extent not funded by the ABL Agent and the ABL Lenders, such Professional Person shall be entitled to be paid any unpaid amount of Allowed Professional Fees in excess of Allowed Professional Fees included in the Approved Budget for such period from a reserve to be funded by the Debtors from all cash on hand constituting proceeds of the DIP ABL Priority Collateral and Term Priority Collateral as of such date and any available cash constituting proceeds of the DIP ABL Priority Collateral and Term Priority Collateral thereafter pursuant to Section 2.3(c) below. Solely as it relates to the ABL Agent and the ABL Lenders, the Carve Out under Section 2.3(a)(iii) above shall be limited to the greater of (x) the sum of the aggregate unpaid amount of Allowed Professional Fees included in such Weekly Statements timely received by the ABL Agent and the DIP Term Agent prior to the Termination Declaration Date (as defined herein), plus the aggregate unpaid amount of Allowed Professional Fees included in the Final Statement timely received by the ABL Agent and the DIP Term Agent pertaining to the period through and including the Termination Declaration Date (as defined herein), and (y) the aggregate unpaid amount of Allowed Professional Fees included in the Budget for the period

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 37 of 92

prior to the Termination Declaration Date (such amount, the "<u>ABL Professional Fee Carve Out</u> <u>Cap</u>"). For the avoidance of doubt, the ABL Agent shall be entitled to maintain, as part of the Reserves (as defined in the ABL Credit Agreement), a reserve (the "<u>ABL Availability</u> <u>Professional Fee Reserve</u>") in an amount (the "<u>ABL Availability Professional Fee Reserve</u> <u>Amount</u>") equal to the following:

- (i) the Minimum Reserve, plus
- (ii) the Post-Carve Out Trigger Notice Cap, plus
- (iii) (iii) \$2,000,000.

"Minimum Reserve" means the Reserve Floor, provided, that, on a weekly basis, the Debtors shall provide Weekly Statements reflecting (i) the actual aggregate amount of unpaid Professional Fees for such week ("Actual Fee Amount"), and (ii) the difference between the actual amount of unpaid Professional Fees for such week and the budgeted amount of Total Accrued & Unpaid Professional Fees (as reflected on the Weekly Professional Fee Schedule attached hereto as **Exhibit 4** and as may be revised, supplemented or amended from time to time, the "Budgeted Fee Amount"), the difference being the "Variance" for such week, which Variance may be positive or negative. In the event that the Actual Fee Amount plus the Variance exceeds the Reserve Floor in any weekly testing period, then the sum of the Actual Fee Amount plus the Variance shall become the new Minimum Reserve. Following an increase in the Minimum Reserve, in the event that the Actual Fee Amount is less than the Budgeted Fee Amount in any weekly testing period, then the new Minimum Reserve shall be reduced by the amount of the Variance; provided, that, in no event shall the Minimum Reserve be less than the Reserve Floor. The "Reserve Floor" means \$10,100,000; provided that in the event Debtors deliver to DIP Agents a revised Weekly Professional Fee Schedule that sets forth Maximum

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 38 of 92

Weekly Projected Accrued & Unpaid Professional Fees in an amount in excess of \$10,100,000 in any week, then the ABL Agent shall increase the Reserve Floor to reflect such increased Maximum Amount. The ABL Availability Professional Fee Reserve will never be less than the ABL Availability Professional Fee Reserve Amount. Notwithstanding anything to the contrary in this Section, the ABL Agent may determine in its discretion to increase the ABL Availability Professional Fee Reserve Amount to include any additional amounts with respect to the projected amount of any success, completion, commission-based or other non-hourly fees billed by or due to any financial adviser, investment banker or other Professional Person engaged by the Debtors or any Committee in the Cases.

(c) <u>Carve Out Reserves</u>.

(i) On the day on which a Carve Out Trigger Notice is given by either the ABL Agent or the DIP Term Agent (at the direction of the Required DIP Term Lenders) to the Debtors with a copy to counsel to the Committee (if appointed) and the ABL Agent or the DIP Term Agent (as applicable) (the "<u>Termination Declaration Date</u>"), the Carve Out Trigger Notice shall be deemed a draw request and notice of borrowing by the Debtors for Postpetition ABL Obligations under the ABL Agreement in an amount equal to the sum of (x) the amounts set forth in Section 2.3(a)(i) and 2.3(a)(a)(ii), above, and (y) the then unpaid amounts of the Allowed Professional Fees incurred on or before the Termination Declaration Date up to the ABL Professional Fee Carve Out Cap (any such amounts actually advanced shall constitute Postpetition ABL Obligations). To the extent amounts under the preceding sentence are not funded as of the end of the second business day after the date on which the Final Statement is delivered to the ABL Agent and DIP Term Agent, the Carve Out Trigger Notice shall also constitute a demand to the Debtors to utilize all cash on hand constituting proceeds of

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 39 of 92

the DIP Term Collateral and ABL Priority Collateral⁵ as of such date and any available cash constituting proceeds of the DIP Term Collateral and ABL Priority Collateral thereafter held by any Debtor to fund a reserve in an amount equal to the amount required to be funded pursuant to such proceeding sentence (which cash amounts shall reduce, on a dollar for dollar basis, the draw requests and applicable Postpetition ABL Obligations pursuant to the first sentence of this Section 2.3(c) not funded as of the end of the second business day after the date on which the Final Statement is delivered to the ABL Agent and DIP Term Agent). The Debtors shall deposit and hold such amounts in one or more segregated accounts at the ABL Agent or DIP Term Agent (as applicable) in trust⁶ exclusively to pay such unpaid Allowed Professional Fees (each, a "Pre Carve Out Trigger Notice Reserve").

(ii) On the Termination Declaration Date, the Carve Out Trigger Notice shall also be deemed a request by the Debtors for Postpetition ABL Obligations under the ABL Agreement in an amount equal to the Post-Carve Out Trigger Notice Cap (any such amounts actually advanced shall constitute Postpetition ABL Obligations), the satisfaction of which request for funding shall be an obligation of the ABL Agent and the ABL Lenders hereunder. To the extent amounts under this Section 2.3(c) are not funded as of the end of the second business day after the date on which the Final Statement is delivered to the ABL Agent

⁵ To the extent that the amount of the Carve Out exceeds the ABL Fee Availability Professional Fee Reserve Amount, (x) to the extent such cash on hand or available cash constitutes proceeds of the DIP Term Collateral, then the ABL Agent, on behalf of the ABL Lenders, will reimburse the DIP Term Agent, on behalf of the DIP Term Lenders, for fifty-percent (50%) of such funded amount, and (y) to the extent such cash on hand or available cash constitutes proceeds of the ABL Priority Collateral, then the DIP Term Agent, on behalf of the DIP Term Lenders, will reimburse the ABL Agent on behalf of the ABL Lenders for fifty-percent (50%) of such funded amount. Further, in the event there is no cash on hand constituting proceeds of either the DIP Term Priority Collateral or the ABL Priority Collateral at the time of such funding, each of ABL Agent and DIP Term Agent agree that the other shall be reimbursed directly from the first proceeds received from the sale or other disposition of either the Term Priority Collateral or the ABL Priority Collateral, after any such proceeds from such sale or other disposition are used to fund the full amount of the Carve Out.

⁶ Nothing in this Interim Order or otherwise shall create or impose any additional or special trust obligations on the depository bank or institution maintaining the segregated account in which such amounts are deposited.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 40 of 92

and DIP Term Agent, the Carve Out Trigger Notice shall also constitute a demand to the Debtors to utilize all cash on hand constituting proceeds of DIP Term Collateral and ABL Priority Collateral as of such date and any available cash constituting proceeds of DIP Term Collateral and ABL Priority Collateral⁷ thereafter held by any Debtor to fund a reserve in an amount equal to the amount required to be funded pursuant to such clause (ii) (which cash amounts shall reduce, on a dollar for dollar basis, the draw requests and applicable Postpetition ABL Obligations pursuant to the first sentence of this Section 2.3(c) not funded as of the end of the second business day after the date on which the Final Statement is delivered to the ABL Agent and DIP Term Agent). The Debtors shall deposit and hold such amounts in one or more segregated accounts at the ABL Agent or DIP Term Agent (as applicable) in trust exclusively to pay such Allowed Professional Fees benefiting from the Post-Carve Out Trigger Notice Cap (each, a "<u>Post-Carve Out Trigger Notice Reserve</u>" and, together with the Pre-Carve Out Trigger Notice Reserves").

(iii) Not later than the end of the next business day following the Termination Declaration Date and the deemed requests for the making of Postpetition ABL Obligations as provided in this Section 2.3(c), notwithstanding anything in the ABL Agreement or the DIP Term Agreement to the contrary, including with respect to (1) the existence of a Default (as defined in the ABL Agreement or the DIP Term Agreement) or DIP Termination

⁷ To the extent that the amount of the Carve Out exceeds the ABL Fee Availability Professional Fee Reserve Amount, (x) to the extent such cash on hand or available cash constitutes proceeds of the DIP Term Collateral, then the ABL Agent, on behalf of the ABL Lenders, will reimburse the DIP Term Agent, on behalf of the DIP Term Lenders, for fifty-percent (50%) of such funded amount, and (y) to the extent such cash on hand or available cash constitutes proceeds of the ABL Priority Collateral, then the DIP Term Agent, on behalf of the DIP Term Lenders, will reimburse the ABL Agent on behalf of the ABL Priority Collateral, then the DIP Term Agent, on behalf of the DIP Term Lenders, will reimburse the ABL Agent on behalf of the ABL Lenders for fifty-percent (50%) of such funded amount. Further, in the event there is no cash on hand constituting proceeds of either the DIP Term Agent and DIP Term Agent agent agree that the other shall be reimbursed directly from the first proceeds received from the sale or other disposition of either the Term Priority Collateral or the ABL Priority Collateral, after any such proceeds from such sale or other disposition are used to fund the full amount of the Carve Out.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 41 of 92

Event, (2) the failure of the Debtors to satisfy any or all of the conditions precedent for the making of any Postpetition ABL Obligation under the ABL Agreement or DIP Term Obligation under the DIP Term Agreement, respectively, (3) any termination of the Commitments (as defined in the ABL Agreement) following a DIP Termination Event, or (4) the occurrence of a DIP Termination Event, each ABL Lender with an outstanding Commitment shall make available to the ABL Agreement) as a result of such deemed requests for the making of Postpetition ABL Obligations as provided in this Section 2.3(c).

(iv) Nothing contained herein shall require the DIP Term Parties to advance or fund DIP Term Loans in an amount greater than the Interim Financing Amount during the Interim Financing Period or to advance or fund DIP Term Loans in an amount greater than \$100,000,000 (inclusive of the Interim Financing Amount and after taking into account any original issue discount) pursuant to the Final Order.

(d) <u>Application of Carve Out Reserves</u>. All funds in the Pre-Carve Out Trigger Notice Reserve shall be used first to pay the obligations set forth in Sections 2.3(a)(i) through 2.3(a)(iii) of the definition of Carve Out set forth above (the "<u>Pre-Carve Out Amounts</u>"), but not, for the avoidance of doubt, the Post-Carve Out Trigger Notice Cap, until paid in full. If the Pre Carve Out Trigger Notice Reserve has not been reduced to zero, subject to clause (iii), below, all remaining funds in the account funded by the ABL Lenders shall be distributed first to the ABL Agent on account of the Postpetition ABL Obligations until indefeasibly paid in full, in cash, all Commitments have been terminated, and all Letters of Credit have been cancelled (or all such Letters of Credit have been fully cash collateralized or otherwise back-stopped, in each case to the satisfaction of the ABL Agent), and thereafter for

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 42 of 92

permanent application to any Prepetition ABL Obligations in accordance with the ABL Agreement as of the Petition Date until indefeasibly paid in full in cash.

All funds in the Post-Carve Out Trigger Notice Reserve (other than (e) up to \$500,000, which may be used to pay Pre-Carve Out Amounts to the extent they exceed the ABL Professional Fee Carve Out Cap) shall be used first to pay the obligations set forth in clause (iv) of the definition of Carve Out set forth above (the "Post-Carve Out Amounts"). If, after such application, the Post Carve Out Trigger Notice Reserve has not been reduced to zero, subject to clause (iii) below, all remaining funds (x) in respect of the account funded by the ABL Lenders, shall be distributed first to the ABL Agent on account of the Postpetition ABL Obligations until indefeasibly paid in full, in cash, all commitments to extend credit under the ABL Facility have been terminated, and all letters of credit have been cancelled (or all such letters of credit have been fully cash collateralized or otherwise back-stopped, in each case to the satisfaction of the ABL Agent), and thereafter for application to the Prepetition ABL Obligations in accordance with the ABL Agreement as of the Petition Date until indefeasibly paid in full in cash and (y) all remaining funds in the account funded by the DIP Term Lenders shall be distributed to the DIP Term Agent, which shall apply such funds to the DIP Term Obligations in accordance with the DIP Term Agreement until indefeasibly paid in full, in cash.

(f) Notwithstanding anything to the contrary in the Prepetition ABL Loan Documents, the DIP Loan Documents, or this Interim Order, if either of the Carve Out Reserves required to be funded by the ABL Lenders is not funded in full in the amounts set forth in this Section 2.3(f), then any excess funds in one of the Carve Out Reserves held in any account following the payment of the Pre-Carve Out Amounts and Post Carve Out Amounts (subject to the limits contained in the ABL Professional Fee Carve Out Cap and the Post-Carve

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 43 of 92

Out Trigger Notice Cap, respectively), respectively, shall be used to fund the other Carve Out Reserve to the extent of any shortfall in funding by the ABL Lenders prior to making any payments to the ABL Agent or the ABL Lenders, as applicable.

(g) Notwithstanding anything to the contrary in the DIP Loan Documents, or this Interim Order, following the end of the second business day after delivery of a Final Statement is delivered to the ABL Agent and DIP Term Agent, (x) the ABL Agent shall not sweep or foreclose on cash (including cash received as a result of the sale or other disposition of any assets) of the Debtors until the Carve Out Reserves required to be funded by the ABL Lenders have been fully funded, but shall have a security interest in any residual interest in the Carve Out Reserves held in accounts by the ABL Agent, with any excess paid as provided in Sections 2.3(c)(ii) and 2.3(c)(iii) above and (y) the DIP Term Agent shall not sweep or foreclose on cash (including cash received as a result of the sale or other disposition of any assets) of the Debtors until the Carve Out Reserves required to be funded by the DIP Term Lenders have been fully funded, but shall have a security interest in any residual interest have been fully funded, but shall have a security interest in any residual interest in the Carve Out Reserves held in accounts by the DIP Term Agent, with any excess paid as provided in Sections 2.3(c)(ii) and 2.3(c)(iii).

(h) Notwithstanding anything to the contrary in this Interim Order, (i) the failure of the Carve Out Reserves to satisfy in full the Allowed Professional Fees shall not affect the priority of the Carve Out with respect to any shortfall (as described below) (provided that, in all cases, the Carve Out priority with regard to the ABL Priority Collateral will always be subject to the limitations applicable thereto set forth in the definition of Carve Out), and (ii) subject to the limitations with respect to the ABL Agent and the ABL Lenders set forth in Section 2.3(b) above, in no way shall any Approved Budget, Carve Out, Post-Carve Out Trigger

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 44 of 92

Notice Cap, Carve Out Reserves, or Priority Carve Out (as defined in the ABL Agreement) be construed as a cap or limitation on the amount of the Allowed Professional Fees due and payable by the Debtors. For the avoidance of doubt and notwithstanding anything to the contrary herein or in the DIP Facilities, or in any Prepetition Secured Facilities, to the extent of any shortfall in the Carve Out Reserves, the Carve Out shall be senior to all liens and claims securing the DIP Facilities, the Adequate Protection Liens, and the Diminution in Value claims, and any and all other forms of adequate protection, liens, or claims securing the Postpetition ABL Obligations and DIP Term Obligations or the Prepetition ABL Obligations; provided that in all cases, the Carve Out priority with regard to the Prepetition ABL Priority Collateral and the ABL Priority Collateral will always be subject to the limitations applicable thereto set forth in the definition of Carve Out.

(i) <u>No Direct Obligation To Pay Allowed Professional Fees</u>. The DIP Agent and the DIP Lenders shall not be responsible for the payment or reimbursement of any fees or disbursements of any Professional Person incurred in connection with the Cases or any successor cases under any chapter of the Bankruptcy Code. Nothing in this Interim Order or otherwise shall be construed to obligate the ABL Agent, the DIP Term Agent, the ABL Lenders, the DIP Term Lenders, or the Issuing Bank, in any way, to pay compensation to, or to reimburse expenses of, any Professional Person or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement.

(j) <u>Payment of Allowed Professional Fees Prior to the Termination</u>
 <u>Declaration Date</u>. Any payment or reimbursement made prior to the occurrence of the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 45 of 92

Termination Declaration Date in respect of any Allowed Professional Fees shall not reduce the Carve Out.

(k) Payment of Carve Out On or After the Termination Declaration

<u>Date</u>. Any payment or reimbursement made on or after the occurrence of the Termination Declaration Date in respect of any Allowed Professional Fees shall permanently reduce the Carve Out on a dollar-for-dollar basis. Any funding of the Carve Out shall be added to, and made a part of, the DIP Term Obligations and ABL Obligations secured by the DIP Collateral and shall be otherwise entitled to the protections granted under this Interim Order, the DIP Documents, the Bankruptcy Code, and applicable law.

(1) For the avoidance of doubt, nothing herein shall be construed to impair the ability to any party to object to the fees, expenses, reimbursement or compensation described herein.

2.4 <u>Excluded Professional Fees</u>. Notwithstanding anything to the contrary in this Interim Order, neither the (i) DIP Facilities, (ii) DIP Collateral, Prepetition Collateral, DIP Loans, Cash Collateral, Revolving Loans, Letters of Credit, or (iii) any other credit or financial accommodations provided under or in connection with any of the DIP Loan Documents, nor the proceeds of any of the foregoing may be used to pay any Allowed Professional Fees or any other fees or expenses incurred by any Professional Person, directly or indirectly, or by any of the Debtors, the Committee, if any, or any trustee or other estate representative appointed in the Chapter 11 Cases or any Successor Cases or any other person or entity (or to pay any professional fees, disbursements, costs or expenses incurred in connection therewith) in connection with any of the following:

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 46 of 92

(i) an assertion or joinder in any claim, counter-claim, action, proceeding, application, motion, objection, defense or other contested matter seeking any order, judgment, determination or similar relief: (i) challenging the legality, validity, priority, perfection, or enforceability of the (A) Prepetition Secured Obligations or Prepetition Secured Parties' liens on and security interests in the Prepetition Collateral, and (B) the DIP Obligations or any DIP Agent's or DIP Lender's liens on and security interests in the DIP Collateral; (ii) invalidating, setting aside, avoiding or subordinating, in whole or in part, (A) the Prepetition Secured Obligations or Prepetition Secured Parties' liens on and security interests in the Prepetition Collateral, and (B) the DIP Obligations or any DIP Agent's or DIP Lender's liens or any DIP Agent's or DIP Lender's liens on and security interests in the Prepetition Collateral, and (B) the DIP Obligations or any DIP Agent's or DIP Lender's liens on and security interests in the DIP Collateral; or (iii) preventing, hindering or delaying either DIP Agent's or DIP Lenders' enforcement of any lien, claim, right or security interest or realization upon any DIP Collateral in accordance with the terms and conditions of either DIP Loan Agreement or any DIP Loan Documents, respectively, and this Interim Order;

(ii) to object to, contest, prevent, hinder, delay or interfere with, in any way, the DIP Agents' or the DIP Lenders' enforcement or realization upon any of the DIP Collateral once a DIP Termination Event occurs (other than with respect to rights otherwise expressly granted herein with respect to the Enforcement Notice Period); (2) to use or seek to use Cash Collateral; (3) to sell or otherwise dispose of the DIP Collateral (or any portion thereof) other than in accordance with the terms of this Interim Order; (4) except to the extent expressly permitted by the DIP Loan Documents, to use or seek to use any insurance proceeds constituting DIP Collateral without the consent of the DIP Agents and the applicable Required DIP Lenders; (5) except to the extent expressly permitted by the DIP Loan Documents, to seek authorization to incur Indebtedness (as defined in the DIP Loan Agreements) or to incur liens or

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 47 of 92

security interests; (6) to object to or challenge in any way the DIP Liens, DIP Obligations, Prepetition Liens, Prepetition Secured Obligations, DIP Collateral (including Cash Collateral or Prepetition Collateral or any other claims or liens held by or on behalf of any of the DIP Agents, the DIP Lenders or the Prepetition Secured Parties, respectively, (7) to investigate (including by way of examinations or discovery proceedings, whether formal or informal), prepare, assert, join, commence, support or prosecute any action for any claim, counter-claim, action, proceeding, application, motion, objection, defense, or other contested matter seeking any order, judgment, determination or similar relief against, or adverse to the interests of, in any capacity, against any of the Released Parties with respect to any transaction, occurrence, omission, action or other matter arising under, in connection with or related to this Interim Order, the DIP Facility, the DIP Loan Documents, the Prepetition Loan Documents or the transactions contemplated therein or thereby, including, without limitation, (A) any claims or causes of action arising under chapter 5 of the Bankruptcy Code, (B) any so-called "lender liability" claims and causes of action, (C) any claim or cause of action with respect to the validity, enforceability, priority and extent of, or asserting any defense, counterclaim, or offset to, the DIP Obligations, the DIP Superpriority Claims, the DIP Liens, the DIP Loan Documents, the Prepetition Loan Documents or the Prepetition Secured Obligations, (D) any claim or cause of action seeking to challenge, invalidate, modify, set aside, avoid, marshal, subordinate, disallow, or recharacterize in whole or in part, the DIP Obligations, the DIP Liens, the DIP Superpriority Claim, the DIP Collateral, the Prepetition Collateral, the Prepetition Secured Obligations, (E) any action seeking to modify any of the rights, remedies, priorities, privileges, protections and benefits granted to either the DIP Agents or the DIP Lenders hereunder or under any of the DIP Loan Documents, the Prepetition Secured Parties under any of the Prepetition Loan Documents (in each case,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 48 of 92

including, without limitation, claims, proceedings or actions that might prevent, hinder or delay any of the DIP Agent's or the DIP Lenders' assertions, enforcements, realizations or remedies on or against the DIP Collateral in accordance with the applicable DIP Loan Documents and this Order and/or the Final Order (as applicable) (provided, however, that an amount no more than, \$50,000 in the aggregate, to be incurred solely by the Committee (the "Investigation Budget <u>Amount</u>") of the DIP Collateral, the Carve Out, or proceeds from the borrowings under the DIP Facilities or any other amounts, may be used for allowed fees and expenses in investigating (but not objecting to, challenging, litigating (including by way of discovery), opposing, or seeking to subordinate or recharacterize) the validity, enforceability, perfection, priority or extent of the Prepetition Liens, the Prepetition Loan Documents, or the Prepetition Secured Obligations, within the applicable challenge period as set forth in Section 4.1.

(iii) any act which has or could directly, materially, and adversely modify or compromise the rights and remedies of any DIP Agent or any DIP Lender under this Interim Order, or which directly results in the occurrence of a DIP Termination Event under any DIP Loan Documents, or this Interim Order.

2.5 <u>Use of Cash Collateral; Adequate Protection.</u>

(a) <u>Authorization to Use Cash Collateral</u>. Subject to the terms and conditions of this Interim Order and the DIP Loan Documents, and in accordance with the Approved Budget, the Debtors shall be and are hereby authorized to use the Cash Collateral (as defined in section 363 of the Bankruptcy Code) for the period commencing on the date of this Interim Order and terminating upon the occurrence of a DIP Termination Event. Nothing in this Interim Order shall authorize the disposition of any assets of the Debtors or their Estates outside the ordinary course of business, or any Debtor's use of Cash Collateral or other proceeds

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 49 of 92

resulting therefrom, except as expressly permitted in this Interim Order, the DIP Loan Documents and in accordance with the Approved Budget. Any failure by the Debtors on or after the Petition Date to comply with the segregation requirements of section 363(c)(4) of the Bankruptcy Code in respect of any Cash Collateral shall not be used as a basis to challenge any of the Prepetition Secured Obligations, or the extent, validity, enforceability or perfected status of any of the Prepetition Liens.

(b) ABL Replacement Liens. As adequate protection for the diminution in value of their interests in the Prepetition ABL Collateral (including Cash Collateral) on account of the Debtors' sale, use, or lease of such Collateral (including Cash Collateral), the imposition of the automatic stay, the subordination to the Carve Out and the priming of their interests in the Term Loan Priority Collateral (collectively, the "ABL Diminution in Value"), the ABL Agent, for the benefit of itself and the ABL Lenders, is hereby granted pursuant to sections 361 and 363 of the Bankruptcy Code, and solely to the extent of the ABL Diminution in Value, valid, binding, enforceable and perfected replacement liens upon and security interests in all DIP Collateral (the "ABL Replacement Lien"). The ABL Replacement Lien shall be junior and subordinate only to (i) the Carve Out, (ii) the Permitted ABL Liens, (iii) ABL Agent's and ABL Lenders' liens on the DIP Collateral to secure the ABL Obligations, (iv) the DIP Term Agent's and DIP Term Lenders DIP Lien on the Term Priority Collateral, and (v) the First Lien Notes Parties' liens upon and security interests in the Term Priority Collateral; provided, however, that the ABL Replacement Lien and the First Lien Notes Adequate Protection Lien shall share in any proceeds of DIP Collateral that is neither Term Priority Collateral nor ABL Priority Collateral on a pro rata basis.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 50 of 92

(c) <u>ABL 507(b) Priority Claims</u>. As additional adequate protection for the ABL Diminution in Value, the ABL Agent, for the benefit of itself and the ABL Lenders, is hereby granted as and to the extent provided by section 507(b) of the Bankruptcy Code, and solely to the extent of the ABL Diminution in Value, an allowed superpriority administrative expense claim in each of the Cases and any successor bankruptcy cases (the "<u>ABL Adequate</u> <u>Protection Superpriority Claim</u>"). The ABL Adequate Protection Superpriority Claim shall be *pari passu* with the First Lien Adequate Protection Claim, and junior only to (i) the Carve Out, and (ii) the DIP ABL Superpriority Claim, and (iii) the DIP Term Superpriority Claim, and shall otherwise have priority over all administrative expense claims and unsecured claims against Debtors and their Estates now existing or hereafter arising, of any kind or nature whatsoever.

(d) <u>First Lien Adequate Protection</u>. The First Lien Notes Trustee, for the benefit of itself and the First Lien Noteholders (collectively, the "<u>First Lien Notes Adequate</u> <u>Protection Parties</u>"), shall be granted the following protection, pursuant to sections 361, 507, and 363(e) of the Bankruptcy Code for the diminution in value of their interests in the First Lien Notes Collateral (including Cash Collateral) on account of the Debtors' sale, use, or lease of such DIP Collateral (including Cash Collateral), the imposition of the automatic stay, the subordination to the Carve Out, and the priming of the First Lien Notes Trustee's and First Lien Noteholders' liens and security interests in the First Lien Noteholder Collateral (collectively, the "<u>First Lien Notes Diminution in Value</u>").

(i) <u>First Lien Adequate Protection Lien</u>. The First Lien Notes Trustee, on behalf of the First Lien Notes Adequate Protection Parties, shall be granted for their benefit, effective and perfected as of the date of entry of this Interim Order and without the necessity of the execution or filing of mortgages, security agreements, pledge agreements,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 51 of 92

financing statements or other agreements, and solely to the extent of the First Lien Notes Diminution in Value, a security interest in and lien on the DIP Collateral (the "<u>First Lien Notes</u> <u>Adequate Protection Liens</u>"), subject and subordinate only to (A) the Carve Out, (B) Permitted Liens, and (C) the liens of the DIP Agents and DIP Lenders securing DIP Obligations arising under or in connection with the DIP Loan Agreements and the other DIP Loan Documents and/or this Interim Order in and upon the DIP Collateral; <u>provided</u> that, the First Lien Notes Adequate Protection Liens shall be (1) senior to the (a) ABL Liens and (b) the ABL Replacement Lien, in each case solely with respect to the Term Priority Collateral, and (2) junior to the liens and security interests of the (x) the ABL Liens on the DIP Collateral, (y) the ABL Replacement Lien, and (z) the FILO Noteholder Replacement Lien with respect to the ABL Priority Collateral; <u>provided</u>, <u>however</u>, that the ABL Replacement Lien and the First Lien Notes Adequate Protection Lien shall share in any proceeds of DIP Collateral that is neither Term Priority Collateral nor ABL Priority Collateral on a pro rata basis.

(ii) <u>First Lien Noteholder 507(b) Claim</u>. As adequate protection for the First Lien Notes Diminution in Value, First Lien Notes Trustee, on behalf of the First Lien Notes Adequate Protection Parties, is hereby granted as and to the extent provided by Section 507(b) of the Bankruptcy Code, and solely to the extent of the First Lien Notes Diminution in Value, an allowed super-priority administrative expense claim in each of the Cases and any successor bankruptcy cases (the "<u>First Lien Noteholder Adequate Protection</u> <u>Superpriority Claim</u>"). The First Lien Adequate Protection Superpriority Claim shall be *pari passu* with the ABL Adequate Protection Superpriority Claim, and shall be junior only to (i) the Carve Out, and (ii) the DIP ABL Superpriority Claim, and (iii) the DIP Term Superpriority Claim, and shall otherwise have priority over all administrative expense claims and unsecured

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 52 of 92

claims against Debtors and their Estates now existing or hereafter arising, of any kind or nature whatsoever.

(iii) In exchange for the provision of adequate protection, and immediately upon entry of this Interim Order, but in no event, later than five (5) calendar days, the First Lien Notes Trustee shall deliver the "Pledged Equity and the Pledge Debt" (as defined in the certain Pledge and Security Agreement between the Borrower and the DIP Term Parties) to the DIP Term Agent.

(e) <u>Second Lien Adequate Protection</u>. The Second Lien Notes Trustee, for the benefit of itself and the Second Lien Noteholders (collectively, the "<u>Second Lien</u> <u>Adequate Protection Parties</u>"), shall be granted the following protection, pursuant to sections 361, 507, and 363(e) of the Bankruptcy Code for the diminution in value of their interests in the DIP Collateral (including Cash Collateral) on account of the Debtors' sale, use, or lease of such DIP Collateral (including Cash Collateral), the imposition of the automatic stay, the subordination to the Carve Out, and the priming of their interest in the Prepetition Term Loan Collateral (collectively, the "<u>Second Lien Diminution in Value</u>"):

(i) <u>Second Lien Adequate Protection Lien</u>. The Second Lien Notes Trustee, on behalf of the Second Lien Adequate Protection Parties, shall be granted for their benefit, effective and perfected as of the date of entry of this Interim Order and without the necessity of the execution or filing of mortgages, security agreements, pledge agreements, financing statements or other agreements, and solely to the extent of the Second Lien Diminution in Value, a security interest in and lien on the DIP Collateral (the "<u>Second Lien Adequate Protection Liens</u>"), subject and subordinate only to (A) the Carve Out, (B) Permitted Liens, (C) the ABL Liens on the DIP Collateral, (D) the DIP Term Liens, (E) ABL Replacement Lien,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 53 of 92

(F) the First Lien Adequate Protection Lien, (G) the FILO Adequate Protection Lien (as defined below) and (H) the Prepetition First Lien Notes Liens.

(ii) Second Lien Noteholder 507(b) Claim. As adequate protection for the Second Lien Diminution in Value, Second Lien Notes Trustee, on behalf of the Second Lien Adequate Protection Parties, is hereby granted as and to the extent provided by Section 507(b) of the Bankruptcy Code, and solely to the extent of the Second Lien Diminution in Value, a super-priority administrative expense claim in each of the Cases and any successor cases (the "Second Lien Noteholder Adequate Protection Claim" together with the ABL Adequate Protection Superpriority Claim, the First Lien Adequate Protection Superpriority Claim, and the FILO Noteholder Adequate Protection Superpriority Claim, the "Adequate Protection Claims"). The Second Lien Noteholder Adequate Protection Claims shall be junior to (i) the Carve Out, and (ii) the DIP ABL Superpriority Claim, and (iii) the DIP Term Superpriority Claim, (iv) the ABL Adequate Protection Superpriority Claim; (v) the First Lien Adequate Protection Claim (vi) the FILO Adequate Protection Claims; and (vii) the Prepetition First Lien Notes Claims, and shall otherwise have priority over all administrative expense claims and unsecured claims against Debtors and their Estates now existing or hereafter arising, of any kind or nature whatsoever); provided that the Second Lien Notes Trustee and the Second Lien Noteholders shall not receive or retain any payments, property or other amounts in respect of such superpriority claims unless and until all DIP Obligations under the DIP Loan Agreements and the other DIP Loan Documents have indefeasibly been paid in cash in full in accordance with the terms of the DIP Loan Agreements and subject to the terms of the Prepetition Intercreditor Agreements.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 54 of 92

(f) <u>FILO Notes Adequate Protection</u>. The FILO Notes Trustee, for the benefit of itself and the FILO Noteholders (collectively, the "<u>FILO Notes Adequate Protection</u> <u>Parties</u>"), shall be granted the following protection, pursuant to sections 361, 507, and 363(e) of the Bankruptcy Code for the diminution in value of their interests in the DIP Collateral (including Cash Collateral) on account of the Debtors' sale, use, or lease of such DIP Collateral (including Cash Collateral), the imposition of the automatic stay, the subordination to the Carve Out, and the priming of their interest in the Prepetition Term Loan Priority Collateral (collectively, the "<u>FILO Notes Diminution in Value</u>"):

(i) <u>FILO Notes Adequate Protection Lien</u>. FILO Notes Trustee, on behalf of the FILO Notes Adequate Protection Parties, shall be granted for their benefit, effective and perfected as of the date of entry of this Interim Order and without the necessity of the execution or filing of mortgages, security agreements, pledge agreements, financing statements or other agreements, and solely to the extent of the FILO Notes Diminution in Value, a security interest in and lien on the DIP Collateral (the "<u>Secured Note Adequate</u> <u>Protection Lien</u>"), subject and subordinate only to (A) the Carve Out, (B) Permitted Liens, and (C) the ABL Liens on the DIP Collateral, (D) the DIP Term Liens, (E) the ABL Replacement Lien and (F) the First Lien Replacement Lien with respect to Term Priority Collateral.

(ii) <u>FILO Notes 507(b) Claim</u>. As adequate protection for the FILO Noteholder Diminution in Value, FILO Notes Trustee, on behalf of the FILO Notes Adequate Protection Parties, is hereby granted and as to the extent provided by Section 507(b) of the Bankruptcy Code, and solely to the extent of the FILO Notes Diminution in Value, a super-priority administrative expense claim in each of the Cases and any successor bankruptcy cases. The FILO Adequate Protection Superpriority Claim shall be *pari passu* with the First

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 55 of 92

Lien Adequate Protection Claim, and shall be junior only to (i) the Carve Out, and (ii) the DIP ABL Superpriority Claim, (iii) the DIP Term Superpriority Claim, (iv) the ABL Adequate Protection Superpriority Claims, and shall otherwise have priority over all administrative expense claims and unsecured claims against Debtors and their Estates now existing or hereafter arising, of any kind or nature whatsoever.

(g) Additional Adequate Protection. As further adequate protection for the Prepetition First Lien Noteholders and the Prepetition FILO Secured Parties: (i) until all of the Prepetition FILO Notes Obligations have been satisfied in full, the Debtors shall (A) pay to the Prepetition FILO Notes Trustee, for the benefit of the Prepetition FILO Noteholders, cash interest, as and when due, at the non-default rate set forth in the Prepetition FILO Notes Agreement, and (B) pay all prepetition and postpetition reasonable and documented fees, costs and expenses incurred by the Prepetition FILO Notes Trustee (including all reasonable and documented fees and out-of-pocket expenses of one legal counsel, the "FILO Notes Professional"), solely in connection with any reasonable and documented fees and out-of-pocket expenses incurred in its capacity as the trustee for Prepetition FILO Notes, and not any other fees and expenses, including any fees or expenses on account of the 6.00% Senior Unsecured Notes due 2024, or any other claims), and (ii) until all of the Prepetition First Lien Notes Obligations have been satisfied in full, the Debtors shall pay all prepetition and postpetition reasonable and documented fees, costs and expenses incurred by the Prepetition First Lien Notes Trustee (including all reasonable and documented fees, costs and expenses of one counsel) and the members of the ad hoc committee of certain Prepetition First Lien Noteholders (the "Ad Hoc Committee") and the reasonable and documented fees, costs and expenses of their respective legal and financial advisors (including (A) Stroock, (B) Ducera, and (C) such other consultants

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 56 of 92

or other professionals as may be retained by the Ad Hoc Committee with the consent of the Debtors (such consent not to be unreasonably withheld, delayed or conditioned) (together with the FILO Notes Professional, the "Lender Professionals"), in each case, without duplication of the DIP Lenders' advisor fees as provided in paragraph 1.3(c) of this Interim Order. The invoices for the fees and expenses of the Lender Professionals to be paid pursuant to this paragraph 2.5(g) shall not be required to comply with the U.S. Trustee guidelines and shall be provided to the Fee Notice Parties (defined below) The invoices to be provided to the Fee Notice Parties shall contain reasonable detail (and, in the case of the FILO Notes Professional, shall be accompanied by a certification of counsel that all such fees and expenses have been incurred solely in connection with secured claims of the Prepetition FILO Notes Trustee and the FILO Noteholders against the Debtors, and not any other fees and expenses), may be in summary form and which may be redacted or modified to the extent necessary to delete any information subject to attorney-client privilege, any information constituting attorney work product or any other confidential information, and the provision of such invoices shall not constitute any waiver of any such privileges or doctrines. If no objection to payment is made within ten (10) calendar days after delivery of such invoices (the "Fee Objection Period"), then, without further order of, or application to, the Court or notice to any other party, such fees and expenses shall be promptly paid by the Debtors. If an objection (in the case of all Lender Professionals, solely as to reasonableness, and in the case of the FILO Notes Professionals, as to whether such fees and expenses have been incurred in compliance with this Interim Order) is made by any of the Fee Notice Parties within the Fee Objection Period to payment of the requested fees and expenses, then only the disputed portion of such fees and expenses shall not be paid until the objection is resolved by the applicable parties in good faith or by order of the Court, and the undisputed

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 57 of 92

portion shall be promptly paid by the Debtors; provided that Debtors shall not pay fees or expenses pursuant to this Section 2.5(g) during the Enforcement Notice Period or at any time thereafter without further order of this Court.

(h) <u>Adequate Protection Reservation</u>. The receipt by the Prepetition Secured Parties of the adequate protection provided pursuant to paragraph 2.5 of this Interim Order shall not be deemed an admission that the interests of the Prepetition Secured Parties are indeed adequately protected. Further, nothing contained in this Interim Order shall prejudice or limit the rights of the Prepetition Secured Parties, subject to the Prepetition Intercreditor Agreements, to seek additional relief with respect to the use of Cash Collateral or for additional adequate protection. Without limiting the foregoing, nothing herein shall impair or modify the application of section 507(b) of the Bankruptcy Code in the event that the adequate protection provided hereunder is insufficient to compensate for any First Lien Notes Diminution in Value of the First Lien Notes Collateral or ABL Diminution in Value of the ABL Priority Collateral during the Chapter 11 Cases.

Section 3. Default; Rights and Remedies; Relief from Stay.

3.1 <u>Events of Default</u>. Each of the following occurrences or events shall constitute a termination event under this Interim Order (each, a "<u>DIP Termination Event</u>") unless waived in writing by the applicable DIP Agents: (a) the occurrence of an "Event of Default" under and as defined in the ABL Credit Agreement or the DIP Term Loan Agreement, as applicable; (b) the occurrence of the scheduled maturity date of either DIP Facility; (c) 40 days from the Petition Date, if the Final Order, in form and substance satisfactory to the DIP Agents and the applicable DIP Agents has not been entered by such date; (d) the date of acceleration of any of the DIP Loans; (e) the Debtors seek any amendment, modification or extension of this Interim Order without the prior written consent of the DIP Agents (and no such consent shall be

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 58 of 92

implied by any other action, inaction or acquiescence of the DIP Secured Parties); (f) the failure by the Debtors to timely perform any of the terms, provisions, conditions, covenants, or other obligations under this Interim Order; or (g) the filing of any action, claim, defense, complaint, motion or other pleading by the Debtors that would affect the validity or amount of the Prepetition ABL Obligations and the First Lien Notes Obligations, as of the Petition Date (including as to the value of the secured claim of the ABL Agent and ABL Lenders solely with regards to the Prepetition ABL Obligations, subject to the rights of the Committee and other parties in interest set forth in Section 4.1(a) hereof and the value of the secured claim of the First Lien Indenture Trustee and First Lien Noteholders solely with regards to the First Lien Notes Obligations, subject to the rights of the Committee and other parties in interest set forth in Section 4.1(b) hereof).

3.2 <u>Rights and Remedies upon a DIP Termination Event.</u>

(a) Immediately upon the occurrence of and during the continuance of a DIP Termination Event, the ABL Agent and/or the DIP Term Agent (at the direction of the Required DIP Term Lenders), as applicable, shall be entitled to take any act or exercise any right or remedy (subject to Section 3.4 below) as provided in this Interim Order or any DIP Loan Document, as applicable, including, without limitation, (a) declare all DIP Obligations owing under their respective DIP Loan Documents to be immediately due and payable; (b) terminate, reduce or restrict any commitment to extend additional credit to the Debtors to the extent any such commitment remains (including provision of any Letters of Credit); (c) terminate the DIP Facility and any DIP Loan Document as to any future liability or obligation of the DIP Secured Parties, but without affecting any of the DIP Obligations or the DIP Liens securing the DIP Obligations; (d) terminate and/or revoke the Debtors' right, if any, under this Interim Order and

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 59 of 92

the other DIP Loan Documents to use any Cash Collateral and all authority to use Cash Collateral shall cease; (e) invoke the right to charge interest at the default rate under the DIP Loan Documents; (f) freeze monies or balances in the Debtors' accounts; (g) immediately setoff any and all amounts in accounts maintained by the Debtors with the applicable DIP Agent or the DIP Lenders against the DIP Obligations; (h) otherwise enforce any and all rights against the DIP Collateral in the possession of any of the applicable DIP Lenders, including, without limitation, disposition of the DIP Collateral solely for application towards the DIP Obligations; and (i) take any other actions or exercise any other rights or remedies permitted under this Order, the DIP Loan Documents or applicable law; provided, however, that prior to the exercise of any right in clauses (d) and (f) though (i) of this paragraph, the applicable DIP Agent or DIP Agents shall be required to provide five (5) days' written notice to counsel to the Debtors, counsel to the Committee and the U.S. Trustee of its intent to exercise its rights and remedies (the "Enforcement Notice Period") other than with respect to funding of the Carve Out. The DIP Agents and DIP Lenders shall have no obligation to lend or advance any additional funds to or on behalf of Debtors, or provide any other financial accommodations to Debtors, immediately upon or after the occurrence of a DIP Termination Event, or upon the occurrence of any act, event, or condition that, with the giving of notice or the passage of time, or both, would constitute a DIP Termination Event.

(b) Unless during such Enforcement Notice Period the Court determines that a Termination Event has not occurred, the DIP Agents and the DIP Lenders shall be deemed to have received relief from the automatic stay, and may foreclose on all or any portion of the DIP Collateral, collect accounts receivable, and apply the proceeds thereof to the DIP Obligations, occupy the Debtors' premises to sell or otherwise dispose of the DIP Collateral,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 60 of 92

or otherwise exercise all rights and remedies available against the DIP Collateral permitted by the DIP Loan Documents in accordance with applicable law or equity, without further notice to, hearing of, or order from this Court, and without restriction or restraint by any stay under sections 105 or 362 of the Bankruptcy Code, or otherwise. During the Enforcement Notice Period, the Debtors may use Cash Collateral only the following amounts and expenses: (i) the Carve Out Reserves, as set forth in paragraph 2.3; and (ii) subject to the Approved Budget, amounts that the Debtors have determined in good faith are, in the ordinary course, critical to the preservation of the Debtors and their estates and have approved in advance in writing by the applicable Required DIP Lenders. The Debtors shall cooperate with the DIP Agents and the DIP Lenders in their exercise of rights and remedies, whether against the DIP Collateral or otherwise, and the Debtors shall waive any right to seek relief under the Bankruptcy Code, including under section 105 thereof, to the extent such relief would restrict or impair the rights and remedies of the DIP Agents and the DIP Lenders set forth in this Interim Order and in the DIP Loan Documents, other than to dispute whether a Termination Event has in fact occurred; provided, that none of the DIP Secured Parties shall object to a request by the Debtors for an expedited hearing before the Court to contest whether a DIP Termination Event has in fact occurred.

3.3 <u>Modification of Automatic Stay</u>.

(a) The automatic stay provisions of section 362 of the Bankruptcy Code and any other restriction imposed by an order of the Court or applicable law are hereby modified without further notice, application or order of the Court to the extent necessary to permit the DIP Agents to perform any act authorized or permitted under or by virtue of this Interim Order, the DIP Loan Agreements or the other DIP Loan Documents, as applicable, including, without limitation, (i) to implement the postpetition financing arrangements

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 61 of 92

authorized by this Interim Order, (ii) to take any act to create, validate, evidence, attach or perfect any lien, security interest, right or claim in the DIP Collateral, (iii) to assess, charge, collect, advance, deduct and receive payments with respect to the ABL Obligations and DIP Term Obligations (or any portion thereof), including, without limitation, all interests, fees, costs and expenses permitted under any of the DIP Loan Documents and apply such payments to the Prepetition ABL Obligations, and (iv) immediately following the expiration of the Enforcement Notice Period (as referred to above), unless stayed, to take any action and exercise all rights and remedies provided to it by this Interim Order, the DIP Loan Documents or applicable law.

3.4 <u>DIP Agent Intercreditor Provisions</u>.

(a) With respect to the DIP Collateral, each of the ABL Agent, on behalf of the ABL Lenders, and DIP Term Agent, on behalf of the DIP Term Lenders, shall be bound by this Section 3.5.

(b) Lien Priorities. The ABL Agent shall, for the benefit of itself and the ABL Lenders, have a first priority priming postpetition lien and security interest in the ABL Priority Collateral, which secures the ABL Obligations, and DIP Term Agent shall, for the benefit of itself and the Term DIP Lenders, have a first priority priming postpetition lien and security interest in the Term Priority Collateral, which secures the DIP Term Obligations. For the avoidance of doubt, the DIP Agents and DIP Lenders shall have liens on and security interests in all DIP Collateral. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the proceeds of any avoidance actions brought pursuant to Chapter 5 of the Bankruptcy Code or applicable state law equivalents shall be shared DIP Collateral, and the ABL Liens on the DIP

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 62 of 92

Collateral and the DIP Term Liens on such proceeds shall rank equally in priority and share in such proceeds on a pro rata basis (based on the outstanding principal amount of the respective DIP Facilities at the time of the distribution of such proceeds to the respective DIP Agents for the benefit of the respective DIP Lenders).

(c) <u>Enforcement</u>.

(i) Until the ABL Obligations (including the Prepetition ABL Obligations) have been indefeasibly paid and satisfied in full, (A) the ABL Agent and the ABL Lenders shall have the sole and exclusive right to enforce rights, exercise remedies, and make determinations regarding the sale, release, or other disposition of any of the ABL Priority Collateral; and (B) DIP Term Agent and DIP Term Lenders (x) will not exercise any right or remedies with respect to any ABL Priority Collateral (including any right of setoff or any right under any lockbox or control agreement with respect to Deposit Accounts or Securities Accounts) or seek relief from the automatic stay in order to commence any action with respect to such rights and remedies, and (y) will not contest, protest or object to or otherwise interfere with any action or proceeding brought by the ABL Agent or any DIP ABL Lender to enforce its rights and remedies relating to the ABL Priority Collateral.

(ii) Until the DIP Term Loan Obligations have been indefeasibly paid and satisfied in full, (A) the DIP Term Agent and the DIP Term Lenders shall have the sole and exclusive right to enforce rights, exercise remedies, and make determinations regarding the sale, release, or other disposition of any of the Term Priority Collateral; and (B) ABL Agent and ABL Lenders (x) will not exercise any rights or remedies with respect to the Term Loan Priority Collateral or seek relief from the automatic stay in order to commence any action with respect to such rights and remedies, and (y) will not contest, protest or object to or

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 63 of 92

otherwise interfere with any action or proceeding brought by DIP Term Agent or any Term Lender to enforce its rights and remedies relating to the Term Loan Priority Collateral.

(d) Access Rights. If the DIP Term Agent, or any agent or representative thereof, or any trustee shall lift the automatic stay with respect to the Term Loan Priority Collateral and is exercising remedies with respect thereto, DIP Term Agent shall provide ABL Agent with a non-exclusive, rent free right to access and use such Term Priority Collateral, including, without limitation, (i) any of the Mortgaged Premises, for a period of up to 180 days from the date the automatic stay is lifted, to enable the ABL Agent to use, take physical possession of, or otherwise control the use or access to the ABL Priority Collateral, for the purpose of arranging for and effecting the sale or other disposition of the ABL Priority Collateral, including, without limitation, the production, compiling, packaging and other preparation of the ABL Priority Collateral for sale or disposition; (ii) a royalty free, rent free, non-exclusive license and leave to use all Term Priority Collateral, consisting of all Intellectual Property to complete the sale of inventory, and (iii) in the event of a sale or other disposition of any Term Priority Collateral during the access period, such purchaser shall agree to provide ABL Agent for continued access to the purchased Term Priority Collateral.

(e) For all other purposes, each of ABL Agent, ABL Lenders, DIP Term Agent and DIP Term Lenders shall be bound by the terms and conditions of the First Lien Intercreditor Agreement.

Section 4. <u>Representations; Covenants; and Waivers.</u>

4.1 <u>Reservation of Third Party Challenge Rights</u>.

(a) <u>Objections to Prepetition ABL Obligations</u>. Notwithstanding anything to the contrary in the Interim Order (including the Reaffirmation), any action, claim, defense, complaint, motion or other written opposition (hereinafter, an "<u>Prepetition ABL</u>

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 64 of 92

Objection") that seeks to object to, challenge, contest or otherwise invalidate or reduce, whether by setoff, recoupment, counterclaim, deduction, disgorgement or claim of any kind regarding: (i) the existence, validity or amount of the Prepetition ABL Obligations, as of the Petition Date, (with the exception of the value of a secured claim of the ABL Agent and ABL Lenders solely with respect to Prepetition ABL Obligations, pursuant to section 506(a) of the Bankruptcy Code) or (ii) the extent, legality, validity, perfection or enforceability of the ABL Agent's and ABL Lenders' respective prepetition liens and security interests in the Prepetition ABL Collateral solely with respect to Prepetition ABL Obligations, shall be properly filed with the Court (x) by the Committee within seventy-five (75) calendar days from the date of entry of the Final Order, (y) if no Committee is appointed, by any party in interest with requisite standing within ninety (90) calendar days from the date of entry of the Final Order; provided, however, that nothing herein shall permit any party to challenge the extent or validity of the Postpetition ABL Obligations or any liens that secure such Postpetition ABL Obligations except as a "Roll-Up." If any such Prepetition ABL Objection is timely and properly filed and a final, nonappealable order is entered by a court of competent jurisdiction sustaining and ordering some or all of the relief requested in such Prepetition ABL Objection, nothing in this Interim Order shall prevent the Court from granting appropriate relief with respect to the Prepetition ABL Obligations or the ABL Agent's or the ABL Lenders' liens on the Prepetition ABL Collateral. If no Prepetition ABL Objection is timely and properly filed, or if a Prepetition ABL Objection is timely and properly filed but denied, (i) the Prepetition ABL Obligations shall be deemed allowed in full, shall not be subject to any setoff, recoupment, counterclaim, deduction or claim of any kind, and shall not be subject to any further objection or challenge by any party at any time, and the ABL Agent's and the ABL Lenders' prepetition liens on and security interest in the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 65 of 92

Prepetition Collateral shall be deemed legal, valid, perfected, enforceable, and non-avoidable for all purposes, and (ii) the ABL Agent and the ABL Lenders, and each of their respective participants, agents, officers, directors, employees, attorneys, professionals, successors, and assigns (each in their respective capacities as such) shall be deemed released and discharged from any and all claims and causes of action related to or arising out of the Prepetition ABL Loan Documents and the Prepetition ABL Obligations, and shall not be subject to any further objection or challenge relating thereto or arising therefrom by any party at any time. Nothing contained in this Section 4.1(a) or otherwise shall or shall be deemed or construed to impair, prejudice or waive any rights, claims or protections afforded to ABL Agent or any ABL Lenders in connection with all postpetition Revolving Loans and Letters of Credit, and any other postpetition financial and credit accommodations provided by ABL Agent and ABL Lenders to Debtors in reliance on section 364(e) of the Bankruptcy Code and in accordance with the terms and provisions of this Interim Order and the Prepetition ABL Loan Documents.

(b) <u>Objections to Prepetition First Lien Notes Obligations</u>. Notwithstanding anything to the contrary in the Interim Order, any action, claim, defense, complaint, motion or other written opposition (hereinafter, a "<u>First Lien Objection</u>") that seeks to object to, challenge, contest or otherwise invalidate or reduce, whether by setoff, recoupment, counterclaim, deduction, disgorgement or claim of any kind regarding: (i) the existence, validity or amount of the First Lien Notes Obligations, as of the Petition Date (with the exception of as to the value of a secured claim of the First Lien Notes Trustee and First Lien Noteholders solely with respect to First Lien Notes Obligations, pursuant to section 506(a) of the Bankruptcy Code), or (ii) the extent, legality, validity, perfection or enforceability of the First Lien Notes Trustee's and First Lien Noteholders' prepetition liens and security interests in the Prepetition Collateral

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 66 of 92

shall be properly filed with the Court (x) by the Committee within seventy-five (75) calendar days from the date of entry of the Final Order, (y) if no Committee is appointed, by any party in interest with requisite standing within ninety (90) calendar days from the date of entry of the Final Order; provided, however, that nothing herein shall permit any party to challenge the extent or validity of the DIP Term Liens or the DIP Term Obligations. If any such First Lien Objection is timely and properly filed and a final, non-appealable order is entered by a court of competent jurisdiction sustaining and ordering some or all of the relief requested in such First Lien Objection, nothing in this Interim Order shall prevent the Court from granting appropriate relief with respect to the First Lien Notes Obligations or the First Lien Note Trustee's or the First Lien Noteholders' liens on the Prepetition Collateral. If no First Lien Objection is timely and properly filed, or if a First Lien Objection is timely and properly filed but denied, (i) the First Lien Notes Obligations shall be deemed allowed in full, shall not be subject to any setoff, recoupment, counterclaim, deduction or claim of any kind, and shall not be subject to any further objection or challenge by any party at any time, and the First Lien Notes Parties' prepetition liens on and security interest in the Prepetition Collateral shall be deemed legal, valid, perfected, enforceable, and non-avoidable for all purposes; and (ii) the First Lien Notes Parties, and each of their respective participants, agents, officers, directors, employees, attorneys, professionals, successors, and assigns (each in their respective capacities as such) shall be deemed released and discharged from any and all claims and causes of action that may be asserted by the Debtors' or their Estates, related to or arising out of the First Lien Notes Documents and the First Lien Notes Obligations, and shall not be subject to any further objection or challenge relating thereto or arising therefrom by any party at any time. Nothing contained in this Section 4.1(b) or otherwise shall or shall be deemed or construed to impair, prejudice or waive any rights, claims or

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 67 of 92

protections afforded to the DIP Term Parties in connection with all postpetition financial and credit accommodations provided by the DIP Term Parties in reliance on section 364(e) of the Bankruptcy Code and in accordance with the terms and provisions of this Interim Order and the DIP Loan Documents.

(c) Nothing in this Interim Order vests or confers on the Committee or any person (as defined in the Bankruptcy Code) standing or authority to pursue any cause of action belonging to the Debtors or their estates, and all rights to object to such standing are expressly reserved.

4.2 Debtors' Waivers. Prior to the indefeasible payment in full of all ABL Obligations and all DIP Term Obligations, any request by the Debtors with respect to the following shall constitute a DIP Termination Event: (a) to use Cash Collateral of DIP Agents, the DIP Lenders, or the Prepetition Secured Parties under section 363 of the Bankruptcy Code; (b) to obtain post-petition loans or other financial accommodations pursuant to section 364(c) or 364(d) of the Bankruptcy Code, other than as provided in this Interim Order or as may be otherwise expressly permitted pursuant to the DIP Loan Documents; (c) to challenge the application of any payments authorized by this Interim Order as pursuant to section 506(b) of the Bankruptcy Code, or to assert that the value of the Prepetition ABL Collateral is less than the Prepetition ABL Obligations; (d) to propose, support or have a plan of reorganization or liquidation that does not provide for the indefeasible payment in cash in full and satisfaction of all DIP Obligations on the effective date of such plan in accordance with the terms and conditions set forth in the DIP Loan Documents; or (e) to seek relief under the Bankruptcy Code, including without limitation, under section 105 of the Bankruptcy Code, to the extent any such relief would in any way restrict or impair (i) the rights and remedies of any of the DIP Agents or

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 68 of 92

the DIP Lenders against the Debtors as provided in this Interim Order or any of the DIP Loan Documents or (ii) the exercise of such rights or remedies by any of the DIP Agents or the DIP Lenders against the Debtors in accordance with the DIP Loan Agreements or this Interim Order (other than to object to the exercise of the rights and remedies within the Enforcement Notice Period on the grounds set forth in Section 3.2 of this Interim Order); <u>provided</u>, <u>however</u>, that the DIP Agents may otherwise consent in writing, but no such consent shall be implied from any other action, inaction, or acquiescence by DIP Agents or any DIP Lender.

4.3 Collateral Rights. Until all ABL Obligations and DIP Term Obligations shall have been indefeasibly paid and satisfied in full in accordance with the terms of the ABL Loan Documents and DIP Term Documents, subject to the terms of the Prepetition Intercreditor Agreements, it shall be an Event of Default pursuant to section 3.1(e) of this Interim Order if any party shall take any of the following actions, except to the extent permitted by, and subject in all respects to, the DIP Loan Documents: (a) take any action to, foreclose upon or recover, in connection with the liens granted thereto pursuant to the Prepetition Loan Documents, this Interim Order, or otherwise seek or exercise any enforcement rights or remedies against any DIP Collateral or in connection with the debt and obligations underlying the Prepetition Loan Documents or the Adequate Protection Liens, including, without limitation, in respect of the occurrence or continuance of any DIP Termination Event, (b) contest any release of DIP Collateral authorized under the DIP Loan Documents, (c) file any further financing statements, patent filings, trademark filings, copyright filings, mortgages, memoranda of lease, notices of lien or similar instruments, or otherwise take any action to perfect their security interests in the DIP Collateral unless, solely as to this clause (c), the DIP Lenders also file financing statements or other documents to perfect the liens granted pursuant to the DIP Loan Documents and/or this

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 69 of 92

Interim Order, or as may be required by applicable state law to continue the perfection of valid and unavoidable liens or security interests as of the date of filing, or (d) fail to deliver or cause to be delivered, at the Debtors' cost and expense, any termination statements, releases and/or assignments (to the extent provided for herein) in favor of the DIP Agents and the DIP Lenders or other documents necessary to effectuate and/or evidence the release, termination and/or assignment of the Adequate Protection Liens on any portion of the DIP Collateral subject to any sale or disposition approved or arranged for by the DIP Agents.

4.4 Release. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that in consideration of the ABL Agent, the ABL Lenders, the DIP Term Agent, and the DIP Term Lenders, permitting Debtors to use the Prepetition Collateral (including Cash Collateral) and providing other credit and financial accommodations to the Debtors pursuant to the provisions of the ABL Loan Documents, the DIP Term Documents, and this Interim Order, each Debtor, on behalf of itself and its successors and assigns, (collectively, the "ABL and DIP Term Releasors"), shall, forever release, discharge and acquit each ABL Agent, each ABL Lender, each DIP Term Agent, and each DIP Term Lender, and each of their respective successors and assigns, and their present and former shareholders, affiliates, subsidiaries, divisions, predecessors, directors, officers, attorneys, employees and other representatives in their respective capacities as such (collectively, the "Prepetition Releasees") of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, of every kind, nature and description, with respect to the ABL Obligations, ABL Loan Documents, the DIP Term Obligations, and the DIP Term Loan Documents, and any loans,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 70 of 92

advances, Letters of Credit, Bank Products, or other financial accommodations made by any of the ABL Lender Parties to Debtors (or any of them) pursuant to the ABL Loan Documents and Term Loan Documents including, without limitation, any so-called "lender liability" claims or defenses, that ABL and DIP Term Releasors had, have or hereafter can or may have against Prepetition Releases as of the date hereof, in respect of events that occurred on or prior to the date hereof. In addition, upon the repayment in full of all ABL Obligations owed to the ABL Agent and the ABL Lenders and the DIP Term Obligations owed to the DIP Term Agent and the DIP Term Lenders, by Debtors and the termination of the rights and obligations arising under the ABL Loan Documents and Term Loan Documents (which payment and termination shall be on terms and conditions acceptable to the ABL Agent and DIP Term Agent, respectively), the ABL Agent, the ABL Lenders, the DIP Term Agent, the DIP Term Lenders, shall be released from any and all obligations, liabilities, actions, duties, responsibilities and causes of action arising or occurring in connection with or related to the ABL Loan Documents, the DIP Loan Agreements, this Interim Order or the Final Order (including without limitation any obligation or responsibility (whether direct or indirect, absolute or contingent, due or not due, primary or secondary, liquidated or unliquidated) to pay or otherwise fund the Carve Out (or any portion thereof), on terms and conditions acceptable to the ABL Agent and DIP Term Agent, but for the avoidance of doubt, excluding any claims or causes of action for gross negligence or willful misconduct).

4.5 <u>Landlord Agreements; Access</u>. The Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that (1) all landlord agreements to which the Prepetition Agents are a party shall be deemed amended

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 71 of 92

to include the DIP Agents as beneficiaries thereunder, and such agreements shall thereafter be additionally enforceable by the DIP Agents against, and binding upon, each landlord party thereto; (2) any title, landlord's lien, right of distraint or levy, security interest or other interest that any landlord or mortgagee may have in any DIP Collateral or Prepetition Collateral of the Debtors located on such leased premises, to the extent the same is not void under section 545 of the Bankruptcy Code, is hereby expressly subordinated to the liens of the DIP Lenders and the Prepetition Lenders; and (3) without limiting any other rights or remedies of the DIP Agents or the other DIP Secured Parties, or otherwise available at law or in equity, and subject to the terms of the DIP Loan Documents, upon three (3) business days' written notice to counsel to the Debtors and any landlord, lienholder, licensor, or other third party owner of any leased or licensed premises or intellectual property, after the expiration of the Enforcement Notice Period, that a DIP Termination Event has occurred and is continuing, the DIP Agents, (i) may, unless otherwise expressly provided in any separate agreement by and between the applicable landlord or licensor and the DIP Agents (the terms of which shall be reasonably acceptable to the parties thereto), enter upon any leased or licensed premises of the Debtors for the purpose of exercising any remedy with respect to DIP Collateral located thereon, and (ii) shall be entitled to all of the Debtors' rights and privileges as lessee or licensee under the applicable license and to use any and all trademarks, trade names, copyrights, licenses, patents, or any other similar assets of the Debtors, which are owned by or subject to a lien of any third party and which are used by Debtors in their businesses, in either the case of subparagraph (i) or (ii) of this Section Error! **Reference source not found.**, without interference from lienholders or licensors thereunder; provided, however, that the DIP Agents (on behalf of the applicable DIP Lenders) shall pay only rent and additional rent, fees, royalties, or other monetary obligations of the Debtors that first

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 72 of 92

arise after the written notice referenced above from the DIP Agents and that accrue during the period of such occupancy or use by DIP Agents calculated on a per diem basis. Nothing herein shall require the Debtors, the DIP Agents or the other DIP Secured Parties, to assume any lease or license under Bankruptcy Code section 365(a) as a precondition to the rights afforded to the DIP Agents and the other DIP Secured Parties herein.

Section 5. <u>Other Rights and DIP Obligations.</u>

5.1 No Modification or Stay of This Interim Order. The DIP Agents and the DIP Lenders have acted in good faith in connection with the DIP Facility, the Interim Financing, and with this Interim Order, and their reliance on this Interim Order is in good faith. Based on the findings set forth in this Interim Order and the record made during the Interim Hearing, and in accordance with Section 364(e) of the Bankruptcy Code, in the event any or all of the provisions of this Interim Order are hereafter modified, reversed, amended or vacated by a subsequent order of the Court or any other court, the DIP Agents, the DIP Lenders, the Prepetition Agents and the Prepetition Lenders are entitled to the protections provided in section 364(e) of the Bankruptcy Code. Notwithstanding (a) any stay, modification, amendment, supplement, vacating, revocation or reversal of this Interim Order, any of the DIP Loan Documents or any term hereunder or thereunder, (b) the failure to obtain a Final Order pursuant to Bankruptcy Rule 4001(c)(2), or (c) the dismissal or conversion of one or more of the Cases (each, a "Subject Event"), (x) the acts taken by each of the DIP Agents and DIP Lenders in accordance with this Interim Order, and (y) the DIP Obligations incurred or arising prior to DIP Agents' actual receipt of written notice (email being sufficient) from the Debtors expressly describing the occurrence of such Subject Event shall, in each instance, be governed in all respects by the original provisions of this Interim Order, and the acts taken by each such DIP Agent and DIP Lender in accordance with this Interim Order, and the liens granted to each DIP

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 73 of 92

Agent and DIP Lender in the DIP Collateral, and all other rights, remedies, privileges, and benefits in favor of each DIP Agent and each DIP Lender pursuant to this Interim Order and the DIP Loan Documents shall remain valid and in full force and effect pursuant to section 364(e) of the Bankruptcy Code. For purposes of this Interim Order, the term "appeal", as used in section 364(e) of the Bankruptcy Code, shall be construed to mean any proceeding for reconsideration, amending, rehearing, or re-evaluating this Interim Order by this Court or any other tribunal.

5.2 [Reserved.]

5.3 <u>Rights of Access and Information</u>. The Debtors shall comply with the rights of access and information afforded to the DIP Agents and the DIP Lenders under the DIP Loan Documents, the Debtors shall be and hereby are required to afford representatives, agents and/or employees of the DIP Agents and the DIP Lenders reasonable access to the Debtors' premises during normal business hours and without interference with the proper operation of the Debtors' businesses and their books and records, and shall reasonably cooperate, reasonably consult with, and reasonably provide to such persons all reasonable information as may be reasonably requested with respect to the business, results of operations and financial condition of any of the Debtors. In addition, the Debtors, upon prior written request to the Debtors' proposed counsel, authorize their independent certified public accountants, financial advisors, investment bankers and consultants to consult with, and provide to the DIP Agents and to each DIP Lender and the Prepetition Agents and their respective professionals all such reasonable information as may be reasonably requested.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 74 of 92

5.4 <u>Power to Waive Rights; Duties to Third Parties</u>.

(a) The ABL Agent shall have the right to waive any of the terms, rights and remedies provided or acknowledged in this Interim Order that are in favor of ABL Agent or the ABL Lenders (the "<u>ABL DIP Lender Rights</u>"), and shall have no obligation or duty to any other party with respect to the exercise or enforcement, or failure to exercise or enforce, any ABL DIP Lender Right(s). Any waiver by ABL Agent of any ABL DIP Lender Rights shall not be or constitute a continuing waiver. Any delay in or failure to exercise or enforce any ABL DIP Lender Right shall neither constitute a waiver of such ABL DIP Lender Right, subject the ABL Agent or any ABL DIP Lender to any liability to any other party, nor cause or enable any party other than the Debtors to rely upon or in any way seek to assert as a defense to any obligation owed by the Debtors to the ABL Agent or any ABL Lender.

(b) The DIP Term Agent (at the direction of the Required DIP Term Lenders) shall have the right to waive any of the terms, rights and remedies provided or acknowledged in this Interim Order that are in favor of DIP Term Agent or the DIP Term Lenders (the "<u>DIP Term Lender Rights</u>"), and shall have no obligation or duty to any other party with respect to the exercise or enforcement, or failure to exercise or enforce, any DIP Term Lender Right(s). Any waiver by DIP Term Agent of any DIP Term Lender Rights shall not be or constitute a continuing waiver. Any delay in or failure to exercise or enforce any DIP Term Lender Right shall neither constitute a waiver of such DIP Term Lender Right, subject DIP Term Agent or any Term Lender to any liability to any other party, nor cause or enable any party other than the Debtors to rely upon or in any way seek to assert as a defense to any obligation owed by the Debtors to DIP Term Agent or any Term Lender

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 75 of 92

5.5 <u>Disposition of Collateral</u>. The Debtors shall not sell, transfer, lease, encumber or otherwise dispose of any portion of the DIP Collateral (including inventory), other than pursuant to the terms of this Interim Order and the DIP Loan Documents.

5.6 [Reserved.]

5.7 <u>Maintenance of DIP Collateral</u>. Unless the DIP Agents, acting at the direction of the applicable Required DIP Lenders, otherwise consent in writing, until (x) the indefeasible payment in full or otherwise acceptable satisfaction of all DIP Obligations and (y) the termination of the DIP Agents' and the DIP Lenders' obligations to extend credit under the DIP Facilities, the Debtors shall continue to maintain all property, operational and other insurance as required and as specified in the DIP Loan Documents and in each case in accordance with the DIP Loan Documents. Upon entry of this Interim Order and to the fullest extent provided by applicable law, each of the DIP Agents (on behalf of the applicable DIP Lenders) shall be, and shall be deemed to be, without any further action or notice, named as additional insureds and loss payees on each insurance policy maintained by the Debtors that in any way relates to the DIP Collateral. The Debtors shall also maintain the cash management system in effect as of the Petition Date, as modified by this Interim Order and any order of the Court authorizing the continued use of the Debtors' cash management system that is acceptable to the DIP Agents in all material respects.

5.8 <u>Reservation of Rights</u>. The terms, conditions and provisions of this Interim Order are in addition to and without prejudice to the rights of each DIP Agent and DIP Lender to pursue any and all rights and remedies under the Bankruptcy Code, the DIP Loan Documents or any other applicable agreement or law, including, without limitation, rights to seek adequate protection and/or additional or different adequate protection, to seek relief from the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 76 of 92

automatic stay, to seek an injunction, to oppose any request for use of cash collateral or granting of any interest in the DIP Collateral, as applicable, or priority in favor of any other party, to object to any sale of assets, and to object to applications for allowance and/or payment of compensation of Professionals or other parties seeking compensation or reimbursement from the Estates.

5.9 <u>Binding Effect</u>.

(a) The provisions of this Interim Order and the DIP Loan Documents, the Postpetition ABL Obligations, the DIP Term Obligations, all Liens and Claims granted hereunder in favor of each of the DIP Agents and DIP Lenders, and any and all rights, remedies, privileges and benefits in favor of each DIP Agent and DIP Lenders provided or acknowledged in this Interim Order, and any actions taken pursuant thereto, shall be effective and enforceable immediately upon entry of this Interim Order notwithstanding Bankruptcy Rules 4001(a)(3), 6004(h) and 7062, shall continue in full force and effect, and shall survive entry of any such other order, including without limitation any order which may be entered confirming any plan of reorganization, converting one or more of the Cases to any other chapter under the Bankruptcy Code, or dismissing one or more of the Cases; <u>provided</u> that the terms and conditions of the Final Order shall control over this Interim Order except with respect to any DIP Obligations incurred by the Debtors before entry of the Final Order.

(b) Any order dismissing one or more of the Cases under section 1112 or otherwise shall be deemed to provide (in accordance with §§ 105 and 349 of the Bankruptcy Code) that (i) the DIP ABL Superpriority Claim and ABL Agent's and ABL Lenders' liens on and security interests in the DIP Collateral and all other claims, liens, adequate protections and other rights granted pursuant to the terms of this Interim Order shall continue in full force and

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 77 of 92

effect notwithstanding such dismissal until the ABL Obligations are indefeasibly paid and satisfied in full, (ii) the DIP Term Superpriority Claim and DIP Term Agent's and DIP Term Lenders' liens on and security interests in the DIP Collateral and all other claims, liens, adequate protections and other rights granted pursuant to the terms of this Interim Order shall continue in full force and effect notwithstanding such dismissal until the DIP Term Obligations are indefeasibly paid and satisfied in full, and (iii) this Court shall retain jurisdiction, notwithstanding such dismissal, for the purposes of enforcing all such claims, liens, protections and rights.

(c) In the event this Court modifies any of the provisions of this Interim Order or any of the DIP Loan Documents following a Final Hearing, such modifications shall not affect the rights or priorities of any DIP Agent or DIP Lender pursuant to this Interim Order with respect to the DIP Collateral or any portion of the DIP Obligations which arises or is incurred or is advanced prior to such modifications.

(d) This Interim Order shall be binding upon Debtors, all parties in interest in the Cases and their respective successors and assigns, including any trustee or other fiduciary appointed in the Cases or any subsequently converted bankruptcy case(s) of any Debtor. This Interim Order shall also inure to the benefit of Debtors, DIP Agents, DIP Lenders, First Lien Notes Parties, Second Lien Notes Parties, Secured Creditors and each of their respective successors and assigns. This Interim Order shall also inure to the benefit of Debtors, DIP Agents, DIP Lenders, First Lien Notes Parties, Second Lien Notes Parties, Secured Creditors and each of their respective successors and assigns.

5.10 <u>Discharge</u>. The DIP Obligations and the obligations of the Debtors with respect to adequate protection hereunder shall not be discharged by the entry of an order

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 78 of 92

confirming any plan of reorganization in any of the Chapter 11 Cases, notwithstanding the provisions of section 1141(d) of the Bankruptcy Code, unless such obligations have been indefeasibly paid in full in cash, on or before the effective date of such confirmed plan of reorganization, or each of the DIP Secured Parties has otherwise agreed in writing. Prior to the indefeasible payment in full of either DIP Facility, it shall be a DIP Termination Event under such DIP Facility if the Debtors propose or support any Chapter 11 plan or sale of all or substantially all of the Debtors' assets, or order confirming such plan or approving such sale, that is not conditioned upon the indefeasible payment of the DIP Obligations in full in cash and the payment of the Debtors' obligations with respect to the adequate protection hereunder, in full in cash, within a commercially reasonable period of time, and in any event no later than the effective date of such Chapter 11 plan or sale, without the written consent of the DIP Agents and the applicable Required DIP Lenders.

5.11 Section 506(c) Waiver. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that in consideration for, among other things, the Carve Out and the payments to be made under the Approved Budget to administer the Chapter 11 Cases with the use of Cash Collateral, no costs or expenses of administration which have been or may be incurred in the Chapter 11 Cases at any time (including, without limitation, any costs and expenses incurred in connection with the preservation, protection or enhancement of realization by the DIP Lenders upon the DIP Collateral, or by the Prepetition Lenders upon the Prepetition Collateral, as applicable) shall be charged against any of the DIP Agents, DIP Lenders, Prepetition Agent or any Prepetition Lenders or any of the DIP Obligations or Prepetition Secured Obligations or the

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 79 of 92

DIP Collateral or the Prepetition Collateral pursuant to sections 105 or 506(c) of the Bankruptcy Code or otherwise without the prior express written consent of the affected Prepetition Agent and/or affected Prepetition Lender, in their sole discretion, and no such consent shall be implied, directly or indirectly, from any other action, inaction, or acquiescence by any such agents or creditors (including, without limitation, consent to the Carve Out or the approval of any budget hereunder).

5.12 <u>Section 552(b) Waiver</u>. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the Prepetition Agents and the Prepetition Lenders are and shall each be entitled to all of the rights and benefits of section 552(b) of the Bankruptcy Code and the "equities of the case" exception under section 552(b) shall not apply to the Prepetition Agents, the Prepetition Lenders or the Prepetition Secured Obligations.

5.13 <u>No Marshaling/Application of Proceeds</u>. Notwithstanding the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that in no event shall the DIP Agents, the DIP Lenders, the Prepetition Agents or the Prepetition Lenders be subject to the equitable doctrine of "marshaling" or any similar doctrine with respect to the DIP Collateral or the Prepetition Collateral, as applicable, and all proceeds shall be received and applied in accordance with the DIP Loan Documents and the Prepetition Loan Documents, as applicable.

5.14 <u>Right of Setoff</u>. To the extent any funds were on deposit with ABL Agent as of the Petition Date, including, without limitation, all funds deposited in, or credited to, an

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 80 of 92

account of any Debtor with the ABL Agent or any ABL Lender immediately prior to the filing of the Cases (regardless of whether, as of the Petition Date, such funds had been collected or made available for withdrawal by any such Debtor), such funds (the "<u>Deposited Funds</u>") are subject to rights of setoff to the extent provided in the ABL Loan Documents. By virtue of such setoff rights, the Deposited Funds are subject to a lien in favor of ABL Agent or the applicable ABL Lenders pursuant to section 506(a) and 553 of the Bankruptcy Code.

Right to Credit Bid. Subject to section 363(k) of the Bankruptcy Code and 5.15 the terms of the DIP Loan Documents, the DIP Agents shall have the right to credit bid as part of any asset sale process and shall have the right to credit bid all or any portion of the full amount of the DIP Obligations during the sale of the DIP Collateral (in whole or in part), including without limitation, sales pursuant to section 363 of the Bankruptcy Code or included as part of any plan of reorganization subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code; provided that any credit bid by (x) the ABL Agent or any of the ABL Loan Parties during the sale of any DIP Collateral on which the ABL Agent holds a lien junior in priority to the lien of the DIP Term Agent on such DIP Collateral, whether pursuant to section 363 of the Bankruptcy Code or included as part of any plan of reorganization subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code, shall include the payment in full in cash of all DIP Term Obligations to the DIP Term Parties; and (y) the DIP Term Agent or any of the other DIP Term Parties during the sale of any DIP Collateral on which such of the DIP Term Agent or any of the DIP Term Parties holds a lien junior in priority to the lien of the ABL Agent on such DIP Collateral, whether pursuant to section 363 of the Bankruptcy Code or included as part of any plan of reorganization subject to confirmation under section 1129(b)(2)(A)(ii)–(iii) of the Bankruptcy Code, shall include the payment in full in cash

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 81 of 92

of all ABL Obligations to the ABL Loan Parties. In addition to the foregoing, the Prepetition Secured Parties intend to require that the Final Order shall provide, and the Debtors have agreed as a condition to obtaining financing under the DIP Facilities, to seek Court approval at the Final Hearing that the Prepetition Secured Parties shall have the right to credit bid as part of any asset sale process and shall have the right to credit bid all or any portion of the full amount of their respective claims, including, for the avoidance of doubt, Adequate Protection Claims, if any, during any sale of the Debtors' assets (in whole or part) with respect to any asset subject to a duly perfected lien in favor of the Prepetition Secured Parties as of the Petition Date, including without limitation, sales occurring pursuant to Bankruptcy Code section 363 or included as part of any restructuring plan subject to confirmation under Bankruptcy Code section 1129(b)(2)(A)(ii)–(iii).

5.16 Payment and Review of Lender Fees and Expenses. Each Debtor shall pay all fees that may be reasonably required or necessary for the Debtors' performance of their obligations under the ABL Credit Agreement or the DIP Term Loan Agreement, as applicable, including, without limitation, the non-refundable payment to the ABL Agent, the ABL Lenders, the DIP Term Agent, and the DIP Term Lenders, as the case may be, of the reasonable and documented fees and expenses set forth in the DIP Loan Documents, as applicable, whether incurred before or after the Petition Date; provided, that the Debtors shall pay all such reasonable and documented fees and expenses within ten (10) days of delivery of a statement or invoice for such fees and expenses (it being understood that such statements or invoices shall not be required to be maintained in accordance with the U.S. Trustee Guidelines, nor shall any such counsel or other professional be required to file any interim or final fee applications with the Court or otherwise seek Court's approval of any such payments) (the "Review Period") to (a) the Debtors,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 82 of 92

(b) the U.S. Trustee (c) the Committee, and (d) counsel to the Ad Hoc Committee (collectively, the "<u>Fee Notice Parties</u>"), unless prior to 10:00 a.m. (prevailing Eastern Time) on the final day of the Review Period, a Fee Notice Party serves a written objection (a "<u>Fee Objection</u>") detailing (x) the specific fees or expenses to which Fee Notice Party objects and (y) a good faith detailed explanation of the legal and factual basis for such Fee Objection, in which case, the Debtors shall pay only such amounts that are not the subject of any Fee Objection and the withheld amount subsequently agreed by the Fee Notice Parties or ordered by the Court to be paid.

5.17 <u>Limits on Lender Liability</u>.

(a) In determining to make any loan under the DIP Loan Agreements or in exercising any rights or remedies as and when permitted pursuant to this Interim Order or the DIP Loan Documents, the DIP Agents and the DIP Lenders shall not be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors, so long as the DIP Lenders' actions do not constitute, within the meaning of 42 U.S.C. § 9601(20)(F), actual participation in the management or operational affairs of a vessel or facility owned or operated by a Debtor, or otherwise cause liability to arise to the federal or state government or the status of responsible person or managing agent to exist under applicable law (as such terms, or any similar terms, are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. §§ 9601 et seq., as amended, or any similar federal or state statute).

(b) Nothing in this Interim Order or the DIP Loan Documents shall permit the Debtors to violate 28 U.S.C. § 959(b).

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 83 of 92

(c) As to the United States, its agencies, departments, or agents, nothing in this Interim Order or the DIP Loan Documents shall discharge, release or otherwise preclude any valid right of setoff or recoupment that any such entity may have.

Survival. The provisions of this Interim Order and any actions taken 5.18 pursuant hereto shall survive, and shall not be modified, impaired or discharged by, entry of any order that may be entered (a) confirming any plan of reorganization in any of the Chapter 11 Cases, (b) converting any or all of the Chapter 11 Cases to a case under chapter 7 of the Bankruptcy Code, (c) dismissing any or all of the Chapter 11 Cases, or (d) pursuant to which the Court abstains from hearing any of the Chapter 11 Cases. The terms and provisions of this Interim Order, including the claims, liens, security interests, and other protections (as applicable) granted to the DIP Agents, the DIP Lenders, the Prepetition Agents and the Prepetition Lenders pursuant to this Interim Order, notwithstanding the entry of any such order, shall continue in any of the Chapter 11 Cases, following dismissal of any of the Chapter 11 Cases or any Successor Cases, and shall maintain their priority as provided by this Interim Order until (i) in respect of the DIP Facility, all of the DIP Obligations, pursuant to the DIP Loan Documents and this Interim Order, have been indefeasibly paid in full in cash (such payment being without prejudice to any terms of provisions contained in the DIP Facility which survive such discharge by their terms) and all commitments to extend credit under the DIP Facility are terminated, and (ii) in respect of the Prepetition Secured Obligations, all of the adequate protection obligations owed to the Prepetition Agents and the Prepetition Lenders provided for in this Interim Order have been indefeasibly paid in full in cash.

5.19 <u>Entry of this Order/Waiver of Applicable Stay</u>. This Interim Order shall be effective and enforceable upon its entry as of the Petition Date and not subject to any stay of

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 84 of 92

execution or effectiveness (all of which are hereby waived), notwithstanding anything to the contrary contained in Bankruptcy Rule 4001(a)(3).

5.20 <u>Nunc Pro Tunc Effect of this Order</u>. This Interim Order shall constitute findings of fact and conclusions of law and shall take effect and be fully enforceable *nunc pro tunc* to the Petition Date immediately upon entry thereof. Notwithstanding Bankruptcy Rules 4001(a)(3), 6004(h), 6006(d), 7062 and 9024 or any other Bankruptcy Rule, or Rule 62(a) of the Federal Rules of Civil Procedure, this Order shall be immediately effective and enforceable upon its entry and there shall be no stay of execution or effectiveness of this Interim Order.

5.21 <u>Limited Effect</u>. In the event of a conflict between the terms and provisions of any of the DIP Loan Documents and this Interim Order, the terms and provisions of this Interim Order shall govern.

5.22 <u>Retention of Jurisdiction</u>. The Court retains exclusive jurisdiction and power with respect to all matters arising from or related to the implementation or interpretation of this Interim Order, the DIP Loan Agreements, and the other DIP Loan Documents.

Section 6. <u>Final Hearing and Response Dates.</u>

The Final Hearing on the Motion pursuant to Bankruptcy Rule 4001(c)(2) is scheduled for March 6, 2018, at 10:00 a.m. (prevailing Eastern Time) before this Court. The Debtors shall promptly mail copies of this Interim Order to the Notice Parties, and to any other party that has filed a request for notices with this Court and to any Committee after it has been appointed, or the Committee's counsel, if it shall have filed a request for notice. The Debtors may serve the Motion and the Interim Order without the exhibits attached thereto, as such exhibits are voluminous and available, free of charge, at https://cases.primeclerk.com/cenveo, and such notice is deemed good and sufficient and no further notice need be given. Any party in interest objecting to the relief sought at the Final Hearing shall serve and file written objections, which

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 85 of 92

objections shall be served upon (a) proposed counsel to the Debtors, Kirkland & Ellis LLP, 601 Lexington Avenue, New York, N.Y. 10022, Attn: Jonathan S. Henes, Esq., Gregory Pesce, Esq., Melissa Koss, Esq., and Natasha Hwangpo, Esq., (b) counsel to ABL Agent, Otterbourg, P.C. 230 Park Avenue, New York N.Y. 10169, Attn: Daniel F. Fiorillo, Esq., Andrew M. Kramer, Esq., and Chad B. Simon, Esq., (c) counsel to DIP Term Agent, the DIP Term Lenders and the Ad Hoc Committee, Stroock & Stroock & Lavan LLP, 180 Maiden Lane, New York, NY 10038, Attn: Erez E. Gilad, Esq. and Gabriel Sasson, Esq., (d) counsel to First Lien Term Loan Agent, (e) counsel to Second Lien Term Loan Agent, (f) counsel to FILO Notes Trustee, (g) counsel to the indenture trustee of the 6.00% Senior Unsecured Notes due 2024 (i) counsel to any Committee; and (h) the U.S. Trustee, Attn: Paul Schwartzberg, Shannon Scott and Brian Masumoto; and shall be filed with the Clerk of the United States Bankruptcy Court for the Southern District of New York, with a copy to the Court's chambers, in each case, to allow actual receipt of the foregoing no later than 4:00 p.m. (prevailing Eastern Time), on February 27, 2018.

Dated: White Plains, New York February 6, 2018

> /s/ Robert D. Drain HONORABLE ROBERT D. DRAIN UNITED STATES BANKRUPTCY JUDGE

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 86 of 92

APPENDIX 1

"<u>ABL Priority Collateral</u>" means all Prepetition Collateral and all DIP Collateral that is of the same type, kind, nature or character as the ABL Priority Collateral (as defined in the First Lien Intercreditor Agreement).

"DIP Collateral" means all assets and properties (whether tangible, intangible, real, personal or mixed) of each of the Debtors and their estates, whether now owned by or owing to, or hereafter acquired by (including the Debtors' interest in property owned or consigned by or to, or leased from or to the Debtors), or arising in favor of, the Debtors (including under any trade names, styles, or derivations thereof), and whether hereafter acquired, owned or consigned by or to, or leased from or to, each of the Debtors, and regardless of where located, before or after the Petition Date, including, without limitation, all of the Debtors' rights, title and interests in: (i) all Prepetition ABL Collateral, First Lien Notes Collateral, and all assets and properties securing other Indebtedness of the Debtors and their estates in existence on the Petition Date; (ii) all cash and cash equivalents; (iii) all funds in any deposit account, securities account or other account of the Debtors, and all cash, cash equivalents, securities, instruments, investments, and other property deposited therein or credited thereto from time to time; (iv) all accounts and other receivables (including those owed to the Debtors generated by intercompany transactions or credit card receivables); (v) all contracts and contract rights; (vi) all instruments, documents and chattel paper; (vii) all securities and other financial assets (whether or not marketable); (viii) all goods, as-extracted collateral, equipment (including documents of title), furniture, inventory and fixtures (in each case, wherever located); (ix) all real property interests; (x) all interests in leaseholds and the proceeds thereof, (xi) all franchise rights; (xii) all patents, tradenames, trademarks (other than intent-to-use trademarks), servicemarks, copyrights, licenses and all other intellectual property; (xiii) all general intangibles (including payment intangibles and all rights to payment, including tax refunds and any other extraordinary receipts or payments); (xiv) all equity interests, capital stock, limited liability company interests, partnership interests and financial assets; (xv) all investment property; (xvi) all supporting obligations; (xvii) all letters of credit issued to the Debtors and letter of credit rights; (xviii) all commercial tort claims; (xix) all actions brought under section 549 of the Bankruptcy Code to recover any postpetition transfer of property; (xx) all other claims and causes of action and the proceeds thereof (including, subject to entry of the Final Order, the proceeds of avoidance actions arising under chapter 5 of the Bankruptcy Code) (xxi) all books and records (including, without limitation, customer lists, credit files, computer programs, printouts and other computer materials and records) pertaining to any of the foregoing, all property in which such books and records and information are stored, recorded and maintained, and all rights of access to such books, records and information; (xxii) to the extent not covered by the foregoing, all other goods, assets or properties of the Debtors, whether tangible, intangible, real, personal or mixed; (xxiii) all products, offspring, profits, and proceeds of each of the foregoing and all accessions to, substitutions and replacements for, and rents, profits and products of, each of the foregoing, and any and all proceeds of any insurance, refunds, premiums, rebates, indemnity, warranty or guaranty payable to such Debtor from time to time with respect to any of the foregoing; and (xxiv) all collateral security guarantees, liens, rights, remedies, and privileges pertaining to any of the foregoing, including the right of stoppage in transit; provided, however, that the DIP Collateral shall not include the Excluded Property (as defined in each of the DIP Loan Agreements); provided,

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 87 of 92

further, however, that the DIP Collateral shall include all proceeds and products of Excluded Property.

"<u>FILO Notes</u>" means the 4% Senior Secured Notes due 2021 pursuant to the FILO Notes Indenture, by and among the Borrower, the guarantors party thereto and the FILO Notes Trustee and FILO Notes Documents in favor of the FILO Notes Trustee or any holder of the FILO Notes.

"<u>FILO Notes Collateral</u>" means "Collateral" as defined in the FILO Notes Indenture.

"FILO Noteholder" means any holder of the FILO Notes.

"<u>FILO Notes Documents</u>" means the FILO Notes Indenture, and all other agreements, documents and instruments executed and/or delivered with, to, or in favor of FILO Notes Trustee or any holder of the FILO Notes in connection with the FILO Notes Indenture, including, without limitation, all security agreements, notes, guarantees, mortgages, Uniform Commercial Code financing statements and all other related agreements, documents and instruments executed and/or delivered in connection therewith or related thereto (all of the foregoing, together with the FILO Notes Indenture and FILO Notes, as all of the same have heretofore been amended, supplemented, modified, extended, renewed, restated and/or replaced at any time prior to the Petition Date.

"<u>FILO Notes Indenture</u>" means that certain Indenture date June 10, 2016 issue the FILO Notes.

"<u>FILO Notes Obligations</u>" means as of the Petition Date, the indebtedness to the FILO Notes Parties under the FILO Notes Documents in respect of loans and other obligations evidenced thereby, plus interest accrued and accruing thereon, together with all costs, fees, expenses (including attorneys' fees and legal expenses) and other charges accrued, accruing or chargeable with respect thereto.

"FILO Notes Parties" means the FILO Noteholders and the FILO Notes Trustee.

"<u>FILO Notes Trustee</u>" means The Bank of New York Mellon, in its capacities as trustee and as collateral agent pursuant to the FILO Notes Indenture.

"<u>First Lien Notes Parties</u>" means the First Lien Noteholders and the First Lien Notes Trustee.

"<u>Prepetition Collateral</u>" means collectively, the FILO Notes Collateral, the Prepetition ABL Collateral, the First Lien Notes Collateral, and the Second Lien Notes Collateral.

"<u>Prepetition ABL Credit Agreement</u>" means that certain Credit Agreement, dated as of April 16, 2013 (as amended, supplemented, amended and restated or otherwise modified from time to time prior to the Petition Date in accordance with the terms thereof), by and among Cenveo Corporation, Borrower, the Guarantors, Bank of America, N.A., as administrative agent and collateral agent, and the lenders party thereto.

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 88 of 92

"<u>Second Lien Notes</u>" means 8.500% Junior Priority Secured Notes due 2022 pursuant to the Second Lien Notes Indenture, by and among the Borrower, the guarantors party thereto and the Second Lien Notes Trustee and Second Lien Notes Documents in favor of Second Lien Notes Trustee or any holder of the Second Lien Notes.

"<u>Second Lien Notes Collateral</u>" means "Collateral" as defined in the Second Lien Notes Indenture.

"Second Lien Notes Documents" means the Second Lien Notes Indenture, and all other agreements, documents and instruments executed and/or delivered with, to, or in favor of Second Lien Notes Trustee or any holder of the Second Lien Notes in connection with the Second Lien Notes Indenture, including, without limitation, all security agreements, notes, guarantees, mortgages, Uniform Commercial Code financing statements and all other related agreements, documents and instruments executed and/or delivered in connection therewith or related thereto (all of the foregoing, together with the Second Lien Notes Indenture and Second Lien Notes, as all of the same have heretofore been amended, supplemented, modified, extended, renewed, restated and/or replaced at any time prior to the Petition Date.

"Second Lien Notes Indenture" means that certain Indenture dated June 26, 2014 issuing the Second Lien Notes

"Second Lien Noteholder" means any holder of the Second Lien Notes.

"<u>Second Lien Notes Obligations</u>" means as of the Petition Date, the indebtedness to the Second Lien Notes Parties under the Second Lien Notes Documents in respect of loans and other obligations evidenced thereby, plus interest accrued and accruing thereon, together with all costs, fees, expenses (including attorneys' fees and legal expenses) and other charges accrued, accruing or chargeable with respect thereto.

"Second Lien Notes Parties" means the Second Lien Noteholders and the Second Lien Notes Trustee.

"Second Lien Notes Trustee" means The Bank of New York Mellon, in its capacities as trustee and as collateral agent pursuant to the Second Lien Notes Indenture.

"<u>Term Priority Collateral</u>" means all Prepetition Collateral and DIP Collateral that is of the same type, kind, nature or character as the Fixed Asset Priority Collateral (as defined in the First Lien Intercreditor Agreement).

[Remainder of Page Intentionally Left Blank.]

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 89 of 92

EXHIBIT 1

ABL Credit Agreement

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 90 of 92

EXHIBIT 2

DIP Term Loan Agreement

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 91 of 92

EXHIBIT 3

Initial Budget

18-22178-rdd Doc 51 Filed 02/06/18 Entered 02/06/18 17:00:35 Main Document Pg 92 of 92

EXHIBIT 4

Weekly Professional Fee Schedule

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 1 of 432

EXHIBIT 1

ABL Credit Agreement

[Execution]

RATIFICATION AND AMENDMENT AGREEMENT

This RATIFICATION AND AMENDMENT AGREEMENT (this "<u>Agreement</u>"), dated as of February 5, 2018, is by and among CENVEO CORPORATION, a Delaware corporation, as a debtor and debtor-in-possession under Chapter 11 of the Bankruptcy Code ("<u>Borrower</u>"), each of the other LOAN PARTIES party hereto, each as a debtor and debtor-in-possession under Chapter 11 of the Bankruptcy Code, BANK OF AMERICA, N.A., as administrative agent (in such capacity, "<u>Administrative Agent</u>") and each of the LENDERS party hereto.

$\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$:

WHEREAS, Borrower, the other Loan Parties, Administrative Agent and the Lenders entered into the Credit Agreement, dated as of April 16, 2013, as amended by Amendment No. 1 to the Credit Agreement, dated as of December 11, 2013, Amendment No. 2 to the Credit Agreement, dated as of June 10, 2014, Amendment No. 3 to the Credit Agreement, dated as of January 30, 2015, Amendment No. 4 to the Credit Agreement, dated as of June 10, 2016 and Amendment No. 5 to the Credit Agreement, dated as of April 27, 2017 (as so amended and otherwise in effect immediately prior to the Ratification Effective Date, the "Existing Credit Agreement"; and as amended by this Agreement on the Ratification Effective Date and as otherwise amended and in effect after the Ratification Effective Date, the "Credit Agreement").

WHEREAS, Borrower and each Guarantor (collectively, the "<u>Debtors</u>") has commenced a case under Chapter 11 of Title 11 of the United States Code in the United States Bankruptcy Court for the Southern District of New York, and each Debtor has retained possession of its assets and is authorized under the Bankruptcy Code to continue the operation of its businesses as a debtor-in-possession;

WHEREAS, prior to the commencement of the Chapter 11 Cases (as defined below), Lenders made loans and advances and provided other financial accommodations to Borrower secured by substantially all assets and properties of Borrower and Guarantors as set forth in the Loan Documents (as defined in the Existing Credit Agreement, the "Existing Loan Documents");

WHEREAS, Borrower and Guarantors have requested that Administrative Agent and Lenders make post-petition loans and advances and provide other financial accommodations to Borrower secured by substantially all the assets and properties of Borrower and Guarantors as set forth in the Financing Orders (as defined below) and the Loan Documents and make certain amendments to the Existing Credit Agreement, and Administrative Agent and Lenders are willing to do so, subject to (a) entry by the Bankruptcy Court (as defined below) of an Interim Financing Order (as defined below) and (b) the other terms and conditions contained herein; and

WHEREAS, Borrower and Guarantors desire to reaffirm their obligations to Administrative Agent and Lenders pursuant to the Existing Loan Documents and acknowledge their continuing liabilities to Administrative Agent and Lenders thereunder in order to induce Administrative Agent and Lenders to make such post-petition loans and advances, and provide such other financial accommodations, to Borrower;

NOW, THEREFORE, in consideration of the foregoing, the mutual covenants and agreements contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Administrative Agent, Lenders, Borrower and Guarantors mutually covenant, warrant and agree as follows:

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 3 of 432

1. Definitions and Interpretation.

1.1 <u>Definitions</u>. For purposes of this Agreement, all terms used herein which are not otherwise defined herein, including but not limited to, those terms used in the recitals hereto, shall have the respective meanings assigned thereto in the Existing Credit Agreement, as amended by this Agreement.

1.2 <u>Additional Definitions</u>. In addition, the following terms, as used in this Agreement, shall have the following meanings:

"<u>Approved Budget</u>" means the Initial Budget, at all times until updated pursuant to <u>Section 5.3(b)</u> by an Updated Budget during the continuance of the Chapter 11 Cases.

"<u>Avoidance Action</u>" means any of the actions described in clause (g) of the definition of "Post-Petition Collateral".

"<u>Bankruptcy Court</u>" means the United States Bankruptcy Court for the Southern District of New York.

"Budget Delivery Date" has the meaning set forth in Section 5.3(b).

"<u>Carve-Out</u>" has the meaning set forth in the Financing Orders.

"Carve-Out Reserve" has the meaning set forth in the Financing Orders.

"<u>Chapter 11 Cases</u>" means the Chapter 11 cases of the Debtors, which are being jointly administered under the Bankruptcy Code and are pending in the Bankruptcy Court.

"<u>Collateral</u>" means all of the "Collateral" and "Mortgaged Property" referred to in the Collateral Documents and all of the other property that is or is intended under the terms of the Collateral Documents to be subject to Liens in favor of Administrative Agent for the benefit of the Secured Parties (including both the Pre-Petition Collateral and the Post-Petition Collateral).

"<u>Cumulative Net Operating Cash Flow</u>" means, on any date of determination, the amount equal to (a) Total Cash Receipts minus (b) Total Operating Disbursements, measured in each case, on a cumulative basis, for the period commencing on the Ratification Effective Date through and including the last day of the then most recently ended Testing Period.

"<u>DIP ABL Facility</u>" means the financing arrangements and other transactions evidenced by the Credit Agreement and the other Loan Documents.

"<u>DIP Superpriority Claims</u>" has the meaning as defined in the Financing Orders in effect on the date of determination.

"<u>Final Financing Order</u>" means an order of the Bankruptcy Court entered in the Chapter 11 Cases authorizing the Credit Agreement after a final hearing of the Bankruptcy Court, in form and substance satisfactory to Administrative Agent and the Required Lenders, and from which no appeal or motion to reconsider has been filed, together with all extension, modifications, and amendments thereto consented to by Administrative Agent, which, among other matters but not by way of limitation, authorizing, on a final basis, Borrower to execute and perform under the terms of Existing Credit Agreement and the other Existing Loan Documents, in each case, as ratified and amended by the terms and conditions of this Agreement, granting to Administrative Agent and Lenders the senior security interests and liens described

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 4 of 432

therein and super-priority administrative expense claims described therein, and modifying the automatic stay and other provisions reasonably required by Administrative Agent and the Required Lenders. The Final Financing Order shall be substantially in the form of the Interim Financing Order, with only such changes as may be satisfactory in form and substance to the Administrative Agent and the Required Lenders.

"<u>Financing Order</u>" means the Interim Financing Order, the Final Financing Order and such other orders relating thereto or authorizing the granting of credit by Administrative Agent and Lenders to Borrower on an emergency, interim or permanent basis pursuant to Section 364 of the Bankruptcy Code as may be issued or entered by the Bankruptcy Court in the Chapter 11 Cases.

"Initial 9-Month Budget" means the budget attached as Exhibit C hereto

"Initial Budget" has the meaning set forth in Section 5.3(a).

"Interim Financing Order" means an order of the Bankruptcy Court entered in the Chapter 11 Cases after an interim hearing, in form and substance reasonably satisfactory to Administrative Agent and the Required Lenders, together with all extension, modifications, and amendments thereto consented to by Administrative Agent and the Required Lenders, which, among other matters but not by way of limitation, authorizing, on an interim basis, Borrower to execute and perform under the terms of Existing Credit Agreement and the other Existing Loan Documents, in each case, as ratified and amended by the terms and conditions of this Agreement.

"Milestones" means each of the covenants set forth in Section 5.6.

"Obligations" means (a) all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding and (b) Secured Bank Product Obligations, excluding, with respect to any Guarantor at any time, Excluded Swap Obligations (and including the Pre-Petition Obligations and the Post-Petition Obligations).

"<u>Permitted Variance</u>" means for any Testing Period set forth in the Approved Budget, (a) the Total Cash Receipts shall not be less than 80% of the budgeted amount as set forth in the Approved Budget with respect to the one calendar week period ending February 9, 2018, 85% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 90% of the budgeted amount as set forth in the Approved Budget with respect to each Testing Period thereafter, and (b) the Total Operating Disbursements shall not exceed 120% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 9, 2018, 115% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 110% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 110% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 110% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 110% of the budgeted amount as set forth in the Approved Budget with respect to each Testing Period thereafter (without giving effect to the making of Revolving Loans or DIP Term Loans or the repayments or prepayments of Revolving Loans or DIP Term Loans).

"Petition Date" means the date of the commencement of the Chapter 11 Cases.

"<u>Post-Petition Collateral</u>" means, collectively, all now existing and hereafter acquired real and personal property of each Debtor's estate, wherever located, of any kind, nature or description, including

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 5 of 432

any such property in which a lien is granted to Agent and Lenders pursuant to the Loan Documents or the Financing Orders, and shall include, without limitation:

(a) all of the Pre-Petition Collateral;

(b) all accounts, chattel paper, deposit accounts, documents (as defined in the UCC), equipment, general intangibles, instruments, inventory, investment property and any supporting obligations related to any of the foregoing;

(c) the commercial tort claims described on Schedule 2 to the Security Agreement and on any Security Agreement Supplement thereto received by Administrative Agent pursuant to the Security Agreement;

(d) all books and records pertaining to the Collateral;

(e) all property of such Debtor held by any Secured Party, including all property of every description, in the custody of or in transit to such Secured Party for any purpose, including safekeeping, collection or pledge, for the account of such Debtor or as to which such Debtor may have any right or power, including but not limited to cash;

(f) all other goods (including but not limited to fixtures) and personal property of such Debtor, whether tangible or intangible and wherever located;

(g) all proceeds recovered by or on behalf of each Borrower and Guarantor or any trustee of such Borrower or Guarantor (whether in the Chapter 11 Cases or any subsequent case to which any of the Chapter 11 Cases is converted) as a result of transfers or obligations avoided or actions maintained or taken pursuant to Sections 542, 545, 547, 548, 549, 550, 551, 552 and 553 of the Bankruptcy Code, subject to the entry of the Final Financing Order;

(h) all assets of such Debtor not otherwise subject to a valid and enforceable Lien in favor of Administrative Agent on the Petition Date, subject to the entry of the Interim Financing Order; and

(i) to the extent not otherwise included, all proceeds of the foregoing;

<u>provided</u>, that "Post-Petition Collateral" shall not include any Excluded Property (as defined in the Security Agreement) or Equity Interests that do not constitute Pledged Equity.

"<u>Post-Petition Obligations</u>" means all Obligations arising on and after the Petition Date and whether arising on or after the conversion or dismissal of the Chapter 11 Cases, or before, during and after the confirmation of any plan of reorganization in the Chapter 11 Cases, and whether arising under or related to this Agreement, the Credit Agreement, the other Loan Documents, a Financing Order, by operation of law or otherwise, and whether incurred by such Borrower or Guarantor as principal, surety, endorser, guarantor or otherwise and including, without limitation, all principal, interest, financing charges, letter of credit fees, unused line fees, servicing fees, line increase fees, debtor-in-possession facility fees, early termination fees, other fees, commissions, costs, expenses and attorneys', accountants' and consultants' fees and expenses incurred in connection with any of the foregoing, and for the avoidance of doubt excluding any Excluded Swap Obligations.

"<u>Pre-Petition Collateral</u>" means, collectively, (a) all "Collateral" as such term is defined in the Existing Credit Agreement as in effect immediately prior to the Petition Date, (b) all "Collateral" and "Pledged Collateral" as such terms are defined in each of the other Existing Loan Documents as in effect

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 6 of 432

immediately prior to the Petition Date, and (c) all other security for the Pre-Petition Obligations as provided in the Existing Credit Agreement and the other Existing Loan Documents immediately prior to the Petition Date.

"Pre-Petition Obligations" means all Obligations arising at any time before the Petition Date.

"Ratification Effective Date" means February 5, 2018.

"<u>Testing Period</u>" means (a) the one (1) calendar week period ending February 9, 2018, (b) the two (2) calendar week period ending February 16, 2018, (c) the three (3) calendar week period ending February 23, 2018, and (d) thereafter, each rolling four (4) calendar week period ending one (1) week after the end of the preceding Testing Period.

"Total Cash Receipts" means the "total cash receipts" detailed in the Approved Budget.

"Total Operating Disbursements" means the aggregate amount of "total operating cash disbursements" as detailed in the Approved Budget. For the avoidance of doubt, Total Operating Disbursements shall include disbursements related to plant and facility closures and consolidations, and shall exclude each of the following: professional fees (including professional fees of the Borrower and its Subsidiaries), expenses, costs and disbursements (including legal fees, expenses, costs and disbursements) of the Loan Parties, any official committee appointed in the Chapter 11 Cases, or any creditor of the Loan Parties in the Chapter 11 Cases or otherwise, interest and financing fees, and the transition services agreement related to the purchase of the Borrower's Quality Park assets by LSC Communications.

"<u>Updated Budget</u>" has the meaning set forth in <u>Section 5.3(b)</u>.

"<u>Variance</u>" means a difference in the amount contained in the Approved Budget with respect to projected Total Cash Receipts and projected Total Operating Disbursements for any Testing Period, in each case compared to the actual Total Cash Receipts and actual Total Operating Disbursements, respectively, for such Testing Period.

"<u>Variance Report</u>" has the meaning set forth in <u>Section 5.3(c)</u>.

"Variance Reporting Delivery Date" has the meaning set forth in Section 5.3(c).

1.3 Interpretation.

(a) All references to the terms "Borrower", "Guarantors" or "Debtors" in any of the Existing Loan Documents shall be deemed and each such reference is hereby amended to mean and include (as applicable) the Debtors, each as defined herein, and their successors and assigns (including any trustee or other fiduciary hereafter appointed as any Debtor's legal representative, as applicable, or with respect to any Debtor the property of the estate of such Debtor whether under Chapter 11 of the Bankruptcy Code or any subsequent Chapter 7 case or cases and its successor upon conclusion of the Chapter 11 Case of such Debtor, as the case may be).

(b) All terms not expressly defined herein which are defined in the Uniform Commercial Code, as in effect in the State of New York as of the date hereof, shall have the meaning set forth therein, except that the term "Lien" or "lien" shall have the meaning set forth in Section 101(37) of the Bankruptcy Code.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 7 of 432

(c) All limits and sublimits set forth in this Agreement and the Credit Agreement, and any formula or other provision to which a limit or sublimit may apply, shall be determined on an aggregate basis considering together both the Pre-Petition Obligations and the Post-Petition Obligations.

2. <u>Amendments</u>. Effective as of the Ratification Effective Date, the Existing Credit Agreement is hereby amended in its entirety to delete the stricken text (indicated textually in the same manner as the following example: <u>stricken text</u>) and to add the bold and double-underlined text (indicated textually in the same manner as the following example: <u>double underlined text</u>) as set forth in <u>Exhibit A</u> hereto. All schedules and exhibits to the Credit Agreement, as in effect immediately prior to the date of this Agreement, shall constitute schedules and exhibits to the Credit Agreement is hereby each replaced in its entirety with the corresponding Schedule attached to <u>Exhibit A</u> hereto. By executing this Agreement, Borrower, Guarantors and Lenders hereby each consents and agrees to the other amendments and modifications to the Credit Agreement contained in <u>Exhibit A</u>.

3. Acknowledgment.

3.1 <u>Pre-Petition Obligations</u>. Borrower and each Guarantor hereby acknowledges, confirms and agrees that, as of January 31, 2018, Borrower is indebted to Administrative Agent and Lenders in respect of all Pre-Petition Obligations in the aggregate principal amount of not less than \$142,183,541.19, consisting of (a) Revolving Loans to Borrower made pursuant to the Existing Loan Documents in the aggregate principal amount of not less than \$125,617,955.47, together with interest accrued and accruing thereon, and (b) Letter of Credit Obligations in the amount of \$16,565,585.72, together with interest accrued and accruing thereon, and all costs, expenses, fees (including attorneys' fees and legal expenses) accrued and owing by Borrower to Administrative Agent and Lenders, without offset, defense or counterclaim of any kind, nature and description whatsoever.

3.2 <u>Guaranteed Obligations</u>. Each Guarantor hereby acknowledges, confirms and agrees that:

(a) all obligations of such Guarantor under the Existing Loan Documents are unconditionally owing by such Guarantor to Administrative Agent and Lenders without offset, defense or counterclaim of any kind, nature and description whatsoever, and

(b) the absolute and unconditional guarantee of the payment of the Pre-Petition Obligations by such Guarantor pursuant to the Existing Loan Documents includes, in addition and not in limitation, all Post-Petition Obligations.

3.3 Acknowledgment of Security Interests. Borrower and each Guarantor hereby acknowledges, confirms and agrees that upon entry of, and subject to, an Interim Financing Order (and when applicable, the Final Financing Order), and subject to the Carve-Out in all respects, Administrative Agent, for the benefit of itself and the other Secured Parties, (a) has and shall continue to have valid, enforceable and perfected first priority and senior security interests in and Liens (subject as to priority to those Permitted Liens that have priority by operation of law and with respect to the claims of Lenders subject to the Carve-Out) upon all Collateral that is Revolver Priority Collateral heretofore granted to Administrative Agent and Lenders pursuant to the Existing Loan Documents as in effect immediately prior to the Petition Date to secure all of the Obligations, (b) has and shall continue to have valid, enforceable and perfected junior priority and subordinate security interests in and Liens, (i) subject as to priority as to those Permitted Liens that have priority by operation of law, (ii) subject to the Carve-Out and (iii) subject to the senior security interests and Liens in favor of the Fixed Asset Facility Collateral Agent upon all Collateral that is Fixed Asset Facility Collateral, and (c) shall have valid and enforceable

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 8 of 432

security interests in and Liens upon all Collateral granted to Administrative Agent, for the benefit of itself and the other Lenders, under the Financing Orders or hereunder or under any of the other Loan Documents or otherwise granted to or held by Administrative Agent and Lenders having the priorities set forth in the Loan Documents and the Financing Orders, as the case may be, and (i) subject as to priority as to those Permitted Liens that have priority by operation of law, (ii) subject to the Carve-Out, (iii) subject to the senior security interests and Liens in favor of the Senior Secured Notes Trustee upon all Collateral that is Fixed Asset Facility Collateral, and (iv) any other Liens expressly permitted by the Financing Orders that under the terms thereof have priority over the security interests and Liens in favor of the Secured Parties. The Secured Parties acknowledge and agree that notwithstanding anything to the contrary in this Agreement, the Credit Agreement or the other Loan Documents, but subject in all respects to the terms and conditions of the Financing Orders, the Carve-Out shall be prior to the claims in the Chapter 11 Cases of the Secured Parties with respect to the Collateral, and to the extent provided in the Financing Orders, the Carve-Out shall not constitute a breach of any representation, warranty or covenant in the Credit Agreement or the other Loan Event of Default thereunder.

3.4 <u>Binding Effect of Documents</u>. Subject to the entry of the Interim Financing Order (and when applicable, the Final Financing Order) and the terms thereof, Borrower and each Guarantor hereby acknowledges, confirms and agrees that: (a) each of the Existing Loan Documents to which it is a party was duly executed and delivered to Administrative Agent and Lenders by such Borrower or Guarantor and each is in full force and effect as of the date hereof, (b) the agreements and obligations of such Borrower or Guarantor enforceable against it in accordance with the terms thereof, and such Borrower or Guarantor has no valid defense, offset or counterclaim to the enforcement of such obligations, and (c) the Secured Parties are and shall be entitled to all of the rights, remedies and benefits provided for in (i) subject to the entry and terms of the Financing Orders, the Loan Documents, and (ii) the Financing Orders.

3.5 <u>Adoption and Ratification for DIP ABL Facility</u>. Borrower and each Guarantor hereby (a) ratifies, restates, assumes, adopts, affirms and confirms all of the terms and conditions of the Existing Loan Documents, as amended and supplemented pursuant hereto and the Financing Orders, and Borrower and each Guarantor agrees to be fully bound, as Debtor and Debtor-in-Possession, by the terms of the Loan Documents to which such Borrower or Guarantor is a party and (b) agrees to pay all of the Pre-Petition Obligations in accordance with the terms of such Existing Loan Documents, as amended by this Agreement, and in accordance with the Financing Orders. All of the Existing Loan Documents are hereby incorporated herein by reference and hereby are and shall be deemed adopted and assumed in full by Borrower and Guarantors, each as Debtor and Debtor-in-Possession, and considered as agreements between Borrower or such Guarantors, on the one hand, and Administrative Agent and Lenders, on the other hand.

4. <u>Grant of Security Interest for DIP ABL Facility</u>. As collateral security for the prompt performance, observance and payment in full of all of the Obligations (including the Pre-Petition Obligations and the Post-Petition Obligations), Borrower and each Guarantor, each as debtor and debtor-in-possession, hereby grants, pledges and assigns to Administrative Agent, for the benefit of itself and the other Secured Parties, and also confirms, reaffirms and restates the prior grant to Administrative Agent and the other Secured Parties of, continuing security interests in and liens upon, and rights of setoff against, all of the Collateral.

5. <u>Additional Representations, Warranties and Covenants for DIP ABL Facility</u>. In addition to the continuing representations, warranties and covenants heretofore and hereafter made by Borrower and Guarantors to Administrative Agent and Lenders, whether pursuant to the Loan Documents or otherwise, and not in limitation thereof, pursuant to the DIP ABL Facility, effective upon the satisfaction of the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 9 of 432

conditions set forth in <u>Section 7</u> for the effectiveness of the DIP ABL Facility, Borrower and each Guarantor hereby represents, warrants and covenants to Administrative Agent and Lenders the following (which shall survive the execution and delivery of this Agreement), the truth and accuracy of which, or compliance with, to the extent such compliance does not violate the terms and provisions of the Bankruptcy Code, shall be a continuing condition of the making of Revolving Loans by Administrative Agent and Lenders:

5.1 <u>Financing Order</u>. The Interim Financing Order (and, following the expiration of the Interim Financing Period defined therein, the Final Financing Order) has been duly entered and (a) has not been vacated, reversed on appeal, or stayed, or modified or amended (other than, in the case of a modification or amendment, as consented to by Administrative Agent), and (b) is not subject to any pending appeal or motion for leave to appeal or other proceeding to set aside any such order or other challenge to the relief provided for in such order, except in each case as consented to by Administrative Agent.

5.2 Use of Proceeds. All Revolving Loans and Letters of Credit provided by Administrative Agent or any Lender to Borrower under the DIP ABL Facility pursuant to the Financing Orders, the Credit Agreement or otherwise, shall be used by Borrower to repay the Obligations under the Existing Credit Agreement in accordance with the Financing Orders, and for working capital and general corporate purposes of Borrower, including: (i) to pay obligations arising from or related to the Carve-Out, (ii) to pay professional fees in connection with the Chapter 11 Cases, (iii) to make adequate protection payments, (iv) to pay fees and expenses incurred in connection with the transactions contemplated by this Agreement and (v) any other allowed administrative expenses incurred during the Chapter 11 Cases, all substantially consistent with the Budget pursuant to Section 5.3 of this Agreement; provided, that, in no event shall the use of such proceeds result in variances from the Budget that would be an Event of Default. Notwithstanding the foregoing, proceeds of the DIP ABL Facility shall not be used by Borrower or Guarantors to affirmatively commence or support, or to pay any professional fees incurred in connection with, any adversary proceeding, motion or other action that seeks to challenge, contest or otherwise seek to impair or object to the validity, extent, enforceability or priority of any liens or other rights under the Existing Credit Agreement or the Credit Agreement as amended by this Agreement. Notwithstanding anything to the contrary herein, the Carve-Out shall be senior to all Liens and claims (including administrative and superpriority claims) securing the Obligations, the Loan Parties' prepetition obligations, adequate protection Liens, and all other Liens or claims (including administrative claims and DIP Superpriority Claims), including all other forms of adequate protection, Liens, or claims (including administrative claims and DIP Superpriority Claims) securing the Obligations and pre-petition obligations granted or recognized as valid, including the Liens, security interests, and claims (including administrative claims and DIP Superpriority Claims) granted to the Administrative Agent and the other Secured Parties.

5.3 Budget.

(a) <u>Initial Budget</u>. The Borrower has prepared and delivered to the Administrative Agent and the Lenders an initial budget, attached as <u>Exhibit B</u> hereto, which reflects the Loan Parties' anticipated cash receipts and anticipated disbursements for each calendar week during the period from the Petition Date through and including the end of the thirteenth (13th) calendar week following the Petition Date (the "<u>Initial Budget</u>"). The Initial Budget (and each subsequent Approved Budget hereunder) shall be deemed the "Approved Budget" for all purposes hereof until superseded by another Approved Budget pursuant to the provisions set forth herein.

(b) <u>Updated Budget</u>. Not later than 5:00 p.m. New York City time on February 23, 2018 and every fourth Friday thereafter (or as more frequently requested by the Administrative Agent) (each, a "<u>Budget Delivery Date</u>"), the Loan Parties shall deliver (email shall suffice) to the Administrative Agent

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 10 of 432

and the Lenders a supplement to the Approved Budget covering the subsequent 13-week period, consistent with the form and level of detail set forth in the Initial Budget and otherwise in form and substance satisfactory to the Administrative Agent in all material respects (each such supplement, solely to the extent approved by the Administrative Agent and the Required Lenders, an "<u>Updated Budget</u>") reflecting the Loan Parties' anticipated cash receipts and anticipated disbursements for each calendar week during the thirteen (13) calendar week period beginning on the Saturday following the applicable Budget Delivery Date, together with a report setting forth the amount by which any line item has changed since the prior Approved Budget. Upon (and subject to) the approval of any such Updated Budget by the Administrative Agent, such Updated Budget shall constitute the then-Approved Budget for purposes hereof; <u>provided</u>, however, that in the event the Administrative Agent provides written notice to the Loan Parties (including, by email) that the proposed Updated Budget is not acceptable, and the Administrative Agent and the Loan Parties are unable to reach agreement regarding such Updated Budget, then the Approved Budget most recently in effect shall remain the Approved Budget.

(c) Variance and Net Operating Cash Flow Reporting. Beginning on February 16, 2018 and every Friday thereafter (each, a "Variance Reporting Delivery Date"), the Loan Parties shall deliver to the Administrative Agent and the Lenders (email shall suffice) a report, in form and substance satisfactory to the Administrative Agent in all material respects (each a "Variance Report"), setting forth (i) actual results for the following items identified in the Approved Budget: (A) Total Cash Receipts, (B) Total Operating Disbursements, (C) "total financing cash receipts/disbursements", (D) "total QP disbursements / adjustments", (E) "total other disbursements / adjustments", and (F) "liquidity" of the Loan Parties for the previous calendar week ended on the immediately preceding Friday, and setting forth all the variances, on a line item and aggregate basis, as compared to the corresponding amounts set forth in the then-current Approved Budget for such week/period, in each case, on a cumulative basis from the beginning of the period covered by the then-current Approved Budget, together with an explanation of (1) in reasonable detail, all material variances from the then current Approved Budget for such period and (2) in the case of a Variance Report immediately following the end of any Testing Period other than a Permitted Variance, the Loan Parties' compliance with the Permitted Variances for such Testing Period or identifying and explaining in reasonable detail any non-compliance by the Loan Parties with the Permitted Variances for such Testing Period, and (ii) the variance, if any, in the amount contained in the Initial 9-Month Budget with respect to projected Cumulative Net Operating Cash Flow compared to the actual Cumulative Net Operating Cash Flow as of the end of the last day covered by the Variance Report. Variances other than any Permitted Variance, if any, from the applicable Approved Budget, and any proposed changes to the applicable Approved Budget, shall be subject to the prior written consent of the Administrative Agent and the Required Lenders.

(d) Each Loan Party hereby confirms, acknowledges and agrees that, unless waived in writing by Administrative Agent and the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), each of the following events shall constitute an Event of Default under the Credit Agreement and the other Loan Documents: (i) any Variance occurs other than a Permitted Variance, (ii) at any time the variance between actual Cumulative Net Operating Cash Flow and projected Cumulative Net Operating Cash Flow as set forth in the Initial 9-Month Budget exceeds \$20,000,000 (adverse to the Borrower), (iii) the failure to deliver any Approved Budget or Variance Report as required by this Section 5.3, and (iv) the failure to comply with any of the Milestones, except to the extent such Milestone is extended to a later date with the prior consent of the Administrative Agent and Required Lenders. Notwithstanding any approval by Administrative Agent or any Lender of any Approved Budget, the Lenders shall only provide Revolving Loans and Letters of Credit in accordance with the terms and conditions set forth in the Existing Credit Agreement as amended by this Agreement, the other Loan Documents and the Financing Order. Administrative Agent and Lenders are relying upon Borrower's delivery of, and compliance with, each Budget in accordance with this <u>Section 5.3</u> in determining to enter into the post-petition financing arrangements provided for herein.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 11 of 432

(e) Notwithstanding any projected amounts set forth in the Budget relating to the costs and expenses of Administrative Agent and the other Secured Parties that are reimbursable by Borrower or any other amounts owing by Borrower to Administrative Agent and the other Secured Parties (including, without limitation, reasonable attorneys and consulting fees and expenses) in accordance with the Credit Agreement and the other Loan Documents, such projections shall not limit, impair, modify or waive the Loan Parties' obligation to pay the actual amounts due to Administrative Agent and/or the other Secured Parties in respect of such costs, expenses and other amounts owing to Administrative Agent and the other Secured Parties, in each case, in accordance with the Credit Agreement and the other Loan Documents.

5.4 <u>Ratification of Deposit Account Control Agreement</u>. To the extent Administrative Agent deems it necessary in its reasonable discretion and upon Administrative Agent's request, Borrower and Guarantors shall use reasonable best efforts to promptly provide Administrative Agent with evidence, in form and substance satisfactory to Administrative Agent, that the Deposit Account Control Agreement (as defined in the Financing Order) and other deposit account arrangements provided for under <u>Section 6.20(c)</u> of the Credit Agreement with banks other than Administrative Agent or an affiliate of Administrative Agent have been ratified and amended by the parties thereto, or their respective successors in interest to reflect the commencement of the Chapter 11 Cases, and including, without limitation, that any reference to a Borrower and Guarantor therein is to such Borrower or Guarantor as a debtor and debtor-in-possession under the Chapter 11 Cases and is in all respects the successor in interest to such Borrower or Guarantor and any references to the Obligations therein include both the Pre-Petition Obligations and any references to the Collateral therein includes both the Pre-Petition Collateral and the Post-Petition Collateral.

5.5 <u>Waiver of Certain Rights</u>. For purposes of the DIP ABL Facility, each Loan Party hereby agrees that it shall not, and hereby waives any right that it may have to, seek authority (a) to use cash collateral of Administrative Agent and Lenders under Section 363 of the Bankruptcy Code, except to the extent provided in any Financing Order, without the prior written consent of Administrative Agent, (b) to obtain post-petition loans or other financial accommodations, other than from Administrative Agent and Lenders not otherwise permitted by the Credit Agreement, as amended by this Agreement, pursuant to Sections 364(c) or (d) of the Bankruptcy Code without the prior written consent of Administrative Agent, (c) to challenge, contest or otherwise seek to impair or object to the validity, extent, enforceability or priority of Administrative Agent's post-petition liens and claims, (d) to challenge the application of any payments or collections received by Administrative Agent or Lenders to the obligations of the Loan Parties as provided for herein, (e) to propose or support a plan of reorganization in the Chapter 11 Cases that does not provide for (or does not provide each Lender at its election to receive) the indefeasible payment in full in cash and satisfaction of all obligations on the effective date of such plan, (f) to surcharge the Collateral pursuant to Section 506(c) of the Bankruptcy Code, or (g) to seek relief under the Bankruptcy Code, including, without limitation, under Section 105 of the Bankruptcy Code, to the extent any such relief would in any way restrict or impair the rights and remedies of Administrative Agent or any Lender as provided herein, in the Loan Documents or the Financing Orders.

5.6 Additional Covenants; Milestones.

(a) Borrowers and Guarantors shall diligently pursue confirmation of a plan of reorganization and shall:

(i) file a plan of reorganization that provides for the indefeasible payment in full in cash of the Obligations on terms and conditions acceptable to the Administrative Agent and the Lenders and related disclosure statement with the Bankruptcy Court on or before April 2, 2018;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 12 of 432

(ii) commence a hearing to confirm the Plan of Reorganization ("<u>Plan</u>"), which shall be in form and substance satisfactory to the Administrative Agent and Lenders, no later than July 2, 2018;

(iii) cause the "effective date" of the Plan to occur no later than July 20, 2018; and

(iv) the Administrative Agent may agree to extend any of the dates set forth above by up to sixty (60) days, with the approval of Required Lenders.

(b) Borrowers and Guarantors shall cause all proposed "First Day" and "Second Day" orders and all other orders establishing procedures for administration of the Chapter 11 Cases (including, without limitation, any orders authorizing KEIP/KERP) or approving significant transactions submitted to the Bankruptcy Court to be in accordance with and permitted by the terms of this Agreement and reasonably acceptable to the Administrative Agent in all respects, it being understood and agreed that the forms of orders approved by the Administrative Agent prior to the Petition Date are in accordance with and permitted by the terms of this Agreement in all respects, and Borrowers and Guarantors shall provide the Administrative Agent with not less than two (2) Business Days' notice (or such lesser time as may be acceptable to the Administrative Agent) and the opportunity to review and reasonably comment on the form of order.

(c) Within one (1) Business Day after the Final Financing Order, Borrower shall have received not less than \$50,000,000 (net of fees and expenses set forth in the Approved Budget) in proceeds of loans under the DIP Term Credit Agreement (which, for the avoidance of doubt, shall be in addition to the amount set forth in <u>Section 6.8</u>), and the aggregate amount of such net proceeds shall be paid to the Administrative Agent on such date to repay the Revolving Loans under the Credit Agreement.

5.7 <u>Letters of Credit</u>. All Letters of Credit on account of Borrower or Guarantors issued and outstanding immediately prior to the Ratification Effective Date shall be deemed to be Letters of Credit under the Credit Agreement (as amended on the Ratification Effective Date) for all purposes of the Credit Agreement and the other Loan Documents.

6. <u>Conditions Precedent to Ratification Effective Date</u>. The effectiveness of the amendments, representations and warranties, covenants, defaults and all other agreements and obligations contained in this Agreement is subject to the satisfaction or waiver of each of the following conditions precedent:

6.1 The Loan Parties shall have commenced the Chapter 11 Cases in the Bankruptcy Court by no later than February 5, 2018 and the Loan Parties shall have complied in full with the notice and other requirements of the Bankruptcy Code in a manner acceptable to Administrative Agent and its counsel, with respect to the Interim Financing Order and Administrative Agent shall have received such evidence thereof as it shall reasonably require. No trustee, or other disinterested person with expanded powers pursuant to Section 1104(c) of the Bankruptcy Code, shall have been appointed or designated with respect to any Loan Party or their respective business, properties or assets and no motion shall be pending seeking any such relief. All of the "first day orders" and "second day orders" intended to be entered by the Bankruptcy Court at the time of the commencement of the Chapter 11 Cases and prior to the Interim Financing Order shall be in form and substance satisfactory to Administrative Agent, and Borrowers and Guarantors shall have provided to the Administrative Agent not less than two (2) Business Days' notice (or such lesser time as may be acceptable to the Administrative Agent) to review and reasonably comment on the form of the order. A cash management order approving the cash management arrangements of Loan Parties consistent with the requirements under the DIP ABL Facility, and otherwise in form and substance reasonably satisfactory to Administrative Agent, the bank product providers under the Existing Credit Agreement and the DIP ABL Facility, shall have been entered and shall be in full force and effect,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 13 of 432

and shall reflect that each Debtor, as debtor and debtor-in-possession, is the successor in interest to such Borrower or Guarantor, as the case may be, that the Obligations include both the Pre-Petition Obligations and the Post-Petition Obligations, and that the Collateral includes both the Pre-Petition Collateral and the Post-Petition Collateral as provided for in this Ratification Agreement.

6.2 The Interim Financing Order shall have been entered by the Bankruptcy Court authorizing the secured financing under the DIP ABL Facility on the terms and conditions contemplated by this Agreement and otherwise on terms acceptable to Administrative Agent and Lenders, and, inter alia, modifying the automatic stay, authorizing and granting the security interests and liens described above, and granting a super-priority administrative expense claim to Administrative Agent and Lenders with respect to all obligations to Administrative Agent and Lenders, subject to no priority claim or administrative expenses of the Chapter 11 Cases (but subject to the Carve-Out), and any future proceeding which may develop out of any such cases, including liquidation in bankruptcy, shall have been entered within five (5) Business Days after the commencement of the Chapter 11 Cases and be in full force and effect and not have been vacated, reversed, or stayed or modified or amended (except in the case of a modification or amendment as consented to by Administrative Agent) and not be subject to a stay pending appeal or motion for leave to appeal or other proceeding to set aside any such order or the challenge to the relief provided for in it, except as consented to by Administrative Agent.

6.3 Administrative Agent shall have received, each in form and substance reasonably satisfactory to Administrative Agent, of (i) the Initial Budget, (ii) the Initial 9-Month Budget, and (iii) projected monthly balance sheets, income statements, statements of cash flows and availability of the Loan Parties for the period through the end of the term of the DIP ABL Facility, in each case as to the projections, with the results and assumptions set forth in all of such projections in form and substance reasonably satisfactory to Administrative Agent, and an opening pro forma balance sheet for the Loan Parties in form and substance reasonably satisfactory to Administrative Agent.

6.4 There shall be no Default or Event of Default under the Existing Credit Agreement other than under Section 8.01(e) as to other Indebtedness as a result of the commencement of the Chapter 11 Cases, 8.01(f), and 8.01(g) of the Existing Credit Agreement in each case due to the commencement of the Chapter 11 Cases (it being agreed that the Loan Documents shall provide that all such Defaults and Events of Default are waived on and after the Ratification Effective Date and no default rate interest shall accrue with respect thereto).

6.5 Execution and delivery of all Loan Documents by the parties thereto. Administrative Agent, for the benefit of itself, the Lenders and the other Secured Parties, shall hold perfected, first priority security interests in and liens upon the Revolver Priority Collateral and perfected, junior priority security interests in and liens upon the Fixed Asset Facility Collateral (subordinate only to the security interests and liens that secure the Dip Term Documents and the valid, non-avoidable prepetition perfected security interests and liens that secure the First Lien Notes), and in each case subject to the Carve-Out and other valid, non-avoidable prepetition perfected security interests and liens priority by operation of law, and Administrative Agent shall have received such evidence of the foregoing as it reasonably requires.

6.6 Administrative Agent shall have received at least five (5) Business Days prior to the Ratification Effective Date all documentation and information as is reasonably requested by Administrative Agent that is required by regulatory authorities under applicable "know your customer" and anti-money-laundering rules and regulations in applicable jurisdictions, including, without limitation, the PATRIOT Act, in each case to the extent requested in writing at least seven (7) Business Days prior to the Ratification Effective Date, <u>provided</u>, that, the Loan Parties will use reasonable efforts to promptly provide any additional information requested thereafter. Administrative Agent and each Lender shall

have received, in form and substance satisfactory to Administrative Agent and each Lender, flood certificates and other insurance and related materials as to any real property subject to a mortgage or lien as is required in accordance with statutes and regulations applicable to such Lender and the applicable policies of such Lender.

6.7 The sum of the opening Availability plus the amount of Qualified Cash at closing after the application of proceeds of the initial Revolving Loans, if any, on such date and after provision for payment of all fees and expenses of the transaction paid or payable on or about the Ratification Effective Date, shall be not less than \$30,000,000.

6.8 Upon confirmation of the Interim Financing Order, Administrative Agent shall have received evidence that Borrower has entered into the DIP Term Credit Agreement, which shall be on terms and conditions reasonably satisfactory to Administrative Agent and Lenders, and Borrower shall have received not less than \$50,000,000 in proceeds of loans under the DIP Term Credit Agreement (net of fees and expenses as set forth in the Approved Budget), and the aggregate amount of such net proceeds shall be paid to the Administrative Agent on the Ratification Effective Date to repay the Revolving Loans under the Credit Agreement.

6.9 No material adverse change in the business, operations, profits, assets or financial condition of Borrower and Guarantors shall have occurred since December 31, 2016 (it being understood that the commencement of the Chapter 11 Cases, the events that customarily occur following the commencement of a proceeding under Chapter 11 of the Bankruptcy Code, any defaults under agreements that have no material effect on the debtors under the terms of the Bankruptcy Code as a result of the commencement of the Chapter 11 Cases, reduction in payment terms by suppliers, and reclamation claims shall not be deemed a material adverse change). No Default or Events of Default shall exist as of the Ratification Effective Date under any of the Loan Documents, as amended by this Agreement. Administrative Agent and Lenders shall have received the payment of all fees required to be paid on the Ratification Effective Date under the terms hereof or otherwise under the Loan Documents (including, without limitation, the Fee Letters). Administrative Agent and Lenders shall have received payment of all reasonable and documented out-of-pocket costs and expenses that have been invoiced prior to the Ratification Effective Date (including, without limitation, the reasonable and documented fees and expenses of counsel for Administrative Agent) incurred for the period through and including the Ratification Effective Date.

6.10 There shall be no material misstatements in (or omissions of material facts necessary in order to make the statements contained in this Agreement, taken as a whole, not materially misleading in light of the circumstances under which such statements are made) the materials previously furnished to Administrative Agent by the Loan Parties. Administrative Agent shall not have become aware of any material information or other matter that is inconsistent in a material and adverse manner with any previously provided due diligence, information or materials (including any financial information).

6.11 Administrative Agent and the Lenders shall have received, in form and substance reasonably satisfactory to Administrative Agent, a five-year exit plan of the Loan Parties.

7. <u>Additional Conditions Precedent to the Revolving Loans Under the DIP ABL Facility</u>. In addition to the satisfaction of the conditions precedent under <u>Section 6</u> of this Agreement with respect to the effectiveness of the noted Sections of this Agreement and the making of the initial Revolving Loans under the DIP ABL Facility, and the satisfaction of the conditions precedent in <u>Section 4.02</u> of the Credit Agreement with respect to all Loans and Letters of Credit available to Borrower (such conditions in <u>Section 4.02</u>, except as modified or made pursuant to this Agreement shall remain applicable to the Revolving Loans, Swingline Loans and Letters of Credit), the following are conditions to Administrative

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 15 of 432

Agent's and Lenders' obligation to extend further Loans or Letters of Credit to Borrower pursuant to the Credit Agreement:

7.1 With respect to further Loans or Letters of Credit after expiration of the Interim Financing Order, on or before the expiration of the Interim Financing Order, the Bankruptcy Court shall have entered a Final Financing Order. Neither Administrative Agent nor any Lender shall provide any Revolving Loans, Swingline Loans or Letters of Credit other than those authorized under the Interim Financing Order unless, on or before the fortieth (40th) day following the date of the commencement of the Chapter 11 Cases (or such later date as Administrative Agent and the Required Lenders may agree), the Final Financing Order shall have been entered, and shall not have been reversed, modified, amended, stayed or vacated (except in the case of any modification or amendment as consented to by Administrative Agent) and shall not be subject to a stay pending appeal, and there shall be no appeal or other contest with respect to any material matter in such order (except as Administrative Agent and the Required Lenders may otherwise agree based on its review of such appeal or other contest).

8. <u>Release</u>.

8.1 Release of Pre-Petition Claims.

(a) Upon the earlier of (i) the entry of the Final Financing Order or (ii) the entry of an Order extending the term of the Interim Financing Order beyond thirty (30) calendar days after the date of the Interim Financing Order, in consideration of the agreements of Administrative Agent and Lenders contained herein and the making of any Revolving Loans by Administrative Agent and Lenders, each Loan Party, pursuant to the Credit Agreement, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, on behalf of itself and its respective successors, assigns, and other legal representatives, hereby absolutely, unconditionally and irrevocably releases, remises and forever discharges Administrative Agent, each Lender and their respective successors and assigns, and their present and former shareholders, affiliates, subsidiaries, divisions, predecessors, directors, officers, attorneys, employees and other representatives (Administrative Agent, each Lender and all such other parties being hereinafter referred to collectively as the "Releasees" and individually as a "Releasee"), of and from all demands, actions, causes of action, suits, covenants, contracts, controversies, agreements, promises, sums of money, accounts, bills, reckonings, damages and any and all other claims, counterclaims, defenses, rights of set-off, demands and liabilities whatsoever, except, for the avoidance of doubt, excluding any claims or causes of action for gross negligence or willful misconduct (individually, a "Pre-Petition Released Claim" and collectively, "Pre-Petition Released Claims") of every name and nature, known or unknown, suspected or unsuspected, both at law and in equity, which any Borrower or Guarantor, or any of their respective successors, assigns, or other legal representatives may now or hereafter own, hold, have or claim to have against the Releasees or any of them for, upon, or by reason of any nature, cause or thing whatsoever which arises at any time on or prior to the day and date of this Agreement, including, without limitation, for or on account of, or in relation to, or in any way in connection with the Credit Agreement, as amended and supplemented through the date hereof, and the other Loan Documents.

(b) Upon the entry of the Final Financing Order, each Loan Party, on behalf of itself and its successors, assigns, and other legal representatives, as Debtors, but not on behalf of the bankruptcy estate of any of the Loan Parties, hereby absolutely, unconditionally and irrevocably, covenants and agrees with each Releasee that it will not sue (at law, in equity, in any regulatory proceeding or otherwise) any Releasee on the basis of any Pre-Petition Released Claim released, remised and discharged by Borrower and each Guarantor pursuant to this <u>Section 8.1</u>. If any Loan Party violates the foregoing covenant, the Loan Parties agree to pay, in addition to such other damages as any Releasee may sustain as a result of such violation, all attorneys' fees and costs incurred by any Releasee as a result of such violation.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 16 of 432

8.2 <u>Release of Post-Petition Claims</u>. Upon (a) the receipt by Administrative Agent, on behalf of itself and the other Lenders, of payment in full of all Obligations (other than contingent and indemnification obligations for which no claim has been asserted) in cash or other immediately available funds, plus cash collateral or other collateral security acceptable to Administrative Agent to secure any Obligations that survive or continue beyond the termination of the Loan Documents, and (b) the termination of the Loan Documents (the "Payment Date"), in consideration of the agreements of Administrative Agent and Lenders contained herein and the making of any Revolving Loans by Administrative Agent and Lenders, each Loan Party hereby covenants and agrees to execute and deliver in favor of Administrative Agent and Lenders a valid and binding termination and release agreement, in form and substance reasonably satisfactory to Administrative Agent. If any Loan Party violates such covenant, the Loan Parties agree to pay, in addition to such other damages as any Releasee may sustain as a result of such violation, all attorneys' fees and costs incurred by any Releasee as a result of such violation.

8.3 Releases Generally.

(a) Each Loan Party understands, acknowledges and agrees that the releases set forth above in <u>Sections 8.1</u> and <u>8.2</u> hereof may be pleaded as a full and complete defense and may be used as a basis for an injunction against any action, suit or other proceeding which may be instituted, prosecuted or attempted in breach of the provisions of such releases.

(b) Each Loan Party agrees that no fact, event, circumstance, evidence or transaction which could now be asserted or which may hereafter be discovered shall affect in any manner the final and unconditional nature of the releases set forth in <u>Section 8.1</u> hereof and, when made, <u>Section 8.2</u> hereof.

9. Miscellaneous.

9.1 <u>Amendments and Waivers</u>. Neither this Agreement nor any other instrument or document referred to herein or therein may be changed, waived, discharged or terminated orally, but only by an instrument in writing executed in accordance with the requirements of <u>Section 11.01</u> of the Existing Credit Agreement.

9.2 Further Assurances. Borrower and each Guarantor shall, at its expense, at any time or times duly execute and deliver, or shall use its commercially reasonable efforts to cause to be duly executed and delivered, such further agreements, instruments and documents, including, without limitation, additional security agreements, collateral assignments, UCC financing statements or amendments or continuations thereof, landlord's or mortgagee's waivers of liens and consents to the exercise by Administrative Agent and Lenders of all the rights and remedies hereunder, under any of the other Loan Documents, any Financing Order or applicable law with respect to the Collateral, and do or use its commercially reasonable efforts to cause to be done such further acts as may be reasonably necessary or proper in Administrative Agent's reasonable opinion to evidence, perfect, maintain and enforce the security interests of Administrative Agent and Lenders, and the priority thereof, in the Collateral and to otherwise effectuate the provisions or purposes of this Agreement, any of the other Loan Documents or the Financing Order. Upon the request of Administrative Agent, at any time and from time to time, Borrower and each Guarantor shall, at its cost and expense, do, make, execute, deliver and record, register or file updates to the filings of Administrative Agent and Lenders with respect to the Intellectual Property with the United States Patent and Trademark Office, the financing statements, mortgages, deeds of trust, deeds to secure debt, and other instruments, acts, pledges, assignments and transfers (or use its best efforts to cause the same to be done) and will deliver to Administrative Agent and Lenders such instruments evidencing items of Collateral as may be available and reasonably requested by Administrative Agent.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 17 of 432

9.3 <u>Headings</u>. The headings used herein are for convenience only and do not constitute matters to be considered in interpreting this Agreement.

9.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all of which shall together constitute one and the same agreement. In making proof of this Agreement, it shall not be necessary to produce or account for more than one counterpart thereof signed by each of the parties hereto. Delivery of an executed counterpart of this Agreement by telefacsimile or other means of electronic transmission shall have the same force and effect as delivery of an original executed counterpart of this Agreement. Any party delivering an executed counterpart of this Agreement by telefacsimile or other means of electronic transmission also shall deliver an original executed counterpart of this Agreement, but the failure to deliver an original executed counterpart of this Agreement, but the failure to deliver an original executed counterpart of this Agreement as to such party or any other party.

9.5 <u>Additional Events of Default</u>. The parties hereto acknowledge, confirm and agree that the failure of any Borrower or Guarantor to comply with any of the covenants, conditions and agreements by such Loan Party contained herein shall constitute an Event of Default under the Loan Documents.

9.6 Costs and Expenses. Subject to the terms of the Financing Orders, Borrower shall pay to Administrative Agent on demand all costs and expenses that Administrative Agent pays or incurs in connection with the negotiation, preparation, consummation, administration, enforcement, and termination of this Agreement and the other Loan Documents and the Financing Orders, including, without limitation: (a) reasonable attorneys' and paralegals' fees and disbursements of counsel to, and reasonable fees and expenses of consultants, accountants and other professionals retained by, Administrative Agent; (b) costs and expenses (including reasonable attorneys' and paralegals' fees and disbursements) for any amendment, supplement, waiver, consent, or subsequent closing in connection with this Agreement, the other Loan Documents, the Financing Orders and the transactions contemplated thereby; (c) taxes, fees and other charges for recording any agreements or documents with any governmental authority, and the filing of UCC financing statements and continuations, and other actions to perfect, protect, and continue the security interests and liens of Administrative Agent and Lenders in the Collateral; (d) sums paid or incurred to pay any amount or take any action required of Borrower and Guarantors under the Loan Documents or the Financing Orders that Borrower and Guarantors fail to pay or take; (e) costs of appraisals, inspections and verifications of the Collateral and including travel, lodging, and meals for inspections of the Collateral and the Debtors' operations by Administrative Agent or its agent and to attend court hearings or otherwise in connection with the Chapter 11 Cases; (f) costs and expenses of preserving and protecting the Collateral; (g) all out-of-pocket expenses and costs heretofore and from time to time hereafter incurred by Administrative Agent during the course of periodic field examinations of the Collateral and Debtors' operations, plus a fee of \$1,050 per day, per auditor, plus reasonable out-of-pocket expenses for each field examination of the Loan Parties performed by personnel employed by Administrative Agent; (h) if implemented, a fee of \$1,050 per day, per applicable individual, plus reasonable out-of-pocket expenses for the establishment of electronic collateral reporting; (i) in accordance with the terms of this Agreement, the actual charges paid or incurred by Administrative Agent if it elects to employ the services of one or more third persons as a consultant or financial advisor during the Chapter 11 Cases; (j) the actual charges paid or incurred by Administrative Agent if it elects to employ the services of one or more third-persons to perform field examinations, to appraise the Collateral, or any portion thereof, or to assess the Loan Parties' business valuation; and (k) costs and expenses (including attorneys' and paralegals' fees and disbursements) paid or incurred to obtain payment of the Obligations, enforce the security interests and liens of Administrative Agent and Lenders, sell or otherwise realize upon the Collateral, and otherwise enforce the provisions of this Agreement, the other Loan Documents and the Financing Orders, or to defend any claims made or threatened against Administrative Agent or any Lender arising out of the transactions contemplated hereby (including,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 18 of 432

without limitation, preparations for and consultations concerning any such matters). The foregoing shall not be construed to limit any other provisions of the Loan Documents regarding costs and expenses to be paid by Borrower. Subject to the terms of the Financing Order(s), all sums provided for in this <u>Section</u> <u>9.6</u> shall be part of the Obligations, shall be payable on demand, and shall accrue interest after demand for payment thereof at the highest rate of interest then payable under the Loan Documents. Administrative Agent is hereby irrevocably authorized to charge any amounts payable hereunder directly to any of the account(s) maintained by Administrative Agent with respect to any Borrower or Guarantor.

9.7 <u>Effectiveness</u>. This Agreement (except those Sections hereof that by the express terms thereof only become effective upon satisfaction or waiver of the conditions set forth in <u>Section 6</u>) shall become effective upon the execution hereof by Administrative Agent and Lenders and Loan Parties; <u>provided</u>, that if the Ratification Effective Date shall not occur by the close of business on the date that is forty (40) days following the Petition Date, the obligations of all parties under this Agreement shall terminate.

9.8 <u>Financing Orders</u>. In the event of any inconsistency between the terms of the Financing Orders and the Loan Documents, the terms of the Financing Orders shall control.

9.9 <u>Conflicts</u>. In the event of any conflict between the terms of this Agreement, on the one hand and the Credit Agreement and the other Loan Documents, on the other, or in the event a matter is addressed in this Agreement but not in the Credit Agreement or the other Loan Documents, the terms of this Agreement shall govern.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 19 of 432

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

AGENT

BANK OF AMERICA, N.A., as Administrative Agent and a Lender

By: Robert G Mahang Name: Robert Q. Mohang Title: S. Vin President

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 20 of 432

BORROWER:

CENVEO CORPORATION < FR. By Name: Robert G. Burton, Sr. Title: Chief Executive Officer

GUARANTORS:

CENVEO, INC. CNMW INVESTMENTS, INC. CENVEO SERVICES, LLC DISCOUNT LABELS, LLC COLORHOUSE CHINA, INC. RX JV HOLDING, INC. CRX JV, LLC CRX HOLDING, INC. RX TECHNOLOGY CORP. CADMUS PRINTING GROUP, INC. CADMUS FINANCIAL DISTRIBUTION, INC. GARAMOND/PRIDEMARK PRESS, INC. CADMUS JOURNAL SERVICES, INC. CADMUS DELAWARE, INC. CADMUS UK, INC. EXPERT GRAPHICS, INC. CADMUS MARKETING GROUP, INC. CADMUS MARKETING, INC. CADMUS/O'KEEFE MARKETING, INC. OLD TSI, INC. PORT CITY PRESS, INC. CADMUS INTERNATIONAL HOLDINGS, INC. MADISON/GRAHAM COLORGRAPHICS, INC. VSUB HOLDING COMPANY VAUGHAN PRINTERS INCORPORATED MADISON/GRAHAM COLORGRAPHICS INTERSTATE SERVICES, INC. COMMERCIAL ENVELOPE MANUFACTURING CO. INC. CENVEO CEM, INC. CENVEO CEM. LLC LIGHTNING LABELS, LLC NASHUA CORPORATION NASHUA INTERNATIONAL, INC. ENVELOPE PRODUCT GROUP, LLC B Name: Robert G. Burton, Sr. Title: Chief Executive Officer CDMS MANAGEMENT, LLC **CENVEO OMEMEE, LLC** R. By

Name: Robert G. Burton, Sr. Title: Chairman

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 22 of 432

LENDER

JPMORGAN CHASE BANK, N.A., as a Lender

By: Donna Di Tores

Name:Donna DiForioTitle:Authorized Officer

LENDER

BARCLAYS BANK PLC, as a Lender By Name: Jøe Jordan

Title: Managing Director

LENDER

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By:_

Name: Mark Bradford Title: Senior Vice President

LENDER

PNC BANK, NATIONAL ASSOCIATION, as a Lender

N By:__

Name: Matthew Leighton Title: Vice President 18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 26 of 432

EXHIBIT A to <u>RATIFICATION AND AMENDMENT AGREEMENT</u>

Credit Agreement

Exhibit A to Ratification and Amendment Agreement

CREDIT AGREEMENT

Dated as of April 16, 2013

among

CENVEO CORPORATION, as a debtor and debtor-in-possession under Chapter 11 of the Bankruptcy Code, as Borrower,

CENVEO, INC.,

as a debtor and debtor-in-possession under Chapter 11 of the Bankruptcy Code, as a Guarantor,

BANK OF AMERICA, N.A., as Administrative Agent, an Issuing Bank and Swingline Lender

BARCLAYS BANK PLC,

GENERAL ELECTRIC CAPITAL CORPORATION and IPMORGAN CHASE BANK, N.A., and WELLS FARGO BANK, NATIONAL ASSOCIATION,-

as Co-Syndication Agents,

PNC and

the other lenders party hereto

BANK OF AMERICA, N.A., JPMORGAN CHASE BANK, N.A., and WELLS FARGO BANK, NATIONAL ASSOCIATION as Documentation Agent,

and

the other lenders party hereto

BANK OF AMERICA, N.A., BARCLAYS BANK PLC,

GENERAL ELECTRIC CAPITAL CORPORATION and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Joint Lead Arrangers and Joint Book Managers

Cahill Gordon & Reindel LLP 80 Pine Street New York, NY 10005 (212) 701-3000

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 28 of 432

TABLE OF CONTENTS

Page

ARTICLE I

DEFINITIONS AND ACCOUNTING TERMS 1

1.01	Defined Terms.	1
1.02	Other Interpretive Provisions.	36<u>32</u>
1.03	Accounting Terms.	37<u>33</u>
1.04	Rounding.	<u>3834</u>
1.05	Times of Day.	39<u>34</u>
1.06	Uniform Commercial Code and PPSA.	39<u>34</u>
1.07	Currency Equivalents Generally.	39<u>34</u>
1.08	Currency Fluctuations.	39<u>34</u>
1.09	Interpretation (Quebec).	39<u>35</u>

ARTICLE II

AMOUNT AND TERMS OF CREDIT-40_35

2.01	Commitments.	<u>4035</u>
2.02	Revolving Loans.	40 <u>35</u>
2.03	Borrowing Procedure.	41 <u>37</u>
2.04	Evidence of Debt; Repayment of Revolving Loans.	4 <u>237</u>
2.05	Fees.	4 <u>338</u>
2.06	Interest on Revolving Loans.	4 <u>339</u>
2.07	Termination and Reduction of Commitments.	44 <u>39</u>
2.08	Interest Elections.	44 <u>40</u>
2.09	Optional and Mandatory Prepayments of Revolving Loans.	4 <u>541</u>
2.10	Payments Generally; Pro Rata Treatment; Sharing of Set-offs.	47 <u>42</u>
2.11	Defaulting Lenders.	48 <u>43</u>
2.12	Swingline Loans.	49 <u>44</u>
2.13	Letters of Credit.	<u>5045</u>
2.14	Settlement Amongst Lenders.	55<u>50</u>
2.15	Revolving Commitment Increase.	56[Reserved]51
2.16	[Reserved]	<u>5751</u>
2.17	Overadvances.	<u>5751</u>
2.18	Protective Advances.	57<u>51</u>
2.19	-Extended Loans.	58

ARTICLE III

TAXES, YIELD PROTECTION AND ILLEGALITY-60_51

3.01	Taxes.	<u>60<u>51</u></u>
3.02	Illegality.	<u>6254</u>
3.03	Inability to Determine Rates.	<u>6254</u>
3.04	Increased Costs; Reserves on Eurodollar Rate Loans.	62<u>54</u>
3.05	Compensation for Losses.	<u>64<u>56</u></u>
3.06	Mitigation Obligations; Replacement of Lenders.	<u>64<u>56</u></u>
3.07	Survival	65<u>56</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 29 of 432

ARTICLE IV

CONDITIONS PRECEDENT TO BORROWINGS-65_57

4.01	Conditions of Initial Borrowing.	65<u>57</u>
4.02	Conditions to All Credit Extensions.	<u>6860</u>

ARTICLE V

REPRESENTATIONS AND WARRANTIES 69 61

5.01	Existence, Qualification and Power.	69<u>61</u>
5.02	Authorization; No Contravention.	<mark>69<u>61</u></mark>
5.03	Governmental Authorization; Other Consents.	<mark>69<u>61</u></mark>
5.04	Binding Effect.	70<u>61</u>
5.05	Financial Statements; No Material Adverse Effect; No Internal Control Event.	70<u>62</u>
5.06	Litigation.	71<u>62</u>
5.07	No Default.	71<u>62</u>
5.08	Ownership of Property; Liens; Investments.	71<u>63</u>
5.09	Environmental Compliance.	71<u>63</u>
5.10	Insurance.	72<u>64</u>
5.11	Taxes.	72<u>64</u>
5.12	ERISA Compliance.	73<u>64</u>
5.13	Canadian Benefit Plans; Canadian Pension Plans.	73<u>65</u>
5.14	Subsidiaries; Equity Interests; Loan Parties.	73<u>65</u>
5.15	Margin Regulations; Investment Company Act.	74 <u>65</u>
5.16	Disclosure.	7466
5.17	Compliance with Laws.	74<u>66</u>
5.18	Intellectual Property; Licenses, Etc.	74 <u>66</u>
5.19	Solvency Reserved.	7566
5.20	Casualty, Etc	75<u>66</u>
5.21	Labor Matters.	75<u>66</u>
5.22	Collateral Documents.	7567
5.23	Anti-Terrorism Laws.	75<u>67</u>

ARTICLE VI

AFFIRMATIVE COVENANTS-76_67

6.01	Financial Statements.	76<u>67</u>
6.02	Certificates; Other Information.	77 <u>68</u>
6.03	Notices.	79<u>70</u>
6.04	Payment of Obligations.	79<u>71</u>
6.05	Preservation of Existence, Etc.	
6.06	Maintenance of Properties.	<u>8071</u>
6.07	Maintenance of Insurance.	<u>8071</u>
6.08	Compliance with Laws.	<u>8072</u>
6.09	Books and Records.	80 <u>72</u>
6.10	Inspection Rights.	80 <u>72</u>
6.11	Use of Proceeds.	<u>8173</u>
6.12	Covenant to Guarantee Obligations and Give Security.	<u>8173</u>
6.13	Compliance with Environmental Laws.	
6.14	Preparation of Environmental Reports.	<u>8476</u>
6.15	Further Assurances.	<u>8576</u>
6.16	Compliance with Terms of Leaseholds.	85 77

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 30 of 432

6.17	[Reserved] .	85<u>77</u>
6.18	Material Contracts.	<u>8577</u>
6.19	[Reserved] .	<u>8577</u>
6.20	Collateral Monitoring and Reporting.	<u>8677</u>
6.21	Post-Closing Actions Related to Real Estate Collateral.	<u>8879</u>

ARTICLE VII

NEGATIVE COVENANTS-89_81

7.01	Liens.	<u>8981</u>
7.02	Indebtedness.	<u>9183</u>
7.03	Investments.	94 <u>84</u>
7.04	Fundamental Changes.	95<u>85</u>
7.05	Dispositions.	96<u>85</u>
7.06	Restricted Payments.	97 85
7.07	Change in Nature of Business.	<mark>98</mark> 86
7.08	Transactions with Affiliates.	<mark>98</mark> 86
7.09	Burdensome Agreements.	<mark>98</mark> 86
7.10	Use of Proceeds.	99<u>87</u>
7.11	Financial Covenant.	99[Reserved].87
7.12	[Reserved].	<u></u>
7.13	Amendments of Organization Documents.	99<u>87</u>
7.14	Accounting Changes.	99<u>87</u>
7.15	Prepayments, Etc. of Indebtedness.	<u>9987</u>
7.16	Amendment, Etc. of Related Documents and Indebtedness.	<u>+0187</u>
7.17	Holding Company.	<u>10188</u>
7.18	Sanctions.	<u>10188</u>

ARTICLE VIII

EVENTS OF DEFAULT AND REMEDIES-101_88

8.01	Events of Default.	101<u>88</u>
8.02	Remedies upon Event of Default.	104<u>91</u>
8.03	Application of Funds.	105<u>92</u>

ARTICLE IX

ADMINISTRATIVE AGENT-106_93

9.01	Appointment and Authority.	106<u>93</u>
9.02	Rights as a Lender.	<u>10794</u>
9.03	Exculpatory Provisions.	108 <u>94</u>
9.04	Reliance by Administrative Agent.	108 95
9.05	Delegation of Duties.	109 95
9.06	Resignation of Administrative Agent.	109<u>96</u>
9.07	Non-Reliance on Administrative Agent and Other Lenders.	109<u>96</u>
9.08	No Other Duties, Etc.	110 <u>96</u>
9.09	Administrative Agent May File Proofs of Claim.	110<u>97</u>
9.10	Collateral and Guaranty Matters.	110 97
9.11	Bank Product Providers.	111<u>98</u>
9.12	Tax Indemnity.	<u>11198</u>

ARTICLE X

CONTINUING GUARANTY-112_98

10.01	Guaranty.	<u>11298</u>
10.02	Rights of Lenders.	112<u>99</u>
10.03	Certain Waivers.	112<u>99</u>
10.04	Obligations Independent.	113 99
10.05	Subrogation.	113<u>99</u>
10.06	Termination; Reinstatement.	<u>113100</u>
10.07	Subordination.	113 100
10.08	Stay of Acceleration.	<u>113100</u>
10.09	Condition of Borrower.	<u>113100</u>
10.10	Additional Guarantor Waivers and Agreements.	<u>114100</u>
10.11	Keepwell.	114<u>101</u>

ARTICLE XI

MISCELLANEOUS-115_101

11.01	Amendments, Etc.	115 101
11.02	Notices; Effectiveness; Electronic Communications.	116<u>103</u>
11.03	No Waiver; Cumulative Remedies.	$\frac{117104}{117104}$
11.04	Expenses; Indemnity; Damage Waiver.	118<u>105</u>
11.05	Payments Set Aside.	<u>119106</u>
11.06	Successors and Assigns.	$\frac{120106}{120106}$
11.07	Treatment of Certain Information; Confidentiality.	<u>123110</u>
11.08	Right of Setoff.	124<u>110</u>
11.09	Interest Rate Limitation.	<u>+24111</u>
11.10	Counterparts; Effectiveness.	124 111
11.11	Survival of Representations and Warranties.	<u>+24111</u>
11.12	Severability.	123<u>111</u>
11.13	Replacement of Lenders.	+ 123 112
11.14	Governing Law; Jurisdiction; Etc.	<u> 126113</u>
11.15	Waiver of Jury Trial.	126<u>113</u>
11.16	No Advisory or Fiduciary Responsibility.	<u>127<u>113</u></u>
11.17	Judgment Currency.	<u> 127114</u>
11.18	USA PATRIOT Act Notice.	$\frac{128}{128}$
11.19	Canadian Anti-Money Laundering Legislation.	<u>+128_115</u>
11.20	ENTIRE AGREEMENT.	<u> 128115</u>
11.21	Second Lien Indenture	128[Reserved]115
11.22	Subject to Intercreditor Agreements Acknowledgement and Consent to Bail-In of EEA Financial Institutions	<u> 128115</u>
11.23	Acknowledgement and Consent to Bail-In of EEA Financial Institutions.	129<u>116</u>
<u>11.24</u>	Conflict with Ratification Agreement	116

SCHEDULES

1.02		Material Contracts
	-	
1.03	-	Consolidated Adjusted EBITDA
1.04	-	icouro i icoluuis
1.05	-	Accounts
2.01	-	Revolving Commitments and Applicable Percentages
2.13	-	Existing Letters or Credit
5.08(b)	-	Existing Liens
5.08(c)	-	Owned Real Property
5.08(d)(i)	-	Leased Real Property (Lessee)
5.08(d)(ii)	-	Leased Real Property (Lessor)
5.08(e)	-	Existing Investments
5.09	-	Environmental Matters
5.13	-	Subsidiaries and Other Equity Investments; Loan Parties
5.18	-	Intellectual Property Matters
5.21	-	Labor Matters
6.12	-	Guarantors
6.20	-	Deposit Accounts
7.01	-	Existing Liens
7.02	-	Existing Indebtedness
7.09		Restrictions
11.02	-	Administrative Agent's Office, Certain Addresses for Notices
11.06	-	Processing and Recordation Fees
11.07	_	Canadian Existing Liens
11.07	-	Canadian Existing Lions

EXHIBITS

Form of

A-1	_	Notice of Borrowing
A-2	_	Notice of Conversion/Continuation
B-1	-	Revolving Note
B-2	-	Swingline Note
C	_	Compliance Certificate
D	_	Assignment and Assumption
E-1	_	Guaranty
E-2	_	Canadian Guaranty
F-1	_	Security Agreement
F-2	_	Canadian Security Agreement
G	_	Mortgage
H-1	_	Intellectual Property Security Agreement
H-2	_	Canadian Intellectual Property Security Agreement
I-1	-	Opinion Matters – Ian R. Scheinmann, Esq., Vice President, Legal Affairs of Holdings
I-2	-	Opinion Matters – Hughes Hubbard & Reed LLP, Special New York Counsel to Loan Parties
I-3	-	Opinion Matters – Davis Graham & Stubbs LLP, Colorado Counsel to Loan Parties
I-4	_	Opinion Matters – DLA Piper LLP, Georgia, Maryland, Massachusetts, North Carolina and
		Virginia Counsel to Loan Parties
I-5	-	Opinion Matters – Bose McKinney & Evans LLP, Indiana Counsel to Loan Parties
I-6	-	Opinion Matters – Fasken Martineau DuMoulin LLP, Ontario Counsel to the Canadian Guarantors
I-7	-	Opinion Matters – Stewart McKelvey, Nova Scotia Counsel to the Canadian Guarantors
J	_	Joinder Agreement
ĸ	-	Pro Forma Adjustment Compliance Certificate
L	_	Tax Status Certificate
M	_	Borrowing Base Certificate

CREDIT AGREEMENT

This CREDIT AGREEMENT ("<u>Agreement</u>") is entered into as of April 16, 2013, among CENVEO CORPORATION, a Delaware corporation<u>as a debtor and debtor-in-possession under Chapter 11 of the Bankruptcy</u> <u>Code</u> (the "<u>Borrower</u>"), CENVEO, INC., a Colorado corporation<u>as a debtor and debtor-in-possession under</u> <u>Chapter 11 of the Bankruptcy Code</u> ("<u>Holdings</u>"), each lender from time to time party hereto (collectively, the "<u>Lenders</u>" and individually, a "<u>Lender</u>"), BANK OF AMERICA, N.A., as Administrative Agent, Issuing Bank and Swingline Lender, <u>BARCLAYS BANK PLC, GENERAL ELECTRIC CAPITAL CORPORATION-</u> <u>ANDJPMORGAN CHASE BANK, N.A., and</u> WELLS FARGO BANK, NATIONAL ASSOCIATION, as Co-Syndication Agents and PNC, and BANK OF AMERICA, N.A., JPMORGAN CHASE BANK, N.A., and WELLS FARGO BANK, NATIONAL ASSOCIATION, as <u>Documentation AgentJoint Lead Arrangers and Joint</u> <u>Book Managers</u>.

PRELIMINARY STATEMENTS:

On the Closing Date, Holdings, the Borrower, Bank of America, N.A., as administrative agent, and the lenders from time to time party thereto will enter into the Second Amended and Restated Credit Agreement and on the Closing Date, the Borrower will use the proceeds of borrowings thereunder to refinance the term loans under the Existing Credit Agreement and terminate the revolving commitments under the Existing Credit Agreement simultaneously therewith.

In furtherance of the foregoing, (a) the Borrower has requested that the Lenders extend credit in the form of Revolving Loans in an aggregate principal amount at any time outstanding not to exceed \$200,000,000, (b) the Borrower has requested that the Issuing Bank issue Letters of Credit in an aggregate stated amount at any time outstanding not to exceed \$35,000,000 and (c) the Borrower has requested the Swingline Lender to extend credit in the form of Swingline Loans in an aggregate principal amount at any time outstanding not to exceed \$20,000,000.

The Lenders are willing to extend such credit to the Borrower, the Swingline Lender is willing to make Swingline Loans to the Borrower and the Issuing Bank is willing to issue Letters of Credit for the account of the Borrower on the terms and subject to the conditions set forth herein.

ARTICLE I

DEFINITIONS AND ACCOUNTING TERMS

1.01 <u>Defined Terms</u>.

As used in this Agreement, the following terms shall have the meanings set forth below:

<u>"2014 Intercreditor Agreement</u>" means that certain Intercreditor Agreement, dated as of June 26, 2014, among the Borrower, Holdings, the other Guarantors, Bank of America, N.A., as administrative agent for the holders of the "Revolving Credit Obligations" (as defined therein), The Bank of New York Mellon, as collateral agent for the holders of the "Senior Priority Fixed Asset Obligations" (as defined therein), and The Bank of New York-Mellon, as collateral agent for the holders of the "Junior Priority Obligations" (as defined therein), as the same maybe amended, amended and restated, supplemented or otherwise modified from time to time.

"2014 Notes" means, collectively, the First Lien Notes and the 2014 Second Lien Notes.

<u>"2014 Second Lien Notes</u>" means the 8.500% Junior Priority Secured Notes due 2022 issued by the Borrower pursuant to the 2014 Second Lien Notes Indenture.

<u>"2014 Second Lien Notes Indenture</u>" means the Indenture, dated as of June 26, 2014, among the Borrower, the guarantors party thereto and The Bank of New York Mellon, in its capacities as trustee and as collateral agent, which provides for the issuance of the 2014 Second Lien Notes.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 34 of 432

"2016 Intercreditor Agreement" means that certain Intercreditor Agreement, dated as of June 10, 2016, among the Borrower, Holdings, the other Guarantors, Bank of America, N.A., as the "Senior Priority Collateral Agent" (as defined therein), and The Bank of New York Mellon, as collateral agent for the 2016 Secured Note-Purchasersholders of the 4% Notes, as the same may be amended, amended and restated, supplemented or otherwise modified from time to time. "2016 Notes" means, collectively, the 2016 Secured Notes and the 2016 Unsecured-Notes.

"4% Notes" means the 4% Senior Secured Notes due 2021 issued by the Borrower pursuant to the 4% Notes Documents.

<u>"4% Notes Documents" means the 4% Notes Indenture, the 4% Notes and all other agreements, instruments</u> and other documents pursuant to which the 4% Notes have been or will be issued or otherwise setting forth the terms of the 4% Notes.

"2016 Secured Note Purchase Agreement4% Notes Indenture" means the Indenture and Note Purchase Agreement, dated as of June 10, 2016,2016 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, Holdings, the other guarantors party thereto, the 2016 Secured Note Purchasers and The-Bank of New York Mellon, as trustee, which provides for the issuance of the 2016 Secured Notes.-guarantors named therein, Allianz US High Yield Fund and Allianz Income And Growth Fund, as purchasers, each noteholder party hereto and the 4% Notes Indenture Trustee.

"<u>2016 Secured Notes</u>" means the 4% Senior Secured Notes due December 2021 issued by the Borrower to the 2016 Secured Note Purchasers pursuant to the 2016 Secured Note Purchase Agreement. <u>4% Notes Indenture</u> <u>Trustee</u>" means The Bank of New York Mellon, in its capacities as trustee and collateral agent under the 4% Notes <u>Indenture</u>.

<u>"6% Notes" means the 6.000% Senior Priority Secured Notes due 2019 issued by the Borrower pursuant to</u> the 6% Notes Documents.

"2016 Secured Note Purchasers" means Allianz Global Investors U.S. LLC, a Delaware limited liabilitycompany, and the other purchasers from time to time of the 2016 Secured Notes under the 2016 Secured Note Purchase Agreement, together with their successors and assigns.<u>6% Notes Documents</u>" means the <u>6% Notes</u> Indenture, the <u>6% Notes and all other agreements</u>, instruments and other documents pursuant to which the <u>6% Notes</u> have been or will be issued or otherwise setting forth the terms of the <u>6% Notes</u>.

<u>"6% Notes Indenture" means the Indenture, dated as of June 26, 2014 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, the guarantors party thereto and the 6% Notes Indenture Trustee.</u>

<u>"6% Notes Indenture Trustee" means The Bank of New York Mellon, in its capacities as trustee and collateral agent under the 6% Notes Indenture.</u>

"2016 Unsecured 6% Unsecured Notes" means the 6.000% Senior Notes due 2024 issued by the Borrower pursuant to the 6% Unsecured Notes Documents.

<u>"6% Unsecured Notes Documents" means the 6% Unsecured Notes Indenture, the 6% Unsecured Notes and all other agreements, instruments and other documents pursuant to which the 6% Unsecured Notes have been or will be issued or otherwise setting forth the terms of the 6% Unsecured Notes.</u>

<u>"6% Unsecured Notes Indenture</u>" means the Indenture, dated as of June 10, <u>2016,2016 (as amended,</u> <u>supplemented or otherwise modified from time to time), by and</u> among the Borrower, the guarantors party thereto and <u>the 6% Unsecured Notes Indenture Trustee.</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 35 of 432

<u>"6% Unsecured Notes Indenture Trustee" means The Bank of New York Mellon, in its capacity as trustee</u> under the 6% Unsecured Notes Indenture.

<u>"8.5% Notes" means the 8.500% Junior Priority Secured Notes due 2022 issued by the Borrower pursuant</u> to the 8.5% Notes Documents.

<u>"8.5% Notes Documents" means the 8.5% Notes Indenture, the 8.5% Notes and all other agreements, instruments and other documents pursuant to which the 8.5% Notes have been or will be issued or otherwise setting forth the terms of the 8.5% Notes.</u>

<u>"8.5% Notes Indenture" means the Indenture, dated as of June 26, 2014 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, the guarantors named therein and the 8.5% Notes Indenture Trustee.</u>

<u>"8.5% Notes Indenture Trustee" means</u> The Bank of New York Mellon, in its capacities as trustee and as collateral agent, which provides for the issuance of the 2016 Unsecured Notes. "<u>2016 Unsecured Notes</u>" means the 6.000% Senior Notes due May 2024 issued by the Borrower pursuant to the 2016 Unsecured under the 8.5% Notes Indenture.

"Account Debtor" means any Person who may become obligated to another Person under, with respect to, or on account of, an Account.

"<u>Accounts</u>" means all "accounts," as such term is defined in the UCC as in effect on the date hereof in the State of New York (or, with respect to any Canadian Guarantor, as such term is defined in the PPSA), in which any Person now or hereafter has rights.

"<u>Acquisition</u>" by any Person, means the purchase or acquisition in a single transaction or a series of related transactions by any such Person, individually or, together with its Affiliates, of (a) any Equity Interest of any other Person (other than an existing Subsidiary of the Borrower) which are sufficient such that such other Person becomes a direct or indirect Subsidiary of the Borrower or (b) all or a substantial portion of the Property, including, without limitation, all or a substantial portion of the property comprising a division, business unit or line of business, of any other Person (other than a Subsidiary of the Borrower), whether involving a merger or consolidation with such other Person. "Acquire" has a meaning correlative thereto.

"Acquisitions and Investment Conditions" means, as to any relevant action contemplated in this Agreement, (i) no Event of Default has then occurred and is continuing or would result from any action and (ii) either (1) (x)(a)-Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 15.0% of the Aggregate Commitments and (b) over the 30 consecutive days prior to consummation of such action, Availability ona Pro Forma Basis for such action averaged no less than 15.0% of the Aggregate Commitments (for the avoidance of doubt, Availability for each of the foregoing clauses (a) and (b) shall be calculated to include any applicableacquired entity's working capital assets, subject to receipt by the Administrative Agent of satisfactory inventoryappraisals and field examination information of such acquired entity from examiners and appraisers reasonablysatisfactory to the Administrative Agent) and (y) the Consolidated Fixed Charge Coverage Ratio would be at least-1.0 to 1.0 on a Pro Forma Basis for such action, or (2) (x) Availability on a Pro Forma Basis immediately aftergiving effect to such action would be at least 20.0% of the Aggregate Commitments and (y) over the 30 consecutivedays prior to consummation of such action, Availability on a Pro Forma Basis for such action averaged no less than 20.0% of the Aggregate Commitments (for the avoidance of doubt, Availability for each of the foregoing clauses (x)and (y) shall be calculated to include any applicable acquired entity's working capital assets, subject to receipt by the Administrative Agent of satisfactory inventory appraisals and field examination information of such acquired entity from examiners and appraisers reasonably satisfactory to the Administrative Agent).

"<u>Additional Senior Unsecured Notes</u>" means any senior notes evidencing Indebtedness permitted under <u>Section 7.02(j)</u>.

"Adjustment Date" means the first day of January, April, July and October of each fiscal year.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 36 of 432

"<u>Administrative Agent</u>" means Bank of America in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.

"<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule 11.02</u>, or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.

"<u>Administrative Questionnaire</u>" means an Administrative Questionnaire in a form supplied by the Administrative Agent.

"<u>Affiliate</u>" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.

"<u>Agent Parties</u>" has the meaning specified in <u>Section 11.02(c)</u>.

"<u>Agent-Related Person</u>" means the Administrative Agent and its affiliates and the officers, directors, employees, agents and attorneys-in-fact of the Administrative Agent and its affiliates.

"<u>Aggregate Commitments</u>" means, at any time, the aggregate amount of the Revolving Commitments of all Lenders.

"<u>Aggregate Exposures</u>" means, at any time, the sum of (a) the aggregate Outstanding Amount of all Revolving Loans *plus* (b) the LC Exposure, each determined at such time.

"Agreement" means this Credit Agreement.

"<u>Amendment No. 1</u>" means Amendment No. 1 to the Credit Agreement, dated as of December 11, 2013, by and among the Borrower, the Administrative Agent, the Increase Loan Lenders (as defined therein) party thereto, the Lenders party thereto and the other Loan Parties party thereto.

"Amendment No. 1 Effective Date" means December 11, 2013.

"<u>Amendment No. 2</u>" means Amendment No. 2 to the Credit Agreement, dated as of June 10, 2014, by and among the Borrower, the Lenders party thereto and the other Loan Parties party thereto, and acknowledged by the Administrative Agent.

"<u>Amendment No. 3</u>" means Amendment No. 3 to the Credit Agreement, dated as of January 30, 2015, by and among the Borrower, the Lenders party thereto and the other Loan Parties party thereto, and acknowledged by the Administrative Agent.

"Amendment No. 3 Effective Date" means January 30, 2015.

"<u>Amendment No. 4</u>" means Amendment No. 4 to the Credit Agreement, dated as of June 10, 2016, by and among the Borrower, the Lenders party thereto, the Issuing Bank and the other Loan Parties party thereto, and acknowledged by the Administrative Agent.

"Amendment No. 4 Effective Date" means June 10, 2016.

"<u>Amendment No. 5</u>" means Amendment No. 5 to the Credit Agreement, dated as of April 27, 2017, by and among the Borrower, the Lenders party thereto and the other Loan Parties party thereto, and acknowledged by the Administrative Agent.

"Amendment No. 5 Effective Date" means April 27, 2017.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 37 of 432

"<u>Anti-Terrorism Laws</u>" means Law related to terrorism financing or money laundering including the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act ("<u>USA PATRIOT Act</u>") of 2001 (Title III of Pub. L. 107-56), The Currency and Foreign Transactions Reporting Act (also known as the "<u>Bank Secrecy Act</u>," 31 U.S.C. §§ 5311-5330 and 12 U.S.C. §§ 1818(s), 1820(b) and 1951-1959), the Trading With the Enemy Act (50 U.S.C. § 1 <u>et seq</u>., as amended), Executive Order 13224 (effective September 24, 2001) and Canadian Anti-Terrorism and AML Legislation.

"<u>Applicable Margin</u>" means, (a) with respect to any Type of Revolving Loan, (x) from and after the Amendment No. 4 Effective Date and prior to June 10, 2017, the per annum margin set forth below for Level III, and (y) prior to the Amendment No. 4 Effective Date and from and after June 10, 2017, the per annum margin set forthbelow, as determined by the Average Availability as of the most recent Adjustment Date: <u>Base Rate Loans, 1.75%</u> per annum, and (b) with respect to Eurodollar Rate Loans, 2.75% per annum.

Level	Average- Availability (percentage of Aggregate Commitments)	Base Rate- Loans	Eurodoll ar Rate Loans
Ŧ	<u> </u>	1.00%	2.00%
Ħ	\geq 33% but < 66%	1.25%	2.25%
Ħ	< 33%	1.50%	2.50%

Subject to clause (x) above, the Applicable Margin shall be subject to increase or decrease on the first-Business Day of each fiscal quarter based on Average Availability, and each such increase or decrease in the-Applicable Margin shall be effective on the Adjustment Date occurring immediately after the last day of the fiscalquarter most recently ended. If the Borrower fails to deliver any Borrowing Base Certificate on or before the date required for delivery thereof, then, at the option of the Required Lenders, the Applicable Margin shall be determinedas if Level III were applicable, from the first day of the calendar month following the date such Borrowing Base Certificate was required to be delivered until the date of delivery of such Borrowing Base Certificate.

"<u>Applicable Percentage</u>" means, with respect to any Lender at any time, the percentage (carried out to the ninth decimal place) of the Aggregate Commitments represented by the principal amount of such Lender's Revolving Commitment at such time. The initial Applicable Percentage of each Lender is set forth opposite the name of such Lender on <u>Schedule 2.01</u> hereto or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"Approved Budget" has the meaning set forth in the Ratification Agreement.

"<u>Approved Fund</u>" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"<u>Approved Processor</u>" means a processor at which Inventory in an amount in excess of \$100,000 is located and has entered into a Landlord Lien Waiver and Access Agreement.

"<u>Assignee Group</u>" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.

"<u>Assignment and Assumption</u>" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by <u>Section 11.06(b)</u>), and accepted by the Administrative Agent, in substantially the form of <u>Exhibit D</u> or any other form approved by the Administrative Agent.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 38 of 432

"Assignment Taxes" has the meaning specified in the definition of "Other Taxes."

"<u>Attributable Indebtedness</u>" means, on any date, (a) in respect of any Capitalized Lease of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease or similar payments under the relevant lease or other applicable agreement or instrument that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease or other agreement or instrument were accounted for as a Capitalized Lease and (c) all Synthetic Debt of such Person.

"<u>Audited Financial Statements</u>" means the audited consolidated balance sheet of Holdings and its Subsidiaries for the fiscal year ended December 29, 2012, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of Holdings and its Subsidiaries, including the notes thereto, in each case as included in Holdings' Annual Report on Form 10-K for the fiscal year ended December 29, 2012.

"<u>Availability</u>" means, as of any applicable date, the amount by which the Line Cap at such time exceeds the Aggregate Exposures on such date.

"<u>Average Availability</u>" means, at any Adjustment Date, the average daily Availability for the fiscal quarter immediately preceding such Adjustment Date.

"<u>Average Usage</u>" means, on any date of determination, the average utilization of Revolving Commitments during the immediately preceding fiscal <u>quartermonth</u>.

"<u>Bail-In Action</u>" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"<u>Bail-In Legislation</u>" means, with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"Bank of America" means Bank of America, N.A. and its successors.

"<u>Bank Product</u>" means any of the following products, services or facilities extended to the Borrower or any of its Subsidiaries: (a) Cash Management Services; (b) products under Swap Contracts; (c) commercial credit card and merchant card services; (d) foreign exchange banking products and services; and (e) other banking products or services as may be requested by the Borrower, other than Letters of Credit.

"Bank Product Debt" means Indebtedness and other obligations of the Borrower or any of its Subsidiaries relating to Bank Products.

"<u>Bank Product Reserve</u>" means the aggregate amount of reserves established by the Administrative Agent from time to time in its Permitted Discretion in respect of Secured Bank Product Obligations.

"Bankruptcy Court" means the United States Bankruptcy Court for the Southern District of New York.

<u>"Base Rate</u>" means, for any day, a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate" and (c) the Eurodollar Rate for an Interest Period of one month plus 1.00%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in the Base Rate due to a change in the "prime rate" or the Federal Funds Rate shall take effect at the opening of business on the day specified in the public announcement of such change. For the purposes of <u>subsection (c)</u> above, the Eurodollar Rate shall be determined daily and any change in the Eurodollar Rate shall take effect on the day of such change.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 39 of 432

"Base Rate Loan" means a Revolving Loan that bears interest based on the Base Rate.

"Borrower" has the meaning specified in the introductory paragraph hereto<u>(and including its successors</u> and assigns, including any trustee or other fiduciary hereafter appointed as its legal representative or with respect to the property of the estate of such entity whether under Chapter 11 of the Bankruptcy Code or any subsequent Chapter 7 case and its successor upon conclusion of the Chapter 11 Case of such entity).

"Borrower Materials" has the meaning specified in Section 6.02.

"<u>Borrowing</u>" means a borrowing consisting of simultaneous Revolving Loans of the same Type and, in the case of Eurodollar Rate Loans, having the same Interest Period made by each of the Lenders pursuant to <u>Section</u> <u>2.02</u>.

"Borrowing Base" means, at any time of calculation, the Dollar Equivalent sum of, without duplication:

(a) 85% of Eligible Accounts of the Loan Parties, *plus*

(b) the lesser of (x) 65% of the lesser of Cost (on a basis consistent with the Loan Parties' historical accounting practices) or market value of Eligible Inventory of the Loan Parties and (y) 85% of the appraised NOLV Percentage of Eligible Inventory of the Loan Parties; *minus*

(c) any Reserves established from time to time by the Administrative Agent in accordance herewith.

The Administrative Agent shall (i) promptly notify the Borrower in writing (including via e-mail) whenever it determines that the Borrowing Base set forth on a Borrowing Base Certificate differs from the Borrowing Base, (ii) discuss the basis for any such deviation and any changes proposed by the Borrower, including the reasons for any impositions of or changes in Reserves or any change in advance rates with respect to Eligible Accounts (in the Administrative Agent's Permitted Discretion and subject to the definition thereof) or eligibility criteria, with the Borrower, (iii) consider, in the exercise of its Permitted Discretion, any additional factual information provided by the Borrower relating to the determination of the Borrower. Pending a decision by the Administrative Agent to make any requested change, the initial determination of the Borrowing Base by the Administrative Agent shall continue to constitute the Borrowing Base.

"<u>Borrowing Base Certificate</u>" means a certificate of a Responsible Officer of the Borrower substantially in the form of <u>Exhibit M</u>.

"<u>Business Day</u>" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, the state where the Administrative Agent's Office is located, Stamford, Connecticut or New York, New York and, if such day relates to any Eurodollar Rate Loan, means any such day that is a London Banking Day.

"<u>Canadian Anti-Terrorism and AML Legislation</u>" means the anti-terrorist provisions of the *Criminal Code* (Canada), the *Proceeds of Crime Act*, the *Terrorist Financing Act* (Canada), the *United Nations Suppression of Terrorism Regulations* and the *Anti-terrorism Act* (Canada) and all regulations and orders made thereunder.

"<u>Canadian Benefit Plan</u>" means any employee benefit, health, welfare, supplemental unemployment benefit, bonus, pension, supplemental pension, profit sharing, retiring allowance, severance, deferred compensation, stock compensation, stock purchase, unit purchase, retirement, life, hospitalization insurance, medical, dental, disability or other employee group or similar benefit or employment plans or supplemental arrangements applicable to the Canadian Employees other than statutory employee plans or the Canadian Pension Plans.

"Canadian Dollars" and "Cdn\$" mean lawful currency of Canada.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 40 of 432

"Canadian Employee" means any employee or former employee of a Canadian Guarantor.

"<u>Canadian Existing Lien Reserve</u>" means, until the date that the Canadian Guarantor delivers (i) an estoppel letter in form and substance reasonably satisfactory to the Administrative Agent, or (ii) confirmation of discharge, in each case in respect of those PPSA registrations listed on <u>Schedule 11.07</u>, a reserve equal to 150% of the liabilities owing to the secured party under such registration.

"<u>Canadian Guarantor</u>" means Cenveo Canada and each other Canadian Subsidiary of Holdings that is, or becomes, a Guarantor hereunder.

"Canadian Intellectual Property Security Agreement" has the meaning specified in Section 4.01(a)(v).

"<u>Canadian IP Security Agreement Supplement</u>" has the meaning specified in <u>Section 1.03</u> of the Canadian Security Agreement.

"Canadian Mortgage" has the meaning specified in Section 6.21(a)(i).

"<u>Canadian Multi-Employer Pension Plan</u>" means any multi-employer pension plan as that term is defined under Section 147.1(1) of the *Income Tax Act* (Canada).

"Canadian Pension Legislation" means the Pension Benefits Act (Ontario), and any Canadian federal, provincial, territorial or local counterparts or equivalents.

"<u>Canadian Pension Plan</u>" means any pension plan that is required to be registered under the Income Tax Act (Canada) and contributed to by a Canadian Guarantor for its Canadian Employees, including a pension plan that is subject to the Pension Benefits Act (Ontario) or other Canadian Pension Legislation and includes any Canadian Multi-Employer Pension Plan to which a Canadian Guarantor is required by contract to contribute to for its Canadian Employees but does not include the Canada Pension Plan maintained by the Government of Canada or Quebec.

"Canadian Priority Payables Reserve" means, on any date of determination, reserves established by the Administrative Agent in its Permitted Discretion for amounts payable by Canadian Guarantors and secured by any Liens, choate or inchoate, which rank or which would reasonably be expected to rank in priority to or pari passu with the Administrative Agent's Liens on Collateral in the Borrowing Base, including, without duplication, amounts deemed to be held in trust, or held in trust, pursuant to applicable Law and/or for amounts which represent costs in connection with the preservation, protection, collection or realization of the Collateral included in the Borrowing Base, any such amounts due and not paid for wages, vacation pay, amounts (including severance pay) payable under the Wage Earner Protection Program Act (Canada) or under the Bankruptcy and Insolvency Act (Canada) or the Companies' Creditors Arrangement Act (Canada), amounts due and not paid under any legislation relating to workers' compensation or to employment insurance, all amounts deducted or withheld and not paid and remitted when due under the Income Tax Act (Canada), sales tax, goods and services tax, value added tax, harmonized tax, excise tax, tax payable pursuant to Part IX of the Excise Tax Act (Canada) or similar applicable provincial legislation, government royalties, amounts currently or past due and not paid for realty, municipal or similar taxes and all amounts currently or past due and not contributed, remitted or paid in respect of any Canadian Pension Plan, or any similar statutory or other claims that would have or would reasonably be expected to have priority over or pari passu with any Liens on Collateral in the Borrowing Base granted to the Administrative Agent in the future (including, as the case may be, in respect of any wind up deficiency or solvency deficiency, whether or not yet due or pavable).

"Canadian Security Agreement" has the meaning specified in Section 4.01(a)(iii).

"<u>Canadian Security Agreement Supplement</u>" has the meaning specified in <u>Section 1.03</u> of the Canadian Security Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 41 of 432

"Canadian Subsidiary" means any Subsidiary that is organized under the laws of Canada or any province or territory thereof.

"<u>Capital Expenditures</u>" means, with respect to any Person for any period, any expenditure in respect of the purchase or other acquisition of any fixed or capital asset (excluding normal replacements and maintenance which are properly charged to current operations). For purposes of this definition, the purchase price of equipment that is purchased simultaneously with the trade-in of existing equipment or with Insurance Proceeds shall be included in Capital Expenditures only to the extent of the gross amount by which such purchase price exceeds the credit granted by the seller of such equipment for the equipment being traded in at such time or the amount of such Insurance Proceeds, as the case may be. It is understood and agreed that for purposes of calculating the Consolidated Fixed Charge Coverage Ratio, Capital Expenditures shall never be less than zero.

"<u>Capitalized Leases</u>" means all leases that have been or should be, in accordance with GAAP, recorded as capitalized leases.

"Carve-Out" has the meaning set forth in the Financing Orders.

"Carve-Out Reserves" has the meaning set forth in the Financing Orders.

<u>"Cash Collateralize</u>" means to pledge and deposit with or deliver to the Administrative Agent for deposit into the LC Collateral Account, for the benefit of the Administrative Agent, the Issuing Bank or the Swingline Lender (as applicable) and the Lenders, cash as collateral for the LC Exposure, Obligations in respect of Swingline Loans, or obligations of Lenders to fund participations in respect of either thereof (as the context may require) in accordance with <u>Section 2.13(j)</u>. "<u>Cash Collateral</u>" has a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"<u>Cash Equivalents</u>" means any of the following types of Investments, to the extent owned by the Borrower or any of its Subsidiaries free and clear of all Liens (other than Liens created under the Collateral Documents):

(a) readily marketable obligations issued or directly and fully guaranteed or insured by the United States of America or Canada or any agency or instrumentality thereof having maturities of not more than one year from the date of acquisition thereof; <u>provided</u> that the full faith and credit of the United States of America or Canada, as the case may be, is pledged in support thereof;

(b) time deposits with, or insured certificates of deposit or bankers' acceptances of, any nationally-recognized securities dealer or any commercial bank, trust company, savings and loan association or savings bank that (i) (A) is a Lender or (B) is (x) organized under the laws of the United States of America, any state thereof or the District of Columbia, or is organized under the laws of Canada, or any province or territory thereof, or a political subdivision thereof, (y) a U.S. branch of any such institution organized under the laws of any other country that is a member of the OECD, or (z) the principal banking subsidiary of a bank holding company organized under the laws of the United States of America, any state thereof or the District of Columbia, and is a member of the Federal Reserve System, (ii) issues (or the parent of which issues) commercial paper rated as described in <u>clause (c)</u> of this definition and (iii) has combined capital and surplus of at least \$1,000,000,000, in each case with maturities of not more than 180 days from the date of acquisition thereof;

(c) commercial paper issued by any Person organized under the laws of any state of the United States of America or any province or territory of Canada and rated at least "Prime-2" (or the then equivalent grade) by Moody's or at least "A-2" (or the then equivalent grade) by S&P, in each case with maturities of not more than 180 days from the date of acquisition thereof;

(d) Investments, classified in accordance with GAAP as current assets of the Borrower or any of its Subsidiaries, in money market investment programs registered under the Investment Company Act of 1940, which are administered by financial institutions that have the highest rating obtainable from either

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 42 of 432

Moody's or S&P, and the portfolios of which are limited solely to Investments of the character, quality and maturity described in <u>clauses (a)</u>, (b) and (c) of this definition; and

(e) obligations with any Lender, any other bank or trust company described in <u>clause (b)</u> above, or any nationally-recognized securities dealer, in respect of the repurchase of obligations of the type described in <u>clause (a)</u> above, <u>provided</u> that such repurchase obligations shall be fully secured by obligations of the type described in said clause and the possession of such obligations shall be transferred to, and segregated from other obligations owned by, such Lender, such other bank or trust company or such securities dealer.

"<u>Cash Management Services</u>" means any services provided from time to time to the Borrower or any of its Subsidiaries in connection with operating, collections, payroll, trust, or other depository or disbursement accounts, including automated clearinghouse, e-payable, electronic funds transfer, wire transfer, controlled disbursement, overdraft, depository, information reporting, lockbox and stop payment services.

"Cenveo Canada" means Cenveo McLaren Morris and Todd Company, a Nova Scotia unlimited company.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980.

"CFC" means a Person that is a controlled foreign corporation under Section 957 of the Code.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Change of Control" means an event or series of events by which:

(a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of such person or its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan), except the Permitted Holders, becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934), directly or indirectly, of 35% or more of the equity securities of Holdings entitled to vote for members of the board of directors or equivalent governing body of Holdings on a fully-diluted basis; or

(b) during any period of <u>twelve (12)</u> consecutive months, a majority of the members of the board of directors or other equivalent governing body of Holdings cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in <u>clause (i)</u> above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body was approved by individuals referred to in <u>clauses (i)</u> above constituting at the time of such election or nomination to that board or other equivalent governing body was approved by individuals referred to in <u>clauses (i)</u> and <u>(ii)</u> above constituting at the time of such election or nomination to that board or other equivalent governing body was approved by individuals referred to in <u>clauses (i)</u> and <u>(ii)</u> above constituting at the time of such election or nomination to that board or other equivalent governing body was approved by individuals referred to in <u>clauses (i)</u> and <u>(ii)</u> above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body; or

(c) any Person or two or more Persons acting in concert (excluding the Permitted Holders) shall have acquired by contract or otherwise, the power to exercise, directly or indirectly, a controlling influence over the management or policies of Holdings, or control over the equity securities of Holdings

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 43 of 432

entitled to vote for members of the board of directors or equivalent governing body of Holdings on a fully-diluted basis representing 35% or more of the combined voting power of such securities; or

(d) Holdings shall cease, directly or indirectly, to own and control legally and beneficially all of the Equity Interests in the Borrower.

<u>"Chapter 11 Cases" means the Chapter 11 cases of the Debtors, which are being jointly administered under</u> the Bankruptcy Code and are pending in the Bankruptcy Court.

"<u>Chattel Paper</u>" has the meaning provided in Article 9 of the UCC (or, with respect to any Canadian Guarantor, as such term is defined in the PPSA).

"Closing Date" means April 16, 2013.

"Co-Syndication Agents" has the meaning assigned to such term in the preamble hereto.

"Code" means the Internal Revenue Code of 1986.

"<u>Collateral</u>" means all of the "Collateral" and "Mortgaged Property" referred to in the Collateral Documents and all of the other property that is or is intended under the terms of the Collateral Documents to be subject to Liens in favor of the Administrative Agent for the benefit of the Secured Parties <u>(including both the</u><u>Pre-Petition Collateral and the Post-Petition Collateral)</u>.

"Collateral Documents" means, collectively, the Security Agreement, the Canadian Security Agreement, the Intellectual Property Security Agreement, the Canadian Intellectual Property Security Agreement, the Mortgages, the Canadian Mortgages, the Deposit Account Control Agreements, the Security Agreement Supplements, the Canadian Security Agreement Supplements, the Canadian IP Security Agreement Supplements, or other similar agreements delivered to the Administrative Agent pursuant to Section 6.12, and each of the other agreements, instruments or documents that creates or purports to create a Lien in favor of the Administrative Agent for the benefit of the Secured Parties.

"<u>Commitment</u>" means, with respect to any Lender, such Lender's Revolving Commitment, LC Commitment or Swingline-Commitment, or any Extended Revolving Loan Commitment.

"<u>Commodity Exchange Act</u>" means the Commodity Exchange Act (7 U.S.C. § 1 <u>et seq</u>.), as amended from time to time, and any successor statute.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C.

"Consolidated Adjusted EBITDA" means, at any date of determination, an amount equal to Consolidated-Net Income of Holdings and its Subsidiaries on a consolidated basis for the most recently completed Measurement-Period plus (a) the following to the extent deducted in calculating such Consolidated Net Income for such-Measurement Period: (i) Consolidated Interest Charges, (ii) the provision for Federal, state, local and foreignincome taxes payable and tax contingencies, (iii) depreciation and amortization expense and (iv) non cash chargesresulting from the write down of goodwill, furniture, fixtures, equipment and software, minus (b) the following to the extent added in calculating such Consolidated Net Income for such Measurement Period: (i) Federal, state, local and foreign income tax credits and tax contingency credits and (ii) all non-cash items increasing Consolidated Net-Income (in each case of or by Holdings and its Subsidiaries for such Measurement Period), and excluding (c) without duplication, the following to the extent included in the calculation of Consolidated Net Income for such Measurement Period: (i) cash restructuring, integration, impairment and related fees, expenses and charges in anaggregate amount of up to the lesser of (x) 5.0% of Consolidated Adjusted EBITDA (prior to giving effect to anyaddbacks pursuant to this clause (i)) and (y) \$10,000,000 during any fiscal year, (ii) non cash restructuring, integration, impairment and related fees, expenses and charges, (iii) the results of operations and gain (loss) on the Disposition of non-strategic assets, businesses and discontinued operations, (iv) stock based compensation expense, (v) gain (loss) on the early extinguishment of Indebtedness, (vi) cumulative effect of changes in accounting-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 44 of 432

principles, (vii) non-cash extraordinary gains and non-cash extraordinary losses for such Measurement Period and (viii) other non-cash non-recurring charges and expenses (in each case of or by Holdings and its Subsidiaries forsuch Measurement Period). For the purpose of calculating the Financial Covenant, Consolidated Adjusted EBITDAshall be calculated on a Pro-Forma Basis in accordance with the provisions in <u>Section 1.03</u>. Notwithstanding the foregoing, Consolidated Adjusted EBITDA shall be the amount set forth on <u>Schedule 1.03</u> for each of the respectiveperiods referenced therein.

"<u>Consolidated First Lien Leverage Ratio</u>" has the meaning assigned to such term in the Second Amended and Restated Credit Agreement as in effect on the Closing Date.

"Consolidated Fixed Charge Coverage Ratio" means, for any period of four consecutive fiscal quarters forwhich financial statements were required to have been delivered in accordance with Section 6.01, the ratio of (a)-Consolidated Adjusted EBITDA for such period, minus (x) Capital Expenditures paid in cash (excluding any Capital-Expenditures to the extent financed with the proceeds of any Indebtedness (other than Indebtedness hereunder)) forsuch period, (y) the amount of cash payments made during such period by Holdings, the Borrower and theirrespective Subsidiaries in respect of federal, state, provincial, local and foreign income taxes during such period and (z) Restricted Payments permitted by Section 7.06(d) paid in cash for such period to (b) Consolidated Fixed Chargesfor such period. In making any determination of the Consolidated Fixed Charge Coverage Ratio, in the event that the Borrower or any of its Subsidiaries incurs, assumes, guarantees, repays, repurchases, redeems, defeases or otherwisedischarges any Indebtedness (other than in the case of revolving credit borrowings, in which case interest expensewill be computed based upon the average daily balance of such Indebtedness during the Measurement Period), ineach case, subsequent to the commencement of the period for which the Consolidated Fixed Charge Coverage Ratiois being calculated and on or prior to the date on which the event for which the calculation of the Consolidated Fixed Charge Coverage Ratio is made (the "Calculation Date"), then the Consolidated Fixed Charge Coverage Ratio willbe calculated on a Pro Forma Basis as if such incurrence, assumption, guarantee, repayment, repurchase, redemption, defeasance or other discharge of Indebtedness, and the use of the proceeds therefrom, had occurred at the beginningof the Measurement Period.

For purposes of this definition, if any Indebtedness bears a floating rate of interest and is being calculated on a Pro Forma Basis, the interest on such Indebtedness will be calculated as if the rate in effect on the Calculation Date had been the applicable rate for the entire period (taking into account any hedging obligations applicable tosuch Indebtedness if such hedging obligations have a remaining term in excess of 12 months as of the Calculation Date). For purposes of this definition, interest on obligations in respect of Capitalized Leases will be deemed toaccrue at an interest rate reasonably determined by a Responsible Officer of the Borrower to be the rate of interestimplicit in such obligations in accordance with GAAP. For purposes of making the computation referred to above, interest on any Indebtedness under a revolving credit facility computed on a pro-forma basis will be computed basedupon the average daily balance of such Indebtedness during the applicable period. Interest on Indebtedness that mayoptionally be determined at an interest rate based upon a factor of a prime or similar rate, a eurocurrency interbankoffered rate, or other rate, will be deemed to have been based upon the rate actually chosen, or, if none, then basedupon such optional rate chosen as the Borrower may designate.

"<u>Consolidated Fixed Charges</u>" means, for any period of four consecutive fiscal quarters for which financial statements were required to have been delivered in accordance with <u>Section 6.01</u>, for Holdings, the Borrower and their respective Subsidiaries on a consolidated basis, the sum, without duplication, of (a) Consolidated Interest-Charges for such period to the extent paid in cash (or accrued and payable during such period on a current basis in cash), (b) the aggregate amount of scheduled amortization payments of principal made during such period in respect of Consolidated Funded Indebtedness of Holdings, the Borrower and their respective Subsidiaries and (c) to the extent not deducted from Consolidated Net Income to determine Consolidated Adjusted EBITDA during such period, amounts paid during such period with respect to any pension and other post employment benefit liabilities. Notwithstanding the foregoing, for purposes of calculating Consolidated Fixed Charges for any period that includes a fiscal quarter (or portion thereof) prior to the Closing Date, Consolidated Fixed Charges shall be calculated from the period from the Closing Date to the date of determination divided by the number of days in such period and multiplied by 365.

"<u>Consolidated Funded Indebtedness</u>" means, as of any date of determination, for Holdings and its Subsidiaries on a consolidated basis, the sum of (a) the outstanding principal amount of (i) all obligations, whether

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 45 of 432

current or long term, for borrowed money (including Obligations hereunder) and (ii) all obligations evidenced by bonds, debentures, notes, loan agreements or other similar instruments, (b) all purchase money Indebtedness, (c) all direct obligations arising under letters of credit (including standby and commercial), bankers' acceptances, bankguaranties, surety bonds and similar instruments, (d) all obligations in respect of the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business), (e) all Attributable-Indebtedness, (f) without duplication, all Guarantees with respect to outstanding Indebtedness of the types specified in clauses (a) through (e) above of Persons other than Holdings or any Subsidiary thereof, and (g) all Indebtedness of the types referred to in clauses (a) through (e) above of any partnership or joint venture (other than a joint venturethat is itself a corporation, limited liability company or similar legal entity) in which Holdings or a Subsidiary thereof is a general partner or joint venturer, except to the extent such Indebtedness is expressly made non-recourse to-Holdings or such Subsidiary. For the purpose of calculating the Financial Covenant, Consolidated Funded-Indebtedness shall be calculated on a Pro Forma Basis in accordance with the provisions in <u>Section 1.03</u>.

"<u>Consolidated Interest Charges</u>" means, at any date of determination, the sum of (a) all cash interest, premium payments, debt discount, fees, charges and related expenses in connection with borrowed money (including capitalized interest) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP, (b) all interest that is treated as "interest" under GAAP that is paid or payable with respect to discontinued operations, and (c) the portion of rent expense under Capitalized Leases that is treated as interest expense in accordance with GAAP, in each case, of or by Holdings and its Subsidiaries on a consolidated basis for the most recently completed Measurement Period. For the purpose of calculating the Financial Covenant, Consolidated Interest Charges shall be calculated on a Pro Forma Basis in accordance with the provisions in <u>Section</u> <u>1.03</u>.

"<u>Consolidated Leverage Ratio</u>" has the meaning assigned to such term in the Second Amended and Restated Credit Agreement as in effect on the Closing Date.

"Consolidated Net Income" means, at any date of determination, the net income (or loss) of Holdings and its Subsidiaries on a consolidated basis for the most recently completed Measurement Period; <u>provided</u> that-Consolidated Net Income shall exclude (a) the net income of any Subsidiary (other than any Loan Party) during such-Measurement Period to the extent that the declaration or payment of dividends or similar distributions by such-Subsidiary of such income is not permitted by operation of the terms of its Organization Documents or anyagreement, instrument or Law applicable to such Subsidiary during such Measurement Period, except that Holdings'equity in any net loss of any such Subsidiary for such Measurement Period shall be included in determining. Consolidated Net Income, and (b) any income (or loss) for such Measurement Period of any Person if such Person isnot a Subsidiary, except that Holdings' equity in the net income of any such Person for such Measurement Periodshall be included in Consolidated Net Income up to the aggregate amount of cash actually distributed by such Personduring such Measurement Period to Holdings or a Subsidiary as a dividend or other distribution (and in the case of adividend or other distribution to a Subsidiary (other than any Loan Party), such Subsidiary is not precluded fromfurther distributing such amount as described in clause (a) of this proviso). For the purpose of calculating the-Financial Covenant, Consolidated Net Income shall be calculated on a Pro Forma Basis in accordance with theprovisions in <u>Section 1.03</u>.

"<u>Consolidated Secured Debt Ratio</u>" has the meaning assigned to such term in the Senior Notes Indenture as in effect on the Closing Date.

"<u>Consolidated Senior Secured Debt Ratio</u>" has the meaning assigned to such term in the First Lien Notes-Indenture as in effect on the Amendment No. 3 Effective Date.

"<u>Consolidated Total Secured Debt Ratio</u>" has the meaning assigned to such term in the 2014 Second Lien-Notes Indenture as in effect on the Amendment No. 3 Effective Date.

"<u>Contractual Obligation</u>" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 46 of 432

"<u>Control</u>" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "<u>Controlling</u>" and "<u>Controlled</u>" have meanings correlative thereto.

"<u>Cost</u>" means, as reasonably determined by the Administrative Agent in good faith, with respect to Inventory, the lower of (a) cost computed on a specific identification or first in first out basis or (b) market value, <u>provided</u> that for purposes of the calculation of Borrowing Base, the Cost of Inventory shall not include (A) the portion of the cost of Inventory equal to the profit earned by any Affiliate on the sale thereof to the Borrower or another Loan Party, (B) write-ups or write-downs in cost with respect to currency exchange rates or (C) any "step-ups" calculated in connection with any Acquisition.

"Credit Event" means the making of any Revolving Loan.

"<u>Credit Extension</u>" means, as the context may require, (i) a Credit Event or (ii) the issuance, amendment, extension or renewal of any Letter of Credit by the Issuing Bank or the amendment, extension or renewal of any Existing Letter of Credit; <u>provided</u> that "Credit Extensions" shall not include (A) conversions and continuations of outstanding Revolving Loans or (B) amendments to any Letter of Credit that do not increase the stated amount thereof or extend the expiration date thereof.

"<u>Debt Rating</u>" means, as of any date of determination, the rating as determined by either S&P or Moody's of the Indebtedness under the <u>Second Amended and Restated Credit Agreement</u><u>6% Notes Indenture</u>.

"Debtor Relief Laws" means the Bankruptcy Code of the United States, the *Bankruptcy and Insolvency Act* (Canada), the *Companies' Creditors Arrangement Act* (Canada), the *Winding-Up Act* (Canada) and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States, Canada, any state, province or territory thereof, or of any other applicable jurisdictions, from time to time in effect and affecting the rights of creditors generally (including, as the case may be, applicable corporate statutes).

"Debtors" means, collectively, Borrower and Guarantors, each as debtor and debtor-in-possession in the <u>Chapter 11 Cases.</u>

"Default" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"Default Rate" has the meaning assigned to such term in Section 2.06(c).

"Defaulting Lender" means any Lender that (a) has failed to perform any funding obligations hereunder, or that has failed to pay to the Administrative Agent, any Issuing Bank, any Swingline Lender or any other Lender any other amount required to be paid by it hereunder (including in respect of its participation in Letters of Credit or Swingline Loans) and, in each case, such failure is not cured within three Business Days; (b) has notified the Administrative Agent or the Borrower that such Lender does not intend to comply with its funding obligations hereunder or has made a public statement to the effect that it does not intend to comply with its funding obligations hereunder or under any other credit facility; (c) has failed, within three Business Days following request by the Administrative Agent, to confirm in a manner satisfactory to the Administrative Agent that such Lender will comply with its funding obligations hereunder; or (d) has, or has a direct or indirect parent company that has, (i) become the subject of an insolvency proceeding or taken any action in furtherance thereof or (ii) become the subject of a Bail-In Action; provided, however, that a Lender shall not be a Defaulting Lender solely by virtue of a Governmental Authority's ownership of any equity interest in such Lender or parent company.

"<u>Deposit Account</u>" has the meaning assigned thereto in Article 9 of the UCC and, with respect to any Canadian Guarantor, means any demand, time, savings, checking, passbook, deposit, collection, lock-box or other similar account maintained with any financial institution.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 47 of 432

"<u>Deposit Account Control Agreement</u>" means a Deposit Account control agreement to be executed by each institution maintaining a Deposit Account (other than an Excluded Deposit Account) for the Borrower or any other Loan Party, in each case as required by and in accordance with the terms of <u>Section 6.20</u>.

"<u>Designated Jurisdiction</u>" means any country or territory to the extent that such country or territory itself is the subject of any Sanction.

"Designated Lenders" means the collective reference to (a) any Defaulting Lenders and (b) any Impacted Lenders with respect to which the Borrower has been required to provide cash collateral or other credit support to the Issuing Bank or the Swingline Lender, as the case may be, or otherwise as a result of which the Borrower cannot obtain a Letter of Credit or a Swingline Loan hereunder.

"<u>Dilution Reserve</u>" means a reserve in an amount equal to the sum of the following: (a) the aggregate amount of rebate accruals described on <u>Schedule 1.04</u> attached hereto established by the Borrower from time to time and reflected on the most recent balance sheet of the Borrower and its Subsidiaries; and (b) an additional amount determined by the Administrative Agent in its Permitted Discretion from time to time.

"<u>DIP Term Agent</u>" means Wilmington Savings Fund Society, FSB, in its capacity as administrative agent under the DIP Term Documents.

<u>"DIP Term Credit Agreement" means the Senior Secured Super-Priority Priming Debtor-in-Possession</u> <u>Credit Agreement, dated as of the Ratification Effective Date, the Borrower, Holdings, the other Guarantors and the</u> <u>DIP Term Agent, as the same may be amended, amended and restated, supplemented or otherwise modified from</u> <u>time to time.</u>

<u>"DIP Term Documents" means the DIP Term Credit Agreement and the other "Loan Documents" (as such term is defined in the DIP Term Credit Agreement).</u>

<u>"DIP Term Intercreditor Agreement" means the provisions of the Financing Orders governing the priority</u> of the security interests in the Collateral in favor of the Administrative Agent and the other Secured Parties, on the one hand and the DIP Term Agent and the lenders parties to the DIP Term Credit Agreement, on the other hand.

"DIP Term Loans" means the term loans borrowed under the DIP Term Credit Agreement.

<u>"Disposition</u>" or "<u>Dispose</u>" means the sale, transfer, license, lease or other disposition (including any sale and leaseback transaction) of any real or personal property by any Person (or the granting of any option or other right to do any of the foregoing), including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"Distress Event" means, with respect to any Person (each, a "Distressed Person"), (i) a voluntary or involuntary case (or comparable proceeding) has been commenced with respect to such Person under any Debtor Relief Law, (ii) a custodian, conservator, receiver, national receiver, trustee, monitor, liquidator, administrator, interim receiver or similar official has been appointed for such Person or for any substantial part of such Person's assets, or (iii) such Person has made a general assignment for the benefit of creditors or has otherwise been adjudicated as, or determined by any Governmental Authority having regulatory authority over such Person or its assets to be, insolvent or bankrupt; provided, that a Lender shall not be a Distressed Person solely as the result of the acquisition or ownership of an equity interest in such Lender or any Person controlling such Lender or the exercise of control over a Lender or any Person controlling such Lender by a Governmental Authority from the jurisdiction of the courts of the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lenders.

"Distressed Person" has the meaning specified in the definition of "Distress Event."

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 48 of 432

"Distribution Conditions" means as to any relevant action contemplated in this Agreement, (i) no Event of Default has then occurred and is continuing or would result from such action and (ii) either (1) (x)(a) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 17.5% of the Aggregate Commitments and (b) over the 30 consecutive days prior to consummation of such action, Availability on a Pro-Forma Basis for such action averaged no less than 17.5% of the Aggregate Commitments and (y) the Consolidated Fixed Charge Coverage Ratio would be at least 1.0 to 1.0 on a Pro-Forma Basis for such action or (2) (x)-Availability on a Pro-Forma Basis immediately after giving effect to such action would be at least 25.0% of the Aggregate Commitments and (y) over the 60 consecutive days prior to consummation of such action, Availability ona Pro-Forma Basis for such action averaged no less than 25.0% of the Aggregate Commitments.

"Dollar" and "\$" mean lawful money of the United States.

"Dollar Equivalent" means on any date, with respect to any amount denominated in Dollars, such amount in Dollars, and with respect to any stated amount denominated in a currency other than Dollars, the equivalent amount thereof in Dollars as determined by the Administrative Agent (which determination shall be conclusive and binding absent manifest error) pursuant to Sections 1.07 and 1.08.

"Domestic Subsidiary" means any Subsidiary that is organized under the laws of the United States, any state thereof or the District of Columbia.

"Dominion Account" means a special concentration account established by the Borrower at Bank of America, N.A. or another bank reasonably acceptable to the Administrative Agent over which the Administrative Agent has exclusive control for withdrawal purposes pursuant to the terms and provisions of this Agreement and the other Loan Documents.

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in <u>clause (a)</u> of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in <u>clauses (a)</u> or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"<u>EEA Resolution Authority</u>" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"<u>EU Bail-In Legislation Schedule</u>" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"Effective Yield" means, as to any Revolving Loans, the effective yield on such Revolving Loans as determined by the Administrative Agent, taking into account the applicable interest rate margins, any interest rate floors or similar devices and all fees, including upfront or similar fees or original issue discount (amortized over the shorter of (x) the Weighted Average Life to Maturity of such Revolving Loans and (y) the four years following the date of incurrence thereof) payable generally to Lenders making such Revolving Loans, but excluding any arrangement, structuring or other fees payable in connection therewith that are not generally shared with the relevant Lenders and customary consent fees paid generally to consenting Lenders.

"Eligible Accounts" means an Account owing to any Loan Party that arises in the ordinary course of business from the sale of goods, is payable in Dollars or Canadian Dollars (or other currencies acceptable to the Administrative Agent) and is deemed by the Administrative Agent, in its Permitted Discretion, to be an Eligible Account. Without limiting the foregoing, no Account shall be an Eligible Account if: (a) it is unpaid for more than 60 days after the original due date, or more than 90 days after the original invoice date; <u>provided</u>, <u>that</u>, notwithstanding the foregoing, up to \$1,000,000 in the aggregate at any time outstanding of Accounts on <u>Schedule</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 49 of 432

1.05, which may remain unpaid up to 120 days after the original invoice date, will not be considered ineligible under this clause (a): (b) 50% or more of the Accounts owing by the Account Debtor are not Eligible Accounts: (c) when aggregated with other Accounts owing by the Account Debtor and its Affiliates, it exceeds 15% of the aggregate Eligible Accounts (or such higher percentage as the Administrative Agent may establish for the Account Debtor from time to time); (d) it is owing by a creditor or supplier, or is otherwise subject to a potential offset, counterclaim, dispute, deduction, discount, recoupment, reserve, defense, chargeback, credit or allowance (but ineligibility shall be limited to the amount thereof); (e) an insolvency proceeding has been commenced by or against the Account Debtor; or the Account Debtor has suspended or ceased doing business, is liquidating, dissolving or winding up its affairs, is not Solvent, or is subject to any country sanctions program or specially designated nationals list maintained by the Office of Foreign Assets Control of the U.S. Treasury Department, or the Borrower is not able to bring suit or enforce remedies against the Account Debtor through judicial process; (f) the Account Debtor is organized or has its principal offices or assets outside the United States or Canada, unless the Account is supported by a letter of credit (delivered to and directly drawable by the Administrative Agent) or credit insurance satisfactory in all respects to the Administrative Agent; (g) it is owing by a Governmental Authority, unless the Account Debtor is the United States or Canada or any department, agency or instrumentality thereof and the Account has been assigned to the Administrative Agent in compliance with the federal Assignment of Claims Act, the Financial Administration Act (Canada) or other applicable Law; (h) it is not subject to a duly perfected, first priority Lien in favor of the Administrative Agent, or is subject to any other Lien (other than Liens permitted under Section 7.01); (i) the goods giving rise to it have not been delivered to the Account Debtor, or it otherwise does not represent a final sale; (j) it is evidenced by Chattel Paper or an Instrument of any kind that has not been delivered to the Administrative Agent, or has been reduced to judgment; (k) it arises from a sale to an Affiliate, from a sale on a cash-on-delivery, sale-or-return, sale-on-approval, consignment, or other repurchase or return basis, or from a sale for personal, family or household purposes; (1) it arises from a sale on a bill-and-hold (unless supported by bill-and-hold letter reasonably acceptable to the Administrative Agent): (m) is subject to offset by customer deposit: (n) it includes a billing for interest, fees or late charges, but ineligibility shall be limited to the extent thereof; or (o) it is unbilled; provided that notwithstanding the foregoing, up to \$12,000,000 in the aggregate at any time outstanding of unbilled Accounts will not be considered ineligible under this clause (o) so long as (i) such unbilled Accounts have a minimum account size of \$200,000 each, (ii) the service period covered by the applicable invoice for such unbilled Accounts does not exceed 31 days and (iii) such unbilled Accounts are billed within 14 days of the end of the relevant service period covered by the applicable invoice. In calculating delinquent portions of Accounts under clauses (a) and (b), credit balances more than 90 days old will be excluded.

"<u>Eligible Assignee</u>" means any Person that meets the requirements to be an assignee under <u>Sections</u> <u>11.06(b)(iii)</u>, (v) and (vi) (subject to such consents, if any, as may be required under <u>Section 11.06(b)(iii)</u>).

"Eligible Inventory" means Inventory owned by any Loan Party that the Administrative Agent, in its Permitted Discretion, deems to be Eligible Inventory, Without limiting the foregoing, no Inventory shall be Eligible Inventory unless it (a) is finished goods or raw materials, and not work-in-process, packaging or shipping materials, labels, samples, display items, bags, replacement parts or manufacturing supplies, provided that work-in-process, packaging or shipping materials, labels, samples, display items, bags, replacement parts, and manufacturing supplies shall not be deemed ineligible under this clause (a) to the extent the most recent inventory appraisal delivered to the Administrative Agent ascribes a value to such work-in-process, packaging or shipping materials, labels, samples, display items, bags, replacement parts, and manufacturing supplies; (b) is not held on consignment, nor subject to any deposit or down payment; (c) is in new and saleable condition and is not damaged, defective, shopworn or otherwise unfit for sale: (d) is not slow-moving, perishable, obsolete or unmerchantable, and does not constitute returned or repossessed goods, provided that slow-moving or obsolete inventory shall not be deemed ineligible under this clause (d) to the extent the most recent inventory appraisal delivered to the Administrative Agent ascribes a value to such slow moving or obsolete inventory; (e) meets all standards imposed by any Governmental Authority, and does not constitute Hazardous Materials; (f) conforms in all material respects with the covenants and representations herein; (g) is subject to the Administrative Agent's duly perfected, first priority Lien, and no other Lien except for Liens in favor of the Fixed Asset Collateral Agent as permitted by the Term Intercreditor. Agreement and the First Lien Intercreditor Agreement and other Liens permitted under Section 7.01; (h) is within the continental United States or Canada and is not consigned to any Person; (i) is not bill & hold inventory; (j) is not subject to any warehouse receipt or negotiable document; (k) is not subject to any license or other property or property right or other arrangement that restricts the applicable Loan Party's or the Administrative Agent's right to dispose of such Inventory, unless the Administrative Agent has received an appropriate Lien Waiver or is otherwise

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 50 of 432

satisfied that it could sell such inventory on favorable terms following a Default; (l) is not located on leased premises or in the possession of a warehouseman, processor, repairman or mechanic, unless the lessor or such Person is an Approved Processor and (i) has delivered a Landlord Lien Waiver and Access Agreement or (ii) an appropriate Rent and Charges Reserve has been established for such location; and (l) is reflected in the details of a current perpetual inventory report.

"<u>Embargoed Person</u>" means any party that (i) is publicly identified on the most current list of "Specially Designated Nationals and Blocked Persons" published by the U.S. Treasury Department's Office of Foreign Assets Control ("<u>OFAC</u>") or (ii) resides, is organized or chartered, or has a place of business in a country or territory subject to OFAC sanctions programs.

"<u>Environment</u>" means ambient air, indoor air, surface water, groundwater, drinking water, soil, surface and subsurface strata, and natural resources such as wetlands, flora and fauna.

"<u>Environmental Laws</u>" means any and all applicable Laws relating to pollution or the protection of the Environment or human health (to the extent related to exposure to Hazardous Materials or the Release into the Environment of Hazardous Materials).

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of the Borrower, any other Loan Party or any of their respective Subsidiaries directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the Release or threatened Release of any Hazardous Materials into the Environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"Environmental Permit" means any permit, approval, identification number, license or other authorization required under any Environmental Law.

<u>"Equipment Loans</u>" means those certain senior unsecured equipment loans owing by Cadmus Journal Services, Inc. (and Guaranteed by the Borrower, as successor to Cadmus Communications Corporation) and set forthon <u>Schedule 7.02</u> in an aggregate amount not to exceed \$1,300,000.

"Equity Interests" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests in such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"<u>ERISA Affiliate</u>" means any trade or business (whether or not incorporated) under common control with the Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"ERISA Event" means (a) a Reportable Event with respect to a Pension Plan; (b) a withdrawal by the Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA; (c) a complete or partial withdrawal by the Borrower, Holdings or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization; (d) the filing of a notice of intent to terminate, the treatment of a Pension Plan amendment as a termination under Section 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 51 of 432

terminate a Pension Plan or Multiemployer Plan; (e) an event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; or (f) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon the Borrower, Holdings or any ERISA Affiliate.

"Eurodollar Rate" means:

(a) for any Interest Period with respect to a Eurodollar Rate Loan, the rate per annum equal to (i) the London Interbank Offered Rate ("<u>LIBOR</u>") or a comparable or successor rate, which rate is approved by the Administrative Agent, as published on the applicable Bloomberg screen page (or such other commercially available source providing such quotations as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two London Banking Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such rate is not available at such time for any reason, then the "Eurodollar Rate" for such Interest Period shall be the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Rate Loan being made, continued or converted by Bank of America and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two London Banking Days prior to the commencement of such Interest Period; and

(b) for any interest calculation with respect to a Base Rate Loan on any date, the rate per annum equal to (i) LIBOR, at approximately 11:00 a.m., London time determined two London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in same day funds in the approximate amount of the Base Rate Loan being made or maintained and with a term equal to one month would be offered by Bank of America's London Branch to major banks in the London interbank Eurodollar market at their request at the date and time of determination;

<u>provided</u> that to the extent a comparable or successor rate is approved by the Administrative Agent in connection herewith, the approved rate shall be applied in a manner consistent with market practice; <u>provided</u>, <u>further</u>, that to the extent such market practice is not administratively feasible for the Administrative Agent, such approved rate shall be applied in a manner as otherwise reasonably determined by the Administrative Agent; <u>provided</u>, <u>further</u>, that to the extent any such rate determined pursuant to this definition is below zero, the Eurodollar Rate shall be deemed to be zero" immediately before the period at the end of such definition.

"Eurodollar Rate Loan" means a Revolving Loan that bears interest at a rate based on the Eurodollar Rate.

"Eurodollar Reserve Percentage" of any Lender for each Interest Period for any Eurodollar Rate Loan means the reserve percentage applicable to such Lender during such Interest Period (or if more than one such percentage shall be so applicable, the daily average of such percentages for those days in such Interest Period during which any such percentage shall be so applicable) under Regulation D or other regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) then applicable to such Lender with respect to liabilities or assets consisting of or including "Eurocurrency liabilities" (as defined in Regulation D of the FRB) having a term equal to such Interest Period.

"Event of Default" has the meaning specified in Section 8.01.

"<u>Exchange Rate</u>" means, on any day with respect to any currency other than Dollars, the exchange rate reported by Bloomberg (or other commercially available source designated by the Administrative Agent) as of the end of the preceding Business Day in the financial market for such currency.

"Exchange Rate Date" has the meaning set forth in Section 1.08.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 52 of 432

"Excluded Deposit Account" means a Deposit Account (i) which is used for the sole purpose of making payroll and withholding tax payments related thereto and other employee wage and benefit payments and accrued and unpaid employee compensation (including salaries, wages, benefits and expense reimbursements), (ii) which is used for paying taxes, including sales taxes, (iii) which is used as an escrow account or as a fiduciary or trust account, (iv) which is a zero balance Deposit Account, or (v) with an average monthly balance of less than \$100,000, not to exceed \$1,000,000 in the aggregate at any time for all Deposit Accounts that are Excluded Deposit Accounts pursuant to this clause (v).

<u>"Excluded Property" has the meaning set forth in the Security Agreement and shall include any</u> Equity Interests in any Person owned by a Loan Party that does not constitute "Pledged Equity" pursuant to the provision in Section 2.01(a) of the Security Agreement.

"Excluded Swap Obligation" means, with respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the Guaranty of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any Guaranty thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act (determined after giving effect to Section 10.11 of this Agreement and Section 5.15 of the Guaranty and any other "keepwell, support or other agreement" for the benefit of such Guarantor of a security interest, becomes effective with respect to such Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such Guaranty or security interest is or becomes excluded in accordance with the first sentence of this definition.

"Excluded Taxes" means, with respect to the Administrative Agent, any Lender or any other recipient of any payment to be made by or on account of any Obligation of the Borrower or any Guarantor hereunder, (a) Taxes imposed on or measured by its net income (however denominated) and franchise Taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Office is located, or by any other jurisdiction in which it is subject to such Taxes as a result of a connection with such jurisdiction unrelated to the Loan Documents, (b) any branch profits Taxes imposed under Section 884(a) of the Code, or any similar Tax imposed by any jurisdiction described in clause (a), (c) in the case of a Foreign Lender (other than an assignee pursuant to a request by the Borrower under Section 11.13), any United States federal withholding Tax that is imposed on amounts payable to such Foreign Lender pursuant to Laws in effect at the time such Foreign Lender (or its assignor, if any) was entitled, immediately prior to the designation of a new Lending Office (or assignment), to receive additional amounts from a Loan Party with respect to such withholding Tax pursuant to Section 3.01(a), (d) any Tax attributable to a failure of a Lender to comply with Section 3.01(e) and (e) any U.S. federal withholding Taxes imposed under FATCA.

"<u>Existing Credit Agreement</u>" means that certain Amended and Restated Credit Agreement, dated as of December 21, 2010, among the Borrower, Holdings, Bank of America, N.A., as Administrative Agent, Swing Line-Lender and L/C Issuer, and a syndicate of lenders, as amended, supplemented or otherwise modified from time totime and in effect on the date hereof.

"Existing Letters of Credit" means those Letters of Credit issued under the Existing Credit Agreement described on Schedule 2.13 hereto.

"Existing Revolving Loans" has the meaning specified in Section 2.19(a).

"Extended Revolving Loans" has the meaning specified in Section 2.19(a).

"<u>Extended Revolving Loan Commitment</u>" means the commitment of any Lender, established pursuant to-Section 2.15, to make Extended Revolving Loans to the Borrower.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 53 of 432

"Extending Lender" has the meaning specified in Section 2.19(c).

"Extension" has the meaning provided in Section 2.19(e).

"Extension Amendment" has the meaning provided in Section 2.19(d).

"Extension Election" has the meaning provided in Section 2.19(c).

"Extension Request" has the meaning provided in Section 2.19(c).

"Extension Series" has the meaning provided in Section 2.19(a).

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code as of the date hereof (and any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or other official interpretations thereof (including any Revenue Ruling, Revenue Procedure, Notice or similar guidance issued by the IRS under such provisions), any agreements entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the Code.

"Federal Funds Rate" means, for any day, the rate calculated by the Federal Reserve Bank of New York based on such day's federal funds transactions by depository institutions (as determined in such manner as the Federal Reserve Bank of New York shall set forth on its public website from time to time) and published on the next succeeding Business Day by the Federal Reserve Bank of New York as the effective federal funds rate; <u>provided</u> that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"<u>Fee Letter Letters</u>" means the letter agreement, dated April 2, 2013, collectively, (a) the Agent Fee Letter, dated the Ratification Effective Date, between the Borrower and the Administrative Agent and (b) the Lenders Fee Letter, dated the Ratification Effective Date, among Borrower and the Lenders party thereto.

"Fees" means all amounts payable pursuant to or referred to in Section 2.05.

"<u>Financial Covenant</u>" means<u>Final Financing Order</u>" has the covenant<u>meaning</u> set forth in <u>Section 7.11</u>.the <u>Ratification Agreement</u>.

"Financing Orders" has the meaning set forth in the Ratification Agreement.

<u>"First Lien Intercreditor Agreement</u>" means that certain Intercreditor Agreement, dated as of June 26, 2014, among the Borrower, Holdings, the other Guarantors, Bank of America, N.A., as administrative agent for the holdersof the "Revolving Credit Obligations" (as defined therein) the Administrative Agent for the benefit of the Secured <u>Parties</u>, and The Bank of New York Mellon, as collateral agent for the holders of the <u>4%</u> Notes, the holders of the <u>6%</u> <u>Notes and the holders of the other</u> "Fixed Asset Obligations" (as defined therein), as the same may be amended, amended and restated, supplemented or otherwise modified from time to time.

<u>"First Lien Notes</u>" means the 6.000% Senior Priority Secured Notes due 2019 issued by the Borrowerpursuant to the First Lien Notes Indenture.

"<u>First Lien Notes Indenture</u>" means the Indenture, dated as of June 26, 2014, among the Borrower, the guarantors party thereto and The Bank of New York Mellon, in its capacities as trustee and as collateral agent, which provides for the issuance of the First Lien Notes.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 54 of 432

"<u>Fixed Asset Collateral Agent</u>" has the meaning assigned to such term in the <u>TermFirst Lien Intercreditor</u> <u>Agreement.</u>

<u>"Fixed Asset Facility Collateral" has the meaning assigned to such term in the First Lien</u> Intercreditor Agreement.

"<u>Flood Insurance Laws</u>" means, collectively, (i) the National Flood Insurance Act of 1968, (ii) the Flood Disaster Protection Act of 1973, (iii) the National Flood Insurance Reform Act of 1994, (iv) the Flood Insurance Reform Act of 2004 and (v) the Biggert-Waters Flood Insurance Reform Act of 2012, or, in each case, any successor statute thereto.

"<u>Foreign Lender</u>" means <u>the Administrative Agent and any Lender, in each case</u>, that is not a <u>U.S.</u> <u>Person</u>"<u>United States person</u>" within the meaning of Section 7701(a)(30) of the Code.

"<u>Foreign Plan</u>" means any defined benefit plan (within the meaning of Section 3(35) of ERISA), other than any Canadian Benefit Plan or Canadian Pension Plan, established or maintained by Borrower, Holdings or any Subsidiary (other than a Plan or Multiemployer Plan) for the benefit of employees employed outside the United States.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"<u>Fronting Exposure</u>" means a Defaulting Lender's Pro Rata Share of LC Exposure or Swingline Loans, as applicable, except to the extent allocated to other Lenders under <u>Section 2.11</u>.

"Fronting Fee" has the meaning specified in Section 2.05(c).

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"<u>GAAP</u>" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"<u>Governmental Authority</u>" means the government of the United States, Canada or any other nation, or of any political subdivision thereof, whether state, provincial, territorial, local, municipal or otherwise, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Guarantee" means, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person. The amount of any Guarantee shall be deemed to be an amount equal to the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 55 of 432

stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "<u>Guarantee</u>" as a verb has a corresponding meaning.

"<u>Guarantors</u>" means, collectively (a) Holdings, (b) the Subsidiaries listed on <u>Schedule 6.12</u> and each other Subsidiary of the Borrower that shall be required to execute and deliver a guaranty or guaranty supplement pursuant to <u>Section 6.12</u> and (c) with respect to (i) Obligations owing by any Loan Party or any Subsidiary of a Loan Party (in each case, other than the Borrower) under any Secured Bank Product Obligations and (ii) the payment and performance by each Specified Loan Party (other than the Borrower) of its obligations under its Guaranty with respect to all Swap Obligations, the Borrower (and including its or their respective successors and assigns, including any trustee or other fiduciary hereafter appointed as its legal representative or with respect to the property of the estate of such entity whether under Chapter 11 of the Bankruptcy Code or any subsequent Chapter 7 case and its successor upon conclusion of the Chapter 11 Case of such entity).

"<u>Guaranty</u>" means, collectively, (i) the Guaranty made by Holdings under <u>Article X</u> in favor of the Secured Parties, (ii) the Guaranty made by the Guarantors (other than Cenveo Canada) in favor of the Secured Parties, substantially in the form of <u>Exhibit E-1</u>, and (iii) the Canadian Guaranty made by Cenveo Canada and each other Canadian Subsidiary of the Borrower in favor of the Secured Parties, substantially in the form of <u>Exhibit E-2</u>, in each case, together with each other guaranty and guaranty supplement delivered pursuant to <u>Section 6.12</u>.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes and all hazardous or toxic substances or wastes, including petroleum or petroleum distillates, asbestos or asbestos-containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other materials, substances, pollutants, contaminants or wastes of any nature in any form regulated pursuant to any Environmental Law.

"Holdings" has the meaning specified in the introductory paragraph hereto.

"Impacted Lender" means any Revolving Lender (a) that has given written notice to the Borrower, the Administrative Agent, the Swingline Lender, the Issuing Bank or any Lender or has otherwise publicly announced that such Revolving Lender believes it will become, or that fails following inquiry promptly to provide to the Borrower, the Administrative Agent, the Swingline Lender or the Issuing Bank making such inquiry reasonably satisfactory assurance that such Revolving Lender will not become, a Defaulting Lender, (b) as to which the Borrower, the Administrative Agent, the Swingline Lender or the Issuing Bank has a good faith belief that such Revolving Lender has defaulted more than once in fulfilling its funding obligations (as a lender, letter of credit issuer or issuer of bank guarantees and including, but not limited to, funding or paying when due loan requests, swing line participations, letter of credit participations, pro rata sharing obligations and expense and indemnification obligations) under any other syndicated credit facility and such Revolving Lender shall not have provided assurances satisfactory to the Borrower, the Administrative Agent, the Swingline Lender and the Issuing Bank that despite such defaults such Revolving Lender will not become a Defaulting Lender hereunder, (c) as to which a Distress Event has occurred with respect to any Affiliate of such Revolving Lender that directly or indirectly controls such Revolving Lender or (d) as to which such Revolving Lender or any Affiliate of such Revolving Lender that directly or indirectly controls such Revolving Lender has consummated or entered into a commitment to consummate a forced (in the good faith judgment of the Administrative Agent) liquidation, merger, sale of assets or other transaction resulting, in the good faith judgment of the Administrative Agent, in a change of ownership or operating control of such Person supported in whole or in part by guaranties, assumption of liabilities or other comparable credit support of (including without limitation the nationalization or assumption of ownership or operating control by) any Governmental Authority and the Administrative Agent (in its good faith judgment) or the Required Lenders (in their respective good faith judgment) believe that such event increases the risk that such Revolving Lender could default in performing its obligations hereunder for so long as the Administrative Agent (in its good faith judgment) or the Required Lenders (in their respective good faith judgment) so believe (it being understood and agreed that the Administrative Agent shall make any determination under this clause (d) promptly after being requested to do so by the Borrower or any Lender); provided, that no Revolving Lender shall be an Impacted Lender solely as the result of the acquisition or ownership of an equity interest in such Revolving Lender or any Person controlling such Revolving Lender or the exercise of control over such Revolving Lender or any Person controlling such Revolving Lender by a Governmental Authority or instrumentality thereof.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 56 of 432

"Increase Loan Lender" has the meaning specified in Section 2.15(b).

"Increase Date" has the meaning specified in Section 2.15(b).

"Increased Inspection Rights Period" means the occurrence of a date when (a) Availability shall have been less than 20.0% of the Aggregate Commitments\$35,000,000 or (b) a Default or Event of Default exists, in each case, until such date as (b) Availability shall have been at least equal to 20.0% of the Aggregate Commitments for 60none of the conditions in the foregoing clauses (a) or (b) exists for 30 consecutive calendar days.

"Increased Reporting Period" means the occurrence of a date when (a) Availability shall have been less than the greater of (i) 10.0% of the Aggregate Commitments and (ii) \$20,000,000, in either case, until such dateas\$25,000,000 at any time during the period commencing on the Ratification Effective Date and ending as of February 16, 2018, (b) Availability shall have been at least equal to the greater of (i) 10.0% of the Aggregate Commitments and (ii) \$20,000,000 less than \$35,000,000 at any time after February 16, 2018 or (c) an Event of Default exists, in each case until such date as none of the conditions in the foregoing clauses (a), (b) or (c) exists for 30 consecutive calendar days.

"Incremental Revolving Commitment Agreement" has the meaning provided in Section 2.15(d).

"<u>Indebtedness</u>" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

(a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;

(b) the maximum amount of all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;

(c) net obligations of such Person under any Swap Contract;

(d) all obligations of such Person to pay the deferred purchase price of property or services (other than trade accounts payable <u>and accrued intercompany accounts payable consistent with past practice</u> and, in each case, in the ordinary course of business and not past due for more than 90 days after the date on which such trade account was created);

(e) indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;

(f) all Attributable Indebtedness in respect of Capitalized Leases and Synthetic Lease Obligations of such Person and all Synthetic Debt of such Person;

(g) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Equity Interest in such Person or any Affiliate of such Person or any warrant, right or option to acquire such Equity Interest, valued, in the case of a redeemable preferred interest, at the greater of its voluntary or involuntary liquidation preference <u>plus</u> accrued and unpaid dividends; and

(h) all Guarantees of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation limited liability company or similar legal entity) in which such Person is a general partner or a joint venturer, unless such Indebtedness is expressly made non-recourse

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 57 of 432

to such Person. The amount of any net obligations of any Person under any Swap Contract on any date shall be deemed to be the Swap Termination Value thereof due and payable by such Person as of such date.

"Indemnified Taxes" means all Taxes other than Excluded Taxes.

"Indemnitees" has the meaning specified in Section 11.04(b).

"Information" has the meaning specified in Section 11.07.

"Instrument" has the meaning provided in Article 9 of the UCC.

"Insurance Proceeds" means all insurance proceeds (other than business interruption insurance proceeds), damages, awards, claims and rights of action with respect to any casualty or other loss, damage or destruction of any real or personal property of Holdings or its Subsidiaries.

"Intellectual Property Security Agreement" has the meaning specified in Section 4.01(a)(v).

"<u>Intercreditor Agreement</u>" means that certain Intercreditor Agreement, dated as of February 5, 2010, as amended, among the Loan Parties, Bank of America, as First Lien Administrative Agent, Bank of America, as control agent, and the Second Lien Indenture Trustee.

"Interest Payment Date" means, (a) as to any Eurodollar Rate Loan, the last day of each Interest Period applicable to such Eurodollar Rate Loan and the Maturity Date; <u>provided</u>, <u>however</u>, that if any Interest Period for a Eurodollar Rate Loan exceeds three (3) months, the respective dates that fall every three (3) months after the beginning of such Interest Period shall also be Interest Payment Dates; and (b) as to any Base Rate Loan, the first Business Day of each January, April, July and October and the Maturity Date.

"Interest Period" means, as to any Borrowing of a Eurodollar Rate Loan, the period commencing on the date of such Borrowing or on the last day of the immediately preceding Interest Period applicable to such Borrowing, as applicable, and ending on the date that is two weeks thereafter (only if available by the ICE Benchmark Administration or the successor thereto if the ICE Benchmark Administration is no longer making a LIBOR rate available) or on the numerically corresponding day (or, if there is no numerically corresponding day, on the last day) in the calendar month that is one (1), two (2), three (3) or six (6) months thereafter, as the Borrower may elect, or the date any Borrowing of a Eurodollar Rate Loan is converted to a Borrowing of a Base Rate Loan in accordance with Section 2.08 or repaid or prepaid in accordance with Section 2.07 or Section 2.09; provided, that if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day. Interest shall accrue from and including the first day of an Interest Period to but excluding the last day of such Interest Period.

"Interim Financing Order" has the meaning set forth in the Ratification Agreement.

"Internal Control Event" means a material weakness in, or fraud that involves management or other employees who have a significant role in, Holdings' and its Subsidiaries' internal controls over financial reporting, in each case as described in the Securities Laws.

"Inventory" has the meaning as defined in the UCC (or, with respect to any Canadian Guarantor, the PPSA), including all goods intended for sale, lease, display or demonstration, all work in process, and all raw materials, and other materials and supplies of any kind that are or could be used in connection with the manufacture, printing, packing, shipping, advertising, sale, lease or furnishing of such goods, or otherwise used or consumed in a Loan Party's business.

"<u>Inventory Reserve</u>" means reserves established by the Administrative Agent in its Permitted Discretion to reflect factors that may negatively impact the value of Inventory, including change in salability, obsolescence, seasonality, theft, shrinkage, imbalance, change in composition or mix, markdowns and vendor chargebacks.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 58 of 432

"Investment" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of Equity Interests of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or interest in, another Person, or ($\subseteq_{\mathbb{C}}$) the purchase or other acquisition (in one transaction or a series of transactions) of assets of another Person that constitute a business unit or all or a substantial part of the business of such Person. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment.

"IP Rights" has the meaning specified in Section 5.18.

"IP Security Agreement Supplement" has the meaning specified in Section 1.03 of the Security Agreement.

"IRS" means the United States Internal Revenue Service.

"Issuing Bank" means, as the context may require, (a) Bank of America (including any branch thereof) with respect to Letters of Credit issued by it; (b) any other Lender that may become an Issuing Bank pursuant to <u>Sections</u> 2.13(i) and 2.13(k), with respect to Letters of Credit issued by such Lender; (c) with respect to the Existing Letters of Credit, the Lender which issued each such Letter of Credit; or (d) collectively, all of the foregoing.

"Joinder Agreement" means a Joinder Agreement executed and delivered in accordance with the provisions of <u>Section 6.12</u>, substantially in the form of <u>Exhibit J</u> hereto.

"Landlord Lien Waiver and Access Agreement" means a Landlord Lien Waiver and Access Agreement, in a form reasonably approved by the Administrative Agent.

"Laws" means, collectively, all international, foreign, federal, state, provincial, territorial, municipal and local statutes, treaties, rules, guidelines, regulations, ordinances, codes, executive orders and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"<u>LC Collateral Account</u>" means a collateral account in the form of a Deposit Account established and maintained by the Administrative Agent for the benefit of the Secured Parties, in accordance with the provisions of <u>Section 2.13</u>.

"<u>LC Commitment</u>" means the commitment of the Issuing Bank to issue Letters of Credit pursuant to <u>Section</u> 2.13.

"<u>LC Disbursement</u>" means a payment or disbursement made by the Issuing Bank pursuant to a Letter of Credit.

"<u>LC Documents</u>" means all documents, instruments and agreements delivered by the Borrower or any other Person to the Issuing Bank or the Administrative Agent in connection with any Letter of Credit.

"<u>LC Exposure</u>" means, at any time, the sum of (a) the aggregate undrawn amount of all outstanding Letters of Credit at such time plus (b) the aggregate principal amount of all LC Disbursements that have not yet been reimbursed at such time. The LC Exposure of any Revolving Lender at any time means its Pro Rata Percentage of the aggregate LC Exposure at such time.

"<u>LC Obligations</u>" means the sum (without duplication) of (a) all amounts owing by the Borrower for any drawings under Letters of Credit (including any bankers' acceptances or other payment obligations arising therefrom); and (b) the stated amount of all outstanding Letters of Credit.

"LC Participation Fee" has the meaning assigned to such term in Section 2.05(c)(i).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 59 of 432

"<u>LC Request</u>" means a request by the Borrower in accordance with the terms of <u>Section 2.13(b)</u> in form and substance satisfactory to the Issuing Bank.

"Lead Arrangers" means Bank of America, N.A., General Electric Capital Corporation, Barclays Bank-PLCJPMorgan Chase Bank, N.A. and Wells Fargo Bank, National Association, each in their respective capacity as joint lead arranger and joint book manager.

"Lender" has the meaning specified in the introductory paragraph hereto.

"<u>Lending Office</u>" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Borrower and the Administrative Agent.

"Letter of Credit" means any letters of credit issued or to be issued by an Issuing Bank for the account of the Borrower or any of its Subsidiaries pursuant to Section 2.13, including each Existing Letter of Credit.

"Letter of Credit Expiration Date" means the date which is five (5) Business Days prior to the Maturity Date.

"Lien" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), trust (constructive, deemed, statutory or otherwise), charge, or preference, priority or other security interest or preferential arrangement in the nature of a security interest of any kind or nature whatsoever (including any conditional sale or other title retention agreement, any easement, right of way or other encumbrance on title to real property, and any financing lease having substantially the same economic effect as any of the foregoing).

"<u>Line Cap</u>" means an amount that is equal to the lesser of (a) the Aggregate Commitments and (b) the then applicable Borrowing Base.

"Liquidity Event" means the occurrence of a date when <u>either (a)</u> Availability shall have been less than the greater of (i) 10.0% of the Aggregate Commitments and (ii) \$20,000,000, in either case, until such date as (b) Availability shall have been at least\$20,000,000 or (b) an Event of Default exists, in either case; provided, that, (i) to the extent that the Liquidity Event has occurred due to clause (a) of this definition, if Availability shall be equal to the greater of (i) 10.0% of the Aggregate Commitments and (ii) \$20,000,000 for greater than the applicable amount for at least 30 consecutive calendar days, the Liquidity Event shall no longer be deemed to exist or be continuing until such time as Availability may again be less than the amount in clause (a) of this definition and (ii) to the extent that the Liquidity Event has occurred due to clause (b) of this definition, if such Event of Default is cured or waived or otherwise no longer exists, the Liquidity Event shall no longer be deemed to exist or be continuing.

"Liquidity Notice" means a written notice delivered by the Administrative Agent at any time during a Liquidity Period to any bank or other depository at which any Deposit Account (other than any Excluded Deposit Account) is maintained directing such bank or other depository (a) to remit all funds in such Deposit Account to a Dominion Account, or in the case of a Dominion Account, to the Administrative Agent on a daily basis, (b) to cease following directions or instructions given to such bank or other depository by any Loan Party regarding the disbursement of funds from such Deposit Account (other than any Excluded Deposit Account), and (c) to follow all directions and instructions given to such bank or other depository by the Administrative Agent in each case, pursuant to the terms of any Deposit Account Control Agreement in place.

"<u>Liquidity Period</u>" means any period throughout which (a) a Liquidity Event has occurred and is continuingor (b) a Specified Event of Default has occurred and is continuing.

"Loan Documents" means, collectively, (a) this Agreement, (b) the Notes, (c) the Guaranty, (d) the Collateral Documents, (e) the Fee LetterLetters, (f) each agreement governing Secured Bank Product Obligations, (g) each Joinder Agreement, (h) Amendment No. 1, (i) Amendment No. 2, (j) Amendment No. 3, (k) the 2014Senior/Junior Intercreditor Agreement, (l) the First Lien Intercreditor Agreement, (m) Amendment No. 4, (n)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 60 of 432

the 2016 Intercreditor Agreement, (o) Amendment No. $\frac{5 \text{ and } (p5, (p) \text{ the DIP Term Intercreditor Agreement, (q) the Ratification Agreement, (r) the Financing Orders and (s) all other documents and agreements executed and delivered in connection with the Obligations hereunder.$

"Loan Parties" means, collectively, the Borrower and each Guarantor, and "Loan Party" means any one such Person.

"London Banking Day" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank Eurodollar market.

"<u>Material Adverse Effect</u>" means (a) a material adverse change in, or a material adverse effect upon, the operations, business, properties, liabilities (actual or contingent) or condition (financial or otherwise) of the Loan Parties taken as a whole; (b) a material impairment of the rights and remedies of the Administrative Agent or any Lender under any Loan Document, or of the ability of any Loan Party to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party (it being understood that the commencement of the Chapter 11 Cases, the events that customarily occur following the commencement of a proceeding under Chapter 11 of the Bankruptcy Code, any defaults under agreements that have no material effect on Borrower and Guarantors under the terms of the Bankruptcy Code as a result of the commencement of the Chapter 11 Cases, and reclamation claims, individually or collectively, shall not be deemed a Material Adverse Effect).

"<u>Material Contract</u>" means, with respect to any Loan Party, each contract or agreement listed on <u>Schedule</u> <u>1.02</u> hereto and each other contract or agreement entered into after the Closing Date by any Loan Party the breach or termination of which could reasonably be expected to have a Material Adverse Effect.

"<u>Maturity Date</u>" means June 10, 2021, as such date may be extended from time to time with respect to-Extended Revolving Loans pursuant to <u>Section 2.19</u>; provided that if the Borrower has not, prior to May 2, 2019, purchased, redeemed, defeased or otherwise refinanced the First Lien Notes, such that no more than \$10 million of the First Lien Notes remains outstanding, with cash on hand and/or proceeds of indebtedness, then the Maturity Dateshall instead be May 2, 2019. the earliest of: (a) the date that is nine (9) months after the commencement of the Petition Date, (b) forty (40) days (or such later date as the Administrative Agent and Required Lenders may agree) after the entry of the Interim Financing Order if the Final Financing Order has not been entered prior to the expiration of such forty (40)-day period, or (c) the "effective date" of a plan of reorganization filed in the Chapter 11 Cases that is acceptable to the Administrative Agent and the Lenders and that is confirmed pursuant to an order entered by the Bankruptcy Court.

"Maximum Cap Ex Amount" has the meaning specified in Section 7.12.

"Maximum Rate" has the meaning specified in Section 11.09.

"<u>Measurement Period</u>" means, at any date of determination, the most recently completed four fiscal quarters of the Borrower.<u>Milestones</u>" has the meaning set forth in the Ratification Agreement.

"Moody's" means Moody's Investors Service, Inc. and any successor thereto.

"Mortgage" has the meaning specified in Section 6.21(a)(i).

"Mortgage Policies" has the meaning specified in Section 6.21(c)(i).

"Mortgaged Property" means each parcel of real (immovable) property (together with all improvements, immovables and fixtures thereon and rights appurtenant thereto) required to be encumbered by a Mortgage or Canadian Mortgage pursuant to Section 6.12 or 6.20 hereof.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 61 of 432

"<u>Multiemployer Plan</u>" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA, to which the Borrower, Holdings or any ERISA Affiliate makes or is obligated to make contributions, or during the preceding five plan years, has made or been obligated to make contributions.

"<u>NOLV Percentage</u>" means the net orderly liquidation value of Eligible Inventory, expressed as a percentage, expected to be realized at an orderly, negotiated sale held within a reasonable period of time, net of all liquidation expenses, as determined from the most recent satisfactory inventory appraisals and field examination information of the Loan Parties' Eligible Inventory received by the Administrative Agent and performed by examiners and appraisers reasonably satisfactory to the Administrative Agent.

"<u>Non-Operating Subsidiary</u>" means any non-operating, inactive Subsidiary with assets having a net book value of less than \$100,000.

"<u>Note</u>" means a promissory note made by the Borrower in favor of a Revolving Lender or Swingline Lender, evidencing Revolving Loans or Swingline Loans, as applicable, made by such Revolving Lender or Swingline Lender, as the case may be, substantially in the form of <u>Exhibit B-1</u> or <u>Exhibit B-2</u>, as applicable.

"Notice of Borrowing" means a notice substantially in the form of Exhibit A-1 hereto.

"Notice of Conversion/Continuation" means a notice substantially in the form of Exhibit A-2 hereto.

"<u>Noticed Hedge</u>" means any Secured Bank Product Obligations arising under a Swap Contract with respect to which the Borrower and the Secured Bank Product Provider thereof have notified the Administrative Agent of the intent to include such Secured Bank Product Obligations as a Noticed Hedge hereunder and with respect to which a Bank Products Reserve has subsequently been established in the maximum amount thereof.

"NPL" means the National Priorities List under CERCLA.

"Obligations" means (x) all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party or any Affiliate thereof of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding and (y) Secured Bank Product Obligations, excluding, with respect to any Guarantor at any time, Excluded Swap Obligations (and including the Pre-Petition Obligations and the Post-Petition <u>Obligations</u>).

"OECD" means the Organization for Economic Cooperation and Development.

"OFAC" has the meaning specified in the definition of "Embargoed Person."

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; (c) with respect to any unlimited company, the memorandum of association and articles of association; and (d) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"<u>Other Taxes</u>" means all present or future stamp, court, filing, recording, intangible or documentary Taxes or any other excise or property Taxes arising from any payment made hereunder or under any other Loan Document or from the execution, delivery, performance, registration or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document, but shall not include Taxes imposed as a result of a transfer by a Lender or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 62 of 432

a Participant of any interest in a Loan Document ("<u>Assignment Taxes</u>"), other than Assignment Taxes imposed in respect of a transfer pursuant to <u>Section 11.13(a)</u>, if such Assignment Taxes are imposed as a result of a connection between the assignor or assignee and the taxing jurisdiction that is unrelated to the Loan Documents.

"Outstanding Amount" means (a) with respect to all Revolving Loans and Swingline Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of Revolving Loans and Swingline Loans occurring on such date and (b) with respect to any LC Obligations on any date, the amount of such LC Obligations on such date after giving effect to any LC Disbursement occurring on such date and any other changes in the aggregate amount of the LC Obligations as of such date, including as a result of any reimbursements by the Borrower of Unreimbursed Amounts.

"Overadvance" has the meaning specified in Section 2.17.

"<u>Overadvance Loan</u>" means a Base Rate Loan made when an Overadvance exists or is caused by the funding thereof.

"Participant" has the meaning specified in Section 11.06(d).

"Participant Register" has the meaning specified in Section 11.06(d).

"<u>Payment Office</u>" means the office of the Administrative Agent located at 101 S. Tryon Street, Charlotte, NC 28255-0001, Attention: Operations Manager, Telephone No.: 860-952-6829, Telecopier No.: 860-952-6830, or such other office as the Administrative Agent may hereafter designate in writing as such to the other parties hereto.

"Payor" has the meaning specified in Section 7.06(a).

"PBGC" means the Pension Benefit Guaranty Corporation.

"PCAOB" means the Public Company Accounting Oversight Board.

"Pension Event" means, solely with respect to a Canadian Pension Plan that contains a defined benefit provision, (a) the whole or partial withdrawal of a Canadian Guarantor from participation in a Canadian Pension Plan that is a Canadian Multi-Employer Pension Plan registered in Quebec during a plan year; or (b) the filing of a notice of proposal to terminate a Canadian Pension Plan or the treatment of a Canadian Pension Plan amendment as a termination; or (c) the issuance of a notice of proposal by any Governmental Authority to terminate or to appoint an administrator or like body appointed to administer a Canadian Pension Plan that is not a Canadian Multi-Employer Pension Plan; or (d) any other event or condition which constitutes grounds for the termination of any Canadian Pension Plan, to the extent any relevant Governmental Authority has so notified a Loan Party, unless such grounds are being contested by a Canadian Guarantor or any of its Subsidiaries in good faith or (e) except for any Canadian Pension Plan that is a Canadian Multi-Employer Pension Plan or any successor thereto, the establishment or participation by a Canadian Guarantor in a defined benefit pension plan on or after the Closing Date.

"<u>Pension Plan</u>" means any "employee pension benefit plan" (as such term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, that is subject to Title IV of ERISA and is sponsored or maintained by the Borrower, Holdings or any ERISA Affiliate or to which the Borrower, Holdings or any ERISA Affiliate contributes or has an obligation to contribute, or in the case of a multiple employer or other plan described in Section 4064(a) of ERISA, has made contributions at any time during the immediately preceding five plan years.

"<u>Perfection Certificate</u>" has the meaning given such term in <u>Section 1.03</u> of the Security Agreement or the Canadian Security Agreement, as applicable.

"<u>Permitted Discretion</u>" means reasonable credit judgment in accordance with customary business practices for comparable asset-based lending transactions, and as it relates to the establishment of reserves or the imposition of exclusionary criteria shall require that (x) the contributing factors to the imposition of any reserves shall not

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 63 of 432

duplicate (i) the exclusionary criteria set forth in the definitions of Eligible Accounts or Eligible Inventory as applicable (and vice versa) or (ii) any reserves deducted in computing book value and (y) the amount of any such reserve so established or the effect of any adjustment or imposition of exclusionary criteria be a reasonable quantification of the incremental dilution of the Borrowing Base attributable to such contributing factors.

"<u>Permitted Encumbrances</u>" has the meaning (if any) specified in the Mortgages and includes any Liens on any real property interests of any Loan Party permitted to be created, incurred or assumed, or otherwise permitted to exist, pursuant to the terms of any Mortgage.

"<u>Permitted Holders</u>" means Robert G. Burton, Sr., his immediate family and their heirs and estates and any Person (other than a natural Person and Holdings and its Subsidiaries) that is wholly-owned or otherwise directly Controlled by any of the foregoing, including, without limitation, Burton Capital Management, LLC.

"<u>Person</u>" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"Petition Date" means the date of the filing by the Loan Parties of their petitions to commence the Chapter <u>11 Cases.</u>

<u>"Plan</u>" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA), other than any Canadian Benefit Plan, Canadian Pension Plan, Canadian Multi-Employer Pension Plan, Multiemployer Plan or Foreign Plan, established or maintained by the Borrower, Holdings or any Subsidiary.

"Platform" has the meaning specified in Section 6.02.

"<u>Pledged Debt</u>" has the meaning specified in <u>Section 2.01</u> of the Security Agreement or the Canadian Security Agreement, as applicable.

"<u>Pledged Equity</u>" has the meaning specified in <u>Section 2.01</u> of the Security Agreement or the Canadian Security Agreement, as applicable.

"<u>PPSA</u>" means the *Personal Property Security Act* (Ontario) and the regulations thereunder; <u>provided</u>, <u>however</u>, if validity, perfection and effect of perfection and non-perfection of the Administrative Agent's Lien on any Collateral are governed by the personal property security laws of any Canadian jurisdiction other than Ontario, PPSA shall mean those personal property security laws (including the Civil Code of Quebec) in such other jurisdiction for the purposes of the provisions hereof relating to such validity, perfection and effect of perfection and non-perfection and for the definitions related to such provisions, as from time to time in effect.

"Prepayment Conditions" means, as to any relevant action contemplated in this Agreement, (i) no Event of Default has then occurred and is continuing or would result from such action and (ii) either (1) (x)(a) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 15.0% of the Aggregate Commitments and (b) over the 30 consecutive days prior to consummation of such action, Availability averaged on a Pro Forma Basis for such action no less than 15.0% of the Aggregate Commitments and (y) the Consolidated Fixed Charge Coverage Ratio would be at least 1.0 to 1.0 on a Pro Forma Basis for such action or (2) (x) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 20.0% of the Aggregate-Commitments and (y) over the 30 consecutive days prior to consummation of such action, Availability averaged on a Pro Forma Basis for such action no less than 20.0% of the Aggregate Commitments.

"Pro Forma Adjustment Compliance Certificate" means a certificate substantially in the form of Exhibit K.

"Pro Forma Basis" has the meaning specified in Section 1.03.

"<u>Pro Rata Percentage</u>" of any Revolving Lender at any time means the percentage of the Aggregate Commitments represented by such Lender's Revolving Commitment.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 64 of 432

"<u>Pro Rata Share</u>" means, with respect to each Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Aggregate Exposures of such Lender at such time and the denominator of which is the Aggregate Exposures at such time. The initial Pro Rata Shares of each Lender are set forth opposite the name of such Lender on <u>Schedule 2.01</u> or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"Protective Advances" has the meaning specified in Section 2.18.

"Public Lender" has the meaning specified in Section 6.02.

"Qualified Cash" means unrestricted cash of the Borrower and the Subsidiaries that (i) is subject to Deposit Account Control Agreements in form and substance reasonably satisfactory to the Administrative Agent (which will not prohibit withdrawal of such funds by the Borrower or such Subsidiaries in the absence of an Event of Default), (ii) does not include customer deposits or unapplied cash and (iii) does not otherwise constitute proceeds from the Disposition of any Revolver Priority Collateral.

"Qualified ECP Guarantor" shall mean, at any time, each Loan Party with total assets exceeding 10,000,000 or that qualifies at such time as an "eligible contract participant" under the Commodity Exchange Act and can cause another Person to qualify as an "eligible contract participant" at such time under 1a(18)(A)(v)(II) of the Commodity Exchange Act.

<u>"Ratification Agreement" means the Ratification and Amendment Agreement, dated as of the Ratification</u> Effective Date, by and among the Borrower, the Administrative Agent, the Lenders party thereto, the Issuing Bank and the other Loan Parties party thereto.

"Ratification Effective Date" means February 5, 2018.

"Register" has the meaning specified in Section 11.06(c).

"<u>Registered Public Accounting Firm</u>" has the meaning specified by the Securities Laws and shall be independent of Holdings as prescribed by the Securities Laws.

"<u>Related Documents</u>" means (a) the <u>Second Lien Notes Documents, (b) the Unsecured Term Loan Facility</u> <u>Documents, (c) the Term Documents and (d8.5% Notes Indenture and the 8.5% Notes, (b) the 4% Notes Indenture</u> and the 4% Notes, (c) 6% Notes Indenture and the 6% Notes, (d) the 6% Unsecured Notes Indenture and the 6% <u>Unsecured Notes and (e)</u> all Material Contracts.

"<u>Related Parties</u>" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"<u>Release</u>" means any release, spill, emission, discharge, deposit, disposal, leaking, pumping, pouring, dumping, emptying, injection or leaching into the Environment, or into, from or through any building, structure or facility.

"<u>Reportable Event</u>" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the 30 day notice period has been waived.

"<u>Required Lenders</u>" means, as of any date of determination, Lenders holding without duplication more than 50% of the Outstanding Amount and unused Revolving Commitments; <u>provided</u> that the portion of the Outstanding Amount and unused Revolving Commitments held or deemed held by any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"<u>Rent and Charges Reserve</u>" means the aggregate of (a) all past due rent and other amounts owing by a Loan Party to any landlord, warehouseman, processor, repairman, mechanic, shipper, freight forwarder, broker or other Person who possesses any Collateral or could assert a Lien on any Collateral; and (b) a reserve at least equal to

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 65 of 432

three (3) months' rent and other charges that could be payable to any such Person, unless it has executed a Landlord Lien Waiver and Access Agreement.

"Reserves" means the sum (without duplication) of (a) the Inventory Reserve; (b) the Rent and Charges Reserve; (c) reserves for accrued and unpaid Royalties, whether or not then due and payable; (d) the Bank Product Reserve; (e) to the extent that the Borrowing Base includes any Eligible Account and/or Eligible Inventory of a Canadian Guarantor, the Canadian Priority Payables Reserve; (f) the Dilution Reserve; (g) reserves for amounts owed by any Loan Party to any processor (including, without limitation, the Approved Processors); (h) the aggregate amount of liabilities secured by Liens upon Collateral included in the Borrowing Base that are senior to the Administrative Agent's Liens (but imposition of any such reserve shall not waive an Event of Default arising therefrom); (i) the Canadian Existing Lien Reserve; (j) the Senior Notes Stub Reserve; [reserved]; (k) the Carve-Out. Reserves, and (k]) such additional reserves, in such amounts and with respect to such matters, as the Administrative Agent in its Permitted Discretion may elect to impose from time to time.

Notwithstanding anything to the contrary in this Agreement, (i) such Reserves shall not be established or changed except upon not less than three (3) Business Days' prior written notice to the Borrower (or such shorterperiod as may be agreed by the Borrower), which notice shall include a reasonably detailed description of such-Reserve being established (during which period (a) the Administrative Agent shall, if requested, discuss any such-Reserve or change with the Borrower and (b) the Borrower may take such action as may be required so that the event, condition or matter that is the basis for such Reserve or change thereto no longer exists or exists in a manner that would result in the establishment of a lower Reserve or result in a lesser change thereto, in a manner and to the extent reasonably satisfactory to the Administrative Agent), and (ii) the amount of any Reserve established by the Administrative Agent, and any change in the amount of any Reserve or such change. Notwithstanding clause (i) of the preceding sentence, changes to the Reserves solely for purposes of correcting mathematical or clerical errors shallnot be subject to such notice period, it being understood that no Default or Event of Default shall be deemed to result therefrom, if applicable, for a period of three (3) Business Days, and with respect to the Carve-Out and Carve-Out Reserves, the maintenance, priority, terms of and funding of such Carve-Out and Carve-Out Reserves shall be set forth in the Financing Orders.

"<u>Responsible Officer</u>" means the chief executive officer, president, chief financial officer, treasurer, assistant treasurer or controller of a Loan Party. Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"Restricted Payment" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other Equity Interest of any Person or any of its Subsidiaries, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, defeasance, acquisition, cancellation or termination of any such capital stock or other Equity Interest, or on account of any return of capital to any Person's stockholders, partners or members (or the equivalent of any thereof), or any option, warrant or other right to acquire any such dividend or other distribution or payment.

"<u>Revolver Priority Collateral</u>" has the meaning assigned to the term "ABL Collateral" in the <u>TermFirst Lien</u> Intercreditor Agreement.

"<u>Revolving Availability Period</u>" means the period from and including the Closing Date to but excluding the earlier of the Maturity Date and the date of termination of the Revolving Commitments.

"Revolving Borrowing" means a Borrowing comprised of Revolving Loans.

"<u>Revolving Commitment</u>" means, with respect to each Lender, the commitment, if any, of such Lender to make Revolving Loans hereunder up to the amount set forth on <u>Schedule 2.01</u>, or in the Assignment and Assumption Agreement pursuant to which such Lender assumed its Revolving Commitment, as applicable, as the same may be

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 66 of 432

(a) reduced from time to time pursuant to <u>Section 2.07</u>, (b) increased from time to time pursuant to <u>Section 2.15</u> and (c2.07 and (b) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to <u>Section 11.06</u>. The aggregate amount of the Lenders' Revolving Commitments on the <u>Amendment No. 4Ratification</u> Effective Date is \$190,000,000.

"Revolving Commitment Increase" has the meaning specified in Section 2.15(a).

"Revolving Commitment Increase Notice" has the meaning specified in Section 2.15(b).

"<u>Revolving Exposure</u>" means, with respect to any Lender at any time, the aggregate principal amount at such time of all outstanding Revolving Loans of such Lender, plus the aggregate amount at such time of such Lender's LC Exposure, plus the aggregate amount at such of such Lender's Swingline Exposure.

"Revolving Lender" means a Lender with a Revolving Commitment.

"<u>Revolving Loan</u>" means a loan made pursuant to <u>Article II</u> hereof including, without duplication, any Swingline Loan (to the extent the context so requires the same), Overadvance Loan, and Protective Advance and Extended Revolving Loan.

"Revolving Note" means each revolving note substantially in the form of Exhibit B-1 hereto.

"<u>Royalties</u>" means all royalties, fees, expense reimbursement and other amounts payable by a Loan Party under a license.

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., and any successor thereto.

"<u>Sanction(s)</u>" means any international economic sanction administered or enforced by the United States Government (including without limitation, OFAC), the United Nations Security Council, the European Union, Her Majesty's Treasury or the Canadian Government.

"Sarbanes-Oxley" means the Sarbanes-Oxley Act of 2002.

"<u>SEC</u>" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Second Amended and Restated Credit Agreement" means that certain Second Amended and Restated Credit Agreement, dated as of the date hereof, among the Borrower, Holdings, Bank of America, N.A., as administrative agent, and a syndicate of lenders, as the same may be amended, restated, amended and restated, modified, supplemented, replaced or refinanced from time to time in accordance herewith and the Term Intercreditor-Agreement.

"<u>Second Lien Indenture Trustee</u>" means Wells Fargo Bank, National Association, in its capacity as trusteeunder the Second Lien Notes Indenture.

"<u>Second Lien Notes</u>" means the Senior Second Lien Notes due 2018 issued pursuant to the Second Lien Notes Documents.

"Second Lien Notes Documents" means the Second Lien Notes Indenture, the Second Lien Notes and allother agreements, instruments and other documents pursuant to which the Second Lien Notes have been or will beissued or otherwise setting forth the terms of the Second Lien Notes." Second Lien Notes Indenture" means the-Indenture, dated as of February 5, 2010, by and among the Borrower, the guarantors party thereto and the Second Lien Indenture Trustee.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 67 of 432

"Section 6.01 Financials" means the quarterly and annual financial statements required to be delivered pursuant to Sections 6.01(a) and (b).

"Secured Bank Product Obligations" means Bank Product Debt owing to a Secured Bank Product Provider, up to the maximum amount (in the case of any Secured Bank Product Provider other than Bank of America, N.A. and its Affiliates) specified by such provider in writing to the Administrative Agent, which amount may be established or increased (by further written notice by the Borrower to the Administrative Agent from time to time) as long as no Default or Event of Default then exists and no Overadvance would result from establishment of a Bank Product Reserve for such amount and all other Secured Bank Product Obligations.

"Secured Bank Product Provider" means, at the time of entry into a Bank Product (or, if such Bank Product exists on the Closing Date, as of the Closing Date), the Administrative Agent, any Lender or any of their respective Affiliates that is providing a Bank Product; provided such provider delivers written notice to the Administrative Agent, in form and substance satisfactory to the Administrative Agent, by the later of the Closing Date or ten (10) days following creation of the Bank Product, (i) describing the Bank Product and setting forth the maximum amount to be secured by the Collateral and the methodology to be used in calculating such amount, and (ii) agreeing to be bound by Section 9.11. It is hereby understood that a Person may not be a Secured Bank Product Provider to the extent it is similarly treated as such under the Second Amended and Restated Credit Agreement in respect of such Bank Product.

"Secured Parties" means, collectively, the Administrative Agent, the Lenders, the Secured Bank Product Providers, each co-agent or sub-agent appointed by the Administrative Agent from time to time pursuant to <u>Section</u> <u>9.05</u>, and the other Persons the Obligations owing to which are or are purported to be secured by the Collateral under the terms of the Collateral Documents.

"<u>Securities Laws</u>" means the Securities Act of 1933, the Securities Exchange Act of 1934, Sarbanes-Oxley, and the applicable accounting and auditing principles, rules, standards and practices promulgated, approved or incorporated by the SEC or the PCAOB.

"Security Agreement" has the meaning specified in Section 4.01(a)(iii).

"Security Agreement Supplement" has the meaning specified in Section 1.03 of the Security Agreement.

"Senior Notes" means the (i) 11.5% Senior Notes due 2017 issued pursuant to the Senior Notes Indentureand (ii) 7.0% Exchangeable Senior Notes due 2017 issued pursuant to the Senior Exchangeable Notes-Indenture./Junior Intercreditor Agreement" means that certain Intercreditor Agreement, dated as of June 10, 2016, among the Borrower, Holdings, the other Guarantors, Bank of America, N.A., as the Revolving Credit Collateral Agent (as defined therein) and The Bank of New York Mellon, as collateral agent for the holders of the 4% Notes and as the collateral agent for the holders of the Junior Priority Obligations (as defined therein).

"<u>Senior Notes Documents</u>" means the Senior Notes Indenture, the Senior Exchangeable Notes Indenture, the Senior Notes and all other agreements, instruments and other documents pursuant to which the Senior Notes have been or will be issued or otherwise setting forth the terms of the Senior Notes.

<u>"Senior Notes Indenture</u>" means the Indenture, dated as of March 28, 2012, among the Borrower, the guarantors party thereto and U.S. Bank National Association, as trustee, which provides for the issuance of the Borrower's 11.5% Senior Notes due 2017.

"<u>Senior Notes Stub Reserve</u>" means on each day during the period beginning on February 13, 2017 and ending upon the repayment of the Senior Notes, a reserve equal to the aggregate principal amount then outstandingunder the Senior Notes on such day.

"<u>Senior Exchangeable Notes Indenture</u>" means the Indenture, dated as of March 28, 2012, among the Borrower, the guarantors party thereto and U.S. Bank National Association, as trustee, which provides for the issuance of the Borrower's 7.0% Exchangeable Senior Notes due 2017.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 68 of 432

"Settlement Date" has the meaning specified in Section 2.14(b).

"Solvent" and "Solvency" mean, with respect to any Person on any date of determination, that on such date (a) the fair value of the property of such Person is greater than the total amount of liabilities, including contingent liabilities, of such Person, (b) the present fair salable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured, (c) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay such debts and liabilities as they mature, (d) such Person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such Person's property would constitute an unreasonably small capital, (e) such Person is able to pay its debts and liabilities, contingent obligations and other commitments as they mature in the ordinary course of business, and (f) as regards the Canadian Guarantors, none is an "insolvent person" as defined in the *Bankruptcy and Insolvency Act* (Canada). The amount of contingent liabilities at any time shall be computed as the amount that, in the light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"Specified Event of Default" means any Event of Default arising under Section 8.01(a), 8.01(b)(i) (solely relating to a failure to comply with Section 6.20(c)), 8.01(b)(iv) or 8.01(f).

"<u>Specified Loan Party</u>" means any Loan Party that is not an "eligible contract participant" under the Commodity Exchange Act (determined prior to giving effect to <u>Section 10.11</u> of this Agreement and <u>Section 5.15</u> of the Guaranty).

"Spot Rate" means the exchange rate, as determined by the Administrative Agent, that is applicable to conversion of one currency into another currency, which is (a) the exchange rate reported by Bloomberg (or other commercially available source designated by the Administrative Agent) as of the end of the preceding Business Day in the financial market for the first currency; or (b) if such report is unavailable for any reason, the spot rate for the purchase of the first currency with the second currency as in effect during the preceding Business Day in the Administrative Agent's principal foreign exchange trading office for the first currency as of the end of such preceding Business Day.

"Subsidiary" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a Subsidiary or Subsidiaries of Holdings.

"<u>Supermajority Lenders</u>" means those non-Defaulting Lenders which would constitute the Required Lenders under, and as defined in, this Agreement if the percentage "50%" contained therein were changed to "66²/₃%."

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement governing any transaction described in <u>clause (a)</u> above (any such master agreement, together with any related schedules, a "<u>Master Agreement</u>"), including any such obligations or liabilities under any Master Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 69 of 432

"<u>Swap Obligation</u>" means, with respect to any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"Swap Termination Value" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) for any date on or after the date such Swap Contracts have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in <u>clause (a)</u>, the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts (which may include a Lender or any Affiliate of a Lender).

"<u>Swingline Commitment</u>" means the commitment of the Swingline Lender to make loans pursuant to <u>Section 2.12</u>, as the same may be reduced from time to time pursuant to <u>Section 2.07</u> or <u>Section 2.12</u>.

"<u>Swingline Exposure</u>" means, at any time, the aggregate principal amount at such time of all outstanding Swingline Loans. The Swingline Exposure of any Revolving Lender at any time shall equal its Pro Rata Percentage of the aggregate Swingline Exposure at such time.

"Swingline Lender" means Bank of America, N.A.

"Swingline Loan" means any Revolving Loan made by the Swingline Lender pursuant to Section 2.12.

"Swingline Note" means each swingline note substantially in the form of Exhibit B-2 hereto.

"Synthetic Debt" means, with respect to any Person as of any date of determination thereof, all obligations of such Person in respect of transactions entered into by such Person that are intended to function primarily as a borrowing of funds (including any minority interest transactions that function primarily as a borrowing) but are not otherwise included in the definition of "Indebtedness" or as a liability on the consolidated balance sheet of such Person and its Subsidiaries in accordance with GAAP.

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property (including sale and leaseback transactions), in each case, creating obligations that do not appear on the balance sheet of such Person but which, upon the application of any Debtor Relief Laws to such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"Tax Status Certificate" has the meaning set forth in Section 3.01(e).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>Term Agent</u>" means Bank of America, N.A., in its capacity as administrative agent under the Term-Documents.

"<u>Term Documents</u>" means the Second Amended and Restated Credit Agreement and the other "Loan-Documents" (as such term is defined in the Second Amended and Restated Credit Agreement).

<u>"Term Intercreditor Agreement</u>" means the Intercreditor Agreement, dated as of the date hereof, among the Borrower, Holdings, the other Guarantors, the Administrative Agent and the Term Agent, as the same may be amended, amended and restated, supplemented or otherwise modified from time to time.

"Term Loans" means the term loans borrowed under the Second Amended and Restated Credit Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 70 of 432

"<u>Threshold Amount</u>" means \$35,000,000.5,000,000.

"Trade Date" has the meaning specified in Section 11.06(b)(i)(B).

"<u>Transaction</u>" means, collectively, (a) the entering into by the Loan Parties and their applicable Subsidiaries of the Loan Documents to which they are or are intended to be a party, (b) the entry into the Second Amended and Restated Credit Agreement, (c) the refinancing of the term loans under the Existing Credit Agreement and the termination of the revolving commitments under the Existing Credit Agreement and (d) the payment of the fees and expenses incurred in connection with the consummation of the foregoing.

"<u>Type</u>" means, with respect to a Revolving Loan, its character as a Base Rate Loan or a Eurodollar Rate Loan.

"<u>UCC</u>" means the Uniform Commercial Code as in effect in the State of New York; <u>provided</u> that, if perfection or the effect of perfection or non-perfection or the priority of any security interest in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of New York, "<u>UCC</u>" means the Uniform Commercial Code as in effect from time to time in such other jurisdiction for purposes of the provisions hereof relating to such perfection, effect of perfection or non-perfection or priority.

"<u>Unfunded Pension Liability</u>" means the excess of a Pension Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the current value of that Pension Plan's assets, determined in accordance with the assumptions used for funding the Pension Plan pursuant to Section 412 of the Code for the applicable plan year.

"<u>United States</u>" and "<u>U.S</u>." mean the United States of America.

"Unreimbursed Amount" has the meaning specified in Section 2.13(e).

"<u>Unsecured Term Loan Facility</u>" means that certain unsecured credit agreement, dated as of January 18, 2013, among the Borrower, Holdings, as a guarantor, Macquarie US Trading LLC, as administrative agent, and the lenders named therein and from time to time party thereto, together with any refinancing, refunding, renewal or extension thereof pursuant to <u>Section 7.02(m)</u>.

"<u>Unsecured Term Loan Facility Documents</u>" means the Unsecured Term Loan Facility and all other agreements, instruments and other documents executed in connection therewith.

<u>"Unsecured Term Loan Prepayment Conditions</u>" means, as to any relevant action contemplated in this-Agreement, (i) no Event of Default has then occurred and is continuing or would result from such action, (ii) (a) the sum of (x) the amount of Qualified Cash on the date of such action plus (y) Availability on a Pro Forma Basisimmediately after giving effect to such action would be at least \$35,000,000 and (b) over the 30 consecutive daysprior to consummation of such action, the sum of (x) the amount of Qualified Cash-plus (y) Availability averaged noless than \$30,000,000 and (iii) the Consolidated Fixed Charge Coverage Ratio would be at least 1.0 to 1.0 on a Pro-Forma Basis for such action.

"<u>Unused Line Fee</u>" has the meaning specified in <u>Section 2.05(a)</u>.

"<u>Unused Line Fee Rate</u>" means, (i) initially, <u>0.500.375</u>% per annum and (ii) from and after the delivery by the Borrower to the Administrative Agent of the Borrowing Base Certificate for the <u>second_first</u> full fiscal <u>quartermonth</u> completed after the <u>ClosingRatification Effective</u> Date, the rate determined by reference to the following grid on a per annum basis based on the Average Usage as a percentage of the Aggregate Commitments during the immediately preceding fiscal <u>quartermonth</u>:

Average Usage (percentage of Aggregate Commitments)		Unused Line Fee Rate

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 71 of 432

< 50%	0.50%
$\geq 50\%$	0.375%

"USA PATRIOT Act" has the meaning specified in the definition of "Anti-Terrorism Laws."

"<u>U.S. Loan Party</u>" means any Loan Party that is organized under the laws of one of the states of the United States of America and that is not a CFC.

<u>"U.S. Person" means any Person that is a "United States person" as defined in Section 7701(a)(30) of the</u> <u>Code.</u>

"Variance Report" has the meaning set forth in the Ratification Agreement.

"Weighted Average Life to Maturity" means, when applied to any Indebtedness at any date, the number of years obtained by dividing (i) the then outstanding principal amount of such Indebtedness into (ii) the product obtained by multiplying (x) the amount of each then remaining installment or other required scheduled payments of principal, including payment at final maturity, in respect thereof, by (y) the number of years (calculated to the nearest one-twelfth) that will elapse between such date and the making of such payment.

"<u>Write-Down and Conversion Powers</u>" means, with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.

1.02 <u>Other Interpretive Provisions</u>.

With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

The definitions of terms herein shall apply equally to the singular and plural forms of the terms (a) defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "will" shall be construed to have the same meaning and effect as the word "shall." Unless the context requires otherwise, (i) any definition of or reference to any agreement, instrument or other document (including any Organization Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein or in any other Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "herein," "hereof" and "hereunder," and words of similar import when used in any Loan Document, shall be construed to refer to such Loan Document in its entirety and not to any particular provision thereof, (iv) all references in a Loan Document to Articles, Sections, Preliminary Statements, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Preliminary Statements, Exhibits and Schedules to, the Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time, and (vi) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

(b) In the computation of periods of time from a specified date to a later specified date, the word "<u>from</u>" means "<u>from and including</u>"; the words "<u>to</u>" and "<u>until</u>" each mean "<u>to but excluding</u>"; and the word "<u>through</u>" means "<u>to and including</u>."

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 72 of 432

(c) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

(d) The Borrower has a fiscal year and fiscal quarters not ending on the calendar year and calendar quarter, respectively. For the avoidance of doubt for so long as such situation exists, when interpreting any provision of this Agreement relating to a fiscal year or fiscal quarter or period ended at or as of the end of a fiscal year or fiscal quarter, including, without limitation, the financial covenant set forth in Section 7.11, any reference to such a period shall be deemed to refer to the Borrower's fiscal quarter or year ended nearest the given date. In addition, any reference to a fiscal quarter of the Borrower contained in Section 7.11 (e.g., "Q3 2010") shall be deemed to refer to the specified fiscal year (e.g., in the case of Q3 2010, such reference means the third fiscal quarter of the Borrower's 2010 fiscal year).

(e) On and after the initial issuance date of the 2014 Notes, (i) each reference to the term. "Intercreditor Agreement" in this Agreement and the other Loan Documents shall mean and be deemed to be a reference to the 2014 Intercreditor Agreement, and (ii) each reference to the term "Term Intercreditor-Agreement" in this Agreement and the other Loan Documents shall mean and be deemed to be a reference to the First Lien Intercreditor Agreement.

1.03 Accounting Terms.

(a) <u>Generally</u>. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, except as otherwise specifically prescribed herein.

(b) <u>Changes in GAAP</u>. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders shall so request, the Administrative Agent, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); <u>provided</u> that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

(c) <u>Pro Forma Calculations</u>. Notwithstanding anything herein to the contrary, any calculation of the Financial Covenant for any Measurement Period during which an Acquisition or Disposition shall have occurred (or shall be deemed to have occurred for the purposes described in clause (ii) of this <u>Section 1.03(c)</u>) shall each be made on a Pro Forma Basis for purposes of making the following determinations:

(i) determining compliance with the Financial Covenant (other than whether the conditionsprecedent for a proposed transaction have been satisfied as contemplated by clause (ii) of this <u>Section</u>. <u>1.03(c)</u>); and

(ii) determining whether the conditions precedent have been satisfied for a proposed transaction which is permitted hereunder only so long as no Default will result from the consummation thereof, including, without limitation, any Disposition or any Investment which results in an Acquisition.

Any calculation made on a Pro Forma Basis or to determine pro forma compliance with any Financial-Covenant or financial ratio shall similarly count any other Acquisition or Disposition consummated and any other-Indebtedness and related interest expense otherwise incurred or repaid after the latest Measurement Period and priorto the calculation date. Any time this Agreement requires pro forma compliance with any Financial Covenant orfinancial ratio, such compliance shall be demonstrated by a certified calculation of such pro forma complianceprovided by the Borrower to the Administrative Agent.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 73 of 432

"Pro Forma Basis" means, for purposes of calculating any financial ratio (including the (c) Consolidated Leverage Ratio, the Consolidated Secured Debt Ratio, the Consolidated First Lien Leverage Ratio and the Consolidated Fixed Charge Coverage Ratio) or financial amount for any Measurement Period for any of thepurposes specified in this Section 1.03(c), and with respect to each proposed Acquisition or Disposition and each such transaction actually consummated in such Measurement Period, that such financial ratio or financial amountshall be calculated on a pro forma basis based on the following assumptions: (a) each such transaction shall bedeemed to have occurred on the first day of such Measurement Period; (b) any funds to be used by any Person inconsummating any such transaction will be assumed to have been used for that purpose as of the first day of such-Measurement Period; (c) any Indebtedness to be incurred by any Person in connection with the consummation of any such transaction will be assumed to have been incurred on the first day of such Measurement Period; (d) the grossinterest expenses, determined in accordance with GAAP, with respect to such Indebtedness assumed to have beenincurred on the first day of such Measurement Period that bears interest at a floating rate shall be calculated at the eurrent rate (as of the date of such calculation) under the agreement governing such Indebtedness (including this-Agreement if the Indebtedness is incurred hereunder); and (e) any gross interest expense, determined in accordancewith GAAP, with respect to Indebtedness outstanding during such Measurement Period that was or is to berefinanced with proceeds of a transaction assumed to have been incurred as of the first day of the Measurement-Period will be excluded from such calculations (and to the extent not already excluded pursuant to clause (a) or (b)above, the principal amount of such Indebtedness shall be excluded). "Pro Forma Basis" may also include suchadjustments for expected cost savings in connection with an Acquisition or Disposition as forecasted by the Borrower to occur within twelve months after such Acquisition or Disposition in a reasonable manner withappropriate supporting documentation and representations by management, reasonably satisfactory to the Administrative Agent; provided, that, (i) "Pro Forma Basis" shall not include any expected cost savings arising from-(A) any Acquisition of property by any Loan Party or any of its Subsidiaries from any Affiliate thereof or (B) any Disposition of any Property by any Loan Party or any of its Subsidiaries to any Affiliate thereof, and (ii) prior to taking into account any such expected cost savings in calculating any financial ratio or financial amount pursuant tothis paragraph, the Borrower shall deliver to the Administrative Agent a Pro Forma Adjustment Compliance Certificate; provided, further, that the aggregate amount of such adjustments for expected cost savings during any-Measurement Period (excluding cost savings arising from an Acquisition with respect to which at least 85% of the purchase price is financed with Equity) shall not exceed 15.0% of Consolidated Adjusted EBITDA for such-Measurement Period.[Reserved].

(d) <u>Consolidation of Variable Interest Entities</u>. All references herein to consolidated financial statements of Holdings and its Subsidiaries or to the determination of any amount for Holdings and its Subsidiaries on a consolidated basis or any similar reference shall, in each case, be deemed to include each variable interest entity that Holdings is required to consolidate pursuant to FASB Interpretation No. 46 – Consolidation of Variable Interest Entities: an interpretation of ARB No. 51 (January 2003) as if such variable interest entity were a Subsidiary as defined herein.

(e) All references to the terms "Borrower", "Guarantors" or "Debtors" in this Agreement and any other Loan Documents shall be deemed and each such reference is hereby amended to mean and include (as applicable) the Debtors, and their successors and assigns (including any trustee or other fiduciary hereafter appointed as any Debtor's legal representative, as applicable, or with respect to any Debtor the property of the estate of such Debtor whether under Chapter 11 of the Bankruptcy Code or any subsequent Chapter 7 case or cases and its successor upon conclusion of the Chapter 11 Case of such Debtor, as the case may be).

1.04 <u>Rounding</u>.

Any financial ratios required to be maintained by the Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 74 of 432

1.05 <u>Times of Day</u>.

Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

1.06 <u>Uniform Commercial Code and PPSA</u>.

As used herein, the following terms are defined in accordance with the UCC in effect in the State of New York (and with respect to any Canadian Guarantor, such definition (or correlative terms (if existing) under the PPSA) shall be defined in accordance with the PPSA) from time to time: "Commercial Tort Claim," "Document" and "Document of Title," "Equipment," "General Intangibles" and "Intangibles," "Goods," "Instrument" and "Supporting Obligations."

1.07 <u>Currency Equivalents Generally</u>.

Any amount specified in this Agreement or any of the other Loan Documents to be in Dollars shall also include the equivalent of such amount in any currency other than Dollars to the extent necessary to give effect to the intent, where applicable, that this Agreement apply to Canadian Guarantors, such equivalent amount thereof in the applicable currency to be determined by the Administrative Agent at such time on the basis of the Spot Rate for the purchase of such currency with Dollars. Unless expressly provided otherwise, all references in the Loan Documents to Revolving Loans, Letters of Credit, Obligations, Revolving Commitments, Borrowing Base components and other amounts shall be denominated in Dollars. The Dollar Equivalent of any amounts denominated or reported under a Loan Document in a currency other than Dollars shall be determined by the Administrative Agent at its discretion, based on the current Spot Rate. The Borrower shall report Borrowing Base components to the Administrative Agent in the currency invoiced by the applicable Loan Parties or shown in their financial records, and unless expressly provided otherwise herein, shall deliver financial statements and calculate financial covenants in Dollars.

1.08 <u>Currency Fluctuations</u>.

On such dates determined by the Administrative Agent (the "<u>Exchange Rate Date</u>"), the Administrative Agent shall determine the Exchange Rate as of such date. The Exchange Rate so determined shall become effective on the first Business Day immediately following such determination (a "<u>Reset Date</u>") and shall remain effective until the next succeeding Reset Date.

1.09 Interpretation (Quebec).

For purposes of any Collateral located in the Province of Quebec or charged by any deed of hypothec (or any other Loan Document) and for all other purposes pursuant to which the interpretation or construction of a Loan Document may be subject to the laws of the Province of Quebec or a court or tribunal exercising jurisdiction in the Province of Quebec, (a) "personal property" shall be deemed to include "movable property," (b) "real property" shall be deemed to include "immovable property," (c) "tangible property" shall be deemed to include "corporeal property," (d) "intangible property" shall be deemed to include "incorporeal property," (e) "security interest," "mortgage" and "lien" shall be deemed to include a "hypothec," "prior claim" and a "resolutory clause," (f) all references to filing, registering or recording under the UCC or the PPSA shall be deemed to include publication under the Civil Code of Quebec, (g) all references to "perfection" of or "perfected" Liens shall be deemed to include a reference to an "opposable" or "set up" Liens as against third parties, (h) any "right of offset," "right of setoff" or similar expression shall be deemed to include a "right of compensation," (i) "goods" shall be deemed to include "corporeal movable property" other than chattel paper, documents of title, instruments, money and securities, (j) an "agent" shall be deemed to include a "mandatary," (k) "construction liens" shall be deemed to include "legal hypothecs," (1) "joint and several" shall be deemed to include "solidary," (m) "gross negligence or willful misconduct" shall be deemed to be "intentional or gross fault," (n) "beneficial ownership" shall be deemed to include "ownership on behalf of another as mandatary," (o) "easement" shall be deemed to include "servitude," (p) "priority" shall be deemed to include "prior claim," (q) "survey" shall be deemed to include "certificate of location and plan" and (r) "fee simple title" shall be deemed to include "absolute ownership." The parties hereto confirm that it is their wish that this Agreement and any other document executed in connection with the transactions contemplated herein

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 75 of 432

be drawn up in the English language only (except if another language is required under any applicable Law) and that all other documents contemplated thereunder or relating thereto, including notices, may also be drawn up in the English language only. Les parties aux présentes confirment que c'est leur volonté que cette convention et les autres documents de crédit soient rédigés en langue anglaise seulement et que tous les documents, y compris tous avis, envisagés par cette convention et les autres documents peuvent être rédigés en la langue anglaise seulement (sauf si une autre langue est requise en vertu d'une loi applicable).

ARTICLE II

AMOUNT AND TERMS OF CREDIT

2.01 <u>Commitments</u>.

Subject to the terms and conditions and relying upon the representations and warranties herein set forth, each Lender agrees, severally and not jointly, to make Revolving Loans to the Borrower, at any time and from time to time on and after the Closing Date until the earlier of one Business Day prior to the Maturity Date and the termination of the Commitment of such Lender in accordance with the terms hereof, in an aggregate principal amount at any time outstanding that will not result in such Lender's Revolving Exposure exceeding the lesser of (A) such Lender's Revolving Commitment, and (B) such Lender's Pro Rata Percentage multiplied by the Borrowing Base then in effect. Within the limits set forth above and subject to the terms, conditions and limitations set forth herein, the Borrower may borrow, pay or prepay and reborrow Revolving Loans.

2.02 <u>Revolving Loans</u>.

(a) Each Revolving Loan (other than Swingline Loans) shall be made as part of a Borrowing consisting of Revolving Loans made by the Lenders ratably in accordance with their applicable Commitments; <u>provided</u> that the failure of any Lender to make any Revolving Loan shall not in itself relieve any other Lender of its obligation to lend hereunder (it being understood, however, that no Lender shall be responsible for the failure of any other Lender to be made by such other Lender). Except for Revolving Loans deemed made pursuant to <u>Section 2.02(f)</u>, Revolving Loans (other than Swingline Loans) comprising any Borrowing shall be in an aggregate principal amount that is (i) (A) in the case of Base Rate Loans, not less than \$500,000 and (B) in the case of Eurodollar Rate Loans, an integral multiple of \$250,000 and not less than \$1,000,000, or (ii) equal to the remaining available balance of the applicable Revolving Commitments.

(b) Subject to <u>Sections 3.02</u>, <u>3.03</u> and <u>3.04</u>, each Borrowing shall be comprised entirely of Base Rate Loans or Eurodollar Rate Loans as the Borrower may request pursuant to <u>Section 2.03</u>. Each Lender may at its option make any Eurodollar Rate Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Revolving Loan; <u>provided</u> that any exercise of such option shall not affect the obligation of the Borrower to repay such Revolving Loan in accordance with the terms of this Agreement or cause the Borrower to pay additional amounts pursuant to <u>Sections 3.02</u>, <u>3.03</u> and <u>3.04</u>. Borrowings of more than one Type may be outstanding at the same time; <u>provided further</u> that the Borrower shall not be entitled to request any Borrowing that, if made, would result in more than ten Borrowings of Eurodollar Rate Loans outstanding hereunder at any one time. For purposes of the foregoing, Borrowings having different Interest Periods, regardless of whether they commence on the same date, shall be considered separate Borrowings.

(c) Except with respect to Revolving Loans deemed made pursuant to <u>Section 2.02(f)</u>, each Lender shall make each Revolving Loan (other than Swingline Loans) to be made by it hereunder on the proposed date thereof by wire transfer of immediately available funds to such account in New York City as the Administrative Agent may designate not later than 3:00 p.m., New York City time, and the Administrative Agent shall promptly credit the amounts so received to an account as directed by the Borrower in the applicable Notice of Borrowing maintained with the Administrative Agent or, if a Borrowing shall not occur on such date because any condition precedent herein specified shall not have been met or waived, return the amounts so received to the respective Lenders.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 76 of 432

(d) Unless the Administrative Agent shall have received notice from a Lender prior to the date of any Borrowing that such Lender will not make available to the Administrative Agent such Lender's portion of such Borrowing, the Administrative Agent may assume that such Lender has made such portion available to the Administrative Agent on the date of such Borrowing in accordance with <u>paragraph (c)</u> above, and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If the Administrative Agent shall have so made funds available then, to the extent that such Lender shall not have made such portion available to the Administrative Agent forthwith on demand such corresponding amount together with interest thereon, for each day from the date such amount is made available to the Borrower until the date such amount is repaid to the Administrative Agent at (i) in the case of the Borrower, the interest rate applicable at the time to the Revolving Loans comprising such Borrowing and (ii) in the case of such Lender, a rate determined by the Administrative Agent to represent its cost of overnight or short-term funds (which determination shall be conclusive absent manifest error). If such Lender shall repay to the Administrative Agent such corresponding amount, such amount shall constitute such Lender's Revolving Loan as part of such Borrowing for purposes of this Agreement.

(e) Notwithstanding any other provision of this Agreement, the Borrower shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

If the Issuing Bank shall not have received from the Borrower the payment required to be made by (f) Section 2.13(e) within the time specified in such Section, the Issuing Bank will promptly notify the Administrative Agent of the LC Disbursement and the Administrative Agent will promptly notify each Revolving Lender of such LC Disbursement and its Pro Rata Percentage thereof. Each Revolving Lender shall pay by wire transfer of immediately available funds to the Administrative Agent on such date (or, if such Revolving Lender shall have received such notice later than 12:00 (noon), New York City time, on any day, not later than 11:00 a.m., New York City time, on the immediately following Business Day), an amount equal to such Lender's Pro Rata Percentage of such LC Disbursement (it being understood that such amount shall be deemed to constitute a Base Rate Loan of such Lender, and such payment shall be deemed to have reduced the LC Exposure), and the Administrative Agent will promptly pay to the Issuing Bank amounts so received by it from the Revolving Lenders. The Administrative Agent will promptly pay to the Issuing Bank any amounts received by it from the Borrower pursuant to Section 2.13(e) prior to the time that any Revolving Lender makes any payment pursuant to this paragraph (f); any such amounts received by the Administrative Agent thereafter will be promptly remitted by the Administrative Agent to the Revolving Lenders that shall have made such payments and to the Issuing Bank, as their interests may appear. If any Revolving Lender shall not have made its Pro Rata Percentage of such LC Disbursement available to the Administrative Agent as provided above, such Lender and the Borrower severally agree to pay interest on such amount, for each day from and including the date such amount is required to be paid in accordance with this paragraph (f) to but excluding the date such amount is paid, to the Administrative Agent for the account of the Issuing Bank at (i) in the case of the Borrower, a rate per annum equal to the interest rate applicable to Revolving Loans pursuant to Section 2.06(a), and (ii) in the case of such Lender, for the first such day, the Federal Funds Rate, and for each day thereafter, the Base Rate.

2.03 <u>Borrowing Procedure</u>.

To request a Revolving Borrowing, the Borrower shall notify the Administrative Agent of such request by telecopy or electronic transmission (if arrangements for doing so have been approved by the Administrative Agent, which approval shall not be unreasonably withheld, conditioned or delayed) or telephone (promptly confirmed by telecopy) (i) in the case of a Borrowing of Eurodollar Rate Loans, not later than 11:00 a.m., New York City time, three Business Days before the date of the proposed Borrowing or (ii) in the case of a Borrowing of Base Rate Loans (other than Swingline Loans), not later than 11:00 a.m., New York City time, on the Business Day of the proposed Borrowing. Each such telephonic Notice of Borrowing shall be irrevocable, subject to <u>Sections 2.09</u> and <u>3.01</u>, and shall be confirmed promptly by hand delivery or telecopy to the Administrative Agent of a written Notice of Borrowing in a form approved by the Administrative Agent and signed by the Borrower. Each such telephonic and written Notice of Borrowing information in compliance with <u>Section 2.02</u>:

(a) the aggregate amount of such Borrowing;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 77 of 432

(b) the date of such Borrowing, which shall be a Business Day;

(c) whether such Borrowing is to be a Borrowing of Base Rate Loans or a Borrowing of Eurodollar Rate Loans;

(d) in the case of a Borrowing of Eurodollar Rate Loans, the initial Interest Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Interest Period";

(e) the location and number of the account to which funds are to be disbursed, which shall comply with the requirements of <u>Section 2.02</u>; and

(f) that the conditions set forth in <u>Section 4.01</u> or <u>Section 4.02</u>, as applicable, are satisfied or waived as of the date of the notice.

If no election as to the Type of Borrowing is specified, then the requested Borrowing shall be a Borrowing of Base Rate Loans. If no Interest Period is specified with respect to any requested Borrowing of Eurodollar Rate Loans, then the Borrower shall be deemed to have selected an Interest Period of one month's duration (subject to the proviso in <u>clause (d)</u> above). Promptly following receipt of a Notice of Borrowing in accordance with this <u>Section</u> <u>2.03</u>, the Administrative Agent shall advise each Lender of the details thereof and of the amount of such Lender's Revolving Loan to be made as part of the requested Borrowing.

2.04 Evidence of Debt; Repayment of Revolving Loans.

(a) The Borrower hereby unconditionally promises to pay (i) to the Administrative Agent for the account of each Revolving Lender, the then unpaid principal amount of each Revolving Loan of such Lender on the Maturity Date and (ii) to the Swingline Lender the then unpaid principal amount of each Swingline Loan on the Maturity Date.

(b) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender resulting from each Revolving Loan made by such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement. The Borrower shall be entitled to review records of such accounts with prior reasonable notice during normal business hours.

(c) The Administrative Agent shall maintain accounts in which it will record (i) the amount of each Revolving Loan made hereunder, the Type thereof and the Interest Period applicable thereto; (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder; and (iii) the amount of any sum received by the Administrative Agent hereunder for the account of the Lenders and each Lender's share thereof. Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender. The Borrower shall be entitled to review records of such accounts with prior reasonable notice during normal business hours.

(d) The entries made in the accounts maintained pursuant to <u>paragraphs</u> (b) and (c) above shall be *prima facie* evidence of the existence and amounts of the obligations therein recorded absent manifest error; <u>provided</u> that the failure of any Lender or the Administrative Agent to maintain such accounts or any error therein shall not in any manner affect the obligations of the Borrower to repay the Revolving Loans in accordance with their terms.

(e) Any Lender may request that Revolving Loans made by it be evidenced by a promissory note. In such event, the Borrower shall promptly prepare, execute and deliver to such Lender a promissory note payable to such Lender (or, if requested by such Lender, to such Lender and its registered assigns) substantially in the form of Exhibit B-1 or Exhibit B-2, as applicable.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 78 of 432

2.05 <u>Fees</u>.

(a) <u>Unused Line Fee</u>. The Borrower shall pay to the Administrative Agent, for the ratable benefit of the Lenders (other than any Defaulting Lender), a fee equal to the Unused Line Fee Rate multiplied by the amount by which the <u>RevolvingAggregate</u> Commitments (other than Revolving Commitments of a Defaulting Lender) exceed the average daily balance of outstanding Revolving Loans (other than Swingline Loans) and stated amount of outstanding Letters of Credit during any fiscal <u>quartermonth</u> (such fee, the "<u>Unused Line Fee</u>"). Such fee shall accrue commencing on the <u>ClosingRatification Effective</u> Date, and will be payable in arrears, on the first day of each fiscal <u>quartermonth</u>, commencing <u>JulyMarch</u> 1, <u>2013.2018</u>.

(b) <u>Administrative Agent Fees</u>. The Borrower agrees to pay to the Administrative Agent, for its own account, the fees set forth in the Fee <u>LetterLetters</u> or such other fees payable in the amounts and at the times separately agreed upon between the Borrower and the Administrative Agent (the "<u>Administrative Agent Fees</u>").

(c) LC and Fronting Fees. The Borrower agrees to pay (i) to the Administrative Agent for the account of each Revolving Lender a participation fee ("LC Participation Fee") with respect to its participations in Letters of Credit, which shall accrue at a rate per annum equal to the Applicable Margin from time to time used to determine the interest rate on Eurodollar Rate Loans pursuant to Section 2.06, on the average daily amount of such Lender's LC Exposure (excluding any portion thereof attributable to unreimbursed LC Disbursements) during the period from and including the ClosingRatification Effective Date to but excluding the later of the date on which such Lender's Revolving Commitment terminates and the date on which such Lender ceases to have any LC Exposure, and (ii) to the Issuing Bank a fronting fee ("Fronting Fee"), which shall accrue at the rate of 0.125% per annum on the average daily amount of the LC Exposure (excluding any portion thereof attributable to unreimbursed LC Disbursements) during the period from and including the ClosingRatification Effective Date to but excluding the later of the date of termination of the Revolving Commitments and the date on which there ceases to be any LC Exposure, as well as the Issuing Bank's standard and reasonable fees with respect to the issuance, amendment, renewal or extension of any Letter of Credit or processing of drawings thereunder as agreed among the Borrower and the Issuing Bank from time to time. LC Participation Fees and Fronting Fees accrued through and including the last day of March, June, September and December of each yearmonth shall be payable on the third Business Dayfirst day of the following such last daymonth, commencing on the first such date to occur after the ClosingRatification Effective Date; provided that all such fees shall be payable on the date on which the Revolving Commitments terminate and any such fees accruing after the date on which the Revolving Commitments terminate shall be payable on demand (including documentation reasonably supporting such request). Any other fees payable to the Issuing Bank pursuant to this paragraph shall be payable within 10 days after written demand (together with backup documentation supporting such reimbursement request). All LC Participation Fees and Fronting Fees shall be computed on the basis of a year of 360 days and shall be payable for the actual number of days elapsed (including the first day but excluding the last day).

(d) All fees shall be paid on the dates due, in immediately available funds, to the Administrative Agent for distribution, if and as appropriate, among the Lenders (other than Defaulting Lenders), except that the Fronting Fees shall be paid directly to the Issuing Bank. Once paid, none of the fees shall be refundable under any circumstances.

2.06 Interest on Revolving Loans.

(a) Subject to the provisions of <u>Section 2.06(c)</u>, the Revolving Loans comprising each Borrowing of Base Rate Loans, including each Swingline Loan, shall bear interest at a rate per annum equal to the Base Rate *plus* the Applicable Margin in effect from time to time.

(b) Subject to the provisions of <u>Section 2.06(c)</u>, the Revolving Loans comprising each Borrowing of Eurodollar Rate Loans shall bear interest at a rate per annum equal to the Eurodollar Rate for the Interest Period in effect for such Borrowing *plus* the Applicable Margin in effect from time to time.

(c) Notwithstanding the foregoing, if any principal of or interest on any Revolving Loan or any fees or other amount payable by the Borrower hereunder is not paid when due, whether at stated maturity, upon acceleration

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 79 of 432

or otherwise, such overdue amount shall bear interest, after as well as before judgment, at a rate per annum equal to (i) in the case of overdue principal of, or interest on, any Revolving Loan, 2% *plus* the rate otherwise applicable to such Revolving Loan or (ii) in the case of any other amount, 2% *plus* the rate applicable to Base Rate Loans (in each case, the "<u>Default Rate</u>").

(d) Accrued interest on each RevolvingEurodollar Rate Loan and each Base Rate Loan shall be payable in each case in arrears on the first day of each Adjustment Datecalendar month, commencing with July 1, 2013, for such Revolving LoanMarch 1, 2018 and upon termination of the Revolving Commitments; provided that ($i\underline{A}$) interest accrued pursuant to paragraph (c) of this Section 2.06 shall be payable on demand and, absent demand, on the first day of each Adjustment Datecalendar month and upon termination of the Revolving Commitments, ($i\underline{iB}$) in the event of any repayment or prepayment of any Revolving Loan (other than a prepayment of a Base Rate Loan prior to the end of the Revolving Availability Period), accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment and ($i\underline{iii}$) in the event of any conversion of any Eurodollar Rate Loan prior to the end of the current Interest Period therefor, accrued interest on such Revolving Loan shall be payable on the effective date of such conversion.

(e) All interest hereunder shall be computed on the basis of a year of 365/366 days, except that interest computed by reference to the Eurodollar Rate (other than Base Rate Loans determined by reference to the Eurodollar Rate) and all fees shall be computed on the basis of a year of 360 days, and in each case shall be payable for the actual number of days elapsed (including the first day but excluding the last day). The applicable Base Rate or Eurodollar Rate shall be determined by the Administrative Agent in accordance with the provisions of this Agreement and such determination shall be conclusive absent manifest error.

2.07 <u>Termination and Reduction of Commitments</u>.

(a) The Revolving Commitments, the Swingline Commitment, and the LC Commitment shall automatically terminate on the Maturity Date.

(b) The Borrower may at any time terminate, or from time to time reduce, the Revolving Commitments; <u>provided</u> that (i) any such reduction shall be in an amount that is an integral multiple of \$1,000,000 and (ii) the Revolving Commitments shall not be terminated or reduced if after giving effect to any concurrent prepayment of the Revolving Loans in accordance with <u>Section 2.09</u>, the Aggregate Exposures would exceed the Aggregate Commitments.

(c) The Borrower shall notify the Administrative Agent of any election to terminate or reduce the Aggregate Commitments under <u>paragraph (b)</u> of this <u>Section 2.07</u> at least two Business Days prior to the effective date of such termination or reduction, specifying such election and the effective date thereof. Promptly following receipt of any notice, the Administrative Agent shall advise the Lenders of the contents thereof. Each notice delivered by the Borrower pursuant to this <u>Section 2.07</u> shall be irrevocable except that, to the extent delivered in connection with a refinancing of the Obligations, such notice shall not be irrevocable until such refinancing is closed and funded. Any effectuated termination or reduction of the Aggregate Commitments shall be made ratably among the Lenders in accordance with their respective Revolving Commitments.

2.08 Interest Elections.

(a) Each Revolving Borrowing initially shall be of the Type specified in the applicable Notice of Borrowing and, in the case of a Borrowing of Eurodollar Rate Loans, shall have an initial Interest Period as specified in such Notice of Borrowing. Thereafter, the Borrower may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Borrowing of Eurodollar Rate Loans, may elect Interest Periods therefor, all as provided in this <u>Section 2.08</u>. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Revolving Loans comprising such Borrowing, and the Revolving Loans comprising each such portion shall be considered a separate Borrowing. Notwithstanding anything to the contrary, the Borrower shall not be entitled to request any conversion or continuation that, if made, would result in more than ten Borrowings of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 80 of 432

Eurodollar Rate Loans outstanding hereunder at any one time. This <u>Section 2.08</u> shall not apply to Swingline Loans, which may not be converted or continued.

(b) To make an election pursuant to this <u>Section 2.08</u>, the Borrower shall notify the Administrative Agent of such election by telephone or electronic transmission (if arrangements for doing so have been approved by the Administrative Agent, which approval shall not be unreasonably withheld, delayed or conditioned) by the time that a Notice of Borrowing would be required under <u>Section 2.03</u> if the Borrower was requesting a Revolving Borrowing of the Type resulting from such election to be made on the effective date of such election, subject to <u>Section 3.05</u>. Each such telephonic Notice of Conversion/Continuation shall be confirmed promptly by hand delivery or telecopy to the Administrative Agent of a written Notice of Conversion/Continuation substantially in the form of <u>Exhibit A-2</u>, unless otherwise agreed to by the Administrative Agent and the Borrower.

(c) Each telephonic and written Notice of Conversion/Continuation shall specify the following information in compliance with <u>Section 2.02</u>:

(i) the Borrowing to which such Notice of Conversion/Continuation applies and, if different options are being elected with respect to different portions thereof, the portions thereof to be allocated to each resulting Borrowing (in which case the information to be specified pursuant to <u>clauses (iii)</u> and <u>(iv)</u> below shall be specified for each resulting Borrowing);

(ii) the effective date of the election made pursuant to such Notice of Conversion/Continuation, which shall be a Business Day;

(iii) whether the resulting Borrowing is to be a Borrowing of Base Rate Loans or a Borrowing of Eurodollar Rate Loans; and

(iv) if the resulting Borrowing is a Borrowing of Eurodollar Rate Loans, the Interest Period to be applicable thereto after giving effect to such election, which shall be a period contemplated by the definition of the term "Interest Period."

If any such Notice of Conversion/Continuation requests a Borrowing of Eurodollar Rate Loans but does not specify an Interest Period, then the Borrower shall be deemed to have selected an Interest Period of one month's duration.

(d) Promptly following receipt of a Notice of Conversion/Continuation, the Administrative Agent shall advise each Lender of the details thereof and of such Lender's portion of each resulting Borrowing.

(e) If a Notice of Conversion/Continuation with respect to a Borrowing of Eurodollar Rate Loans is not timely delivered prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein, at the end of such Interest Period such Borrowing shall be converted to a Borrowing of Base Rate Loans. Notwithstanding any contrary provision hereof, if an Event of Default has occurred and is continuing and the Administrative Agent, at the request of the Required Lenders, so notifies the Borrower, then, after the occurrence and during the continuance of such Event of Default (i) no outstanding Borrowing may be converted to or continued as a Borrowing of Eurodollar Rate Loans and (ii) unless repaid, each Borrowing of Eurodollar Rate Loans shall be converted to a Borrowing of Base Rate Loans at the end of the Interest Period applicable thereto.

2.09 Optional and Mandatory Prepayments of Revolving Loans.

(a) <u>Optional Prepayments</u>. The Borrower shall have the right at any time and from time to time to prepay, without premium or penalty, any Borrowing, in whole or in part, subject to the requirements of this <u>Section</u> 2.09; provided that each partial prepayment shall be in an amount that is an integral multiple of \$100,000.

(b) <u>Revolving Loan Prepayments</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 81 of 432

(i) In the event of the termination of all the Revolving Commitments, the Borrower shall, on the date of such termination, repay or prepay all the outstanding Revolving Borrowings and all outstanding Swingline Loans and Cash Collateralize or backstop on terms reasonably satisfactory to the Administrative Agent the LC Exposure in accordance with Section 2.13(j).

(ii) In the event of any partial reduction of the Revolving Commitments, then (A) at or prior to the effective date of such reduction, the Administrative Agent shall notify the Borrower and the Revolving Lenders of the Aggregate Exposures after giving effect thereto and (B) if the Aggregate Exposures would exceed the Line Cap then in effect, after giving effect to such reduction, then the Borrower shall, on the date of such reduction (or, if such reduction is due to the imposition of new Reserves or a change in the methodology of calculating existing Reserves, within five Business Days following such notice), *first*, repay or prepay all Swingline Loans, *second*, repay or prepay Revolving Borrowings and *third*, replace or Cash Collateralize outstanding Letters of Credit in accordance with the procedures set forth in <u>Section 2.13(j)</u>, in an amount sufficient to eliminate such excess.

(iii) In the event that the Aggregate Exposures at any time exceeds the Line Cap then in effect, the Borrower shall, immediately after demand (or, if such overadvance is due to the imposition of new Reserves or a change in the methodology of calculating existing Reserves, or change in eligibility standards, within five Business Days following notice), apply an amount equal to such excess to prepay the Revolving Loans and any interest accrued thereon, in accordance with this <u>Section 2.09(b)(iii)</u>. The Borrower shall, *first*, repay or prepay all Swingline Loans, *second*, repay or prepay Revolving Borrowings, and *third*, replace or Cash Collateralize outstanding Letters of Credit in accordance with the procedures set forth in <u>Section 2.13(j)</u>, in an amount sufficient to eliminate such excess.

(iv) In the event that the aggregate LC Exposure exceeds the LC Commitment then in effect, the Borrower shall, without notice or demand, immediately replace or Cash Collateralize outstanding Letters of Credit in accordance with the procedures set forth in Section 2.13(j), in an amount sufficient to eliminate such excess.

(c) <u>Application of Prepayments</u>.

(i) Prior to any optional or mandatory prepayment of Borrowings hereunder, the Borrower shall select the Borrowing or Borrowings to be prepaid and shall specify such selection in the notice of such prepayment pursuant to paragraph (i) of this Section 2.09(c). Unless during a Liquidity Period, except as provided in Section 2.09(b)(iii) hereof, all mandatory prepayments shall be applied as follows: *first*, to fees and reimbursable expenses of the Administrative Agent then due and payable pursuant to the Loan Documents; *second*, to interest then due and payable on the Borrower's Swingline Loans; *third*, to the principal balance of the Swingline Loans outstanding until the same has been prepaid in full; *fourth*, to interest then due and payable on the Revolving Loans and other amounts due pursuant to Section 3.05; *fifth*, to the principal balance of the Revolving Loans until the same have been prepaid in full; *sixth*, to Cash Collateralize all LC Exposure plus any accrued and unpaid interest thereon (to be held and applied in accordance with Section 2.13(j) hereof); *seventh*, to all other Obligations *pro rata* in accordance with the amounts that such Lender certifies is outstanding; and *eighth*, as required by the TermFirst Lien Intercreditor Agreement or, in the absence of any such requirement, returned to the Borrower or to such party as otherwise required by law.

(ii) Amounts to be applied pursuant to this <u>Section 2.09</u> to the prepayment of Revolving Loans shall be applied, as applicable, first to reduce outstanding Base Rate Loans. Any amounts remaining after each such application shall be applied to prepay Eurodollar Rate Loans. Notwithstanding the foregoing, if the amount of any prepayment of Revolving Loans required under this <u>Section 2.09</u> shall be in excess of the amount of the Base Rate Loans at the time outstanding, only the portion of the amount of such prepayment that is equal to the amount of such outstanding Base Rate Loans shall be immediately prepaid and, at the election of the Borrower, the balance of such required prepayment shall be either (A) deposited in the LC Collateral Account and applied to the prepayment of Eurodollar Rate Loans on the last day of the then next-expiring Interest Period for Eurodollar Rate Loans (with all interest accruing thereon for the account of the Borrower) or (B) prepaid immediately, together with any amounts owing to the Lenders under <u>Section 2.10</u>. Notwithstanding any such deposit in the LC Collateral Account, interest shall continue to accrue on such Revolving Loans until prepayment.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 82 of 432

Notice of Prepayment. The Borrower shall notify the Administrative Agent (and, in the case of (d) prepayment of a Swingline Loan, the Swingline Lender) by telephone (confirmed by telecopy) of any prepayment hereunder (i) in the case of prepayment of a Borrowing of Eurodollar Rate Loans, not later than 1:00 p.m., New York City time, three Business Days before the date of prepayment, (ii) in the case of prepayment of a Borrowing of Base Rate Loans, not later than 4:00 p.m., New York City time, on the date of prepayment or (iii) in the case of prepayment of a Swingline Loan, not later than 1:00 p.m., New York City time, on the date of prepayment. Each such notice shall specify the prepayment date, the principal amount of each Borrowing or portion thereof to be prepaid and, in the case of a mandatory prepayment, a reasonably detailed calculation of the amount of such prepayment. Each notice of prepayment pursuant to this Section shall be irrevocable, except that the Borrower may, by subsequent notice to the Administrative Agent, revoke any such notice of prepayment if such notice of revocation is received not later than 10:00 a.m. (New York City time) on the day on which such prepayment is scheduled to occur and, provided that (i) the Borrower reimburses each Lender pursuant to Section 3.05 for any funding losses within five Business Days after receiving written demand therefor and (ii) the amount of Revolving Loans as to which such revocation applies shall be deemed converted to (or continued as, as applicable) Base Rate Loans in accordance with the provisions of Section 2.08 as of the date of notice of revocation (subject to subsequent conversion in accordance with the provisions of this Agreement). Promptly following receipt of any such notice (other than a notice relating solely to Swingline Loans), the Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment of any Borrowing shall be in an amount that would be permitted in the case of an advance of a Borrowing of the same Type as provided in Section 2.02, except as necessary to apply fully the required amount of a mandatory prepayment. Each prepayment of a Borrowing shall be applied ratably to the Revolving Loans included in the prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by Section 2.06.

2.10 Payments Generally; Pro Rata Treatment; Sharing of Set-offs.

(a) The Borrower shall make each payment required to be made by it hereunder or under any other Loan Document (whether of principal, interest, fees or reimbursement of LC Disbursements, or of amounts payable under Sections 3.02, 3.03, 3.04 and 3.05 or otherwise) at or before the time expressly required hereunder or under such other Loan Document for such payment (or, if no such time is expressly required, prior to 2:00 p.m., New York City time), on the date when due, in immediately available funds, without setoff or counterclaim. Any amounts received after such time on any date may, in the reasonable discretion of the Administrative Agent, be deemed to have been received on the next succeeding Business Day for purposes of calculating interest thereon. All such payments shall be made to the Administrative Agent at the Payment Office, except payments to be made directly to the Issuing Bank or Swingline Lender as expressly provided herein and except that payments pursuant to Sections 3.02, 3.03, 3.04, 3.05 and 11.04 shall be made to the Administrative Agent for the benefit of to the Persons entitled thereto and payments pursuant to other Loan Documents shall be made to the Administrative Agent for the benefit of the Persons specified therein. The Administrative Agent shall distribute any such payments received by it for the account of any other Person to the appropriate recipient promptly following receipt thereof. If any payment under any Loan Document shall be due on a day that is not a Business Day, the date for payment shall be extended to the next succeeding Business Day, and, in the case of any payment accruing interest, interest thereon shall be payable for the period of such extension. All payments under each Loan Document shall be made in Dollars.

(b) If at any time insufficient funds are received by and available to the Administrative Agent to pay fully all amounts of principal, unreimbursed LC Disbursements, interest and fees then due hereunder, such funds shall be applied in the manner as provided in <u>Section 2.09(c)</u> or 8.03 hereof, as applicable, ratably among the parties entitled thereto.

(c) If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of its Revolving Loans or participations in LC Disbursements or Swingline Loans resulting in such Lender receiving payment of a greater proportion of the aggregate amount of its Revolving Loans and participations in LC Disbursements and Swingline Loans and accrued interest thereon than the proportion received by any other Lender, then the Lender receiving such greater proportion shall purchase (for cash at face value) participations in the Revolving Loans and participations in LC Disbursements and Swingline Loans of other Lenders to the extent necessary so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Revolving Loans and participations in LC Disbursements and Swingline Loans; provided that (i) if any such participations are

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 83 of 432

purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest, and (ii) the provisions of this paragraph shall not be construed to apply to any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Revolving Loans or participations in LC Disbursements to any assignee or participant, other than to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this paragraph shall apply). The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Loan Parties rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of a Loan Party in the amount of such participation.

(d) Unless the Administrative Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Administrative Agent for the account of the Lenders or the Issuing Bank hereunder that the Borrower will not make such payment, the Administrative Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders or the Issuing Bank, as the case may be, the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders or the Issuing Bank, as the case may be, the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders or the Issuing Bank, as the case may be, severally agrees to repay to the Administrative Agent forthwith on demand the amount so distributed to such Lender or Issuing Bank with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Administrative Agent, at the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation.

(e) If any Lender shall fail to make any payment required to be made by it pursuant to <u>Section 2.02(c)</u>, <u>2.02(f)</u>, <u>2.10(d)</u>, <u>2.12(d)</u> or <u>2.13(d)</u>, then the Administrative Agent may, in its discretion (notwithstanding any contrary provision hereof), apply any amounts thereafter received by the Administrative Agent for the account of such Lender to satisfy such Lender's obligations under such Sections until all such unsatisfied obligations are fully paid.

2.11 <u>Defaulting Lenders</u>.

(a) <u>Reallocation of Pro Rata Share; Amendments</u>. For purposes of determining the Lenders' obligations to fund or acquire participations in Revolving Loans (including Swingline Loans) or Letters of Credit, the Administrative Agent may exclude the Revolving Commitments and Revolving Loans of any Defaulting Lender(s) from the calculation of Pro Rata Shares and any Revolving Commitments or Fronting Exposure of any such Defaulting Lender shall automatically be reallocated among the non-Defaulting Lenders pro rata in accordance with their Revolving Commitments up to an amount such that the Revolving Exposure of each non-Defaulting Lender does not exceed its Revolving Commitments, so long as the conditions set forth in <u>Section 4.02</u> are satisfied at the time of such reallocation. A Defaulting Lender shall have no right to vote on any amendment, waiver or other modification of a Loan Document, except as provided in <u>Section 11.01</u>.

Payments; Fees. Any payment of principal, interest, fees or other amounts received by the (b) Administrative Agent for the account of a Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Article VIII or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 11.08) shall be applied at such time or times as may be determined by the Administrative Agent as follows: *first*, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to any applicable Issuing Banks and Swingline Lenders hereunder; *third*, if so reasonably determined by the Administrative Agent or reasonably requested by the applicable Issuing Bank or Swingline Lender, to be held as Cash Collateral for the Fronting Exposure of such Defaulting Lender; *fourth*, to the funding of any Revolving Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so determined by the Administrative Agent or the Borrower, to be held in a deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Revolving Loans under this Agreement and to Cash Collateralize any Issuing Bank's Fronting Exposure with respect to such Defaulting Lender; sixth, to the payment of any amounts owing to the Lenders, the Issuing Banks or Swingline Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender, any Issuing Bank or any Swingline Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 84 of 432

under this Agreement; seventh, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and *eighth*, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Revolving Loans or LC Obligations in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Revolving Loans or LC Obligations were made at a time when the conditions set forth in Section 4.02 were satisfied or waived, such payment shall be applied solely to pay the Revolving Loans of, and LC Obligations owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Revolving Loans of, or LC Obligations owed to, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this Section 2.11(b) shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto. A Lender shall not be entitled to receive any fees accruing hereunder during the period in which it is a Defaulting Lender, and the unfunded portion of its Revolving Commitment shall be disregarded for purposes of calculating the Unused Line Fee Rate under Section 2.05(a). To the extent any LC Obligations owing to a Defaulting Lender are reallocated to other Lenders, Letter of Credit fees attributable to such LC Obligations under Section 2.05(c) shall be paid to such other Lenders. The Borrower shall not be required to pay any Letter of Credit fees attributable to LC Obligations that are not so reallocated; provided that if the Borrower has not Cash Collateralized the Fronting Exposure associated with a Defaulting Lender pursuant to Section 2.13(j)(ii), then the Borrower shall pay such Letter of Credit fees to the Administrative Agent.

(c) <u>Cure</u>. The Borrower, the Administrative Agent and the Issuing Bank may agree in writing that a Lender is no longer a Defaulting Lender. At such time, Pro Rata Shares shall be reallocated without exclusion of such Lender's Revolving Commitments and Revolving Loans, and all outstanding Revolving Loans, LC Obligations and other exposures under the Revolving Commitments shall be reallocated among the Lenders and settled by the Administrative Agent (with appropriate payments by the reinstated Lender) in accordance with the readjusted Pro Rata Shares. Unless expressly agreed in writing by the Borrower, the Administrative Agent and the Issuing Bank (each of which shall make such determination, in its sole discretion), no reinstatement of a Defaulting Lender shall constitute a waiver or release of claims against such Lender. The failure of any Lender to fund a Revolving Loan, to make a payment in respect of LC Obligations or otherwise to perform its obligations hereunder shall not relieve any other Lender of its obligations, and no Lender shall be responsible for default by another Lender. Subject to <u>Section 11.23</u>, no reallocation hereunder shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender arising from that Lender having become a Defaulting Lender, including any claim of a non-Defaulting Lender as a result of such non-Defaulting Lender's increased exposure following such reallocation.

2.12 <u>Swingline Loans</u>.

(a) <u>Swingline Commitment</u>. Subject to the terms and conditions set forth herein, the Swingline Lender may, but shall not be obligated to, make Swingline Loans to the Borrower from time to time during the Revolving Availability Period, in an aggregate principal amount at any time outstanding that will not result in (i) the aggregate principal amount of outstanding Swingline Loans exceeding \$20,000,000 or (ii) the Aggregate Exposures exceeding the Line Cap; <u>provided</u> that the Swingline Lender shall not be required to make a Swingline Loan to refinance an outstanding Swingline Loan. Within the foregoing limits and subject to the terms and conditions set forth herein, the Borrower may borrow, repay and reborrow Swingline Loans.

(b) Swingline Loans. To request a Swingline Loan, the Borrower shall notify the Administrative Agent of such request by telephone (confirmed by telecopy), not later than 4:00 p.m., New York City time, on the day of a proposed Swingline Loan. Each such notice shall be irrevocable and specify the requested date (which shall be a Business Day) and amount of the requested Swingline Loan. The Administrative Agent will promptly advise the Swingline Lender of any such notice received from the Borrower. The Swingline Lender shall make each Swingline Loan available to the Borrower by means of a credit to the general deposit account of the Borrower with the Swingline Lender (or, in the case of a Swingline Loan made to finance the reimbursement of an LC Disbursement as provided in Section 2.13(e), by remittance to the Issuing Bank) by 5:00 p.m., New York City time, on the requested date of such Swingline Loan. The Borrower shall not request a Swingline Loan if at the time of and immediately after giving effect to such request a Default has occurred and is continuing. Swingline Loans shall be made in minimum amounts of \$100,000.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 85 of 432

(c) <u>Prepayment</u>. The Borrower shall have the right at any time and from time to time to repay, without premium or penalty, any Swingline Loan, in whole or in part, upon giving written or telecopy notice (or telephone notice promptly confirmed by written, or telecopy notice) to the Swingline Lender and to the Administrative Agent before 4:00 p.m., New York City time on the date of repayment at the Swingline Lender's address for notices specified in the Swingline Lender's administrative questionnaire. All principal payments of Swingline Loans shall be accompanied by accrued interest on the principal amount being repaid to the date of payment.

(d) Participations. The Swingline Lender may by written notice given to the Administrative Agent not later than 4:00 p.m., New York City time, on any Business Day require the Revolving Lenders to acquire participations on such Business Day in all or a portion of the Swingline Loans outstanding. Such notice shall specify the aggregate amount of Swingline Loans in which Revolving Lenders will participate. Promptly upon receipt of such notice, the Administrative Agent will give notice thereof to each Revolving Lender, specifying in such notice such Lender's Pro Rata Percentage of such Swingline Loan(s). Each Revolving Lender hereby absolutely and unconditionally agrees, upon receipt of notice as provided above, to pay to the Administrative Agent, for the account of the Swingline Lender, such Lender's Pro Rata Percentage of such Swingline Loan(s). Each Revolving Lender acknowledges and agrees that its obligation to acquire participations in Swingline Loans pursuant to this paragraph is absolute and unconditional and shall not be affected by any circumstance whatsoever, including the occurrence and continuance of a Default or reduction or termination of the Aggregate Commitments or whether an Overadvance exists or is created thereby, and that each such payment shall be made without any offset, abatement, withholding or reduction whatsoever (provided that such payment shall not cause such Lender's Revolving Exposure to exceed such Lender's Revolving Commitment). Each Revolving Lender shall comply with its obligation under this paragraph by wire transfer of immediately available funds, in the same manner as provided in Section 2.02(f) with respect to Revolving Loans made by such Lender (and Section 2.02 shall apply, *mutatis mutandis*, to the payment obligations of the Revolving Lenders), and the Administrative Agent shall promptly pay to the Swingline Lender the amounts so received by it from the Revolving Lenders. The Administrative Agent shall notify the Borrower of any participations in any Swingline Loan acquired pursuant to this paragraph, and thereafter payments in respect of such Swingline Loan shall be made to the Administrative Agent and not to the Swingline Lender. Any amounts received by the Swingline Lender from the Borrower (or other party on behalf of the Borrower) in respect of a Swingline Loan after receipt by the Swingline Lender of the proceeds of a sale of participations therein shall be promptly remitted to the Administrative Agent; any such amounts received by the Administrative Agent shall be promptly remitted by the Administrative Agent to the Revolving Lenders that shall have made their payments pursuant to this paragraph and to the Swingline Lender, as their interests may appear. The purchase of participations in a Swingline Loan pursuant to this paragraph shall not relieve the Borrower of any default in the payment thereof.

(e) If the Maturity Date shall have occurred at a time when Extended Revolving Loan Commitmentsare in effect, then on the Maturity Date all then outstanding Swingline Loans shall be repaid in full on such date (andthere shall be no adjustment to the participations in such Swingline Loans as a result of the occurrence of such-Maturity Date); <u>provided</u> that, if on the occurrence of the Maturity Date (after giving effect to any repayments of Revolving Loans and any reallocation of Letter of Credit participations as contemplated in <u>Section 2.13(o)</u>), there shall exist sufficient unutilized Extended Revolving Loan Commitments so that the respective outstanding Swingline Loans could be incurred pursuant to the Extended Revolving Loan Commitments which will remain in effect after the occurrence of the Maturity Date, then there shall be an automatic adjustment on such date of the participations in such Swingline Loans and same shall be deemed to have been incurred solely pursuant to the Extended Revolving Loan Commitments and such Swingline Loans shall not be so required to be repaid in full on the Maturity Date.

2.13 Letters of Credit.

(a) <u>General</u>. Subject to the terms and conditions set forth herein, the Borrower may request the issuance of Letters of Credit for the Borrower's account or the account of a Subsidiary of the Borrower in a form reasonably acceptable to the Administrative Agent and the Issuing Bank, at any time and from time to time during the Revolving Availability Period (<u>provided</u> that the Borrower shall be a co-applicant with respect to each Letter of Credit issued for the account of or in favor of a Subsidiary). All Existing Letters of Credit shall be deemed, without further action by any party hereto, to have been issued on the Closing Date pursuant to this Agreement, and the Lenders shall thereupon acquire participations in the Existing Letters of Credit as if so issued without further action by any party hereto, to be acquired by the Lenders hereto. In the event of any inconsistency between the terms and conditions of this Agreement and the terms and conditions of any form of letter of credit application or other

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 86 of 432

agreement submitted by the Borrower to, or entered into by the Borrower with, the Issuing Bank relating to any Letter of Credit, the terms and conditions of this Agreement shall control.

Request for Issuance, Amendment, Renewal, Extension; Certain Conditions. To request the (b) issuance of a Letter of Credit or the amendment, renewal or extension of an outstanding Letter of Credit, the Borrower shall hand deliver or telecopy (or transmit by electronic communication, if arrangements for doing so have been approved by the Issuing Bank) a LC Request to the Issuing Bank and the Administrative Agent not later than 1:00 p.m. on the second Business Day preceding the requested date of issuance, amendment, renewal or extension (or such later date and time as is reasonably acceptable to the Issuing Bank). A request for an initial issuance of a Letter of Credit shall specify in form and detail reasonably satisfactory to the Issuing Bank: (i) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day); (ii) the amount thereof; (iii) the expiry date thereof; (iv) the name and address of the beneficiary thereof; (v) the documents to be presented by such beneficiary in case of any drawing thereunder; (vi) the full text of any certificate to be presented by such beneficiary in case of any drawing thereunder; and (vii) such other matters as the Issuing Bank may reasonably require. A request for an amendment, renewal or extension of any outstanding Letter of Credit shall specify in form and detail reasonably satisfactory to the Issuing Bank (w) the Letter of Credit to be amended, renewed or extended, (x) the proposed date of amendment, renewal or extension thereof (which shall be a Business Day), (y) the nature of the proposed amendment, renewal or extension, and (z) such other matters as the Issuing Bank may reasonably require. If requested by the Issuing Bank, the Borrower also shall submit a letter of credit application substantially on the Issuing Bank's standard form in connection with any request for a Letter of Credit. A Letter of Credit shall be issued, amended, renewed or extended only if, and upon issuance, amendment, renewal or extension of each Letter of Credit the Borrower shall be deemed to represent and warrant that, after giving effect to such issuance, amendment, renewal or extension (A) the LC Exposure shall not exceed \$35,000,000, (B) the total Revolving Exposures shall not exceed the lesser of (1) the total Revolving Commitments and (2) the Borrowing Base then in effect and (C) if a Defaulting Lender exists, either such Lender or the Borrower has entered into arrangements satisfactory to the Administrative Agent and Issuing Bank to eliminate any Fronting Exposure associated with such Lender. Unless the Issuing Bank shall otherwise agree, no Letter of Credit shall be denominated in a currency other than Dollars.

(c) Expiration Date. Each Letter of Credit shall expire at or prior to the close of business on the earlier of the date which is one year after the date of the issuance of such Letter of Credit (or such other longer period of time as the Administrative Agent and the applicable Issuing Bank may agree and, in the case of any renewal or extension thereof, one (1) year after such renewal or extension) and, unless Cash Collateralized or otherwise credit supported to the reasonable satisfaction of the Administrative Agent and the applicable Issuing Bank (in which case the expiry may extend no longer than twelve (12) months after the Letter of Credit Expiration Date), the Letter of Credit Expiration Date. Each Letter of Credit may, upon the request of the Borrower, include a provision whereby such Letter of Credit shall be renewed automatically for additional consecutive periods of twelve (12) months or less (but, subject to the foregoing, not beyond the date that is after the Letter of Credit Expiration Date) unless the applicable Issuing Bank notifies the beneficiary thereof at least thirty (30) days prior to the then-applicable expiration date that such Letter of Credit will not be renewed.

(d) Participations. By the issuance of a Letter of Credit (or an amendment to a Letter of Credit increasing the amount thereof) and without any further action on the part of the Issuing Bank or the Lenders, the Issuing Bank hereby grants to each Revolving Lender, and each Revolving Lender hereby acquires from the Issuing Bank, a participation in such Letter of Credit equal to such Lender's Pro Rata Percentage of the aggregate amount available to be drawn under such Letter of Credit. In consideration and in furtherance of the foregoing, each Revolving Lender hereby absolutely and unconditionally agrees to pay to the Administrative Agent, for the account of the Issuing Bank, such Lender's Pro Rata Percentage of each LC Disbursement made by the Issuing Bank and not reimbursed by the Borrower on the date due as provided in paragraph (e) of this Section 2.13, or of any reimbursement payment required to be refunded to the Borrower for any reason. Each Lender acknowledges and agrees that its obligation to acquire participations pursuant to this paragraph in respect of Letters of Credit is absolute and unconditional and shall not be affected by any circumstance whatsoever, including any amendment, renewal or extension of any Letter of Credit or the occurrence and continuance of a Default or reduction or termination of the Aggregate Commitments or whether or not an Overadvance exists or is created thereby, and that each such payment shall be made without any offset, abatement, withholding or reduction whatsoever.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 87 of 432

Reimbursement. If the Issuing Bank shall make any LC Disbursement in respect of a Letter of (e) Credit, the Borrower shall reimburse such LC Disbursement by paying to the Issuing Bank an amount equal to such LC Disbursement not later than 2:00 p.m., New York City time, on the Business Day after receiving notice from the Issuing Bank of such LC Disbursement; provided that, whether or not the Borrower submits a Notice of Borrowing, the Borrower shall be deemed to have requested (except to the extent the Borrower makes payment to reimburse such LC Disbursement when due) a Borrowing of Base Rate Loans in an amount necessary to reimburse such LC Disbursement. If the Borrower fails to make such payment when due, the Issuing Bank shall notify the Administrative Agent and the Administrative Agent shall notify each Revolving Lender of the applicable LC Disbursement, the payment then due from the Borrower in respect thereof and such Lender's Pro Rata Percentage thereof. Promptly following receipt of such notice, each Revolving Lender shall pay to the Administrative Agent its Pro Rata Percentage of the unreimbursed LC Disbursement (such aggregate amount, the "Unreimbursed Amount") in the same manner as provided in Section 2.02(f) with respect to Revolving Loans made by such Lender, and the Administrative Agent shall promptly pay to the Issuing Bank the amounts so received by it from the Revolving Lenders. Promptly following receipt by the Administrative Agent of any payment from the Borrower pursuant to this paragraph, the Administrative Agent shall, to the extent that Revolving Lenders have made payments pursuant to this paragraph to reimburse the Issuing Bank, distribute such payment to such Lenders and the Issuing Bank as their interests may appear. Any payment made by a Revolving Lender pursuant to this paragraph to reimburse the Issuing Bank for any LC Disbursement (other than the funding of Base Rate Loans or a Swingline Loan as contemplated above) shall not constitute a Revolving Loan and shall not relieve the Borrower of its obligation to reimburse such LC Disbursement.

(f) <u>Obligations Absolute</u>.

<u>(i)</u> Subject to the limitations set forth below, the obligation of the Borrower to reimburse LC Disbursements as provided in paragraph (e) of this Section 2.13 shall be absolute, unconditional and irrevocable, and shall be performed strictly in accordance with the terms of this Agreement under any and all circumstances whatsoever and irrespective of (i) any lack of validity or enforceability of any Letter of Credit or this Agreement, or any term or provision therein, (ii) any draft or other document presented under a Letter of Credit proving to be forged, fraudulent or invalid in any respect or any statement therein being untrue or inaccurate in any respect, (iii) payment by the Issuing Bank under a Letter of Credit against presentation of a draft or other document that does not strictly comply with the terms of such Letter of Credit, (iv) the existence of any claim, setoff, defense or other right which the Borrower may have at any time against a beneficiary of any Letter of Credit, or (v) any other event or circumstance whatsoever, whether or not similar to any of the foregoing, that might, but for the provisions of this Section 2.13, constitute a legal or equitable discharge of, or provide a right of setoff against, the obligations of the Borrower hereunder; provided that the Borrower shall have no obligation to reimburse the Issuing Bank to the extent that such payment was made in error due to the gross negligence, bad faith, or willful misconduct of the Issuing Bank (as determined by a court of competent jurisdiction or another independent tribunal having jurisdiction). Neither the Administrative Agent, the Lenders nor the Issuing Bank, nor any of their Affiliates, shall have any liability or responsibility by reason of or in connection with the issuance or transfer of any Letter of Credit or any payment or failure to make any payment thereunder (irrespective of any of the circumstances referred to in the preceding sentence), or any error, omission, interruption, loss or delay in transmission or delivery of any draft, notice or other communication under or relating to any Letter of Credit (including any document required to make a drawing thereunder), any error in interpretation of technical terms or any consequence arising from causes beyond the control of the Issuing Bank: provided that the foregoing shall not be construed to excuse the Issuing Bank from liability to the Borrower to the extent of any direct damages (as opposed to consequential damages, claims in respect of which are hereby waived by the Borrower to the extent permitted by applicable Law) suffered by the Borrower that are caused by the Issuing Bank's failure to exercise care when determining whether drafts and other documents presented under a Letter of Credit comply with the terms thereof. The parties hereto expressly agree that, in the absence of gross negligence, willful misconduct, or bad faith on the part of the Issuing Bank (as determined by a court of competent jurisdiction or another independent tribunal having jurisdiction), the Issuing Bank shall be deemed to have exercised care in each such determination. In furtherance of the foregoing and without limiting the generality thereof, the parties agree that, with respect to documents presented which appear on their face to be in substantial compliance with the terms of a Letter of Credit, the Issuing Bank may, in its sole discretion, either accept and make payment upon such documents without responsibility for further investigation, regardless of any notice or information to the contrary, or refuse to accept and make payment upon such documents if such documents are not in strict compliance with the terms of such Letter of Credit.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 88 of 432

(ii) The Issuing Bank does not assume any responsibility for any failure or delay in performance or any breach by the Borrower or other Person of any obligations under any LC Document. The Issuing Bank does not make to the Lenders any express or implied warranty, representation or guaranty with respect to the Collateral, such documents or any Loan Party. The Issuing Bank shall not be responsible to any Lender for any recitals, statements, information, representations or warranties contained in, or for the execution, validity, genuineness, effectiveness or enforceability of any LC Document; the validity, genuineness, enforceability, collectability, value or sufficiency of any Collateral or the perfection of any Lien therein; or the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any Loan Party.

(iii) No Issuing Bank or any of its Affiliates, or any of their respective officers, directors, employees, agents and investment advisors, shall be liable to any Lender or other Person for any action taken or omitted to be taken in connection with any LC Documents except as a result of its actual gross negligence or willful misconduct as determined by court of competent jurisdiction in a final nonappealable judgment. The Issuing Bank shall not have any liability to any Lender if the Issuing Bank refrains from any action under any Letter of Credit or such LC Documents until it receives written instructions from the Required Lenders.

(g) <u>Disbursement Procedures</u>. The Issuing Bank shall, promptly following its receipt thereof, examine all documents purporting to represent a demand for payment under a Letter of Credit. The Issuing Bank shall promptly notify the Administrative Agent and the Borrower by telephone (confirmed by telecopy) of such demand for payment and whether the Issuing Bank has made or will make an LC Disbursement thereunder; <u>provided</u> that any failure to give or delay in giving such notice shall not relieve the Borrower of its obligation to reimburse the Issuing Bank and the Revolving Lenders with respect to any such LC Disbursement (other than with respect to the timing of such reimbursement obligation set forth in <u>Section 2.13(e)</u>).

(h) Interim Interest. If the Issuing Bank shall make any LC Disbursement, then, unless the Borrower shall reimburse such LC Disbursement in full on the date such LC Disbursement is made (whether with the proceeds of a Revolving Loan or otherwise), the unpaid amount thereof shall bear interest, for each day from and including the date such LC Disbursement is made to but excluding the date that the Borrower reimburses such LC Disbursement, at the rate per annum then applicable to Base Rate Loans; provided that, if the Borrower fails to reimburse such LC Disbursement when due pursuant to Section 2.13(e), then Section 2.06(c) shall apply. Interest accrued pursuant to this paragraph shall be for the account of the Issuing Bank, except that interest accrued on and after the date of payment by any Revolving Lender pursuant to Section 2.13(e) to reimburse the Issuing Bank shall be for the account of such Lender to the extent of such payment.

Resignation or Removal of the Issuing Bank. The Issuing Bank may resign as Issuing Bank (i) hereunder at any time upon at least 30 days' prior written notice to the Lenders, the Administrative Agent and the Borrower. The Issuing Bank may be replaced at any time by agreement between the Borrower and the Administrative Agent, provided that so long as no Default or Event of Default exists, such successor Issuing Bank shall be reasonably acceptable to the Borrower. One or more Lenders may be appointed as additional Issuing Banks in accordance with subsection (k) below. The Administrative Agent shall notify the Lenders of any such replacement of the Issuing Bank or any such additional Issuing Bank. At the time any such resignation or replacement shall become effective, the Borrower shall pay all unpaid fees accrued for the account of the replaced Issuing Bank pursuant to <u>Section 2.05(c)</u>. From and after the effective date of any such resignation or replacement or addition, as applicable, (i) the successor or additional Issuing Bank shall have all the rights and obligations of the Issuing Bank under this Agreement with respect to Letters of Credit to be issued thereafter and (ii) references herein to the term "Issuing Bank" shall be deemed to refer to such successor or such addition or to any previous Issuing Bank, or to such successor or such additional Issuing Bank and all previous Issuing Banks, as the context shall require. After the resignation or replacement of an Issuing Bank hereunder, the replaced Issuing Bank shall remain a party hereto and shall continue to have all the rights and obligations of an Issuing Bank under this Agreement with respect to Letters of Credit issued by it prior to such resignation or replacement, but shall not be required to issue additional Letters of Credit. If at any time there is more than one Issuing Bank hereunder, the Borrower may, in its discretion, select which Issuing Bank is to issue any particular Letter of Credit.

(j) <u>Cash Collateralization</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 89 of 432

If any Specified Event of Default shall occur and be continuing, on the Business Day that the (i) Borrower receives notice from the Administrative Agent (acting at the request of the Required Lenders) demanding the deposit of Cash Collateral pursuant to this paragraph, the Borrower shall deposit in the LC Collateral Account, in the name of the Administrative Agent and for the benefit of the Secured Parties, an amount in cash equal to 103.0% of the LC Exposure as of such date. Each such deposit shall be held by the Administrative Agent as collateral for the payment and performance of the obligations of the Borrower under this Agreement, but shall be immediately released and returned to the Borrower (in no event later than two (2) Business Days) once all Specified Events of Default are cured or waived. The Administrative Agent shall have exclusive dominion and control, including the exclusive right of withdrawal, over such account. Other than any interest earned on the investment of such deposits, which investments shall be made only in Cash Equivalents and at the direction of the Borrower and at the Borrower's risk and expense, such deposits shall not bear interest. Interest or profits, if any, on such investments shall accumulate in such account. Moneys in such account shall be applied by the Administrative Agent to reimburse the Issuing Bank for LC Disbursements for which it has not been reimbursed and, to the extent not so applied, shall be held for the satisfaction of the reimbursement obligations of the Borrower for the LC Exposure at such time or, if the maturity of the Revolving Loans has been accelerated (but subject to the consent of Revolving Lenders with LC Exposure representing greater than 50% of the total LC Exposure), be applied to satisfy other Obligations of the Borrower.

(ii) The Borrower shall, on demand by the Issuing Bank or the Administrative Agent from time to time, Cash Collateralize the Fronting Exposure associated with any Defaulting Lender.

(k) Additional Issuing Banks. The Borrower may, at any time and from time to time with the consent of the Administrative Agent (which consent shall not be unreasonably withheld, delayed or conditioned) and such Lender, designate one or more additional Lenders to act as an issuing bank under the terms of this Agreement. Any Lender designated as an issuing bank pursuant to this <u>paragraph (k)</u> shall be deemed (in addition to being a Lender) to be the Issuing Bank with respect to Letters of Credit issued or to be issued by such Lender, and all references herein and in the other Loan Documents to the term "Issuing Bank," shall, with respect to such Letters of Credit, be deemed to refer to such Lender in its capacity as Issuing Bank, as the context shall require.

(l) The Issuing Bank shall be under no obligation to issue any Letter of Credit if:

(i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuing Bank from issuing such Letter of Credit, or any law applicable to the Issuing Bank or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuing Bank shall prohibit, or request that the Issuing Bank refrain from, the issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon the Issuing Bank with respect to such Letter of Credit any restriction, reserve or capital requirement (for which the Issuing Bank is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon the Issuing Bank any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the Issuing Bank in good faith deems material to it; or

(ii) the issuance of such Letter of Credit would violate one or more policies of the Issuing Bank.

(m) The Issuing Bank shall be under no obligation to amend any Letter of Credit if (i) the Issuing Bank would have no obligation at such time to issue such Letter of Credit in its amended form under the terms hereof, or (ii) the beneficiary of such Letter of Credit does not accept the proposed amendment to such Letter of Credit.

(n) <u>LC Collateral Account</u>.

(i) The Administrative Agent is hereby authorized to establish and maintain at the Payment Office, in the name of the Administrative Agent and pursuant to a dominion and control agreement, a restricted deposit account designated "The Borrower LC Collateral Account." Each Loan Party shall deposit into the LC Collateral Account from time to time the Cash Collateral required to be deposited under <u>Section 2.13(j)</u> hereof.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 90 of 432

(ii) The balance from time to time in such LC Collateral Account shall constitute part of the Collateral and shall not constitute payment of the Obligations until applied as hereinafter provided. Notwithstanding any other provision hereof to the contrary, all amounts held in the LC Collateral Account shall constitute collateral security first for the liabilities in respect of Letters of Credit outstanding from time to time and second for the other Obligations hereunder until such time as all Letters of Credit shall have been terminated and all of the liabilities in respect of Letters of Credit in full. All funds in "The Borrower LC Collateral Account" may be invested in accordance with the provisions of Section 2.13(j).

(o) <u>Extended Commitments</u>. If the Maturity Date shall have occurred at a time when Extended-Revolving Loan Commitments are in effect, then (i) such Letters of Credit shall automatically be deemed to havebeen issued (including for purposes of the obligations of the Lenders to purchase participations therein and to make payments in respect thereof pursuant to <u>Sections 2.13(d)</u> and (e)) under (and ratably participated in by Lendersproviding) the Extended Revolving Loan Commitments, up to an aggregate amount not to exceed the aggregateprincipal amount of the unutilized Extended Revolving Loan Commitments thereunder at such time (it being understood that no partial face amount of any Letter of Credit may be so reallocated) and (ii) to the extent notreallocated pursuant to the immediately preceding clause (i), the Borrower shall Cash Collateralize any such Letterof Credit in accordance with <u>Section 2.13(j)</u>. Except to the extent of reallocations of participations pursuant to the prior sentence, the occurrence of the Maturity Date with respect to Existing Revolving Loans shall have no effectupon (and shall not diminish) the percentage participations of the Lenders of Extended Revolving Loans in any-Letter of Credit issued before the Maturity Date.

2.14 <u>Settlement Amongst Lenders</u>.

(a) The Swingline Lender may, at any time (but, in any event shall weekly), on behalf of the Borrower (which hereby authorizes the Swingline Lender to act on its behalf in that regard) request the Administrative Agent to cause the Lenders to make a Revolving Loan (which shall be a Base Rate Loan) in an amount equal to such Lender's Pro Rata Percentage of the Outstanding Amount of Swingline Loans, which request may be made regardless of whether the conditions set forth in <u>Section 4.02</u> have been satisfied. Upon such request, each Lender shall make available to the Administrative Agent the proceeds of such Revolving Loan for the account of the Swingline Lender. If the Swingline Lender requires a Revolving Loan to be made by the Lenders and the request therefor is received prior to 12:00 Noon on a Business Day, such transfers shall be made in immediately available funds no later than 3:00 p.m. that day; and, if the request therefor is received after 12:00 Noon, then no later than 3:00 p.m. on the next Business Day. The obligation of each such Lender to transfer such funds is irrevocable, unconditional and without recourse to or warranty by the Administrative Agent or the Swingline Lender. If and to the extent any Lender shall not have so made its transfer to the Administrative Agent, such Lender agrees to pay to the Administrative Agent, forthwith on demand, such amount, together with interest thereon, for each day from such date until the date such amount is paid to the Administrative Agent, at the Federal Funds Rate.

(b) The amount of each Lender's Pro Rata Percentage of outstanding Revolving Loans (including Swingline Loans) shall be computed weekly (or more frequently in the Administrative Agent's discretion) and shall be adjusted upward or downward based on all Revolving Loans (including Swingline Loans) and repayments of Revolving Loans (including Swingline Loans) received by the Administrative Agent as of 3:00 p.m. on the first Business Day (such date, the "Settlement Date") following the end of the period specified by the Administrative Agent.

(c) The Administrative Agent shall deliver to each of the Lenders promptly after a Settlement Date a summary statement of the amount of outstanding Revolving Loans (including Swingline Loans) for the period and the amount of repayments received for the period. As reflected on the summary statement, (i) the Administrative Agent shall transfer to each Lender its applicable Pro Rata Percentage of repayments, and (ii) each Lender shall transfer to the Administrative Agent (as provided below) or the Administrative Agent shall transfer to each Lender, such amounts as are necessary to insure that, after giving effect to all such transfers, the amount of Revolving Loans made by each Lender with respect to Revolving Loans to the Borrower (including Swingline Loans) shall be equal to such Lender's applicable Pro Rata Percentage of Revolving Loans (including Swingline Loans) outstanding as of such Settlement Date. If the summary statement requires transfers to be made to the Administrative Agent by the Lenders and is received prior to 12:00 Noon on a Business Day, such transfers shall be made in immediately

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 91 of 432

available funds no later than 3:00 p.m. that day; and, if received after 12:00 Noon, then no later than 3:00 p.m. on the next Business Day. The obligation of each Lender to transfer such funds is irrevocable, unconditional and without recourse to or warranty by the Administrative Agent. If and to the extent any Lender shall not have so made its transfer to the Administrative Agent, such Lender agrees to pay to the Administrative Agent, forthwith on demand such amount, together with interest thereon, for each day from such date until the date such amount is paid to the Administrative Agent, at the Federal Funds Rate.

2.15 <u>Revolving Commitment Increase</u>.

(a) Subject to the terms and conditions set forth herein, after the Closing Date, the Borrower shall have the right to request, by written notice to the Administrative Agent, an increase in the Revolving Commitments-(a "<u>Revolving Commitment Increase</u>") in an aggregate amount not to exceed \$10,000,000; <u>provided</u> that (i) the Borrower shall only be permitted to request two (2) Revolving Commitment Increases during the term of this Agreement and (ii) any Revolving Commitment Increase shall be in a minimum amount of \$5,000,000.

(b) Each notice submitted pursuant to this Section 2.15 (a "Revolving Commitment Increase Notice")requesting a Revolving Commitment Increase shall specify the amount of the increase in the Revolving-Commitments being requested. Upon receipt of a Revolving Commitment Increase Notice, the Administrative Agentmay (at the direction of the Borrower) promptly notify the Lenders and each Lender may (subject to the Borrower'sconsent, not to be unreasonably withheld) have the right to elect to have its Revolving Commitment increased by its-Pro Rata Share (it being understood and agreed that a Lender may elect to have its Revolving Commitment increasedin excess of its Pro Rata Share in its discretion if any other Lender declines to participate in the Revolving-Commitment Increase) of the requested increase in Revolving Commitments; provided that (i) each Lender may elector decline, in its sole discretion, to have its Revolving Commitment increased in connection with any requested-Revolving Commitment Increase, it being understood that no Lender shall be obligated to increase its Revolving-Commitment unless it, in its sole discretion, so agrees and, if a Lender fails to respond to any Revolving-Commitment Increase Notice within five (5) Business Days after such Lender's receipt of such request, such Lendershall be deemed to have declined to participate in such Revolving Commitment Increase, (ii) if any Lender declinesto participate in any Revolving Commitment Increase and, as a result, commitments from additional financial institutions are required in connection with the Revolving Commitment Increase, any Person or Persons providingsuch commitment shall be subject to the written consent of the Administrative Agent, the Swingline Lender and the Issuing Bank (such consent not to be unreasonably withheld or delayed), if such consent would be required pursuantto the definition of Eligible Assignee and (iii) in no event shall a Defaulting Lender be entitled to participate in such-Revolving Commitment Increase. In the event that any Lender or other Person agrees to participate in any Revolving Commitment Increase (each an "Increase Loan Lender"), such Revolving Commitment Increase shallbecome effective on such date as shall be mutually agreed upon by the Increase Loan Lenders and the Borrower, which date shall be as soon as practicable after the date of receipt of the Revolving Commitment Increase Notice-(such date, the "Increase Date"); provided that the establishment of such Revolving Commitment Increase shall besubject to the satisfaction of each of the following conditions: (1) the conditions set forth in Section 4.02(a) and (b) shall be satisfied; (2) the Revolving Commitment Increase shall be effected pursuant to one or more joinderagreements executed and delivered by the Borrower, the Administrative Agent, and the Increase Loan Lenders, eachof which shall be reasonably satisfactory to the Borrower, the Administrative Agent, and the Increase Loan Lenders; (3) the Loan Parties shall execute and deliver or cause to be executed and delivered to the Administrative Agent suchamendments to the Loan Documents, customary legal opinions and other documents as the Administrative Agentmay reasonably request in connection with any such transaction, which amendments, legal opinions and otherdocuments shall be reasonably satisfactory to the Administrative Agent; (4) the Borrower shall have paid to the Administrative Agent and the Lenders such additional fees as may be agreed to be paid by the Borrower inconnection therewith; (5) such Revolving Commitment Increase shall be permitted under (A) for so long as any First-Lien Notes remain outstanding, the First Lien Notes Indenture and (B) for so long as any 2014 Second Lien Notesremain outstanding, the 2014 Second Lien Notes Indenture; (6) for so long as any 2016 Secured Notes remainoutstanding, holders of more than 50% of the outstanding principal amount of the 2016 Secured Notes shall consentin writing to the establishment of such Revolving Commitment Increase; and (7) the definition of "Maximum Senior-Priority Principal Amount" in the 2016 Intercreditor Agreement shall be amended to include an aggregate principalamount equal to the amount of such Revolving Commitment Increase.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 92 of 432

(c) On the Increase Date, upon fulfillment of the conditions set forth in this Section 2.15, (i) the Administrative Agent shall effect a settlement of all outstanding Revolving Loans among the Lenders that will reflect the adjustments to the Revolving Commitments of the Lenders as a result of the Revolving Commitment Increase, (ii) the Administrative Agent shall notify the Lenders and Loan Parties of the occurrence of the Revolving Commitment Increase to be effected on the Increase Date, (iii) <u>Schedule 2.01</u> shall be deemed modified to reflect the revised Revolving Commitments of the affected Lenders and (iv) Notes will be issued, at the expense of the Borrower, to any Lender participating in the Revolving Commitment Increase and requesting a Note.

(d) The terms and provisions of the Revolving Commitment Increase shall be identical to the-Revolving Loans and the Revolving Commitments and, for purposes of this Agreement and the other Loan-Documents, all Revolving Loans made under the Revolving Commitment Increase shall be deemed to be Revolving-Loans. Without limiting the generality of the foregoing, (i) the rate of interest applicable to the Revolving Loans, (ii) unusedline fees applicable to the Revolving Commitment Increase shall be calculated using the same Unused Line Fee-Rates applicable to the existing Revolving Loans, (iii) the Revolving Commitment Increase shall share ratably in anymandatory prepayments of the Revolving Loans, (iv) after giving effect to such Revolving Commitment Increases, Revolving Commitment shall be reduced based on each Lender's Pro Rata Percentage, and (v) the Revolving Commitment Increase shall rank *pari passu* in right of payment and security with the existing Revolving Loans. Each joinder agreement and any amendment to any Loan Document requested by the Administrative Agent inconnection with the establishment of the Revolving Commitment Increase may, without the consent of any of the Lenders, effect such amendments to this Agreement (an "Incremental Revolving Commitment Agreement") and the other Loan Documents as may be reasonably necessary or appropriate, in the opinion of the Administrative Agent and the Borrower, to effect the provisions of this Section 2.15.

2.15 2.16 [Reserved]

2.16 [Reserved]

2.17 <u>Overadvances</u>.

If the aggregate Revolving Loans outstanding exceed the Line Cap (an "Overadvance") at any time, the excess amount shall be payable by the Borrower on demand by the Administrative Agent, but all such Revolving Loans shall nevertheless constitute Obligations secured by the Collateral and entitled to all benefits of the Loan Documents. The Administrative Agent may require the Lenders to honor requests for Overadvance Loans and to forbear from requiring the Borrower to cure an Overadvance, (a) when no other Event of Default is known to the Administrative Agent, as long as (i) the Overadvance does not continue for more than 30 consecutive days (and no Overadvance may exist for at least five consecutive days thereafter before further Overadvance Loans are required) and (ii) the aggregate amount of all Overadvances and Protective Advances is not known by the Administrative Agent to exceed 10.0% of the Borrowing Base, (b) regardless of whether an Event of Default exists, if the Administrative Agent discovers an Overadvance not previously known by it to exist, as long as from the date of such discovery the Overadvance (i) is not increased by more than \$500,000, and (ii) does not continue for more than 30 consecutive days. In no event shall Overadvance Loans be required that would cause the aggregate outstanding Revolving Loans and LC Obligations to exceed the aggregate Revolving Commitments. The making of any Overadvance shall not create nor constitute a Default or Event of Default; it being understood that the making or continuance of an Overadvance shall not constitute a waiver by the Administrative Agent or the Lenders of the then existing Event of Default. In no event shall the Borrower or other Loan Party be permitted to require any Overadvance Loan to be made.

2.18 <u>Protective Advances</u>.

The Administrative Agent shall be authorized, in its discretion, following notice to and consultation with the Borrower, at any time, to make Base Rate Loans ("<u>Protective Advances</u>") (a) in an aggregate amount, together with the aggregate amount of all Overadvance Loans, not to exceed 10.0% of the Borrowing Base, if the Administrative Agent deems such Protective Advances necessary or desirable to preserve and protect the Collateral, or to enhance the collectability or repayment of the Obligations; or (b) to pay any other amounts chargeable to Loan Parties under

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 93 of 432

any Loan Documents, including costs, fees and expenses; <u>provided</u> that, the aggregate amount of outstanding Protective Advances plus the outstanding amount of Revolving Loans and LC Obligations shall not exceed the aggregate Revolving Commitments. Each Lender shall participate in each Protective Advance in accordance with its Pro Rata Percentage. Required Lenders may at any time revoke the Administrative Agent's authority to make further Protective Advances under <u>clause (a)</u> by written notice to the Administrative Agent. Absent such revocation, the Administrative Agent's determination that funding of a Protective Advance is appropriate shall be conclusive. The Administrative Agent may use the proceeds of such Protective Advances to (a) protect, insure, maintain or realize upon any Collateral; or (b) defend or maintain the validity or priority of the Administrative Agent's Liens in any Collateral, including any payment of a judgment, insurance premium, warehouse charge, finishing or processing charge, or landlord claim, or any discharge of a Lien; <u>provided</u> that the Administrative Agent shall use reasonable efforts to notify the Borrower after paying any such amount or taking any such action and shall not make payment of any item that is being properly contested.

2.19 <u>Extended Loans</u>.

(a) -Notwithstanding anything to the contrary in this Agreement, subject to the terms of this Section-2.19, the Borrower may at any time and from time to time when no Event of Default then exists request that all or a portion of the Revolving Loans (the "Existing Revolving Loans"), together with any related outstandings, beconverted to extend the scheduled maturity date(s) of any payment of principal with respect to all or any portion of the principal amount (and related outstandings) of such Revolving Loans (any such Revolving Loans which have been so converted, "Extended Revolving Loans") and to provide for other terms consistent with this Section 2.19. Inorder to establish any Extended Revolving Loans, the Borrower shall provide a notice to the Administrative Agent-(who shall provide a copy of such notice to each of the Lenders) (each, an "Extension Request") setting forth the proposed terms of the Extended Revolving Loans to be established, which shall (x) be identical as offered to each Lender (including as to the proposed interest rates and fees payable) and (v) be identical to the Existing Revolving-Loans, except that: (i) repayments of principal of the Extended Revolving Loans may be delayed to later dates thanthe Maturity Date; (ii) the Effective Yield with respect to the Extended Revolving Loans (whether in the form of interest rate margin, upfront fees, original issue discount or otherwise) may be different than the Effective Yield forthe Existing Revolving Loans to the extent provided in the applicable Extension Amendment; and (iii) the Extension-Amendment may provide for other covenants and terms that apply solely to any period after the Maturity Date that isin effect on the effective date of the Extension Amendment (immediately prior to the establishment of such Extended-Revolving Loans); provided, however, that (A) in no event shall the final maturity date of any Extended Revolving-Loans at the time of establishment thereof be earlier than the then Maturity Date of any other Revolving Loanshereunder and (B) the Weighted Average Life to Maturity of any Extended Revolving Loans at the time of establishment thereof shall be no shorter than the remaining Weighted Average Life to Maturity of any other-Revolving Loans then outstanding. Any Extended Revolving Loans converted pursuant to any Extension Requestshall be designated a series (each, an "Extension Series") of Extended Revolving Loans, as applicable, for allpurposes of this Agreement; provided that any Extended Revolving Loans converted from Existing Revolving Loansmay, to the extent provided in the applicable Extension Amendment, be designated as an increase in any previouslyestablished Extension Series with respect to such Revolving Loans.

(b) With respect to any Extended Revolving Loans, subject to the provisions of <u>Sections 2.12(e)</u> and <u>2.13(o)</u>, to the extent dealing with Swingline Loans and Letters of Credit which mature or expire after the Maturity-Date, all Swingline Loans and Letters of Credit shall be participated in on a pro rata basis by all Lenders with Revolving Loan Commitments and/or Extended Revolving Loan Commitments in accordance with their Pro Rata Share of the Aggregate Commitments (and, except as provided in <u>Sections 2.12(e)</u> and <u>2.13(o)</u>, without giving effectto changes thereto on the Maturity Date with respect to Swingline Loans and Letters of Credit theretofore incurred orissued) and all borrowings under the Aggregate Commitments and repayments thereunder shall be made on a pro ratabasis (except for (x) payments of interest and fees at different rates on Extended Revolving Loan Commitments (andrelated outstandings) and (y) repayments required upon any Maturity Date of any Revolving Commitments or <u>Extended Revolving Loan Commitments</u>).

(c) The Borrower shall provide the applicable Extension Request at least ten (10) Business Days priorto the date on which Lenders under the Existing Revolving Loans are requested to respond, and shall agree to suchprocedures, if any, as may be established by, or acceptable to, the Administrative Agent, in each case actingreasonably to accomplish the purposes of this <u>Section 2.19</u>. No Lender shall have any obligation to agree to have-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 94 of 432

any of its Existing Revolving Loans converted into Extended Revolving Loans pursuant to any Extension Request. Any Lender (each, an "<u>Extending Lender</u>") wishing to have all or a portion of its Existing Revolving Loans subject to such Extension Request converted into Extended Revolving Loans shall notify the Administrative Agent (each, an "<u>Extension Election</u>") on or prior to the date specified in such Extension Request of the amount of its Existing Revolving Loans which it has elected to request be converted into Extended Revolving Loans (subject to anyminimum denomination requirements imposed by the Administrative Agent). Any Lender that does not respond to the Extension Request on or prior to the date specified therein shall be deemed to have rejected such Extension-Request. In the event that the aggregate principal amount of Existing Revolving Loans subject to Extension-Elections relating to a particular Extension Request exceeds the amount of Extended Revolving Loans requestedpursuant to such Extension Request, Revolving Loans subject to such Extension Elections shall be converted to Extended Revolving Loans, on a pro rata basis based on the aggregate principal amount of Revolving Loansincluded in each such Extension Election or to the extent such option is expressly set forth in the respective-Extension Request, the Borrower shall have the option to increase the amount of Extended Revolving Loans so that such excess does not exist.

(d) Extended Revolving Loans shall be established pursuant to an amendment (each, a "<u>Extension</u> <u>Amendment</u>") to this Agreement among the Borrower, the Administrative Agent and each Extending Lenderproviding Extended Revolving Loans thereunder which shall be consistent with the provisions set forth in <u>Section</u> <u>2.19(a)</u> above (but which shall not require the consent of any other Lender). The Administrative Agent shall promptly notify each relevant Lender as to the effectiveness of each Extension Amendment.

(e) With respect to any extension (an "Extension") consummated by the Borrower pursuant to this-Section 2.19, (i) such Extension shall not constitute voluntary or mandatory payments or prepayments for purposesof this Agreement, (ii) with respect to Extended Revolving Loan Commitments, if the aggregate amount extended isless than (A) the LC Commitment, the LC Commitment shall be reduced upon the date that is five (5) Business Daysprior to the Maturity Date to the extent needed so that the LC Commitment does not exceed the aggregate Revolving-Commitment which would be in effect after the Maturity Date, and, if applicable, the Borrower shall Cash-Collateralize obligations under any issued Letters of Credit in an amount equal to 103.0% of the stated amount of such Letters of Credit, or (B) the Swingline Commitment, the Swingline Commitment shall be reduced upon the datethat is five (5) Business Days prior to the Maturity Date (to the extent needed so that the Swingline Commitmentdoes not exceed the aggregate Revolving Commitment which would be in effect after the Maturity Date, and, if applicable, the Borrower shall prepay any outstanding Swingline Loans. The Administrative Agent and the Lendershereby consent to each Extension and the other transactions contemplated by this Section 2.19 (including, for the avoidance of doubt, payment of any interest or fees in respect of any Extended Revolving Loan Commitments onsuch terms as may be set forth in the Extension Request) and hereby waive the requirements of any provision of this-Agreement or any other Loan Document that may otherwise prohibit any Extension or any other transactioncontemplated by this Section 2.19, provided that such consent shall not be deemed to be an acceptance of the Extension Request.

-Each of the parties hereto hereby agrees that this Agreement and the other Loan Documents may be amended pursuant to an Extension Amendment, without the consent of any other Lenders, to the extent (but onlyto the extent) necessary to (i) reflect the existence and terms of any Extended Revolving Loans incurred pursuantthereto, (ii) establish new tranches or sub tranches in respect of Revolving Loan Commitments so extended and suchtechnical amendments as may be necessary in connection with the establishment of such new tranches orsub tranches, in each case on terms consistent with this Section 2.19, and (iii) effect such other amendments to this-Agreement and the other Loan Documents as may be necessary or appropriate, in the reasonable opinion of the Administrative Agent and the Borrower, to effect the provisions of this Section, and the Required Lenders herebyexpressly authorize the Administrative Agent to enter into any such Extension Amendment. Notwithstanding theforegoing, the Administrative Agent shall have the right (but not the obligation) to seek the advice or concurrence of the Required Lenders with respect to any matter contemplated by this Section 2.19 and, if the Administrative Agentseeks such advice or concurrence, the Administrative Agent shall be permitted to enter into such amendments withthe Borrower in accordance with any instructions actually received from such Required Lenders and shall also beentitled to refrain from entering into such amendments with the Borrower unless and until it shall have received such advice or concurrence; provided, however, that whether or not there has been a request by the Administrative Agentfor any such advice or concurrence, all such amendments entered into with the Borrower by the Administrative-Agent hereunder shall be binding and conclusive on the Lenders.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 95 of 432

ARTICLE III

TAXES, YIELD PROTECTION AND ILLEGALITY

3.01 <u>Taxes</u>.

(a) <u>Payments Free of Taxes</u>. Any and all payments by, or on account of any Obligation of, any Loan Party hereunder or under any other Loan Document shall be made free and clear of and without reduction or withholding for any Taxes, <u>provided</u> that if the applicable withholding agent shall be required by applicable Law to deduct any Taxes in respect of such payments, then (i) if the Tax in question is an Indemnified Tax or Other Tax, the sum payable by the applicable Loan Party shall be increased as necessary so that after all required deductions have been made (including deductions applicable to additional sums payable under this <u>Section 3.01</u>) the Administrative Agent or any Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions been made, (ii) the applicable withholding agent shall make such deductions and (iii) the applicable withholding agent shall timely pay the full amount deducted to the relevant Governmental Authority in accordance with applicable Law.

(b) <u>Payment of Other Taxes by the Loan Parties</u>. Without limiting the provisions of <u>subsection (a)</u> above, the Loan Parties shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Law.

(c) Indemnification by the Loan Parties. The Loan Parties shall jointly and severally indemnify the Administrative Agent and each Lender, within 10 days after demand therefor, for the full amount of any Indemnified Taxes with respect to amounts payable under the Loan Documents or any Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this <u>Section 3.01</u>) paid by the Administrative Agent or such Lender, as the case may be, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

(d) <u>Evidence of Payments</u>. As soon as practicable after any payment of Indemnified Taxes or Other Taxes by any Loan Party to a Governmental Authority, the Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

Status of Lenders. Any Lender that is entitled to an exemption from or reduction of withholding (e) tax under the law of the jurisdiction in which the Borrower, Holdings or any other Loan Party, as the case may be, is resident for tax purposes, or any treaty to which such jurisdiction is a party, with respect to any payments hereunder or under any other Loan Document shall deliver to the Borrower. Holdings, such other Loan Party and the Administrative Agent, at the time or times reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent, such properly completed and executed documentation prescribed by applicable Law, or reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent, as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if requested by the Borrower, Holdings, any other Loan Party or the Administrative Agent, shall deliver such other documentation prescribed by applicable Law or reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent as will enable the Borrower, Holdings, such other Loan Party or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Each Lender shall, whenever a lapse in time or change in circumstances renders any documentation (including any specific documentation required below in this Section 3.01(e)) obsolete, expired or inaccurate in any respect, deliver promptly to the Borrower, Holdings, such other Loan Party and the Administrative Agent updated or other appropriate documentation or promptly notify the Borrower, Holdings, such other Loan Party and the Administrative Agent of its legal ineligibility to do so.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 96 of 432

Without limiting the generality of the foregoing, if the Borrower or Holdings, as the case may be, is resident for tax purposes in the United States, any Foreign Lender shall deliver to the Borrower, Holdings and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement whichever of the following is applicable:

(i) duly completed copies of Internal Revenue Service Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,

(ii) duly completed copies of Internal Revenue Service Form W-8ECI,

(iii) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (A) a certificate substantially in the form of <u>Exhibit L</u> (a "<u>Tax</u><u>Status Certificate</u>") to the effect that such Foreign Lender is not (1) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (2) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (3) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code, and that no interest payments will be considered effectively connected income and (B) duly completed copies of Internal Revenue Service Form W-8BEN,

(iv) where such Foreign Lender is a partnership (for U.S. federal income tax purposes) or otherwise not a beneficial owner (e.g., where such Lender has sold a participation), IRS Form W-8IMY (or any successor thereto) and all required supporting documentation (including, where one or more of the underlying beneficial owner(s) is claiming the benefits of the Portfolio Interest Exemption, a Tax Status Certificate of such beneficial owner(s) (provided that, if the Foreign Lender is a partnership and not a participating Lender, the Tax Status Certificate from the beneficial owner(s) may be provided by the Foreign Lender on the beneficial owner(s) behalf), and/or

(v) any other form prescribed by applicable Law as a basis for claiming exemption from or a reduction in United States federal withholding tax duly completed together with such supplementary documentation as may be prescribed by applicable Law to permit the Borrower to determine the withholding or deduction required to be made.

If a payment made to a Foreign Lender or the Administrative Agent hereunder or under any other Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender or the Administrative Agent (as the case may be) were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable, or in any agreement or request entered into or issued pursuant to such Sections), such Lender or the Administrative Agent (as the case may be) shall deliver to the Borrower, Holdings and (in the case of such Lender) to the Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower, Holdings or the Administrative Agent, such documentation, certifications or other information prescribed by applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation, certifications or other information reasonably requested by the Borrower, Holdings or the Administrative Agent as may be necessary for the Borrower, Holdings or the Administrative Agent to comply with its obligations under FATCA, to determine whether such Lender or the Administrative Agent (as the case may be) has complied with its obligations under FATCA or to determine the amount, if any, to deduct and withhold from such payment. Solely for purposes of this paragraph, "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Notwithstanding anything to the contrary in this <u>Section 3.01(e)</u>, no Lender shall be required to deliver any documentation that it is not legally eligible to deliver.

(f) <u>Authorization to Share Documentation</u>. Each Lender hereby authorizes the Administrative Agent to deliver to the Loan Parties and to any successor Administrative Agent any documentation provided by such Lender to the Administrative Agent pursuant to <u>Section 3.01(e)</u> of this Agreement.

(g) <u>FATCA Non-Grandfathered</u>. For purposes of FATCA, from and after the Amendment No. 4 Effective Date, the Borrower and the Administrative Agent shall treat (and the Lenders hereby authorize the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 97 of 432

Administrative Agent to treat) this Agreement and any Loan made hereunder (including any Loan that has been already outstanding) as not qualifying as a "grandfathered obligation" within the meaning of Treasury Regulation Section 1.1471-2(b)(2)(i).

(h) <u>Treatment of Certain Refunds</u>. If the Administrative Agent or any Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by any Loan Party, or with respect to which any Loan Party has paid additional amounts pursuant to this <u>Section</u> <u>3.01</u>, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnify payments made, or additional amounts paid, by the Loan Parties under this <u>Section 3.01</u> with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of the Administrative Agent or such Lender, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), <u>provided</u> that the Borrower, upon the request of the Administrative Agent or such Lender or the Issuing Bank, agrees to repay the amount paid over to such Loan Party (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent or such Lender if the Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This Section shall not be construed to require the Administrative Agent or any Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Borrower, Holdings or any other Person.

(i) For purposes of this <u>Section 3.01</u>, the term "Lender" shall include any Swingline Lender and the Issuing Bank.

3.02 <u>Illegality</u>.

If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund Eurodollar Rate Loans, or to determine or charge interest rates based upon the Eurodollar Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the London interbank market, then, on notice thereof by such Lender to the Borrower through the Administrative Agent, any obligation of such Lender to make or continue Eurodollar Rate Loans or to convert Base Rate Loans to Eurodollar Rate Loans shall be suspended until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrower shall, upon demand from such Lender to Base Rate Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such Eurodollar Rate Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such Eurodollar Rate Loans. Upon any such prepayment or conversion, the Borrower shall also pay accrued interest on the amount so prepaid or converted.

3.03 <u>Inability to Determine Rates</u>.

If the Required Lenders determine that for any reason in connection with any request for a Eurodollar Rate Loan or a conversion to or continuation thereof that (a) Dollar deposits are not being offered to banks in the London interbank eurodollar market for the applicable amount and Interest Period of such Eurodollar Rate Loan, (b) adequate and reasonable means do not exist for determining the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan, or (c) the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan does not adequately and fairly reflect the cost to such Lenders of funding such Loan, the Administrative Agent will promptly so notify the Borrower and each Lender. Thereafter, the obligation of the Lenders to make or maintain Eurodollar Rate Loans shall be suspended until the Administrative Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, the Borrower may revoke any pending request for a Borrowing of, conversion to or continuation of Eurodollar Rate Loans or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans in the amount specified therein.

3.04 Increased Costs; Reserves on Eurodollar Rate Loans.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 98 of 432

(a) <u>Increased Costs Generally</u>. If any Change in Law shall:

(i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement contemplated by <u>Section</u> <u>3.04(e)</u>), the Swingline Lender or the Issuing Bank;

(ii) subject any Lender, the Swingline Lender or the Issuing Bank to any Tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any participation in a Letter of Credit or any Eurodollar Rate Loan made by it, or change the basis of Taxation of payments to such Lender, the Swingline Lender or the Issuing Bank in respect thereof (except for Indemnified Taxes or Other Taxes indemnifiable under <u>Section 3.01</u> and any Excluded Tax); or

(iii) impose on any Lender, the Swingline Lender or the Issuing Bank or the London interbank market any other condition, cost or expense affecting this Agreement or Eurodollar Rate Loans made by such Lender, the Swingline Lender or any Letter of Credit or participation therein;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Eurodollar Rate Loan (or of maintaining its obligation to make any such Revolving Loan), or to increase the cost to such Lender or the Issuing Bank of participating in, issuing or maintaining any Letter of Credit (or of maintaining its obligation to participate in or to issue any Letter of Credit), or to reduce the amount of any sum received or receivable by such Lender or the Issuing Bank hereunder (whether of principal, interest or any other amount) then, upon request of such Lender or the Issuing Bank, the Borrower will pay to such Lender or the Issuing Bank, as the case may be, such additional amount or amounts as will compensate such Lender or the Issuing Bank, as the case may be, for such additional costs incurred or reduction suffered.

(b) <u>Capital Requirements</u>. If any Lender or the Issuing Bank determines that any Change in Law affecting such Lender or the Issuing Bank or any Lending Office of such Lender or such Lender's or the Issuing Bank's holding company, if any, regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's or the Issuing Bank's capital or on the capital of such Lender's or the Issuing Bank's holding company, if any, as a consequence of this Agreement, the Revolving Commitments of such Lender or the Revolving Loans made by, or participations in Letters of Credit held by, such Lender, or the Letters of Credit issued by the Issuing Bank, to a level below that which such Lender or the Issuing Bank's holding company could have achieved but for such Change in Law (taking into consideration such Lender's or the Issuing Bank's policies and the policies of such Lender's or the Issuing Bank, as the case may be, such additional amount or amounts as will compensate such Lender or the Issuing Bank or such Lender's or the Issuing Bank's holding company for any such reduction suffered.

(c) Certificates for Reimbursement. A certificate of a Lender or the Issuing Bank setting forth the amount or amounts necessary to compensate such Lender or the Issuing Bank or its holding company, as the case may be, as specified in <u>subsection (a)</u> or (b) of this Section and delivered to the Borrower shall be conclusive absent manifest error. The Borrower shall pay such Lender or the Issuing Bank, as the case may be, the amount shown as due on any such certificate within 10 days after receipt thereof.

(d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender or the Issuing Bank to demand compensation pursuant to the foregoing provisions of this Section shall not constitute a waiver of such Lender's or the Issuing Bank's right to demand such compensation, <u>provided</u> that the Borrower shall not be required to compensate a Lender or the Issuing Bank pursuant to the foregoing provisions of this Section for any increased costs incurred or reductions suffered more than 120 days prior to the date that such Lender or the Issuing Bank, as the case may be, notifies the Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's or the Issuing Bank's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the 120 day period referred to above shall be extended to include the period of retroactive effect thereof).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 99 of 432

(e) <u>Reserves on Eurodollar Rate Loans</u>. The Borrower shall pay to each Lender, as long as such Lender shall be required to maintain reserves with respect to liabilities or assets consisting of or including Eurocurrency funds or deposits (currently known as "Eurocurrency liabilities," as defined in Regulation D of the FRB), additional interest on the unpaid principal amount of each Eurodollar Rate Loan equal to the actual costs of such reserves allocated to such Eurodollar Rate Loan by such Lender (as determined by such Lender in good faith, which determination shall be conclusive), which shall be due and payable on each date on which interest is payable on such Eurodollar Rate Loan, provided the Borrower shall have received at least 10 days' prior notice (with a copy to the Administrative Agent) of such additional interest from such Lender together with a schedule containing a reasonably detailed explanation and calculation thereof; and provided, further, that such additional interest shall not exceed an amount based upon an interest rate per annum equal to the remainder obtained by subtracting (i) the Eurodollar Rate for the Interest Period for such Eurodollar Rate Loan from (ii) the rate obtained by dividing such Eurodollar Rate by a percentage equal to 100% minus the Eurodollar Reserve Percentage of such Lender for such Interest Period. If a Lender fails to give notice 10 days prior to the relevant Interest Payment Date, such additional interest shall be due and payable 10 days from receipt of such notice.

3.05 <u>Compensation for Losses</u>.

Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrower shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:

(a) any continuation, conversion, payment or prepayment of any Eurodollar Rate Loan on a day other than the last day of the Interest Period for such Eurodollar Rate Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);

(b) any failure by the Borrower (for a reason other than the failure of a Lender to make a Revolving Loan) to prepay, borrow, continue or convert any Eurodollar Rate Loan on the date or in the amount notified by the Borrower; or

(c) any assignment of a Eurodollar Rate Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Borrower pursuant to <u>Section 11.13</u>;

excluding any loss of anticipated profits but including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Eurodollar Rate Loan or from fees payable to terminate the deposits from which such funds were obtained. The Borrower shall also pay any customary administrative fees charged by such Lender in connection with the foregoing.

For purposes of calculating amounts payable by the Borrower to the Lenders under this <u>Section 3.05</u>, each Lender shall be deemed to have funded each Eurodollar Rate Loan made by it at the Eurodollar Rate for such Eurodollar Rate Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Rate Loan was in fact so funded.

3.06 <u>Mitigation Obligations; Replacement of Lenders.</u>

(a) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section</u> <u>3.04</u>, or the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, or if any Lender gives a notice pursuant to <u>Section 3.02</u>, then such Lender shall use reasonable efforts to designate a different Lending Office for funding or booking its Revolving Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.01</u> or <u>3.04</u>, as the case may be, in the future, or eliminate the need for the notice pursuant to <u>Section 3.02</u>, as applicable, and (ii) in each case, would not subject such Lender to any unreimbursed cost or expense and would not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 100 of 432

(b) <u>Replacement of Lenders</u>. If any Lender requests compensation under <u>Section 3.04</u>, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, the Borrower may replace such Lender in accordance with <u>Section 11.13</u>.

3.07 <u>Survival</u>.

All of the Borrower's obligations under this <u>Article III</u> shall survive <u>the resignation or replacement of the</u> <u>Administrative Agent</u>, any assignment of rights by or replacement of a Lender (including a Swingline Lender or the <u>Issuing Bank)</u>, termination of the Aggregate Commitments and <u>the</u> repayment, <u>satisfaction or discharge</u> of all other Obligations hereunder.

ARTICLE IV

CONDITIONS PRECEDENT TO BORROWINGS

4.01 <u>Conditions of Initial Borrowing</u>.

The Administrative Agent, the Swingline Lender, the Issuing Bank and the Lenders shall not be required to fund any Revolving Loans or Swingline Loans, or arrange for the issuance of any Letters of Credit on the Closing Date, until the following conditions are satisfied or waived:

(a) <u>Documents, Certificates, Opinions and Other Instruments</u>. The Administrative Agent's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to the Administrative Agent and each of the Lenders:

(i) (x) with respect to each Loan Party (other than the Canadian Guarantors), to the extent that it is a party thereto, executed counterparts of this Agreement, the Second Amended and Restated Credit Agreement, a Guaranty and the Term Intercreditor Agreement, sufficient in number for distribution to the Administrative Agent, each Lender and the Borrower, and (y) with respect to each Canadian Guarantor, to the extent it is a party thereto, executed counterparts of a Canadian Guaranty, the Term Intercreditor Agreement, sufficient in number for distribution to the Administrative Agent, each Lender and the Borrower;

(ii) a Note executed by the Borrower in favor of each Lender requesting a Note;

(iii) a pledge and security agreement, in substantially the form of Exhibit F-1 (together with each other pledge and security agreement and pledge and security agreement supplement delivered by any Loan Party (other than the Canadian Guarantors) pursuant to Section 6.12, in each case as amended, the "Security Agreement"), duly executed by each Loan Party (other than the Canadian Guarantors), and a pledge and security agreement, in substantially the form of Exhibit F-2 (together with each other pledge and security agreement and pledge and security agreement supplement delivered by a Canadian Guarantor pursuant to Section 6.12, in each case as amended, the "Canadian Security Agreement"), duly executed by each Canadian Guarantor, together with:

(A) certificates (if any) representing the Pledged Equity referred to therein accompanied by undated stock powers executed in blank and instruments evidencing the Pledged Debt (if any) indorsed in blank,

(B) proper financing statements in form appropriate for filing under the Uniform Commercial Code or the PPSA of all jurisdictions that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Security Agreement or the Canadian Security Agreement, covering the Collateral described therein,

(C) certified copies of UCC, PPSA, Bank Act, insolvency, tax, judgment lien and execution searches, or equivalent reports or searches, each of a recent date listing all effective financing statements, lien notices or comparable documents (together with copies of such financing statements and documents) that name any Loan Party as debtor and that are filed in those state, province, territory and county jurisdictions in which any Loan Party is organized or maintains its principal place of business or maintains assets in Canada, and such other searches that are required by the Perfection Certificate or that the Administrative Agent deems necessary or appropriate,

(D) evidence of the completion of all other actions, recordings and filings of or with respect to the Security Agreement and the Canadian Security Agreement that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created thereby,

(E) a Perfection Certificate, duly executed by each Loan Party, and

(F) evidence that all other action that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Security Agreement and the Canadian Security Agreement has been taken (including receipt of duly executed payoff letters, UCC-3 termination statements, PPSA termination and financing change statements and landlords' and bailees' waiver and consent agreements);

(iv) with respect to each Mortgaged Property located in the United States, (x) a completed "Life-of-Loan" Federal Emergency Management Agency Standard Flood Hazard Determination and (y) if any portion of any Mortgaged Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), a notice about special flood hazard area status and flood disaster assistance duly executed by the Borrower and each Loan Party relating thereto and the evidence of flood insurance required pursuant to <u>Section 6.07</u> hereof;

(v) an intellectual property security agreement, in substantially the form of Exhibit <u>H-1</u> (together with each other intellectual property security agreement and intellectual property security agreement supplement delivered by any Loan Party (other than the Canadian Guarantors) pursuant to <u>Section 6.12</u> or for the avoidance of doubt pursuant to the Existing Credit Agreement, in each case as amended, the "<u>Intellectual Property Security Agreement</u>"), duly executed by the applicable Loan Parties (other than the Canadian Guarantors), and, to the extent that any Canadian Guarantor owns any patents, trademarks, copyrights or other registered intellectual property rights, an intellectual property security agreement, in substantially the form of <u>Exhibit H-2</u> (together with each other intellectual property security agreement and intellectual property security agreement supplement delivered by a Canadian Guarantor pursuant to <u>Section 6.12</u>, the "<u>Canadian Intellectual Property Security Agreement</u>"), duly executed by each Canadian Guarantor, together with evidence that all action that the Administrative Agent may reasonably deem necessary in order to perfect the Liens created under the Intellectual Property Security Agreement and the Canadian Intellectual Property Security Agreement has been taken;

(vi) such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as the Administrative Agent may require evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party or is to be a party;

(vii) such documents and certifications as the Administrative Agent may reasonably require to evidence that each Loan Party is duly organized or formed and is validly existing, in good standing and qualified to engage in business in each jurisdiction where its ownership, lease or operation of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 102 of 432

properties or the conduct of its business requires such qualification, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect;

(viii) a favorable opinion of Ian R. Scheinmann, Esq., Senior Vice President, Legal Affairs of Holdings, addressed to the Administrative Agent and each Lender, as to the matters set forth in <u>Exhibit I-1</u> and such other matters concerning the Loan Parties and the Loan Documents as the Administrative Agent may reasonably request;

(ix) a favorable opinion of Hughes Hubbard & Reed LLP, special counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to the matters set forth in Exhibit <u>I-2</u> and such other matters concerning the Loan Parties and the Loan Documents as the Administrative Agent may reasonably request;

(x) a favorable opinion of Davis Graham & Stubbs LLP, special Colorado counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to the matters set forth in <u>Exhibit I-3</u> and such other matters concerning the Loan Parties and the Loan Documents as the Administrative Agent may reasonably request;

(xi) a favorable opinion of DLA Piper LLP, special Georgia, Maryland, Massachusetts, North Carolina and Virginia counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to the matters set forth in <u>Exhibit I-4</u> and such other matters concerning the Loan Parties and the Loan Documents as the Administrative Agent may reasonably request;

(xii) a favorable opinion of Bose McKinney & Evans LLP, special Indiana counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to the matters set forth in <u>Exhibit I-5</u> and such other matters concerning the Loan Parties and the Loan Documents as the Administrative Agent may reasonably request;

(xiii) a favorable opinion of Fasken Martineau DuMoulin LLP, special Ontario counsel to the Canadian Guarantors, addressed to the Administrative Agent and each Lender, as to the matters set forth in Exhibit I-6 and such other matters concerning the Canadian Guarantors and the Loan Documents as the Administrative Agent may reasonably request;

(xiv) a favorable opinion of Stewart McKelvey, special Nova Scotia counsel to the Canadian Guarantors, addressed to the Administrative Agent and each Lender, as to the matters set forth in <u>Exhibit I-7</u> and such other matters concerning the Canadian Guarantors and the Loan Documents as the Administrative Agent may reasonably request;

(xv) a certificate of a Responsible Officer of each Loan Party either (A) attaching copies of all consents, licenses and approvals required in connection with the execution, delivery and performance by such Loan Party and the validity against such Loan Party of the Loan Documents to which it is a party, and such consents, licenses and approvals shall be in full force and effect, or (B) stating that no such consents, licenses or approvals are so required;

(xvi) a certificate signed by a Responsible Officer of the Borrower certifying that the conditions specified in <u>Sections 4.02(a)</u> and (b) have been satisfied;

(xvii) certificates attesting to the Solvency of each Loan Party, individually and together with its Subsidiaries on a consolidated basis, before and after giving effect to the Transaction, from its chief financial officer;

(xviii) evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect, together with the certificates of insurance, naming the Administrative Agent, on behalf of the Lenders, as an additional insured or loss payee, as the case may be,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 103 of 432

under all insurance policies maintained with respect to the assets and properties of the Loan Parties that constitutes Collateral;

(xix) such other assurances, certificates, documents, consents or opinions as the Administrative Agent, the Issuing Bank, the Swingline Lender or any Lender reasonably may require; and

(xx) a Borrowing Base Certificate prepared as of the last day of the most recent fiscal month ended at least 15 consecutive Business Days prior to the Closing Date (or such shorter period as may be elected by the Borrower).

(b) <u>Fees</u>. (i) All fees required to be paid to the Administrative Agent and the Lead Arrangers on or before the Closing Date shall have been paid and (ii) all fees required to be paid to the Lenders on or before the Closing Date shall have been paid.

(c) <u>Counsel's Fees</u>. Unless waived by the Administrative Agent, the Borrower shall have paid all reasonable fees, charges and disbursements of counsel to the Administrative Agent (directly to such counsel if requested by the Administrative Agent) to the extent invoiced prior to the Closing Date, plus such additional amounts of such fees, charges and disbursements as shall constitute its reasonable estimate of such reasonable fees, charges and disbursements or to be incurred by it through the closing proceedings (provided that such estimate shall not thereafter preclude a final settling of accounts between the Borrower and the Administrative Agent).

(d) Availability on the Closing Date, after giving effect to the Transaction, shall not be less than \$35.0 million.

(e) Substantially concurrently with the initial Borrowing on the Closing Date, the Second Amended and Restated Credit Agreement shall become effective and (ii) an amendment to the Unsecured Term Loan Facility-shall have become effective.

Without limiting the generality of the provisions of <u>Section 9.04</u>, for purposes of determining compliance with the conditions specified in this <u>Section 4.01</u>, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

4.02 <u>Conditions to All Credit Extensions</u>.

The obligation of each Lender and each Issuing Bank to make any Credit Extension (including the initial Credit Extension) shall be subject to the satisfaction (or waiver) of each of the conditions precedent set forth below:

(a) <u>Representations and Warranties</u>. The representations and warranties of the Borrower and each other Loan Party contained in <u>Article V</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct in all material respects (except to the extent already qualified by materiality, in which case such representation and warranty shall be true and correct in all material respects) on and as of the date of such Borrowing, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects (except to the extent already qualified by materiality, in which case such representation and warranty shall be true and correct in all respects) as of such earlier date, and except that for purposes of this <u>Section 4.02</u>, the representations and warranties contained in <u>Sections 5.05(a)</u> and (b) shall be deemed to refer to the most recent statements furnished pursuant to <u>Sections 6.01(a)</u> and (b), respectively.

(b) <u>No Default</u>. No Default shall exist, or would result from such proposed Borrowing or from the application of proceeds thereof.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 104 of 432

(c) <u>Notice of Borrowing</u>. The Administrative Agent shall have received a Notice of Borrowing as required by <u>Section 2.03</u> (or such notice shall have been deemed given in accordance with <u>Section 2.03</u>) if Revolving Loans are being requested or, in the case of the issuance, amendment, extension or renewal of a Letter of Credit, the Issuing Bank and the Administrative Agent shall have received a notice requesting the issuance, amendment, extension or renewal of such Letter of Credit as required by <u>Section 2.13(b)</u> or, in the case of the Borrowing of a Swingline Loan, the Swingline Lender and the Administrative Agent shall have received a notice requesting such Swingline Loan as required by <u>Section 2.12(b)</u>.

(d) <u>Availability</u>. Availability on the proposed date of such Credit Extension shall be adequate to cover the amount of such Credit Extension.

Each Notice of Borrowing (other than a Notice of Borrowing requesting only a conversion of Revolving Loans of one Type to another Type or a continuation of Eurodollar Rate Loans) submitted by the Borrower shall be deemed to be a representation and warranty that the conditions specified in <u>Sections 4.02(a)</u> and (b) have been satisfied on and as of the date of the applicable Borrowing.

ARTICLE V

REPRESENTATIONS AND WARRANTIES

Each of Holdings and the Borrower represents and warrants to the Administrative Agent and the Lenders that:

5.01 Existence, Qualification and Power.

EachSubject to any entry of any required orders of the Bankruptcy Court including, without limitation, the entry of the Interim Financing Order and the Final Financing Order, as applicable, each Loan Party and each of its Subsidiaries (a) is duly organized or formed, validly existing and, as applicable, in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) has all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own or lease its assets and carry on its business and (ii) execute, deliver and perform its obligations under the Loan Documents or Material Contracts to which it is a party and consummate the Transaction, and (c) is duly qualified and is licensed and, as applicable, in good standing under the Laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification or license; except in each case referred to in <u>clause (b)(i)</u> or <u>(c)</u>, to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.

5.02 Authorization; No Contravention.

TheSubject to the entry and terms of the Financing Orders, the execution, delivery and performance by each Loan Party of each Loan Document to which such Person is or is to be a party have been duly authorized by all necessary corporate or other organizational action, and do not and will not (a) contravene the terms of any of such Person's Organization Documents; (b) conflict with or result in any breach or contravention of, or result in the creation of any Lien under (i) any Contractual Obligation to which such Person is a party or binding upon such Person or the properties of such Person or any of its Subsidiaries or (ii) any order, injunction, writ or decree of any Governmental Authority or any arbitral award to which such Person or its property is subject, except in each case where enforcement is stayed upon commencement of the Chapter 11 Cases; or (c) violate any Law, except in each case referred to in the foregoing clauses (b) and (c), to the extent that such conflict, breach, contravention or violation could not reasonably be expected to have a Material Adverse Effect.

5.03 <u>Governmental Authorization; Other Consents.</u>

No approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any Governmental Authority or any other Person is necessary or required in connection with (a) the execution, delivery or performance by, or enforcement against, any Loan Party of this Agreement or any other Loan Document or Material Contract, or for the consummation of the Transaction, (b) the grant by any Loan Party of the Liens granted

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 105 of 432

by it pursuant to the Collateral Documents, (c) the perfection or maintenance of the Liens created under the Collateral Documents (including the first priority nature thereof (subject to the <u>TermDIP Term Intercreditor</u> <u>Agreement and the First Lien</u> Intercreditor Agreement)) or (d) the exercise by the Administrative Agent or any Lender of its rights under the Loan Documents or the remedies in respect of the Collateral pursuant to the Collateral Documents, except for (i) filings and recordings necessary to perfect Liens created under the Collateral Documents, (ii) with respect to <u>clause (d)</u> above, any approvals, authorizations or filings that may be required under the Securities Laws and (iii) such authorizations, approvals, actions, notices and filings that have been duly obtained, taken, given or made and are in full force and effect<u>including the applicable Financing Order</u>. All applicable waiting periods in connection with the Transaction have expired without any action having been taken by any Governmental Authority restraining, preventing or imposing materially adverse conditions upon the Transaction or the rights of the Loan Parties or their Subsidiaries freely to transfer or otherwise dispose of, or to create any Lien on, any properties now owned or hereafter acquired by any of them.

5.04 <u>Binding Effect</u>.

ThisSubject to the entry and terms of the Financing Orders, this Agreement has been, and each other Loan Document, when delivered hereunder, will have been, duly executed and delivered by each Loan Party that is party thereto. ThisSubject to the entry and terms of the Financing Orders, this Agreement constitutes, and each other Loan Document when so delivered will constitute, a legal, valid and binding obligation of each Loan Party that is party thereto, enforceable against such Loan Party that is party thereto in accordance with its terms, except as enforceability may be limited by applicable Debtor Relief Laws and by equitable principles regardless of whether considered in a proceeding in equity or at law.

5.05 <u>Financial Statements; No Material Adverse Effect; No Internal Control Event.</u>

(a) The Audited Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present the financial condition of Holdings and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; and (iii) show all material indebtedness and other liabilities, direct or contingent, of Holdings and its Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness.

(b) The unaudited consolidated balance sheet of Holdings and its Subsidiaries as at September 29, 2012, and the related consolidated statements of income or operations, and cash flows for the fiscal quarter ended on that date (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein, and (ii) fairly present the financial condition of Holdings and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby, subject, in the case of <u>clauses (i)</u> and <u>(ii)</u>, to the absence of footnotes and to normal quarter-end or year-end adjustments.

(c) Since the date of the Audited Financial Statements, except as disclosed in Holdings' public filings with the SEC made prior to the Closing Date, there has been no event or circumstance, either individually or in the aggregate, that has had or could reasonably be expected to have a Material Adverse Effect.

(d) To the best knowledge of Holdings and the Borrower, no Internal Control Event exists or has occurred since the date of the Audited Financial Statements that has resulted in or could reasonably be expected to result in a misstatement in any material respect, in any financial information delivered or to be delivered to the Administrative Agent or the Lenders, of (i) covenant compliance calculations provided hereunder or (ii) the assets, liabilities, financial condition or results of operations of Holdings and its Subsidiaries on a consolidated basis.

(e) The consolidated forecasted balance sheets, statements of income and cash flows of Holdings and its Subsidiaries delivered by or on behalf of the Borrower to the Administrative Agent or the Lenders during the period prior to the Closing Date in connection with the syndication of the credit facilities evidenced by this Agreement were prepared in good faith on the basis of the assumptions stated therein, which assumptions were reasonable in light of the conditions existing at the time of delivery of such forecasts.

5.06 <u>Litigation</u>.

ThereOther than the commencement of the Chapter 11 Cases, or as stayed upon commencement of the Chapter 11 Cases, there are no actions, suits, proceedings, investigations, claims or disputes pending or, to the knowledge of Holdings or the Borrower after due and diligent investigation, threatened or contemplated, at law, in equity, in arbitration or before any Governmental Authority, by or against Holdings, the Borrower or any of Holding's other Subsidiaries or against any of their properties or revenues that (a) purport to affect or pertain to this Agreement, any other Loan Document, any Related Document or the consummation of the Transaction, or (b) either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.07 <u>No Default</u>.

Neither Except where enforcement is stayed upon the commencement of the Cases, neither any Loan Party nor any Subsidiary thereof is in default under or with respect to any Contractual Obligation in any manner that could, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. No Default or <u>Event of Default</u> has occurred and is continuing or would result from the consummation of the transactions contemplated by this Agreement or any other Loan Document.

5.08 <u>Ownership of Property; Liens; Investments</u>.

(a) <u>EachSubject to the rights of counterparties under the applicable provisions of the Bankruptcy</u> <u>Code, including, without limitation, Section 365 thereof, each</u> Loan Party and each of its Subsidiaries has good record and marketable title in fee simple to, or valid leasehold interests in, all real property necessary or used in the ordinary conduct of its business, except for such defects in title as could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(b) <u>Schedule 5.08(b)</u> sets forth a complete and accurate list as of the Closing Date of all Liens (except Liens securing the Obligations) on the property or assets of each Loan Party and each of its Subsidiaries, showing as of the Closing Date the property or assets of such Loan Party or such Subsidiary subject thereto. The property of each Loan Party and each of its Subsidiaries is subject to no Liens, other than Liens set forth on <u>Schedule 5.08(b)</u>, and as otherwise permitted by <u>Section 7.01</u>.

(c) <u>Schedule 5.08(c)</u> sets forth a complete and accurate list as of the Closing Date of all real property owned by each Loan Party and each of its Subsidiaries, showing as of the Closing Date the street address, county or other relevant jurisdiction, state, record owner and fair market value thereof (as reasonably determined by the Borrower). Each Loan Party and each of its Subsidiaries has good, marketable and insurable fee simple title to the real property owned by such Loan Party or such Subsidiary, free and clear of all Liens, other than Liens created or permitted by the Loan Documents.

(d) <u>Schedules 5.08(d)(i)</u> and (ii) set forth as of the Closing Date a complete and accurate list of all leases of real property under which any Loan Party or any Subsidiary of a Loan Party is the lessee or the lessor, respectively, showing as of the Closing Date the street address, county or other relevant jurisdiction, state, lessor, lessee and annual rental cost thereof. Each such lease is the legal, valid and binding obligation of such Loan Party or Subsidiary party thereto, enforceable against such Loan Party or Subsidiary in accordance with its terms, except as enforceability may be limited by applicable Debtor Relief Laws and by equitable principles regardless of whether considered in a proceeding in equity or at law.

(e) <u>Schedule 5.08(e)</u> sets forth a complete and accurate list of all Investments held by any Loan Party or any Subsidiary of a Loan Party on the Closing Date (other than Investments in Subsidiaries), showing as of the Closing Date the amount, obligor or issuer and maturity, if any, thereof.

5.09 <u>Environmental Compliance</u>.

Except as specifically disclosed in <u>Schedule 5.09</u> or for any matters, individually or in the aggregate, that could not reasonably be expected to have a Material Adverse Effect:

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 107 of 432

(a) the Loan Parties and their respective Subsidiaries conduct in the ordinary course of business a review of the effect of existing Environmental Laws and claims alleging potential liability or responsibility for violation of any Environmental Law on their respective businesses, operations and properties, and as a result thereof the Borrower has reasonably concluded that such Environmental Laws and claims could not reasonably be expected to result in an Environmental Liability;

(b) (i) none of the properties currently or, to the best knowledge of Holdings and the Borrower, formerly owned or operated by any Loan Party or any of its Subsidiaries is listed or formally proposed for listing on the NPL or any analogous foreign, state, provincial, territorial or local list; (ii) there are no and never have been any underground or above-ground storage tanks or any surface impoundments, septic tanks, pits, sumps or lagoons in which Hazardous Materials are being or have been treated, stored or disposed on any property currently owned or operated by any Loan Party or, to the best of the knowledge of the Loan Parties, on any property formerly owned or operated by any Loan Party or any of its Subsidiaries, in each case except in compliance with all applicable Environmental Laws; (iii) there is no asbestos or asbestos-containing material on any property currently owned or operated by any Loan Party or any of its Subsidiaries, in each case except in compliance with all applicable Environmental Laws, and (iv) there has been no Release of Hazardous Materials on, at, under or from any property currently or formerly owned or operated by any Loan Party owned or operated by any Loan Party or any of its and the subsidiaries of the Subsidiaries that could reasonably be expected to result in an Environmental Liability; and

(c) neither any Loan Party nor any of its Subsidiaries is undertaking, and has not completed, either individually or together with other potentially responsible parties, any investigation or assessment or remedial or response action relating to any actual or threatened Release of Hazardous Materials at any site, location or operation, either voluntarily or pursuant to the order of any Governmental Authority or the requirements of any Environmental Law; and all Hazardous Materials generated, used, treated, handled or stored at, or transported to or from, any property currently or, to the knowledge of Holdings and the Borrower, formerly owned or operated by any Loan Party or any of its Subsidiaries have been disposed of in a manner not reasonably expected to result in an Environmental Liability.

5.10 <u>Insurance</u>.

(a) The properties of the Borrower, Holdings and its other Subsidiaries are insured with financially sound and reputable insurance companies having an A.M. Best Financial Strength Rating of at least A which are not Affiliates of the Borrower, in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where the Borrower or the applicable Subsidiary operates.

(b) With respect to any portion of any Mortgaged Property which is located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), the Borrower or the applicable Loan Party (x) maintains, or causes to be maintained, with a financially sound and reputable insurer, flood insurance in an amount and otherwise sufficient to comply with all applicable rules and regulations promulgated pursuant to the Flood Insurance Laws and (y) has delivered to the Administrative Agent evidence of such compliance in form and substance reasonably acceptable to the Administrative Agent.

5.11 <u>Taxes</u>.

The Borrower, Holdings and each of their respective Subsidiaries have filed all federal, state, provincial, territorial and other material tax returns and reports required to be filed, and have paid all federal, state, provincial, territorial and other material Taxes, assessments, fees and other governmental charges levied or imposed upon them or their properties, income or assets otherwise due and payable, except those which are being contested in good faith by appropriate proceedings diligently conducted and for which adequate reserves have been established in accordance with GAAP. To the best knowledge of Holdings and the Borrower, there is no proposed Tax assessment against the Borrower, Holdings or any other Subsidiary that would, if made, reasonably be expected, individually or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 108 of 432

in the aggregate, to have a Material Adverse Effect. Neither any Loan Party nor any Subsidiary thereof is party to any tax sharing agreement.

5.12 ERISA Compliance.

(a) <u>Compliance Generally</u>. Each Plan is in compliance in all material respects with the applicable provisions of ERISA, the Code and other Federal or state Laws. Each Plan that is intended to qualify under Section 401(a) of the Code has received a favorable determination letter from the IRS (or is a prototype plan that is the subject of a favorable opinion letter from the IRS) or an application for such a letter is currently being processed by the IRS with respect thereto and, to the best knowledge of the Borrower, nothing has occurred which would prevent, or cause the loss of, such qualification. <u>The Borrower, Holdings and each ERISA Affiliate have made all required</u> contributions to each Pension Plan, and no application for a funding waiver or an extension of any amortization period pursuant to Section 412 of the Code has been made with respect to any Pension Plan. Each Foreign Plan is in compliance in all material respects with the applicable requirements of Law. No Loan Party nor any of its Subsidiaries has incurred any obligation in connection with the termination of or withdrawal from any Foreign Plan.

(b) <u>Pending Claims</u>. There are no pending or, to the best knowledge of the Borrower, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Plan that could reasonably be expected to have a Material Adverse Effect. There has been no prohibited transaction or violation of the fiduciary responsibility rules with respect to any Plan that has resulted or could reasonably be expected to result in a Material Adverse Effect.

(c) <u>No ERISA or Unfunded Pension Liability.</u> (i) <u>Event.</u> No ERISA Event has occurred or is reasonably expected to occur that could reasonably be expected to have a Material Adverse Effect; (ii) as of the Closing Date, no Pension Plan has any Unfunded Pension Liability that could reasonably be expected to result in a liability in an amount in excess of \$75,000,000; and (iii) neither the Borrower nor any ERISA Affiliate has incurred, or reasonably expects to incur, any liability (A) under Title IV of ERISA with respect to any Pension Plan (other than premiums due and not delinquent under Section 4007 of ERISA), (B) under Section 4201 or 4243 of ERISA with respect to a Multiemployer Plan (and no event has occurred which, with the giving of notice under Section 4219 of ERISA, would result in such liability), or (C) as a result of a transaction that could be subject to Section 4069 or 4212(c) of ERISA, which liability described in the foregoing clauses (A) through (C), individually or in the aggregate, could reasonably be expected to exceed \$50,000,000.

5.13 <u>Canadian Benefit Plans; Canadian Pension Plans</u>.

(a) Except as, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, each Canadian Pension Plan that is not a Canadian Multi-Employer Pension Plan is, and has been, established, registered, funded, administered and invested in compliance in all material respects with the terms of such Canadian Pension Plan, all applicable Laws and any collective agreements, as applicable.

(b) Except as, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, there are no pending or, to the knowledge of Holdings or the Borrower, threatened claims (other than routine claims for benefits), actions or lawsuits, or other action by any Governmental Authority, with respect to any Canadian Benefit Plan.

(c) Except for any Canadian Pension Plan that is a Canadian Multi-Employer Pension Plan or any successor thereto, no Canadian Guarantor has at any time prior to the Closing Date established or participated or currently participates in a defined benefit pension plan.

(d) Except as, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, no Pension Event in respect of a Canadian Guarantor has occurred and is continuing.

(e) No Lien has arisen in respect of any Canadian Guarantor in connection with any Canadian Pension Plan (save for contribution amounts not yet due) which could reasonably be expected to secure assets of such Canadian Guarantor exceeding \$1,000,000 in value.

5.14 Subsidiaries; Equity Interests; Loan Parties.

As of the Closing Date, no Loan Party has any Subsidiaries other than those specifically disclosed in Part (a) of <u>Schedule 5.13</u>, and all of the outstanding Equity Interests in such Subsidiaries have been validly issued, are fully paid and non-assessable and are owned directly or indirectly by a Loan Party in the amounts specified in Part (a) of <u>Schedule 5.13</u> free and clear of all Liens except those created under the Collateral Documents. As of the Closing Date, no Loan Party has any equity investments in any other corporation or entity other than such Subsidiaries and those specifically disclosed in Part (b) of <u>Schedule 5.13</u>. All of the outstanding Equity Interests in the Borrower have been validly issued, are fully paid and non-assessable and are owned by Holdings free and clear of all Liens except those created under the Colsing Date, the jurisdiction of its incorporation, the address of its principal place of business and its U.S. taxpayer identification number or, in the case of any non-U.S. Loan Party that does not have a U.S. taxpayer identification number, its unique identification number issued to it by the jurisdiction of its incorporation. The copy of the charter of each Loan Party and each amendment thereto provided pursuant to <u>Section 4.01(a)</u> is a true and correct copy of each such document, each of which is valid and in full force and effect as of the Closing Date.

5.15 <u>Margin Regulations; Investment Company Act</u>.

(a) <u>Margin Stock</u>. The Borrower is not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (within the meaning of Regulation U issued by the FRB), or extending credit for the purpose of purchasing or carrying margin stock.

(b) <u>Investment Company Act</u>. None of the Borrower, any Person Controlling the Borrower, or any Subsidiary is or is required to be, or after giving effect to the Transaction will be required to be, registered as an "investment company" under the Investment Company Act of 1940.

5.16 <u>Disclosure</u>.

The Borrower has disclosed to the Administrative Agent and the Lenders all agreements, instruments and corporate or other restrictions to which it or any of its Subsidiaries or any other Loan Party is subject, and all other matters known to it, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect. No report, financial statement, certificate or other information furnished (whether in writing or orally) by or on behalf of any Loan Party to the Administrative Agent or any Lender in connection with the transactions contemplated hereby and the negotiation of this Agreement or delivered hereunder or under any other Loan Document, including any confidential information memorandum (in each case as modified or supplemented by other information so furnished), taken as a whole, contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not materially misleading; <u>provided</u> that, with respect to projected financial information, each of Holdings and the Borrower represents only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.

5.17 <u>Compliance with Laws</u>.

EachSubject to the entry of any required orders of the Bankruptcy Court, each Loan Party and each Subsidiary thereof is in compliance in all material respects with the requirements of all Laws and all orders, writs, injunctions and decrees applicable to it or to its properties, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted or (b) the failure to comply therewith, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

5.18 Intellectual Property; Licenses, Etc.

Each Loan Party and each of its Subsidiaries own, or possess the right to use, all of the trademarks, service marks, trade names, copyrights, patents, patent rights, franchises, licenses and other intellectual property rights

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 110 of 432

(collectively, "<u>IP Rights</u>") that are reasonably necessary for the operation of their respective businesses, without conflict with the rights of any other Person, and <u>Schedule 5.18</u> sets forth a complete and accurate list of all such IP Rights owned or licensed by each Loan Party and each of its Subsidiaries as of the Closing Date. To the best knowledge of the Borrower and Holdings, no slogan or other advertising device, product, process, method, substance, part or other material now employed, or now contemplated to be employed, by any Loan Party or any of its Subsidiaries infringes upon any rights held by any other Person. No claim or litigation regarding any of the foregoing is pending or, to the best knowledge of the Borrower and Holdings, threatened, which, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.19 <u>Solvency.[Reserved].</u>

Each Loan Party is, individually and together with its Subsidiaries on a consolidated basis, Solvent.

5.20 <u>Casualty, Etc</u>.

Neither the businesses nor the properties of any Loan Party or any of its Subsidiaries are affected by any fire, explosion, accident, strike, lockout or other labor dispute, drought, storm, hail, earthquake, embargo, act of God or of the public enemy or other casualty (whether or not covered by insurance) that, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.21 Labor Matters.

Except as described on <u>Schedule 5.21</u> hereto, there are no collective bargaining agreements or Multiemployer Plans covering the employees of the Borrower or any of its Subsidiaries as of the Closing Date and to the best knowledge of any Responsible Officer of the Borrower and Holdings, neither the Borrower nor any Subsidiary has suffered any strikes, walkouts, work stoppages or other material labor difficulty within the last five years.

5.22 <u>Collateral Documents</u>.

TheSubject in all respects to the Carve-Out, the provisions of the Collateral Documents are effective to create in favor of the Administrative Agent for the benefit of the Secured Parties a legal, valid and enforceable first priority Lien (subject to Liens permitted by Section 7.01 and the Term7.01, the Financing Orders (with the priorities set forth in the applicable order), the DIP Term Intercreditor Agreement and the First Lien Intercreditor Agreement) on all right, title and interest of the respective Loan Parties in the Collateral described therein. Except for filings and recordings completed prior to the Closing Date and as contemplated hereby and by the Collateral Documents, no filing or other action will be necessary to perfect or protect such Liens.

5.23 <u>Anti-Terrorism Laws</u>.

(a) No Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or agents of such Loan Party or such Subsidiary (i) has violated or is in violation of Anti-Terrorism Laws or (ii) has engaged or engages in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of offenses designated in the "Forty Recommendations" and "Nine Special Recommendations" published by the Organisation for Economic Co-operation and Development's Financial Action Task Force on Money Laundering.

(b) No Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or agents of such Loan Party or such Subsidiary that is acting or benefiting in any capacity in connection with the Revolving Loans is an Embargoed Person.

(c) Except as otherwise authorized by OFAC, no Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 111 of 432

agents of such Loan Party or such Subsidiary acting or benefiting in any capacity in connection with the Revolving Loans (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Embargoed Person, (ii) deals in, or otherwise engages in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law or (iii) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.

ARTICLE VI

AFFIRMATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder or any Revolving Loan or other Obligation hereunder shall remain unpaid or unsatisfied, or any Letter of Credit shall remain outstanding (unless Cash Collateralized or backstopped on terms reasonably satisfactory to the Administrative Agent), each of Holdings and the Borrower shall, and shall (except in the case of the covenants set forth in <u>Sections 6.01, 6.02, 6.03</u> and <u>6.11</u>) cause each Subsidiary to:

6.01 <u>Financial Statements</u>.

Deliver to the Administrative Agent and each Lender, in form and detail reasonably satisfactory to the Administrative Agent and the Required Lenders (<u>provided</u> that the Required Lenders shall be deemed to find the following items satisfactory unless the Administrative Agent shall have received notice from the Required Lenders specifying their objections thereto within five Business Days of the Lenders' receipt of such items):

(a) <u>Annual Financials</u>. As soon as available, but in any event within 110 days (or such shorter period as required by the SEC) after the end of each fiscal year of Holdings, on or prior to April 30, 2018, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal year, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year, setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail and prepared in accordance with GAAP, audited and accompanied by a report and opinion of a Registered Public Accounting Firm of nationally recognized standing reasonably acceptable to the Required Lenders, which report and opinion shall be prepared in accordance with generally accepted auditing standards and applicable Securities Laws and shall not be subject to any "going concern" or like qualification or exception or any qualification or exception as to the scope of such audit or with respect to the absence of any material misstatement; provided that the report and opinion of a Registered Public Accounting Firm delivered in connection with the financial statements of Holdings and its Subsidiaries as at the end of the 2016 fiscal year may be subject to a "going concern" or like qualification or exception as it relates to the maturity of the Senior Notes; and

(b) Quarterly Financials. As soon as available, but in any event within 45 days after the end of each of the first three fiscal quarters of each fiscal year of Holdings, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal quarter, and the related consolidated statements of income or operations and cash flows for such fiscal quarter and for the portion of Holdings' fiscal year then ended, setting forth in each case in comparative form the figures for the corresponding fiscal quarter of the previous fiscal year and the corresponding portion of the previous fiscal year, all in reasonable detail, such consolidated statements to be certified by the chief executive officer, chief financial officer, treasurer or controller of Holdings as fairly presenting the financial condition, results of operations and cash flows of Holdings and its Subsidiaries in accordance with GAAP, subject only to normal quarter-end or year-end adjustments and the absence of footnotes.

(c) <u>Monthly Financials</u>. As soon as available, but in any event within 30 days after the end of each fiscal month, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal month, and the related consolidated statements of income or operations and cash flows for such fiscal month and for the portion of Holdings' fiscal year then ended, setting forth in each case in comparative form the figures for the corresponding month of the previous fiscal year and the corresponding portion of the previous fiscal year, all in reasonable detail.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 112 of 432

(d) Budget. Each Approved Budget and Variance Report required in accordance with Section 5.3 of the Ratification Agreement.

As to any information contained in materials furnished pursuant to <u>Section 6.02(d)</u>, neither Holdings nor the Borrower shall be separately required to furnish such information under <u>Section 6.01(a)</u> or (b) above, but the foregoing shall not be in derogation of the obligation of Holdings and the Borrower to furnish the information and materials described in <u>Sections 6.01(a)</u> and (b) above at the times specified therein.

6.02 <u>Certificates; Other Information</u>.

Deliver to the Administrative Agent and each Lender, in form and detail reasonably satisfactory to the Administrative Agent and the Required Lenders (<u>provided</u> that the Required Lenders shall be deemed to find the following items satisfactory unless the Administrative Agent shall have received notice from the Required Lenders specifying their objections thereto within five Business Days of the Lenders' receipt of such items):

(a) [<u>Intentionally Omitted</u>];

(b) <u>Compliance Certificate</u>. Concurrently with the delivery of the financial statements referred to in <u>Sections 6.01(a)</u> and (b), a duly completed Compliance Certificate signed by the chief executive officer, chief financial officer, treasurer or controller of the Borrower, which certificate shall set forth reasonably detailed calculations with respect to the Consolidated Fixed Charge Coverage Ratio for such period, whether or not the Financial Covenant is then in effect, and in the event of any change in generally accepted accounting principles used in the preparation of such financial statements, the Borrower shall also provide, if necessary for the determination of compliance with <u>Section 7.11</u>, a statement of reconciliation conforming such financial statements to GAAP;

(c) <u>Annual Budget</u>. Concurrently with the delivery of the financial statements referred to in <u>Section</u> 6.01(a), an annual budget of Holdings and its Subsidiaries on a consolidated basis, including forecasts prepared by management of the Borrower, in form reasonably satisfactory to the Administrative Agent, of consolidated balance sheets and statements of income or operations and cash flows of Holdings and its Subsidiaries on a quarterly basis for the then-current fiscal year;

(d) <u>SEC Filings</u>. Promptly after the same are available, copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of the Borrower, and copies of all annual, regular, periodic and special reports and registration statements which the Borrower may file or be required to file with the SEC under Section 13 or 15(d) of the Securities Exchange Act of 1934, or with any national securities exchange, and in any case not otherwise required to be delivered to the Administrative Agent pursuant hereto;

(e) <u>Other Reports</u>. Promptly after the furnishing thereof, copies of any statement or report furnished to any holder of debt securities of any Loan Party or of any of its Subsidiaries pursuant to the terms of any indenture, loan or credit or similar agreement and not otherwise required to be furnished to the Lenders pursuant to <u>Section</u> <u>6.01</u> or any other clause of this <u>Section 6.02</u>;

(f) <u>Insurance Coverage</u>. As soon as available, but in any event within 30 days after the end of each fiscal year of the Borrower, a report summarizing the insurance coverage (specifying type, amount and carrier) in effect for each Loan Party and its Subsidiaries and containing such additional information as the Administrative Agent, or any Lender through the Administrative Agent, may reasonably specify;

(g) <u>Investigations</u>. Promptly, and in any event within ten Business Days after receipt thereof by any Loan Party or any Subsidiary thereof, copies of each notice or other correspondence received from the SEC (or comparable agency in any applicable non-U.S. jurisdiction) concerning any investigation or possible investigation or other inquiry by such agency regarding financial or other operational results of any Loan Party or any Subsidiary thereof;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 113 of 432

(h) <u>Related Document Notices</u>. Not later than five Business Days after receipt thereof by any Loan Party or any Subsidiary thereof, copies of all notices, requests and other documents (including amendments, waivers and other modifications) so received under or pursuant to any Related Document regarding or related to any breach or default by any party thereto or any other event that could materially impair the value of the interests or the rights of any Loan Party or otherwise have a Material Adverse Effect and, from time to time upon request by the Administrative Agent, such information and reports regarding the Related Documents as the Administrative Agent may reasonably request;

(i) <u>Environmental Notices</u>. Promptly after the assertion or occurrence thereof, notice of any action or proceeding alleging an Environmental Liability against any Loan Party or any of its Subsidiaries that could (i) reasonably be expected to have a Material Adverse Effect or (ii) cause any property described in the Mortgages or Canadian Mortgages to be subject to any restrictions on ownership, occupancy, use or transferability under any Environmental Law;

(j) <u>Schedule Supplements</u>. As soon as available, but in any event within 90 days after the end of each fiscal year of the Borrower, (i) a report supplementing Schedules <u>5.08(c)</u>, <u>5.08(d)(i)</u> and <u>5.08(d)(ii)</u>, including an identification of all owned and leased real property disposed of by any Loan Party or any Subsidiary thereof during such fiscal year, a list and description (including the street address, county or other relevant jurisdiction, state, record owner, book value thereof and, in the case of leases of property, lessor, lessee, expiration date and annual rental cost thereof) of all real property acquired or leased during such fiscal year and a description of such other changes in the information included in such schedules as may be necessary for such schedules to be accurate and complete and (ii) a report supplementing <u>Schedules 5.08(e)</u> and <u>5.13</u> and each schedule to the Security Agreement and the Canadian Security Agreement containing a description of all changes in the information included in such schedules to be accurate and complete, each such report to be signed by a Responsible Officer of the Borrower and to be in a form reasonably satisfactory to the Administrative Agent; and

(k) <u>Additional Information</u>. Promptly, such additional information regarding the business, financial, legal or corporate affairs of any Loan Party or any Subsidiary thereof, or compliance with the terms of the Loan Documents, as the Administrative Agent or any Lender may from time to time reasonably request.

Documents required to be delivered pursuant to Section 6.01(a), (b) or (c) or Section 6.02(d) (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date (i) on which the Borrower posts such documents, or provides a link thereto on the Borrower's website on the Internet at the website address listed on <u>Schedule 11.02</u>; or (ii) on which such documents are posted on the Borrower's behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether sponsored by the Administrative Agent); provided that (i) the Borrower shall deliver paper copies of such documents to the Administrative Agent and each Lender (by facsimile or electronic mail) of the posting of any such documents and provide to the Administrative Agent by electronic mail electronic versions (<u>i.e.</u>, soft copies) of such documents. Except for such Compliance Certificates, the Administrative Agent shall have no obligation to request the delivery or to maintain copies of the documents referred to above, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery, and each Lender shall be solely responsible for requesting delivery to it or maintaining its copies of such documents.

The Borrower hereby acknowledges that (a) the Administrative Agent and/or the Lead Arrangers will make available to the Lenders and the Issuing Bank materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform") and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or its securities) (each, a "Public Lender"). The Borrower hereby agrees that it will use commercially reasonable efforts to identify that portion of the Borrower Materials that may be distributed to the Public Lenders and that (w) all such Borrower Materials shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, means that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 114 of 432

to have authorized the Administrative Agent, the Lead Arrangers, the Issuing Bank and the Lenders to treat such Borrower Materials as not containing any material non-public information (although it may be sensitive and proprietary) with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (<u>provided</u>, <u>however</u>, that to the extent such Borrower Materials constitute Information, they shall be treated as set forth in <u>Section 11.07</u>); (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Investor"; and (z) the Administrative Agent and the Lead Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding the foregoing, the Borrower shall be under no obligation to mark any Borrower Materials "PUBLIC."

6.03 <u>Notices</u>.

Promptly notify the Administrative Agent and each Lender:

(a) of the occurrence of any Default. <u>Event of Default or any breach of any provision in the Financing</u>.

(b) of any matter that has resulted or could reasonably be expected to result in a Material Adverse Effect, including (i) breach or non-performance of, or any default under, a Contractual Obligation of any Loan Party or any Subsidiary thereof; (ii) any dispute, litigation, investigation, proceeding or suspension between any Loan Party or any Subsidiary thereof and any Governmental Authority; or (iii) the commencement of, or any material development in, any litigation or proceeding affecting any Loan Party or any Subsidiary thereof, including pursuant to any Environmental Laws, but only if any of the matters described in the foregoing <u>clauses (i)</u> through (<u>iii</u>) has resulted or reasonably could be expected to result in a Material Adverse Effect;

(c) after a Responsible Officer of the Borrower or Holdings knows or has reason to know of the occurrence of any ERISA Event or Pension Event that could reasonably be expected to result in liability of one or more Loan Parties or any Subsidiary thereof in excess of \$2,500,000;

(d) of any material change in accounting policies or financial reporting practices by any Loan Party or any Subsidiary thereof;

(e) of the Borrower's determination at any time of the occurrence or existence of any Internal Control Event; and

(f) of any announcement by Moody's or S&P of any change in a Debt Rating.

Each notice pursuant to Section 6.03 (other than Section 6.03(f)) shall be accompanied by a statement of a Responsible Officer of the Borrower setting forth details of the occurrence referred to therein and stating what action the Borrower has taken and proposes to take with respect thereto. Each notice pursuant to Section 6.03(a) shall describe with particularity any and all provisions of this Agreement and any other Loan Document that have been breached.

6.04 <u>Payment of Obligations</u>.

Pay and discharge as the same shall become due and payable, all its obligations and liabilities, including (a) all Tax liabilities, assessments and governmental charges or levies upon it or its properties or assets, unless the same are being contested in good faith by appropriate proceedings diligently conducted and adequate reserves in accordance with GAAP are being maintained by Holdings, the Borrower or such Subsidiary; and (b) all lawful claims which, if unpaid, would by law become a Lien upon its property, unless the same are being contested in good faith by appropriate proceedings diligently conducted, adequate reserves in accordance with GAAP are being maintained by Holdings, the Borrower or such Subsidiary and such contest could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect<u>provided that</u>, obligations of the Loan Parties hereunder are subject to the provisions of the Bankruptcy Code and any required approvals by an applicable order of the Bankruptcy Court; and further provided that the Loan Parties shall be obligated only to pay all of the foregoing to

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 115 of 432

the extent constituting post-petition obligations that constitute administrative expenses under Section 503(b) of the Bankruptcy Code in the Chapter 11 Cases and to perform all material obligations arising from Contractual Obligations entered into after the Petition Date or from Contractual Obligations entered into prior to the Petition Date and assumed.

6.05 <u>Preservation of Existence, Etc.</u>

(a) Preserve, renew and maintain in full force and effect its legal existence and good standing under the Laws of the jurisdiction of its organization except in a transaction permitted by Section 7.04 or 7.05;

(b) take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent that failure to do so could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and

(c) preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

6.06 <u>Maintenance of Properties</u>.

(a) Maintain, preserve and protect all of its material properties and equipment necessary in the operation of its business in good working order and condition, ordinary wear and tear excepted;

(b) make all necessary repairs thereto and renewals and replacements thereof except where the failure to do so could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and

(c) use the standard of care typical in the industry in the operation and maintenance of its facilities.

6.07 <u>Maintenance of Insurance</u>.

(a) Maintain with financially sound and reputable insurance companies having an A.M. Best Financial Strength Rating of at least A which are not Affiliates of the Borrower, insurance with respect to its properties and business against interruption, loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts as are customarily carried under similar circumstances by such other Persons and providing for such insurer to endeavor to give not less than 30 days' prior notice to the Administrative Agent of termination, lapse or cancellation of such insurance. The Loan Parties shall cause the Administrative Agent to be named on each policy evidencing such insurance as secured party, mortgagee, loss payee or additional insured, as applicable, in a manner acceptable to the Administrative Agent.

(b) If any portion of any Mortgaged Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), (x) maintain, or cause to be maintained, with a financially sound and reputable insurer, flood insurance in an amount and otherwise sufficient to comply with all applicable rules and regulations promulgated pursuant to the Flood Insurance Laws and (y) deliver to the Administrative Agent evidence of such compliance in form and substance reasonably acceptable to the Administrative Agent.

6.08 <u>Compliance with Laws</u>.

Comply in all material respects with the requirements of all Laws and all orders, writs, injunctions and decrees applicable to it or to its business or property, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted or (b) the failure to comply therewith could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

6.09 Books and Records.

Maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of Holdings, the Borrower or such Subsidiary, as the case may be.

6.10 Inspection Rights.

(a) Permit the Administrative Agent (and any Lender, at the expense of any such Lender), subject to reasonable advance notice to, and reasonable coordination with, the Borrower and during normal business hours, to visit and inspect the properties of the Loan Parties, at the Borrower's expense as provided in clause (b) below, inspect, audit and make extracts from the Loan Parties' corporate, financial or operating records, and discuss with its officers, employees, agents, advisors and independent accountants (subject to such accountants' customary policies and procedures) the Loan Parties' business, financial condition, assets and results of operations (it being understood that a representative of the Borrower is allowed to be present in any discussions with officers, employees, agent, advisors and independent accountants); provided that the Administrative Agent shall only be permitted to conduct one field examination and one inventory appraisal (in each case, exclusive of any field examination or appraisal initiated during an Increased Inspection Rights Period or during the existence and continuance of ana Default or Event of Default) with respect to any Collateral comprising the Borrowing Base per 12 month period during each consecutive 12 month period (for the avoidance of doubt, the field examination to occur in February 2018 is the one field examination for the period for the purposes of the period ending April 16, 2018); provided, further, that during any Increased Inspection Rights Period, one additional field examination and one additional inventory appraisal of Revolver Priority Collateral will be permitted to be commenced in such 12-month period (and once commenced may be completed regardless of whether such Increased Inspection Rights Period shall terminate), except that during the existence and continuance of ana Default or Event of Default, there shall be no limit on the number of additional field examinations and inventory appraisals of Revolver Priority Collateral that shall be permitted at the Administrative Agent's request. The Administrative Agent may conduct any number of additional field exams and appraisals at its own expense. No such inspection or visit shall unduly interfere with the business or operations of the Loan Parties, nor result in any damage to the property or other Collateral. Neither the Administrative Agent nor any Lender shall have any duty to the Borrower to make any inspection, nor to share any results of any inspection, appraisal or report with the Borrower. The Borrower acknowledges that all inspections, appraisals and reports are prepared by the Administrative Agent and Lenders for their purposes, and the Borrower shall not be entitled to rely upon them.

(b) Reimburse the Administrative Agent for all reasonable and documented out-of-pocket costs and expenses (other than any legal fees or costs and expenses covered under <u>Section 11.04</u>) of the Administrative Agent in connection with (i) two examinations per fiscal year of the Loan Parties' books and records or any other financial or Collateral matters as the Administrative Agent deems appropriate and (ii) field examinations and inventory appraisals of Collateral comprising the Borrowing Base, in each case subject to the limitations on such examinations, audits and appraisals permitted under the preceding paragraph. Subject to and without limiting the foregoing, the Borrower specifically agrees to pay the Administrative Agent's then standard charges for examination activities, including the standard charges of the Administrative Agent's internal appraisal group. This Section shall not be construed to limit the Administrative Agent's right to use third parties for such purposes.

6.11 <u>Use of Proceeds</u>.

Use the proceeds of the Borrowings to consummate the Transactions and for general corporate purposes not in contravention of any Law or of any Loan Document<u>and</u>, after the Ratification Effective Date, in accordance with the Approved Budget; provided, that no more than \$10,000,000 of Revolving Loans may be used to prepay, purchase, redeem or defease the 7.0% Exchangeable Senior Notes due 2017. the funding of the Carve-Out shall not be subject to the Approved Budget.

6.12 <u>Covenant to Guarantee Obligations and Give Security.</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 117 of 432

(a) Upon the formation or acquisition of any new direct or indirect Domestic Subsidiary (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) or Canadian Subsidiary by any Loan Party, then the Borrower shall, at the Borrower's expense:

(i) within 30 days after such formation or acquisition, cause such Domestic Subsidiary or Canadian Subsidiary, as the case may be, and cause each direct and indirect parent of such Domestic Subsidiary or Canadian Subsidiary, as the case may be (if it has not already done so), to duly execute and deliver to the Administrative Agent a Joinder Agreement, pursuant to which such other Loan Party shall guaranty the other Loan Parties' obligations under the Loan Documents and pledge a security interest in and to all of its assets in support of such guaranty in accordance with the terms and conditions of the Security Agreement (for each Domestic Subsidiary) or the Canadian Subsidiary maintains its chief executive office, registered office or domicile (within the meaning of such term in the Civil Code of Québec) or any property or assets within the Province of Québec, also in accordance with a deed of hypothec),

(ii) within 15 days after such formation or acquisition, furnish to the Administrative Agent a description of the real and personal properties of such Domestic Subsidiary or Canadian Subsidiary, as the case may be, in detail satisfactory to the Administrative Agent,

(iii) within 30 days after such formation or acquisition, cause such Domestic Subsidiary or Canadian Subsidiary, as the case may be, and each direct and indirect parent of such Domestic Subsidiary or Canadian Subsidiary, as the case may be (if it has not already done so) to duly execute and deliver to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements, Canadian Subsidiary or Canadian Supplements, IP Security Agreement Supplements, Canadian IP Security Agreement Supplements, as specified by and in form and substance reasonably satisfactory to the Administrative Agent (including delivery of all Pledged Equity in and of such Domestic Subsidiary or Canadian Subsidiary, as the case may be, and other instruments of the type specified in Section 4.01(a)(iii)), securing payment of all the Obligations of such Domestic Subsidiary or Canadian Subsidiary, as the case may be, under the Loan Documents and constituting Liens on all such real and personal properties, and with respect to each Mortgaged Property encumbered by a Mortgage or a Canadian Mortgage, as the case may be, pursuant to this Section 6.12(a)(iii), deliver the items set forth in Sections 4.01(a)(iv) and 6.21 hereof,

(iv) within 30 days after such formation or acquisition, cause such Domestic Subsidiary or Canadian Subsidiary, as the case may be, and each direct and indirect parent of such Domestic Subsidiary or Canadian Subsidiary, as the case may be (if it has not already done so) to take whatever action (including the recording of mortgages, the filing of Uniform Commercial Code or PPSA financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on the properties purported to be subject to the deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), hypothecs, Security Agreement Supplements, Canadian Security Agreement Supplements and security and pledge agreements delivered pursuant to this Section 6.12, enforceable against all third parties in accordance with their terms,

(v) within 60 days after such formation or acquisition, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the Administrative Agent's discretion, may be given by in-house counsel) as to the matters contained in <u>clauses (i)</u>, (iii) and (iv) above, and as to such other matters as the Administrative Agent may reasonably request, and

(vi) as promptly as practicable after such formation or acquisition, deliver, upon the request of the Administrative Agent in its sole discretion, to the Administrative Agent with respect to each parcel of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 118 of 432

real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of \$3,000,000 or more that is owned or held by the entity that is the subject of such formation or acquisition title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent, provided, <u>however</u>, that to the extent that any Loan Party or any of its Subsidiaries shall have otherwise received any of the foregoing items with respect to such real property, such items shall, promptly after the receipt thereof, be delivered to the Administrative Agent.

(b) Upon the acquisition of any property (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) by any Loan Party, if such property, in the judgment of the Administrative Agent, shall not already be subject to a perfected first priority security interest and Lien in favor of the Administrative Agent for the benefit of the Secured Parties (subject to the TermDIP Term Intercreditor Agreement and the First Lien Intercreditor Agreement), then the Borrower shall, at the Borrower's expense:

(i) within 15 days after such acquisition, furnish to the Administrative Agent a description of the property so acquired in detail satisfactory to the Administrative Agent,

(ii) within 30 days after such acquisition, cause the applicable Loan Party to duly execute and deliver to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements, Canadian Security Agreement Supplements, IP Security Agreement Supplements, Canadian IP Security Agreement Supplements and other security and pledge agreements (including any deed of hypothec), as specified by and in form and substance reasonably satisfactory to the Administrative Agent, securing payment of all the Obligations of the applicable Loan Party under the Loan Documents and constituting Liens on all such properties, and with respect to each Mortgaged Property encumbered by a Mortgage or a Canadian Mortgage pursuant to this Section 6.12(b)(ii), deliver the items set forth in Sections 4.01(a)(iv) and 6.21 hereof,

(iii) within 30 days after such acquisition, cause the applicable Loan Party to take whatever action (including the recording of mortgages, the filing of Uniform Commercial Code or PPSA financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on such property, enforceable against all third parties,

(iv) within 60 days after such acquisition, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the Administrative Agent's discretion, may be given by in-house counsel) as to the matters contained in <u>clauses (ii)</u> and <u>(iii)</u> above and as to such other matters as the Administrative Agent may reasonably request, and

(v) as promptly as practicable after any acquisition of any real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of \$3,000,000 or more, deliver, upon the request of the Administrative Agent in its sole discretion, to the Administrative Agent with respect to such real property title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent, <u>provided</u>, <u>however</u>, that to the extent that any Loan Party or any of its Subsidiaries shall have otherwise received any of the foregoing items with respect to such real property, such items shall, promptly after the receipt thereof, be delivered to the Administrative Agent.

(c) Upon the request of the Administrative Agent following the occurrence and during the continuance of a Default, the Borrower shall, at the Borrower's expense:

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 119 of 432

(i) within 10 days after such request, furnish to the Administrative Agent a description of the real and personal properties of the Loan Parties and their respective Subsidiaries in detail reasonably satisfactory to the Administrative Agent,

(ii) within 15 days after such request, duly execute and deliver, and cause each Domestic Subsidiary (other than any Subsidiary that is held directly or indirectly by a CFC) and Canadian Subsidiary of the Borrower (if it has not already done so) to duly execute and deliver, to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements, Canadian IP Security Agreement Supplements and other security and pledge agreements (including any deed of hypothec), as specified by and in form and substance reasonably satisfactory to the Administrative Agent (including delivery of all Pledged Equity and Pledged Debt in and of such Subsidiary, and other instruments of the type specified in <u>Section 4.01(a)(iii)</u>), securing payment of all the Obligations of such Domestic Subsidiary or Canadian Subsidiary, as applicable, under the Loan Documents and constituting Liens on all such properties,

(iii) within 30 days after such request, take, and cause each Domestic Subsidiary (other than any Subsidiary that is held directly or indirectly by a CFC) and Canadian Subsidiary of the Borrower to take, whatever action (including the recording of mortgages, the filing of Uniform Commercial Code or PPSA financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on the properties purported to be subject to the deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements, Canadian Security Agreement Supplements, IP Security Agreement Supplements, Canadian IP Security Agreement Supplements and other security and pledge agreements (including any deed of hypothec) delivered pursuant to this <u>Section 6.12</u>, enforceable against all third parties in accordance with their terms,

(iv) within 60 days after such request, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the discretion of the Administrative Agent, may be given by in-house counsel) as to the matters contained in <u>clauses (ii)</u> and <u>(iii)</u> above, and as to such other matters as the Administrative Agent may reasonably request, and

(v) as promptly as practicable after such request, deliver to the Administrative Agent with respect to each parcel of real property owned or held by the Borrower and its Subsidiaries, title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent.

(d) At any time upon request of the Administrative Agent, promptly execute and deliver any and all further instruments and documents and take all such other action as the Administrative Agent may deem necessary or desirable in obtaining the full benefits of, or (as applicable) in perfecting and preserving the Liens of, such guaranties, deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements, Canadian Security Agreement Supplements, IP Security Agreements (including any deed of hypothec).

Notwithstanding anything in this <u>Section 6.12</u> to the contrary (i) no Loan Party shall be required to deliver a mortgage on any real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of less than \$3,000,000 and (ii) any Subsidiary that pledges assets to secure obligations under the <u>Second Amended and Restated Credit Agreement6% Notes Indenture</u> or that guarantees the obligations under the <u>Second Amended and Restated Credit Agreement6% Notes Indenture</u> shall similarly guarantee and grant Liens to secure the Obligations even if such Subsidiary is a CFC.

6.13 <u>Compliance with Environmental Laws</u>.

Comply, and use commercially reasonable efforts to cause all lessees and other Persons operating or occupying its properties to comply, in all material respects, with all Environmental Laws and Environmental Permits; obtain and renew all Environmental Permits necessary for its operations and properties; and conduct any investigation, study, sampling and testing, and undertake any cleanup, removal, remedial or other action necessary to remove and clean up all Hazardous Materials from any of its properties, in accordance with the requirements of all Environmental Laws; <u>provided</u>, <u>however</u>, that neither Holdings nor any of its Subsidiaries shall be required to undertake any such cleanup, removal, remedial or other action to the extent that its obligation to do so is being contested in good faith and by proper proceedings and appropriate reserves are being maintained with respect to such circumstances in accordance with GAAP.

6.14 <u>Preparation of Environmental Reports</u>.

At the request of the Required Lenders from time to time upon the occurrence of any Release of Hazardous Materials or other event governed by Environmental Law that could reasonably be expected to materially impair the interests of the Secured Parties in any property of the Loan Parties, provide to the Lenders within 90 days after such request, at the expense of the Borrower, an environmental site assessment report for any of such properties, prepared by an environmental consulting firm reasonably acceptable to the Administrative Agent indicating the presence or absence of Hazardous Materials and the estimated cost of any compliance, removal or remedial action in connection with any Hazardous Materials on such properties; without limiting the generality of the foregoing, if the Administrative Agent determines at any time that a material risk exists that any such report will not be provided within the time referred to above, the Administrative Agent may retain an environmental consulting firm to prepare such report at the expense of the Borrower, and the Borrower hereby grants and agrees to cause any Subsidiary that owns any property described in such request to grant at the time of such request to the Administrative Agent, the Lenders, such firm and any agents or representatives thereof an irrevocable non-exclusive license, subject to the rights of tenants, to enter onto their respective properties to undertake such an assessment.

6.15 <u>Further Assurances</u>.

Promptly upon request by the Administrative Agent, or any Lender through the Administrative Agent, (a) correct any material defect or error that may be discovered in any Loan Document or in the execution, acknowledgment, filing or recordation thereof, (b) do, execute, acknowledge, deliver, record, re-record, file, re-file, register and re-register any and all such further acts, deeds, certificates, assurances and other instruments as the Administrative Agent, or any Lender through the Administrative Agent, may reasonably require from time to time in order to (i) carry out more effectively the purposes of the Loan Documents, (ii) to the fullest extent permitted by applicable Law, subject any Loan Party's or any of its Domestic Subsidiaries' (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) or Canadian Subsidiaries' properties, assets, rights or interests to the Liens now or hereafter intended to be covered by any of the Collateral Documents, (iii) perfect and maintain the validity, effectiveness and priority of any of the Collateral Documents and any of the Liens intended to be created thereunder and (iv) assure, convey, grant, assign, transfer, preserve, protect and confirm more effectively unto the Secured Parties the rights granted or now or hereafter intended to be granted to the Secured Parties under any Loan Document or under any other instrument executed in connection with any Loan Document to which any Loan Party or any of its Subsidiaries is or is to be a party, and cause each of its Domestic Subsidiaries (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) or Canadian Subsidiaries to do so, and (c) if an Event of Default has occurred and is continuing, deliver to the Administrative Agent such surveys, appraisals, environmental assessments, title insurance updates or policies and other related documents with respect to each parcel of real estate subject to a Mortgage or Canadian Mortgage.

6.16 <u>Compliance with Terms of Leaseholds</u>.

Make all payments and otherwise perform all obligations in respect of all leases of real property to which the Borrower or any of its Subsidiaries is a party, keep such leases in full force and effect and not allow such leases to lapse or be terminated or any rights to renew such leases to be forfeited or cancelled, notify the Administrative Agent of any default by any party with respect to such leases and cooperate with the Administrative Agent in all

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 121 of 432

respects to cure any such default, and cause each of its Subsidiaries to do so, except, in any case, where the failure to do so, either individually or in the aggregate, could not be reasonably likely to have a Material Adverse Effect.

6.17 [<u>Reserved</u>].

6.18 <u>Material Contracts</u>.

Perform and observe all the terms and provisions of each Material Contract to be performed or observed by it, maintain each such Material Contract in full force and effect, enforce each such Material Contract in accordance with its terms, take all such action to such end as may be from time to time requested by the Administrative Agent and, upon request of the Administrative Agent, make to each other party to each such Material Contract such demands and requests for information and reports or for action as any Loan Party or any of its Subsidiaries is entitled to make under such Material Contract, and cause each of its Subsidiaries to do so, except, in any case, where the failure to do so, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

6.19 [<u>Reserved</u>].

6.20 <u>Collateral Monitoring and Reporting</u>.

Borrowing Base Certificates. By the 20th day after the end of each fiscal month, the Borrower (a) shall deliver to the Administrative Agent (and the Administrative Agent shall promptly deliver same to the Lenders) a Borrowing Base Certificate prepared as of the close of business on the last Business Day of the previous month (provided that, during an Increased Reporting Period, the Borrower shall deliver to the Administrative Agent weekly Borrowing Base Certificates by Wednesday of every week prepared as of the close of business on Friday of the previous week, which weekly Borrowing Base Certificates shall be in standard form unless otherwise reasonably agreed to by the Administrative Agent; it being understood that (i) Inventory amounts shown in the Borrowing Base Certificates delivered on a weekly basis will be based on the Inventory amount (x) set forth in the most recent weekly report, where possible, and (y) for the most recently ended fiscal month for which such information is available with regard to locations where it is impracticable to report Inventory more frequently, and (ii) the amount of Eligible Accounts shown in such Borrowing Base Certificate will be based on the amount of the gross Accounts set forth in the most recent weekly report, less the amount of ineligible Accounts reported for the most recently ended fiscal month). All calculations of Availability in any Borrowing Base Certificate shall be made by the Borrower and certified by a Responsible Officer, provided that the Administrative Agent may from time to time review and adjust any such calculation in consultation with the Borrower to the extent the calculation is not made in accordance with this Agreement or does not accurately reflect the Reserves.

(b) Records and Schedules of Accounts. Each Loan Party shall keep accurate and complete records of its Accounts, including all payments and collections thereon, and shall submit to the Administrative Agent sales, collection, reconciliation and other reports in form reasonably satisfactory to the Administrative Agent on a periodic basis (but not more frequently than at the time of delivery of each of the financial statements required pursuant to Sections 6.01(a) and (b)). The Borrower shall also provide to the Administrative Agent, on or before the 15th day after the end of each fiscal month, a detailed aged trial balance of all Accounts of the Loan Parties as of the end of the preceding month, specifying each Account's Account Debtor name and the amount, invoice date and due date as the Administrative Agent may reasonably request. If Accounts of Loan Parties owing from any single Account Debtor in an aggregate face amount of \$500,000 or more cease to be Eligible Accounts, the Borrower shall notify the Administrative Agent of such occurrence promptly (and in any event within five Business Days) after any Responsible Officer of the Borrower has actual knowledge thereof.

(c) <u>Maintenance of Dominion Account</u>. Within sixty (60) days (or such later date as the Administrative Agent may agree in its reasonable discretion) of the Closing Date (or, with respect to any Deposit Account other than Excluded Deposit Accounts opened following the Closing Date, within thirty (30) days (or such later date as the Administrative Agent may agree in its reasonable discretion) after the date such Loan Party notifies the Administrative Agent of the opening of such Deposit Account or the date any Person becomes a Loan Party hereunder), (i) each Loan Party shall cause each bank or other depository institution at which any Deposit Account is

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 122 of 432

maintained (other than any in respect of Excluded Deposit Account is maintained Accounts), to enter into a Deposit Account Control Agreement that provides for such bank or other depository institution to transfer to a Dominion Account, on a daily basis, all balances in each Deposit Account (other than any Excluded Deposit Account) maintained by any Loan Party with such depository institution for application to the Obligations then outstanding following the receipt by such bank or other depository institution of a Liquidity Notice (it being understood that the Administrative Agent shall reasonably promptly deliver a copy of such Liquidity Notice to the Borrower), (ii) the Borrower shall establish the Dominion Account and obtain an agreement (in form reasonably satisfactory to the Administrative Agent) from the Dominion Account bank, establishing the Administrative Agent's control over and Lien in the Dominion Account, which will be exercised by the Administrative Agent during any Liquidity Period, requiring immediate deposit of all remittances received to a Dominion Account, (iii) each Loan Party shall irrevocably appoint the Administrative Agent as such Loan Party's attorney-in-fact to collect such balances during a Liquidity Period to the extent any such delivery is not so made and (iv) each Loan Party shall instruct each Account Debtor to make all payments with respect to Revolver Priority Collateral into Deposit Accounts subject to Deposit Account Control Agreements, or the Loan Parties shall immediately direct any such payments into Deposit Accounts subject to Deposit Account Control Agreements. Following the commencement of a Liquidity Period, the Administrative Agent shall promptly deliver Liquidity Notices to banks or depository institutions that have entered into Deposit Account Control Agreements, unless otherwise directed by the Required Lenders. The Administrative Agent and the Lenders assume no responsibility to the Borrower for any lockbox arrangement or Dominion Account, including any claim of accord and satisfaction or release with respect to any check, draft or other item of payment payable to a Loan Party (including those constituting proceeds of Collateral) accepted by any bank. <u>All amounts</u> received by Borrowers and Guarantors on account of the Obligations in any Deposit Account prior to the delivery of a Liquidity Notice shall be deemed to be the proceeds of Pre-Petition Collateral and shall be applied by Administrative Agent to the Pre-Petition Obligations until the Pre-Petition Obligations are paid and satisfied in full

(d) <u>Proceeds of Collateral</u>. Each Loan Party shall request in writing and otherwise take all necessary steps to ensure that all payments on Accounts or otherwise relating to Revolver Priority Collateral are made directly to a Deposit Account subject to a Deposit Account Control Agreement (or a lockbox relating to a Dominion Account). If any Loan Party receives cash or any check, draft or other item of payment payable to a Loan Party with respect to any Revolver Priority Collateral, it shall hold the same in trust for the Administrative Agent and promptly deposit the same into any such Deposit Account or Dominion Account.

Administration of Deposit Accounts. Schedule 6.20 sets forth all Deposit Accounts (other than-(e) Excluded Deposit Accounts) maintained by the Loan Parties, including all Dominion Accounts, as of the Closing Date. Subject to Section 6.20(c), each Loan Party shall take all actions necessary to establish the Administrative Agent's control (within the meaning of the UCC) over each such Deposit Account (other than Excluded Deposit Accounts) at all times. Each Loan Party shall be the sole account holder of each Deposit Account and shall not allow any other Person (other than (i) subject to the terms of the DIP Term Intercreditor Agreement, the First Lien Intercreditor Agreement and the 2014 Senior/Junior Intercreditor Agreement, the Administrative Agent, the trustee and/or collateral agent (as the case may be) for the First Lien6% Notes, the trustee and/or collateral agent (as the case may be) for the 2014 Second Lien 4% Notes, the 8.5% Notes, and the applicable depositary bank, and (ii) on and after the date of issuance of the 2016 Secured 4% Notes, subject to the terms of the DIP Term Intercreditor Agreement, the First Lien Intercreditor Agreement, the 2014 Senior/Junior Intercreditor Agreement and the 2016 Intercreditor Agreement (as applicable), the Administrative Agent, the trustee and/or collateral agent (as the case may be) for the First Lien6% Notes, the trustee and/or collateral agent (as the case may be) for the 2014 Second-Lien4% Notes, the 8.5% Notes, the trustee and/or collateral agent (as the case may be) for the 2016 Secured4% Notes, and the applicable depositary bank) to have control over a Deposit Account or any deposits therein. The Borrower shall promptly notify the Administrative Agent of any opening or closing of a Deposit Account by any Loan Party (other than any Excluded Deposit Accounts), and no Loan Party shall open any Deposit Accounts (other than any Excluded Deposit Accounts Account) at a bank not reasonably acceptable to the Administrative Agent.

(f) <u>Account Verification</u>. Whether or not a Default or Event of Default exists, the Administrative Agent shall have the right at any time, in the name of the Administrative Agent, any designee of the Administrative Agent or any Loan Party, to verify the validity, amount or any other matter relating to any Accounts of the Loan Parties by mail, telephone or otherwise. Loan Parties shall cooperate fully with the Administrative Agent in an effort to facilitate and promptly conclude any such verification process.

(g) <u>Administration of Inventory</u>.

(i) <u>Records and Reports of Inventory</u>. Each Loan Party shall keep accurate and complete records of its Inventory, including costs and daily withdrawals and additions, and, prior to the 15th day after the end of each fiscal month, shall submit or otherwise make available to the Administrative Agent inventory and reconciliation reports for such month in form reasonably satisfactory to the Administrative Agent. Each Loan Party shall conduct a physical inventory at least once per calendar year (and on a more frequent basis if requested by the Administrative Agent when an Event of Default exists) and periodic cycle counts consistent with historical practices, and shall provide to the Administrative Agent a report based on each such inventory and count promptly upon completion thereof, together with such supporting information as the Administrative Agent may reasonably request. The Administrative Agent may participate in and observe each physical count.

(ii) <u>Returns of Inventory</u>. No Loan Party shall return any Inventory to a supplier, vendor or other Person, whether for cash, credit or otherwise, unless (a) such return is in the ordinary course of business; (b) no Default, Event of Default or Overadvance exists or would result therefrom; (c) the Administrative Agent is promptly notified if the aggregate value of all Inventory returned in any month exceeds \$250,000; and (d) any payment received by a Loan Party for a return is promptly remitted to the Administrative Agent for application to the Obligations.

(iii) <u>Acquisition, Sale and Maintenance</u>. No Loan Party shall acquire or accept any Inventory on consignment or approval from any individual vendor in excess of \$1,000,000 without notifying the Administrative Agent, and shall take all steps to assure that all Inventory is produced in accordance with applicable Law, including the Fair Labor Standards Act and that any such Inventory is segregated from all other Inventory of the Loan Parties. No Loan Party shall sell any Inventory on consignment or approval or any other basis under which the customer may return or require a Loan Party to repurchase such Inventory to any individual customer in excess of \$1,000,000 without notifying the Administrative Agent. Loan Parties shall use, store and maintain all Inventory with reasonable care and caution, in accordance with applicable standards of any insurance and in conformity in all material respects with all applicable Law, and shall make current rent payments (within applicable grace periods provided for in leases) at all locations where any Collateral is located.

6.21 <u>Post-Closing Actions Related to Real Estate Collateral.</u>

(a) Within ninety (90) days after the Closing Date or such longer period as agreed to by the Administrative Agent in its reasonable discretion, deliver to the Administrative Agent the following items (unless the requirement for any such item is waived by the Administrative Agent due to costs that are excessive in relation to the benefits afforded thereby or on such other basis as the Administrative Agent may determine in its sole discretion):

(i) deeds of trust, trust deeds, deeds to secure debt and mortgages (other than leasehold mortgages and leasehold deeds of trust), in substantially the form of Exhibit G (with such changes as may be satisfactory to the Administrative Agent and its counsel to account for local law matters) and covering the properties identified to be mortgaged on <u>Schedule 5.08(c)</u> (together with the assignments of leases and rents contained therein and each other mortgage delivered pursuant to <u>Section 6.12</u>, in each case as amended, the "<u>Mortgages</u>"), duly executed by the appropriate Loan Party (other than Canadian Guarantors), as well as deeds of trust), in form and substance satisfactory to the Administrative Agent (with such changes as may be satisfactory to the Administrative Agent and its counsel to account for local law matters, including, as the case may be, for purposes of creating a hypothec) and covering the properties identified to be mortgage delivered pursuant to <u>Section 6.12</u>, in each case and each other mortgage delivered pursuant to <u>Section 6.12</u>, and the <u>sectore</u> (with such changes as may be satisfactory to the Administrative Agent and its counsel to account for local law matters, including, as the case may be, for purposes of creating a hypothec) and covering the properties identified to be mortgage delivered pursuant to <u>Section 6.12</u>, in each case as amended, the "<u>Canadian Mortgages</u>"), duly executed by the appropriate Canadian Guarantor;

(ii) evidence that counterparts of the Mortgages and the Canadian Mortgages have been duly executed, acknowledged and delivered and are in form suitable for filing or recording in all filing or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 124 of 432

recording offices that the Administrative Agent may deem necessary or desirable in order to create a valid first and subsisting Lien (subject only to Liens set forth on <u>Schedule 5.08(b)</u> and Liens otherwise permitted by <u>Section 7.01</u>) on the property described therein in favor of the Administrative Agent for the benefit of the Secured Parties and that all filing, documentary, stamp, intangible and recording taxes and fees have been paid (or that arrangements for the payment thereof satisfactory to the Administrative Agent have been made);

(iii) with respect to each Mortgage and Canadian Mortgage, as the case may be, a title search dated contemporaneous with the delivery of such Mortgage or Canadian Mortgage conducted by a title insurance company in the case of such Mortgage and by the Canadian Guarantor's counsel in the case of such Canadian Mortgage which reflects that the Mortgaged Property subject to such Mortgage or Canadian Mortgage, as the case may be, is free and clear of all defects and encumbrances other than Liens set forth on <u>Schedule 5.08(b)</u> and Liens otherwise permitted by <u>Section 7.01</u>;

(iv) evidence of the insurance required by the terms of the Mortgages and the Canadian Mortgages and <u>Section 6.07</u> hereof; and

(v) evidence that all other action that the Administrative Agent may deem necessary or desirable in order to create valid first and subsisting Liens (subject only to Liens set forth on Schedule 5.08(b) and Liens otherwise permitted by <u>Section 7.01</u>) on the property described in the Mortgages and the Canadian Mortgages has been taken.

(b) Within thirty (30) days after the recording of each Mortgage or Canadian Mortgage, as the case may be, delivered pursuant to Section 6.12 or Section 6.21(a) or such longer period as agreed to by the Administrative Agent in its reasonable discretion, a confirmatory lien search conducted by a title insurance company in the case of such Mortgage and by the Canadian Guarantor's counsel in the case of such Canadian Mortgage which reflects that such Mortgage or Canadian Mortgage, as the case may be, was duly recorded and that the Mortgaged Property subject to such Mortgage or Canadian Mortgage, as the case may be, was free and clear of all defects and encumbrances other than Liens set forth on Schedule 5.08(b) and Liens otherwise permitted by Section 7.01 as of the date of such recording.

(c) If any Event of Default shall have occurred and be continuing, promptly following written request by the Administrative Agent, deliver to the Administrative Agent the following items with respect to each Mortgage or Canadian Mortgage delivered pursuant to Section 6.12 or 6.21(a) hereof or the applicable Mortgaged Property:

(i) if requested by the Administrative Agent, fully paid American Land Title Association Lender's Extended Coverage title insurance policies (the "<u>Mortgage Policies</u>") in form and substance, with endorsements (other than zoning, creditors rights and environmental) and in amounts reasonably acceptable to the Administrative Agent (but in any event not greater than the fair market value of the applicable Mortgaged Property, as reasonably determined by the Borrower), issued, coinsured and reinsured by title insurers reasonably acceptable to the Administrative Agent, insuring the Mortgages to be valid first and subsisting Liens on the property described therein, free and clear of all defects (including, but not limited to, mechanics' and materialmen's Liens) and encumbrances, excepting only Permitted Encumbrances and Liens otherwise permitted by <u>Section 7.01</u>, and providing for such other affirmative insurance (including endorsements for future advances under the Loan Documents and for mechanics' and materialmen's Liens) and such coinsurance and direct access reinsurance as the Administrative Agent may deem necessary or desirable;

(ii) if requested by the Administrative Agent, American Land Title Association/American Congress on Surveying and Mapping form surveys (or with respect to any Canadian property, a new survey prepared by a qualified land surveyor), for which all necessary fees (where applicable) have been paid, certified to the Administrative Agent and the issuer of the Mortgage Policies in a manner satisfactory to the Administrative Agent by a land surveyor duly registered and licensed in the States, provinces or territories in which the property described in such surveys is located and acceptable to the Administrative Agent, showing all buildings and other improvements, any off-site improvements, the location of any easements,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 125 of 432

parking spaces, rights of way, building set-back lines and other dimensional regulations and the absence of encroachments, either by such improvements or on to such property, and other defects, other than encroachments and other defects acceptable to the Administrative Agent or that do not materially adversely affect the value of the applicable Mortgaged Property;

(iii) if requested by the Administrative Agent, Phase I environmental site assessments as to the properties described in the Mortgages, in form and substance and from professional firms reasonably acceptable to the Administrative Agent; and

(iv) if requested by the Administrative Agent, an appraisal of each of the properties described in the Mortgages complying with the requirements of the Federal Financial Institutions Reform, Recovery and Enforcement Act of 1989.

ARTICLE VII

NEGATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder or any Revolving Loan or other Obligation hereunder shall remain unpaid or unsatisfied, or any Letter of Credit shall remain outstanding (unless Cash Collateralized or backstopped on terms reasonably satisfactory to the Administrative Agent), the Borrower shall not, nor shall it permit any Subsidiary to, directly or indirectly:

7.01 <u>Liens</u>.

Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, or sign or file or suffer to exist under the Uniform Commercial Code or the PPSA of any jurisdiction a financing statement that names the Borrower, Holdings or any of its other Subsidiaries as debtor, or assign any accounts or other right to receive income, other than the following (including any financing statements filed in connection with any of the following):

(a) (i) Liens pursuant to any Loan Document;-and (ii) Liens securing Indebtedness and other obligations under the <u>6% Notes so long as such Liens are subject to the First Lien Intercreditor Agreement; and (iii)</u> Liens securing Indebtedness and other obligations under the <u>DIP</u> Term Documents so long as such Liens are subject to the <u>DIP</u> Term Intercreditor Agreement;

(b) Liens <u>not otherwise set forth in clauses (a) through (r) of this Section 7.01 and existing on the</u> <u>ClosingRatification Effective</u> Date and listed on <u>Schedule 7.01 and to the Ratification Agreement</u>;

(c) Liens existing on the Closing Date and immediately prior to the commencement of the Chapter 11 Cases, and listed on Schedule 5.08(b), any renewals or extensions thereof, provided that (i) the property covered thereby is not changed, (ii) the amount secured or benefited thereby is not increased except as contemplated by Section 7.02(d), as applicable, (iii) the direct or any contingent obligor with respect thereto is not changed, and (iv) any renewal or extension of the obligations secured or benefited thereby is permitted by Section 7.02(d) or Section 7.15(d);hereunder.

(d) Liens for taxes not yet due or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;

(e) carriers', warehousemen's, mechanics', materialmen's, repairmen's, construction or other like Liens arising in the ordinary course of business which are not overdue for a period of more than 30 days or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 126 of 432

(f) pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA or Canadian Pension Legislation;

(g) deposits to secure the performance of bids, trade contracts and leases (other than Indebtedness), statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business;

(h) (i) easements, (ii) rights-of-way, (iii) servitudes, (iv) restrictions, (v) reservations, limitations, provisos and conditions expressed in an original grant from the Crown and (vi) other similar encumbrances affecting real property which, in the aggregate, are not substantial in amount, and which do not in any case materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Person;

(i) Liens securing judgments for the payment of money not constituting an Event of Default under <u>Section 8.01(h);</u>

(j) Liens securing Indebtedness permitted under <u>Section 7.02(f)</u>; <u>provided</u> that (i) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and (ii) the Indebtedness secured thereby does not exceed the cost or fair market value, whichever is lower, of the property being acquired on the date of acquisition;

(k) Liens on property of a Person existing at the time such Person is merged into, amalgamated with or consolidated with the Borrower or any Subsidiary of the Borrower or becomes a Subsidiary of the Borrower; provided that such Liens were not created in contemplation of such merger, amalgamation, consolidation or Investment and do not extend to any assets other than those of the Person merged into, amalgamated with or consolidated with the Borrower or such Subsidiary or acquired by the Borrower or such Subsidiary, and the applicable Indebtedness secured by such Lien is permitted under <u>Section 7.02(g)</u>;

(I) other Liens securing Indebtedness outstanding in an aggregate principal amount not to exceed \$25,000,000,1,000,000, provided that no such Lien shall extend to or cover Revolver Priority Collateral with a book or fair market value (whichever is higher) of over \$1,000,000250,000 in the aggregate and any such Lien on the Revolver Priority Collateral shall be on a junior priority basis;

(m) [reserved];

(n) the replacement, extension or renewal of any Lien permitted by <u>clauses (i)</u> through (k) above upon or in the same property theretofore subject thereto;

(o) Liens securing Indebtedness evidenced by the Second Lien Notes outstanding on the Closing Dateor Indebtedness incurred pursuant to Section 7.02(n), so long as such Liens are subject and subordinate to the Liensupon the Collateral under the Loan Documents in accordance with the terms and conditions of the Intercreditor-Agreement; the Carve-Out;

(n) Liens in favor of a depositary institution on cash deposits and other funds maintained with such depository institution in each case arising in the ordinary course of business by virtue of any statutory or common law provision relating to banker's liens, including Section 4-208 of the UCC;

(o) [reserved];

(p) Liens on Collateral securing Indebtedness and other obligations evidenced by the 2014 Second-Lien Notes and other Indebtedness permitted under Section 7.02(p)6% Notes and the documents executed and delivered by the Loan Parties in connection therewith, so long as such Liens are subject and subordinate to the Liens-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 127 of 432

upon the Collateral under the Loan Documents in accordance with the terms and conditions of the 2014<u>to the First</u> Lien Intercreditor Agreement;

(p)—Liens on Collateral securing Indebtedness and other obligations evidenced by the First-Lien Notes and other Indebtedness permitted under <u>Section 7.02(o)</u> and <u>Section 7.02(q)</u> and the documentsexecuted and delivered by the Loan Parties in connection therewith, so long as such Liens are subject to the First Lien Intercreditor Agreement; and

(q) Liens on Collateral securing Indebtedness and other obligations evidenced by the 2016 Secured Notes and the documents executed and delivered by the Loan Parties in connection therewith, so long as such Liens are subject to the 2014 Intercreditor Agreement, the First Lien Intercreditor Agreement and the 2016 Intercreditor Agreement; and

(r) any other Liens consented to in writing by the Administrative Agent and Required Lenders.

7.02 Indebtedness.

Create, incur, assume or suffer to exist any Indebtedness, except:

(a) obligations (contingent or otherwise) of the Borrower existing or arising under any Swap Contract; <u>provided</u> that (A) such obligations are (or were) entered into by such Person in the ordinary course of business for the purpose of directly mitigating risks associated with liabilities, commitments, investments, assets, or property held or reasonably anticipated by such Person, or changes in the value of securities issued by such Person, and not for purposes of speculation or taking a "market view," and (B) such Swap Contract does not contain any provision exonerating the non-defaulting party from its obligation to make payments on outstanding transactions to the defaulting party;

(b) unsecured Indebtedness of a Subsidiary of the Borrower owed to the Borrower or a wholly-owned Subsidiary of the Borrower <u>as of the Ratification Effective Date</u>, which Indebtedness shall (i) in the case of Indebtedness owed to a Loan Party, constitute "Pledged Debt" under the Security Agreement or the Canadian Security Agreement, as the case may be, (ii) be on terms (including subordination terms) acceptable to the Administrative Agent and (iii) be otherwise permitted under the provisions of <u>Section 7.03</u>;

(i) Indebtedness under the Loan Documents; and (ii) (x) prior to the issuance of the 2014 Notes-(c) pursuant to Section 7.02(o). Indebtedness under the Second Amended and Restated Credit Agreement not to exceed-\$360,000,000 at any one time outstanding and (y) on and after the date of issuance of the 2014 Notes, this clause (ii) and the two provisos of this Section 7.02(c) shall have no further force or effect; provided that the total Indebtednessincurred pursuant to this clause (c) shall not exceed \$560,000,000 at any one time outstanding; provided further,however, that notwithstanding the foregoing provisions of this clause (c), the Borrower and its Subsidiaries mayincur Indebtedness under the Loan Documents and the Second Amended and Restated Credit Agreement in anaggregate amount in excess of \$560,000,000 (and, for the avoidance of doubt, Indebtedness under the Second-Amended and Restated Credit Agreement in excess of \$360,000,000) so long as (A) immediately after giving effectto the incurrence of such Indebtedness, Holdings and its Subsidiaries shall have a Consolidated First Lien Leverage-Ratio calculated on a Pro Forma Basis, as determined on the basis of the financial information most recentlydelivered to the Administrative Agent and the Lenders pursuant to Section 6.01(a) or (b), as applicable, less than 2.25 to 1.0, (B) after giving effect to the incurrence of such Indebtedness, the sum of (x) the Aggregate-Commitments and (y) the aggregate outstanding principal amount of the Term Loans shall not exceed the Maximum-First Lien Principal Indebtedness (as defined in the Intercreditor Agreement), (C) any such Indebtedness in excess of \$560,000,000 shall be used solely to finance all or a portion of the purchase price of any Acquisition permitted under-Section 7.03 and related fees, costs and expenses, and (D) no Default exists immediately prior to, or would result from, the issuance or incurrence of such Indebtedness; (ii) Indebtedness under the 6% Notes Indenture not to exceed \$540,000,000 at any one time outstanding, and (iii) Indebtedness under the DIP Term Documents not to exceed \$100,000,000 at any one time outstanding:

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 128 of 432

Indebtedness (excluding Indebtedness permitted to be incurred under Section 7.02(k)) outstanding (d) on the Closing Date and listed on Schedule 7.02 and any refinancings, refundings, renewals or extensions of such-Indebtedness or of Indebtedness permitted under Section 7.02(g); provided that (i) no Default shall have occurredand be continuing or would result from any such refinancing, refunding, renewal or extension on a Pro Forma Basis, (ii)) the amount of such Indebtedness is not increased at the time of such refinancing, refunding, renewal orextension except by an amount equal to any premiums required to be paid, and fees and expenses reasonablyincurred, in connection with such refinancing and by an amount equal to any existing commitments unutilizedthereunder and any accrued and unpaid interest thereon, (iii) the direct or any contingent obligor with respect theretois not changed as a result of or in connection with such refinancing, refunding, renewal or extension, and (iv) the terms relating to principal amount, amortization, maturity, collateral (if any) and subordination (if any), and other material terms taken as a whole, of any such refinancing, refunding, renewing or extending Indebtedness, and of any agreement entered into and of any instrument issued in connection therewith, are no less favorable in any material respect to the Loan Parties or the Lenders than the terms of any agreement or instrument governing the Indebtedness being refinanced, refunded, renewed or extended and the interest rate applicable to any such refinancing, refunding, renewing or extending Indebtedness does not exceed the then applicable market interest rate;

(e) Guarantees of Holdings, the Borrower, or any Subsidiary in respect of Indebtedness otherwise permitted hereunder of the Borrower or any of its Subsidiaries;

(f) Indebtedness in respect of Capitalized Leases, Synthetic Lease Obligations, Synthetic Debt and purchase money obligations for fixed or capital assets within the limitations set forth in <u>Section 7.01(i)</u>; provided, <u>however</u>, that the aggregate amount of all such Indebtedness at any one time outstanding shall not exceed \$50,000,00025,000,000;

(g) Indebtedness of any Person that becomes a Subsidiary of the Borrower after the Closing Date inaccordance with the terms of Section 7.03(h), which Indebtedness is existing at the time such Person becomes a-Subsidiary of the Borrower (other than Indebtedness incurred in contemplation of such Person's becoming a-Subsidiary of the Borrower);other Indebtedness in any aggregate amount of up to \$1,000,000 which may be secured by Liens permitted by Section 7.01(k), to the extent permitted thereunder;

(h) (i) Indebtedness of any Subsidiary that is not a Loan Party owing to any other Subsidiary that is not a Loan Party and (ii) other Indebtedness of Subsidiaries which are not Loan Parties in an aggregate principal amount not to exceed \$5,000,000 at any time outstanding;

(i) <u>otherunsecured</u> Indebtedness in <u>anyan</u> aggregate amount of up to \$25,000,000 which may besecured by Liens permitted under <u>Section 7.01(k)1,000,000</u>, provided, that, such Indebtedness is in the Approved Budget;

(j) other unsecured Indebtedness, provided that (A) no Default exists immediately prior to, or would result from, on a Pro Forma Basis, the incurrence of such Indebtedness, (B) no portion of such Indebtedness is scheduled to be paid (either at maturity or as amortization or as a mandatory prepayment (except, in the case of the initial Unsecured Term Loan Facility only, any mandatory prepayments thereunder that are paid only after any mandatory prepayments then due and payable hereunder are satisfied) prior to 91 days after the Maturity Date of any Revolving Loan hereunder, (C) such Indebtedness is not Indebtedness of Holdings to the Borrower or any of its-Subsidiaries, (D) the material terms and conditions of such Indebtedness are not more restrictive than the terms and conditions of this Agreement, (E) [reserved] and (F) if such Indebtedness is subordinated Indebtedness, such-Indebtedness is subordinated in right of payment to the Obligations pursuant to terms of subordination in form and substance acceptable to the Administrative Agent[reserved];

(k) Indebtedness evidenced by (i) the Senior Notes, (ii) the 2016 Unsecured Notes issued in exchangefor all or any portion of the Senior Notes and (iii) the 2016 Secured Notes; unsecured Indebtedness of a Subsidiary of the Borrower owed to the Borrower or a wholly-owned Subsidiary of the Borrower, which Indebtedness shall (i) in the case of Indebtedness owed to a Loan Party, constitute "Pledged Debt" under the Security Agreement, (ii) be on terms (including subordination terms) acceptable to the Administrative Agent and (iii) be otherwise permitted under the provisions of Section 7.03;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 129 of 432

(I) Indebtedness issued as consideration for all or any portion of the purchase price of any Acquisitionpermitted under <u>Section 7.03(h)</u>, provided that (A) no Default exists immediately prior to, or would result from, on a-Pro Forma Basis, the incurrence of such Indebtedness, (B) such Indebtedness is not Indebtedness of a Loan Party toany other Loan Party and (C) the aggregate amount of such Indebtedness shall not exceed \$20,000,000;evidenced by (i) the 6% Notes, the (ii) 4% Notes, (iii) the 6% Unsecured Notes and (iv) the 8.5% Notes;

(m) any refinancings, refundings, replacements or exchanges of the Senior Notes in whole or in partfollowing the Amendment No. 4 Effective Date; <u>provided</u> that (i) no Default exists immediately prior to (except tothe extent such Default would be cured by), or would result from, any such refinancing, refunding, replacement or exchange on a Pro Forma Basis, (ii) the amount of such Indebtedness is not increased at the time of such refinancing, refunding, replacement or exchange except by an amount equal to fees and expenses reasonably incurred in connection with such refinancing, refunding, replacement or exchange and any accrued and unpaid interest thereon, (iii) the direct or any contingent obligor with respect thereto is not changed as a result of or in connection with suchrefinancing, refunding, replacement or exchange, (iv) such Indebtedness is unsecured, (v) no portion of such-Indebtedness is scheduled to be paid (either at maturity or as amortization or as a mandatory prepayment) prior to 91days after the Maturity Date of any Revolving Loan hereunder and (vi) the material terms and conditions of such-Indebtedness are consistent with, or, taken as a whole, no less favorable in any material respect to the Borrower than, the terms of the Senior Notes Indenture; [reserved]:

(n) [reserved]:

(n) Indebtedness issued or incurred from time to time in respect of the Second Lien Notes after the Closing Date or other Indebtedness secured by Liens on the Collateral that are junior to the Liens securing the Obligations (including, without limitation, Permitted Additional Pari Passu Obligations (as defined in the Second Lien Notes Indenture)); provided, that (x) immediately after giving effect to such issuance or incurrence, Holdings and its Subsidiaries shall have a Consolidated Secured Debt Ratio not greater than 4.5 to 1.0, (y) no Default exists immediately prior to, or would result from, on a Pro Forma Basis, the incurrence of such Indebtedness and (z) such-Indebtedness is in compliance with the terms and conditions of the Interceditor Agreement; [reserved]:

(o) Indebtedness in respect of the First Lien Notes or, after the Amendment No. 3 Effective Date,. Indebtedness issued or incurred from time to time in respect of Permitted Additional Pari Passu Obligations (as defined in the First Lien Notes Indenture); <u>provided</u>, that, with respect to any such issuance or incurrence of such-Permitted Additional Pari Passu Obligations, (i) immediately after giving effect to such issuance or incurrence, Holdings and its Subsidiaries shall have a Consolidated Senior Secured Debt Ratio not greater than 4.0 to 1.0, (ii) no-Default exists immediately prior to, or would result from, on a Pro Forma Basis, the incurrence of such Indebtedness, (iii) such Indebtedness is in compliance with the terms and conditions of the First Lien Intercreditor Agreement, (iv) the proceeds of such Indebtedness are used to refinance, or such Indebtedness is issued in exchange for, all or anyportion of the Borrower's 11.50% Senior Notes due 2017, (v) such Indebtedness does not contain any mandatoryprepayments or mandatory redemptions (other than customary asset sale, event of loss and change of control offercovenants) or require any amortization, (vi) the Weighted Average Life to Maturity of such Indebtedness at the time of incurrence thereof shall not be shorter than the period from the date of such incurrence to the date that is 180 daysafter the Maturity Date of any then outstanding Revolving Commitments and (vii) such Indebtedness is otherwisepermitted under the First Lien Notes Indenture; [reserved];

(p) Indebtedness in respect of the 2014 Second Lien Notes or, after the Amendment No. 3 Effective-Date, Indebtedness issued or incurred from time to time in respect of Permitted Additional Pari Passu Obligations (as defined in the 2014 Second Lien Notes Indenture); <u>provided</u>, that, with respect to any such issuance or incurrence of such Permitted Additional Pari Passu Obligations, (i) immediately after giving effect to such issuance or incurrence, Holdings and its Subsidiaries shall have a Consolidated Total Secured Debt Ratio not greater than 5.25 to 1.00, (ii)no Default exists immediately prior to, or would result from, on a Pro Forma Basis, the incurrence of such-Indebtedness, (iii) such Indebtedness is in compliance with the terms and conditions of the 2014 Intercreditor-Agreement, (iv) the proceeds of such Indebtedness are used to refinance, or such Indebtedness is issued in exchangefor, all or any portion of the Borrower's 11.50% Senior Notes due 2017, (v) such Indebtedness does not contain any-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 130 of 432

mandatory prepayments or mandatory redemptions (other than customary asset sale, event of loss and change of control offer covenants) or require any amortization, (vi) the Weighted Average Life to Maturity of such-Indebtedness at the time of incurrence thereof shall not be shorter than the period from the date of such incurrence tothe date that is 180 days after the Maturity Date of any then outstanding Revolving Commitments and (vii) such-Indebtedness is otherwise permitted under the 2014 Second Lien Notes Indenture; and [reserved]:

(q) any refinancings, refundings, replacements or exchanges of the First Lien Notes in whole or in partfollowing the Amendment No. 4 Effective Date; <u>provided</u> that (i) no Default exists immediately prior to (except tothe extent such Default would be cured by), or would result from, any such refinancing, refunding, replacement or exchange on a Pro Forma Basis, (ii) the amount of such Indebtedness is not increased at the time of such refinancing, refunding, replacement or exchange except by an amount equal to any premiums required to be paid, and fees and expenses reasonably incurred, in connection with such refinancing, refunding, replacement or exchange and anyaccrued and unpaid interest thereon, (iii) the direct or any contingent obligor with respect thereto is not changed as a result of or in connection with such refinancing, replacement or exchange, (iv) such Indebtedness is in compliance with the terms and conditions of the First Lien Intercreditor Agreement, (v) no portion of such-Indebtedness is scheduled to be paid (either at maturity or as amortization or as a mandatory prepayment) prior to 91days after the Maturity Date of any Revolving Loan hereunder and (vi) the material terms and conditions of such-Indebtedness are consistent with, or, taken as a whole, no less favorable in any material respect to the Borrower than, the terms of the First Lien Notes Indenture.[reserved]: and

(r) any other Indebtedness consented to in writing by the Administrative Agent and Required Lenders.

7.03 <u>Investments</u>.

Make or hold any Investments, except:

(a) Investments held by Holdings and its Subsidiaries in the form of Cash Equivalents;

(b) advances to officers, directors and employees of Holdings and its Subsidiaries in an aggregate amount not to exceed $\frac{2,000,000500,000}{2,000,000}$ at any time outstanding, for travel, entertainment, relocation and analogous ordinary business purposes, in each case in accordance with the Approved Budget;

(c) (i) Investments by Holdings and its Subsidiaries in their respective Subsidiaries outstanding on the Closing Date, (iiRatification Effective Date, (ii) Investments by any Loan Party in any Loan Party (other than Holdings), (iii) Investments by any Subsidiary that is not a Loan Party in any Subsidiary that is not a Loan Party, and (iv) additional Investments by Holdings and its Subsidiaries in Loan Parties (other than Holdings), and (iii) additional Investments by Subsidiaries of the Borrower that are not Loan Parties in other Subsidiaries that are not Loan Parties; provided that each such provided, that (A) in the case of the foregoing clause (iv), each such Investment is in the Approved Budget and (B) each Investment that consists of intercompany Indebtedness owing to a Loan Party shall be evidenced by an intercompany note which shall be delivered to, and in form reasonably acceptable to, the Administrative Agent and shall constitute "Pledged Debt" pursuant to the terms of the Security Agreement or the Canadian Security Agreement, as the case may be;

(d) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business, and Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors to the extent reasonably necessary in order to prevent or limit loss;

(e) Guarantees permitted by <u>Section 7.02</u>;

(f) Investments existing on the $\frac{\text{ClosingRatification Effective}}{\text{Section 7.03(c)(i)}}$ and set forth on $\frac{\text{Schedule 5.08(e)}}{\text{Schedule 5.08(e)}}$;

(g) Investments by the Borrower in Swap Contracts permitted under Section 7.02(a);

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 131 of 432

(h) any Acquisition; provided that each of the following conditions shall be satisfied:[reserved];

any Subsidiary created or acquired in connection with such Acquisition shall comply with the requirements of <u>Section 6.12</u>;

the lines of business of the Person to be (or the property of which is to be) so Acquired shall besubstantially similar or related to one or more of the principal businesses of the Borrower and its-Subsidiaries in the ordinary course;

immediately before and immediately after giving effect to any such Acquisition, the Acquisitions and Investment Conditions shall be satisfied on a Pro Forma Basis for such Acquisition; and

the Borrower shall have delivered to the Administrative Agent and each Lender, at least two-Business Days prior to the date on which any such Acquisition is to be consummated, a certificateof a Responsible Officer, in form and substance reasonably satisfactory to the Administrative-Agent, certifying that all of the requirements set forth in this clause (h) have been satisfied or willbe satisfied on or prior to the consummation of such Acquisition;

(i) Investments by the Borrower and its Subsidiaries not otherwise permitted under this <u>Section 7.03</u>, so long as the Acquisitions and Investment Conditions are satisfied on a Pro Forma Basis immediately after giving effect to such Investment; provided that, with respect to each Investment made pursuant to this <u>Section</u>. <u>7.03(i):[reserved]:</u>

> such Investment shall not include or result in any contingent liabilities that could reasonably beexpected to be material to the business, financial condition, operations or prospects of the Borrower and its Subsidiaries, taken as a whole (as determined in good faith by the board of directors (or persons performing similar functions) of the Borrower or such Subsidiary if the board of directors is otherwise approving such transaction and, in each other case, by a Responsible-Officer); and

such Investment shall be in property that is part of, or in lines of business that are substantiallysimilar or related to, one or more of the principal businesses of the Borrower and its Subsidiariesin the ordinary course; and

(j) Investments by the Borrower and its Subsidiaries not otherwise permitted under this <u>Section 7.03</u>not to exceed \$3,000,000 in any fiscal year (provided that no more than \$6,000,000 in the aggregate shall bepermitted during the term of this Agreement), so long as the Acquisitions and Investment Conditions are satisfied ona Pro Forma Basis immediately after giving effect to such Investment.[reserved]:

(k) [reserved]; and

(1) any other Investments consented to in writing by the Administrative Agent and Required Lenders.

7.04 **Fundamental Changes**.

Merge, dissolve, liquidate, consolidate or amalgamate with or into another Person, or Dispose of (whether in one transaction or in a series of transactions) all or substantially all of its assets (whether now owned or hereafter acquired) to or in favor of any Person, except that, so long as no Default exists or would result therefrom on a Pro-Forma Basispro forma basis:

(a) any Domestic Subsidiary may merge with (i) the Borrower, <u>provided</u> that the Borrower shall be the continuing or surviving Person, or (ii) any one or more other Domestic Subsidiaries, <u>provided</u> that when any Loan Party is merging with another Subsidiary, such Loan Party shall be the continuing or surviving Person; <u>and</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 132 of 432

any Canadian Subsidiary may amalgamate with any one or more other Canadian Subsidiaries, <u>provided</u> that when any Canadian Guarantor is amalgamating with another Canadian Subsidiary, such Canadian Guarantor shall be the continuing or surviving Person;

any Loan Party may Dispose of all or substantially all of its assets (upon voluntary liquidation or otherwise) to the Borrower or to another Loan Party (other than Holdings);

in connection with any Acquisition permitted under <u>Section 7.03</u>, any Subsidiary of the Borrower may merge into or amalgamate or consolidate with any other Person or permit any other Person to merge into or amalgamate or consolidate with it; <u>provided</u> that (i) the Person surviving or continuing suchmerger, amalgamation or consolidation shall be a wholly owned Subsidiary of the Borrower and (ii) in the case of any such merger, amalgamation or consolidation to which any Loan Party is a party, such Loan-Party is the surviving or continuing Person;

<u>so long as no Default has occurred and is continuing or would result therefrom, each of</u> the Borrower and any of its Subsidiaries may merge into or amalgamate or consolidate with any other-Person or permit any other Person to merge into or amalgamate or consolidate with it; provided, however, that in each case, immediately after giving effect thereto (i) in the case of any such merger, amalgamation or consolidation to which the Borrower is a party, the Borrower is the surviving corporation and (ii) in the case of any such merger, amalgamation or consolidation to which any Loan Party (other than the Borrower) is a party, such Loan Party is the surviving or continuing corporation; and <u>any Non Operating Subsidiary maybe dissolved or liquidated so long as such dissolution or liquidation would not reasonably be expected to result in a Material Adverse Effect.(b) <u>any Loan Party may Dispose of all or substantially all of its</u> <u>assets (upon voluntary liquidation or otherwise) to the Borrower or to another Loan Party (other than Holdings).</u></u>

7.05 <u>Dispositions</u>.

Make any Disposition or enter into any agreement to make any Disposition, except<u>(in each case, in accordance with the Approved Budget)</u>:

(a) Dispositions of obsolete or worn out property, whether now owned or hereafter acquired, in the ordinary course of business; and

(b) Dispositions of inventory in the ordinary course of business;

(c) Dispositions of equipment or real property to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property or (ii) the net proceeds of such Disposition are reasonably promptly applied to the purchase price of such replacement property;

(d) Dispositions of property by any Subsidiary to the Borrower or to a wholly-owned Subsidiary; provided that if the transferor of such property is a Guarantor, the transferee thereof must either be the Borrower or a Guarantor;

(e) Dispositions permitted by <u>Section 7.04</u>;

(f) Dispositions by the Borrower and its Subsidiaries of property pursuant to sale leasebacktransactions, of equipment in connection with plant closures and consolidations not to exceed \$3,000,000 in the aggregate: provided, that the book value of all property so Disposed of shall not exceed \$15,000,000 in the aggregate from and after the Closing Date (excluding Dispositions by the Borrower and its Subsidiaries of property pursuant to sale leaseback transactions in connection with the incurrence of any Indebtedness in respect of Capitalized Leases or Synthetic Lease Obligations permitted under Section 7.02(f))to the extent the aggregate amount of net cash proceeds.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 133 of 432

of all equipment sold pursuant to this clause (f) exceeds \$1,000,000, such excess proceeds are applied to repay the Indebtedness outstanding in respect of the DIP Term Credit Agreement;

(g) non-exclusive licenses of IP Rights in the ordinary course of business and substantially consistent with past practice for terms not exceeding ten years; and

Dispositions by the Borrower and its Subsidiaries not otherwise permitted under this Section 7.05:-(h) provided that (i) at the time of such Disposition, no Default shall exist or would result from such Disposition on a Pro Forma Basis, (ii) in connection with the Disposition of any Revolver Priority Collateral, (x) the Borrower shallreceive not less than 75% of consideration in connection with such Disposition in the form of cash and (y) a new-Borrowing Base Certificate shall be delivered and (iii) the aggregate book value of all property Disposed of inreliance on this clause (h) in any fiscal year shall not exceed \$35,000,000 (provided, that this clause (iii) shall not apply to any Disposition for which the net proceeds thereof are applied in accordance with Section 4.07 of the First-Lien Notes Indenture so long as either (A) (x)(1) Availability on a Pro Forma Basis immediately after giving effectto such action would be at least 15.0% of the Aggregate Commitments and (2) over the 30 consecutive days prior toconsummation of such action, Availability on a Pro Forma Basis for such action averaged no less than 15.0% of the Aggregate Commitments and (v) the Consolidated Fixed Charge Coverage Ratio would be at least 1.0 to 1.0 on a Pro-Forma Basis for such action, or (B) (1) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 20.0% of the Aggregate Commitments and (2) over the 30 consecutive days prior toconsummation of such action, Availability on a Pro Forma Basis for such action averaged no less than 20.0% of the Aggregate Commitments);[reserved];

provided, however, that any Disposition pursuant to <u>Section 7.05(a)</u> through <u>Section 7.05(h)</u> shall be for fair market value (as reasonably determined by the Borrower or the applicable Subsidiary).

7.06 <u>Restricted Payments</u>.

Declare or make, directly or indirectly, any Restricted Payment, or incur any obligation (contingent or otherwise) to do so, except that <u>(in each case, in accordance with the Approved Budget)</u>:

(a) each Subsidiary (hereinafter, the "Payor") may make Restricted Payments to (i) the Borrower, (ii) any Subsidiaries of the Borrower that are Guarantors and, (iii) so long as no Default has occurred and is continuing, any other Person that owns a direct Equity Interest in such Subsidiarythe Payor, ratably according to their respective holdings of the type of Equity Interest in respect of which such Restricted Payment is being made; and (iv) in the event the Payor files a consolidated, combined, unitary or similar type income tax return with Holdings, the Borrower, or any other Subsidiaries that are Guarantors and own a direct or indirect Equity Interest in Payor, to such parties to permit such parties to pay federal and state income taxes then due and payable, franchise taxes and other similar licensing expenses incurred in the ordinary course of business; provided, that the amount of such Restricted Payment shall not be greater than the amount of such taxes or expenses that would have been due and payable by such parties and any Subsidiaries of such parties had such parties not filed a consolidated, combined, unitary or similar type return with the Payor.

(b) Holdings and each Subsidiary may declare and make dividend payments or other distributionspayable solely in the common stock or other common Equity Interests of such Person; Holdings and each Subsidiary may purchase, redeem or otherwise acquire its common Equity Interests with the proceeds received from the substantially concurrent issue of new common Equity Interests; [reserved];

(c) Holdings and the Borrower may declare and make any Restricted Payments so long as the Distribution Conditions are satisfied on a Pro Forma Basis immediately after giving effect to such Restricted Payment; [reserved]:

(d) [reserved]; and

(e) the Borrower may declare and pay cash dividends to Holdings (i) in the amounts of and at the times necessary to permit Holdings to make payments permitted pursuant to Section 7.06(d) above and (ii) so long as

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 134 of 432

no Default exists or would be caused thereby, in an aggregate amount not to exceed \$10,000,0001,000,000 in any fiscal year to permit Holdings to pay (Ai) reasonable and customary corporate and operating expenses (including reasonable out-of-pocket expenses for legal, administrative and accounting services provided by third parties, and compensation, benefits and other amounts payable to officers and employees in connection with their employment in the ordinary course of business and to board of director observers), and (Bii) franchise fees or similar Taxes and fees required to maintain its corporate existence, and (C) its proportionate share of the tax liability of the affiliated groupof corporations that file consolidated Federal income tax returns determined as if the Borrower and its Subsidiarieshad filed a separate consolidated federal income tax return (or that file state and local income tax returns on a consolidated basis), in each case under <u>clauses (i)</u> and (ii) above, only to the extent such payments are actually made by Holdings: and (f) the Borrower may declare and make dividend payments or other distributions to Holdings in an aggregate amount of up to \$2,500,000 in order for Holdings to purchase, redeem or otherwise acquire itsoutstanding common Equity Interests, so long as either (A) (x)(i) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least 15.0% of the Aggregate Commitments and (ii) over the 30consecutive days prior to consummation of such action. Availability on a Pro Forma Basis for such action averaged no less than 15.0% of the Aggregate Commitments and (y) the Consolidated Fixed Charge Coverage Ratio would beat least 1.0 to 1.0 on a Pro Forma Basis for such action or (B) (i) Availability on a Pro Forma Basis immediatelyafter giving effect to such action would be at least 20.0% of the Aggregate Commitments and (ii) over the 30consecutive days prior to consummation of such action, Availability on a Pro Forma Basis for such action averaged no less than 20.0% of the Aggregate Commitments.

7.07 <u>Change in Nature of Business</u>.

Engage in any material line of business substantially different from those lines of business conducted by the Borrower and its Subsidiaries on the Closing Date or any business substantially related or incidental thereto.

7.08 <u>Transactions with Affiliates</u>.

Enter into any transaction of any kind with any Affiliate of the Borrower, whether or not in the ordinary course of business, other than <u>in compliance with the Approved Budget and</u> on fair and reasonable terms substantially as favorable to the Borrower or such Subsidiary as would be obtainable by the Borrower or such Subsidiary at the time in a comparable arm's length transaction with a Person other than an Affiliate; <u>provided</u> that the foregoing restriction shall not apply to (i) transactions between or among the Loan Parties (other than Holdings) and (ii) transactions contemplated by that certain offering memorandum, dated May 10, 2016, delivered in connection with the 2016 Unsecured Notes.

7.09 Burdensome Agreements.

Enter into or permit to exist any Contractual Obligation (other than this Agreement or any other Loan Document) that limits the ability (i) of any Subsidiary to make Restricted Payments to the Borrower or any Guarantor or to otherwise transfer property to or invest in the Borrower or any Guarantor, except for any agreement in effect (A) on the Closing Date and set forth on Schedule 7.09 (including the Term Documents, the Senior Notes-Indenture, the Senior Exchangeable Notes Indenture and the Second Lien Notes Documents), 7.09, (B) at the time any Subsidiary becomes a Subsidiary of the Borrower, so long as such agreement was not entered into solely in contemplation of such Person becoming a Subsidiary of the Borrower, (C) on the effective date of any refinancing, refunding, replacement or exchange of the Senior Notes pursuant to Section 7.02(m) and contained in the indenturesor other documents that evidence, or that are executed and delivered by the Loan Parties in connection with, suchrefinancing, refunding, replacement or exchange[reserved], (D) on the date of issuance of the 20146% Notes and contained in the indentures or other documents that evidence, or that are executed and delivered by the Loan Parties in connection with the 2014 Notes. (E) on the date of issuance of the 2016 Notes and contained in the indentures. note purchase agreements or other documents that evidence, or that are executed and delivered by the Loan Parties in connection with, the 2016 Notes or (F) on the effective date of any refinancing, refunding, replacement or exchange of the First Lien6% Notes pursuant to Section 7.02(g) and contained in the indentures or other documents that evidence, or that are executed and delivered by the Loan Parties in connection with, such refinancing, refunding, replacement or exchange, (ii) of any Subsidiary to Guarantee the Indebtedness of the Borrower or (iii) of the Borrower or any Subsidiary to create, incur, assume or suffer to exist Liens on property of such Person, except for (A) any agreement in effect on the Closing Date and set forth on <u>Schedule 7.09</u> (including the <u>DIP</u> Term Documents-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 135 of 432

and the <u>DIP</u> Term Intercreditor Agreement, the Intercreditor Agreement, the Senior Notes Indenture, the Senior-Exchangeable Notes Indenture and the Second Lien Notes Documents) or (B) the First Lien Intercreditor Agreement, the 2014<u>Senior/Junior</u> Intercreditor Agreement, the 2016 Intercreditor Agreement and the indentures, note purchase agreements or other documents that evidence, or that are executed and delivered by the Loan Parties in connection with, the 2014 Notes and the 2016 Notes; provided, however, that this clause (iii) shall not prohibit any negative pledge incurred or provided in favor of any holder of Indebtedness permitted under (x) <u>Section 7.02(f)</u> solely to the extent any such negative pledge relates to the property financed by or the subject of such Indebtedness or (y) <u>Sections 7.02(d)</u>, (i), (j), (k), (m), (n) and (q) so long as such negative pledge does not limit the ability of the Borrower or any Subsidiary to create, incur, assume or suffer to exist Liens on property of such Person to secure the Obligations.

7.10 <u>Use of Proceeds</u>.

Use the proceeds of any Borrowing, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock (within the meaning of Regulation U of the FRB) or to extend credit to others for the purpose of purchasing or carrying margin stock or to refund indebtedness originally incurred for such purpose.

7.11 <u>Financial Covenant.</u>

The Borrower and its Subsidiaries shall, on any date when Availability is less than the greater of (a) 10.0% of the Aggregate Commitments and (b) \$20,000,000 (the "FCCR Test Amount"), have a Consolidated Fixed Charge-Coverage Ratio of at least 1.0 to 1.0, tested for the four fiscal quarter period ending on the last day of the most-recently ended fiscal quarter for which the Borrower has delivered Section 6.01 Financials, and at the end of each-succeeding fiscal quarter thereafter until the date on which Availability has exceeded the FCCR Test Amount for 30-consecutive days.

7.11 7.12 [Reserved].

<u>7.12</u> [Reserved].

7.13 Amendments of Organization Documents.

Amend any of its Organization Documents in any material respect adverse to the Administrative Agent or the Lenders.

7.14 <u>Accounting Changes</u>.

Make any change in (a) accounting policies or reporting practices, except as required by GAAP or applicable Laws, or (b) its fiscal year, except in the case of this <u>clause (b)</u> to a fiscal year ending December 31 or otherwise upon 30 days prior written notice to the Administrative Agent.

7.15 <u>Prepayments, Etc. of Indebtedness</u>.

Prepay, redeem, purchase, defease or otherwise satisfy prior to the scheduled maturity thereof in any manner, or make any payment in violation of any subordination terms of, any Indebtedness, except: <u>(a) the</u> prepayment of the Revolving Loans in accordance with the terms of this Agreement, and (b) mandatory prepayments of the Indebtedness outstanding in respect of the DIP Term Credit Agreement made with Net Cash Proceeds (as such term is defined in the DIP Term Credit Agreement, as in effect on the Ratification Effective Date) received in respect of assets of the Loan Parties (other than assets constituting Revolver Priority Collateral), as required by Section 2.09(b)(ii) of the DIP Term Credit Agreement, as in effect on the Ratification Effective Date.

(i) the prepayment of the Revolving Loans in accordance with the terms of this Agreement and (ii) the repayment and prepayment of Indebtedness under the Second Amended and Restated Credit Agreement in accordance with the terms of the Term Intercreditor Agreement; provided that

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 136 of 432

the Borrower shall not make any voluntary prepayments under the Second Amended and Restated Credit Agreement (other than refinancings of the Second Amended and Restated Credit Agreement permittedunder Section 7.02) unless the Prepayment Conditions are satisfied on a Pro Forma Basis immediately aftergiving effect to such prepayment;

regularly scheduled or required repayments, purchases or redemptions of Indebtedness setforth in <u>Schedule 7.02</u> (except any such repayment, purchase or redemption subject to <u>Section 7.15(d)</u>below) and refinancings, refundings, replacements and exchanges of such Indebtedness and any Senior-Notes or First Lien Notes in compliance with <u>Section 7.02(d)</u>, <u>Section 7.02(m)</u> or <u>Section 7.02(q)</u>, asapplicable;

<u>prepayments</u>, redemptions, defeasances or satisfactions of Indebtedness permitted under <u>Section 7.02(g)</u>; provided that (i) such prepayments, redemptions, defeasances or satisfactions are made on, or within sixty (60) days after, the date on which the Person owing such Indebtedness becomes a Subsidiary of the Borrower, and (ii) the funds required to make such prepayment, redemption, defeasance or satisfaction of such Indebtedness are deposited with a trustee, escrow agent, paying agent or similar agent on the closing date of the Borrower's acquisition of such Person, and such funds reduce, or are otherwiseincluded as a portion of, the purchase price for such Person pursuant to the terms of the purchase and saleagreement governing such acquisition;

the prepayment, purchase, redemption, defeasance or satisfaction of any Indebtedness-(including, without limitation, the 2014 Notes, the Senior Notes and the 2016 Notes) so long as the Prepayment Conditions are satisfied on a Pro Forma Basis immediately after giving effect to such prepayment, purchase, redemption, defeasance or satisfaction;

——prepayments of Indebtedness permitted under <u>Section 7.02(b)</u>;

notwithstanding any other provision contained in this <u>Section 7.15</u> or elsewhere in this Agreement to the contrary, the prepayment, redemption, purchase or defeasance of all Indebtedness under the Term Documents and the Second Lien Notes Documents with the proceeds of the 2014 Notes, in each case with (for the avoidance of doubt) such Indebtedness being permanently retired;

the prepayment, purchase, redemption, defeasance or satisfaction of Indebtedness (i) with the proceeds received from the substantially concurrent issue of new common Equity Interests of Holdings or (ii) to the extent such prepayment, purchase, redemption, defeasance or satisfaction is consummated pursuant to the exchange of Indebtedness for common Equity Interests of Holdings;

the prepayment, purchase, redemption, defeasance or satisfaction of Indebtedness nototherwise permitted under this <u>Section 7.15</u> in an aggregate amount not to exceed \$10,000,000; *provided*that (i) Availability on a Pro Forma Basis immediately after giving effect to such action would be at least \$32,500,000, (ii) over the 30 consecutive days prior to consummation of such action, Availability averagedon a Pro Forma Basis for such action no less than \$32,500,000 and (iii) the Consolidated Fixed Charge-Coverage Ratio would be at least 1.0 to 1.0 on a Pro Forma Basis for such action;

<u>prior to the refinancing in full of the initial Unsecured Term Loan Facility with</u> Indebtedness incurred pursuant to <u>Section 7.02(m)</u>, the prepayment, purchase, redemption, exchange or defeasance of the initial Unsecured Term Loan Facility in an aggregate amount not to exceed the principalamount thereof so long as the Unsecured Term Loan Prepayment Conditions are satisfied on a Pro Forma-Basis immediately after giving effect to such prepayment, purchase, redemption, exchange or defeasance;

(j) notwithstanding any other provision contained in this <u>Section 7.15</u> or elsewhere in this Agreement to the contrary, the prepayment, redemption, purchase, defeasance or exchange of all or any portion of the Senior Notes for the 2016 Unsecured Notes, in each case, with (for the avoidance of doubt)-such prepaid, redeemed, purchased, defeased or exchanged Senior Notes being permanently retired; and

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 137 of 432

(k) prepayments, redemptions, defeasances or satisfactions of Indebtedness made with netproceeds from Dispositions applied in accordance with Section 4.07 of the First Lien Notes Indenture and permitted under Section 7.05(h).

Notwithstanding the foregoing clauses (a) through (i) of this <u>Section 7.15</u>, Holdings and the Borrower shall not, norshall they permit any Subsidiary to, directly or indirectly, make any payment in violation of the Intercreditor-Agreement with respect to any Indebtedness evidenced by the Second Lien Notes.

7.16 <u>Amendment, Etc. of Related Documents and Indebtedness</u>.

(a) Cancel-or terminate any Unsecured Term Loan Facility Documents or Second Lien Notes Documents orconsent to or accept any cancellation or termination thereof, except, in the case of the Unsecured Term Loan Facility, in connection with any transaction permitted under Section 7.15(i), and, in the case of the Second Lien Notes-Documents, in connection with any transaction permitted under Section 7.15(d) or Section 7.15(f), (b) cancel or terminate any Material Contract or consent to or accept any cancellation or termination thereof, unless the cancellation or termination thereof could not reasonably be expected to have a Material Adverse Effect, (e) amend, modify or change in any manner any term or condition of any Unsecured Term Loan Facility Documents or Second-Lien Notes Documents or give any consent, waiver or approval thereunder that, in any such case, could impair the rights and remedies of the Secured Parties under the Loan Documents or otherwise result in a Material Adverse-Effect. (db) amend, modify or change in any manner any term or condition of any Material Contract or give any consent, waiver or approval thereunder unless such amendment, modification or change could not reasonably be expected to have a Material Adverse Effect, (e) <u>Fintentionally omitted</u>, (fc) waive any material default or any breach of any material term or condition of any Material Contract, (gd) take any other action in connection with any Related Document that would impair in any material respect the value of the interest or rights of any Loan Party thereunder or that would impair the rights or interests of the Administrative Agent or any Lender or (he) amend, modify or change in any manner any term or condition of any Indebtedness set forth in Schedule 7.02, except for any refinancing, refunding, renewal or extension thereof permitted by Section 7.02(d), Section 7.02(m), Section 7.15(d)or Section 7.15(i) and except for any such amendment, modification or change that could not reasonably be expected to impair the rights and remedies of the Secured Parties under the Loan Documents or otherwise result in a Material Adverse Effect.

7.17 Holding Company.

In the case of Holdings, engage in any business or activity other than (a) the ownership of all outstanding Equity Interests in the Borrower and CNMW Investments, Inc., (b) maintaining its corporate existence, (c) participating in tax, accounting and other administrative activities as the parent of the consolidated group of companies, including the Loan Parties, (d) the execution and delivery of the Loan Documents, the Related-Documents to which it is a party and the Senior Notes Documents to which it is a party and the performance of its obligations thereunder, (e) activities incidental to the businesses or activities described in <u>clauses (a)</u> through (d) of this Section and (f) any other activities substantially similar or related to the lines of business of Holdings in the ordinary course of business as of the Closing Date to the extent otherwise permitted under the Loan Documents.

7.18 <u>Sanctions.</u>

Directly or indirectly, use the proceeds of any Borrowing, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other individual or entity, to fund any activities of or business with any individual or entity, or in any Designated Jurisdiction that, at the time of such funding, is the subject of Sanctions, or in any other manner that will result in a violation by any individual or entity (including any individual or entity participating in the transaction, whether as Lender, Lead Arranger, Administrative Agent or otherwise) of Sanctions.

ARTICLE VIII

EVENTS OF DEFAULT AND REMEDIES

8.01 <u>Events of Default</u>. Any of the following shall constitute an Event of Default:

(a) <u>Non-Payment</u>. The Borrower or any other Loan Party fails to (i) pay when and as required to be paid herein, any amount of principal of any Revolving Loan or any LC Obligation or deposit any funds as Cash Collateral in respect of LC Obligations, (ii) pay within three Business Days after the same becomes due, any interest on any Revolving Loan or any LC Obligation, or any fee due hereunder, or (iii) pay within five Business Days after the same becomes due, any other amount payable hereunder or under any other Loan Document; or

(b) Specific Covenants. (i) The Borrower fails to perform or observe any term, covenant or agreement contained in any of Section 6.01(a), 6.01(b), 6.02(b), 6.03, 6.05(a), 6.05(c), 6.10, 6.11, 6.12, 6.14, 6.20(c) or 6.21 or Article VII, (ii) any Guarantor fails to perform or observe any term, covenant or agreement set forth in the foregoing subclause (i) that such Guarantor has agreed to perform or observe pursuant to Section 4.01 of any Guaranty to which it is a party, (iii) any Loan Party fails to perform or observe any term, covenant or agreement contained in (xw) SectionSections 5.01(c), 5.01(d), Section 5.01(g), or Section 5.02 of the Security Agreement or the Canadian Security Agreement, as applicable, or (yx) the foregoing subclause (i) that any Loan Party agreed to perform or observe pursuant to any Mortgage or Canadian Mortgage to which it is a party, (y) any Financing Order or the Ratification Agreement or (z) any other material order of the Bankruptcy Court in any material respect, or (iv) the Borrower fails to deliver a Borrowing Base Certificate required to be delivered pursuant to Section 6.20(a) within five (5) Business Days of the date such Borrowing Base Certificate is required to be delivered (other than during an Increased Reporting Period, in which case such period shall be two (2) Business Days); or

(c) Other Defaults. Any Loan Party fails to perform or observe any other covenant or agreement (not specified in Section 8.01(a) or (b) above) contained in any Loan Document on its part to be performed or observed and such failure continues for 30 days from the earlier of (i) the date that any Responsible Officer of a Loan Party has actual knowledge thereof or (ii) the date that the Administrative Agent delivers to Borrower written notice of such failure; or

(d) <u>Representations and Warranties</u>. Any representation, warranty, certification or statement of fact made or deemed made by or on behalf of the Borrower or any other Loan Party herein, in any other Loan Document, or in any document delivered in connection herewith or therewith shall be incorrect or misleading in any material respect when made or deemed made; or

-Cross Default. (i) AnyCross-Default. Except with respect to payments that are (e) unenforceable as a result of the Chapter 11 Cases and defaults that occur solely as a result of the filing of the Chapter 11 Cases, (i) any Loan Party or any Subsidiary thereof (A) fails to make any payment when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) in respect of any Indebtedness or Guarantee (other than Indebtedness hereunder and Indebtedness under Swap Contracts) having an aggregate principal amount (including undrawn committed or available amounts and including amounts owing to all creditors under any combined or syndicated credit arrangement) of more than the Threshold Amount, or (B) fails to observe or perform any other agreement or condition relating to any such Indebtedness or Guarantee or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event occurs, the effect of which default or other event is to cause, or to permit the holder or holders of such Indebtedness or the beneficiary or beneficiaries of such Guarantee (or a trustee or agent on behalf of such holder or holders or beneficiary or beneficiaries) to cause, with the giving of notice if required, such Indebtedness to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made, prior to its stated maturity, or such Guarantee to become payable or cash collateral in respect thereof to be demanded; or (ii) there occurs under any Swap Contract an Early Termination Date (as defined in such Swap Contract) resulting from any event of default under such Swap Contract as to which a Loan Party or any Subsidiary thereof is the Defaulting Party (as defined in such Swap Contract) and the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 139 of 432

Swap Termination Value owed by such Loan Party or such Subsidiary as a result thereof is greater than the Threshold Amount; or

(f) Insolvency Proceedings, Etc. AnyExcept as a result of the Chapter 11 Cases, any Loan Party or any Subsidiary thereof institutes or consents to the institution of any proceeding under any Debtor Relief Law, or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, national receiver, interim receiver, trustee, custodian, conservator, liquidator, rehabilitator, monitor or similar officer for it or for all or any material part of its property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of such Person and the appointment continues undischarged or unstayed for 60 calendar days; or any proceeding under any Debtor Relief Law relating to any such Person or to all or any material part of its property is instituted without the consent of such Person and continues undismissed or unstayed for 60 calendar days, or an order for relief is entered in any such proceeding; or

(g) Inability to Pay Debts; Attachment. (i) Any Loan Party or any Subsidiary thereofbecomes unable or admits in writing its inability or fails generally to pay its debts as they become due, or (ii)Except as a result of the Chapter 11 Cases, any writ or warrant of attachment or execution or similar process is issued or levied against all or any material part of the property of any such Person and is not released, vacated or fully bonded within 30 days after its issue or levy; or

(h) Judgments. There is entered against any Loan Party or any Subsidiary thereof (i) one or more final judgments or orders for the payment of money in an aggregate amount (as to all such judgments and orders) exceeding the Threshold Amount <u>\$750,000</u> (to the extent not covered by independent third-party insurance as to which the insurer has an A.M. Best Financial Strength Rating of at least "A," been notified of the potential claim and does not dispute coverage), or (ii) any one or more non-monetary final judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of 30 consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect; or

(i) <u>ERISA.</u> (i)-An ERISA Event occurs<u>or is reasonably likely to occur</u>, with respect to a Pension Plan or Multiemployer Plan which has resulted or could reasonably be expected to result in liability of any Loan Party under Title IV of ERISA to the Pension Plan, Multiemployer Plan or the PBGC in an aggregate amountin excess of \$50,000,000, or (ii) any Loan Party or any ERISA Affiliate fails to pay when due, after the expiration of any applicable grace period, any installment payment with respect to its withdrawal liability under Section 4201 of ERISA under a Multiemployer Plan in an aggregate amount in excess of the Threshold Amount<u>a Material Adverse</u>.

(j) <u>Pension Event</u>. A Pension Event shall have occurred that when taken either alone or together with all other such Pension Events (i) could reasonably be expected to result in a Material Adverse Effect or (ii) causes any Lien to arise in respect of any Canadian Guarantor in connection with any Canadian Pension Plan (save for contribution amounts not yet due) which could reasonably be expected to secure assets of such Canadian Guarantor exceeding \$1,000,000 in value; or

(k) Invalidity of Loan Documents. Any material provision of any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect; or any Loan Party or any other Person contests in any manner the validity or enforceability of any provision of any Loan Document; or any Loan Party denies that it has any or further liability or obligation under any provision of any Loan Document, or purports to revoke, terminate or rescind any provision of any Loan Document; or

(<u>)</u> <u>Change of Control</u>. There occurs any Change of Control; or

(m) <u>Collateral Documents</u>. Any Collateral Document after delivery thereof pursuant to <u>Section 4.01</u> or <u>6.12 (subject to the Financing Orders)</u> shall for any reason (other than pursuant to the terms hereof of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 140 of 432

thereof) cease to create a valid and perfected first priority Lien (subject to Liens permitted by <u>Section 7.017.01, the</u> <u>DIP Term Intercreditor Agreement</u> and the <u>TermFirst Lien</u> Intercreditor Agreement) on the Collateral purported to be covered thereby; or

Subordination. (i) The Intercreditor Agreement or the Term Intercreditor Agreement shall, in whole or in part, terminate, cease to be effective or cease to be legally valid, binding and enforceable against any holder of the Second Lien Notes or Indebtedness under the Second Amended and Restated Credit Agreement, in each case except pursuant to the express terms thereof; (ii) all or any materialportion of the Obligations cease to constitute "First Lien Obligations" under the Second Lien Notes-Documents; (iii) the Borrower or any other Loan Party shall, directly or indirectly, disavow or contest inany manner (A) the effectiveness, validity or enforceability of the Intercreditor Agreement or the Term-Intercreditor Agreement, (B) that the Intercreditor Agreement or the Term Intercreditor Agreement existsfor the benefit of the Administrative Agent, the Lenders and the Issuing Bank or (C) limitations set forth inthe Intercreditor Agreement or the Term Intercreditor Agreement upon application of proceeds from anysource to payment of principal of, or premium or interest on, the Second Lien Notes or other junior lien-Indebtedness or the Indebtedness under the Second Amended and Restated Credit Agreement; (iv) any holders of the Second Lien Notes or other junior lien Indebtedness shall, directly or indirectly, disavow orcontest in any manner the effectiveness, validity or enforceability of any material term of the Intercreditor-Agreement; or (v) any holder of Indebtedness under the Second Amended and Restated Credit Agreementshall, directly or indirectly, disavow or contest in any manner the effectiveness, validity or enforceability of any material term of the Term Intercreditor Agreement:

provided, however, that upon the issuance of the 2014 Notes and the execution and delivery of the First Lien Intercreditor Agreement and the 2014 Intercreditor Agreement by the parties thereto, this <u>Section</u>. <u>8.01(n)</u> shall be of no further force or effect

(n) [reserved]; or

-Subordination. At any time following the issuance of the 2014 Notes and the execution- $(\mathbf{0})$ (θ) the First Lien Intercreditor Agreement and the 2014 Intercreditor Agreement by the parties thereto, and delivery (i) the <u>DIP Term Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014 Senior/Junior</u> Intercreditor Agreement shall, in whole or in part, terminate, cease to be effective or cease to be legally valid, binding and enforceable against any holderholders of the First LienDIP Term Loans, the 6% Notes or the 2014-Second Lien8.5% Notes, in each case except pursuant to the express terms thereof; (ii) all or any material portion of the Obligations cease to constitute "Senior Priority Obligations" (or any comparable defined term) under the 2014-Second Lien8.5% Notes Indenture; (iii) the Borrower or any other Loan Party shall, directly or indirectly, disavow or contest in any manner (A) the effectiveness, validity or enforceability of the DIP Term Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement, (B) that the DIP Term. Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014 Senior/Junior Intercreditor Agreement exists for the benefit of the Administrative Agent, the Lenders and the Issuing Bank or (C) limitations set forth in the DIP Term Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement upon application of proceeds from any source to payment of principal of, or premium or interest on, the 2014 Second Lien8.5% Notes or other junior lien Indebtedness or the Indebtedness under the First Lien6% Notes; (iv) any holders of the 2014 Second Lien8.5% Notes or other junior lien Indebtedness shall, directly or indirectly, disavow or contest in any manner the effectiveness, validity or enforceability of any material term of the <u>2014Senior/Junior</u> Intercreditor Agreement; or (v) any holder of the First Lien6% Notes shall, directly or indirectly, disavow or contest in any manner the effectiveness, validity or enforceability of any material term of the First Lien Intercreditor Agreement; or (p) v) any holder of the DIP Term Loans shall, directly or indirectly, disavow or contest in any manner the effectiveness, validity or enforceability of any material term of the DIP Term Intercreditor Agreement: or

(p) At any time following the issuance of the $\frac{2016 \text{ Secured 4\%}}{2016 \text{ Intercreditor Agreement shall, in whole or in part, terminate, cease to be effective or cease to be legally valid, binding and enforceable against any holder of the <math>\frac{2016 \text{ Secured 4\%}}{2016 \text{ Secured 4\%}}$ Notes, in each case except pursuant to the express terms thereof; (ii) all or any material portion of the Obligations cease to constitute "Senior Priority Obligations" (or any comparable defined term) under the 2016

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 141 of 432

Intercreditor Agreement; (iii) the Borrower or any other Loan Party shall, directly or indirectly, disavow or contest in any manner (A) the effectiveness, validity or enforceability of the 2016 Intercreditor Agreement, (B) that the 2016 Intercreditor Agreement exists for the benefit of the Administrative Agent, the Lenders and the Issuing Bank or (C) limitations set forth in the 2016 Intercreditor Agreement upon application of proceeds from any source to payment of principal of, or premium or interest on, the 2016 Secured4% Notes; or (iv) any holders of the 2016 Secured4% Notes shall, directly or indirectly, disavow or contest in any manner the effectiveness, validity or enforceability of any material term of the 2016 Intercreditor Agreement; or

(q) <u>Conversion of any Chapter 11 Case to a Chapter 7 case under the Bankruptcy Code; or</u>

(r) Dismissal of any Chapter 11 Case or any subsequent Chapter 7 case either voluntarily or involuntarily; or Any Financing Order shall be revoked, remanded, vacated, reversed, stayed or rescinded or modified (other than, in the case of a modification as consented to by Administrative Agent and Required Lenders); or

(s) Appointment of a trustee, examiner or disinterested person with expanded powers relating to the operations or the business of any Loan Party in the Chapter 11 Cases; or

(t) Any administrative expense claim is allowed having priority or ranking in parity with the rights of Administrative Agent; or

(u) Any plan of reorganization is confirmed that does not provide for the indefeasible payment in full in cash of the Obligations on terms and conditions acceptable to the Administrative Agent and Lenders, including customary releases and exculpation agreements by the Loan Parties in favor of the Administrative Agent and Lenders; or

 (\underline{v}) Payment of or granting adequate protection with respect to any of the existing secured debt of Borrower, other than to the extent permitted, and in a manner consistent with, the existing intercreditor agreement applicable thereto, as approved by the Bankruptcy Court; or

(w) Liens or super-priority claims with respect to the Loan Documents shall at any time cease to be valid, perfected and enforceable in all respects with the priority described herein; or

(x) The Loan Parties' exclusive period to file a plan of reorganization and disclosure statement shall have terminated without the Loan Parties having filed a plan of reorganization that provides for the indefeasible payment in full in cash of the Obligations on terms and conditions acceptable to the Administrative Agent and Lenders, including customary releases and exculpation agreements by the Loan Parties in favor of the Administrative Agent and Lenders; or

(y) The failure to comply with any of the Milestones, except to the extent such Milestone is extended to a later date with the prior consent of the Administrative Agent and Required Lenders.

8.02 <u>Remedies upon Event of Default</u>.

If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:

(a) declare the Aggregate Commitments and the LC Commitment to be terminated, whereupon such commitments shall be terminated;

(b) declare the unpaid principal amount of all outstanding Revolving Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 142 of 432

(c) subject to the <u>TermDIP Term Intercreditor Agreement and the First Lien</u> Intercreditor Agreement, exercise on behalf of itself, the Lenders and the Issuing Bank all rights and remedies available to it and the Lenders under the Loan Documents or applicable Law;

(d) terminate, reduce or condition any Revolving Commitment, or make any adjustment to the Borrowing Base; and

(e) require the Loan Parties to Cash Collateralize the LC Obligations, and, if the Loan Parties fail promptly to deposit such Cash Collateral, the Administrative Agent may (and shall upon the direction of the Required Lenders) advance the required Cash Collateral as Revolving Loans (whether or not an Overadvance exists or is created thereby, or the conditions in Section 4.02 are satisfied);

provided, however, that upon the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under the Bankruptcy Code of the United States, the obligation of each Lender to make Revolving Loans and any obligation of the Issuing Bank to issue, amend, extend or renew Letters of Credit shall automatically terminate, the unpaid principal amount of all outstanding Revolving Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of the Loan Parties to Cash Collateralize the LC Obligations as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender.

8.03 <u>Application of Funds</u>.

After the exercise of remedies provided for in <u>Section 8.02</u> (or after the Revolving Loans have automatically become immediately due and payable and the LC Exposure has automatically been required to be Cash Collateralized as set forth above), any amounts received on account of the Obligations (including without limitation, proceeds received by the Administrative Agent in respect of any sale of, collection from, or other realization upon, all or any part of the Collateral (including, without limitation, pursuant to the exercise by the Administrative Agent of its remedies during the continuance of an Event of Default) or otherwise received on account of the Obligations) shall, subject to the provisions of <u>Sections 2.11</u> and 2.13(j), be applied in the following order:

<u>First</u>, to the payment of all reasonable costs and out-of-pocket expenses, fees, commissions and taxes of such sale, collection or other realization including, without limitation, compensation to the Administrative Agent and its agents and counsel payable pursuant to the Loan Documents, and all expenses, liabilities and advances made or incurred by the Administrative Agent in connection therewith;

Second, to the payment of all other reasonable costs and out-of-pocket expenses of such sale, collection or other realization including, without limitation, costs and expenses incurred by the other Secured Parties (other than in respect of Secured Bank Product Obligations) and all expenses, liabilities and advances made or incurred by the other Secured Parties in connection therewith (other than in respect of Secured Bank Product Obligations);

Third, to interest then due and payable on the Borrower's Swingline Loans;

<u>Fourth</u>, to the principal balance of the Swingline Loans outstanding until the same has been prepaid in full;

<u>Fifth</u>, to interest then due and payable on Revolving Loans and other amounts due pursuant to <u>Sections 3.02, 3.03, 3.04</u> and <u>3.05</u>;

Sixth, to Cash Collateralize all LC Exposures (to the extent not otherwise Cash Collateralized pursuant to the terms hereof) plus any accrued and unpaid interest thereon;

Seventh, to the principal balance of Revolving Borrowings then outstanding and all Obligations on account of Noticed Hedges with Secured Parties, *pro rata*;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 143 of 432

Eighth, to all other Obligations pro rata; and

<u>Ninth</u>, the balance, if any, as required by the <u>Term DIP Term Intercreditor Agreement and the First</u>. <u>Lien</u> Intercreditor Agreement or, in the absence of any such requirement, to the Person lawfully entitled thereto (including the applicable Loan Party or its successors or assigns).

Notwithstanding anything to the contrary contained in this Agreement or in any of the other Loan Documents, Administrative Agent may, in its discretion, apply such payments or proceeds first, to the Pre-Petition Obligations until such Pre-Petition Obligations are paid and satisfied in full and second to the Post-Petition Obligations until such Post-Petition Obligations are paid and satisfied in full.

Amounts used to Cash Collateralize the aggregate undrawn amount of Letters of Credit pursuant to clause <u>Sixth</u> above shall be applied to satisfy drawings under such Letters of Credit as they occur. If any amount remains on deposit as Cash Collateral after all Letters of Credit have either been fully drawn or expired, such remaining amount shall be applied to the other Obligations, if any, in the order set forth above. Amounts distributed with respect to any Secured Bank Product Obligations shall be the lesser of the maximum Secured Bank Product Obligations last reported to the Administrative Agent or the actual Secured Bank Product Obligations as calculated by the methodology reported to the Administrative Agent for determining the amount due. The Administrative Agent shall have no obligation to calculate the amount to be distributed with respect to any Secured Bank Product Obligations, and may request a reasonably detailed calculation of such amount from the applicable Secured Party. If a Secured Party fails to deliver such calculation within five days following request by the Administrative Agent, the Administrative Agent may assume the amount to be distributed is zero.

In the event that any such proceeds are insufficient to pay in full the items described in clauses <u>First</u> through <u>Eighth</u> of this <u>Section 8.03</u>, the Loan Parties shall remain liable for any deficiency. Notwithstanding the foregoing provisions, this <u>Section 8.03</u> is subject to the provisions of the <u>DIP</u> Term Intercreditor Agreement and the First Lien Intercreditor Agreement.

ARTICLE IX

ADMINISTRATIVE AGENT

9.01 Appointment and Authority.

(a) Each of the Lenders and the Issuing Bank hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the Issuing Bank, and none of Holdings, the Borrower nor any other Loan Party shall have rights as a third party beneficiary of any of such provisions.

(b) Each of the Lenders (including in its capacity as a Secured Bank Product Provider) hereby further authorizes the Administrative Agent to enter into the <u>DIP Term Intercreditor Agreement, the 2016</u> Intercreditor Agreement, the First Lien Intercreditor Agreement and the <u>2014Senior/Junior</u> Intercreditor Agreement, and any respective amendments thereto on behalf of such Lender. Without limiting the generality of the foregoing, each of the Lenders hereby authorizes and directs the Administrative Agent to bind each Lender to the actions required by such Lender under the terms of the <u>DIP Term Intercreditor Agreement, the 2016</u> Intercreditor Agreement, the First Lien Intercreditor Agreement and the <u>2014Senior/Junior</u> Intercreditor Agreement, the First Lien Intercreditor Agreement and the <u>2014Senior/Junior</u> Intercreditor Agreement, the First Lien Intercreditor Agreement and the <u>2014Senior/Junior</u> Intercreditor Agreement and any respective amendments thereto.

(c) The Administrative Agent shall also act as the "collateral agent" under the Loan Documents, and each of the Lenders (in its capacities as a Lender, Swingline Lender (if applicable), the Issuing Bank and potential Secured Bank Products Provider) hereby irrevocably appoints and authorizes the Administrative Agent to act as the agent of such Lender and the Issuing Bank for purposes of acquiring, holding and enforcing any and all Liens on

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 144 of 432

Collateral granted by any of the Loan Parties to secure any of the Obligations, together with such powers and discretion as are reasonably incidental thereto. In this connection, the Administrative Agent, as "collateral agent" and any co-agents, sub-agents and attorneys-in-fact appointed by the Administrative Agent pursuant to <u>Section 9.05</u> for purposes of holding or enforcing any Lien on the Collateral (or any portion thereof) granted under the Collateral Documents, or for exercising any rights and remedies thereunder at the direction of the Administrative Agent, shall be entitled to the benefits of all provisions of this <u>Article IX</u> and <u>Article XI</u> (including <u>Section 11.04(c)</u>, as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto.

(d) <u>Québec Liens</u>.

Without in any way limiting any of the foregoing provisions of this Section 9.01:

Solidary Interests. Should the Administrative Agent choose to create Liens (hypothecs) pursuant to the Civil Code of Quebec through the *solidarité* structure, then for the purposes of creating a solidarité active in accordance with Article 1541 of the Civil Code of Quebec between each Secured Party, taken individually, on the one hand, and the Administrative Agent, on the other hand, each Loan Party granting a Lien (hypothec) to the Administrative Agent under the Civil Code of Quebec and each such Secured Party acknowledges and agrees with the Administrative Agent that such Secured Party and the Administrative Agent are hereby conferred the legal status of solidary creditors of each such Loan Party in respect of all Indebtedness, liabilities and other obligations, present and future, owed by each such Loan Party to the Administrative Agent and such Secured Party hereunder and under the other Loan Documents (collectively, the "Solidary Claim") and that, accordingly, but subject (for the avoidance of doubt) to Article 1542 of the Civil Code of Quebec, each such Loan Party is irrevocably bound towards the Administrative Agent and each Secured Party in respect of the entire Solidary Claim of the Administrative Agent and such Secured Party. As a result of the foregoing, the parties hereto acknowledge that the Administrative Agent and each Secured Party shall at all times have a valid and effective right of action for the entire Solidary Claim of the Administrative Agent and such Secured Party and the right to give full acquittance for it. Accordingly, and without limiting the generality of the foregoing, the Administrative Agent, as solidary creditor with each Secured Party, shall at all times have a valid and effective right of action in respect of the Solidary Claim and the right to give a full acquittance for same. By its execution of the Loan Documents to which it is a party, each such Loan Party not a party hereto shall also be deemed to have accepted the stipulations hereinabove provided. The parties further agree and acknowledge that such Liens (hypothecs) under the Loan Documents shall be granted to the Administrative Agent, for its own benefit and for the benefit of Secured Parties, as solidary creditor as hereinabove set forth.

(ii) *Fondé de Pouvoir*. Should the Administrative Agent choose to create Liens (hypothecs) pursuant to the Civil Code of Quebec through the *fondé de pouvoir* structure, then for the purposes of holding any Liens (hypothecs) that secure the payment of any bond (or similar instrument), granted by any Loan Party pursuant to any Quebec law governed Collateral Documents, the Secured Parties hereby acknowledge that the Administrative Agent shall be and act as the Person holding the power of attorney of all present and future Secured Parties for all purposes of Article 2692 of the Civil Code of Québec, and, more specifically, all present and future holders of bonds or similar instruments. Each Secured Party therefore appoints, to the extent necessary, the Administrative Agent as its irrevocable fondé de pouvoir to hold the Liens created pursuant to such Quebec law governed Collateral Documents in order to secure the payment of any bonds or similar instruments. By executing an Assignment and Acceptance, each future Secured Party shall be deemed to ratify the power of attorney granted to the Administrative Agent hereunder. Notwithstanding the provisions of Section 32 of An Act Respecting the Special Powers of Legal Persons (Québec), the Administrative Agent may acquire bonds and similar instruments. Finally, each Loan Party acknowledges that any bond or other similar instrument executed by it shall constitute a "title of indebtedness" as such expression is defined in Article 2692 of the Civil Code of Quebec. The Administrative Agent accepts to act as *fondé de pouvoir* of the Secured Parties.

9.02 <u>Rights as a Lender</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 145 of 432

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

9.03 <u>Exculpatory Provisions</u>.

The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:

(a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

(b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), provided that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable Law; and

(c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in <u>Sections 11.01</u> and <u>8.02</u>) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to the Administrative Agent by the Borrower, a Lender or the Issuing Bank.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Collateral Documents, (v) the value or the sufficiency of any Collateral, or (v) the satisfaction of any condition set forth in <u>Article IV</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

9.04 <u>Reliance by Administrative Agent.</u>

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Revolving Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a

Lender or the Issuing Bank, the Administrative Agent may presume that such condition is satisfactory to such Lender or the Issuing Bank unless the Administrative Agent shall have received notice to the contrary from such Lender or the Issuing Bank prior to the making of such Revolving Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

9.05 <u>Delegation of Duties</u>.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

9.06 <u>Resignation of Administrative Agent.</u>

The Administrative Agent may at any time give notice of its resignation to the Lenders, the Issuing Bank and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower (and, so long as no Event of Default has occurred and is continuing, with the prior written consent of the Borrower, which consent shall not be unreasonably withheld or delayed), to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders or the Issuing Bank, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by the Administrative Agent on behalf of the Lenders or the Issuing Bank under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and the Issuing Bank directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 11.04 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as Issuing Bank and Swingline Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (i) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuing Bank and Swingline Lender, (ii) the retiring Issuing Bank and Swingline Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (iii) the successor Issuing Bank shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuing Bank to effectively assume the obligations of the retiring Issuing Bank with respect to such Letters of Credit.

9.07 <u>Non-Reliance on Administrative Agent and Other Lenders.</u>

Each Lender and the Issuing Bank acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender and the Issuing Bank also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

9.08 <u>No Other Duties, Etc.</u>

Anything herein to the contrary notwithstanding, none of the Lead Arrangers, and Syndication-Agents or-Documentation Agents listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent, a Lender or the Issuing Bank hereunder.

9.09 Administrative Agent May File Proofs of Claim.

In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to any Loan Party, the Administrative Agent (irrespective of whether the principal of any Revolving Loan or LC Exposure shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:

(a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Revolving Loans, LC Exposure and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the Issuing Bank and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the Issuing Bank and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the Issuing Bank and the Administrative Agent under <u>Sections 2.05</u> and <u>11.04</u>) allowed in such judicial proceeding; and

(b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same; and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender and the Issuing Bank to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Lenders and the Issuing Bank, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.05 and 11.04.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender or the Issuing Bank any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or the Issuing Bank to authorize the Administrative Agent to vote in respect of the claim of any Lender or the Issuing Bank in any such proceeding.

9.10 <u>Collateral and Guaranty Matters</u>.

The Lenders and the Issuing Bank irrevocably authorize the Administrative Agent, at its option and in its discretion,

(a) to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (i) upon termination of the Aggregate Commitments and payment in full of all Obligations (other than (x)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 148 of 432

obligations under Secured Bank Product Obligations and (y) contingent indemnification obligations) and the expiration or termination of all Letters of Credit (unless Cash Collateralized or backstopped on terms reasonably satisfactory to the Administrative Agent), (ii) that is Disposed of or to be Disposed of as part of or in connection with any Disposition permitted hereunder or under any other Loan Document, or (iii) if approved, authorized or ratified in writing in accordance with <u>Section 11.01</u>;

(b) to release any Guarantor from its obligations under the Guaranty if such Person ceases to be a Subsidiary as a result of a transaction permitted hereunder;

(c) to subordinate any Lien on any property granted to or held by the Administrative Agent under any Loan Document to the holder of any Lien on such property that is permitted by Section 7.01(i); and

(d) to enter into the <u>DIP Term Intercreditor Agreement, the</u> 2016 Intercreditor Agreement, the 2014 <u>Senior/Junior</u> Intercreditor Agreement and the First Lien Intercreditor Agreement and any respective amendments thereto.

Upon request by the Administrative Agent at any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guaranty pursuant to this <u>Section 9.10</u>. In each case as specified in this <u>Section 9.10</u>, the Administrative Agent will, at the Borrower's expense, execute and deliver to the applicable Loan Party such documents as such Loan Party may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Collateral Documents or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guaranty, in each case in accordance with the terms of the Loan Documents and this <u>Section 9.10</u>.

9.11 Bank Product Providers.

Each Secured Bank Product Provider, by delivery of a notice to the Administrative Agent of such agreement, agrees to be bound by this <u>Article IX</u>. Each such Secured Bank Product Provider shall indemnify and hold harmless Agent-Related Persons, to the extent not reimbursed by the Loan Parties, against all claims that may be incurred by or asserted against any Agent-Related Person in connection with such provider's Secured Bank Product Obligations.

9.12 <u>Tax Indemnity</u>.

To the extent required by any applicable Laws (as determined in good faith by the Administrative Agent). the Administrative Agent may withhold from any payment to any Lender under any Loan Document an amount equivalent to any applicable withholding Tax. Without limiting or expanding the provisions of Section 3.01, each Lender shall indemnify and hold harmless the Administrative Agent against, and shall make payable in respect thereof within 10 days after demand therefor, any and all Taxes and any and all related losses, claims, liabilities and expenses (including fees, charges and disbursements of any counsel for the Administrative Agent) incurred by or asserted against the Administrative Agent by the IRS or any other Governmental Authority as a result of the failure of the Administrative Agent to properly withhold Tax from amounts paid to or for the account of such Lender for any reason (including because the appropriate form was not delivered or not properly executed, or because such Lender failed to notify the Administrative Agent of a change in circumstance that rendered the exemption from or reduction of withholding Tax ineffective), whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document against any amount due the Administrative Agent under this Section 9.12. The agreements in this Section 9.12 shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender and the repayment, satisfaction or discharge of all other Obligations.

ARTICLE X

CONTINUING GUARANTY

10.01 <u>Guaranty</u>.

Holdings hereby absolutely and unconditionally guarantees, as a guaranty of payment and performance and not merely as a guaranty of collection, prompt payment when due, whether at stated maturity, by required prepayment, upon acceleration, demand or otherwise, and at all times thereafter, of any and all of the Obligations, whether for principal, interest, premiums, fees, indemnities, damages, costs, expenses or otherwise, of the Borrower to the Secured Parties, arising hereunder and under the other Loan Documents (including all renewals, extensions, amendments, refinancings and other modifications thereof and all costs, reasonable attorneys' fees and expenses incurred by the Secured Parties in connection with the collection or enforcement thereof), excluding, with respect to any Guarantor at any time, Excluded Swap Obligations with respect to such Guarantor at such time. The Administrative Agent's books and records showing the amount of the Obligations shall be admissible in evidence in any action or proceeding, and shall be binding upon Holdings, and conclusive for the purpose of establishing the amount of the Obligations, absent manifest error. This Guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Obligations or any instrument or agreement evidencing any Obligations, or by the existence, validity, enforceability, perfection, non-perfection or extent of any collateral therefor, or by any fact or circumstance relating to the Obligations which might otherwise constitute a defense to the obligations of Holdings under this Guaranty, and Holdings hereby irrevocably waives any defenses it may now have or hereafter acquire in any way relating to any or all of the foregoing, whether arising as a result of any law or regulation of any jurisdiction or any other event affecting any term of the Obligations.

10.02 <u>Rights of Lenders</u>.

Holdings consents and agrees that the Secured Parties may, at any time and from time to time, without notice or demand, and without affecting the enforceability or continuing effectiveness hereof: (a) amend, extend, renew, compromise, discharge, accelerate or otherwise change the time for payment or the terms of the Obligations or any part thereof; (b) take, hold, exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any security for the payment of this Guaranty or any Obligations; (c) apply such security and direct the order or manner of sale thereof as the Administrative Agent, the Issuing Bank and the Lenders in their sole discretion may determine; and (d) release or substitute one or more of any endorsers or other guarantors of any of the Obligations. Without limiting the generality of the foregoing, Holdings consents to the taking of, or failure to take, any action which might in any manner or to any extent vary the risks of Holdings under this Guaranty or which, but for this provision, might operate as a discharge of Holdings.

10.03 <u>Certain Waivers</u>.

Holdings waives (a) any defense arising by reason of any disability or other defense of the Borrower or any other guarantor, or the cessation from any cause whatsoever (including any act or omission of any Secured Party) of the liability of the Borrower; (b) any defense based on any claim that Holdings' obligations exceed or are more burdensome than those of the Borrower; (c) the benefit of any statute of limitations affecting Holdings' liability hereunder; (d) subject to Section 10.05, any right to proceed against the Borrower, proceed against or exhaust any security for the Obligations, or pursue any other remedy in the power of any Secured Party whatsoever; (e) subject to Section 10.05, any benefit of and any right to participate in any security now or hereafter held by any Secured Party; and (f) to the fullest extent permitted by law, any and all other defenses or benefits that may be derived from or afforded by applicable Law limiting the liability of or exonerating guarantors or sureties. Holdings expressly waives all setoffs and counterclaims and all presentments, demands for payment or performance, notices of nonpayment or nonperformance, protests, notices of protest, notices of dishonor and all other notices or demands of any kind or nature whatsoever with respect to the Obligations, and all notices of acceptance of this Guaranty or of the existence, creation or incurrence of new or additional Obligations. Holdings waives any rights and defenses that are or may become available to Holdings by reason of §§ 2787 to 2855, inclusive, and §§ 2899 and 3433 of the California Civil Code. As provided below, this Guaranty shall be governed by, and construed in accordance with, the laws of the State of New York. The foregoing waivers and the provisions hereinafter set forth in this Guaranty which pertain to

California law are included solely out of an abundance of caution, and shall not be construed to mean that any of the above-referenced provisions of California law are in any way applicable to this Guaranty or the Obligations.

10.04 Obligations Independent.

The obligations of Holdings hereunder are those of primary obligor, and not merely as surety, and are independent of the Obligations and the obligations of any other guarantor, and a separate action may be brought against Holdings to enforce this Guaranty whether or not the Borrower or any other person or entity is joined as a party.

10.05 <u>Subrogation</u>.

Holdings shall not exercise any right of subrogation, contribution, indemnity, reimbursement or similar rights with respect to any payments it makes under this Guaranty until all of the Obligations and any amounts payable under this Guaranty have been paid in full in cash and performed in full and the Aggregate Commitments and this Agreement are terminated. If any amounts are paid to Holdings in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of the Secured Parties and shall forthwith be paid to the Secured Parties to reduce the amount of the Obligations, whether matured or unmatured.

10.06 <u>Termination; Reinstatement</u>.

This Guaranty is a continuing and irrevocable guaranty of all Obligations now or hereafter existing and shall remain in full force and effect until all Obligations and any other amounts payable under this Guaranty are paid in full in cash and the Aggregate Commitments and this Agreement are terminated. Notwithstanding the foregoing, this Guaranty shall continue in full force and effect or be revived, as the case may be, if any payment by or on behalf of the Borrower or Holdings is made, or any of the Secured Parties exercises its right of setoff, in respect of the Obligations and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by any of the Secured Parties in their discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Laws or otherwise, all as if such payment had not been made or such setoff had not occurred and whether or not the Secured Parties are in possession of or have released this Guaranty and regardless of any prior revocation, rescission, termination or reduction. The obligations of Holdings under this paragraph shall survive termination of this Guaranty.

10.07 <u>Subordination</u>.

Holdings hereby subordinates the payment of all obligations and indebtedness of the Borrower owing to Holdings, whether now existing or hereafter arising, including but not limited to any obligation of the Borrower to Holdings as subrogee of the Secured Parties or resulting from Holdings' performance under this Guaranty, to the payment in full in cash of all Obligations. If the Secured Parties so request, any such obligation or indebtedness of the Borrower to Holdings shall be enforced and performance received by Holdings as trustee for the Secured Parties and the proceeds thereof shall be paid over to the Secured Parties on account of the Obligations, but without reducing or affecting in any manner the liability of Holdings under this Guaranty.

10.08 Stay of Acceleration.

If acceleration of the time for payment of any of the Obligations is stayed, in connection with any case commenced by or against Holdings or the Borrower under any Debtor Relief Laws, or otherwise, all such amounts shall nonetheless be payable by Holdings immediately upon demand by the Secured Parties.

10.09 <u>Condition of Borrower</u>.

Holdings acknowledges and agrees that it has the sole responsibility for, and has adequate means of, obtaining from the Borrower and any other guarantor such information concerning the financial condition, business and operations of the Borrower and any such other guarantor as Holdings requires, and that none of the Secured

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 151 of 432

Parties has any duty, and Holdings is not relying on the Secured Parties at any time, to disclose to Holdings any information relating to the business, operations or financial condition of the Borrower or any other guarantor (Holdings waiving any duty on the part of the Secured Parties to disclose such information and any defense relating to the failure to provide the same).

10.10 Additional Guarantor Waivers and Agreements.

Holdings understands and acknowledges that if the Secured Parties foreclose judicially or (a) nonjudicially against any real property security for the Obligations, that foreclosure could impair or destroy any ability that Holdings may have to seek reimbursement, contribution, or indemnification from the Borrower or others based on any right Holdings may have of subrogation, reimbursement, contribution, or indemnification for any amounts paid by Holdings under this Guaranty. Holdings further understands and acknowledges that in the absence of this paragraph, such potential impairment or destruction of Holdings' rights, if any, may entitle Holdings to assert a defense to this Guaranty based on Section 580d of the California Code of Civil Procedure as interpreted in Union Bank v. Gradsky, 265 Cal. App. 2d 40 (1968). By executing this Guaranty, Holdings freely, irrevocably, and unconditionally: (i) waives and relinquishes that defense and agrees that Holdings will be fully liable under this Guaranty even though the Secured Parties may foreclose, either by judicial foreclosure or by exercise of power of sale, any deed of trust securing the Obligations; (ii) agrees that Holdings will not assert that defense in any action or proceeding which the Secured Parties may commence to enforce this Guaranty; (iii) acknowledges and agrees that the rights and defenses waived by Holdings in this Guaranty include any right or defense that Holdings may have or be entitled to assert based upon or arising out of any one or more of §§ 580a, 580b, 580d, or 726 of the California Code of Civil Procedure or § 2848 of the California Civil Code; and (iv) acknowledges and agrees that the Secured Parties are relying on this waiver in creating the Obligations, and that this waiver is a material part of the consideration which the Secured Parties are receiving for creating the Obligations.

(b) Holdings waives all rights and defenses that Holdings may have because any of the Obligations is secured by real property. This means, among other things, (i) the Secured Parties may collect from Holdings without first foreclosing on any real or personal property collateral pledged by the other Loan Parties; and (ii) if the Secured Parties foreclose on any real property collateral pledged by the other Loan Parties: (A) the amount of the Obligations may be reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is worth more than the sale price, and (B) the Secured Parties may collect from Holdings may have to collect from the Borrower. This is an unconditional and irrevocable waiver of any rights and defenses Holdings may have because any of the Obligations is secured by real property. These rights and defenses include, but are not limited to, any rights or defenses based upon § 580a, 580b, 580d, or 726 of the California Code of Civil Procedure.

(c) Holdings waives any right or defense it may have at law or equity, including California Code of Civil Procedure § 580a, to a fair market value hearing or action to determine a deficiency judgment after a foreclosure.

10.11 <u>Keepwell</u>.

Each Loan Party that is a Qualified ECP Guarantor at the time the Guaranty or the grant of the security interest under the Loan Documents, in each case, by any Specified Loan Party, becomes effective with respect to any Swap Obligation, hereby jointly and severally, absolutely, unconditionally and irrevocably undertakes to provide such funds or other support to each Specified Loan Party with respect to such Swap Obligation as may be needed by such Specified Loan Party from time to time to honor all of its obligations under its Guaranty and the other Loan Documents in respect of such Swap Obligation (but, in each case, only up to the maximum amount of such liability that can be hereby incurred without rendering such Qualified ECP Guarantor's obligations and undertakings under this <u>Article X</u> voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations have been paid and performed in full. Each Qualified ECP Guarantor intends this Section to constitute, and this Section shall be deemed to constitute, a guarantee of the obligations of, and a "keepwell, support, or other agreement" for the benefit of, each Specified Loan Party for all purposes of the Commodity Exchange Act.

ARTICLE XI

MISCELLANEOUS

11.01 <u>Amendments, Etc</u>.

(a) <u>Required Lender and Unanimous Consent</u>. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by the Required Lenders and the Borrower or the applicable Loan Party, as the case may be, and acknowledged by the Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; <u>provided</u>, <u>however</u>, that no such amendment, waiver or consent shall:

(i) [reserved];

(ii) [reserved];

(iii) subject to <u>Section 2.19</u>, extend or increase the Revolving Commitment of any Lender (or reinstate any Revolving Commitment terminated pursuant to <u>Section 8.02</u>) without the written consent of such Lender (and the Issuing Bank, if applicable);

(iv) subject to <u>Section 2.19</u>, postpone any date fixed by this Agreement or any other Loan Document for any payment (excluding mandatory prepayments) of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under such other Loan Document or modify the definition of "Maturity Date" (or waive any provision thereof) without the written consent of each Lender directly and adversely affected thereby;

(v) reduce the principal of, or the rate of interest specified herein on, any Revolving Loan or LC Disbursement, or (subject to <u>clause (D)</u> of the second proviso to this <u>Section 11.01(a)</u>) any fees or other amounts payable hereunder or under any other Loan Document, without the written consent of each Lender entitled to such amount; <u>provided</u>, <u>however</u>, that only the consent of the Required Lenders shall be necessary (i) to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest or LC Participation Fees at the Default Rate or (ii) to amend any financial covenant hereunder (or any defined term used therein) even if the effect of such amendment would be to reduce the rate of interest on any Revolving Loan or LC Disbursement or to reduce any fee payable hereunder;

(vi) change <u>Section 2.10</u> or <u>Section 8.03</u> in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender;

(vii) change any provision of this <u>Section 11.01</u> or the definition of "Required Lenders," "Supermajority Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;

(viii) subject to <u>Section 9.10</u>, release all or substantially all of the Collateral in any transaction or series of related transactions, without the written consent of each Lender;

(ix) subject to <u>Section 9.10</u>, release all or substantially all of the value of the Guaranty, without the written consent of each Lender;

(x) subordinate the Obligations or the Liens securing the Obligations without the written consent of each Lender; or

(xi) modify the definition of "Interest Period" without the consent of each affected Lender;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 153 of 432

and <u>provided</u>, <u>further</u>, that (A) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document; (B) <u>theany</u> Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto<u>and the Administrative Agent</u>; (C) no such amendment, waiver or consent shall, without the consent of the Issuing Bank or the Swingline Lender, amend, modify or waive any provision relating to the rights or obligations of the Issuing Bank or the Swingline Lender, respectively; (D) no such amendment, waiver or consent shall, without the prior written consent of the Supermajority Lenders, change the definition of the terms "Availability" or "Borrowing Base" or any component definition used therein (including, without limitation, the definitions of "Eligible Accounts" and "Eligible Inventory") if, as a result thereof, the amounts available to be borrowed by the Borrower would be increased; <u>provided</u> that the foregoing shall not limit the discretion of the Administrative Agent to change, establish or eliminate any Reserves or to add Accounts and Inventory acquired in a permitted Acquisition to the Borrowing Base as provided herein; and (E) no such amendment, waiver or consent shall, without the prior written consent of the Supermajority Lenders, increase the percentages set forth in the term "Borrowing Base" or add any new classes of eligible assets thereto.

(b) <u>Defaulting Lenders</u>. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except those described in <u>Section 11.01(a)(iii)</u>, (iv) and (v).

(c) <u>Replacement of Non-Consenting Lenders</u>. If any Lender does not consent to a proposed amendment, waiver, consent or release with respect to any Loan Document that requires the consent of each Lender and that has been approved by the Required Lenders, the Borrower may replace such non-consenting Lender in accordance with <u>Section 11.13</u>; <u>provided</u> that such amendment, waiver, consent or release can be effected as a result of the assignment contemplated by such Section (together with all other such assignments required by the Borrower to be made pursuant to this paragraph).

11.02 Notices; Effectiveness; Electronic Communications.

(a) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in <u>subsection (b)</u> below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

(i) if to Holdings, the Borrower, the Administrative Agent, the Issuing Bank or the Swingline Lender, to the address, facsimile number, electronic mail address or telephone number specified for such Person on <u>Schedule 11.02</u>; and

(ii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications to the extent provided in <u>subsection (b)</u> below shall be effective as provided in such <u>subsection (b)</u>.

(b) <u>Electronic Communications</u>. Notices and other communications to the Lenders and the Issuing Bank hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall not apply to notices to any Lender or the Issuing Bank pursuant to <u>Article II</u> if such Lender or the Issuing Bank, as applicable, has notified the Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 154 of 432

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), provided that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing <u>clause (i)</u> of notification that such notice or communication is available and identifying the website address therefor.

The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE (c) AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to Holdings, the Borrower, any Lender, the Issuing Bank or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to Holdings, the Borrower, any Lender, the Issuing Bank or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

(d) <u>Change of Address, Etc.</u> Each of Holdings, the Borrower, the Administrative Agent, the Issuing Bank and the Swingline Lender may change its address, facsimile or telephone number for notices and other communications hereunder by notice to the other parties hereto. Each Lender may change its address, facsimile or telephone number for notices and other communications hereunder by notice to the Borrower, the Administrative Agent, the Issuing Bank and the Swingline Lender. In addition, each Lender agrees to notify the Administrative Agent from time to time to ensure that the Administrative Agent has on record (i) an effective address, contact name, telephone number, facsimile number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender.

(e) <u>Reliance by Administrative Agent, the Issuing Bank and Lenders</u>. The Administrative Agent, the Issuing Bank and the Lenders shall be entitled to rely and act upon any notices (including any telephonic Notice of Borrowing) purportedly given by or on behalf of the Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify the Administrative Agent, the Issuing Bank, each Lender and the Related Parties of each of them from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower, except to the extent that such losses, costs, expenses or liabilities are determined by a final judgment of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of such Person. All telephonic notices to and other telephonic communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

11.03 <u>No Waiver; Cumulative Remedies</u>.

No failure by any Lender, the Issuing Bank or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder or under any other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided, and provided under each other Loan Document, are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 155 of 432

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Loan Parties or any of them shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent in accordance with Section 8.02 for the benefit of all the Lenders and the Issuing Bank; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) the Issuing Bank or the Swingline Lender from exercising the rights and remedies that inure to its benefit (solely in its capacity as Issuing Bank or Swingline Lender, as the case may be) hereunder and under the other Loan Documents, (c) any Lender from exercising setoff rights in accordance with Section 11.08 (subject to the terms of Section 2.10), or (d) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to any Loan Party under any Debtor Relief Law; and provided, further, that if at any time there is no Person acting as Administrative Agent hereunder and under the other Loan Documents, then (i) the Required Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 8.02 and (ii) in addition to the matters set forth in <u>clauses (b)</u>, (c) and (d) of the preceding proviso and subject to <u>Section 2.10</u>, any Lender may, with the consent of the Required Lenders, enforce any rights and remedies available to it and as authorized by the Required Lenders.

11.04 Expenses; Indemnity; Damage Waiver.

(a) <u>Costs and Expenses</u>. The Borrower shall pay (i) all reasonable out-of-pocket expenses incurred by the Administrative Agent, the Issuing Banks and their respective Affiliates (including the reasonable fees, charges and disbursements of counsel for the Administrative Agent and the Issuing Banks, but excluding any costs of maintaining the Platform), in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated) and (ii) all out-of-pocket expenses incurred by the Administrative Agent, the Issuing Banks or any Lender (including the reasonable fees, charges and disbursements of any counsel for the Administrative Agent, the Issuing Banks or any Lender, in connection with the enforcement or protection of its rights (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with Revolving Loans made or Letters of Credit issued hereunder, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Revolving Loans or Letters of Credit.

Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Issuing Bank, each Lender and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, liabilities, penalties and related expenses (including the reasonable fees, charges and disbursements of any counsel for any Indemnitee), and shall indemnify and hold harmless each Indemnitee from all reasonable, allocated fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any third party or by the Borrower or any other Loan Party arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or the consummation of the transactions contemplated hereby or thereby, or, in the case of the Administrative Agent (and any sub-agent thereof) and its Related Parties only, the administration of this Agreement and the other Loan Documents, (ii) any Revolving Loan or Letter of Credit or the use or proposed use of the proceeds therefrom (including any refusal by the Issuing Bank to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (iii) any actual or alleged presence or Release of Hazardous Materials on, at, under or from any property owned or operated by the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by the Borrower or any other Loan Party or any of the Borrower's or such Loan Party's directors, shareholders or creditors, and regardless of whether any Indemnitee is a party thereto, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 156 of 432

COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF THE INDEMNITEE; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by the Borrower or any other Loan Party against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, if the Borrower or such Loan Party has obtained a final judgment in its favor on such claim as determined by a court of competent jurisdiction.

(c) <u>Reimbursement by Lenders</u>. To the extent that the Borrower for any reason fails to pay any amount required under <u>subsection (a)</u> or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), the Issuing Bank or any Related Party of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the Issuing Bank or such Related Party, as the case may be, such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnify payment is sought) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) or the Issuing Bank in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent) or Issuing Bank in connection with such capacity.

Waiver of Consequential Damages, Etc. To the fullest extent permitted by applicable Law, neither (d) the Borrower nor any Indemnitee shall assert, and each of the Borrower and the Indemnitees hereby waives, any claim against the Borrower, any other Loan Party or any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Revolving Loan or Letter of Credit or the use of the proceeds thereof. No Indemnite referred to in subsection (b) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final judgment of a court of competent jurisdiction. For the avoidance of doubt, this subsection (d) shall not limit the obligation of the Borrower to indemnify each Indemnitee for any liabilities or damages incurred by such Indemnitee that are asserted against such Indemnitee by a third party and that are payable by the Borrower pursuant to Section 11.04(b).

(e) <u>Payments</u>. All amounts due under this Section shall be payable not later than ten Business Days after demand therefor.

(f) <u>Survival</u>. The agreements in this Section shall survive the resignation of the Administrative Agent, the Issuing Bank and the Swingline Lender, the replacement of any Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all the other Obligations.

11.05 Payments Set Aside.

To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent, the Issuing Bank or any Lender, or the Administrative Agent, the Issuing Bank or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, the Issuing Bank or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender and the Issuing Bank severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, <u>plus</u> interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 157 of 432

to time in effect. The obligations of the Lenders and the Issuing Bank under <u>clause (b)</u> of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

11.06 <u>Successors and Assigns</u>.

(a) Successors and Assigns Generally. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that neither the Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder or under the other Loan Documents (except in connection with any transaction permitted by <u>Section 7.04(a)</u>, (d) or (e) without the prior written consent of the Administrative Agent and each Lender, and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of <u>Section 11.06(b)</u>, (ii) by way of participation in accordance with the provisions of <u>Section 11.06(b)</u>, (ii) by way of participation in accordance with the provisions of <u>Section 11.06(f)</u> (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in <u>subsection (d)</u> of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Issuing Bank and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Revolving Commitment(s) and the Revolving Loans (including for purposes of this <u>Section 11.06(b)</u>, participations in LC Obligations and in Swingline Loans) at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

(i) <u>Minimum Amounts</u>.

(A) in the case of an assignment of the entire remaining amount of the assigning Lender's Revolving Commitment and the Revolving Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and

(B) in any case not described in <u>subsection (b)(i)(A)</u> of this Section, the aggregate amount of the Revolving Commitment (which for this purpose includes Revolving Loans outstanding thereunder) or, if the Revolving Commitment is not then in effect, the outstanding principal balance of the Revolving Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); <u>provided</u>, <u>however</u>, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met;

(ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Revolving Loans or the Revolving Commitment assigned, except that this <u>clause (ii)</u> shall not apply to the Swingline Lender's rights and obligations in respect of Swingline Loans;

(iii) <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by <u>subsection (b)(i)(B)</u> of this Section and, in addition:

(A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (1) an Event of Default has occurred and is continuing at the time

of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; <u>provided</u>, that if the Borrower fails to respond to a request for a consent to an assignment within five Business Days following the date such request is received by the Borrower, then the Borrower shall be deemed to have consented to such assignment;

(B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of any Revolving Commitment or any Revolving Loan if such assignment is to a Person that is not a Lender, an Affiliate of a Lender or an Approved Fund;

(C) the consent of the Issuing Bank (such consent not to be unreasonably withheld or delayed) shall be required for any assignment that increases the obligation of the assignee to participate in exposure under one or more Letters of Credit (whether or not then outstanding); and

(D) the consent of the Swingline Lender (such consent not to be unreasonably withheld or delayed) shall be required for any assignment in respect of the Revolving Commitments.

(iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount, if any, required as set forth in <u>Schedule 11.06</u>; provided, however, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.

(v) <u>No Assignment to the Borrower.</u> No such assignment shall be made to the Borrower or any of the Borrower's Affiliates or Subsidiaries.

(vi) <u>No Assignment to Natural Persons</u>. No such assignment shall be made to a natural person.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to <u>subsection (c)</u> of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of <u>Sections 3.01</u>, <u>3.04</u>, <u>3.05</u> and <u>11.04</u> with respect to facts and circumstances occurring prior to the effective date of such assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with <u>Section 11.06(d)</u>.

(c) <u>Register</u>. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Revolving Commitments of, and principal and interest amounts of the Revolving Loans and LC Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and, with respect to itself, any Lender, at any reasonable time and from time to time upon reasonable prior notice. <u>No assignment shall be effective unless recorded in the Register</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 159 of 432

Participations. Any Lender may at any time, without the consent of, or notice to, the Borrower or (d) the Administrative Agent, sell participations to any Person (other than a natural person or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Revolving Commitment and/or the Revolving Loans (including such Lender's participations in LC Obligations and/or Swingline Loans) owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the Issuing Bank shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in clauses (iii) through (vi) of the first proviso to Section 11.01(a) that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.01, 3.04 and 3.05 (subject to the requirements and limitations of those Sections, including the documentation requirements in Section 3.01(e), and Section 3.06) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 11.06(b). To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.08 as though it were a Lender, provided such Participant complies with Section 2.10 as though it were a Lender.

Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal and interest amounts of each Participant's interest in the Revolving Loans or other obligations under this Agreement (the "<u>Participant Register</u>"). The entries in the Participant Register shall be conclusive and such Lender (and the Borrower, to the extent that the Participant requests payment from the Borrower) shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. No Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations.

<u>The parties hereto agree and intend that Section 11.06(c) and Section 11.06(d) shall be construed so that the</u> <u>Obligations are treated as in "registered form" for the purposes of the Code (including Code Sections 163(f).</u> <u>871(h)(2), and 881(c)(2)), and the Register and Participant Register shall be maintained in accordance with such</u> <u>intention.</u>

(e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Section 3.01, 3.04</u> or <u>3.05</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent or except to the extent such entitlement to a greater payment results from a Change in Law occurring after such Participant became a Participant.

(f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank or other central bank having jurisdiction over such Lender; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto.

(g) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 160 of 432

Resignation as Issuing Bank or Swingline Lender after Assignment. Notwithstanding anything to (h) the contrary contained herein, if at any time Bank of America assigns all of its Revolving Commitments and Revolving Loans pursuant to Section 11.06(b), Bank of America may, (i) upon 30 days' notice to the Borrower and the Lenders, resign as Issuing Bank and/or (ii) upon 30 days' notice to the Borrower, resign as Swingline Lender. In the event of any such resignation as Issuing Bank or Swingline Lender, the Borrower shall be entitled to appoint from among the Lenders a successor Issuing Bank or Swingline Lender hereunder; provided, however, that no failure by the Borrower to appoint any such successor shall affect the resignation of Bank of America as Issuing Bank or Swingline Lender, as the case may be. If Bank of America resigns as Issuing Bank, it shall retain all the rights, powers, privileges and duties of the Issuing Bank hereunder with respect to all Letters of Credit outstanding as of the effective date of its resignation as Issuing Bank and all LC Obligations with respect thereto (including the right to require the Lenders to make Base Rate Loans or fund risk participations pursuant to Section 2.13). If Bank of America resigns as Swingline Lender, it shall retain all the rights of the Swingline Lender provided for hereunder with respect to Swingline Loans made by it and outstanding as of the effective date of such resignation, including the right to require the Lenders to make Base Rate Loans or fund risk participations in outstanding Swingline Loans pursuant to Section 2.12(d). Upon the appointment of a successor Issuing Bank and/or Swingline Lender, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuing Bank or Swingline Lender, as the case may be, and (b) the successor Issuing Bank shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to Bank of America to effectively assume the obligations of Bank of America with respect to such Letters of Credit.

11.07 <u>Treatment of Certain Information; Confidentiality.</u>

Each of the Administrative Agent, the Lenders and the Issuing Bank agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by applicable Laws or regulations or by any subpoena or similar legal process (provided, that in the event of any such disclosure under this clause (c), the Administrative Agent, such Lender or the Issuing Bank, as the case may be, agrees to use commercially reasonable efforts to inform the Borrower of such disclosure to the extent not prohibited by Law), (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section or (ii) becomes available to the Administrative Agent, any Lender, the Issuing Bank or any of their respective Affiliates on a nonconfidential basis from a source other than the Borrower (other than through a Person whom the Administrative Agent, such Lender or the Issuing Bank actually knows to be acting in violation of his or its obligations to the Borrower or any other Loan Party).

For purposes of this Section, "<u>Information</u>" means all information received from any Loan Party or any Subsidiary thereof relating to any Loan Party or any Subsidiary thereof or their respective businesses, other than any such information that is available to the Administrative Agent, any Lender or the Issuing Bank on a nonconfidential basis prior to disclosure by any Loan Party or any Subsidiary thereof, <u>provided</u> that, in the case of information received from a Loan Party or any such Subsidiary after the Closing Date, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 161 of 432

Each of the Administrative Agent, the Lenders and the Issuing Bank acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including Federal and state securities Laws.

11.08 <u>Right of Setoff.</u>

If an Event of Default shall have occurred and be continuing, each Lender, the Issuing Bank and each of their respective Affiliates is hereby authorized at any time and from time to time, after obtaining the prior written consent of the Administrative Agent, to the fullest extent permitted by applicable law, to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by such Lender, the Issuing Bank or any such Affiliate to or for the credit or the account of Holdings, the Borrower or any other Loan Party against any and all of the obligations of Holdings, the Borrower or such Loan Party now or hereafter existing under this Agreement or any other Loan Document to such Lender or the Issuing Bank, irrespective of whether or not such Lender or the Issuing Bank shall have made any demand under this Agreement or any other Loan Document and although such obligations of Holdings, the Borrower or such Loan Party may be contingent or unmatured or are owed to a branch or office of such Lender or the Issuing Bank different from the branch or office holding such deposit or obligated on such indebtedness; provided, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.11(b) and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff. The rights of each Lender, the Issuing Bank and their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff) that such Lender, the Issuing Bank or their respective Affiliates may have. Each Lender and the Issuing Bank agrees to notify the Borrower and the Administrative Agent promptly after any such setoff and application, provided that the failure to give such notice shall not affect the validity of such setoff and application.

11.09 Interest Rate Limitation.

Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law, or that would result in the receipt by any Lender of "interest" at a "criminal rate" as such terms are construed under the *Criminal Code* (Canada) (the "<u>Maximum Rate</u>"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Revolving Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

11.10 <u>Counterparts; Effectiveness</u>.

This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Except as provided in <u>Section 4.01</u>, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or electronic transmission shall be effective as delivery of a manually executed counterpart of this Agreement.

11.11 <u>Survival of Representations and Warranties</u>.

All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or any Lender or on their behalf and notwithstanding that the Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of any Borrowing, and shall continue in full force and effect as long as any Revolving Loan or any other Obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.

11.12 <u>Severability</u>.

If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

11.13 <u>Replacement of Lenders</u>.

(a) If any Lender requests compensation under <u>Section 3.04</u>, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section</u> <u>3.01</u>, if any Lender is a Defaulting Lender, or if any other circumstance exists hereunder that gives the Borrower the right to replace a Lender as a party hereto, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, <u>Section 11.06</u> (notwithstanding the foregoing, it being understood that such assignment shall not require the consent of such Lender), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), <u>provided</u> that:

(i) the Borrower shall have paid to the Administrative Agent the processing and recordation fee specified in <u>Section 11.06(b)</u>;

(ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Revolving Loans and LC Disbursements, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>Sections 3.01</u>, <u>3.04</u> or <u>3.05</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);

(iii) in the case of any such assignment resulting from a claim for compensation under <u>Section</u> <u>3.04</u> or payments required to be made pursuant to <u>Section 3.01</u>, such assignment will result in a reduction in such compensation or payments thereafter; and

(iv) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

(b) Notwithstanding anything herein to the contrary, the Borrower may at any time terminate in full the Revolving Commitment of any Revolving Lender that is a Designated Lender (without any requirement of any reduction in the Revolving Commitments of any other Revolving Lenders) by giving notice to such Designated Lender and the Administrative Agent (which notice shall specify the effective date of such termination); provided,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 163 of 432

that (i) at the time of such termination, no Event of Default exists; (ii) concurrently with such termination, the Borrower shall pay to such Designated Lender an amount equal to the outstanding principal of its Revolving Loans. accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents; (iii) if, on the effective date of such termination, any Letter of Credit is outstanding, the conditions specified in Section 2.13(b) would be satisfied (after giving effect to such termination) were each such Letter of Credit issued on such date; and (iv) if, on the effective date of such termination, any Swingline Loan is outstanding, the conditions specified in Section 2.12(a) would be satisfied (after giving effect to such termination) were each such Swingline Loan to be made on such date. Upon satisfaction of the conditions specified in the foregoing clauses (i), (ii), (iii) and (iv), the Revolving Commitment of such Designated Lender shall terminate on the effective date specified in such notice, its participation in the LC Obligations and outstanding Swingline Loans (if any) shall terminate on such effective date and the participations of the other Revolving Lenders in the LC Obligations and outstanding Swingline Loans shall be redetermined as of such termination date as if the outstanding Letters of Credit had been issued, the LC Disbursements had been paid by the Issuing Bank, and the Swingline Loans had been advanced by the Swingline Lender, in each case on such termination date. The termination of the Revolving Commitment of a Designated Lender pursuant to this Section 11.13(b) shall not be deemed to be a waiver of any right that the Borrower, the Administrative Agent, the Issuing Bank, the Swingline Lender or any other Lender may have against such Designated Lender.

11.14 <u>Governing Law; Jurisdiction; Etc.</u>

(a) <u>GOVERNING LAW</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.

(b) SUBMISSION TO JURISDICTION. THE BORROWER AND EACH OTHER LOAN PARTY IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THE BANKRUPTCY COURT, THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK, AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT (OTHER THAN AS PROVIDED IN ANY MORTGAGE OR OTHER COLLATERAL DOCUMENT WITH RESPECT TO ITSELF), OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN THE BANKRUPTCY COURT OR SUCH STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW.

(c) <u>WAIVER OF VENUE</u>. THE BORROWER AND EACH OTHER LOAN PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY COURT REFERRED TO IN <u>PARAGRAPH (b)</u> OF THIS SECTION. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.

(d) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 11.02</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

11.15 <u>Waiver of Jury Trial</u>.

EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

11.16 <u>No Advisory or Fiduciary Responsibility</u>.

In connection with all aspects of each transaction contemplated hereby, the Borrower and Holdings each acknowledge and agree, and acknowledge their respective Affiliates' understanding, that: (i) the credit facility provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between the Borrower, Holdings and their respective Affiliates, on the one hand, and the Administrative Agent and the Lead Arrangers, on the other hand, and each of the Borrower and Holdings is capable of evaluating and understanding and understands and accepts the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver or other modification hereof or thereof); (ii) in connection with the process leading to such transaction, each of the Administrative Agent and the Lead Arrangers is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary for the Borrower, Holdings or any of their respective Affiliates, stockholders, creditors or employees or any other Person; (iii) neither the Administrative Agent nor any Lead Arranger has assumed or will assume an advisory, agency or fiduciary responsibility in favor of the Borrower or Holdings with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any other Loan Document (irrespective of whether the Administrative Agent or any Lead Arranger has advised or is currently advising the Borrower, Holdings or any of their respective Affiliates on other matters) and neither the Administrative Agent nor any Lead Arranger has any obligation to the Borrower, Holdings or any of their respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; (iv) the Administrative Agent and the Lead Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower, Holdings and their respective Affiliates, and neither the Administrative Agent nor any Lead Arranger has any obligation to disclose any of such interests by virtue of any advisory, agency or fiduciary relationship; and (v) the Administrative Agent and the Lead Arrangers have not provided and will not provide any legal, accounting, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Loan Document) and each of the Borrower and Holdings has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. Each of the Borrower and Holdings hereby waives and releases, to the fullest extent permitted by law, any claims that it may have against the Administrative Agent and the Lead Arrangers with respect to any breach or alleged breach of agency or fiduciary duty.

11.17 Judgment Currency.

If, for the purposes of obtaining judgment in any court in any jurisdiction with respect to this Agreement or any other Loan Document, it becomes necessary to convert into a particular currency (the "Judgment Currency") any amount due under this Agreement or under any other Loan Document in any currency other than the Judgment Currency (the "<u>Currency Due</u>"), then conversion shall be made at the rate of exchange prevailing on the Business Day before the day on which judgment is given. For this purpose "rate of exchange" means the rate at which the Administrative Agent is able, on the relevant date, to purchase the Currency Due with the Judgment Currency in accordance with its normal practice at its head office in New York. In the event that there is a change in the rate of exchange prevailing between the Business Day before the day on which the judgment is given and the date

of receipt by the Administrative Agent of the amount due, the Borrower will, on the date of receipt by the Administrative Agent, pay such additional amounts, if any, or be entitled to receive reimbursement of such amount, if any, as may be necessary to ensure that the amount received by the Administrative Agent on such date is the amount in the Judgment Currency that when converted at the rate of exchange prevailing on the date of receipt by the Administrative Agent is the amount then due under this Agreement or such other Loan Document in the Currency Due. If the amount of the Currency Due which the Administrative Agent is so able to purchase is less than the amount of the Currency Due originally due to it, the Borrower shall indemnify and save the Administrative Agent and the Lenders harmless from and against all loss or damage arising as a result of such deficiency. This indemnity shall constitute an obligation separate and independent from the other obligations contained in this Agreement and the other Loan Documents, shall give rise to a separate and independent cause of action, shall apply irrespective of any indulgence granted by the Administrative Agent from time to time and shall continue in full force and effect notwithstanding any judgment or order for a liquidated sum in respect of an amount due under this Agreement or any other Loan Document or under any judgment or order.

11.18 USA PATRIOT Act Notice.

Each Lender that is subject to the USA PATRIOT Act and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies each Loan Party that pursuant to the requirements of the USA PATRIOT Act, it is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify each Loan Party in accordance with the USA PATRIOT Act. The Loan Parties shall, promptly following a request by the Administrative Agent or any Lender, provide all documentation and other information that the Administrative Agent or such Lender requests in order to comply with its ongoing obligations under applicable "know your customer" an anti-money laundering rules and regulations, including the USA PATRIOT Act.

11.19 <u>Canadian Anti-Money Laundering Legislation</u>.

If the Administrative Agent has ascertained the identity of any Canadian Guarantor or any authorized signatories of any Canadian Guarantor for the purposes of any Canadian Anti-Terrorism and AML Legislation (including any "know your client" policies, regulations, laws or rules), then the Administrative Agent:

(i) shall be deemed to have done so as an agent for each Lender and this Agreement shall constitute a "written agreement" in such regard between each Lender and the Administrative Agent within the meaning of the applicable Canadian Anti-Terrorism and AML Legislation; and

(ii) shall provide to the Administrative Agent copies of all information obtained in such regard without any representation or warranty as to its accuracy or completeness.

Notwithstanding the preceding sentence and except as may otherwise be agreed in writing, each Lender agrees that the Administrative Agent has no obligation to ascertain the identity of the Canadian Guarantors or any authorized signatories of the Canadian Guarantors on behalf of any Lender or to confirm the completeness or accuracy of any information it obtains from any Canadian Guarantor or any such authorized signatory in doing so.

11.20 ENTIRE AGREEMENT.

THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 166 of 432

11.22 Subject to Intercreditor Agreements.

Notwithstanding anything herein to the contrary, (i) the liens and security interests granted to the Secured Parties pursuant to the Collateral Documents are expressly subject to the DIP Term Intercreditor Agreement, the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement and the 2014 Senior/Junior Intercreditor Agreement and (ii) the exercise of any right or remedy by the Administrative Agent hereunder or under the DIP. Term Intercreditor Agreement, the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement is subject to the limitations and provisions of the DIP Term Intercreditor Agreement, the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement and the 2014 Senior/Junior Intercreditor Agreement. In the event of any conflict between the terms of the DIP Term Intercreditor Agreement. the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014 Senior/Junior Intercreditor Agreement and the terms of this Agreement, the terms of the DIP Term Intercreditor Agreement, the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement (as applicable) shall govern. Except as otherwise expressly set forth in the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement, in the event of any conflict between the terms of the 2016 Intercreditor Agreement and the terms of the First Lien Intercreditor Agreement or the 2014Senior/Junior Intercreditor Agreement, the terms of the First Lien Intercreditor Agreement or the 2014 Senior/Junior Intercreditor Agreement (as applicable) shall govern. Except as otherwise expressly set forth in the DIP Term Intercreditor Agreement, the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the Senior/Junior Intercreditor Agreement, in the event of any conflict between the terms of the DIP Term Intercreditor Agreement and the terms of the 2016 Intercreditor Agreement, the First Lien Intercreditor Agreement or the Senior/Junior Intercreditor Agreement, the terms of the DIP Term Credit Agreement shall govern.

11.23 Acknowledgement and Consent to Bail-In of EEA Financial Institutions.

Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an EEA Financial Institution; and

(b) the effects of any Bail-in Action on any such liability, including, if applicable:

(i) a reduction in full or in part or cancellation of any such liability;

(ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent entity, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or

(iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of any EEA Resolution Authority.

11.24 <u>Conflict with Ratification Agreement</u>. In the event of any conflict between the terms of the Ratification Agreement, on the one hand and this Agreement and the other Loan Documents, on the other, or in the event a matter is addressed in the Ratification Agreement but not in this Agreement or the other Loan Documents, the terms of the Ratification Agreement shall govern.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 167 of 432

[Remainder of Page Intentionally Left Blank]

SCHEDULE 1.02

Material Contracts

- 1. 4% Notes Documents
- 2. 6% Notes Documents
- 3. 6% Unsecured Notes Documents
- 4. 8.5% Notes Documents
- 5. DIP Term Credit Agreement
- 6. General Terms, Conditions, and Specifications For the Procurement of 2020 Census Printing and Mailing as requisitioned from the U.S. Government Publishing Office (GPO) by the U.S. Department of Commerce / U.S. Census Bureau (the "<u>General Terms</u>")
- 7. Purchase Order dated as of October 17, 2017 related to the General Terms

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 169 of 432

SCHEDULE 1.04

Rebate Accruals

- 1. Print Service Group rebate accrual
- 2. Commercial Print Group rebate accrual
- 3. Envelope Group rebate accrual
- 4. Labels Group rebate accrual

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 170 of 432

SCHEDULE 1.05

Accounts

Customer	Terms
RB MANUFACTURING LLC	120
ABBOTT DIAGNOSTICS	90
ABBOTT NUTRITION	90
ABBOTT PROCUREMENT LLC	90
ADARE INTERNATIONAL	90
ALC, LLC	90
AVENUE 4 COMM GRP	90
BEDFORD ST MARTIN'S	90
CAMPBELL EWALD	90
COTY US LLC DBA O P I	90
CREATIVE NETWORK	90
DHL eCommerce	90
FEDEX DISBURSEMENTS	90
HALCYON BUS PUB INC	90
HH GLOBAL	90
LOGICSOURCE, INC.	90
OSPREY PUBLISHING LIMITED	90
PROFORMA SCOTTO PROMOTIONS	90
SHOPPER LOCAL	90
CVS CAREMARK CORPORATION	75
CVS CORPORATION	75
KELLY PAPER	75
LINDENMEYR ENVELOPE	75
LINDENMEYR MUNROE	75
LOGICSOURCE, INC	75
OWENS CORNING SALES LLC	75
SPICERS PAPER INC.	75

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 171 of 432

Revolving Name of Lender **Applicable Percentage** Commitment BANK OF AMERICA, N.A. \$50,000,000 0.263157895% WELLS FARGO BANK, \$42,500,000 0.223684211% NATIONAL ASSOCIATION JPMORGAN CHASE BANK, \$42,500,000 0.223684211% N.A. PNC BANK, NATIONAL \$35,000,000 0.184210526% ASSOCIATION BARCLAYS BANK PLC \$20,000,000 0.105263158% Total \$190,000,000 100.00000000%

Revolving Commitments and Applicable Percentages

SCHEDULE 2.01

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 172 of 432

SCHEDULE 2.13

Borrower	LOC #	Beneficiary	Amount
Cenveo Corporation	3077724	Hartford Fire Insurance Company	\$ 14,000,000
Cenveo Inc. 3048165 United States Fidelity and Guarantee Company, C/O Discovery Managers, Ltd.		\$ 1,375,000	
Nashua Corporation	68030554	Traveler's Indemnity Company	\$ 442,000
Cenveo Corporation	3008927	Ace America Insurance Company, Indemnity Insurance Company of North Amercia, Bankers Standard Insurance Company, Pacific Employers Insurance Co	\$ 164,000
Cenveo Corporation	3099978	Southern California Edison	\$ 375,000
Cenveo Corporation	3126197	Nip Owner LLC	\$ 100,000
Cenveo Corporation	3118554	Appalachian Electric Cooperative	\$ 80,000
Cenveo Corporation	3091899	Snohomis County PUD No 1	\$ 29,586

Existing Letters or Credit

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 173 of 432

SCHEDULE 5.08(b)

Existing Liens

Liens on assets described on the UCC lien summary report attached as Attachment I hereto.

Easements, rights-of-way, restrictions and other similar encumbrances affecting real property which, in the aggregate, are not substantial in amount, and which do not in any case materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Person.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 174 of 432

Attachment I

			CENVEO CO	ORPORATION				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	VFI KR SPE I LLC 6340 South 3000 East, Suite 400 Salt Lake City,			UCC- 1 as of 12-08- 2017	05-06- 2014	2014 1783372	Leased equipment
Delaware - Secretary of State	WestRock Converting Company 504 Thrasher Street Norcross, GA 30071	UT 84121 VFI KR SPE I LLC 6340 South 3000 East, Suite 400 Salt Lake City, UT 84121			UCC- 3 as of 12-08- 2017	04-06- 2016	2016 2044087	Amendment to original financing statement no. 2014 1783372 changing debtor's name and address
Delaware - Secretary of State	WestRock Converting Company 504 Thrasher Street Norcross, GA 30071	VFI KR SPE I LLC 6340 South 3000 East, Suite 400 Salt Lake City, UT 84121			UCC- 3 as of 12-08- 2017	03-27- 2017	2017 1989299	Amendment to original financing statement no. 2014 1783372 restating collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Peak, LLC, as Representative 15 Main Street, Suite 2 Warrenton, VA 20186			UCC- 1 as of 12-08- 2017	08-07- 2015	2015 3434999	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 175 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street	Peak, LLC, as Representative 15 Main Street, Suite 2	VFI KR SPE I LLC 6340 South 3000 East,	UCC- 3 as of 12-08- 2017	08-21- 2015	2015 3677779	Assignment of original financing statement no. 2015 3434999
	Stamford, CT 06901	Warrenton, VA 20186	Suite 400 Salt Lake City, UT 84121				
Delaware - Secretary of State	WestRock Converting Company 504 Thrasher Street Norcross, GA 30071	VFI KR SPE I LLC 6340 South 3000 East, Suite 400 Salt Lake City, UT 84121		UCC- 3 as of 12-08- 2017	04-05- 2016	2016 2023511	Amendment to original financing statement no. 2015 3434999 changing debtor's name and address
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	10-09- 2013	2013 3951747	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	UCC- 3 as of 12-08- 2017	01-01- 2014	2014 0128306	Assignment of original financing statement no. 2013 3951747

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 176 of 432

Delaware - Secretary of State	WestRock Converting Company 504 Thrasher Street Norcross, GA 30071	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 3 as of 12-08- 2017	04-05- 2016	2016 2004768	Amendment to original financing statement no. 2014 0128306 changing debtor's name and address
Delaware - Secretary of State	Cenveo Corporation 8310 S. Valley Highway, #400 Englewood, CO 80112 Cenveo Commercial, Inc. 8310 S. Valley Highway, #400 Englewood, CO 80112	General Electric Capital Corporation 401 Merritt Seven, Suite 23 Norwalk, CT 06851-1177		UCC- 1 as of 12-08- 2017	08-26- 2004	4242045 5	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 8310 S. Valley Highway, #400 Englewood, CO 80112 Cenveo Commercial, Inc. 8310 S. Valley Highway, #400 Englewood, CO 80112	General Electric Capital Corporation 401 Merritt Seven, Suite 23 Norwalk, CT 06851-1177		UCC- 3 as of 12-08- 2017	03-03-2009	2009 0667946	Continuation of original financing statement no. 4242045 5

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 177 of 432

Delaware - Secretary of State	Cenveo Corporation 8310 S. Valley Highway, #400 Englewood, CO 80112 Cenveo Commercial, Inc. 8310 S. Valley Highway, #400 Englewood, CO 80112	General Electric Capital Corporation 401 Merritt Seven, Suite 23 Norwalk, CT 06851-1177		UCC- 3 as of 12-08- 2017	03-04-2014	2014 0818153	Continuation of original financing statement no. 4242045 5
Delaware - Secretary of State	Cenveo Corporation 8310 S. Valley Highway, #400 Englewood, CO 80112	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	09-07- 2005	5275890 3	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 8310 S. Valley Highway, #400 Englewood, CO 80112	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	05-26- 2006	6180220 6	Amendment to original financing statement no. 5275890 3 adding collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	10-23- 2007	2007 4004429	Amendment to original financing statement no. 5275890 3 changing debtor's address

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 178 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	07-02- 2008	2008 2272050	Amendment to original financing statement no. 5275890 3 adding collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	05-21- 2009	2009 1747309	Amendment to original financing statement no. 5275890 3 restating collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	08-13- 2010	2010 2838310	Continuation of original financing statement no. 5275890 3
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	08-07- 2015	2015 3438297	Amendment to original financing statement no. 5275890 3 restating collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 179 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green #201 Stamford, CT 06901	INX International Inc. Co. 150 North Martingale Rd. Suite 700		UCC- 3 as of 12-08- 2017	08-07- 2015	2015 3438305	Continuation of original financing statement no. 5275890 3
		Schaumburg, IL 60173					
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	Harold M. Pitman Company 721 Union Blvd. Totowa, NJ 07512		UCC- 1 as of 12-08- 2017	09-05- 2006	6307777 3	All consigned goods and inventory including photo supplies, etc.
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	AGFA II Acquisition Corp. 721 Union Blvd. Totowa, NJ 07512		UCC- 3 as of 12-08- 2017	12-21- 2010	2010 4522714	Amendment to original financing statement no. 6307777 3 changing secured party's name
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	AGFA II Acquisition Corp. 721 Union Blvd. Totowa, NJ 07512		UCC- 3 as of 12-08- 2017	03-28- 2011	2011 1137069	Continuation of original financing statement no. 6307777 3
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	AGFA Corporation 611 River Drive, Center #3 Elmwood Park, NJ 07407		UCC- 3 as of 12-08- 2017	01-19- 2012	2012 0243594	Amendment to original financing statement no. 6307777 3 changing secured party's name and address

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 180 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	AGFA Corporation 611 River Drive, Center #3 Elmwood Park, NJ 07407		UCC- 3 as of 12-08- 2017	06-13- 2014	2014 2320612	Amendment to original financing statement no. 6307777 3 changing debtor's address
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	AGFA Corporation 611 River Drive, Center #3 Elmwood Park, NJ 07407		UCC- 3 as of 12-08- 2017	04-13- 2016	2016 2174397	Continuation of original financing statement no. 6307777 3
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	E.I. DuPont de Nemours & Company Routes 141 & 48, Barley Mill Plaza 30 Wilmington, DE 19805		UCC- 1 as of 12-08- 2017	03-26- 2007	2007 1123263	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	E.I. DuPont de Nemours & Company Routes 141 & 48, Barley Mill Plaza 30 Wilmington, DE 19805		UCC- 3 as of 12-08- 2017	12-01- 2011	2011 4598408	Continuation of original financing statement no. 2007 1123263

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 181 of 432

Delaware -	Cenveo Corporation	E.I. DuPont de		UCC-	10-20-	2016 6465486	Continuation of
Secretary of State	One Canterbury Green Stamford, CT 06901	Nemours & Company		3 as of 12-08- 2017	2016		original financing statement no. 2007 1123263
	Stannord, CT 00901	Routes 141 & 48, Barley Mill Plaza 30		2017			1123203
		Wilmington, DE 19805					
Delaware - Secretary of State	Cenveo Corporation One Canturbury Green Stamford, CT 06901	RBS Asset Finance, Inc. 480 Jefferson Blvd. Warwick, RI 02886		UCC- 1 as of 12-08- 2017	05-29- 2008	2008 1950631	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canturbury Green Stamford, CT 06901	RBS Asset Finance, Inc. 480 Jefferson Blvd. Warwick, RI 02886		UCC- 3 as of 12-08- 2017	10-08- 2008	2008 3477864	Amendment to original financing statement no. 2008 1950631 adding collateral
Delaware - Secretary of State	Cenveo Corporation One Canturbury Green Stamford, CT 06901	RBS Asset Finance, Inc. 480 Jefferson Blvd. Warwick, RI 02886		UCC- 3 as of 12-08- 2017	11-18- 2008	2008 3849559	Amendment to original financing statement no. 2008 1950631 adding collateral
Delaware - Secretary of State	Cenveo Corporation One Canturbury Green Stamford, CT 06901	RBS Asset Finance, Inc. 480 Jefferson Blvd. Warwick, RI 02886		UCC- 3 as of 12-08- 2017	03-19- 2013	2013 1048769	Continuation of original financing statement no. 2008 1950631

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 182 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	General Electric Capital Corporation 10 Riverview Drive Danbury, CT 06810		UCC- 1 as of 12-08- 2017	10-02- 2009	2009 3259600	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	General Electric Capital Corporation of Tennessee 10 Riverview Drive Danbury, CT 06810		UCC- 3 as of 12-08- 2017	04-19-2010	2010 1334576	Amendment to original financing statement on. 2009 3259600 changing secured party's name.
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	General Electric Capital Corporation of Tennessee PO Box 35713 Billings, MT 59107-571		UCC- 3 as of 12-08- 2017	12-06- 2013	2013 4813094	Amendment to original financing statement on. 2009 3259600 changing secured party's address
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	General Electric Capital Corporation of Tennessee PO Box 35713 Billings, MT 59107-571		UCC- 3 as of 12-08- 2017	05-27- 2014	2014 2052363	Continuation of original financing statement on. 2009 3259600
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	Somerset Leasing Corp. XVII 1087 Broad Street Bridgeport, CT 06604		UCC- 1 as of 12-08- 2017	12-23- 2009	2009 4118219	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 183 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	Somerset Leasing Corp. XVII 1087 Broad Street Bridgeport, CT 06604		UCC- 3 as of 12-08- 2017	11-14- 2014	2014 4608329	Continuation of original financing statement no. 2009 4118219
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Somerset Capital Group, Ltd. Merritt Corporate Woods, 612 Wheelers Farms Road Milford, CT 06461		UCC- 1 as of 12-08- 2017	05-14- 2010	2010 1700305	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Somerset Capital Group, Ltd. Merritt Corporate Woods, 612 Wheelers Farms Road Milford, CT 06461		UCC- 3 as of 12-08- 2017	03-12- 2015	2015 1038800	Continuation of original financing statement no. 2010 1700305
Delaware - Secretary of State	Cenveo Corporation 1635 Coining Drive Toledo, OH 43612	Unisource Worldwide, Inc. 850 North Arlington Heights Road Itasca, IL 60143-141		UCC- 1 as of 12-08- 2017	07-27- 2010	2010 2596033	Consignment stock and other products and supplies

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 184 of 432

Delaware - Secretary of State	Cenveo Corporation 1635 Coining Drive Toledo, OH 43612	Unisource Worldwide, Inc. 850 North Arlington Heights Road Itasca, IL 60143-141		UCC- 3 as of 12-08- 2017	07-06- 2015	2015 2897840	Continuation of original financing statement no. 2015 1038800
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Somerset Leasing Corp. IV Merritt Corporate Woods, 612 Wheelers Farms Road Milford, CT 06461		UCC- 1 as of 12-08- 2017	08-30- 2010	2010 3025024	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Somerset Leasing Corp. IV Merritt Corporate Woods, 612 Wheelers Farms Road Milford, CT 06461		UCC- 3 as of 12-08- 2017	06-02- 2015	2015 2334596	Continuation of original financing statement no. 2010 3025024
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	07-29- 2011	2011 3185280	Leased equipment described on attached Exhibit "A"

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 185 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	11-13- 2012	2012 4515666	Amendment to original financing statement no. 2011 3185280 restating collateral description
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	09-02- 2014	2014 3493566	Amendment to original financing statement no. 2011 3185280 restating collateral description
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	11-05- 2015	2015 5268775	Amendment to original financing statement no. 2011 3185280 restating collateral description
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	06-29- 2016	2016 3927736	Continuation of original financing statement no. 2011 3185280

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 186 of 432

Delaware - Secretary of State	Cenveo Corporation 4500 Tiedeman Road Cleveland, OH 44144	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	09-01- 2011	2011 3393017	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 4500 Tiedeman Road Cleveland, OH 44144	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-08- 2017	06-29- 2016	2016 3927769	Continuation of original financing statement no. 2011 3393017
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Meridian Leasing Corporation Nine Parkway North, Suite 500 Deerfield, IL 60015		UCC- 1 as of 12-08- 2017	01-17- 2013	2013 0236092	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Meridian Leasing Corporation Nine Parkway North, Suite 500 Deerfield, IL 60015		UCC- 3 as of 12-08- 2017	01-31- 2013	2013 0407917	Amendment to original financing statement no. 2013 0236092 adding collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 187 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Commonwealth Capital Corp. 17755 US Hwy 19 N., Suite 400 Clearwater, FL 33764		UCC- 1 as of 12-08- 2017	02-26- 2013	2013 0741810	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459248	Blanket security interest All assets
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	09-30- 2013	2013 3958932	Amendment to original financing statement no. 2013 1459248 deleting collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	09-30- 2015	2015 4409073	Amendment to original financing statement no. 2013 1459248 deleting collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 188 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	04-13- 2017	2017 2423470	Amendment to original financing statement no. 2013 1459248 deleting collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7453873	Amendment to original financing statement no. 2013 1459248 deleting collateral
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8377022	Continuation of original financing statement no. 2013 1459248
Delaware - Secretary of State	Cenveo Corporation 201 Broad Street Stamford, CT 06901	Sun Chemical Corp. 35 Waterview Boulevard Parsippany, NY 07054		UCC- 1 as of 12-08- 2017	05-22- 2013	2013 1961003	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 189 of 432

Delaware - Secretary of State	Cenveo Corporation 201 Broad Street Stamford, CT 06901	Sun Chemical Corp. 35 Waterview Boulevard Parsippany, NY		UCC- 3 as of 12-08- 2017	02-25- 2014	2014 0734541	Amendment to original financing statement no. 2013 1961003 restating collateral
Delaware - Secretary of State	NE Opco, Inc. D/B/A National Envelope 3211 Internet Boulevard Suite 200 Frisco, TX 75034	07054 4th Generation Recycling, Inc. 369 New Britain Road Unit E Kensington, CT 06037	1	UCC- 1 as of 12-08- 2017	05-31- 2013	2013 2060888	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	4th Generation Recycling, Inc. 369 New Britain Road Unit E Kensington, CT 06037		UCC- 3 as of 12-08- 2017	09-30- 2013	2013 3805653	Amendment to original financing statement no. 2013 2060888 changing debtor's name and address
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	General Electric Credit Corporation of Tennessee PO Box 35713 Billings, MT 59107-571	1	UCC- 1 as of 12-08- 2017	06-26- 2013	2013 2464759	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green Stamford, CT 06901	General Electric Credit Corporation of Tennessee PO Box 35713 Billings, MT 59107-571		UCC- 3 as of 12-08- 2017	08-09- 2013	2013 3122257	Amendment to original financing statement no. 2013 2464759 restating collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 190 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY 10010		UCC- 1 as of 12-08- 2017	09-12- 2013	2013 3557080	All inventory pursuant to letter agreement dated August 21, 2013
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY		UCC- 3 as of 12-08- 2017	10-16- 2013	2013 4054707	Amendment to original financing statement no. 2013 3557080 changing debtor's address
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	10010 Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	10-16- 2013	2013 4072782	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 3 as of 12-08- 2017	05-30- 2014	2014 2106870	Amendment to original financing statement no. 2013 4072782 adding collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 191 of 432

Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	UCC- 3 as of 12-08- 2017	05-30- 2014	2014 2108181	Assignment of original financing statement no. 2013 4072782
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Kingsbridge Holdings, LLC 150 N. Field Dr. Suite 193 Lake Forest, IL 60045		UCC- 1 as of 12-08- 2017	11-06- 2013	2013 4377314	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Kingsbridge Holdings, LLC 150 N. Field Dr. Suite 193 Lake Forest, IL 60045	Beverly Bank & Trust Company N.A. 10258 S. Western Avenue Chicago, IL 60643	UCC- 3 as of 12-08- 2017	07-15- 2014	2014 2883304	Assignment of original financing statement no. 2013 4377314

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 192 of 432

Delaware -	Cenveo Corporation,	Kingsbridge	MB	UCC-	02-23-	2016 1092608	Assignment of
Secretary of State	200 First Stamford Place Stamford, CT 06902	Holdings, LLC 150 N. Field Dr. Suite 193 Lake Forest, IL 60045 Beverly Bank & Trust Company N.A. 10258 S. Western Avenue Chicago, IL 60643	Financial Bank, N.A. 6111 North River Road Rosemont, IL 60018	3 as of 12-08- 2017	2016		original financing statement no. 2013 4377314
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	12-18- 2013	2013 5000022	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	UCC- 3 as of 12-08- 2017	04-07- 2014	2014 1362979	Assignment of original financing statement no. 2013 5000022

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 193 of 432

Delaware - Secretary of State	Cenveo Corporation, 121 12th Ave. S Minneapolis, MN 55415	Xerox Corporation 1301 Ridgeview Bldg. 300 Lewisville, TX		UCC- 1 as of 12-08- 2017	01-21- 2014	2014 0253385	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	75057 Crown Credit Company 40 S. Washington Street New Bremen, OH 45869		UCC- 1 as of 12-08- 2017	03-03- 2014	2014 0804534	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527935	Blanket security interest All assets
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 3 as of 12-08- 2017	09-30- 2015	2015 4408984	Amendment to original financing statement no. 2014 2527935 deleting collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 194 of 432

Delaware -	Cenveo Corporation	The Bank of		UCC-	04-13-	2017 2423611	Amendment to
Secretary of State	200 First Stamford	New York		3 as of	2017		original financing
	Place	Mellon, as		12-08-			statement no. 2014
	Stamford, CT 06902	Collateral		2017			2527935 deleting
		Agent					collateral
		101 Barclay Street					
		Attn: Corporate Trust - Cenveo					
		New York, NY 10286					
Delaware -	Cenveo Corporation	The Bank of		UCC-	11-09-	2017 7454269	Amendment to
Secretary of State	200 First Stamford	New York		3 as of	2017		original financing
	Place	Mellon, as		12-08-			statement no. 2014
	Stamford, CT 06902	Collateral Agent		2017			2527935 deleting collateral
		101 Barclay					
		Street					
		Attn: Corporate					
		Trust - Cenveo					
		New York, NY					
		10286					
Delaware -	Cenveo Corporation	The Bank of		UCC-	06-26-	2014 2528099	Blanket security
Secretary of State	200 First Stamford	New York		1 as of	2014		interest
	Place	Mellon, as		12-08-			All assets
	Stamford, CT 06902	Collateral		2017			
		Agent					
		101 Barclay Street					
		Attn: Corporate					
		Trust - Cenveo					
		New York, NY 10286					

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 195 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 3 as of 12-08- 2017	09-30- 2015	2015 4409156	Amendment to original financing statement no. 2014 2528099 deleting collateral
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 3 as of 12-08- 2017	04-13- 2017	2017 2423512	Amendment to original financing statement no. 2014 2528099 deleting collateral
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7454178	Amendment to original financing statement no. 2014 2528099 deleting collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 196 of 432

Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	08-12- 2014	2014 3216769	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	UCC- 3 as of 12-08- 2017	08-12- 2014	2014 3217692	Assignment of original financing statement no. 2014 3216769
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	NMHG Financial Services, Inc. PO Box 35701 Billings, MT 59107-570		UCC- 1 as of 12-08- 2017	09-30- 2014	2014 3910254	Leased equipment
Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	12-31- 2014	2014 5305685	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 197 of 432

Delaware - Secretary of State	Cenveo Corporation, 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	Newstar Commercial Lease Funding I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116	UCC- 3 as of 12-08- 2017	12-31- 2014	2014 5306527	Assignment of original financing statement no. 2014 5305685
Delaware - Secretary of State	Cenveo Corporation 1635 Coining Drive Toledo, OH 43612	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 220 Fort Lauderdale, FL 33308		UCC- 1 as of 12-08- 2017	12-31- 2014	2014 5306733	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 1635 Coining Drive Toledo, OH 43612	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 220 Fort Lauderdale, FL 33308	United Bank 45 Glastonbury Boulevard Glastonbury, CT 06016	UCC- 3 as of 12-08- 2017	01-01- 2015	2015 0000397	Assignment of original financing statement no. 2014 5306733
Delaware - Secretary of State	Cenveo Corporation 705 Baldwin Park Blvd. City of Industry, CA 91746	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 220 Fort Lauderdale, FL 33308		UCC- 1 as of 12-08- 2017	12-31- 2014	2014 5311097	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 198 of 432

Delaware - Secretary of State	Cenveo Corporation 705 Baldwin Park Blvd. City of Industry, CA 91746	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 220 Fort Lauderdale, FL 33308	United Bank 45 Glastonbury Boulevard Glastonbury, CT 06016	UCC- 3 as of 12-08- 2017	01-01- 2015	2015 0000405	Assignment of original financing statement no. 2014 5311097
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0712983	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0713031	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0713106	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 199 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0713155	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0713221	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-08- 2017	02-19- 2015	2015 0713312	Leased equipment
Delaware - Secretary of State	Cenveo Corporation One Canterbury Green 201 Broad Street Stamford, CT 06901	Somerset Capital Group, Ltd. 612 Wheelers Farms Road Milford, CT 06461		UCC- 1 as of 12-08- 2017	07-24- 2015	2015 3205613	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 200 of 432

Delaware - Secretary of State	Cenveo Corporation One Canterbury Green	Somerset Capital Group, Ltd.		UCC- 1 as of 12-08-	07-24- 2015	2015 3205621	Leased equipment
201 Broad Str	201 Broad Street Stamford, CT 06901	612 Wheelers Farms Road Milford, CT 06461		2017			
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Newstar Financial, Inc. 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	09-30- 2015	2015 4409305	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Newstar Financial, Inc. 500 Boylston St., Suite 1200 Boston, MA 02116	ECN Financial LLC 655 Business Center Drive Horsham, PA 19044	UCC- 3 as of 12-08- 2017	12-22- 2016	2016 7978206	Assignment of original financing statement no. 2015 4409305
Delaware - Secretary of State	Cenveo Corporation 1325-1327 Highland Ridge SE Smyrna, GA 30082	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 522 Fort Lauderdale, FL 33308		UCC- 1 as of 12-08- 2017	11-09- 2015	2015 5218556	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 785 Juniata River Road Williamsburg, PA 16693	4th Generation Recycling, Inc. 6550 N. Federal Highway Ste 522 Fort Lauderdale, FL 33308		UCC- 1 as of 12-08- 2017	11-09- 2015	2015 5219075	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 201 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	WestRock Converting Company 504 Thrasher Street NW Norcross, GA 30071		UCC- 1 as of 12-08- 2017	01-22- 2016	2016 0444099	All rights to Escrow Funds defined in Escrow Agreement dated January 19, 2016
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Data Sales Co., Inc. 3450 W. Burnsville Pkwy Burnsville, MN 55337		UCC- 1 as of 12-08- 2017	03-30- 2016	2016 1886843	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Data Sales Co., Inc. 3450 W. Burnsville Pkwy Burnsville, MN 55337		UCC- 1 as of 12-08- 2017	05-20- 2016	2016 3039565	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513247	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 202 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 3 as of 12-08- 2017	04-13- 2017	2017 2423728	Amendment to original financing statement no. 2016 3513247 deleting collateral
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7454046	Amendment to original financing statement no. 2016 3513247 deleting collateral
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Newstar Equipment Finance I, LLC 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 1 as of 12-08- 2017	08-08- 2016	2016 4781876	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Radius Bank 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 3 as of 12-08- 2017	04-13- 2017	2017 2426457	Amendment to original financing statement no. 2016 4781876 changing name of secured party
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Radius Bank 500 Boylston St., Suite 1200 Boston, MA 02116		UCC- 3 as of 12-08- 2017	08-18- 2017	2017 5510968	Amendment to original financing statement no. 2016 4781876 deleting collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 203 of 432

Delaware -	Cenveo Corporation	AGFA		UCC-	11-16-	2016 7099300	Leased equipment
Secretary of State	200 First Stamford Place, 2nd Floor Stamford, CT 06902	Corporation 611 River Drive, Center		1 as of 12-08- 2017	2016		
		#3 Elmwood Park, NJ 07407					
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Toyota Industries Commercial Finance, Inc. P.O. Box 9050		UCC- 1 as of 12-08- 2017	01-27- 2017	2017 0610466	Leased equipment
		Dallas, TX 75019-9050					
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Toyota Industries Commercial Finance, Inc. P.O. Box 9050 Dallas, TX		UCC- 1 as of 12-08- 2017	01-27- 2017	2017 0610847	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place	75019-9050 Toyota Industries Commercial		UCC- 1 as of 12-08-	01-27- 2017	2017 0611639	Leased equipment
	Stamford, CT 06902	Finance, Inc. P.O. Box 9050 Dallas, TX 75019-9050		2017			
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	LaSalle Systems Leasing, Inc. 9550 W. Higgins Road, Suite 600 Rosemont, IL		UCC- 1 as of 12-08- 2017	02-02- 2017	2017 0740693	Leased equipment
		60018					

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 204 of 432

Delaware -	Cenveo Corporation	LaSalle	MB	UCC-	03-29-	2017 2042585	Partial assignment
Secretary of State	200 First Stamford Place	Systems Leasing, Inc.	Financial Bank, N.A.	3 as of 12-08-	2017		of original financing
	Stamford, CT 06902	9550 W. Higgins Road,	6111 North River Road	2017			statement no. 2017 0740693
		Suite 600 Rosemont, IL 60018	Rosemont, IL 60018				
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	LaSalle Systems Leasing, Inc. 9550 W. Higgins Road, Suite 600 Rosemont, IL 60018	MB Financial Bank, N.A. 6111 North River Road Rosemont, IL 60018	UCC- 3 as of 12-08- 2017	08-31- 2017	2017 5817751	Partial assignment of original financing statement no. 2017 0740693 and collateral change
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Wells Fargo Vendor Financial Services, LLC PO Box 35701 Billings, MT 59107		UCC- 1 as of 12-08- 2017	03-07- 2017	2017 1514162	Leased equipment
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Varilease Finance, Inc. 6340 South 3000 East, Suite 400 Salt Lake City, UT 84121		UCC- 1 as of 12-08- 2017	03-27- 2017	2017 1977849	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 205 of 432

Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Varilease Finance, Inc. 6340 South 3000 East, Suite 400 Salt Lake City, UT 84121	Prime Alliance Bank 1868 South 500 West Woods Cross, UT 84087	UCC- 3 as of 12-08- 2017	03-31- 2017	2017 2126438	Assignment of original financing statement no. 2017 1977849
Delaware - Secretary of State	Cenveo Corporation 200 First Stamford Place Stamford, CT 06902	Stonebriar Commercial Finance LLC 5601 Granite Pkwy, Suite 1350 Plano, TX 75024 Key Equipment Finance, a Division of Keybank NA 1000 S. McCaslin Blvd Superior, CO 80027		UCC- 1 as of 12-08- 2017	04-07- 2017	2017 2274592	Leased equipment
Delaware - Secretary of State	Cenveo Corporation c/o CSC 2711 Centerville Road, Suite 400 Wilmington, DE 19808	Flint Group North America Corporation 14909 N. Beck Road Plymouth, MI 48170		UCC- 1 as of 12-08- 2017	05-01- 2017	2017 2862354	Printing equipment pursuant to consignment agreement dated August 5, 2010

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 206 of 432

Delaware - Secretary of State	Cenveo Corporation c/o CSC 2711 Centerville Road, Suite 400 Wilmington, DE 19808	Day International, Inc. 14909 N. Beck Road Plymouth, MI 48170			UCC- 1 as of 12-08- 2017	05-01- 2017	2017 2862636	Printing equipment pursuant to consignment agreement dated August 18, 2008
			CADMUS DE	LAWARE, INC.				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Cadmus Delaware, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459545	Blanket security interest All assets
Delaware - Secretary of State	Cadmus Delaware, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8376623	Continuation of original financing statement no. 2013 1459545

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 207 of 432

Delaware -	Cadmus Delaware,	The Bank of		UCC-	06-26-	2014 2527109	Blanket security
Secretary of State	Inc.	New York		1 as of	2014		interest
	200 First Stamford	Mellon, as		12-08-			All assets
	Place	Collateral		2017			
	Stamford, CT 06902	Agent					
		101 Barclay					
		Street					
		Attn: Corporate					
		Trust - Cenveo					
		New York, NY					
		10286					
Delaware -	Cadmus Delaware,	The Bank of		UCC-	06-26-	2014 2527190	Blanket security
Secretary of State	Inc.	New York		1 as of	2014		interest
	200 First Stamford	Mellon, as		12-08-			All assets
	Place	Collateral		2017			
	Stamford, CT 06902	Agent					
		101 Barclay Street					
		Attn: Corporate					
		Trust - Cenveo					
		New York, NY					
		10286					
Delaware -	Cadmus Delaware,	The Bank of		UCC-	06-10-	2016 3513023	Blanket security
Secretary of State	Inc.	New York		1 as of	2016		interest
	200 First Stamford	Mellon, as		12-08-			All assets
	Place	Collateral		2017			
	Stamford, CT 06902	Agent					
		101 Barclay					
		Street, 7 East					
		New York, NY					
		10286					

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 208 of 432

		CADM	US FINANCIA	L DISTRIBUTION	N, INC.			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus Financial Distribution, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3954-4	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Financial Distribution, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-18- 2017	12-18- 2017	1712185745-0	Continuation of original financing statement no. 130417 3954-4
Virginia - State Corporation Commission	Cadmus Financial Distribution, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4070-8	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 209 of 432

Virginia - State Corporation Commission	Cadmus Financial Distribution, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4080-9	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Financial Distribution, Inc. 200 First Stamford Place Stamford, CT 06902	10286 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4199-5	Blanket security interest All assets
		CADMU	JS INTERNAT	IONAL HOLDING	GS, INC	-		
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus International Holdings, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3960-7	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 210 of 432

Virginia - State Corporation Commission	Cadmus International Holdings, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	1712185747-4	Continuation of original financing statement no. 130417 3960-7
Virginia - State Corporation Commission	Cadmus International Holdings, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4071-0	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus International Holdings, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4081-1	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 211 of 432

Virginia - State Corporation Commission	Cadmus International Holdings, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4208-6	Blanket security interest All assets
		CA	DMUS JOURN	AL SERVICES, I	NC			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus Journal Services, Inc. 2901 Byrdhill Road Richmond, VA 23228	Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ 07922			UCC- 1 as of 12-07- 2017	04-11-2011	110411 4035-1	Leased equipment
Virginia - State Corporation Commission	Cadmus Journal Services, Inc. 2901 Byrdhill Road Richmond, VA 23228	Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ 07922			UCC- 3 as of 12-07- 2017	02-11- 2016	160211 5205-9	Continuation of original financing statement no. 110411 4035-1

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 212 of 432

Virginia - State Corporation Commission	Cadmus Journal Services, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3953-2	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Journal Services, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	1712185802-2	Continuation of original financing statement no. 130417 3953-2
Virginia - State Corporation Commission	Cadmus Journal Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4072-2	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 213 of 432

Virginia - State Corporation Commission	Cadmus Journal Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4082-3	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Journal Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4207-4	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 214 of 432

	CADMUS MARKETING GROUP, INC.										
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION			
Virginia - State Corporation Commission	Cadmus Marketing Group, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3957-0	Blanket security interest All assets			
Virginia - State Corporation Commission	Cadmus Marketing Group, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-18- 2017	12-18- 2017	1712185805-8	Continuation of original financing statement no. 130417 3957-0			
Virginia - State Corporation Commission	Cadmus Marketing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4073-4	Blanket security interest All assets			

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 215 of 432

Virginia - State Corporation Commission	Cadmus Marketing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4083-5	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Marketing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4206-2	Blanket security interest All assets
			CADMUS MA	RKETING, INC.	•	•	•	
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus Marketing, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3958-2	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 216 of 432

Virginia - State Corporation Commission	Cadmus Marketing, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	1712185801-0	Continuation of original financing statement no. 130417 3958-2
Virginia - State Corporation Commission	Cadmus Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4074-6	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4084-7	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 217 of 432

Virginia - State Corporation Commission	Cadmus Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286	ADMUS PRINT	TING GROUP, IN	UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4205-0	Blanket security interest All assets
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus Printing Group, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3952-0	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus Printing Group, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-18- 2017	12-18- 2017	1712185803-4	Continuation of original financing statement no. 130417 3952-0

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 218 of 432

Virginia - State Corporation Commission	Cadmus Printing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4075-8	Blanket security interest All assets
		Trust - Cenveo New York, NY 10286						
Virginia - State Corporation Commission	Cadmus Printing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4085-9	Blanket security interest All assets
		Attn: Corporate Trust - Cenveo New York, NY 10286						
Virginia - State Corporation Commission	Cadmus Printing Group, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4204-8	Blanket security interest All assets
	-		CADMU	S UK, INC	-			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 219 of 432

Virginia - State Corporation Commission	Cadmus UK, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3955-6	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus UK, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	1712185804-6	Continuation of original financing statement no. 130417 3955-6
Virginia - State Corporation Commission	Cadmus UK, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4076-0	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 220 of 432

Virginia - State Corporation Commission	Cadmus UK, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4086-1	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus UK, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4203-6	Blanket security interest All assets
		CAD	MUS/O'KEEF	E MARKETING,	INC			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	Cadmus/O'Keefe Marketing, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3959-4	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 221 of 432

Virginia - State Corporation Commission	Cadmus/O'Keefe Marketing, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street		UCC- 1 as of 01-22- 2018	01-22- 2018	1801226328-5	Continuation of original financing statement no. 130417 3959-4
Virginia - State Corporation Commission	Cadmus/O'Keefe Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	Hartford, CT 06103 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4077-2	Blanket security interest All assets
Virginia - State Corporation Commission	Cadmus/O'Keefe Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	New York, NY 10286 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4087-3	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 222 of 432

Virginia - State Corporation Commission	Cadmus/O'Keefe Marketing, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4202-4	Blanket security interest All assets
			CDMS MANA	GEMENT, LLC				
Delaware - Secretary of State	CDMS Management, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459693	Blanket security interest All assets
Delaware - Secretary of State	CDMS Management, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8376982	Continuation of original financing statement no. 2013 1459693

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 223 of 432

Delaware - Secretary of State	CDMS Management, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527216	Blanket security interest All assets
Delaware - Secretary of State	CDMS Management, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527331	Blanket security interest All assets
Delaware - Secretary of State	CDMS Management, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513106	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 224 of 432

	CENVEO CEM, INC.										
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION			
Delaware - Secretary of State	Cenveo CEM, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459925	Blanket security interest All assets			
Delaware - Secretary of State	Cenveo CEM, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8376990	Continuation of original financing statement no. 2013 1459925			
Delaware - Secretary of State	Cenveo CEM, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527521	Blanket security interest All assets			

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 225 of 432

Delaware - Secretary of State	Cenveo CEM, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527646	Blanket security interest All assets
Delaware - Secretary of State	Cenveo CEM, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513155	Blanket security interest All assets
			CENVEO	CEM, LLC				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Cenveo CEM, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459859	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 226 of 432

Delaware - Secretary of State	Cenveo CEM, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8377006	Continuation of original financing statement no. 2013 1459859
Delaware - Secretary of State	Cenveo CEM, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527794	Blanket security interest All assets
Delaware - Secretary of State	Cenveo CEM, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2527869	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 227 of 432

Delaware - Secretary of State	Cenveo CEM, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513205	Blanket security interest All assets
			CENVEO O	MEMEE LLC				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Cenveo Omemee LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1460626	Blanket security interest All assets
Delaware - Secretary of State	Cenveo Omemee LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8377071	Continuation of original financing statement no. 2013 1460626

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 228 of 432

Delaware -	Cenveo Omemee	The Bank of			UCC-	06-26-	2014 2528206	Blanket security
Secretary of State	LLC 200 First Stamford Place Stamford, CT 06902	New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY			1 as of 12-08- 2017	2014		interest All assets
Delaware - Secretary of State	Cenveo Omemee LLC 200 First Stamford Place Stamford, CT 06902	New York, NY 10286 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2528339	Blanket security interest All assets
Delaware - Secretary of State	Cenveo Omemee LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513304	Blanket security interest All assets
	·		CENVEO SE	CRVICES, LLC				·
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 229 of 432

Colorado - Secretary of State	Cenveo Services, LLC 829 Vandemark Road Sidney, OH 45365	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-18- 2017	04-16- 2013	20132033101	Blanket security interest All assets
Colorado - Secretary of State	Cenveo Services, LLC 829 Vandemark Road Sidney, OH 45365	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	20172117136	Continuation of original financing statement no. 20132033101
Colorado - Secretary of State	Cenveo Services, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-26- 2014	20142061468	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 230 of 432

Colorado - Secretary of State	Cenveo Services, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate			UCC- 1 as of 12-18- 2017	06-26- 2014	20142061470	Blanket security interest All assets
		Trust - Cenveo New York, NY 10286						
Colorado - Secretary of State	Cenveo Services, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-18- 2017	06-10- 2016	20162052933	Blanket security interest All assets
			CENV	EO, INC				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Colorado - Secretary of State	Cenveo, Inc. 201 Broad Street Stamford, CT 06901	Hewlett- Packard Financial Services Company 420 Mountain Ave. Murray Hill, NJ 07974			UCC- 1 as of 12-18- 2017	03-29- 2011	2011F012833	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 231 of 432

Colorado - Secretary of State	Cenveo, Inc. 201 Broad Street Stamford, CT 06901	Hewlett- Packard Financial Services Company 420 Mountain Ave. Murray Hill, NJ 07974		UCC- 3 as of 12-18- 2017	01-29- 2016	20162008994	Continuation of original financing statement no. 2011F012833
Colorado - Secretary of State	Cenveo, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	Ariva Distribution Inc. 50 E. River Center Blvd. Suite 500 Covington, KY 41011		UCC- 1 as of 12-18- 2017	03-27- 2013	20132026896	Leased equipment
Colorado - Secretary of State	Cenveo, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-18- 2017	04-16- 2013	20132033095	Blanket security interest All assets
Colorado - Secretary of State	Cenveo, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	20172117160	Continuation of original financing statement no. 20132033095

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 232 of 432

Colorado - Secretary of State	Cenveo, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-26- 2014	20142061469	Blanket security interest All assets
Colorado - Secretary of State	Cenveo, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-26- 2014	20142061480	Blanket security interest All assets
Colorado - Secretary of State	Cenveo, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-18- 2017	06-10- 2016	20162052943	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 233 of 432

Delaware - Secretary of State	Cenveo, Inc. 1635 Coining Drive Toledo, OH 43612	International Paper Company 4600 Poplar Avenue Memphis, TN 38197			UCC- 1 as of 12-08- 2017	03-28- 2016	2016 1815545	Consigned paper products
			CNMW INVE	STMENTS, INC.				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	CNMW Investments, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459438	Blanket security interest All assets
Delaware - Secretary of State	CNMW Investments, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8378277	Continuation of original financing statement no. 2013 1459438

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 234 of 432

Delaware -	CNMW Investments,	The Bank of			UCC-	06-26-	2014 2528701	Blanket security
Secretary of State	Inc. 200 First Stamford Place Stamford, CT 06902	New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate			1 as of 12-08- 2017	2014		interest All assets
		Trust - Cenveo New York, NY 10286						
Delaware - Secretary of State	CNMW Investments, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2528776	Blanket security interest All assets
Delaware - Secretary of State	CNMW Investments, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513353	Blanket security interest All assets
	·	·	COLORHOUS	SE CHINA, INC.		•	·	-
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 235 of 432

Colorado - Secretary of State	Colorhouse China, Inc. 13010 County Road 6 Minneapolis, MN 55441	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-18- 2017	04-16- 2013	20132033104	Blanket security interest All assets
Colorado - Secretary of State	Colorhouse China, Inc. 13010 County Road 6 Minneapolis, MN 55441	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	20172117161	Continuation of original financing statement no. 20132033104
Colorado - Secretary of State	Colorhouse China, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-26- 2014	20142061524	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 236 of 432

Colorado - Secretary of State	Colorhouse China, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY			UCC- 1 as of 12-18- 2017	06-26- 2014	20142061531	Blanket security interest All assets
Colorado - Secretary of State	Colorhouse China, Inc. 200 First Stamford Place Stamford, CT 06902	10286 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-18- 2017	06-10- 2016	20162052945	Blanket security interest All assets
	•	COMMERCIA	L ENVELOPE	MANUFACTUR	ING CO.,	INC.	•	•
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
New York - Department of State	Commercial Envelope Manufacturing Co. Inc. 350 Wireless Blvd. Hauppauge, NY 11788	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-13- 2017	04-17- 2013	201304170206250	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 237 of 432

New York - Department of State	Commercial Envelope Manufacturing Co. Inc. 350 Wireless Blvd. Hauppauge, NY 11788	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-13- 2017	12-18- 2017	201712186529447	Continuation of original financing statement no. 201304170206250
New York - Department of State	Commercial Envelope Manufacturing Co. Inc. One Canturbury Green, 201 Broad Street Stamford, CT 06901	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY 10010		UCC- 1 as of 12-13- 2017	09-12- 2013	201309120521397	All inventory and consigned goods pursuant to letter agreement dated August 21, 2013
New York - Department of State	Commercial Envelope Manufacturing Co. Inc. 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY 10010		UCC- 3 as of 12-13- 2017	10-16- 2013	201310160580873	Amendment to original financing statement no. 201309120521397 changing debtor's address
New York - Department of State	Commercial Envelope Manufacturing Co., Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-13- 2017	06-26- 2014	201406260351606	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 238 of 432

New York - Department of State	Commercial Envelope Manufacturing Co., Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286	CRX HOI	LDING, INC.	UCC- 1 as of 12-13- 2017	06-26-2014	201406260351618	Blanket security interest All assets
JURISDICTION	DEBTOR	SECURED	ASSIGNEE	JUDGMENTS,	ТҮРЕ	DATE	FILE NO.	DESCRIPTION
		PARTY		SUITS, TAX LIENS	OF UCC	FILED		
Delaware - Secretary of State	CRX Holding, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459297	Blanket security interest All assets
Delaware - Secretary of State	CRX Holding, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8378301	Continuation of original financing statement no. 2013 1459297

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 239 of 432

Delaware - Secretary of State	CRX Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2528909	Blanket security interest All assets
		New York, NY 10286			LUCC.	04.04		
Delaware - Secretary of State	CRX Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2528941	Blanket security interest All assets
		Attn: Corporate Trust - Cenveo New York, NY 10286						
Delaware - Secretary of State	CRX Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513387	Blanket security interest All assets
			CRX	IV, LLC	-		·	
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 240 of 432

Delaware - Secretary of State	CRX JV, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum		UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1460642	Blanket security interest All assets
		Street Hartford, CT 06103					
Delaware - Secretary of State	CRX JV, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8378319	Continuation of original financing statement no. 2013 1460642
Delaware - Secretary of State	CRX JV, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529048	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 241 of 432

Delaware - Secretary of State	CRX JV, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529121	Blanket security interest All assets
Delaware - Secretary of State	CRX JV, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513429	Blanket security interest All assets
			DISCOUNT	LABELS, LLC		•	•	<u> </u>
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Indiana - Secretary of State	Discount Labels, Inc. 4115 Profit Court New Albany, IN 47150	E.I. DuPont de Nemours & Company Routes 141 & 48, Barley Mill Plaza Wilmington, DE 19805			UCC- 1 as of 12-13- 2017	05-21- 2004	200400004950296	Leased equipment

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 242 of 432

Indiana - Secretary of State	Discount Labels, Inc. 4115 Profit Court New Albany, IN 47150	E.I. DuPont de Nemours & Company Routes 141 & 48, Barley Mill Plaza Wilmington, DE 19805		UCC- 3 as of 12-13- 2017	05-11- 2009	200900003909981	Continuation of original financing statement no. 200400004950296
Indiana - Secretary of State	Discount Labels, Inc. 4115 Profit Court New Albany, IN 47150	E.I. DuPont de Nemours & Company Routes 141 & 48, Barley Mill Plaza Wilmington, DE 19805		UCC- 3 as of 12-13- 2017	01-31- 2014	201400000894826	Continuation of original financing statement no. 200400004950296
Indiana - Secretary of State	Discount Labels, LLC 4115 Profit Court New Albany, IN 47150	Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ 07922		UCC- 1 as of 12-13- 2017	10-07- 2011	201100008730779	Leased equipment
Indiana - Secretary of State	Discount Labels, LLC 4115 Profit Court New Albany, IN 47150	Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ 07922		UCC- 3 as of 12-13- 2017	08-08- 2016	201600006134043	Continuation of original financing statement no. 201100008730779

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 243 of 432

Indiana - Secretary of State	Discount Labels, LLC One Canterbury Green 201 Broad Street Stamford, CT 06901	Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ		UCC- 1 as of 12-13- 2017	10-07- 2011	201100008731134	Leased equipment
Indiana - Secretary of State	Discount Labels, LLC One Canterbury Green 201 Broad Street Stamford, CT 06901	07922 Hewlett- Packard Financial Services Company 200 Connell Drive Berkeley Heights, NJ 07922		UCC- 3 as of 12-13- 2017	08-08- 2016	201600006134154	Continuation of original financing statement no. 201100008731134
Indiana - Secretary of State	Discount Labels, LLC 4115 Profit Court New Albany, IN 47150	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-13- 2017	04-17- 2013	201300003577038	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 244 of 432

Indiana - Secretary of State	Discount Labels, LLC 4115 Profit Court New Albany, IN 47150	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 01-23- 2018	01-23- 2018	201800000650173	Continuation of original financing statement no.
Indiana - Secretary of State	Discount Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-13- 2017	06-26- 2014	201400005284288	Blanket security interest All assets
Indiana - Secretary of State	Discount Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-13- 2017	06-26- 2014	201400005284399	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 245 of 432

Indiana - Secretary of State	Discount Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286	VELOPE PROI	DUCT GROUP, L	UCC- 1 as of 12-13- 2017	06-13- 2016	201600004727211	Blanket security interest All assets
	DEDECO				1			DECENTION
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Envelope Product Group, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1460600	Blanket security interest All assets
Delaware - Secretary of State	Envelope Product Group, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-19- 2013	2013 5040473	Amendment to original financing statement no. 2013 1460600 deleting collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 246 of 432

Delaware - Secretary of State	Envelope Product Group, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103	3 a 12	CC- as of 2-08-)17	11-09- 2017	2017 7453972	Amendment to original financing statement no. 2013 1460600 deleting collateral
Delaware - Secretary of State	Envelope Product Group, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103	3 a 12	CC- as of 2-08-)17	12-18- 2017	2017 8378376	Continuation of original financing statement no. 2013 1460600
Delaware - Secretary of State	Envelope Product Group, LLC One Canterbury Green, 201 Broad Street Stamford, CT 06901	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY 10010	1 a 12	CC- as of 2-08-)17	09-12- 2013	2013 3556876	All inventory supplied by secured party
Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place, 2nd Floor Stamford, CT 06902	Southern Paper, LLC 11 Madison Avenue - 14th Floor New York, NY 10010	3 a 12	CC- as of 2-08- 017	10-16- 2013	2013 4055084	Amendment to original financing statement no. 2013 3556876 changing debtor's address

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 247 of 432

Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529204	Blanket security interest All assets
		Trust - Cenveo New York, NY 10286					
Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7454442	Amendment to original financing statement no. 2014 2529204 deleting collateral
Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529410	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 248 of 432

Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7454384	Amendment to original financing statement no. 2014 2529410 deleting collateral	
Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513494	Blanket security interest All assets	
Delaware - Secretary of State	Envelope Product Group, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 3 as of 12-08- 2017	11-09- 2017	2017 7454350	Amendment to original financing statement no. 2016 3513494 deleting collateral	
	EXPERT GRAPHICS, INC.								
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION	

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 249 of 432

Virginia - State Corporation Commission	Expert Graphics, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3956-8	Blanket security interest All assets
Virginia - State Corporation Commission	Expert Graphics, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-18- 2017	12-18- 2017	1712186446-7	Continuation of original financing statement no. 130417 3956-8
Virginia - State Corporation Commission	Expert Graphics, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4078-4	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 250 of 432

Virginia - State Corporation Commission	Expert Graphics, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4088-5	Blanket security interest All assets
Virginia - State Corporation Commission	Expert Graphics, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-07- 2017	06-13- 2016	160613 4201-2	Blanket security interest All assets
	•	GAR	AMOND/PRID	EMARK PRESS,	INC.	•		
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Maryland - State Department of Assessments and Taxation	Garamond/Pridemark Press, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	181469984	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 251 of 432

JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION	
LIGHTNING LABELS, LLC									
		New York, NY 10286							
		Attn: Corporate Trust - Cenveo							
		101 Barclay Street							
	Stamford, CT 06902	Agent							
Assessments and Taxation	200 First Stamford Place	Mellon, as Collateral			12-07- 2017			All assets	
Department of	Press, Inc.	New York			1 as of	2014	101202200	interest	
Maryland - State	Garamond/Pridemark	10286 The Bank of			UCC-	06-26-	181503900	Blanket security	
		Trust - Cenveo New York, NY							
		Attn: Corporate							
		101 Barclay Street							
	Stamford, CT 06902	Agent							
Assessments and Taxation	200 First Stamford Place	Mellon, as Collateral			12-07- 2017			All assets	
Department of	Press, Inc.	New York			1 as of	2014	101202077	interest	
Maryland - State	Garamond/Pridemark	06103 The Bank of			UCC-	06-26-	181503899	Blanket security	
		Hartford, CT							
	Stamford, CT 06901	185 Asylum Street							
Taxation	201 Broad Street	Agent			2017			101407704	
Assessments and Taxation	One Canterbury Green,	as Administrative			12-07- 2017			statement no. 181469984	
Department of	Press, Inc.	America, N.A.,			3 as of	2017		original financing	

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 252 of 432

Delaware - Secretary of State	Lightning Labels, LLC 2369 S. Trenton Way, Unit C Denver, CO 80231	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1460162	Blanket security interest All assets
Delaware - Secretary of State	Lightning Labels, LLC 2369 S. Trenton Way, Unit C Denver, CO 80231	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8381099	Continuation of original financing statement no. 2013 1460162
Delaware - Secretary of State	Lightning Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529600	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 253 of 432

Delaware - Secretary of State	Lightning Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2529881	Blanket security interest All assets
Delaware - Secretary of State	Lightning Labels, LLC 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513544	Blanket security interest All assets
	MAI	DISON/GRAHAM	COLORGRAI	PHICS INTERSTA	ATE SER	VICES, IN	NC.	
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
California - Secretary of State	Madison/Graham ColorGraphics Interstate Services, Inc. 150 N. Myers Street Los Angeles, CA 90033	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-11- 2017	04-16- 2013	13-7356830753	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 254 of 432

California - Secretary of State	Madison/Graham ColorGraphics	Bank of America, N.A.,		UCC- 3 as of	12-19- 2017	17-76234709	Continuation of original financing
	Interstate Services, Inc. 150 N. Myers Street Los Angeles, CA 90033	as Administrative Agent 185 Asylum Street Hartford, CT 06103		12-19- 2017			statement no. 13- 7356830753
California - Secretary of State	Madison/Graham ColorGraphics Interstate Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-11- 2017	06-26- 2014	14-7417681029	Blanket security interest All assets
California - Secretary of State	Madison/Graham ColorGraphics Interstate Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-11- 2017	06-26- 2014	14-7417681524	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 255 of 432

California - Secretary of State	Madison/Graham ColorGraphics Interstate Services, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-11- 2017	06-10- 2016	16-7530690219	Blanket security interest All assets
		MADISO	ON/GRAHAM	COLORGRAPHI	CS, INC.			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	NMHG Financial Services, Inc. 10 Riverview Drive Danbury, CT 06810			UCC- 1 as of 12-11- 2017	02-05- 2007	07-7101427009	Leased equipment
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	NMHG Financial Services, Inc. 10 Riverview Drive Danbury, CT 06810			UCC- 3 as of 12-11- 2017	11-02- 2011	11-72897361	Continuation of original financing statement no. 07- 7101427009
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	NMHG Financial Services, Inc. PO Box 35701 Billings, MT 59107-5701			UCC- 3 as of 12-11- 2017	11-02- 2011	11-72897362	Amendment to original financing statement no. 07- 7101427009 changing secured party's address

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 256 of 432

California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	HYG Financial Services, Inc. PO Box 35701 Billings, MT 59107-5701		UCC- 3 as of 12-11- 2017	09-21- 2016	16-75471922	Amendment to original financing statement no. 07- 7101427009 changing secured party's name
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	HYG Financial Services, Inc. PO Box 35701 Billings, MT 59107-5701		UCC- 3 as of 12-11- 2017	09-21- 2016	16-75473225	Continuation of original financing statement no. 07- 7101427009
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 1 as of 12-11- 2017	01-18- 2011	11-7258641368	Leased equipment
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	07-09- 2015	15-74739837	Amendment to original financing statement no. 11- 7258641368 restating collateral
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	12-09- 2015	15-74987328	Continuation of original financing statement no. 11- 7258641368

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 257 of 432

California -	Madison/Graham	INX		UCC-	07-18-	11-7278293526	Leased equipment
Secretary of State	ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	International Ink Co. 150 North Martingale Rd., Suite 700		1 as of 12-11- 2017	2011		
		Schaumburg, IL 60173					
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	06-11- 2012	1273168618	Amendment to original financing statement no. 11- 7278293526 restating collateral
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	11-04- 2015	15-74935952	Amendment to original financing statement no. 11- 7278293526 restating collateral
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	04-13- 2016	16-75196606	Amendment to original financing statement no. 11- 7278293526 restating collateral

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 258 of 432

California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	06-07- 2016	16-75294471	Continuation of original financing statement no. 11- 7278293526
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	11-15- 2017	17-76167665	Amendment to original financing statement no. 11- 7278293526 restating collateral
California - Secretary of State	Madison/Graham ColorGraphics, Inc. One Canterbury Green 201 Broad Street Stamford, CT 06901	INX International Ink Co. 150 North Martingale Rd., Suite 700 Schaumburg, IL 60173		UCC- 3 as of 12-11- 2017	11-16- 2017	17-76170526	Amendment to original financing statement no. 11- 7278293526 restating collateral
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 150 N. Myers Street Los Angeles, CA 90033	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-11- 2017	04-16- 2013	13-7356838094	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 259 of 432

California - Secretary of State	Madison/Graham ColorGraphics, Inc.	Bank of America, N.A.,		UCC- 3 as of	12-19- 2017	17-76234734	Continuation of original financing
	150 N. Myers Street Los Angeles, CA 90033	as Administrative Agent		12-19- 2017	2017		statement no. 13- 7356838094
		185 Asylum Street Hartford, CT 06103					
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY		UCC- 1 as of 12-11- 2017	06-26- 2014	14-7417681261	Blanket security interest All assets
California - Secretary of State	Madison/Graham ColorGraphics, Inc. 200 First Stamford Place Stamford, CT 06902	10286 The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-11- 2017	06-26- 2014	14-7417681382	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 260 of 432

California - Secretary of State	Madison/Graham ColorGraphics, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-11- 2017	06-10- 2016	16-7530689429	Blanket security interest All assets
			NASHUA CO	ORPORATION				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-18- 2017	04-16- 2013	201303335350	Blanket security interest All assets
Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-19- 2017	12-19- 2017	201742472780	Continuation of original financing statement no. 201303335350

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 261 of 432

Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-27- 2014	201412909990	Blanket security interest All assets
Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-18- 2017	06-27- 2014	201412926870	Blanket security interest All assets
Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation 59 Daniel Webster Highway Merrimack, NH 03054-4831	NMHG Financial Services, Inc. PO Box 35701 Billings, MT 59107-5701		UCC- 1 as of 12-18- 2017	07-08- 2014	201413117790	Leased equipment
Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation c/ Cenveo 200 First Stamford Place Stamford, CT 06902	Verso Corporation 6775 Lenox Center Court Suite 400 Memphis, TN 38115		UCC- 1 as of 12-18- 2017	04-21- 2015	201519344630	Consigned inventory

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 262 of 432

Massachusetts - Secretary of the Commonwealth, UCC Division	Nashua Corporation 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		RNATIONAL, INC	UCC- 1 as of 12-18- 2017	06-13- 2016	201629042280	Blanket security interest All assets
	1				T	T		
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	Nashua International, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1460238	Blanket security interest All assets
Delaware - Secretary of State	Nashua International, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-18- 2017	2017 8404263	Continuation of original financing statement no. 2013 1460238

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 263 of 432

Delaware -	Nashua International,	The Bank of			UCC-	06-26-	2014 2530285	Blanket security
Secretary of State	Inc. 200 First Stamford Place Stamford, CT 06902	New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate			1 as of 12-08- 2017	2014		interest All assets
		Trust - Cenveo New York, NY 10286						
Delaware - Secretary of State	Nashua International, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2530566	Blanket security interest All assets
Delaware - Secretary of State	Nashua International, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513601	Blanket security interest All assets
			OLD T	TSI, INC.		•		
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 264 of 432

Georgia - Cooperative Authority - Clerk of Superior Court, Fulton County	Old TSI, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 1 as of 12-13- 2017	04-17- 2013	060-2013-003346	Blanket security interest All assets
Georgia - Cooperative Authority - Clerk of Superior Court, Fulton County	Old TSI, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-20- 2017	12-20- 2017	0602017-09565	Continuation of original financing statement no. 060- 2013-003346
Georgia - Cooperative Authority- Clerk of Superior Court, Fulton County	Old TSI, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-13- 2017	06-26- 2014	060-2014-005245	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 265 of 432

Georgia - Cooperative Authority- Clerk of Superior Court, Fulton County	Old TSI, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-13- 2017	06-26- 2014	060-2014-005246	Blanket security interest All assets
Georgia - Cooperative Authority - Gwinnett County Clerk of Superior Court	Old TSI, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-13- 2017	06-13- 2016	067-2016-004035	Blanket security interest All assets
			PORT CITY	PRESS, INC.				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Maryland - State Department of Assessments and Taxation	Port City Press, Inc. 1323 Greenwood Road Baltimore, MD 21208	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-15- 2017	04-17- 2013	181469985	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 266 of 432

Maryland - State Department of Assessments and Taxation	Port City Press, Inc. 1323 Greenwood Road Baltimore, MD 21208	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 01-25- 2018	12-19- 2017	171219-1537001	Continuation of original financing statement no. 181469985
Maryland - State Department of Assessments and Taxation	Port City Press, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-15- 2017	06-26- 2014	181503902	Blanket security interest All assets
Maryland - State Department of Assessments and Taxation	Port City Press, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-15- 2017	06-26- 2014	181503906	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 267 of 432

Maryland - State Department of Assessments and Taxation	Port City Press, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286	RX JV HO	LDING, INC.	UCC- 1 as of 12-15- 2017	06-13- 2016	160613-1352006	Blanket security interest All assets
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	RX JV Holding, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459446	Blanket security interest All assets
Delaware - Secretary of State	RX JV Holding, Inc. One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-19- 2017	2017 8404305	Continuation of original financing statement no. 2013 1459446

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 268 of 432

Delaware - Secretary of State	RX JV Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2530699	Blanket security interest All assets
Delaware - Secretary of State	RX JV Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2530822	Blanket security interest All assets
Delaware - Secretary of State	RX JV Holding, Inc. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513643	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 269 of 432

			RX TECHNO	DLOGY CORP.				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Delaware - Secretary of State	RX Technology Corp. 3301 Enterprise Avenue Joplin, MO 64801	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-08- 2017	04-16- 2013	2013 1459354	Blanket security interest All assets
Delaware - Secretary of State	RX Technology Corp. 3301 Enterprise Avenue Joplin, MO 64801	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-08- 2017	12-19- 2017	2017 8404313	Continuation of original financing statement no. 2013 1459354
Delaware - Secretary of State	RX Technology Corp. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2530863	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 270 of 432

Delaware - Secretary of State	RX Technology Corp. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-08- 2017	06-26- 2014	2014 2530913	Blanket security interest All assets
Delaware - Secretary of State	RX Technology Corp. 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286		UCC- 1 as of 12-08- 2017	06-10- 2016	2016 3513684	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 271 of 432

		VAU	GHAN PRINTE	ERS INCORPORA	TED			
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Florida - Secured Transaction Registry	Vaughan Printers Incorporated One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-14- 2017	04-16- 2013	201308851675	Blanket security interest All assets
Florida - Secured Transaction Registry	Vaughan Printers Incorporated One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 3 as of 12-19- 2017	12-19- 2017	201703577653	Continuation of original financing statement no. 201308851675
Florida - Secured Transaction Registry	Vaughan Printers Incorporated 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-14- 2017	06-26- 2014	201401702501	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 272 of 432

Florida - Secured Transaction Registry	Vaughan Printers Incorporated 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286			UCC- 1 as of 12-14- 2017	06-26- 2014	201401703273	Blanket security interest All assets
Florida - Secured Transaction Registry	Vaughan Printers Incorporated 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street, 7 East New York, NY 10286			UCC- 1 as of 12-14- 2017	06-13- 2016	201607870078	Blanket security interest All assets
			VSUB HOLD	ING COMPANY				
JURISDICTION	DEBTOR	SECURED PARTY	ASSIGNEE	JUDGMENTS, SUITS, TAX LIENS	TYPE OF UCC	DATE FILED	FILE NO.	DESCRIPTION
Virginia - State Corporation Commission	VSUB Holding Company One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103			UCC- 1 as of 12-07- 2017	04-17- 2013	130417 3961-9	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 273 of 432

Virginia - State Corporation Commission	VSUB Holding Company One Canterbury Green, 201 Broad Street Stamford, CT 06901	Bank of America, N.A., as Administrative Agent 185 Asylum Street Hartford, CT 06103		UCC- 3 as of 12-19- 2017	12-19- 2017	1712195922-8	Continuation of original financing statement no. 130417 3961-9
Virginia - State Corporation Commission	VSUB Holding Company 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4079-6	Blanket security interest All assets
Virginia - State Corporation Commission	VSUB Holding Company 200 First Stamford Place Stamford, CT 06902	The Bank of New York Mellon, as Collateral Agent 101 Barclay Street Attn: Corporate Trust - Cenveo New York, NY 10286		UCC- 1 as of 12-07- 2017	06-27- 2014	140627 4089-7	Blanket security interest All assets

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 274 of 432

Virginia - State	VSUB Holding	The Bank of		UCC-	06-13-	160613 4200-0	Blanket security
Corporation	Company	New York		1 as of	2016		interest
Commission	200 First Stamford Place Stamford, CT 06902	Mellon, as Collateral Agent		12-07- 2017			All assets
		101 Barclay Street, 7 East					
		New York, NY 10286					

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 275 of 432

SCHEDULE 5.08(c)

Loan Party	Address	City	State	Book Value
Discount Labels, LLC	4115 Profit Court	New Albany	IN	\$4,433,968
Discount Labels, Inc.	4120 Profit Court	New Albany	IN	N/A
Cadmus Journal Services Inc.	2901 Byrdhill Road	Richmond	VA	\$7,673,413
Cadmus Journal Services Inc.	Lot 38 and PT 35, Block A	Richmond	VA	\$156,312
Cadmus Journal Services Inc.	Lot 38 and PT 35, Block A, Shirley Subdivision Sign #5	Richmond	VA	N/A
Cadmus Journal Services Inc.	Lot 38 and PT 35, Block A, Shirley Subdivision Sign #4	Richmond	VA	N/A
Cadmus Journal Services Inc.	Lot 13-14, Block A, Shirley Subdivision	Richmond	VA	\$5,767,714
Envelope Product Group	6302 Churchman Bypass	Indianapolis	IN	\$2,400,000
Cenveo Corp.	3301 Enterprise Ave	Joplin	МО	\$1,033,145
Cenveo Corp.	4500 Tiedeman Road	Cleveland	ОН	\$3,883,557
Envelope Product Group	Route 866	Williamsburg	PA	\$3,820,000
Cenveo Corp.	109 S. Fillmore	Amarillo	TX	\$133,190
Cenveo Corp.	5101 S. Zarzamora	San Antonio	ТХ	\$117,515
Cadums Journal Services, Inc.	3575 Hempland Rd	Easton West Hempfield Township	РА	\$5,407,175
Cenveo Corp.	555 Virginia Drive	Fort Washington	PA	N/A
Nashua Corp.	1995 Rittenhouse Road, and 1925 Slate Road	Jefferson City	TN	\$7,010,000

Owned Real Property

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 276 of 432

SCHEDULE 5.08(d)(i)

Loan Party Lessee	Address	City	State / Country	Annual Rent
Cenveo Corporation	211 East Ocean Blvd, Suite 405	Long Beach	СА	\$ (33,504.00)
Cenveo Corporation	665 Third Street, Suite 505	San Francisco	CA	\$ (96,569.48)
Cenveo Corporation	200 First Stamford Place, 2nd Floor	Stamford	СТ	\$ (1,529,500.00)
Discount Labels, LLC	1955 Corporate Square	Longwood	FL	\$ (81,874.92)
Cenveo Corporation	1325 Highlands Ridge Rd	Smyrna	GA	\$ (479,514.00)
Cenveo Corporation	13900 West 108th Street	Lenexa	KS	\$ (13,644.00)
Cenveo Corporation	5457 Twin Knolls Rd	Columbia	MD	\$ (151,503.12)
Cenveo Corporation	1300 Godward Street NE, Suite 4000	Minneapolis	MN	\$ (346,447.20)
Cenveo Corporation	140 E Ridgewood Ave, Office 415	Paramus	NJ	\$ (16,428.00)
Cenveo Corporation	5332 Broadhollow Rd	Melville	NY	\$ (67,361.76)
Cenveo Corporation	2005 City Line Road, Suite 106	Bethlehem	MD	\$ (64,679.06)
Cenveo Corporation	150 North Myers St.	Los Angeles	СА	\$ (466,272.00)
Cenveo Corporation	150 North Myers St. 2nd Floor	Los Angeles	СА	\$ (57,096.00)
Cenveo Corporation	115 - 121 North Mission Rd	Los Angeles	СА	\$ (221,880.00)
Cenveo Corporation	147 North Mission Rd	Los Angeles	СА	\$ (117,288.00)
Cenveo Corporation	91-280 Hanua Street	Honolulu	HI	\$ (204,240.00)
Cenveo Corporation	3001 N Rockwell and 2950 N Campbell Site	Chicago	IL	\$ (670,346.04)

Leased Real Property (Lessee)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 277 of 432

Loan Party Lessee	Address	City	State / Country	Annual Rent
Cenveo Corporation	4810 Williamsburg Rd	Hurlock	MD	\$ (611,078.40)
Cenveo Corporation	1323 Greenwood Rd	Pikesville	MD	\$ (1,467,207.00)
Cenveo Corporation	3010 North Industrial Rd	Kirksville	МО	\$ (120,000.00)
Cenveo Corporation	301 South 74th Street	Omaha	NE	\$ (420,875.40)
Cenveo Corporation	25 Linden Ave. East	Jersey City	NJ	\$ (689,585.00)
Cenveo Corporation	12990 S.E. Highway 212	Clackamas	OR	\$ (84,000.00)
Cenveo Corporation	Kristel Lane	Altoona	PA	\$ (1,368,512.96)
Cenveo Corporation	1001 Technology Drive	Mt. Pleasant	PA	\$ (2,198,601.24)
Cenveo Corporation	14001 Inwood Rd	Farmers Branch	TX	\$ (468,000.00)
Cenveo Corporation	6520 S. 190th St. Suite 100	Seattle (Kent)	WA	\$ (1,282,350.00)
Cenveo Corporation	3186 Pullman St.	Costa Mesa	CA	\$ (373,404.00)
Madison/Graham ColorGraphics, Inc.	3186 Pullman St.	Costa Mesa	CA	\$ 99,878.00
Cenveo Corporation	6250 S Boyle Avenue	Vernon	CA	\$ (631,855.08)
Lightning Labels, LLC	2369 S Trenton Way, Unit C & E & F	Denver	СО	\$ (104,288.28)
Cenveo Corporation	6220 Purdue Drive	Atlanta	GA	\$ (142,593.00)
Cenveo Corporation	318 Griffith Rd, Unit 1	Chicopee	МА	\$ (332,521.00)
Cenveo Corporation	101 Workman Court	Eureka	МО	\$ (1,210,885.50)
Cenveo Corporation	13507 I Circle, Omaha, NE	Omaha	NE	\$ (103,260.00)
Cenveo, Inc.	777 Westchester	White Plains	NY	\$ (11,628.00)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 278 of 432

Loan Party Lessee	Address	City	State / Country	Annual Rent
	Ave			
Cadmus Journal Services, Inc.	136 Carlin Rd	Conklin	NY	\$ (569,692.00)
Cenveo Corporation	112 S. Pierce	Amarillo	TX	\$ (39,600.00)
Cenveo Corporation	2807 N. Central Ave.	Brownsville	TX	N/A
Cenveo Corporation	2828 Trade Center Drive, Suite 140	Carrollton	TX	N/A
Cenveo Corporation	601 National Drive	Ennis	TX	\$ (1,147,041.00)
Cenveo Corporation	917 Parkway Drive	Grand Prairie	TX	N/A
Cenveo Corporation	250 S. Northwest Highway	Park Ridge	IL	\$ (343,824.00)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 279 of 432

SCHEDULE 5.08(d)(ii)

Leased Real Property (Lessor)

Loan Party Lessor	Address	City	State	Annual Rent
Cenveo Corporation	Broadway Place West, 1300 Godward Street NE, Suite 4000	Minneapolis	MN	\$355,633.30

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 280 of 432

SCHEDULE 5.08(e)

Existing Investments

- 1. 20% ownership interest in Consolidated Converting Services, Inc.
- 2. One share of Common Stock of Banta Corporation
- 3. One share of Common Stock of R.R. Donnelley & Sons Company
- 4. 35% Membership interest in Tec-Print, LLC held by Nashua Corporation
- 5. 2.96% interest in NuLabel Technologies Company
- 6. 35% ownership interest in Young Business Solutions, Inc.

SCHEDULE 5.09

Environmental Matters

None.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 282 of 432

SCHEDULE 5.13

Grantor	Subsidiary	Percentage Interest
Cenveo, Inc.	Cenveo Corporation	100%
	CNMW Investments, Inc.	100%
Cenveo Corporation	Cenveo Services, LLC	100%
	Discount Labels, LLC	100%
	Colorhouse China, Inc.	100%
	CRX JV, LLC	49%
	RX JV Holding, Inc.	100%
	Cenveo CEM, LLC	50%
	Madison/Graham ColorGraphics, Inc.	100%
	Cenveo McLaren Morris and Todd Company	45%
	Port City Press, Inc.	100%
	Cadmus Printing Group, Inc.	100%
	Cadmus Marketing Group, Inc.	100%
	Cadmus Marketing UK Limited	100%
	Cadmus International Holdings, Inc.	100%
	CDMS Management, LLC	100%
	Vaughan Printers Incorporated	100%
	Lightning Labels, LLC	100%
	Nashua Corporation	100%
	Envelope Product Group, LLC	100%
	Cenveo Omemee, LLC	100%
Discount Labels, LLC	CRX JV, LLC	50%
RX JV Holding, Inc.	CRX JV, LLC	1%
	Cenveo CEM, LLC	50%
CRX JV, LLC	CRX Holding, Inc.	100%
CRX Holding, Inc.	Rx Technology Corp.	100%
Cenveo CEM, LLC	Cenveo CEM, Inc.	100%
Cenveo CEM, Inc.	Commercial Envelope Manufacturing Co. Inc.	100%

Subsidiaries and Other Equity Investments; Loan Parties

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 283 of 432

Grantor	Subsidiary	Percentage Interest
Madison/Graham ColorGraphics, Inc.	Madison/Graham ColorGraphics Interstate Services, Inc.	100%
Cadmus Marketing Group, Inc.	Cadmus Marketing, Inc.	100%
	Cadmus/O'Keefe Marketing, Inc.	100%
	Old TSI, Inc.	100%
Cadmus Printing Group, Inc.	Cadmus Financial Distribution, Inc.	100%
	Garamond/Pridemark Press, Inc.	100%
	Cadmus Journal Services, Inc.	100%
	Expert Graphics, Inc.	100%
Cadmus Journal Services, Inc.	Cadmus Delaware, Inc.	100%
Cadmus Delaware, Inc.	Cadmus UK, Inc.	100%
Vaughan Printers Incorporated	VSUB Holding Company	100%
Nashua Corporation	Nashua International, Inc.	100%
Cadmus International Holdings, Inc.	Cadmus KnowledgeWorks International Ltd.	100%
Cenveo Omemee, LLC	3120587 Nova Scotia Company	100%

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 284 of 432

SCHEDULE 5.18

Intellectual Property Matters

None.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 285 of 432

SCHEDULE 5.21

Labor Matters

(a) Multiemployer Plans

GRAPHIC COMMUNICATIONS CONFERENCE OF THE INTERNATIONAL BROTHERHOOD OF TEAMSTERS SUPPLEMENTAL RETIREMENT AND DISABILITY FUND (GCIU – SRDF)

GRAPHIC ARTS INDUSTRY JOINT PENSION TRUST FUND (GCIJPTF)

CWA/ITU NEGOTIATED PENSION PLAN

(b) Collective bargaining agreements with the following:

Location	Union Name		
Clackamas, OR	Graphic Communications Int'l Union		
	Local 747-M		
Cleveland, OH	Int'l Association of Machinists and		
	Aerospace Workers, AFL-CIO Local		
	233		
Honolulu, HI	Graphic Communications Union, Local		
	501-M		
Jersey City, NJ	United Steel Workers, AFL-CIO, CLC,		
	Local 318		
Santa Fe Springs, CA (Envelope) and	Graphic Communications Union,		
(Nashua Label)	District Counsel No. 2, Local 388-M &		
	United Food and Commercial Workers		
	Union Local 770		
St. Louis, MO (Color Art)	Graphic Communications Union Local		
	6505-M		
Lancaster, PA (Pressroom)	Graphics Communications		
	Conference/IBT Union Local 4C		
Lancaster, PA (Bindery)	Graphics Communications		
	Conference/IBTUnion Local 4C		
Lancaster, PA (PrePress)	Communications Workers of America		
	Local 70		
Lancaster, PA (Dockworkers)	Graphics Communications		
	Conference/IBT Union Local 4C		
Lancaster, PA (Maintenance)	Graphic Communications		
	Conference/IBT Union Local 4C		
Altoona, PA	United Steel Workers International		
	Union Local 6521		
Kirksville, MO	United Steel Workers International		
	Union Local 990-Unit 2		
Indianapolis, IN	United Steel Workers, AFL-CIO, CLC,		
-	Local 1999		
Williamsburg, PA	United Steel Workers, AFL-CIO, CLC,		
	Local 2-0308		
Merrimack, NH	United Steel Workers International		
	Union Local 4-00270 International		

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 286 of 432

	Brotherhood of Electricians Union Local 490
Mount Pleasant, PA	United Steel Workers, AFL-CIO,CLC,
	Local 348, Unit 16
Smyrna, GA	GCC/Teamsters, Local 537-S
Chicopee, MA	United Steel Workers, AFL-CIO,CLC,
	Local 4-513

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 287 of 432

SCHEDULE 6.12

Guarantors

- 1. Cenveo, Inc.
- 2. Cadmus Delaware, Inc.
- 3. Cadmus Financial Distribution, Inc.
- 4. Cadmus International Holdings, Inc.
- 5. Cadmus Journal Services, Inc.
- 6. Cadmus Marketing Group, Inc.
- 7. Cadmus Marketing, Inc.
- 8. Cadmus Printing Group, Inc.
- 9. Cadmus UK, Inc.
- 10. Cadmus/O'Keefe Marketing, Inc.
- 11. CDMS Management, LLC
- 12. Cenveo CEM, Inc.
- 13. Cenveo CEM, LLC
- 14. Cenveo Omemee, LLC
- 15. Cenveo Services, LLC
- 16. CNMW Investments, Inc.
- 17. Colorhouse China, Inc.
- 18. Commercial Envelope Manufacturing Co. Inc.
- 19. CRX Holding, Inc.
- 20. CRX JV, LLC
- 21. Discount Labels, LLC
- 22. Envelope Product Group, LLC
- 23. Expert Graphics, Inc.
- 24. Garamond/Pridemark Press, Inc.
- 25. Lightning Labels, LLC
- 26. Madison/Graham ColorGraphics Interstate Services, Inc.
- 27. Madison/Graham ColorGraphics, Inc.
- 28. Nashua Corporation
- 29. Nashua International, Inc.
- 30. Old TSI, Inc.
- 31. Port City Press, Inc.
- 32. RX JV Holding, Inc.
- 33. Rx Technology Corp.
- 34. Vaughan Printers Incorporated
- 35. VSUB Holding Company

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 288 of 432

SCHEDULE 6.20

Deposit Accounts

1. Deposit Accounts:

Account Number	Depositary Institution	Account Name
8026229221	PNC Bank - Concentration	Cenveo Corporation
8026294306	PNC Bank - Cadmus	Cenveo Corporation
8026286293	PNC Bank - Lightning Labels	Cenveo Corporation
8026229213	PNC Bank - Trade Account	Cenveo Corporation
8026232703	PNC Bank - Non-trade Account	Cenveo Corporation
8026241538	PNC Bank - GAC Portland	Cenveo Corporation
8026275906	PNC Bank - Nashua Corp	Cenveo Corporation
8026275893	PNC Bank - Nashua Merrimack	Cenveo Corporation
8026256369	PNC Bank - Cadmus	Cenveo Corporation

2. Securities Accounts:

None.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 289 of 432

SCHEDULE 7.01

Equipment Loans

Origination Date	Business Unit	Lessor	Type of Equipment	Total Equipment Cost	Scheduled Monthly Payment
	INDIANAPOLIS,		W+D 202 refurbished envelope production machine and related		
3/17	IN	VARILEASE	equipment	1,330,256.50	33,256.41
8/16	CITY OF INDUSTRY, CA	ΤΟΥΟΤΑ	16 Forklifts	619,271.00	12,755.05
11/12	VIDVSVILLE MO	RADIUS	1 W+D Used 202 RE Hi Speed reel- fed envelope making Machine	1 521 000 00	25 021 85
11/13	KIRKSVILLE, MO	CAROLINA	2 Raymond 7500- R45TT forklifts with battery & chargers, 2 Raymond 4700- C60TT forklifts with batteries & chargers, 1 Raymond 4700- C50HM forklift with	1,531,000.00	25,031.85
5/15	SMYRNA, GA	HANDLING LLC	battery & charger		4,615.68
5/15	SNAVDNA CA	CAROLINA HANDLING	5 Raymond 7500- R45TT forklift with batteries & chargers, 1 Raymond 4250- C40TT forklift with battery & charger, 2 Raymond 8410- FRE60L forklifts with batteries &		5 202 (5
5/15	SMYRNA, GA JEFFERSON CITY,	LLC RADIUS	chargers 1 ETI Cohesio 26" in-line coating and printing equipment		5,293.65
5/14	TN	BANK	line HM Cross & Sons Model 521 conveyor with Accessories / DHP Bindery Equpment & Supply SP 2S1CC Head	3,155,201.00	48,874.06
12/15	CONKLIN, NY	DATA SALES CO	112A Special Heavy Duty Drill Head Standard Perfect	19,428.00	411.87
		DATA SALES	Standard Perfect Binder / Standard Three-Knife Trimmer plus		
12/15	CONKLIN, NY	CO	accessories	219,138.50	4,627.56
12/15	CONKLIN, NY	DATA SALES CO	Ricoh InfoPrint 5000 MD3/4 InkJet	996,566.00	21,127.20

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 290 of 432

	1			г	
			Solution with		
			Hunkeler equipment		
		CANNION	plus accessories		
,		CANNON	I DII		
renewed	PORT CITY	FINANCIAL	LasermaxRoll	252 20 4 00	15 000 06
2/16/2016	PRESS	SERVICES	StreamFolder	352,386.00	15,298.86
			1 Yale Veracitor		
			Cushion GC100		
			Clamp Truck / 1		
			Yale Veracitor		
		SOMERSET	Cushion Tire		
1 /1 7		CAPITAL	Forklift GC030	0.6 222 00	2 220 00
1/15	RICHMOND	GROUP	Rotator Truck	86,332.00	2,229.09
			3 ERP035VT		
			Forklifts (2 batteries		
			& 1 charger) / 2		
		SOMERSET	ERP035VT Forklifts		
		CAPITAL	(1 battery & 1		
1/15	RICHMOND	GROUP	charger)	174,320.00	3,455.02
			HP Indigo WS6800		
		HEWLETT	Printing Press with		
7/15	NEW ALBANY, IN	PACKARD	accessories	763,577.00	14,029.97
			HP Indigo WS6800		
o // =		HEWLETT	Printing Press with		
8/17	NEW ALBANY, IN	PACKARD	accessories	589,900.00	8,916.92
			HP Indigo WS6600		
		HEWLETT	Printing Press with		
7/15	DENVER, CO	PACKARD	accessories	675,050.00	13,319.22
			Mark Andy		
			Performance Series		
	Express Labels -	RADIUS	P5-13, 8 Color		
6/14	Longwood FL	BANK	Converting System	623,876.00	8,786.23
	Express Labels -	RADIUS	Mark Andy Retrofit		
7/15	Longwood FL	BANK	for P5-13	101,350.00	3,079.44
	SAN ANTONIO,	WELLS	1 Ricoh C9110		
7/16	TX	FARGO	Branding Set	144,305.50	2,815.84
		HEWLETT			
6/15	ST. LOUIS	PACKARD	HP Indigo 7800	450,599.00	8,266.03
			HP Scitex FB10000		
			Industrial Printing		
			Press, Latex 3000		
		HEWLETT	Ink Printer, 1200dpi,		
7/14	ST. LOUIS	PACKARD	6 color	1,998,059.86	44,336.95
		HEWLETT	HP Indigo 12000		
10/17	ST. LOUIS	PACKARD	plus accessories	1,150,719.00	29,256.23
		WELLS	1 Ricoh C9110		
7/16	AMARILLO, TX	FARGO	Branding Set	227,952.00	4,556.77
			2 Heidelberg		
	COLORGRAPHICS	RADIUS	XL105-8+LXUV		
12/14	- Los Angeles, CA	BANK	Printing Presses	2,518,562.88	45,522.77
		HEWLETT			
1/16	Kent Wa	PACKARD	HP Indigo 7800	487,999.00	8,981.43
		HYG			
		FINANCIAL	1 2016 Hyster		
1/17	VERNON, CA	SERVICES	E35ZRS2 Forklift	35,223.73	846.34

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 291 of 432

	-				
		HYG			
1 /1 7		FINANCIAL	1 2016 Hyster	24,220,00	000.04
1/17	VERNON, CA	SERVICES	E35HSD3 Forklift	34,239.00	803.34
			1 MCS Falcon		
			Copier/ 2 Perfect		
			match flowmaster /		
2/17		WELLS	6 cameras &	1 40 00 5 00	0.014.05
2/17	VERNON, CA	FARGO	accessories	140,095.00	2,914.27
		HYG	1 Used Hyster 2014		
0/15	MT. PLEASANT,	FINANCIAL	Forklift Unit Model	57 500 00	1 1 67 46
8/15	PA	SERVICES	S120FT	57,500.00	1,167.46
			2 Raymond Swing Reach Trucks model		
			9600CSR30T w/		
	MT. PLEASANT,	LaSalle			
1/17	PA	Solutions	battery, charger and install	107,569.00	2,489.00
1/1/	WILLIAMSBURG,	Solutions	IIIstaii	107,309.00	2,489.00
10/16	PA	Ryder	1 Yard Tractor	103,716.00	2,597.87
10/10	ГA	Nyuci	Ricoh InfoPrint	105,710.00	2,371.01
			5000 MD3/4 InkJet		
			Solution with		
			Hunkeler equipment		
2/17	CONKLIN, NY	RICOH	plus accessories	1,005,076.00	32,172.48
2/1/		LaSalle	2 Ricoh C9110	1,005,070.00	52,172.40
6/17	HURLOCK, MD	Solutions	Presses	354,161.40	8,174.00
0/1/	VALA CYNWYD,	Bolutions	1105505	55 1,101.10	0,171.00
	PA / WESTFIELD,				
	MA / NEWARK,				
	DE / MOUNT		Konica Minolta		
	PLEASANT, PA /	HEWLETT	BizHub 754e Laser		
10/14	CHICOPEE, MA	PACKARD	Print/Copy/Scan/Fax	78,633.28	2,088.04
	ST. GEORGE,				,
	UT/BERESFORD,		Konica Minolta		
	SD/HURLOCK,	HEWLETT	BizHub 754e Laser		
11/14	MD	PACKARD	Print/Copy/Scan/Fax	44,334.00	1,159.83
			Konica Minolta		
	INDIANAPOLIS,	HEWLETT	BizHub 754e Laser		
4/15	IN	PACKARD	Print/Copy/Scan/Fax	16,528.07	426.42
	OMAHA, NE /				
	JACKSONVILLE,				
	FL /				
	MINNEAPOLIS,				
	MN / HURLOCK,				
	MD / SALT LAKE				
	CITY, UT /		Konica Minolta		
	DALLAS, TX /	HEWLETT	BizHub Laser		
7/15	CLACKAMAS, OR	PACKARD	Print/Copy/Scan/Fax	63,302.07	1,682.99
	AMARILLO, TX /				
	LANCASTER, PA /				
	VERNON, CA /				
	DANVILLE, IL /				
	MT PLEASANT,				
	PA / VERNON, CA	• • • • • • •	Konica Minolta		
o // -	/ JACKSONVILLE,	HEWLETT	BizHub Laser		
8/15	FL	PACKARD	Print/Copy/Scan/Fax	114,866.56	3,007.31

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 292 of 432

			IZ : D: 1 1 007		
			Konico Bizhub 227		
			Laser		
			Printer/Scan/Copy,		
10/15	PORTLAND OR &	HEWLETT	22ppm, 250GB HD,	(1.505.1.1	1 550 10
12/15	PIKES MD	PACKARD	Duplex, GbNIC	61,527.14	1,573.40
	KENT WA,				
	JERSEY CITY NJ,		Konica Minolta		
	DENVER CO,	HEWLETT	BizHub Laser		
4/16	ENNIS TX	PACKARD	Print/Copy/Scan/Fax	70,160.00	1,906.91
		HYG	2 Used Yale		
	CITY OF	FINANCIAL	Forklifts -		
8/16	INDUSTRY, CA	SERVICES	GC070VX		437.97
		HYG			
		FINANCIAL	1 Used Yale Forklift		
8/16	KIRKSVILLE, MO	SERVICES	- ERP030TH		190.57
		YALE			
6/14	VERNON, CA	CAROLINAS	Yale Forklift	23,755.30	349.33
			1 Ricoh Pro		
	SAN ANTONIO,		C7110X Branding		
11/15	TX	RICOH	Set		1,354.15
		PITNEY	DM400C Digital		
12/16	STAMFORD, CT	BOWES	Mailing System		297.00
	MINNEAPOLIS,				
10/17	MN	Kodak	Nexpress Equipment		8,880.00
			Intellilink		
			Subscription &		
			Maintenance		
		PITNEY	Agreement for 2 2-		
12/16	VERNON, CA	BOWES	R760 metering bases		137.00
	,		2015 Ford E350 Cut		
			away Cab plus		
10/17	AMARILLO, TX	Penske	maintenance	19,941.00	720.00
	,		1 New Nissan	,	
	WILLIAMSBURG,		CF80LPS Cushion		
7/14	PA	de lage landen	Forklift		628.97
			1 Xeikon 8600		
			digital press / 1		
			Xeikon 8800 digital		
		RADIUS	press / 1 Xeikon		
7/16	Hurlock / St. Louis	BANK	8500 digital press	1,706,500.00	29,235.76
//10	Turiock / St. Louis	DANK	5500 digital piess	1,700,500.00	47,433.10

SCHEDULE 7.02

Existing Indebtedness

- 1. 4% Notes Documents
- 2. 6% Notes Documents
- 3. 6% Unsecured Notes Documents
- 4. 8.5% Notes Documents
- 5. Second Amended and Restated Credit Agreement
- 6. To the extent constituting Indebtedness, Indebtedness represented by the letters of credit listed on Schedule 2.13 hereto.
- 7. To the extent constituting Indebtedness, Indebtedness represented by the following capital leases and equipment loans:

Origination Date	Business Unit	Lessor	Type of Equipment	Total Equipment Cost	Scheduled Monthly Payment
			W+D 202		
			refurbished		
			envelope production		
	INDIANAPOLIS,		machine and related		
3/17	IN	VARILEASE	equipment	1,330,256.50	33,256.41
	CITY OF				
8/16	INDUSTRY, CA	TOYOTA	16 Forklifts	619,271.00	12,755.05
			1 W+D Used 202		
			RE Hi Speed reel-		
		RADIUS	fed envelope making		
11/13	KIRKSVILLE, MO	BANK	Machine	1,531,000.00	25,031.85
			2 Raymond 7500-		
			R45TT forklifts with		
			battery & chargers,		
			2 Raymond 4700-		
			C60TT forklifts with		
			batteries & chargers,		
		CAROLINA	1 Raymond 4700-		
		HANDLING	C50HM forklift with		
5/15	SMYRNA, GA	LLC	battery & charger		4,615.68
			5 Raymond 7500-		
			R45TT forklift with		
			batteries & chargers,		
			1 Raymond 4250-		
			C40TT forklift with		
			battery & charger, 2		
			Raymond 8410-		
		CAROLINA	FRE60L forklifts		
		HANDLING	with batteries &		
5/15	SMYRNA, GA	LLC	chargers		5,293.65
	JEFFERSON CITY,	RADIUS	1 ETI Cohesio 26"		
5/14	TN	BANK	in-line coating and	3,155,201.00	48,874.06

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 294 of 432

-		1		-	
			printing equipment line		
			HM Cross & Sons		
			Model 521 conveyor		
			with Accessories /		
			DHP Bindery		
			Equpment & Supply		
			SP 2S1CC Head		
		DATA SALES	112A Special Heavy		
12/15	CONKLIN, NY	CO	Duty Drill Head	19,428.00	411.87
			Standard Perfect		
			Binder / Standard		
			Three-Knife		
		DATA SALES	Trimmer plus		
12/15	CONKLIN, NY	CO	accessories	219,138.50	4,627.56
			Ricoh InfoPrint	.,	,
			5000 MD3/4 InkJet		
			Solution with		
		DATA SALES	Hunkeler equipment		
12/15	CONKLIN, NY	СО	plus accessories	996,566.00	21,127.20
		CANNON			
renewed	PORT CITY	FINANCIAL	LasermaxRoll		
2/16/2016	PRESS	SERVICES	StreamFolder	352,386.00	15,298.86
			1 Yale Veracitor		,
			Cushion GC100		
			Clamp Truck / 1		
			Yale Veracitor		
		SOMERSET	Cushion Tire		
		CAPITAL	Forklift GC030		
1/15	RICHMOND	GROUP	Rotator Truck	86,332.00	2,229.09
			3 ERP035VT		
			Forklifts (2 batteries		
			& 1 charger) / 2		
		SOMERSET	ERP035VT Forklifts		
		CAPITAL	(1 battery & 1		
1/15	RICHMOND	GROUP	charger)	174,320.00	3,455.02
			HP Indigo WS6800		
		HEWLETT	Printing Press with		
7/15	NEW ALBANY, IN	PACKARD	accessories	763,577.00	14,029.97
			HP Indigo WS6800		
		HEWLETT	Printing Press with		
8/17	NEW ALBANY, IN	PACKARD	accessories	589,900.00	8,916.92
			HP Indigo WS6600		
		HEWLETT	Printing Press with		10.017.55
7/15	DENVER, CO	PACKARD	accessories	675,050.00	13,319.22
			Mark Andy		
		DADUIC	Performance Series		
r 13 A	Express Labels -	RADIUS	P5-13, 8 Color		0.704.00
6/14	Longwood FL	BANK	Converting System	623,876.00	8,786.23
	Express Labels -	RADIUS	Mark Andy Retrofit	101.050.00	0.070.44
7/15	Longwood FL	BANK	for P5-13	101,350.00	3,079.44
- 11 -	SAN ANTONIO,	WELLS	1 Ricoh C9110	144 005 50	0.015.01
7/16	TX	FARGO	Branding Set	144,305.50	2,815.84
<i>c</i> 11 m		HEWLETT		450 500 00	0.011.00
6/15	ST. LOUIS	PACKARD	HP Indigo 7800	450,599.00	8,266.03

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 295 of 432

			•		
			HP Scitex FB10000		
			Industrial Printing		
			Press, Latex 3000		
		HEWLETT	Ink Printer, 1200dpi,		
7/14	ST. LOUIS	PACKARD	6 color	1,998,059.86	44,336.95
.,		HEWLETT	HP Indigo 12000	_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,
10/17	ST. LOUIS	PACKARD	plus accessories	1,150,719.00	29,256.23
10/17	51. LOUIS	WELLS	1 Ricoh C9110	1,130,717.00	27,230.23
7/16	AMARILLO, TX	FARGO	Branding Set	227,952.00	4,556.77
//10	AWAKILLO, IA	TAKOO	2 Heidelberg	227,952.00	4,550.77
	COLORGRAPHICS	RADIUS	XL105-8+LXUV		
12/14		BANK	Printing Presses	2 5 1 9 5 6 2 9 9	15 500 77
12/14	- Los Angeles, CA		Finding Flesses	2,518,562.88	45,522.77
1/10	V and W/a	HEWLETT	UD L. 1' 7900	497 000 00	0.001.42
1/16	Kent Wa	PACKARD	HP Indigo 7800	487,999.00	8,981.43
		HYG	1.001 (11)		
1 (1 =		FINANCIAL	1 2016 Hyster		046.04
1/17	VERNON, CA	SERVICES	E35ZRS2 Forklift	35,223.73	846.34
		HYG			
		FINANCIAL	1 2016 Hyster		
1/17	VERNON, CA	SERVICES	E35HSD3 Forklift	34,239.00	803.34
			1 MCS Falcon		
			Copier/ 2 Perfect		
			match flowmaster /		
		WELLS	6 cameras &		
2/17	VERNON, CA	FARGO	accessories	140,095.00	2,914.27
		HYG	1 Used Hyster 2014		
	MT. PLEASANT,	FINANCIAL	Forklift Unit Model		
8/15	PA	SERVICES	S120FT	57,500.00	1,167.46
			2 Raymond Swing		
			Reach Trucks model		
			9600CSR30T w/		
	MT. PLEASANT,	LaSalle	battery, charger and		
1/17	PA	Solutions	install	107,569.00	2,489.00
	WILLIAMSBURG,			,	,
10/16	PA	Ryder	1 Yard Tractor	103,716.00	2,597.87
			Ricoh InfoPrint		_,
			5000 MD3/4 InkJet		
			Solution with		
			Hunkeler equipment		
2/17	CONKLIN, NY	RICOH	plus accessories	1,005,076.00	32,172.48
		LaSalle	2 Ricoh C9110	1,000,070.00	22,172.10
6/17	HURLOCK, MD	Solutions	Presses	354,161.40	8,174.00
0/17	VALA CYNWYD,	Solutions	1100000	55 1,101.70	0,174.00
	PA / WESTFIELD,				
	MA / NEWARK,				
	DE / MOUNT		Konica Minolta		
		HEWLETT	BizHub 754e Laser		
10/14	PLEASANT, PA /			78 632 20	2 000 04
10/14	CHICOPEE, MA	PACKARD	Print/Copy/Scan/Fax	78,633.28	2,088.04
	ST. GEORGE,		Varias Marsha		
	UT/BERESFORD,		Konica Minolta		
11/14	SD/HURLOCK,	HEWLETT	BizHub 754e Laser	44 224 00	1 150 02
11/14	MD	PACKARD	Print/Copy/Scan/Fax	44,334.00	1,159.83
	DELANASSI		Konica Minolta		
	INDIANAPOLIS,	HEWLETT	BizHub 754e Laser		
4/15	IN	PACKARD	Print/Copy/Scan/Fax	16,528.07	426.42

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 296 of 432

	MD / SALT LAKE CITY, UT /		Konica Minolta		
	-				
7/15	DALLAS, TX / CLACKAMAS, OR	HEWLETT PACKARD	BizHub Laser Print/Copy/Scan/Fax	63,302.07	1,682.99
//15	AMARILLO, TX /	TACKARD	Time/Copy/Scan/Tax	03,302.07	1,082.99
	LANCASTER, PA /				
	VERNON, CA /				
	DANVILLE, IL /				
	MT PLEASANT,				
	PA / VERNON, CA		Konica Minolta		
	/ JACKSONVILLE,	HEWLETT	BizHub Laser		
8/15	FL	PACKARD	Print/Copy/Scan/Fax	114,866.56	3,007.31
			Konico Bizhub 227		
			Laser		
	PORTLAND OR &	HEWLETT	Printer/Scan/Copy, 22ppm, 250GB HD,		
12/15	PIKES MD	PACKARD	Duplex, GbNIC	61,527.14	1,573.40
12/10	KENT WA,	1110111111		01,02/11	1,070110
	JERSEY CITY NJ,		Konica Minolta		
	DENVER CO,	HEWLETT	BizHub Laser		
4/16	ENNIS TX	PACKARD	Print/Copy/Scan/Fax	70,160.00	1,906.91
		HYG	2 Used Yale		
	CITY OF	FINANCIAL	Forklifts -		
8/16	INDUSTRY, CA	SERVICES	GC070VX		437.97
		HYG FINANCIAL	1 Used Yale Forklift		
8/16	KIRKSVILLE, MO	SERVICES	- ERP030TH		190.57
0/10		YALE	- LRI 050111		170.57
6/14	VERNON, CA	CAROLINAS	Yale Forklift	23,755.30	349.33
			1 Ricoh Pro		
	SAN ANTONIO,		C7110X Branding		
11/15	TX	RICOH	Set		1,354.15
		PITNEY	DM400C Digital		
12/16	STAMFORD, CT	BOWES	Mailing System		297.00
10/17	MINNEAPOLIS,	17.11	Number		0 000 00
10/17	MN	Kodak	Nexpress Equipment		8,880.00
			Intellilink Subscription &		
			Maintenance		
		PITNEY	Agreement for 2 2-		
12/16	VERNON, CA	BOWES	R760 metering bases		137.00
			2015 Ford E350 Cut		
			away Cab plus		
10/17	AMARILLO, TX	Penske	maintenance	19,941.00	720.00
			1 New Nissan		
	WILLIAMSBURG,		CF80LPS Cushion		
7/14	PA	de lage landen	Forklift		628.97
		DADUIG	1 Xeikon 8600		
			digital press / 1		
7/16	Hurlock / St. Louis	RADIUS BANK	Xeikon 8800 digital	1,706,500.00	29,235.76

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 297 of 432

	press / 1 Xeikon	
	8500 digital press	

SCHEDULE 7.09

Restrictions

None.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 299 of 432

SCHEDULE 11.02

Administrative Agent's Office, Certain Addresses for Notices

If to Administrative Agent:

Borrowings, Conversion, and Notices to:

Mr. Robert Q. Mahoney Bank of America, N.A. 185 Asylum Street Hartford, Connecticut 06103 Email: robert.mahoney@baml.com

Financials and any other notices to:

Mr. Robert Q. Mahoney Bank of America, N.A. 185 Asylum Street Hartford, Connecticut 06103 Email: robert.mahoney@baml.com

If to the Borrower to:

Cenveo Corporation Treasurer 200 First Stamford Place, 2nd Floor Stamford, Connecticut 06902 Phone: (203) 595-3046 Fax: (203) 595-3085 Email: Ayman.Zameli@cenveo.com

If to Holdings to:

Cenveo Corporation Treasurer 200 First Stamford Place, 2nd Floor Stamford, Connecticut 06902 Phone: (203) 595-3046 Fax: (203) 595-3085 Email: Ayman.Zameli@cenveo.com

EXHIBIT B to <u>RATIFICATION AND AMENDMENT AGREEMENT</u>

Initial Budget

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 301 of 432

Project Chelsea - Weekly Cash Flow Forecast

Dollars US	D		-	_			_	-		_	-	_			
		Week 1 2/3	Week 2 2/10	Week 3 2/17	Week 4 2/24	Week 5 3/3	Week 6 3/10	Week 7 3/17	Week 8 3/24	Week 9 3/31	Week 10 4/7	Week 11 4/14	Week 12 4/21	Week 13 4/28	Total 2/3
		2/9	2/16	2/23	3/2	3/9	3/16	3/23	3/30	4/6	4/13	4/20	4/27	5/4	5/4
	CONSOLIDATED	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast
s	AR Collection	\$ 15,870	\$ 15,679	\$ 15,299	\$ 18,258	\$ 23,237	\$ 31,505	\$ 26,347	\$ 26,347	\$ 26,391	\$ 26,183	\$ 27,404	\$ 28,056	\$ 22,636	\$ 303,213
Cash Receipts	Non-Trade	119	49	31	22	40	18	531	607	733	1,221	733	1,466	1,059	6,627
Rec	Waste	303	303		607	76	607		683	-	326	407	489		3,800
!	Total Cash Receipts	16,293	16,031	15,330	18,886	23,353	32,129	26,878	27,637	27,124	27,730	28,544	30,010	23,695	313,641
Operating Disbursements	Vendor Payments	(11,374)	(11,169)	(14,124)	(14,279)	(16,546)	(15,758)	(11,069)	(7,201)	(18,775)	(15,812)	(15,087)	(16,683)	(13,809)	(181,685)
me	Payroll	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(15,000)	(2,700)	(73,000)
ias.	Taxes	(100)	(400)	(1,000)	(600)	(100)	(400)	(1,000)	(600)	-	(100)	(1,300)	(500)	(200)	(6,300)
Ē	Benefits	(1,708)	(1,108)	(608)	(1,708)	(1,708)	(1,108)	(708)	(1,708)	(1,708)	(1,708)	(1,192)	(1,608)	(1,808)	(18,384)
lsi	Rent	(1,300)	-	-	(1,400)	-	-	-	(100)	(1,300)	-	-	-	(1,400)	(5,500)
I gl	Other Payments	(500)	(500)	(400)	(800)	(500)	(500)	(500)	(500)	(800)	(500)	(500)	(500)	(800)	(7,300)
Ę.	Transfers to Foreign Subsidiaries	-	-	-	(1,100)	-	-	-	-	(1,100)	-	-	-	(1,100)	(3,300)
era	Plant Closures	(250)	(250)	(250)	(250)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(7,750)
do	Total Operating Cash Disbursements	(17,932)	(19,126)	(19,082)	(31,137)	(22,304)	(24,216)	(16,727)	(21,859)	(27,133)	(24,570)	(21,528)	(35,041)	(22,566)	(303,219)
	OPERATING CASH FLOW	(1,639)	(3,096)	(3,751)	(12,250)	1,049	7,914	10,151	5,778	(9)	3,160	7,016	(5,031)	1,129	10,421
1	Proceeds for Issuance of Debt													-	
Debt Service	Principal														
la ja	Interest and Financing Fees	(5,380)	(10)	(500)	(939)	-		-	(968)	(500)		-	-	(944)	(9,241)
- s	Total Financing Cash Receipts / Disbursements		(10)	(500)	(939)				(968)	(500)				(944)	(9,241)
	Total Financing Cash Receipts / Disbursement	(5,380)	(10)	(500)	(939)		-		(908)	(500)	-	-	-	(944)	(9,241)
-	LSC Vendor Reimbursement	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	-	-	-	-	-	8,000
QP LSC / TSA	LSC / QP Related AP	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	-	-	-		(9,000)
\sim	LSC Customer Collection	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	-	-	-	-	-	16,000
ITSC	LSC / QP Related AR	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	-	-	-	-	(18,000)
£	QP Sale Proceeds	-	-	-	-	-	-	-	-	-	-	-	-	-	-
-	Total QP Disbursements / Adjustments	-	-	-	-	-	-	-	-	(3,000)	-	-	-	-	(3,000)
1	Restructuring Professional Fees			(160)	(160)			(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	(7,820)
	Utility Deposit	(1,400)		-	-	-		-	-	-	-	-	-		(1,400)
Other	DIP Proceeds	50,000		50,000	-				-	-		<u> </u>	-	<u> </u>	100,000
õ	Total Other Disbursements / Adjustments	48,600	-	49,840	(160)	-		(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	90,780
1	NET CASH FLOW	41,580	(3,105)	45,589	(13,349)	1,049	7,914	9,991	4,315	(3,934)	745	4,591	(6,116)	(310)	88,960
1	Borrowing Base	146,000	146,000	146,000	147,953	147,953	147,953	147,953	159,976	159,976	159,976	159,976	159,976	149,957	149,957
	Professional Fee Reserve	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)
	Net Borrowing Base	130,900	130,900	130,900	132,853	132,853	132,853	132,853	144,876	144,876	144,876	144,876	144,876	134,857	134,857
	Starting ABL Outstanding	120,164	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	120,164
Ę.	ABL Draw / (Repayment)	(41,580)	3,105	(45,589)	13,349	(1,049)	(7,914)	(9,991)	(4,315)	3,934	(745)	(4,591)	6,116	310	(88,960)
idi	Ending ABL Outstanding	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	31,204	31,204
Liquidity	Plus : L/Cs Outstanding	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600
-	Borrowings + L/Cs	95.184	98,289	52,700	66.050	65.000	57,086	47,095	42,780	46,714	45,969	41,378	47,494	47,804	47.804
	Ending ABL Availability	35,716	32,611	78,200	66,803	67,853	75,767	85,758	102,096	98,162	98,907	103,498	97,382	87,053	87,053
	Less: Outstanding Check Float	(2,000)	(4,000)	(6,000)	(7,000)	(7,000)	(7,000)	(7,000)	(7,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)
	Plus: Unrestricted Cash	-	(4,000)	-	-	-	-	-	-	-	(4,000)	-	(1,000)	(4,000)	(4,000)
	LIQUIDITY	33,716	28,611	72,200	59,803	60,853	68,767	78,758	95,096	94,162	94,907	99,498	93,382	83,053	83,053
	· · · ·	56,715	10,011	,_ 00	2,000	20,000	20,707	. 0,700	10,070	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,,,,,,,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	50,002	50,000	00,000

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 302 of 432

EXHIBIT C to <u>RATIFICATION AND AMENDMENT AGREEMENT</u>

Initial 9-Month Budget

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 303 of 432

Project Chelsea - Weekly Cash Flow Forecast

ollars USD															
	I	Week 1 2/3	Week 2 2/10	Week 3 2/17	Week 4 2/24	Week 5 3/3	Week 6 3/10	Week 7 3/17	Week 8 3/24	Week 9 3/31	Week 10 4/7	Week 11 4/14	Week 12 4/21	Week 13 4/28	Week 14 5/5
I	CONSOLIDATED	2/9 Forecast	2/16 Forecast	2/23 Forecast	3/2 Forecast	3/9 Forecast	3/16 Forecast	3/23 Forecast	3/30 Forecast	4/6 Forecast	4/13 Forecast	4/20 Forecast	4/27 Forecast	5/4 Forecast	5/11 Forecast
	AR Collection	\$ 15,870	\$ 15,679	\$ 15,299	\$ 18,258	\$ 23,237	\$ 31,505 \$	26,347	\$ 26,347 \$	26,391 \$	26,183	\$ 27,404	\$ 28,056 \$	\$ 22,636 \$	25,568
h pts	Non-Trade	119	49	31	22	¢ 23,237 40	18	531	¢ 20,517 ¢ 607	733	1,221	733	1,466	1,059	733
Cash Receipt	Waste	303	303	-	607	76	607	-	683	-	326	407	489	-	-
R	Total Cash Receipts	16,293	16,031	15,330	18,886	23,353	32,129	26,878	27,637	27,124	27,730	28,544	30,010	23,695	26,301
nts	Vendor Payments	(11,374)	(11,169)	(14,124)	(14,279)	(16,546)	(15,758)	(11,069)	(7,201)	(18,775)	(15,812)	(15,087)	(16,683)	(13,809)	(15,096)
mei	Payroll	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(15,000)	(2,700)	(5,700)
I-BS	Taxes	(100)	(400)	(1,000)	(600)	(100)	(400)	(1,000)	(600)	-	(100)	(1,300)	(500)	(200)	(100)
ng	Benefits	(1,708)	(1,108)	(608)	(1,708)	(1,708)	(1,108)	(708)	(1,708)	(1,708)	(1,708)	(1,192)	(1,608)	(1,808)	(808)
Dis	Rent	(1,300)	-	-	(1,400)	-	-	-	(100)	(1,300)	-	-	-	(1,400)	-
Bu	Other Payments	(500)	(500)	(400)	(800)	(500)	(500)	(500)	(500)	(800)	(500)	(500)	(500)	(800)	(500)
rati	Transfers to Foreign Subsidiaries Plant Closures	(250)	(250)	(250)	(1,100) (250)	(750)	- (750)	(750)	- (750)	(1,100) (750)	(750)	(750)	(750)	(1,100) (750)	-
Operating Disbursements	Total Operating Cash Disbursements	(17,932)	(19,126)	(19,082)	(31,137)	(22,304)	(24,216)	(16,727)	(21,859)	(27,133)	(24,570)	(21,528)	(35,041)	(22,566)	(750) (22,953)
	OPERATING CASH FLOW	(1,639)	(3,096)	(3,751)	(12,250)	1,049	7,914	10,151	5,778	(9)	3,160	7,016	(5,031)	1,129	3,347
		())	(,,,,,,	(,, , ,	())		<i>r</i>	., .			.,			, ,	- ,-
. 8	Proceeds for Issuance of Debt	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Service	Principal Interest and Financing Fees	- (5,380)	(10)	(500)	- (939)	-	-	-	- (968)	- (500)	-	-	-	- (944)	-
Se.	Total Financing Cash Receipts / Disbursement	(5,380)	(10)	(500)	(939)				(968)	(500)				(944)	
1	LSC Vendor Reimbursement	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000						
5	LSC / OP Related AP	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	-	-	-	-	-
2	LSC Customer Collection	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	(1,000)	-	-	-	-	-
2	LSC / QP Related AR	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	-	-	-	-	-
QP LSC / TSA	QP Sale Proceeds	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	-	-	-	-	-
ð	Total QP Disbursements / Adjustments									(3,000)					
i i	Restructuring Professional Fees			(160)	(160)			(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	(605)
.	Utility Deposit	(1,400)	-	(100)	(100)	-	-	(100)	(495)	(423)	(2,413)	(2,423)	(1,005)	(493)	(003)
Other	DIP Proceeds	50,000		50,000			-	-		-	-	-		-	
5	Total Other Disbursements / Adjustments	48,600		49,840	(160)			(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	(605)
	NET CASH FLOW	41,580	(3,105)	45,589	(13,349)	1,049	7,914	9,991	4,315	(3,934)	745	4,591	(6,116)	(310)	2,742
1	Borrowing Base	146,000	146,000	146,000	147,953	147,953	147,953	147,953	159,976	159,976	159,976	159,976	159,976	149,957	149,957
	Professional Fee Reserve	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)
	Net Borrowing Base	130,900	130,900	130,900	132,853	132,853	132,853	132,853	144,876	144,876	144,876	144,876	144,876	134,857	134,857
	Starting ABL Outstanding	120,164	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	31,204
È	ABL Draw / (Repayment)	(41,580)	3,105	(45,589)	13,349	(1,049)	(7,914)	(9,991)	(4,315)	3,934	(745)	(4,591)	6,116	310	(2,742)
Liquidity	Ending ABL Outstanding	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	31,204	28,462
Liq	Plus : L/Cs Outstanding	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600
	Borrowings + L/Cs	95,184	98,289	52,700	66,050	65,000	57,086	47,095	42,780	46,714	45,969	41,378	47,494	47,804	45,062
	Ending ABL Availability	35,716	32,611	78,200	66,803	67,853	75,767	85,758	102,096	98,162	98,907	103,498	97,382	87,053	89,795
	Less: Outstanding Check Float Plus: Unrestricted Cash	(2,000)	(4,000)	(6,000) -	(7,000) -	(7,000) -	(7,000) -	(7,000) -	(7,000) -	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)
	LIQUIDITY	33,716	28,611	72,200	59,803	60,853	68,767	78,758	95,096	94,162	94,907	99,498	93,382	83,053	85,795

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 304 of 432

Project Chelsea - Weekly Cash Flow Forecast

Dolla	ars USD	

Veck 10 Veck 10 Veck 10 Veck 10 Veck 10 Veck 10 Veck 20 Veck 20 <t< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>_</th><th>Oollars USD</th></t<>														_	Oollars USD
State State Order Format Format <th>6 Week 27 8/4</th> <th></th>	6 Week 27 8/4														
Unstantial Invasit Procest	8/4 8/10														
Point Trade 244 570 773 114 477 226 224 570 1156 475 4135 455 Wase Total Cash Receips 18,991 22,647 22,352 25,342 25,568 26,555 24,524 23,868 22,047 21,051 24,054 21,051 24,054 22,047 21,051 24,054 22,047 21,051 24,054 22,047 21,051 24,054 22,047 21,051 24,054 22,047 21,051 24,051<														CONSOLIDATED	ļ
Mon-Trade 244 570 773 114 970 220 224 115 975 1156 1156 <td></td> <td></td> <td>*</td> <td>A 01551</td> <td>*</td> <td>*</td> <td></td> <td></td> <td>A 04750 4</td> <td>*</td> <td>*</td> <td></td> <td>10.055</td> <td></td> <td></td>			*	A 01551	*	*			A 04750 4	*	*		10.055		
Note Case registry in the case is a serie is serie is a serie is															ts _
Note Case registry in the case is a serie is serie is a serie is				495		578		326				570			ast ceir
Mark Under Psymetis (12,244) (12,372) (11,051) (14,000) (12,374) (12,374) (14,000) (12,374) (12,374) (14,000) (12,374)				22.047		24 524		25 568				23.614			Rec
Protection (1,000)	20,000	1,001	21,001	==,017	20,000	21,021	20,000	20,000	20,212	20,002	21,170	20,011	10,071		· · ·
Protection (1,000)															nts
Protection (1,000)						(5,700)				(5,700)					me
Rent Image: constraint of the symmetry interval symmetry inter						-									rse
OPERATING CASE FLOW (1,808) 84 (212) (1,005) 5.079 5.316 (1,71) (6,52) 742 (3,890) (2,12) (1,01) Proceeds for issuance of Deht Principal Meret and Financing Fees Total Financing Cash Receipts / Disbursement			(1,708)	(1,192)	(708)			(908)	(908)			(1,608)	(1,108)		Ē
OPERATING CASE FLOW (1,809) 84 (212) (1,005) 5.079 5.316 (1,171) (6,525) 742 (3,890) (2,126) Proceeds for issuance of Debt			-	-	-			-	-			-	-		Dis
OPERATING CASE FLOW (1,809) 84 (212) (1,005) 5.079 5.316 (1,171) (6,525) 742 (3,890) (2,126) Proceeds for issuance of Debt			(500)	(500)	(500)		(500)	(500)	(500)		(800)	(500)	(500)		g
OPERATING CASE Flow (1,809) 84 (212) (1,005) 5.079 5.316 (1,171) (6,525) 742 (3,890) (2,126) Proceeds for issuance of Debt Principal Interest and Planning Fees Total Financing Case Receipts / Disbursement			-	-	-		-	-			-	-	-		ati
OPERATING CASE FLOW (1,808) 84 (212) (1,005) 5.079 5.316 (1,71) (6,52) 742 (3,890) (2,12) (1,01) Proceeds for issuance of Deht Principal Meret and Financing Fees Total Financing Cash Receipts / Disbursement														—	per
Proceeds for Issuance of Debt	,671) (21,503)	(28,671)	(23,677)	(25,936)	(23,121)	(31,049)	(21,528)	(20,252)	(20,163)	(24,537)	(21,709)	(23,529)	(20,698)	Total Operating Cash Disbursements	õ
Principal Interst and Financing Fees Total Financing Fees	,817) 7,398	(3,817)	(2,126)	(3,890)	742	(6,525)	(1,171)	5,316	5,079	(1,005)	(212)	84	(1,808)	OPERATING CASH FLOW	ľ
Principal Interstand Princing Pees Total Pinancing Cash Receipts / Disbursement .			-	-	-	-	-	-	-	-	-	-	-	Proceeds for Issuance of Debt	. I.
Total Financing Cash Receipts / Disbursement · <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td>-</td> <td></td> <td>-</td> <td></td> <td>ić pt</td>										-	-		-		ić pt
Total Financing Cash Receipts / Disbursement · <td>,039) -</td> <td>(1,039)</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>(1,443)</td> <td>-</td> <td>-</td> <td>-</td> <td>(987)</td> <td>-</td> <td>-</td> <td></td> <td>erv</td>	,039) -	(1,039)	-	-	-	-	(1,443)	-	-	-	(987)	-	-		erv
Verticity LSC / QP Related AP .<	,039) -	(1,039)	-	-	-	-	(1,443)	-	-	-	(987)	-	-	Total Financing Cash Receipts / Disbursement	s
Vert Construction Image: construction														ISC Vondor Poimbursomont	- E
Total QP Disbursements / Adjustments ·		-	-	-	-	-	-	-	-	-	-	-	-		<
Total QP Disbursements / Adjustments ·		-	-	-	-	-	-	-	-	-	-	-	-		TS
Total QP Disbursements / Adjustments ·		-	-	-	-	-	-	-	-	-	-	-	-		2
Total QP Disbursements / Adjustments ·		-	-	-	-	-	-	-	-	-	-	-	-		LS I
Pope Restructuring Professional Fees (2,025) (1,315) (495) (335) (2,260) (735) (410) (520) (1,765) (855) (725) (495) Utility DipProceeds . <	<u> </u>		<u> </u>	<u> </u>		-						-	<u> </u>	QP Sale Proceeds	5
Utility Deposit		-	-	-	-	-	-	-	-	-		-	-	Total QP Disbursements / Adjustments	
OP DIP Proceeds <th< td=""><td>(495) (725)</td><td>(495)</td><td>(725)</td><td>(855)</td><td>(1,765)</td><td>(520)</td><td>(410)</td><td>(735)</td><td>(2,260)</td><td>(335)</td><td>(495)</td><td>(1,315)</td><td>(2,025)</td><td>Restructuring Professional Fees</td><td></td></th<>	(495) (725)	(495)	(725)	(855)	(1,765)	(520)	(410)	(735)	(2,260)	(335)	(495)	(1,315)	(2,025)	Restructuring Professional Fees	
Processional Fee Reserve 149,957 149,957 135,421 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,321 120,32		-	-	-		-	-			-	-		-	Utility Deposit	5
NET CASH FLOW (3,833) (1,231) (1,694) (1,340) 2,819 4,581 (3,024) (7,045) (1,023) (4,745) (2,851) (5,350) NET CASH FLOW (3,833) (1,231) (1,694) (1,340) 2,819 4,581 (3,024) (7,045) (1,023) (4,745) (2,851) (5,350) Borrowing Base 149,957 149,957 135,421 120,321 <td< td=""><td><u> </u></td><td></td><td><u> </u></td><td><u> </u></td><td></td><td></td><td><u> </u></td><td><u> </u></td><td><u> </u></td><td></td><td>-</td><td>-</td><td>-</td><td>—</td><td>the</td></td<>	<u> </u>		<u> </u>	<u> </u>			<u> </u>	<u> </u>	<u> </u>		-	-	-	—	the
Borrowing Base Professional Fee Reserve 149,957 (15,100) 135,421 (15,100) 135,421 (120,321 135,421 (120,321	(495) (725)	(495)	(725)	(855)	(1,765)	(520)	(410)	(735)	(2,260)	(335)	(495)	(1,315)	(2,025)	Total Other Disbursements / Adjustments	Ĭ
Professional Fee Reserve (15,100) (15,1	,350) 6,673	(5,350)	(2,851)	(4,745)	(1,023)	(7,045)	(3,024)	4,581	2,819	(1,340)	(1,694)	(1,231)	(3,833)	NET CASH FLOW	· · ·
Professional Fee Reserve (15,100) (15,1	,421 135,421	135,421	135,421	135,421	135,421	135,421	135,421	135,421	135,421	135,421	135,421	149,957	149,957	Borrowing Base	- E
Key Starting ABL Outstanding 28,462 32,294 33,525 35,219 36,560 33,741 29,160 32,184 39,229 40,253 44,997 47,848 ABL Draw (Repayment) 3,833 1,231 1,664 1,340 (2,819) (4,581) 3,024 7,045 1,023 4,745 2,851 5,350 Ending ABL Outstanding 32,294 33,525 35,219 36,560 33,741 29,160 32,184 39,229 40,253 44,997 4,745 5,350 Plus: L/Cs Outstanding 16,600															
ABL Draw / (Repayment) 3,833 1,231 1,694 1,340 (2,819) (4,581) 3,024 7,045 1,023 4,745 2,851 5,350 Ending ABL Outstanding 32,294 33,525 35,219 36,660 33,741 29,160 32,184 39,229 40,253 44,997 47,848 53,199 Plus: L/Cs Outstanding 16,600 1),321 120,321	120,321	120,321	120,321	120,321	120,321	120,321	120,321	120,321	120,321	120,321	134,857	134,857	Net Borrowing Base	
ABL Draw / (Repayment) 3,833 1,231 1,694 1,340 (2,819) (4,581) 3,024 7,045 1,023 4,745 2,851 5,350 Ending ABL Outstanding 32,294 33,525 35,219 36,560 33,741 29,160 32,184 39,229 40,253 44,997 47,848 53,199 Plus: L/Cs Outstanding 16,600 1	,848 53,199	47,848	44,997	40,253	39,229	32,184	29,160	33,741	36,560	35,219	33,525	32,294	28,462	Starting ABL Outstanding	
Borrowings + L/Cs <u>48,894</u> 50,125 51,819 53,160 50,341 45,760 48,784 55,829 56,853 61,597 64,448 69,799	(6,673)	5,350	2,851	4,745	1,023	7,045	3,024	(4,581)	(2,819)	1,340	1,694	1,231	3,833	ABL Draw / (Repayment)	£
Borrowings + L/Cs <u>48,894</u> 50,125 51,819 53,160 50,341 45,760 48,784 55,829 56,853 61,597 64,448 69,799		53,199	47,848	44,997	40,253	39,229	32,184			36,560	35,219	33,525	32,294		idi
Borrowings + L/Cs <u>48,894</u> 50,125 51,819 53,160 50,341 45,760 48,784 55,829 56,853 61,597 64,448 69,799														0	nbi
															-
							· · · · ·		· · · · · · · · · · · · · · · · · · ·						
Less: Outstanding Check Float (4,000)	,000) (4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)		
Plus: Unrestricted Cash		-	-	-	-	-	-	-	-	-	-	-	-	Plus: Unrestricted Cash	
LIQUIDITY 81,962 80,732 64,502 63,162 65,981 70,562 67,537 60,492 59,469 54,724 51,873 46,523	,523 53,195	46.523	51,873	54.724	59,469	60.492	67.537	70.562	65.981	63.162	64,502	80.732	81.962	LIQUIDITY	1

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 305 of 432

Project Chelsea - Weekly Cash Flow Forecast

s USD		Week 28	Week 29	Week 30	Week 31	Week 32	Week 33	Week 34	Week 35	Week 36	Week 37	Week 38	Total
		8/11	8/18	8/25	9/1	9/8	9/15	9/22	9/29	10/6	10/13	10/20	2/3
_		8/17	8/24	8/31	9/7	9/14	9/21	9/28	10/5	10/12	10/19	10/26	10/26
	CONSOLIDATED	Forecast	Forecas										
s	AR Collection	\$ 20,560	\$ 22,955	\$ 19,900	\$ 20,313	\$ 21,551	\$ 20,230	\$ 26,836	\$ 21,800	\$ 21,720	\$ 21,084	\$ 22,914	\$ 862
Receipts	Non-Trade	495	495	413	495	1,239	248	248	1,909	955	636	2,307	23
ece	Waste	83	330	495		248		743	-	239	80	159	
~	Total Cash Receipts	21,138	23,781	20,808	20,808	23,038	20,478	27,827	23,709	22,914	21,800	25,380	896
ş	Vendor Payments	(16,105)	(15,926)	(14,216)	(14,982)	(15,439)	(8,857)	(10,726)	(17,863)	(16,194)	(16,770)	(15,474)	(54
e e	Payroll	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(11,033)	(21
ser	Taxes	(400)	(1,400)	(200)	-	(500)	(900)	(500)	(200)	(100)	(1,300)	(500)	(1
Operating Disbursements	Benefits	(808)	(1,608)	(1,208)	(1,408)	(808)	(1,325)	(1,908)	(1,808)	(1,208)	(692)	(1,808)	(5
	Rent	-	-	(100)	(1,300)	-	-	(100)	(1,300)	-	-		(1
20	Other Payments	(500)	(500)	(500)	(500)	(500)	(500)	(500)	(500)	(500)	(500)	(500)	(2
	Transfers to Foreign Subsidiaries	-	-	-	(1,100)	-	-	-	(1,100)	-	-	-	
E.	Plant Closures	(750)	(750)	(250)	<u> </u>		<u> </u>		-		-		(2
5	Total Operating Cash Disbursements	(24,263)	(22,884)	(27,474)	(21,990)	(22,946)	(14,282)	(24,733)	(25,470)	(23,702)	(21,962)	(29,315)	(89
1	OPERATING CASH FLOW	(3,124)	897	(6,666)	(1,182)	91	6,196	3,093	(1,761)	(788)	(162)	(3,935)	
Т	Proceeds for Issuance of Debt	-							-			.	
	Principal	-	-	-	-	-	-	-	-	-	-	-	
	Interest and Financing Fees	<u> </u>	-	(1,026)		<u> </u>	<u> </u>	(500)	(1,044)	<u> </u>	-	·	(1
1	Total Financing Cash Receipts / Disbursement	-		(1,026)		-	-	(500)	(1,044)	-		-	(1
1	LSC Vendor Reimbursement	-	-	-	-	-	-	-	-	-	-		
	LSC / QP Related AP	-	-	-	-	-	-	-	-	-	-	· i	
	LSC Customer Collection	-	-	-	-	-	-	-	-	-	-	-	1
	LSC / QP Related AR	-	-	-	-	-	-	-	-	-	-	-	(1
	QP Sale Proceeds		<u> </u>				-	-			-	- i	
	Total QP Disbursements / Adjustments	-				-			-	•		·	(
Т	Restructuring Professional Fees	(3,870)	(955)	(495)	(535)	(3,060)	(775)	(3,285)	(695)	(1,765)	(855)	(14,522)	(5
	Utility Deposit		-	-	-	-	-	-	-	-	-		
	DIP Proceeds	<u> </u>	-			<u> </u>	-		-		-		10
	Total Other Disbursements / Adjustments	(3,870)	(955)	(495)	(535)	(3,060)	(775)	(3,285)	(695)	(1,765)	(855)	(14,522)	4
1	NET CASH FLOW	(6,994)	(58)	(8,187)	(1,717)	(2,969)	5,421	(692)	(3,500)	(2,553)	(1,017)	(18,457)	3
Т	Borrowing Base	135,421	135,421	135,093	135,093	135,093	135,093	135,093	135,093	135,093	135,093	135,093	13
	Professional Fee Reserve	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(1
	Net Borrowing Base	120,321	120,321	119,993	119,993	119,993	119,993	119,993	119,993	119,993	119,993	119,993	11
	Starting ABL Outstanding	46,526	53,520	53,578	61,766	63,482	66,451	61,030	61,721	65,221	67,774	68,791	12
1	ABL Draw / (Repayment)	6,994	58	8,187	1,717	2,969	(5,421)	692	3,500	2,553	1,017	18,457	(3
	Ending ABL Outstanding	53,520	53,578	61,766	63,482	66,451	61,030	61,721	65,221	67,774	68,791	87,248	8
	Plus : L/Cs Outstanding	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	1
	Borrowings + L/Cs	70,120	70,178	78,366	80,082	83,051	77,630	78,321	81,821	84,374	85,391	103,848	10
	Ending ABL Availability	50,201	50,143	41,628	39,911	36,943	42,364	41,672	38,172	35,619	34,602	16,146	1
	Less: Outstanding Check Float	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	
I.	Plus: Unrestricted Cash				-	-	-		-				
1	LIQUIDITY	46.201	46,143	37,628	35,911	32,943	38,364	37,672	34,172	31,619	30.602	12,146	1

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 306 of 432

EXHIBIT 2

DIP Term Loan Agreement

EXECUTION VERSION

SENIOR-SECURED SUPER-PRIORITY PRIMING

DEBTOR-IN-POSSESSION CREDIT AGREEMENT

Dated as of February 5, 2018

among

CENVEO CORPORATION, as Borrower,

CENVEO, INC., as a Guarantor,

WILMINGTON SAVINGS FUND SOCIETY, FSB, as Administrative Agent

and

the Lenders party hereto

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 308 of 432

TABLE OF CONTENTS

Page

ARTICLE I

DEFIN	ITIONS AND ACCOUNTING TERMS	1
1.01	Defined Terms	1
1.02	Other Interpretive Provisions	28
	Accounting Terms	
1.04	Rounding	29
1.05	Times of Day	29
1.06	Uniform Commercial Code	29

ARTICLE II

AMOU	JNT AND TERMS OF CREDIT	
2.01	Commitments	
2.02	Loans	
2.03	Borrowing Procedure	
2.04	Evidence of Debt; Repayment of Loans	
2.05	Fees and Payments	
2.06	Interest on Loans	
2.07	Termination and Reduction of Commitments	
2.08	Interest Elections	
2.09	Optional and Mandatory Prepayments of Loans	
2.10	Payments Generally; Pro Rata Treatment; Sharing of Set-offs	
2.11	Defaulting Lenders	
2.12	DIP Priority Account	

ARTICLE III

TAXES, YIELD PROTECTION AND ILLEGALITY	
--	--

38
41
41
42
43
43
44
•

ARTICLE IV

CONDI	TIONS PRECEDENT TO BORROWINGS	.44
	Conditions of Initial Borrowing	
4.02	Conditions to All Borrowings	.47

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 309 of 432

ARTICLE V

REPRE	SENTATIONS AND WARRANTIES	. 49
5.01	Existence, Qualification and Power	40
5.02	Authorization; No Contravention	
5.02	Governmental Authorization; Other Consents	
5.03	Binding Effect	
5.04	Financial Statements; No Material Adverse Effect; No Internal Control Event	
5.06	Litigation	
5.07	No Default	
5.08	Ownership of Property; Liens; Investments	
5.09	Environmental Compliance	
5.10	Insurance	
5.11	Taxes	. 53
5.12	ERISA Compliance	. 53
5.13	Use of Proceeds	. 53
5.14	Subsidiaries; Equity Interests; Loan Parties	. 54
5.15	Margin Regulations; Investment Company Act	. 54
5.16	Disclosure	. 55
5.17	Compliance with Laws	. 55
5.18	Intellectual Property; Licenses, Etc	. 55
5.19	Initial Budget	. 55
5.20	Casualty, Etc	. 56
5.21	Labor Matters	. 56
5.22	Collateral Documents	. 57
5.23	Anti-Terrorism Laws	. 57
5.24	Orders	. 58
5.25	Bankruptcy Matters	. 58
5.26	Material Contracts	. 58

ARTICLE VI

AFFI	RMATIVE COVENANTS	
6.01	Financial Statements	
6.02	Certificates; Other Information	
6.03	Notices	
6.04	Payment of Obligations	
6.05	Preservation of Existence, Etc	
6.06	Maintenance of Properties	
6.07	Maintenance of Insurance	
6.08	Compliance with Laws	
6.09	Books and Records	
6.10	Inspection Rights	
6.11	Use of Proceeds	
6.12	Covenant to Guarantee Obligations and Give Security	
6.13	Compliance with Environmental Laws	
6.14	Preparation of Environmental Reports	
6.15	Further Assurances	
6.16	Compliance with Terms of Leaseholds	

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 310 of 432

6.17	Additional Chapter 11 Reporting	69
	Material Contracts	
6.19	Restructuring Obligations	70
	Collateral Monitoring and Reporting	
	Post-Closing Actions	

ARTICLE VII

NEGA	TIVE COVENANTS	71
7.01	Liens	71
7.02	Indebtedness	73
7.03	Investments	73
7.04	Fundamental Changes	74
7.05	Dispositions	74
7.06	Restricted Payments	75
7.07	Change in Nature of Business	76
7.08	Transactions with Affiliates	76
7.09	Burdensome Agreements	76
7.10	Use of Proceeds	76
7.11	Financial Covenants	76
7.12	Capital Expenditures	76
7.13	Amendments of Organization Documents or Material Contracts	77
7.14	Accounting Changes	
7.15	Prepayments, Etc. of Indebtedness	77
7.16	Amendment, Etc. of Related Documents and Indebtedness	77
7.17	Holding Company	77
7.18	Sanctions	
7.19	Chapter 11 Cases	

ARTICLE VIII

8.01	Events of Default	
	Remedies upon Event of Default	
	Code and Other Remedies	
8.04	Application of Funds	
8.05	Receiver	Error! Bookmark not defined.
8.06	Deficiency	

ARTICLE IX

9.01	Appointment and Authority	87
	Rights as a Lender	
	Exculpatory Provisions	
	Reliance by Administrative Agent	
9.05	Delegation of Duties	89
9.06	Resignation of Administrative Agent	89

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 311 of 432

9.07	Non-Reliance on Administrative Agent and Other Lenders	
	[Reserved]	
	Administrative Agent May File Proofs of Claim	
	Collateral and Guaranty Matters	
	Credit Bidding	
	Tax Indemnity	
	-	

ARTICLE X

CONTINUING GUARANTY		
10.01	Guaranty	
	Rights of Lenders	
	Certain Waivers	
10.04	Obligations Independent	
10.05	Subrogation	
10.06	Termination; Reinstatement	
10.07		
10.08		
	Condition of Borrower	
10.10	Additional Guarantor Waivers and Agreements	
10.11	Keepwell	

ARTICLE XI

MISCELLANEOUS		
11.01	Amendments, Etc	
11.02	Notices; Effectiveness; Electronic Communications	
11.03	No Waiver; Cumulative Remedies	
11.04	Expenses; Indemnity; Damage Waiver	
11.05	Payments Set Aside	
11.06	Successors and Assigns	
11.07	Treatment of Certain Information; Confidentiality	
11.08	Right of Setoff	
11.09	Interest Rate Limitation	
11.10	Counterparts; Effectiveness	
11.11	Survival of Representations and Warranties	
11.12	Severability	
11.13	Replacement of Lenders	
11.14	Governing Law; Jurisdiction; Etc	
11.15	Waiver of Jury Trial	
11.16	No Advisory or Fiduciary Responsibility	
11.17	Judgment Currency	
11.18	USA PATRIOT Act Notice	
11.19	Acknowledgement and Consent to Bail-In of EEA Financial Institutions	
11.20	ENTIRE AGREEMENT	
11.21	[Reserved]	
11.22	Subject to Orders	

SCHEDULES

1.02	_	Material Contracts
	-	
1.05	-	Accounts
2.01		Commitments and Applicable Percentages
5.08(b)	-	Existing Liens
5.08(c)	-	Owned Real Property
5.08(d)(i)	-	Leased Real Property (Lessee)
5.08(d)(ii)	-	Leased Real Property (Lessor)
5.08(e)	-	Existing Investments
5.09	-	Environmental Matters
5.14	-	Subsidiaries and Other Equity Investments; Loan Parties
5.18	-	Intellectual Property Matters
5.21	-	Labor Matters
5.26	-	Material Contracts
6.12	-	Guarantors
6.20	-	Deposit Accounts
6.21	-	Post-Closing Actions
6.23(c)	-	Milestones
7.02	-	Existing Indebtedness
7.09		Restrictions
11.02	-	Administrative Agent's Office, Certain Addresses for Notices
11.06	-	Processing and Recordation Fees

EXHIBITS

Form of

A-1	-	Notice of Borrowing
A-2	-	Notice of Conversion/Continuation
A-2	-	Notice of Prepayment
В	-	Note
С	-	Compliance Certificate
D	-	Assignment and Assumption
E	-	[Reserved]
F	-	[Reserved]
G	-	Joinder Agreement
Н	-	Pro Forma Adjustment Compliance Certificate
Ι	-	Tax Status Certificate
J	-	Initial Budget

SENIOR-SECURED SUPER-PRIORITY PRIMING DEBTOR-IN-POSSESSION CREDIT AGREEMENT

This SENIOR-SECURED SUPER-PRIORITY PRIMING DEBTOR-IN-POSSESSION CREDIT AGREEMENT (as amended, amended and restated, supplemented or otherwise modified from time to time in accordance with the terms hereof, this "<u>Agreement</u>") is entered into as of February 5, 2018, among CENVEO CORPORATION, a Delaware corporation, as borrower and a debtor-in-possession in the Chapter 11 Cases (as defined below) (the "<u>Borrower</u>"), CENVEO, INC., a Colorado corporation ("<u>Holdings</u>"), as a Guarantor (as defined below) and a debtor-in-possession in the Chapter 11 Cases, each lender from time to time party hereto (collectively, the "<u>Lenders</u>" and individually, a "<u>Lender</u>") and Wilmington Savings Fund Society, FSB, as Administrative Agent (as defined below).

PRELIMINARY STATEMENTS:

WHEREAS, on February 5, 2018 (the "<u>Petition Date</u>") each Loan Party filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code (collectively, the "<u>Chapter 11 Cases</u>") in the United States Bankruptcy Court for the Southern District of New York;

WHEREAS, each Loan Party is continuing in the possession of its assets and in the operation and management of its business and properties as a debtor-in-possession pursuant to section 1107(a) and 1108 of the Bankruptcy Code;

WHEREAS, the Bankruptcy Court has entered an Interim Order pursuant to which the Lenders may make post-petition term loans to the Borrower pursuant to a credit facility in an aggregate principal amount not to exceed \$100,000,000 (the "Facility"), as set forth in the Interim Order;

WHEREAS, the Loan Parties have asked the Lenders to provide the Borrower with Loans under the Facility for the purposes described herein, subject to the Orders;

WHEREAS, the Lenders are willing to extend such credit to the Borrower, subject to the terms and conditions in this Agreement and the Orders; and

WHEREAS, the Obligations of the Borrower are guaranteed by the Guarantors and, subject to the Carve-Out, secured by Liens on the Collateral, in each case, as set forth in, and subject to, the Loan Documents and the Orders.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto covenant and agree as follows.

ARTICLE I

DEFINITIONS AND ACCOUNTING TERMS

1.01 <u>Defined Terms</u>.

As used in this Agreement, the following terms shall have the meanings set forth below:

"<u>4% Notes</u>" means the 4% Senior Secured Notes due 2021 issued by the Borrower pursuant to the 4% Notes Documents.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 314 of 432

"<u>4% Notes Documents</u>" means the 4% Notes Indenture, the 4% Notes and all other agreements, instruments and other documents pursuant to which the 4% Notes have been or will be issued or otherwise setting forth the terms of the 4% Notes.

"<u>4% Notes Indenture</u>" means the Indenture and Note Purchase Agreement, dated as of June 10, 2016 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, Holdings, the guarantors named therein, Allianz US High Yield Fund and Allianz Income And Growth Fund, as purchasers, each noteholder party hereto and the 4% Notes Indenture Trustee.

"<u>4% Notes Indenture Trustee</u>" means The Bank of New York Mellon, in its capacities as trustee and collateral agent under the 4% Notes Indenture.

"<u>6% Notes</u>" means the 6.000% Senior Priority Secured Notes due 2019 issued by the Borrower pursuant to the 6% Notes Documents.

"<u>6% Notes Documents</u>" means the 6% Notes Indenture, the 6% Notes and all other agreements, instruments and other documents pursuant to which the 6% Notes have been or will be issued or otherwise setting forth the terms of the 6% Notes.

"<u>6% Notes Indenture</u>" means the Indenture, dated as of June 26, 2014 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, the guarantors party thereto and the 6% Notes Indenture Trustee.

"<u>6% Notes Indenture Trustee</u>" means The Bank of New York Mellon, in its capacities as trustee and collateral agent under the 6% Notes Indenture.

"<u>6% Unsecured Notes</u>" means the 6.000% Senior Notes due 2024 issued by the Borrower pursuant to the 6% Unsecured Notes Documents.

"<u>6% Unsecured Notes Documents</u>" means the 6% Unsecured Notes Indenture, the 6% Unsecured Notes and all other agreements, instruments and other documents pursuant to which the 6% Unsecured Notes have been or will be issued or otherwise setting forth the terms of the 6% Unsecured Notes.

"<u>6% Unsecured Notes Indenture</u>" means the Indenture, dated as of June 10, 2016 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, the guarantors party thereto and the 6% Unsecured Notes Indenture Trustee.

"<u>6% Unsecured Notes Indenture Trustee</u>" means The Bank of New York Mellon, in its capacity as trustee under the 6% Unsecured Notes Indenture.

"<u>8.5% Notes</u>" means the 8.500% Junior Priority Secured Notes due 2022 issued by the Borrower pursuant to the 8.5% Notes Documents.

"<u>8.5% Notes Documents</u>" means the 8.5% Notes Indenture, the 8.5% Notes and all other agreements, instruments and other documents pursuant to which the 8.5% Notes have been or will be issued or otherwise setting forth the terms of the 8.5% Notes.

"<u>8.5% Notes Indenture</u>" means the Indenture, dated as of June 26, 2014 (as amended, supplemented or otherwise modified from time to time), by and among the Borrower, the guarantors named therein and the 8.5% Notes Indenture Trustee.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 315 of 432

"<u>8.5% Notes Indenture Trustee</u>" means The Bank of New York Mellon, in its capacities as trustee and collateral agent under the 8.5% Notes Indenture.

"<u>ABL DIP Credit Agreement</u>" means that certain Ratification and Amendment Agreement, dated as of February 5, 2018, by and among the Borrower, the Guarantors, Bank of America, N.A., as administrative agent and collateral agent, and the lenders party thereto, ratifying and amending the Prepetition ABL Credit Agreement, including Exhibit A and all other exhibits, annexes and attachments thereto, in each case, as amended, supplemented or otherwise modified from time to time in accordance with the terms and conditions set forth therein and the Orders.

"<u>ABL DIP Facility</u>" means the revolving loan and letter of credit facility established under the ABL DIP Credit Agreement in an aggregate principal amount not exceeding \$190,000,000.

"<u>ABL DIP Loans</u>" means loans and other credit extensions made under the ABL DIP Credit Agreement.

"<u>ABL DIP Priority Collateral</u>" means the portion (if any) of the Collateral securing the ABL DIP Facility in which the lenders thereunder have a first priority security interest by virtue of the Orders.

"<u>ABL DIP Secured Parties</u>" means the "Secured Parties" under, and as such term is defined in, the ABL DIP Credit Agreement.

"<u>Accounts</u>" means all "accounts," as such term is defined in the UCC as in effect on the date hereof in the State of New York, in which any Person now or hereafter has rights.

"<u>Acquisition</u>" by any Person, means the purchase or acquisition in a single transaction or a series of related transactions by any such Person, individually or, together with its Affiliates, of (a) any Equity Interest of any other Person (other than an existing Subsidiary of the Borrower) or (b) all or a substantial portion of the Property, including, without limitation, all or a substantial portion of the property comprising a division, business unit or line of business, of any other Person (other than a Subsidiary of the Borrower), whether involving a merger or consolidation with such other Person. "<u>Acquire</u>" has a meaning correlative thereto.

"<u>Adequate Protection Liens</u>" has the meaning specified in the Order in effect on the date of determination.

"<u>Adequate Protection Payments</u>" means the adequate protection payments required to be made under Section 2.5 of the Interim Order or the Final Order, as applicable.

"<u>Administrative Agent</u>" means Wilmington Savings Fund Society, FSB, in its capacities as administrative agent and collateral agent under this Agreement and the other applicable Loan Documents, or any successor in such capacities.

"<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule 11.02</u>, or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.

"<u>Administrative Questionnaire</u>" means an Administrative Questionnaire in a form supplied, and reasonably satisfactory, to the Administrative Agent.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 316 of 432

"<u>Affiliate</u>" means, with respect to any Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.

"<u>Agent Fee Letter</u>" means that certain letter agreement, dated as of the date hereof, by and among the Borrower and the Administrative Agent, as the same may be amended, amended and restated, supplemented or otherwise modified from time to time.

"Agent Parties" has the meaning specified in Section 11.02(c).

"<u>Agent-Related Person</u>" means the Administrative Agent and its affiliates and the officers, directors, employees, agents and attorneys-in-fact of the Administrative Agent and its affiliates.

"<u>Aggregate Commitments</u>" means, at any time, the aggregate principal amount of the Commitments of all Lenders in effect at such time. The Aggregate Commitments on the Closing Date is \$100,000,000.

"Agreement" has the meaning specified in the preamble hereto.

"<u>Anti-Terrorism Laws</u>" means Law related to terrorism financing or money laundering including the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act ("<u>USA PATRIOT Act</u>") of 2001 (Title III of Pub. L. 107-56), The Currency and Foreign Transactions Reporting Act (also known as the "<u>Bank Secrecy Act</u>," 31 U.S.C. §§ 5311-5330 and 12 U.S.C. §§ 1818(s), 1820(b) and 1951-1959), the Trading With the Enemy Act (50 U.S.C. § 1 <u>et seq</u>., as amended), Executive Order 13224 (effective September 24, 2001).

"<u>Applicable Margin</u>" means, with respect to (a) Base Rate Loans, 7.00% per annum and (b) Eurodollar Rate Loans, 8.00% per annum.

"<u>Applicable Percentage</u>" means, with respect to any Lender at any time, the percentage (carried out to the second decimal place) of the Aggregate Commitments represented by the principal amount of such Lender's Commitment at such time. The initial Applicable Percentage of each Lender is set forth opposite the name of such Lender on <u>Schedule 2.01</u> hereto or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"<u>Approved Budget</u>" means the Initial Budget, at all times until updated pursuant to <u>Section</u> <u>6.17(b)</u> by an Updated Budget during the continuance of the Chapter 11 Cases.

"<u>Approved Fund</u>" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"<u>Assignee Group</u>" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.

"<u>Assignment and Assumption</u>" means an assignment and assumption entered into by a Lender and an Eligible Assignee) and accepted by the Administrative Agent, in substantially the form of <u>Exhibit D</u>.

"Assignment Taxes" has the meaning specified in the definition of "Other Taxes."

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 317 of 432

"<u>Attributable Indebtedness</u>" means, on any date, (a) in respect of any Capitalized Lease of any Person, the capitalized amount thereof that would (or should) appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease or similar payments under the relevant lease or other applicable agreement or instrument that would (or should) appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease or other agreement or instrument were accounted for as a Capitalized Lease and (c) all Synthetic Debt of such Person.

"<u>Audited Financial Statements</u>" means the audited consolidated balance sheet of Holdings and its Subsidiaries for the fiscal year ended December 31, 2016, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of Holdings and its Subsidiaries, including the notes thereto, in each case as included in Holdings' Annual Report on Form 10-K for the fiscal year ended December 31, 2016.

"<u>Availability Period</u>" means the period from and including the date the Interim Order is entered by the Bankruptcy Court to but excluding the Maturity Date.

"<u>Bail-In Action</u>" means the exercise of any Write-Down and Conversion Powers by the applicable EEA Resolution Authority in respect of any liability of an EEA Financial Institution.

"<u>Bail-In Legislation</u>" means with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule.

"<u>Bankruptcy Code</u>" means title 11 of the United States Bankruptcy Code (11 U.S.C. §101, et seq.), as amended and in effect from time to time, and any successor statute.

"<u>Bankruptcy Court</u>" means the Untied Stated Bankruptcy Court for the Southern District of New York, or any other court having jurisdiction over the Chapter 11 Cases.

"<u>Base Rate</u>" means, for any day, a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the Prime Rate in effect for such day and (c) the Eurodollar Rate for an Interest Period of one month plus 1.00%. Any change in the Base Rate due to a change in the Prime Rate or the Federal Funds Rate shall take effect at the opening of business on the day specified in the public announcement of such change. For the purposes of subsection (c) above, the Eurodollar Rate shall be determined daily and any change in the Eurodollar Rate shall take effect on the day of such change.

"<u>Base Rate Loan</u>" means a Loan that bears interest based on the Base Rate.

"<u>Benefit Plan</u>" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) established, sponsored, maintained or contributed to by the Borrower, Holdings or any Subsidiary or for which the Borrower, Holdings or any Subsidiary have an obligation to sponsor, maintain or contribute to or, with respect to any such plan that is subject to Sections 412 or 430 of the Code or Title IV of ERISA, is or was within the prior six years, established, sponsored, maintained, or contributed to or obligated to be contributed to, by the Borrower or any ERISA Affiliate or for which Borrower, Holdings or any Subsidiary or any ERISA Affiliates has any liability (contingent or otherwise).

"<u>Borrower</u>" has the meaning specified in the introductory paragraph hereto.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 318 of 432

"Borrower Materials" has the meaning specified in Section 6.02.

"<u>Borrowing</u>" means a borrowing consisting of simultaneous Loans of the same Type made by each of the Lenders pursuant to <u>Section 2.02</u>; provided that "Borrowings" shall not include conversions and continuations of outstanding Loans.

"<u>Borrowing Date</u>" means any Business Day specified by the Borrower in any Notice of Borrowing as a date on which the Borrower requests the relevant Lenders to make Loans hereunder.

"Budget Delivery Date" has the meaning specified in Section 6.17(b).

"<u>Business Day</u>" means any day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required to close, or are in fact closed; <u>provided</u>, that with respect to notices and determinations in connection with, and payments of principal and interest on, and any other amounts due in respect of, Eurodollar Rate Loans, such day is also a London Banking Day.

"<u>Capital Expenditures</u>" means, with respect to any Person for any period, any expenditure in respect of the purchase or other acquisition of any fixed or capital asset (excluding normal replacements and maintenance which are properly charged to current operations). For purposes of this definition, the purchase price of equipment that is purchased simultaneously with the trade-in of existing equipment or with Insurance Proceeds shall be included in Capital Expenditures only to the extent of the gross amount by which such purchase price exceeds the credit granted by the seller of such equipment for the equipment being traded in at such time or the amount of such Insurance Proceeds, as the case may be.

"<u>Capitalized Leases</u>" means all leases that have been or should be, in accordance with GAAP, recorded as capitalized leases.

"<u>Carve-Out</u>" has the meaning in (a) prior to the date of entry of the Final Order, the Interim Order or (b) for all purposes after the date on which the Final Order is entered, the Final Order, as the case may be.

"<u>Cash Collateral Account</u>" means any Deposit Account or Securities Account established by the Administrative Agent in which cash and Cash Equivalents may from time to time be on deposit or held therein as provided herein.

"<u>Cash Equivalents</u>" means any of the following types of Investments, to the extent owned by the Borrower or any of its Subsidiaries free and clear of all Liens (other than Liens created under the Collateral Documents):

(a) readily marketable obligations issued or directly and fully guaranteed or insured by the United States of America or any agency or instrumentality thereof having maturities of not more than one year from the date of acquisition thereof; <u>provided</u> that the full faith and credit of the United States of America is pledged in support thereof;

(b) time deposits with, or insured certificates of deposit or bankers' acceptances of, any nationally-recognized securities dealer or any commercial bank, trust company, savings and loan association or savings bank that (i) (A) is a Lender or (B) is (x) organized under the laws of the United States of America, any state thereof or the District of Columbia, or a political subdivision thereof, (y) a U.S. branch of any such institution organized under the laws of any other country that is a member of the OECD, or (z) the principal banking subsidiary of a bank holding company organized under the laws of the United States of America, any state thereof or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 319 of 432

the District of Columbia, and is a member of the Federal Reserve System, (ii) issues (or the parent of which issues) commercial paper rated as described in clause (c) of this definition and (iii) has combined capital and surplus of at least \$1,000,000,000, in each case with maturities of not more than 180 days from the date of acquisition thereof;

(c) commercial paper issued by any Person organized under the laws of any state of the United States of America and rated at least "Prime-2" (or the then equivalent grade) by Moody's or at least "A-2" (or the then equivalent grade) by S&P, in each case with maturities of not more than 180 days from the date of acquisition thereof;

(d) Investments, classified in accordance with GAAP as current assets of the Borrower or any of its Subsidiaries, in money market investment programs registered under the Investment Company Act of 1940, which are administered by financial institutions that have the highest rating obtainable from either Moody's or S&P, and the portfolios of which are limited solely to Investments of the character, quality and maturity described in clauses (a), (b) and (c) of this definition; and

(e) obligations with any Lender, any other bank or trust company described in clause (b) above, or any nationally-recognized securities dealer, in respect of the repurchase of obligations of the type described in clause (a) above, <u>provided</u> that such repurchase obligations shall be fully secured by obligations of the type described in said clause and the possession of such obligations shall be transferred to, and segregated from other obligations owned by, such Lender, such other bank or trust company or such securities dealer.

"<u>Cash Management Order</u>" means that certain Order Authorizing Cenveo, Inc., et al. to Continue (I) to Operate its Cash Management System, Honor Certain Prepetition Obligations Related Thereto, and Maintain Existing Business Forms; and (II) its Intercompany Transactions, which shall be in form and substance reasonably satisfactory to the Required Lenders, as the same may be extended, amended, supplemented or modified in a manner reasonably satisfactory to the Required Lenders.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980.

"<u>CFC</u>" means a Person that is a controlled foreign corporation under Section 957 of the Code.

"<u>Change in Law</u>" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority; <u>provided</u> that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Change of Control" means an event or series of events by which:

(a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of such person or

its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan), except the Permitted Holders, becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934), directly or indirectly, of 20% or more of the equity securities of Holdings entitled to vote for members of the board of directors or equivalent governing body of Holdings on a fully-diluted basis; or

(b) a majority of the members of the board of directors or other equivalent governing body of Holdings cease to be composed of individuals (i) who were members of that board or equivalent governing body on the Closing Date, or (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body (excluding, in the case of clause (ii), any individual whose initial nomination for, or assumption of office as, a member of that board or equivalent governing body occurs as a result of an actual or threatened solicitation of proxies or consents for the election or removal of one or more directors by any person or group other than a solicitation for the election of one or more directors by or on behalf of the board of directors); or

(c) any Person or two or more Persons acting in concert (excluding the Permitted Holders) shall have acquired by contract or otherwise, the power to exercise, directly or indirectly, a controlling influence over the management or policies of Holdings, or control over the equity securities of Holdings entitled to vote for members of the board of directors or equivalent governing body of Holdings on a fully-diluted basis representing 20% or more of the combined voting power of such securities; or

(d) Holdings shall cease, directly or indirectly, to own and control legally and beneficially all of the Equity Interests in the Borrower; or

(e) any plan for the liquidation or dissolution of any Loan Party (whether or not otherwise in compliance with the provisions of this Agreement) is approved by holders of the Capital Stock of such Loan Party; or

(f) all or substantially all of the assets of any Loan Party are sold, leased, transferred or otherwise disposed of; or

(g) the occurrence of a change of control (or similar event, however defined) under and as defined in any of the Prepetition Debt Documents.

"Chapter 11 Cases" has the meaning specified in the recitals.

"<u>Chapter 11 Debtors</u>" has the meaning assigned to the term "Debtors" in the Orders.

"Chapter 11 Plan Effective Date" means the effective date of the Plan of Reorganization.

"Chattel Paper" has the meaning provided in Article 9 of the UCC.

"<u>Closing Date</u>" has the meaning specified in <u>Section 4.01</u>.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"<u>Collateral</u>" means the "DIP Collateral", as such term is defined in the Orders and, for the avoidance of doubt, includes all assets and property of the Loan Parties (other than Excluded Property).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 321 of 432

"<u>Collateral Documents</u>" means, collectively, the Security Agreement, the Orders and all other similar documents, instruments and agreements delivered to the Administrative Agent pursuant to <u>Section</u> <u>6.12</u>, and each of the other agreements, instruments or documents that creates, perfects or records, or purports to create, perfect or record, a Lien in favor of the Administrative Agent for the benefit of the Secured Parties.

"<u>Commitment</u>" means, with respect to each Lender, the commitment of such Lender to make Loans, in the aggregate principal amount set forth opposite such Lender's name on <u>Schedule 2.01</u> under the caption "Commitment", as amended from time to time to reflect Assignments (if any), and as such amount may be reduced pursuant to this Agreement.

"<u>Committee</u>" means the official committee of unsecured creditors appointed in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code, if any.

"<u>Commodity Exchange Act</u>" means the Commodity Exchange Act (7 U.S.C. § 1 <u>et seq</u>.), as amended from time to time, and any successor statute.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C.

"<u>Contractual Obligation</u>" means, as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"<u>Control</u>" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "<u>Controlling</u>" and "<u>Controlled</u>" have meanings correlative thereto.

"Control Notice" has the meaning assigned to such term in Section 2.12(c).

"Control Effective Time" has the meaning assigned to such term in Section 2.12(c).

"Credit Date" has the meaning assigned to such term in Section 5.

"<u>Cumulative Net Operating Cash Flow</u>" means, on any date of determination, the amount equal to (a) Total Cash Receipts minus (b) Total Operating Disbursements, measured in each case, on a cumulative basis, for the period commencing on the Closing Date through and including the last day of the then most recently ended Testing Period.

"<u>Debtor Relief Laws</u>" means the Bankruptcy Code of the United States and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States, any state, or territory thereof, or of any other applicable jurisdictions, from time to time in effect and affecting the rights of creditors generally (including, as the case may be, applicable corporate statutes).

"<u>Default</u>" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" has the meaning assigned to such term in <u>Section 2.06(c)</u>.

"<u>Defaulting Lender</u>" means any Lender that (a) has failed to perform any funding obligations hereunder, or that has failed to pay to the Administrative Agent or any other Lender any other amount

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 322 of 432

required to be paid by it hereunder and, in each case, such failure is not cured within three (3) Business Days; (b) has notified the Administrative Agent or the Borrower that such Lender does not intend to comply with its funding obligations hereunder or has made a public statement to the effect that it does not intend to comply with its funding obligations hereunder or under any other credit facility; (c) has failed, within three (3) Business Days following request by the Administrative Agent, to confirm in a manner satisfactory to the Administrative Agent that such Lender will comply with its funding obligations hereunder; or (d) has, or has a direct or indirect parent company that has, become the subject of an insolvency proceeding or taken any action in furtherance thereof, or become the subject of a Bail-In Action; provided, however, that a Lender shall not be a Defaulting Lender solely by virtue of a Governmental Authority's ownership of any equity interest in such Lender or parent company.

"Deposit Account" has the meaning assigned thereto in Article 9 of the UCC.

"<u>Designated Jurisdiction</u>" means any country or territory to the extent that such country or territory itself is the subject of any Sanction.

"<u>DIP Motion</u>" means a motion to be filed by the Chapter 11 Debtors with the Bankruptcy Court seeking approval of the Facility, which motion shall be in form and substance acceptable to the Required Lenders.

"<u>DIP Priority Collateral</u>" means the portion of the collateral securing the Facility in which the Lenders have a first priority perfected security interest by virtue of the Orders.

"<u>DIP Superpriority Claims</u>" has the meaning specified in the Order in effect on the date of determination.

"<u>Disposition</u>" or "<u>Dispose</u>" means the sale, transfer, license, lease or other disposition (including any sale and leaseback transaction) of any real or personal property by any Person (or the granting of any option or other right to do any of the foregoing), including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"<u>Dollar</u>" and "<u>\$</u>" mean lawful money of the United States.

"<u>Dollar Equivalent</u>" means on any date, with respect to any amount denominated in Dollars, such amount in Dollars, and with respect to any stated amount denominated in a currency other than Dollars, the equivalent amount thereof in Dollars as determined by the Administrative Agent (which determination shall be conclusive and binding absent manifest error) pursuant to <u>Sections 1.07</u> and <u>1.08</u>.

"<u>Domestic Subsidiary</u>" means any Subsidiary that is organized under the laws of the United States, any state thereof or the District of Columbia.

"<u>EEA Financial Institution</u>" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"<u>EEA Member Country</u>" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 323 of 432

"<u>EEA Resolution Authority</u>" means any public administrative authority or any person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"<u>Eligible Assignee</u>" means any Person that meets the requirements to be an assignee under <u>Sections 11.06(b)(iii)</u>, (v) and (vi) (subject to such consents, if any, as may be required under <u>Section 11.06(b)(iii)</u>).

"<u>Embargoed Person</u>" means any party that (i) is publicly identified on the most current list of "Specially Designated Nationals and Blocked Persons" published by the U.S. Treasury Department's Office of Foreign Assets Control ("<u>OFAC</u>") or (ii) resides, is organized or chartered, or has a place of business in a country or territory subject to OFAC sanctions programs.

"<u>Environment</u>" means ambient air, indoor air, surface water, groundwater, drinking water, soil, surface and subsurface strata, and natural resources such as wetlands, flora and fauna.

"<u>Environmental Laws</u>" means any and all applicable Laws relating to pollution or the protection of the Environment or human health (to the extent related to exposure to Hazardous Materials or the Release into the Environment of Hazardous Materials).

"<u>Environmental Liability</u>" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of the Borrower, any other Loan Party or any of their respective Subsidiaries directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the Release or threatened Release of any Hazardous Materials into the Environment or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"<u>Environmental Permit</u>" means any permit, approval, identification number, license or other authorization required under any Environmental Law.

"Equity Interests" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests in such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"<u>ERISA</u>" means the Employee Retirement Income Security Act of 1974, as amended from time to time.

"<u>ERISA Affiliate</u>" means any trade or business (whether or not incorporated) under common control with the Borrower within the meaning of Section 414(b) or (c) of the Code (and Sections 414(m) and (o) of the Code for purposes of provisions relating to Section 412 of the Code).

"<u>ERISA Event</u>" means (a) a Reportable Event with respect to a Pension Plan; (b) a withdrawal by the Borrower or any ERISA Affiliate from a Pension Plan subject to Section 4063 of ERISA during a

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 324 of 432

plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA or results in the imposition of liability under Section 4062(e) of ERISA; (c) a complete or partial withdrawal by the Borrower, Holdings or any ERISA Affiliate from a Multiemployer Plan or notification that a Multiemployer Plan is in reorganization, insolvent or in endangered or critical status; (d) other than to the extent expressly permitted by the terms of the Restructuring Support Agreement, the filing of a notice of intent to terminate, the treatment of a Pension Plan amendment as a termination under Section 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan including the receipt by the Borrower or any ERISA Affiliate from the PBGC or plan administrator of any notice relating to an intention to terminate any Pension Plan or Multiemployer Plan or appoint a trustee to administer any such Plan; (e) an event or condition which constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; (f) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon the Borrower, Holdings or any ERISA Affiliate; (g) a determination that any Pension Plan is in, or expected to be in, "at risk" status; or (h) the filing pursuant to Section 412(c) of the Code or Section 302(c) of ERISA of application for a waiver of the minimum funding standard with respect to any Pension Plan.

"<u>EU Bail-In Legislation Schedule</u>" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"Eurodollar Rate" means:

(a) for any Interest Period with respect to a Eurodollar Rate Loan, the rate per annum equal to (i) the offered rate administered by ICE Benchmark Administration (or any successor or substitute for such service, as determined by the Administrative Agent from time to time for purposes of providing quotations of interest rates applicable to Dollar deposits in London or other applicable interbank market in a manner consistent with market practice) ("LIBOR"), as published by Bloomberg (or such other commercially available source providing quotations of LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two London Banking Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such rate is not available at such time for any reason, then the "Eurodollar Rate" for such Interest Period shall be the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Rate Loan being made, continued or converted by the Administrative Agent and with a term equivalent to such Interest Period would be offered to major banks in the London interbank eurodollar market (or other applicable interbank market selected by the Administrative Agent) at their request at approximately 11:00 a.m. (London time) two London Banking Days prior to the commencement of such Interest Period; provided that, the Eurodollar Rate applicable to a Eurodollar Rate Loan shall be deemed to be not less than 1.00% per annum; and

(b) for any interest calculation with respect to a Base Rate Loan on any date, the rate per annum equal to (i) LIBOR, at approximately 11:00 a.m., London time determined two London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market (or other applicable interbank market selected by the Administrative Agent) for a term of one month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in same day funds in the approximate amount of the Base Rate Loan being made or maintained and with a term equal to

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 325 of 432

one month would be offered to major banks in the London interbank Eurodollar market (or other applicable interbank market selected by the Administrative Agent) at their request at the date and time of determination <u>provided</u> that, the Eurodollar Rate applicable to a Base Rate Loan shall be deemed to be not less than 2.00% per annum.

"Eurodollar Rate Loan" means a Loan that bears interest at a rate based on the Eurodollar Rate.

"Eurodollar Reserve Percentage" of any Lender for each Interest Period for any Eurodollar Rate Loan means the reserve percentage applicable to such Lender during such Interest Period (or if more than one such percentage shall be so applicable, the daily average of such percentages for those days in such Interest Period during which any such percentage shall be so applicable) under Regulation D or other regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) then applicable to such Lender with respect to liabilities or assets consisting of or including "Eurocurrency liabilities" (as defined in Regulation D of the FRB) having a term equal to such Interest Period.

"Event of Default" has the meaning specified in Section 8.01.

"<u>Exchange Rate</u>" means, on any day with respect to any currency other than Dollars, the exchange rate reported by Bloomberg (or other commercially available source designated by the Administrative Agent) as of the end of the preceding Business Day in the financial market for such currency.

"Exchange Rate Date" has the meaning set forth in Section 1.08.

"<u>Excluded Property</u>" has the meaning set forth in the Security Agreement and shall include any Equity Interests in any Person owned by a Loan Party that does not constitute "Pledged Equity" pursuant to the provision in <u>Section 2.01(a)</u> of the Security Agreement.

"Excluded Swap Obligation" means, with respect to any Guarantor, any Swap Obligation if, and to the extent that, all or a portion of the Guaranty of such Guarantor of, or the grant by such Guarantor of a security interest to secure, such Swap Obligation (or any Guaranty thereof) is or becomes illegal under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Guarantor's failure for any reason to constitute an "eligible contract participant" as defined in the Commodity Exchange Act (determined after giving effect to Section 10.11 of this Agreement and Section 5.15 of the Guaranty and any other "keepwell, support or other agreement" for the benefit of such Guarantor and any and all guarantees of such Guarantor's Swap Obligations by other Loan Parties) at the time the Guaranty of such Swap Obligation. If a Swap Obligation arises under a master agreement governing more than one swap, such exclusion shall apply only to the portion of such Swap Obligation that is attributable to swaps for which such Guaranty or security interest is or becomes excluded in accordance with the first sentence of this definition.

"<u>Excluded Taxes</u>" means, with respect to the Administrative Agent, any Lender or any other recipient of any payment to be made by or on account of any Obligation of the Borrower or any Guarantor hereunder, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such recipient being organized under the Laws of, or having its principal office (or, in the case of any Lender, its applicable Lending Office) located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, any United States federal withholding Tax that is

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 326 of 432

imposed on amounts payable to such Lender pursuant to Laws in effect at the time (i) such Lender becomes a party hereto (other than pursuant to an assignment request by the Borrower under <u>Section 11.13</u>), or (ii) designates a new Lending Office, except, in each case, to the extent that such Lender (or its assignor, if any) was entitled, immediately prior to the designation of a new Lending Office (or assignment), to receive additional amounts from a Loan Party with respect to such withholding Tax pursuant to <u>Section 3.01(a)</u>, (c) any Tax attributable to a failure of a recipient to comply with <u>Section 3.01(e)</u> and (d) any withholding Taxes imposed under FATCA.

"Facility" has the meaning assigned to such term in the recitals.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code as of the date hereof (and any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the Code.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; <u>provided</u> that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to the Administrative Agent by three (3) federal funds brokers of recognized standing selected by the Administrative Agent on such day on such transactions, as determined by the Administrative Agent.

"Fees" means all amounts payable pursuant to or referred to in Section 2.05.

"<u>Final Order</u>" means a final order of the Bankruptcy Court approving the Loans and the Facility, in form and substance acceptable to the Required Lenders, which I shall be in full force and effect and shall not be reversed, vacated, stayed, amended, supplemented or otherwise modified or subject to the possibility of appeal, in each case, without the prior written consent of the Required Lenders.

"Financial Covenants" means the covenants set forth in Section 7.11.

"<u>Flood Insurance Laws</u>" means, collectively, (i) the National Flood Insurance Act of 1968, (ii) the Flood Disaster Protection Act of 1973, (iii) the National Flood Insurance Reform Act of 1994 and (iv) the Flood Insurance Reform Act of 2004, or, in each case, any successor statute thereto.

"Foreign Lender" means the Administrative Agent and any Lender, in each case, that is not a "U.S. Person".

"<u>Foreign Plan</u>" means any employee benefit plan established or maintained by Borrower, Holdings or any Subsidiary (other than a Plan or Multiemployer Plan) for the benefit of employees employed outside the United States.

"<u>FRB</u>" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 327 of 432

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"<u>GAAP</u>" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"<u>Governmental Authority</u>" means the government of the United States, or any other nation, or of any political subdivision thereof, whether state, territorial, local, municipal or otherwise, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Guarantee" means, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person. The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"<u>Guarantors</u>" means, collectively (a) Holdings, and (b) the Subsidiaries listed on <u>Schedule 6.12</u> and each other Subsidiary of the Borrower that executes the Guaranty or that shall be required to execute and deliver a guaranty or guaranty supplement pursuant to <u>Section 6.12</u>.

"<u>Guaranty</u>" means, collectively, (i) the Guaranty made by Holdings under <u>Article X</u> in favor of the Secured Parties, and (ii) the Guaranty made by the Guarantors in favor of the Secured Parties, dated as of the date hereof, in each case, together with each other guaranty and guaranty supplement delivered pursuant to <u>Section 6.12</u>.

"<u>Hazardous Materials</u>" means all explosive or radioactive substances or wastes and all hazardous or toxic substances or wastes, including petroleum or petroleum distillates, asbestos or asbestos-containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other materials, substances, pollutants, contaminants or wastes of any nature in any form regulated pursuant to any Environmental Law.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 328 of 432

"Holdings" has the meaning specified in the preamble hereto.

"<u>Indebtedness</u>" means, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

(a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;

(b) the maximum amount of all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, bank guaranties, surety bonds and similar instruments;

(c) net obligations of such Person under any Swap Contract;

(d) all obligations of such Person to pay the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business consistent with past practices and not past due for more than ninety (90) days after the date on which such trade account was created);

(e) indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;

(f) all Attributable Indebtedness of such Person;

(g) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Equity Interest in such Person or any Affiliate of such Person or any warrant, right or option to acquire such Equity Interest, valued, in the case of a redeemable preferred interest, at the greater of its voluntary or involuntary liquidation preference <u>plus</u> accrued and unpaid dividends; and

(h) all Guarantees of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture (other than a joint venture that is itself a corporation limited liability company or similar legal entity) in which such Person is a general partner or a joint venturer, unless such Indebtedness is expressly made non-recourse to such Person. The amount of any net obligations of any Person under any Swap Contract on any date shall be deemed to be the Swap Termination Value thereof due and payable by such Person as of such date.

"<u>Indemnified Taxes</u>" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of the Borrower under any Loan Document and (b) to the extent not otherwise described in (a), Other Taxes.

"Indemnitees" has the meaning specified in Section 11.04(b).

"<u>Information</u>" has the meaning specified in <u>Section 11.07</u>.

"Initial Budget" has the meaning specified in Section 5.19.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 329 of 432

"Initial Loan" has the meaning specified in Section 2.01.

"<u>Insurance Proceeds</u>" means all insurance proceeds (other than business interruption insurance proceeds), damages, awards, claims and rights of action with respect to any casualty or other loss, damage or destruction of any real or personal property of Holdings or its Subsidiaries.

"Intellectual Property" has the meaning specified in Section 5.18.

"<u>Interest Payment Date</u>" means, (a) as to each Eurodollar Rate Loan, the last day of each Interest Period applicable to such Eurodollar Rate Loan; <u>provided</u>, <u>however</u>, that if any Interest Period for a Eurodollar Rate Loan exceeds three months, the respective dates that fall every three months after the beginning of such Interest Period shall also be Interest Payment Dates; (b) as to each Base Rate Loan, the first Business Day of each calendar month and (c) as to each Loan, the date of prepayment of such Loan and the Maturity Date of such Loan.

"Interest Period" means, as to any Borrowing of a Eurodollar Rate Loan, the period commencing on the date of such Borrowing or on the last day of the immediately preceding Interest Period applicable to such Borrowing, as applicable, and ending on the date that is one month thereafter, as the Borrower may elect, or the date any Borrowing of a Eurodollar Rate Loan is converted to a Borrowing of a Base Rate Loan in accordance with <u>Section 2.08</u> or repaid or prepaid in accordance with <u>Section 2.07</u> or <u>Section 2.09</u>; <u>provided</u>, that if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day. Interest shall accrue from and including the first day of an Interest Period to but excluding the last day of such Interest Period.

"<u>Interim Order</u>" means the interim order entered by the Bankruptcy Court in the Chapter 11 Cases approving the DIP Motion on an interim basis, which Interim Order shall be in form and substance acceptable to the Required Lenders, shall be in full force and effect, and shall not, subject to entry of the Final Order, be reversed, vacated, stayed, amended, supplemented or otherwise modified without the prior written consent of the Required Lenders.

"<u>Internal Control Event</u>" means a material weakness in, or fraud that involves management or other employees who have a significant role in, Holdings' and its Subsidiaries' internal controls over financial reporting, in each case as described in the Securities Laws.

"Inventory" has the meaning as defined in the UCC, including all goods intended for sale, lease, display or demonstration, all work in process, and all raw materials, and other materials and supplies of any kind that are or could be used in connection with the manufacture, printing, packing, shipping, advertising, sale, lease or furnishing of such goods, or otherwise used or consumed in a Loan Party's business.

"Investment" means, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of Equity Interests of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or interest in, another Person, (c) the purchase or other acquisition (in one transaction or a series of transactions) of assets of another Person that constitute a business unit or all or a substantial part of the business of such Person or (d) any other investment of money or capital outside the ordinary course of business in order to obtain a profitable return. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 330 of 432

"IP Rights" has the meaning specified in Section 5.18.

"<u>IRS</u>" means the United States Internal Revenue Service.

"Joinder Agreement" means a Joinder Agreement executed and delivered in accordance with the provisions of <u>Section 6.12</u>, substantially in the form of <u>Exhibit G</u> hereto.

"<u>Laws</u>" means, collectively, all international, foreign, federal, state, territorial, municipal and local statutes, treaties, rules, guidelines, regulations, ordinances, codes, executive orders and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"<u>Lender</u>" has the meaning specified in the introductory paragraph hereto.

"<u>Lender Advisors</u>" means, collectively, each of the advisors (including Ducera), counsel (with respect to legal fees and expenses, limited to Stroock & Stroock & Lavan LLP and one local counsel in each jurisdiction where necessary), consultants and other professionals as are reasonably required by the Required Lenders in connection with the Facility or any other transactions contemplated by the Restructuring Support Agreement.

"<u>Lender Payment Letter</u>" means that certain letter agreement, dated as of the date hereof, by and among the Borrower, Holdings and the Lenders party thereto, as the same may be amended, amended and restated, supplemented or otherwise modified from time to time.

"Lender Payment" has the meaning specified in Section 2.05.

"<u>Lending Office</u>" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Borrower and the Administrative Agent.

"<u>Lien</u>" means any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), trust (constructive, deemed, statutory or otherwise), charge, or preference, priority or other security interest or preferential arrangement in the nature of a security interest of any kind or nature whatsoever (including any conditional sale or other title retention agreement, any easement, right of way or other encumbrance on title to real property, and any financing lease having substantially the same economic effect as any of the foregoing).

"Loan Documents" means, collectively, (a) this Agreement, (b) the Notes, (c) the Guaranty, (d) the Collateral Documents, (e) the Payment Letters, (f) each Joinder Agreement, (g) each Notice of Borrowing, Notice of Conversion/Continuation and Notice of Prepayment and (h) each other agreement, instrument or document now or hereafter executed and delivered in connection with any of the foregoing.

"Loan Parties" means, collectively, the Borrower and each Guarantor, and "Loan Party" means any one such Person.

"Loans" means the loans made by the Lenders to the Borrower pursuant to this Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 331 of 432

"<u>London Banking Day</u>" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank Eurodollar market or in another applicable interbank Eurodollar market used to determine the Eurodollar Rate in accordance with the definition thereof.

"<u>Make-Whole Amount</u>" means, with respect to optional prepayments of Loans on any date prior to the Maturity Date, an amount (payable in cash) equal to the excess, if any, of (a) the present value as of such prepayment date of (i) the principal amount of such Loans *plus* (ii) the amount of all interest (calculated using the interest rate applicable to such Loans as of the date of such prepayment) that would have accrued on the principal amount of such Loans from such prepayment date through the Maturity Date (excluding any accrued but unpaid interest to such prepayment date), computed using a discount rate equal to the Treasury Rate (as in effect on such prepayment date) *plus* 0.50%, <u>over</u> (b) the principal amount of such Loans.

"Management Conference Call" has the meaning specified in Section 6.17(d).

"Material Adverse Effect" means (a) a material adverse change in, or a material adverse effect upon, the operations, business, properties, liabilities (actual or contingent) or condition (financial or otherwise) of the Loan Parties and their Subsidiaries taken as a whole; (b) a material impairment of the rights and remedies of the Administrative Agent or any Lender under any Loan Document, or of the ability of any Loan Party to perform its obligations under any Loan Document to which it is a party; (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party; or (d) a material adverse effect upon the Collateral or the Administrative Agent's Liens on the Collateral or the priority of such Liens, in each case, other than resulting from the commencement or continuation of the Chapter 11 Cases, the events that customarily occur following the commencement of a proceeding under Chapter 11 of the Bankruptcy Code, any defaults under agreements that have no material effect on Borrower and Guarantors under the terms of the Bankruptcy Code as a result of the commencement of the Chapter 11 Cases. In determining whether any individual event would result in a Material Adverse Effect, notwithstanding that such event in and of itself does not have such effect, a Material Adverse Effect shall be deemed to have occurred if the cumulative effect of such event together with all other events related thereto (whether concurrent or in a series) would result in a Material Adverse Effect.

"<u>Material Contract</u>" means, with respect to any Loan Party, each contract or agreement listed on <u>Schedule 1.02</u> hereto and each other contract or agreement entered into by any Loan Party (x) pursuant to which the aggregate consideration payable thereunder is in excess of \$20,000,000 in any fiscal year or (y) the breach or termination of which could reasonably be expected to have a Material Adverse Effect.

"<u>Maturity Date</u>" means the earliest to occur of: (a) the date that is 9 months after the Petition Date; (b) the date of consummation of any sale of all or substantially all of the assets of the Loan Parties pursuant to section 363 of the Bankruptcy Code; (c) the date that is forty (40) calendar days after the Petition Date, if the Final DIP Order has not been entered by the Bankruptcy Court on or before such date; (d) the date on which the Loans shall become due and payable and Commitments terminated upon acceleration pursuant to <u>Article 8</u> and (e) the Chapter 11 Plan Effective Date.

"Maximum Cap Ex Amount" has the meaning specified in Section 7.12.

"Maximum Rate" has the meaning specified in Section 11.09.

"<u>Measurement Period</u>" means, at any date of determination, the most recently completed four fiscal quarters of the Borrower.

-19-

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 332 of 432

"Milestones" means the events described in Schedule 6.23(c).

"<u>Minimum Liquidity</u>" means, as of any time of determination, the sum of the amount of unrestricted and unreserved cash and Cash Equivalents of the Loan Parties that is not subject to a Lien (other than restricted and reserved cash subject to the Liens securing the Loans and the obligations under the ABL DIP Facility), plus Availability (as defined in the ABL DIP Facility) under the ABL DIP Facility at such time.

"<u>Moody's</u>" means Moody's Investors Service, Inc. and any successor thereto.

"<u>Multiemployer Plan</u>" means any Benefit Plan of the type described in Section 4001(a)(3) of ERISA.

"Net Cash Proceeds" means proceeds received by any Loan Party or any of its Subsidiaries after the Closing Date in cash or Cash Equivalents from any (a) Disposition by any Loan Party or its Subsidiaries (other than (x) pursuant to Sections 7.05(a), 7.05(b), 7.05(c) and 7.05(g) and (y) any Disposition between the Borrower and the Loan Parties or any Disposition among the Loan Parties permitted pursuant to Section 7.05) (provided with respect to any Net Cash Proceeds received by any Loan Party or any of its Subsidiaries in connection with any Disposition made pursuant to Section 7.05(f), solely to the extent such Net Cash Proceeds exceed \$1,000,000) net of (i) the reasonable and documented cash costs of sale, assignment or other disposition, including, without limitation, legal, accounting, investment banking fees and sales commissions incurred as a result thereof, taxes paid as a result thereof, in each case after taking into account any available tax credits or deductions and any tax sharing arrangements and amounts required to be applied to the repayment of Indebtedness secured by a Lien on the asset or assets that were the subject of such Disposition (other than any Lien securing ABL DIP Priority Collateral) by Borrower or any Loan Party as a result thereof and (ii) any amount required by the Bankruptcy Court to be paid or prepaid on Indebtedness (other than the Obligations) secured by a perfected and unavoidable Lien on the assets subject to such Disposition; provided, however, that the evidence of each of (i) and (ii) are provided to the Administrative Agent in form satisfactory to the Required Lenders; (b) Recovery Event (net (i) of any expenses of the type described in clause (a)(ii) above, if any and (ii) any actual out-of-pocket costs incurred by Holdings or any of its Subsidiaries in connection with the adjustment or settlement of any claims of Holdings or such Subsidiary in respect thereof (and any bona fide costs incurred in connection with taking of such assets), (c) (i) sale or issuance of Equity Interests by any Loan Party or any of its Subsidiaries or (ii) the incurrence or issuance of any Indebtedness by any Loan Party or any of its Subsidiaries, other than the incurrence of Indebtedness permitted under Section 7.02, in each case of (c)(i) and (c)(ii), net of reasonably and documented brokers' and advisors' fees and other costs actually incurred in connection with such transaction.

"Note" means a promissory note made pursuant to <u>Section 2.04(e)</u> by the Borrower in favor of a Lender, evidencing the Loans made by such Lender, which promissory note shall be substantially in the form of <u>Exhibit B</u> hereto.

"<u>Notice of Borrowing</u>" means a notice substantially in the form of <u>Exhibit A-1</u> hereto.

"Notice of Conversion/Continuation" means a notice substantially in the form of Exhibit A-2 hereto.

"<u>Obligations</u>" means all advances to, and debts (including interest, principal, fees, costs, expenses and indemnities), liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document or otherwise with respect to any Loan or arising in connection with the Transactions, whether

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 333 of 432

direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising.

"OECD" means the Organization for Economic Cooperation and Development.

"OFAC" has the meaning specified in the definition of "Embargoed Person."

"Orders" means the Interim Order and/or the Final Order, as the context requires.

"<u>Organization Documents</u>" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; (c) with respect to any unlimited company, the memorandum of association and articles of association; and (d) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization and, if applicable, any certificate or articles of formation or organization or organization and, if applicable, any certificate or articles of formation or organization or organization and, if applicable, any certificate or articles of formation or organization or organization and, if applicable, any certificate or articles of formation or organization or organization and any agreement and applicable agreement of formation or organization and, if applicable, any certificate or articles of formation or organization or organization and any agreement and applicable agreement and any agreement and any certificate or articles of formation or organization and any agreement and any certificate or articles of formation or organization and any agreement and any certificate or articles of formation or organization and any agreement and any certificate or articles of formation or organization and any agreement and any certificate or articles of formation or organization and any agreement and any certificate or articles of formation or organization and any agreement and any agreement and agreement and agreement and agreement and agreement and agreement and agreement agreement and agreement agreement agreement agreement agreement agreement agreement agreement agreem

"Other Connection Taxes" means, with respect to any recipient of any payment to be made by or on account of any Obligation of the Borrower or any Guarantor hereunder, Taxes imposed as a result of a present or former connection between such recipient and the jurisdiction imposing such Tax (other than connections arising from such recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"<u>Other Taxes</u>" means all present or future stamp, court, filing, recording, intangible or documentary Taxes or any other excise or property Taxes arising from any payment made hereunder or under any other Loan Document or from the execution, delivery, performance, registration or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document, but shall not include Other Connection Taxes imposed as a result of a transfer by a Lender or a Participant of any interest in a Loan Document ("<u>Assignment Taxes</u>"), other than Assignment Taxes imposed in respect of a transfer pursuant to <u>Section 11.13(a)</u>.

"<u>Outstanding Amount</u>" means with respect to all Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of Loans occurring on such date.

"Participant" has the meaning specified in Section 11.06(d).

"Participant Register" has the meaning specified in Section 11.06(d).

"Payment Letters" means, collectively, the Agent Fee Letter and the Lender Payment Letter.

"Payor" has the meaning specified in Section 7.06(a).

"<u>PBGC</u>" means the Pension Benefit Guaranty Corporation.

"<u>PCAOB</u>" means the Public Company Accounting Oversight Board.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 334 of 432

"<u>Pension Plan</u>" means any Benefit Plan, other than a Multiemployer Plan, that is subject to Title IV of ERISA or Section 412 or 430 of the Code.

"Perfection Certificate" has the meaning given such term in Section 1.03 of the Security Agreement.

"<u>Permitted Holders</u>" means Robert G. Burton, Sr., his immediate family and their heirs and estates and any Person (other than a natural Person and Holdings and its Subsidiaries) that is whollyowned or otherwise directly Controlled by any of the foregoing, including, without limitation, Burton Capital Management, LLC.

"Permitted Variance" means for any Testing Period set forth in the Approved Budget, (a) the Total Cash Receipts shall not be less than 80% of the budgeted amount as set forth in the Approved Budget with respect to the one calendar week period ending February 9, 2018, 85% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 90% of the budgeted amount as set forth in the Approved Budget amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 90% of the budgeted amount as set forth in the Approved Budget with respect to each Testing Period thereafter, and (b) the Total Operating Disbursements shall not exceed 120% of the budgeted amount as set forth in the Approved Budget with respect to the one calendar week period ending February 9, 2018, 115% of the budgeted amount as set forth in the Approved Budget with respect to the two (2) calendar week period ending February 16, 2018 and 110% of the budgeted amount as set forth in the Approved Budget with respect to each Testing Period thereafter (without giving effect to the making of Revolving Loans or DIP Term Loans or the repayments or prepayments of Revolving Loans or DIP Term Loans) (as set forth in the Approved Budget with respect to the applicable Testing Period) and (c) the variance in the actual Cumulative Net Operating Cash Flow to the budgeted amount (as set forth in the Approved Budget with respect to the applicable Testing Period) shall not be more than \$20,000,000 (adverse to the Borrower).

"<u>Person</u>" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"<u>Petition Date</u>" has the meaning given such term in the recitals hereto.

"<u>Plan of Reorganization</u>" means a plan of reorganization under the Chapter 11 Cases (including all related schedules, supplements, exhibits and orders, as applicable), which shall be in form and substance satisfactory to the Required Lenders and consistent with the terms of the Restructuring Support Agreement.

"Platform" has the meaning specified in Section 6.02.

"<u>Pledged Debt</u>" has the meaning specified in <u>Section 2.01</u> of the Security Agreement.

"Pledged Equity" has the meaning specified in Section 2.01 of the Security Agreement.

"Prepetition ABL Credit Agreement" means that certain Credit Agreement, dated as of April 16, 2013 (as amended, amended and restated, supplemented or otherwise modified from time to time), by and among the Borrower, Holdings, the lenders party thereto from time to time and Bank of America, N.A., as Administrative Agent, Issuing Bank and Swingline Lender (each, as defined therein), and the other entities party thereto from time to time.

"<u>Prepetition ABL Credit Agreement</u>" means the revolving loan facility established under the Prepetition ABL Credit Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 335 of 432

"Prepetition ABL Documents" means the Loan Documents, as such term is defined in the Prepetition ABL Credit Agreement.

"<u>Prepetition Debt Documents</u>" means the Prepetition ABL Documents and the Prepetition Notes Documents.

"<u>Prepetition Indebtedness</u>" means all Indebtedness of the Borrower outstanding on the Petition Date immediately prior to the filing of the Chapter 11 Cases, including under the Prepetition Notes Documents and the Prepetition ABL Documents.

"<u>Prepetition Notes Documents</u>" means, collectively, (a) the 4% Notes Documents, (b) the 6% Notes Documents, (c) the 6% Unsecured Notes Documents and (d) the 8.5% Notes Documents.

"<u>Prepetition Secured Parties</u>" means Persons constituting "Secured Parties", as such term is defined in each of the Prepetition Notes Documents and the Prepetition ABL Documents.

"<u>Prime Rate</u>" means for any day, a rate per annum that is publicly quoted from time to time by The Wall Street Journal as the "Prime Rate" in the United States (or, if The Wall Street Journal ceases quoting a base rate of the type described, either (a) the per annum rate quoted as the base rate on such corporate loans in a different national publication as selected by the Administrative Agent or (b) the highest per annum rate of interest published by the Federal Reserve Board in Federal Reserve statistical release H.15 (519) entitled "Selected Interest Rates" as the Bank prime loan rate or its equivalent).

"<u>Pro Rata Share</u>" means, with respect to each Lender on any date, a fraction (expressed as a percentage, carried out to the second decimal place), the <u>numerator</u> of which is the (x) aggregate principal amount of outstanding Loans held by such Lender on such date *plus* (y) the aggregate principal amount of such Lender's unfunded Commitments on such date and the <u>denominator</u> of which is (1) the aggregate principal amount of all outstanding Loans on such date *plus* (2) the unfunded Aggregate Loan Commitments on such date. The Pro Rata Shares of each Lender as in effect on the Closing Date are set forth opposite the name of such Lender on <u>Schedule 2.01</u>.

"Public Lender" has the meaning specified in Section 6.02.

"Qualified ECP Guarantor" shall mean, at any time, each Loan Party with total assets exceeding 10,000,000 or that qualifies at such time as an "eligible contract participant" under the Commodity Exchange Act and can cause another Person to qualify as an "eligible contract participant" at such time under 1a(18)(A)(v)(II) of the Commodity Exchange Act.

"<u>Recovery Event</u>" means any event or circumstance that results in the receipt by any Person of Insurance Proceeds or any taking of property of the Loan Parties or any of its Subsidiaries pursuant to the power of eminent domain, condemnation or otherwise that results in the receipt by such Person of a compensation payment in respect thereof.

"<u>Register</u>" has the meaning specified in <u>Section 11.06(c)</u>.

"<u>Registered Public Accounting Firm</u>" has the meaning specified by the Securities Laws and shall be independent of Holdings as prescribed by the Securities Laws.

"<u>Related Documents</u>" means (a) the Prepetition Debt Documents and (b) all Material Contracts.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 336 of 432

"<u>Related Parties</u>" means, with respect to any Person, such Person's Affiliates and the successors, assigns, partners, directors, officers, employees, agents, trustees, attorneys and advisors of such Person and of such Person's Affiliates.

"<u>Release</u>" means any release, spill, emission, discharge, deposit, disposal, leaking, pumping, pouring, dumping, emptying, injection or leaching into the Environment, or into, from or through any building, structure or facility.

"<u>Reportable Event</u>" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the 30 day notice period has been waived.

"<u>Required Lenders</u>" means, as of any date of determination, Lenders holding more than 50% of the aggregate (a) principal amount of Loans outstanding on such date and (b) Commitments on such date; <u>provided</u> that all Loans and Commitments held or deemed held by any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"<u>Responsible Officer</u>" means the chief executive officer, president, chief financial officer, treasurer, assistant treasurer or controller of a Loan Party. Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"<u>Restricted Payment</u>" means any dividend or other distribution (whether in cash, securities or other property) with respect to any capital stock or other Equity Interest of any Person or any of its Subsidiaries, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, defeasance, acquisition, cancellation or termination of any such capital stock or other Equity Interest, or on account of (x) any return of capital to any Person's stockholders, partners or members (or the equivalent of any thereof), or any option, warrant or other right to acquire any such dividend or other distribution or payment or (y) any principal, premium, interest, expenses or similar payments under any Prepetition Debt Documents (other than as permitted by the Orders).

"<u>Restructuring Support Agreement</u>" means that certain Restructuring Support Agreement (including all exhibits, schedules and attachments thereto) dated on or about the Petition Date, (as amended, supplemented, amended and restated or otherwise modified from time to time in accordance with the terms thereof), by and among the Loan Parties, certain holders of the 6% Notes and the other parties party thereto from time to time.

"<u>S&P</u>" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., and any successor thereto.

"<u>Sanction(s)</u>" means any international economic sanction administered or enforced by the United States Government (including without limitation, OFAC), the United Nations Security Council, the European Union, Her Majesty's Treasury.

"Sarbanes-Oxley" means the Sarbanes-Oxley Act of 2002.

"<u>SEC</u>" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 337 of 432

"Section 6.01 Financials" means the quarterly and annual financial statements required to be delivered pursuant to Sections 6.01(a) and (b).

"<u>Secured Parties</u>" means, collectively, the Administrative Agent, the Lenders, each co-agent or sub-agent appointed by the Administrative Agent from time to time pursuant to <u>Section 9.05</u>, and the other Persons the Obligations owing to which are or are purported to be secured by the Collateral under the terms of the Collateral Documents.

"Securities Account" has the meaning specified in Article 8 of the UCC.

"<u>Securities Laws</u>" means the Securities Act of 1933, the Securities Exchange Act of 1934, Sarbanes-Oxley, and the applicable accounting and auditing principles, rules, standards and practices promulgated, approved or incorporated by the SEC or the PCAOB.

"Security Agreement" has the meaning specified in Section 4.01(a)(iii).

"Settlement Date" has the meaning specified in Section 2.14(b).

"<u>Specified Loan Party</u>" means any Loan Party that is not an "eligible contract participant" under the Commodity Exchange Act (determined prior to giving effect to Section 10.11 of this Agreement and Section 5.15 of the Guaranty).

"Subsidiary" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a Subsidiary or Subsidiaries of Holdings.

"<u>Supermajority Lenders</u>" means those non-Defaulting Lenders which would constitute the Required Lenders under, and as defined in, this Agreement if the percentage "50%" contained therein were changed to "662/3%."

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 338 of 432

"<u>Swap Obligation</u>" means, with respect to any Guarantor, any obligation to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"<u>Swap Termination Value</u>" means, in respect of any one or more Swap Contracts, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Contracts, (a) for any date on or after the date such Swap Contracts have been closed out and termination value(s) determined in accordance therewith, such termination value(s), and (b) for any date prior to the date referenced in clause (a), the amount(s) determined as the mark-to-market value(s) for such Swap Contracts, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Contracts (which may include a Lender or any Affiliate of a Lender).

"Syndication" has the meaning specified in Section 11.06(b).

"Synthetic Debt" means, with respect to any Person as of any date of determination thereof, all obligations of such Person in respect of transactions entered into by such Person that are intended to function primarily as a borrowing of funds (including any minority interest transactions that function primarily as a borrowing) but are not otherwise included in the definition of "Indebtedness" or as a liability on the consolidated balance sheet of such Person and its Subsidiaries in accordance with GAAP.

"<u>Synthetic Lease Obligation</u>" means the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property (including sale and leaseback transactions), in each case, creating obligations that do not appear on the balance sheet of such Person but which, upon the application of any Debtor Relief Laws to such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"Tax Status Certificate" has the meaning set forth in Section 3.01(e).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>Testing Period</u>" means (a) the one (1) calendar week period ending February 9, 2018, (b) the two (2) calendar week period ending February 16, 2018, (c) the three (3) calendar week period ending February 23, 2018, and (d) thereafter, each rolling four (4) calendar week period ending one (1) week after the end of the preceding Testing Period.

"<u>Threshold Amount</u>" means \$5,000,000.

"Total Cash Receipts" means the "total cash receipts" detailed in the Approved Budget.

"Total Operating Disbursements" means the aggregate amount of "total operating cash disbursements" as detailed in the Approved Budget. For the avoidance of doubt, Total Operating Disbursements shall include disbursements related to plant and facility closures and consolidations, and shall exclude each of the following: professional fees, expenses, costs and disbursements (including legal fees, expenses, costs and disbursements) of the Loan Parties, any official committee appointed in the Chapter 11 Cases, or any creditor of the Loan Parties in the Chapter 11 Cases or otherwise, interest and financing fees, and the transition services agreement related to the purchase of the Borrower's Quality Park assets by LSC Communications.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 339 of 432

"Trade Date" has the meaning specified in Section 11.06(b)(i)(B).

"<u>Transactions</u>" (a) the execution, delivery and performance by the Loan Parties and their applicable Subsidiaries of the Loan Documents to which they are or are intended to be a party, (b) the borrowing of Loans and the use of proceeds thereof, (c) the Chapter 11 Cases, (d) the consummation of all other transactions contemplated by or relating to any of the foregoing and (e) the payment of the fees, costs, disbursements and expenses incurred in connection with the consummation of the foregoing.

"Treasury Rate" means the yield to maturity at the time of computation of United States Treasury securities with a constant maturity, as compiled by and published in the most recent statistical release designated as "H.15" under the caption "Treasury constant maturities" or any successor publication which is published at least weekly by the FRB (or companion online data resource published by the FRB) and which establishes yield on actively traded United States Treasury securities adjusted to constant maturity that has become publicly available at least two (2) Business Days prior to date of any prepayment of Loans (or, if such statistical release is no longer published, any publicly available source or similar market data), most nearly equal to the period from the date of such prepayment of Loans to the second anniversary of the Closing Date; provided, that if the period from the date of such prepayment of Loans to the second anniversary of the Closing Date is not equal to the constant maturity of a United States Treasury security for which a weekly average yield is given, the Treasury Rate shall be obtained by linear interpolation (calculated to the nearest one-twelfth of a year) from the weekly average yields of United States Treasury securities for which such yields are given, except that if the period from the date of such prepayment of Loans to the second anniversary of the Closing Date is less than one (1) year, the weekly average yield on actually traded United States Treasury securities adjusted to a constant maturity of one year will be used.

"Type" means, with respect to a Loan, its character as a Base Rate Loan or a Eurodollar Rate Loan.

"<u>UCC</u>" means the Uniform Commercial Code as in effect in the State of New York; <u>provided</u> that, if perfection or the effect of perfection or non-perfection or the priority of any security interest in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of New York, "<u>UCC</u>" means the Uniform Commercial Code as in effect from time to time in such other jurisdiction for purposes of the provisions hereof relating to such perfection, effect of perfection or non-perfection or priority.

"<u>Unfunded Pension Liability</u>" means the excess of a Pension Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the current value of that Pension Plan's assets, determined in accordance with the assumptions used for funding the Pension Plan pursuant to Section 412 of the Code for the applicable plan year.

"<u>United States</u>" and "<u>U.S</u>." mean the United States of America.

"Unreimbursed Amount" has the meaning specified in Section 2.13(e).

"<u>Updated Budget</u>" has the meaning specified in <u>Section 6.17(b)</u>.

"USA PATRIOT Act" has the meaning specified in the definition of "Anti-Terrorism Laws."

"<u>U.S. Loan Party</u>" means any Loan Party that is organized under the laws of one of the states of the United States of America and that is not a CFC.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 340 of 432

"<u>U.S. Person</u>" means any Person that is a "United States person" as defined in Section 7701(a)(30) of the Code.

"<u>Variance</u>" means a difference, expressed as a percentage or number (as applicable), between (a) the amount contained in the Approved Budget with respect to projected Total Cash Receipts, projected Total Operating Disbursements and projected Cumulative Net Operating Cash Flow for any Testing Period and (b) the actual Total Cash Receipts, actual Total Operating Disbursements and actual Cumulative Net Operating Cash Flow, respectively, for such Testing Period.

"<u>Variance Report</u>" has the meaning specified in <u>Section 6.17(b)</u>.

"Variance Reporting Delivery Date" has the meaning specified in Section 6.17(b).

"<u>Write-Down and Conversion Powers</u>": with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule.

1.02 <u>Other Interpretive Provisions</u>.

With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

The definitions of terms herein shall apply equally to the singular and plural (a) forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "will" shall be construed to have the same meaning and effect as the word "shall." Unless the context requires otherwise, (i) any definition of or reference to any agreement, instrument or other document (including any Organization Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein or in any other Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "herein," "hereof" and "hereunder," and words of similar import when used in any Loan Document, shall be construed to refer to such Loan Document in its entirety and not to any particular provision thereof, (iv) all references in a Loan Document to Articles, Sections, Preliminary Statements, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Preliminary Statements, Exhibits and Schedules to, the Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time, and (vi) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

(b) In the computation of periods of time from a specified date to a later specified date, the word "<u>from</u>" means "<u>from and including</u>"; the words "<u>to</u>" and "<u>until</u>" each mean "<u>to but excluding</u>"; and the word "<u>through</u>" means "<u>to and including</u>."

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 341 of 432

(c) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

(d) The Borrower has a fiscal year and fiscal quarters not ending on the calendar year and calendar quarter, respectively. For the avoidance of doubt for so long as such situation exists, when interpreting any provision of this Agreement relating to a fiscal year or fiscal quarter or period ended at or as of the end of a fiscal year or fiscal quarter, including, without limitation, the Financial Covenants, any reference to such a period shall be deemed to refer to the Borrower's fiscal quarter or year ended nearest the given date.

1.03 <u>Accounting Terms</u>.

(a) <u>Generally</u>. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, <u>except</u> as otherwise specifically prescribed herein.

(b) <u>Changes in GAAP</u>. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders shall so request, the Administrative Agent, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); <u>provided</u> that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

(c) <u>Consolidation of Variable Interest Entities</u>. All references herein to consolidated financial statements of Holdings and its Subsidiaries or to the determination of any amount for Holdings and its Subsidiaries on a consolidated basis or any similar reference shall, in each case, be deemed to include each variable interest entity that Holdings is required to consolidate pursuant to FASB Interpretation No. 46 – Consolidation of Variable Interest Entities: an interpretation of ARB No. 51 (January 2003) as if such variable interest entity were a Subsidiary as defined herein.

1.04 <u>Rounding</u>.

Any financial ratios required to be maintained by the Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.05 <u>Times of Day</u>.

Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

1.06 <u>Uniform Commercial Code</u>.

Terms defined in the UCC (as in effect in the State of New York from time to time) and used but not otherwise defined herein shall be used herein in accordance with the UCC in effect in the State of New York from time to time.

ARTICLE II

AMOUNT AND TERMS OF CREDIT

2.01 <u>Commitments</u>.

Subject to the terms and conditions hereof, the Orders and the Approved Budget, and relying upon the representations and warranties herein set forth, the Lenders agree, severally and not jointly, to make term loans to the Borrower (a) once (but not more than once) on the Closing Date in an aggregate principal amount not to exceed \$50,000,000 (the "Initial Loans") and (b) during the Availability Period but on or after the date on which the Final Order is entered by the Bankruptcy Court, in an aggregate principal amount not to exceed the Aggregate Loan Commitments in effect at such time (the "Delayed-Draw Loans" and, together with the Initial Loans, collectively, the "Loans" and, each, a "Loan"); provided, that the Borrower shall not request, and the Lenders shall not be required to make, (x) Delayed-Draw Loans on more than three (3) separate occasions and (y) Delayed-Draw Loans on any Borrowing Date in an aggregate principal amount exceeding the lesser of \$10,000,000 and the Aggregate Loan Commitments in effect at such time. Loans made by the Lenders pursuant to this Section 2.01 shall be made in accordance with their respective Pro Rata Shares. Each Lender's Commitment shall be reduced on a dollar-for dollar basis in an amount equal to the Loans foundered thereunder. Without limiting the generality of the forgoing, in no event shall any Lender be required to make any Loans at any time in an amount that exceeds such Lender's Commitment in effect at such time. Once repaid, the Loans may not be reborrowed. Each Loan not contemplated by the Approved Budget shall only be required to be made if, and only if, requested by the Borrower and approved by the Required Lenders in their reasonable discretion (or the Administrative Agent, acting at the written direction of the Required Lenders). Notwithstanding anything to the contrary contained herein (and without affecting any other provision hereof), the funded portion of Loans to be made on the initial Borrowing Date and on each subsequent Borrowing Date shall be in an amount equal to the Loans requested on such Borrowing Date minus 1.00% of the aggregate principal amount of Loans requested on such Borrowing Date (it being agreed that the full principal amount of any such Loans will be deemed outstanding on and as of the applicable Borrowing Date and the Borrower shall be obligated to repay 100% of the principal amount of each such Loan as provided hereunder).

2.02 <u>Loans</u>.

(a) Each Loan shall be made as part of a Borrowing consisting of Loans made by the Lenders ratably in accordance with their applicable Commitments; <u>provided</u> that the failure of any Lender to make any Loan shall not in itself relieve any other Lender of its obligation to lend hereunder (it being understood, however, that no Lender shall be responsible for the failure of any other Lender to make any Loan required to be made by such other Lender). Loans comprising any Borrowing shall be in an aggregate principal amount that is (i) (A) in the case of Base Rate Loans, not less than \$500,000 and (B) in the case of Eurodollar Rate Loans, an integral multiple of \$250,000 and not less than \$1,000,000, or (ii) equal to the remaining available balance of the applicable Commitments.

(b) Subject to <u>Sections 3.02</u>, <u>3.03</u> and <u>3.04</u>, each Borrowing shall be comprised entirely of Base Rate Loans or Eurodollar Rate Loans as the Borrower may request pursuant to Section 2.03. Each Lender may at its option make any Eurodollar Rate Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 343 of 432

obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement or cause the Borrower to pay additional amounts pursuant to <u>Sections 3.02</u>, <u>3.03</u> and <u>3.04</u>. Borrowings of more than one Type may be outstanding at the same time; provided further that the Borrower shall not be entitled to request any Borrowing that, if made, would result in more than ten Borrowings of Eurodollar Rate Loans outstanding hereunder at any one time. For purposes of the foregoing, Borrowings having different Interest Periods, regardless of whether they commence on the same date, shall be considered separate Borrowings.

(c) Each Lender shall make each Loan to be made by it hereunder on the proposed date thereof by wire transfer of immediately available funds to such account in New York City as the Administrative Agent may designate not later than 3:00 p.m., New York City time, and the Administrative Agent shall promptly credit the amounts so received to the account of the Borrower set forth in the Notice of Borrowing or, if a Borrowing shall not occur on such date because any condition precedent herein specified shall not have been met or waived, return the amounts so received to the respective Lenders.

(d) [Reserved].

(e) Notwithstanding any other provision of this Agreement, the Borrower shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

2.03 <u>Borrowing Procedure</u>.

To request a Borrowing, the Borrower shall notify the Administrative Agent of such request by delivering a Notice of Borrowing in respect of such Loan by telecopy or electronic transmission (if arrangements for doing so have been approved by the Administrative Agent) not later than 11:00 a.m., New York City time, three (3) Business Days before the date of the proposed Borrowing (or such shorter period as the Required Lenders and the Administrative Agent are willing to accommodate in their sole discretion)¹. Each such Notice of Borrowing shall be irrevocable, subject to <u>Sections 2.09</u> and <u>3.01</u>. Each such Notice of Borrowing shall specify the following information in compliance with <u>Section 2.02</u>:

- (a) the aggregate amount of such Borrowing;
- (b) the date of such Borrowing, which shall be a Business Day;

(c) whether such Borrowing is to be a Borrowing of Base Rate Loans or a Borrowing of Eurodollar Rate Loans;

(d) in the case of a Borrowing of Eurodollar Rate Loans, the initial Interest Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Interest Period"; and

(e) that the conditions set forth in <u>Section 4.01</u> or <u>Section 4.02</u>, as applicable, are satisfied or waived as of the date of the notice.

If no election as to the Type of Borrowing is specified, then the requested Borrowing shall be a Borrowing of Base Rate Loans. If no Interest Period is specified with respect to any requested Borrowing of Eurodollar Rate Loans, then the Borrower shall be deemed to have selected an Interest Period of one

¹ NTD: Administrative Agent/Lenders to confirm if shorter notice period is possible for Base Rate Loans.

month's duration (subject to the proviso in clause (d) above). Promptly following receipt of a Notice of Borrowing in accordance with this <u>Section 2.03</u>, the Administrative Agent shall advise each Lender of the details thereof and of the amount of such Lender's Loan to be made as part of the requested Borrowing.

2.04 Evidence of Debt; Repayment of Loans.

(a) The Borrower hereby unconditionally promises to pay to the Administrative Agent, for the account of each Lender, the then unpaid principal amount of each Loan of such Lender on the Maturity Date.

(b) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender resulting from each Loan made by such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement. The Borrower shall be entitled to review records of such accounts with prior reasonable notice during normal business hours.

(c) The Administrative Agent shall maintain accounts in which it will record (i) the amount of each Loan made hereunder, the Type thereof and the Interest Period applicable thereto; (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder; and (iii) the amount of any sum received by the Administrative Agent hereunder for the account of the Lenders and each Lender's share thereof. Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender. The Borrower shall be entitled to review records of such accounts with prior reasonable notice during normal business hours.

(d) The entries made in the accounts maintained pursuant to paragraphs (b) and (c) above shall be *prima facie* evidence of the existence and amounts of the obligations therein recorded absent manifest error; <u>provided</u> that the failure of any Lender or the Administrative Agent to maintain such accounts or any error therein shall not in any manner affect the obligations of the Borrower to repay the Loans in accordance with their terms.

(e) Any Lender may request that Loans made by it be evidenced by a Note. In such event, the Borrower shall promptly prepare, execute and deliver to such Lender a Note payable to such Lender (or, if requested by such Lender, to such Lender and its registered assigns).

2.05 <u>Fees and Payments</u>.

(a) [Reserved].

(b) <u>Lender Payments</u>. The Borrower shall pay to the Administrative Agent, for the ratable benefit of the Lenders (other than any Defaulting Lender), the amounts set forth in the Lender Payment Letter or such other amounts payable at the times separately agreed upon in writing between the Borrower and the Lenders (collectively, the "Lender Payments").

(c) <u>Administrative Agent Fees</u>. The Borrower agrees to pay to the Administrative Agent, for its own account, the fees set forth in the Agent Fee Letter, or such other fees payable in the amounts and at the times separately agreed upon in writing between the Borrower and the Administrative Agent.

(d) All fees, payments and other amounts payable pursuant to this <u>Section 2.05</u> shall be paid on the dates due, in immediately available funds, to the Administrative Agent for distribution, if and as

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 345 of 432

appropriate and on a ratable basis in accordance with their Pro Rata Shares, among the Lenders (other than Defaulting Lenders). Once paid, none of the fees shall be refundable under any circumstances.

2.06 <u>Interest on Loans</u>.

(a) Subject to the provisions of <u>Section 2.06(c)</u>, the Loans comprising each Borrowing of Base Rate Loans shall bear interest at a rate per annum equal to the Base Rate *plus* the Applicable Margin in effect from time to time.

(b) Subject to the provisions of <u>Section 2.06(c)</u>, the Loans comprising each Borrowing of Eurodollar Rate Loans shall bear interest at a rate per annum equal to the Eurodollar Rate for the Interest Period in effect for such Borrowing *plus* the Applicable Margin in effect from time to time.

(c) Notwithstanding the foregoing, if any principal of or interest on any Loan or any fees or other amount payable by the Borrower hereunder is not paid when due, whether at stated maturity, upon acceleration or otherwise, such overdue amount shall bear interest, after as well as before judgment, at a rate per annum equal to (i) in the case of overdue principal of, or interest on, any Loan, 2% *plus* the rate otherwise applicable to such Loan or (ii) in the case of any other amount, 2% *plus* the rate applicable to Base Rate Loans (in each case, the "Default Rate").

(d) Accrued interest on each Loan shall be payable in immediately available funds in arrears on each Interest Payment Date, commencing with the first applicable Interest Payment Date occurring after the Closing Date, for such Loan and upon termination of the Commitments; <u>provided</u> that (i) interest accrued pursuant to paragraph (c) of this <u>Section 2.06</u> shall be payable in cash on demand and, absent demand, on each Interest Payment Date and upon termination of the Commitments, (ii) in the event of any repayment or prepayment of any Loan (other than a prepayment of a Base Rate Loan prior to the end of the Availability Period), accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment and (iii) in the event of any conversion of any Eurodollar Rate Loan prior to the end of the current Interest Period therefor, accrued interest on such Loan shall be payable on the effective date of such conversion.

(e) All interest hereunder shall be computed on the basis of a year of 365/366 days, except that interest computed by reference to the Eurodollar Rate (other than Base Rate Loans determined by reference to the Eurodollar Rate) and all fees shall be computed on the basis of a year of 360 days, and in each case shall be payable for the actual number of days elapsed (including the first day but excluding the last day). The applicable Base Rate or Eurodollar Rate shall be determined by the Administrative Agent in accordance with the provisions of this Agreement and such determination shall be conclusive absent manifest error.

2.07 <u>Termination and Reduction of Commitments.</u>

The Commitments shall automatically terminate on the Maturity Date.

2.08 Interest Elections.

(a) Each Borrowing initially shall be of the Type specified in the applicable Notice of Borrowing and, in the case of a Borrowing of Eurodollar Rate Loans, shall have an initial Interest Period as specified in such Notice of Borrowing. Thereafter, the Borrower may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Borrowing of Eurodollar Rate Loans, may elect Interest Periods therefor, all as provided in this <u>Section 2.08</u>. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 346 of 432

portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing, and the Loans comprising each such portion shall be considered a separate Borrowing. Notwithstanding anything to the contrary, the Borrower shall not be entitled to request any conversion or continuation that, if made, would result in more than three (3) Borrowings of Eurodollar Rate Loans outstanding hereunder at any one time

(b) To make an election pursuant to this <u>Section 2.08</u>, the Borrower shall notify the Administrative Agent of such election by telephone or electronic transmission (if arrangements for doing so have been approved by the Administrative Agent, which approval shall not be unreasonably withheld, delayed or conditioned) by the time that a Notice of Borrowing would be required under <u>Section 2.03</u> if the Borrower was requesting a Borrowing of the Type resulting from such election to be made on the effective date of such election, subject to <u>Section 3.05</u>. Each such telephonic Notice of Conversion/Continuation shall be confirmed promptly by hand delivery or telecopy to the Administrative Agent of a written Notice of Conversion/Continuation substantially in the form of <u>Exhibit A-2</u>, unless otherwise agreed to by the Administrative Agent and the Borrower.

(c) Each telephonic and written Notice of Conversion/Continuation shall specify the following information in compliance with <u>Section 2.02</u>:

(i) the Borrowing to which such Notice of Conversion/Continuation applies and, if different options are being elected with respect to different portions thereof, the portions thereof to be allocated to each resulting Borrowing (in which case the information to be specified pursuant to clauses (iii) and (iv) below shall be specified for each resulting Borrowing);

(ii) the effective date of the election made pursuant to such Notice of Conversion/Continuation, which shall be a Business Day;

(iii) whether the resulting Borrowing is to be a Borrowing of Base Rate Loans or a Borrowing of Eurodollar Rate Loans; and

(iv) if the resulting Borrowing is a Borrowing of Eurodollar Rate Loans, the Interest Period to be applicable thereto after giving effect to such election, which shall be a period contemplated by the definition of the term "Interest Period."

If any such Notice of Conversion/Continuation requests a Borrowing of Eurodollar Rate Loans but does not specify an Interest Period, then the Borrower shall be deemed to have selected an Interest Period of one month's duration.

(d) Promptly following receipt of a Notice of Conversion/Continuation, the Administrative Agent shall advise each Lender of the details thereof and of such Lender's portion of each resulting Borrowing.

(e) If a Notice of Conversion/Continuation with respect to a Borrowing of Eurodollar Rate Loans is not timely delivered prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein, at the end of such Interest Period such Borrowing shall be converted to a Borrowing of Base Rate Loans. Notwithstanding any contrary provision hereof, if an Event of Default has occurred and is continuing and the Administrative Agent, at the request of the Required Lenders, so notifies the Borrower, then, after the occurrence and during the continuance of such Event of Default (i) no outstanding Borrowing may be converted to or continued as a Borrowing of Eurodollar Rate Loans and (ii) unless repaid, each Borrowing of Eurodollar Rate Loans shall be converted to a Borrowing of Base Rate Loans at the end of the Interest Period applicable thereto.

2.09 Optional and Mandatory Prepayments of Loans.

(a) <u>Optional Prepayments</u>. The Borrower shall have the right, at any time and from time to time prior to the Maturity Date, to voluntarily prepay the principal amount of the Loans outstanding at such time, in whole or in part; <u>provided</u>, that each such prepayment shall (i) be accompanied by payment in cash of all accrued and unpaid interest on the principal amount of Loans being prepaid to, but excluding the date of, such prepayment, (ii) be in an amount that is an integral multiple of \$100,000, (iii) be made in accordance with this Agreement (including clauses (c) and (d) of this <u>Section 2.09</u>)) and (iv) be accompanied by payment of the Make-Whole Amount.

(b) <u>Mandatory Prepayments</u>.

(i) [Reserved].

(ii) If on any date any Loan Party or any of its Subsidiaries shall receive any Net Cash Proceeds, then, subject to the Orders, an amount equal to 100% of such Net Cash Proceeds (excluding Net Cash Proceeds in respect of assets of the Loan Parties constituting ABL Priority Collateral that are actually used to prepay the ABL DIP Loans) shall immediately be applied not later than three (3) Business Days after the date of receipt thereof toward the prepayment of the Loans in accordance with this Agreement (including clauses (c) and (d) of this <u>Section 2.09</u>)).

(c) <u>Application of Prepayments</u>.

(i) Prior to any optional or mandatory prepayment of Borrowings hereunder, the Borrower shall select the Borrowing or Borrowings to be prepaid and shall specify such selection in the Notice Of Prepayment delivered pursuant to <u>Section 2.09(d)</u>. Except as provided in <u>Section 2.09(b)(iii)</u> hereof, all mandatory prepayments shall be applied as follows: *first*, to fees and reimbursable expenses of the Administrative Agent and the Lenders then due and payable pursuant to the Loan Documents; *second*, to interest then due and payable on the Loans; *third*, to the principal balance of the Loans outstanding until the same has been prepaid in full; *fourth*, to all other Obligations *pro rata* in accordance with the amounts that such Lender certifies is outstanding; and *fifth*, returned to the Borrower or to such party as otherwise required by law.

(ii) Amounts to be applied pursuant to this <u>Section 2.09</u> to the prepayment of Loans shall be applied, as applicable, first to reduce outstanding Base Rate Loans. Any amounts remaining after each such application shall be applied to prepay Eurodollar Rate Loans. Notwithstanding the foregoing, if the amount of any prepayment of Loans required under this <u>Section 2.09</u> shall be in excess of the amount of the Base Rate Loans at the time outstanding, only the portion of the amount of such prepayment that is equal to the amount of such outstanding Base Rate Loans shall be either (A) applied to the prepayment of Eurodollar Rate Loans on the last day of the then next-expiring Interest Period for Eurodollar Rate Loans (with all interest accruing thereon for the account of the Borrower) or (B) prepaid immediately, together with any amounts owing to the Lenders under <u>Section 2.10</u>. Notwithstanding the foregoing, interest shall continue to accrue on such Loans until prepayment.

(d) <u>Notice of Prepayment</u>. The Borrower shall notify the Administrative Agent of any prepayment hereunder by delivering a Notice of Prepayment in respect of such Loan by telecopy or electronic transmission (if arrangements for doing so have been approved by the Administrative Agent) (i) in the case of prepayment of a Borrowing of Eurodollar Rate Loans, not later than 1:00 p.m., New York City time, three (3) Business Days before the date of prepayment or (ii) in the case of prepayment of a Borrowing of Base Rate Loans, not later than 12:00 noon, New York City time, on the date of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 348 of 432

prepayment. Each such Notice of Prepayment shall specify the prepayment date, the principal amount of each Borrowing or portion thereof to be prepaid and, in the case of a mandatory prepayment, a reasonably detailed calculation of the amount of such prepayment. Each Notice of Prepayment pursuant to this Section shall be irrevocable, except that the Borrower may, by subsequent notice to the Administrative Agent, revoke any such Notice of Prepayment if such notice of revocation is received not later than 10:00 a.m. (New York City time) on the day on which such prepayment is scheduled to occur and, provided that (i) the Borrower reimburses each Lender pursuant to Section 3.05 for any funding losses within five (5) Business Days after receiving written demand therefor and (ii) the amount of Loans as to which such revocation applies shall be deemed converted to (or continued as, as applicable) Base Rate Loans in accordance with the provisions of Section 2.08 as of the date of notice of revocation (subject to subsequent conversion in accordance with the provisions of this Agreement). Promptly following receipt of any such Notice of Prepayment, the Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment of any Borrowing shall be in an amount that would be permitted in the case of an advance of a Borrowing of the same Type as provided in Section 2.02, except as necessary to apply fully the required amount of a mandatory prepayment. Each prepayment of a Borrowing shall be applied ratably to the Loans included in the prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by Section 2.06.

2.10 Payments Generally; Pro Rata Treatment; Sharing of Set-offs.

(a) The Borrower shall make each payment required to be made by it hereunder or under any other Loan Document (whether of principal, interest, fees or of amounts payable under Sections 3.02. 3.03, 3.04 and 3.05 or otherwise) at or before the time expressly required hereunder or under such other Loan Document for such payment (or, if no such time is expressly required, prior to 1:00 p.m., New York City time), on the date when due, in immediately available funds, without setoff or counterclaim. Any amounts received after such time on any date may, in the reasonable discretion of the Administrative Agent, be deemed to have been received on the next succeeding Business Day for purposes of calculating interest thereon. All such payments shall be made to the Administrative Agent at the Payment Office, except that payments pursuant to Sections 3.02, 3.03, 3.04, 3.05 and 11.04 shall be made to the Administrative Agent for the benefit of to the Persons entitled thereto and payments pursuant to other Loan Documents shall be made to the Administrative Agent for the benefit of the Persons specified therein. The Administrative Agent shall distribute any such payments received by it for the account of any other Person to the appropriate recipient promptly following receipt thereof. If any payment under any Loan Document shall be due on a day that is not a Business Day, the date for payment shall be extended to the next succeeding Business Day, and, in the case of any payment accruing interest, interest thereon shall be payable for the period of such extension. All payments under each Loan Document shall be made in Dollars.

(b) If at any time insufficient funds are received by and available to the Administrative Agent to pay fully all amounts of principal, , interest and fees then due hereunder, such funds shall be applied in the manner as provided in Section 2.09(c) or 8.03 hereof, as applicable, ratably among the parties entitled thereto.

(c) If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of its resulting in such Lender receiving payment of a greater proportion of the aggregate amount of its Loans and accrued interest thereon than the proportion received by any other Lender, then the Lender receiving such greater proportion shall purchase (for cash at face value) participations in the Loans of other Lenders to the extent necessary so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Loans; <u>provided</u> that (i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 349 of 432

participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest, and (ii) the provisions of this paragraph shall not be construed to apply to any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans, other than to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this paragraph shall apply). The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Loan Parties rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of a Loan Party in the amount of such participation.

(d) [Reserved].

(e) If any Lender shall fail to make any payment required to be made by it pursuant to <u>Section 2.02(c)</u>, <u>2.02(f)</u>, <u>2.10(d)</u>, <u>2.12(d)</u> or <u>2.13(d)</u>, then the Administrative Agent may, in its discretion (notwithstanding any contrary provision hereof), apply any amounts thereafter received by the Administrative Agent for the account of such Lender to satisfy such Lender's obligations under such Sections until all such unsatisfied obligations are fully paid.

2.11 <u>Defaulting Lenders</u>.

(a) <u>Reallocation of Pro Rata Share; Amendments</u>. For purposes of determining the Lenders' obligations to fund or acquire participations in Loans, the Administrative Agent may exclude the Commitments and Loans of any Defaulting Lender(s) from the calculation of Pro Rata Shares and any Commitments of any such Defaulting Lender shall automatically be reallocated among the non-Defaulting Lenders pro rata in accordance with their Commitments, so long as the conditions set forth in <u>Section 4.02</u> are satisfied at the time of such reallocation. A Defaulting Lender shall have no right to vote on any amendment, waiver or other modification of a Loan Document, except as provided in <u>Section 11.01</u>.

Payments; Fees. Any payment of principal, interest, fees or other amounts received by (b) the Administrative Agent for the account of a Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Article VIII or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 11.08) shall be applied at such time or times as may be determined by the Administrative Agent as follows: *first*, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; third, if so determined by the Administrative Agent or the Borrower, to be held in a Deposit Account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; *fourth*, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and *fifth*, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans were made at a time when the conditions set forth in Section 4.02 were satisfied or waived, such payment shall be applied solely to pay the Loans of, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this Section 2.11(b) shall be deemed paid to and redirected by that

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 350 of 432

Defaulting Lender, and each Lender irrevocably consents hereto. A Lender shall not be entitled to receive any fees accruing hereunder during the period in which it is a Defaulting Lender.

(c) <u>Cure</u>. The Borrower and the Administrative Agent (acting at the direction of the Required Lenders) may agree in writing that a Lender is no longer a Defaulting Lender. At such time, Pro Rata Shares shall be reallocated without exclusion of such Lender's Commitments and Loans, and all outstanding Loans and other exposures under the Commitments shall be reallocated among the Lenders and settled by the Administrative Agent (with appropriate payments by the reinstated Lender) in accordance with the readjusted Pro Rata Shares. Unless expressly agreed in writing by the Borrower, the Administrative Agent (which shall make such determination in its discretion), no reinstatement of a Defaulting Lender shall constitute a waiver or release of claims against such Lender. The failure of any Lender to fund a Loan or otherwise to perform its obligations hereunder shall not relieve any other Lender of its obligations, and no Lender shall be responsible for default by another Lender. No reallocation hereunder shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender as a result of such non-Defaulting Lender's increased exposure following such reallocation.

ARTICLE III

TAXES, YIELD PROTECTION AND ILLEGALITY

3.01 <u>Taxes</u>.

(a) <u>Payments Free of Taxes</u>. Any and all payments by, or on account of any Obligation of, any Loan Party hereunder or under any other Loan Document shall be made free and clear of and without reduction or withholding for any Taxes, provided that if the applicable withholding agent shall be required by applicable Law to deduct any Taxes in respect of such payments, then (i) if the Tax in question is an Indemnified Tax, the sum payable by the applicable Loan Party shall be increased as necessary so that after all required deductions have been made (including deductions applicable to additional sums payable under this <u>Section 3.01</u>) the Administrative Agent or any Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions been made, (ii) the applicable withholding agent shall make such deductions and (iii) the applicable withholding agent shall timely pay the full amount deducted to the relevant Governmental Authority in accordance with applicable Law.

(b) <u>Payment of Other Taxes by the Loan Parties</u>. Without limiting the provisions of subsection (a) above, the Loan Parties shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Law.

(c) <u>Indemnification by the Loan Parties</u>. The Loan Parties shall jointly and severally indemnify the Administrative Agent and each Lender, within 10 days after demand therefor, for the full amount of any Indemnified Taxes with respect to amounts payable under the Loan Documents (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this <u>Section 3.01</u>) paid by the Administrative Agent or such Lender, as the case may be, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 351 of 432

(d) <u>Evidence of Payments</u>. As soon as practicable after any payment of Indemnified Taxes by any Loan Party to a Governmental Authority, the Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

(e) <u>Status of Lenders</u>.

The Administrative Agent and any Lender that is entitled to an exemption from (i) or reduction of withholding Tax with respect to any payments hereunder or under any other Loan Document shall deliver to the Borrower, Holdings, such other Loan Party and the Administrative Agent, at the time or times reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent, such properly completed and executed documentation prescribed by applicable Law, or reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent, as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, the Administrative Agent and any Lender, if requested by the Borrower, Holdings, any other Loan Party or the Administrative Agent, shall deliver such other documentation prescribed by applicable Law or reasonably requested by the Borrower, Holdings, such other Loan Party or the Administrative Agent as will enable the Borrower, Holdings, such other Loan Party or the Administrative Agent to determine whether or not such Administrative Agent or such Lender is subject to backup withholding or information reporting requirements. The Administrative Agent and each Lender shall, whenever a lapse in time or change in circumstances renders any documentation (including any specific documentation required below in this Section 3.01(e)) obsolete, expired or inaccurate in any respect, deliver promptly to the Borrower, Holdings, such other Loan Party and the Administrative Agent updated or other appropriate documentation or promptly notify the Borrower, Holdings, such other Loan Party and the Administrative Agent of its legal inability to do so.

(ii) Without limiting the generality of the foregoing, if the Borrower or Holdings, as the case may be, is a U.S. Person:

(A) any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or about the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed copies of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax;

(B) any Foreign Lender shall deliver to the Borrower, Holdings and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender or a party under this Agreement whichever of the following is applicable:

(1) duly completed copies of Internal Revenue Service Form W-8BEN or W-8BEN-E claiming eligibility for benefits of an income tax treaty to which the United States is a party,

(2) duly completed copies of Internal Revenue Service Form W-8ECI or W-8EXP,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 352 of 432

(3) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (A) a certificate (a "<u>Tax Status</u> <u>Certificate</u>") to the effect that such Foreign Lender is not (1) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (2) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (3) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code, and that no interest payments will be considered effectively connected income and (B) duly completed copies of Internal Revenue Service Form W-8BEN or W-8BEN-E,

(4) where such Foreign Lender is a partnership (for U.S. federal income tax purposes) or otherwise not a beneficial owner (e.g., where such Lender has sold a participation), IRS Form W-8IMY (or any successor thereto) and all required supporting documentation, including IRS Form W-8ECI, IRS Form W-8BEN, IRS Form W-8BEN-E, a Tax Status Certificate, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable (provided that, if the Foreign Lender is a partnership and not a participating Lender, the Tax Status Certificate from the beneficial owner(s) may be provided by the Foreign Lender on the beneficial owner(s) behalf), and/or

(5) any other form prescribed by applicable Law as a basis for claiming exemption from or a reduction in United States federal withholding tax duly completed together with such supplementary documentation as may be prescribed by applicable Law to permit the Borrower to determine the withholding or deduction required to be made.

If a payment made to a Foreign Lender or the Administrative Agent (C) hereunder or under any other Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender or the Administrative Agent (as the case may be) were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable, or in any agreement or request entered into or issued pursuant to such Sections), such Lender or the Administrative Agent (as the case may be) shall deliver to the Borrower, Holdings and (in the case of such Lender) to the Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower, Holdings or the Administrative Agent, such documentation, certifications or other information prescribed by applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation, certifications or other information reasonably requested by the Borrower, Holdings or the Administrative Agent as may be necessary for the Borrower, Holdings or the Administrative Agent to comply with its obligations under FATCA, to determine whether such Lender or the Administrative Agent (as the case may be) has complied with its obligations under FATCA or to determine the amount, if any, to deduct and withhold from such payment. Solely for purposes of this paragraph, "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Notwithstanding anything to the contrary in this <u>Section 3.01(e)</u>, neither the Administrative Agent nor any Lender shall be required to deliver any documentation that it is not legally able to deliver.

(f) <u>Treatment of Certain Refunds</u>. If the Administrative Agent or any Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes as to which it has been indemnified by any Loan Party, or with respect to which any Loan Party has paid additional amounts

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 353 of 432

pursuant to this <u>Section 3.01</u>, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Loan Parties under this <u>Section 3.01</u> with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of the Administrative Agent or such Lender, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), <u>provided</u> that the Borrower, upon the request of the Administrative Agent or such Lender, agrees to repay the amount paid over to such Loan Party (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent or such Lender if the Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This Section shall not be construed to require the Administrative Agent or any Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to the Borrower, Holdings or any other Person.

(g) <u>Survival</u>. Each party's obligations under this Section shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.

3.02 <u>Illegality</u>.

If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund Eurodollar Rate Loans, or to determine or charge interest rates based upon the Eurodollar Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the London interbank market, then, on notice thereof by such Lender to the Borrower through the Administrative Agent, any obligation of such Lender to make or continue Eurodollar Rate Loans or to convert Base Rate Loans to Eurodollar Rate Loans shall be suspended until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrower shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all Eurodollar Rate Loans of such Lender to Base Rate Loans, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such Eurodollar Rate Loans. Upon any such prepayment or conversion, the Borrower shall also pay accrued interest on the amount so prepaid or converted.

3.03 <u>Inability to Determine Rates</u>.

If the Required Lenders determine that for any reason in connection with any request for a Eurodollar Rate Loan or a conversion to or continuation thereof that (a) Dollar deposits are not being offered to banks in the London interbank eurodollar market for the applicable amount and Interest Period of such Eurodollar Rate Loan, (b) adequate and reasonable means do not exist for determining the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan, or (c) the Eurodollar Rate for any requested Interest Period with respect to a proposed Eurodollar Rate Loan does not adequately and fairly reflect the cost to such Lenders of funding such Loan, the Administrative Agent will promptly so notify the Borrower and each Lender. Thereafter, the obligation of the Lenders to make or maintain Eurodollar Rate Loans shall be suspended until the Administrative Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, the Borrower may revoke any pending request for a Borrowing of, conversion to or continuation of Eurodollar Rate Loans or, failing that, will be deemed to have converted such request into a request for a Borrowing of Base Rate Loans in the amount specified therein.

3.04 Increased Costs; Reserves on Eurodollar Rate Loans.

(a) <u>Increased Costs Generally</u>. If any Change in Law shall:

(i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement contemplated by Section 3.04(e));

(ii) subject any Lender to any Tax of any kind whatsoever with respect to this Agreement, or any Eurodollar Rate Loan made by it, in respect thereof (in each case, except for Indemnified Taxes, any Taxes described in clauses (b) through (d) of the definition of Excluded Tax and Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes); or

(iii) impose on any Lender or the London interbank market any other condition, cost or expense affecting this Agreement or Eurodollar Rate Loans made by such Lender;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Eurodollar Rate Loan (or of maintaining its obligation to make any such Loan), or to increase the cost to such Lender, or to reduce the amount of any sum received or receivable by such Lender hereunder (whether of principal, interest or any other amount) then, upon request of such Lender, the Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender for such additional costs incurred or reduction suffered.

(b) <u>Capital Requirements</u>. If any Lender determines that any Change in Law affecting such Lender or any Lending Office of such Lender or such Lender's holding company, if any, regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of this Agreement, the Commitments of such Lender or the Loans made by such Lender, to a level below that which such Lender or such Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to capital adequacy), then from time to time the Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender or such Lender's holding company for any such reduction suffered.

(c) <u>Certificates for Reimbursement.</u> A certificate of a Lender setting forth the amount or amounts necessary to compensate such Lender or its holding company, as the case may be, as specified in subsection (a) or (b) of this Section and delivered to the Borrower shall be conclusive absent manifest error. The Borrower shall pay such Lender the amount shown as due on any such certificate within 10 days after receipt thereof.

(d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender to demand compensation pursuant to the foregoing provisions of this Section shall not constitute a waiver of such Lender's right to demand such compensation, <u>provided</u> that the Borrower shall not be required to compensate a Lender pursuant to the foregoing provisions of this Section for any increased costs incurred or reductions suffered more than 270 days prior to the date that such Lender notifies the Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the 270 day period referred to above shall be extended to include the period of retroactive effect thereof).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 355 of 432

Reserves on Eurodollar Rate Loans. The Borrower shall pay to each Lender, as long as (e) such Lender shall be required to maintain reserves with respect to liabilities or assets consisting of or including Eurocurrency funds or deposits (currently known as "Eurocurrency liabilities," as defined in Regulation D of the FRB), additional interest on the unpaid principal amount of each Eurodollar Rate Loan equal to the actual costs of such reserves allocated to such Eurodollar Rate Loan by such Lender (as determined by such Lender in good faith, which determination shall be conclusive), which shall be due and payable on each date on which interest is payable on such Eurodollar Rate Loan; provided, the Borrower shall have received at least 10 days' prior notice (with a copy to the Administrative Agent) of such additional interest from such Lender together with a schedule containing a reasonably detailed explanation and calculation thereof; and provided, further, that such additional interest shall not exceed an amount based upon an interest rate per annum equal to the remainder obtained by subtracting (i) the Eurodollar Rate for the Interest Period for such Eurodollar Rate Loan from (ii) the rate obtained by dividing such Eurodollar Rate by a percentage equal to 100% minus the Eurodollar Reserve Percentage of such Lender for such Interest Period. If a Lender fails to give notice 10 days prior to the relevant Interest Payment Date, such additional interest shall be due and payable 10 days from receipt of such notice.

3.05 <u>Compensation for Losses</u>.

Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrower shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:

(a) any continuation, conversion, payment or prepayment of any Eurodollar Rate Loan on a day other than the last day of the Interest Period for such Eurodollar Rate Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);

(b) any failure by the Borrower (for a reason other than the failure of a Lender to make a Loan) to prepay, borrow, continue or convert any Eurodollar Rate Loan on the date or in the amount notified by the Borrower; or

(c) any assignment of a Eurodollar Rate Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Borrower pursuant to <u>Section 11.13;</u>

excluding any loss of anticipated profits but including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Eurodollar Rate Loan or from fees payable to terminate the deposits from which such funds were obtained. The Borrower shall also pay any customary administrative fees charged by such Lender in connection with the foregoing.

For purposes of calculating amounts payable by the Borrower to the Lenders under this <u>Section 3.05</u>, each Lender shall be deemed to have funded each Eurodollar Rate Loan made by it at the Eurodollar Rate for such Eurodollar Rate Loan by a matching deposit or other borrowing in the London interbank eurodollar market for a comparable amount and for a comparable period, whether or not such Eurodollar Rate Loan was in fact so funded.

3.06 <u>Mitigation Obligations; Replacement of Lenders.</u>

(a) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.04</u>, or the Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, or if any Lender gives a notice pursuant to <u>Section 3.02</u>, then such Lender shall use commercially reasonable efforts to designate a different Lending Office for funding or booking its Loans hereunder or to assign its

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 356 of 432

rights and obligations hereunder to another of its offices, branches or affiliates, if, in the good faith judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.01</u> or <u>3.04</u>, as the case may be, in the future, or eliminate the need for the notice pursuant to <u>Section 3.02</u>, as applicable, and (ii) in each case, in the good faith judgment of such Lender, would not subject such Lender to any unreimbursed cost or expense and would not cause such Lender to suffer any economic, regulatory or other disadvantage (except to a *de minimis* extent).. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment. For the avoidance of doubt, nothing in this <u>Section 3.06</u> shall affect or postpone any of the obligations of the Borrower or the rights of any Lender pursuant to <u>Sections 3.01</u>, <u>3.04 and 3.05</u>.

(b) <u>Replacement of Lenders</u>. If any Lender requests compensation under <u>Section 3.04</u>, or if the Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, the Borrower may replace such Lender in accordance with <u>Section 11.13</u>.

3.07 <u>Survival</u>.

All of the Borrower's obligations under this <u>Article III</u> shall survive the resignation or replacement of the Administrative Agent, any assignment of rights by or replacement of a Lender, termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all other Obligations hereunder.

ARTICLE IV

CONDITIONS PRECEDENT TO BORROWINGS

4.01 <u>Conditions of Initial Borrowing</u>.

This Agreement and the obligations of the Administrative Agent and the Lenders hereunder shall not become effective until the Business Day (the "<u>Closing Date</u>") on which the following conditions are satisfied in a manner satisfactory to, or waived in writing by, the Required Lenders:

(a) <u>Documents, Certificates, Opinions and Other Instruments</u>. The Administrative Agent's receipt of a copy of the following, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to the Required Lenders and the Administrative Agent:

(i) counterparts of this Agreement, the Payment Letters and the Guaranty, in each case, duly and properly executed and delivered by each of the parties thereto;

(ii) a Note duly and properly executed by a Responsible Officer of the Borrower in favor of each Lender requesting a Note;

(iii) a pledge and security agreement (together with each other pledge and security agreement and pledge and security agreement supplement delivered by any Loan Party pursuant to <u>Section 6.12</u>, in each case as amended, the "<u>Security Agreement</u>"), duly and properly executed by each of the parties thereto, together with:

(A) certificates (if any) representing the Pledged Equity referred to therein accompanied by undated stock powers executed in blank by the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 357 of 432

applicable Loan Party and instruments evidencing the Pledged Debt (if any) indorsed in blank by the applicable Loan Party,

(B) proper financing statements in form appropriate for filing under the Uniform Commercial Code of all jurisdictions that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Security Agreement covering the Collateral described therein,

(C) certified copies of UCC, insolvency, tax, judgment lien and execution searches, or equivalent reports or searches, each of a recent date listing all effective financing statements, lien notices or comparable documents (together with copies of such financing statements and documents) that name any Loan Party as debtor and that are filed in those state, territory and county jurisdictions in which any Loan Party is organized or maintains its principal place of business, and such other searches that are required by the Perfection Certificate or that the Administrative Agent deems necessary or appropriate,

(D) evidence of the completion of all other actions, recordings and filings of or with respect to the Security Agreement that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created thereby,

(E) a Perfection Certificate, duly executed and properly completed by a Responsible Officer of each Loan Party, and

(F) evidence that all other action that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Security Agreement and/or the Orders, as applicable, has been taken;

- (iv) [reserved];
- (v) [reserved];

(vi) such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as the Administrative Agent may require evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party or is to be a party;

(vii) such documents and certifications as the Administrative Agent may reasonably require, including without limitation copies of each Loan Party's Organizational Documents and certificates of good standing, to evidence that each Loan Party is duly organized or formed and is validly existing, in good standing and qualified to engage in business in each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect;

(viii) [reserved];

(ix) a favorable opinion of Kirkland & Ellis LLP, special counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to such matters concerning the Loan Parties and the Loan Documents as the Required Lenders may reasonably request;

(x) a favorable opinion of Polsinelli, special Colorado, Georgia, Maryland and Virginia counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to such matters concerning the Loan Parties and the Loan Documents as the Required Lenders may reasonably request;

(xi) [reserved];

(xii) a favorable opinion of Bose McKinney & Evans LLP, special Indiana counsel to the Loan Parties, addressed to the Administrative Agent and each Lender, as to such matters concerning the Loan Parties and the Loan Documents as the Required Lenders may reasonably request;

(xiii) each other Loan Document required by the Administrative Agent or the Required Lenders to be effective on the Closing Date, in each case, duly and properly executed by each of the parties thereto;

(xiv) [reserved];

(xv) a certificate of a Responsible Officer of each Loan Party either (A) attaching copies of all consents, licenses and approvals required in connection with the execution, delivery and performance by such Loan Party and the validity against such Loan Party of the Loan Documents to which it is a party, and such consents, licenses and approvals shall be in full force and effect, or (B) stating that no such consents, licenses or approvals are so required;

(xvi) a certificate signed by a Responsible Officer of the Borrower certifying that the conditions specified in Sections 4.02(a) and (b) have been satisfied;

(xvii) [reserved];

(xviii) evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect, together with the certificates of insurance, naming the Administrative Agent, on behalf of the Lenders, as an additional insured or loss payee, as the case may be, under all insurance policies maintained with respect to the assets and properties of the Loan Parties that constitutes Collateral; and

(xix) such other assurances, certificates, documents, consents or opinions as the Administrative Agent or any Required Lender reasonably may require.

(b) <u>Initial Budget</u>. Administrative Agent shall have received the Initial Budget, in form and substance satisfactory to the Required Lenders.

(c) <u>Patriot Act.</u> Each Loan Party shall have provided such documentation and other information reasonably requested by the Lenders and/or the Administrative Agent in connection with regulatory requirements under the applicable "know your customer" and anti-money-laundering rules and regulations, including without limitation the PATRIOT Act.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 359 of 432

(d) <u>ABL DIP Facility.</u> The ABL DIP Facility shall have been entered into prior to or concurrently with entry into this Agreement and the Administrative Agent shall have received a copy of the ABL DIP Credit Agreement and the Ratification Agreement, duly and properly executed and delivered by each of the parties thereto, and the ABL DIP Credit Agreement and the Ratification Agreement and the Ratification Agreement shall be in full force and effect.

Without limiting the generality of the provisions of <u>Section 9.04</u>, for purposes of determining compliance with the conditions specified in this <u>Section 4.01</u>, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

4.02 <u>Conditions to All Borrowings</u>.

The obligation of each Lender to make any Loans (including the Initial Loans and any Delayed-Draw Loans) shall be subject to the occurrence of the Closing Date and the satisfaction in a manner satisfactory to, or waiver in writing by), the Required Lenders of each of the conditions precedent set forth below (and, with respect to <u>Section 4.02(e)</u> only, the Administrative Agent):

(a) <u>Representations and Warranties</u>. The representations and warranties of the Borrower and each other Loan Party contained in <u>Article V</u> or any other Loan Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct in all material respects (except to the extent already qualified by materiality, in which case such representation and warranty shall be true and correct in all respects) on and as of the date of such Borrowing, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects (except to the extent already qualified by materiality, in which case such representation and warranties specifically refer to the extent already qualified by materiality, in which case such representation and warranty shall be true and correct in all respects) as of such earlier date, and except that for purposes of this <u>Section 4.02</u>, the representations and warranties contained in <u>Sections 5.05(a)</u> and (b) shall be deemed to refer to the most recent statements furnished pursuant to <u>Sections 6.01(a)</u> and (b), respectively.

(b) <u>No Default or Violation of Law</u>. No Default or Event of Default shall exist, or would result from such proposed Borrowing or from the application of proceeds thereof. The making of such Loan shall not violate any applicable Law in any material respect.

(c) <u>Notice of Borrowing</u>. The Administrative Agent shall have received a Notice of Borrowing in respect of each applicable Loan in accordance with the requirements of <u>Section</u> 2.03.

(d) <u>Material Adverse Change</u>. Since December 31, 2017, there shall not have occurred any event, development, circumstance or change that could reasonably be expected to, individually or in the aggregate with any other such event, development, circumstance or change, have a Material Adverse Effect.

(e) <u>Fees and Other Payments</u>. The Borrower shall have paid in full in cash all reasonable and documented fees, costs, charges, expenses and disbursements payable pursuant to the Loan Documents or otherwise in connection with the Transactions of (i) the Administrative Agent, (ii) the Lenders, (iii) (x) counsel to the Administrative Agent and (y) the Lender Advisors, in each case, to the extent invoiced prior to the proposed Borrowing Date.

(f) [Reserved].

(g) <u>Restructuring Support Agreement</u>. The Administrative Agent shall have received a copy of the Restructuring Support Agreement, duly executed and delivered by each of the parties thereto, and such Restructuring Support Agreement shall be in full force and effect and shall not have been amended, waived, terminated or otherwise modified without the prior written consent of the Required Lenders.

(h) <u>Approved Budget</u>. The Initial Budget or such other Approved Budget shall be in full force and effect on and as of the proposed Borrowing Date, and the making of any Loan on any Borrowing Date shall be in accordance with the Initial Budget or such other Approved Budget.

(i) <u>Motions and Documents</u>. All (i) first day motions filed by the Loan Parties on the Petition Date and related orders entered by the Bankruptcy Court in the Chapter 11 Cases shall be in form and substance satisfactory to the Required Lenders and (ii) DIP Motions and all documents to be filed with and submitted to the Bankruptcy Court related to this Agreement or the Loans (including relating to the approval hereof and thereof) shall be in form and substance satisfactory to the Required Lenders.

(j) Interim Order/Final Order. With respect to Initial Loans, the Bankruptcy Court shall have entered the Interim Order not later than five (5) calendar days after the Petition Date, which Interim Order shall, at all times and on each Borrowing Date prior to the entry of the Final Order, be in full force and effect and shall not have been reversed, vacated or stayed, and shall not have been amended, supplemented or otherwise modified without the prior written consent of the Required Lenders. With respect to Delayed-Draw Loans, the Bankruptcy Court shall have entered the Final Order, which Final Order shall, on each Borrowing Date on and after the entry of the Final Order, be in full force and effect and shall not have been reversed, vacated or stayed, and shall not have been amended, supplemented or otherwise modified without the prior written consent of the Final Order, be in full force and effect and shall not have been reversed, vacated or stayed, and shall not have been amended, supplemented or otherwise modified without the prior written consent of the Required Lenders.

(k) <u>Validity and Priority of Liens</u>. The Administrative Agent, for the benefit of the Secured Parties, shall have valid, binding, enforceable, non-avoidable, and automatically and fully and properly perfected Liens on, and security interests in, the Collateral, in each case, having the priorities set forth in the Interim Order and/or the Final Order and subject only to the Carve-Out.

(1) <u>Cash Management System</u>. The Loan Parties shall have established a cash management system approved by the Cash Management Order and satisfactory to the Required Lenders in all material respects, and all documentation relating to the foregoing shall have been executed and delivered to the Administrative Agent.

(m) <u>Milestones</u>. The Loan Parties shall have satisfied all Milestones which are required to be satisfied on or prior to the applicable Borrowing Date.

(n) <u>Other</u>. The Borrower shall have delivered to the Administrative Agent and the Lenders such other documents and information with respect to the business, property, condition (financial or otherwise), legal, financial or corporate or similar affairs or operations of any Loan Party or any of their Subsidiaries reasonably requested by the Administrative Agent or such Lender reasonably in advance of any such Borrowing.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 361 of 432

(o) <u>Cleansing Materials</u>. The Borrower shall have taken all actions (including publicly disclose all information, documents and other materials) as are required pursuant to applicable confidentiality and nondisclosure agreements or requested by the Required Lenders to ensure that, as of the Closing Date, none of the Lenders or any of their Affiliates are in possession of any material non-public information relating to the Loan Parties or any of their Subsidiaries or any of their respective securities, or otherwise constitute Public Lenders.

Each Notice of Borrowing (other than a Notice of Borrowing requesting only a conversion of Loans of one Type to another Type or a continuation of Eurodollar Rate Loans) submitted by the Borrower shall be deemed to be a representation and warranty that the conditions specified in <u>Sections 4.02(a)</u> and (b) have been satisfied on and as of the date of the applicable Borrowing.

ARTICLE V

REPRESENTATIONS AND WARRANTIES

Each of Holdings and the Borrower represents and warrants to the Administrative Agent and the Lenders that as of the Closing Date, each Borrowing Date and each date of delivery of a quarterly or annual financial statement pursuant to <u>Section 6.01</u> (each, a "<u>Credit Date</u>"):

5.01 Existence, Qualification and Power.

Each Loan Party is a Chapter 11 Debtor. Each Loan Party and each of its Subsidiaries (a) is duly organized or formed, validly existing and, as applicable, in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) subject to any necessary Bankruptcy Court approval, including, without limitation, entry of the Interim Order or the Final Order, as the case may be, has all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own or lease its assets and carry on its business and (ii) execute, deliver and perform its obligations under the Loan Documents or Material Contracts to which it is a party and consummate the Transaction, and (c) is duly qualified and is licensed and, as applicable, in good standing under the Laws of each jurisdiction where its ownership, lease or operation of properties or the conduct of its business requires such qualification or license; except in each case referred to in clause (b)(i) or (c), to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.

5.02 <u>Authorization; No Contravention</u>.

Subject to the entry and terms of the Orders, the execution, delivery and performance by each Loan Party of each Loan Document to which such Person is or is to be a party have been duly authorized by all necessary corporate or other organizational action, and do not and will not (a) contravene the terms of any of such Person's Organization Documents; (b) conflict with or result in any breach or contravention of, or result in the creation of any Lien under (i) any Contractual Obligation to which such Person is a party or binding upon such Person or the properties of such Person or any of its Subsidiaries or (ii) any order, injunction, writ or decree of any Governmental Authority or any arbitral award to which such Person or its property is subject, except, in each case, where enforcement is stayed upon commencement of the Cases; or (c) violate any applicable Law in any material respect.

5.03 <u>Governmental Authorization; Other Consents.</u>

No approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any Governmental Authority or any other Person is necessary or required in connection with (a) the execution, delivery or performance by, or enforcement against, any Loan Party of this Agreement or any

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 362 of 432

other Loan Document or Material Contract, or for the consummation of the Transaction, (b) the grant by any Loan Party of the Liens granted by it pursuant to the Collateral Documents, (c) the perfection or maintenance of the Liens created under the Collateral Documents (including the first priority nature thereof (subject to the Orders)) or (d) the exercise by the Administrative Agent or any Lender of its rights under the Loan Documents or the remedies in respect of the Collateral pursuant to the Collateral Documents, except for (i) filings and recordings necessary to perfect Liens created under the Collateral Documents, (ii) with respect to clause (d) above, any approvals, authorizations or filings that may be required under the Securities Laws and (iii) such authorizations, approvals, actions, notices and filings that have been duly obtained, taken, given or made and are in full force and effect, including the applicable Order. All applicable waiting periods in connection with the Transaction have expired without any action having been taken by any Governmental Authority restraining, preventing or imposing materially adverse conditions upon the Transaction or the rights of the Loan Parties or their Subsidiaries freely to transfer or otherwise dispose of, or to create any Lien on, any properties now owned or hereafter acquired by any of them.

5.04 <u>Binding Effect</u>.

Subject to the entry of and the terms of the Orders, this Agreement has been, and each other Loan Document, when delivered hereunder, will have been, duly executed and delivered by each Loan Party that is party thereto. Subject to the entry of and the terms of the Orders, this Agreement constitutes, and each other Loan Document when so delivered will constitute, a legal, valid and binding obligation of each Loan Party that is party thereto, enforceable against such Loan Party that is party thereto in accordance with its terms, except as enforceability may be limited by applicable Debtor Relief Laws and by equitable principles regardless of whether considered in a proceeding in equity or at law.

5.05 <u>Financial Statements; No Material Adverse Effect; No Internal Control Event.</u>

(a) The Audited Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present the financial condition of Holdings and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; and (iii) show all material indebtedness and other liabilities, direct or contingent, of Holdings and its Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness.

(b) The unaudited consolidated balance sheet of Holdings and its Subsidiaries as at December 31, 2016, and the related consolidated statements of income or operations, and cash flows for the fiscal quarter ended on that date (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein, and (ii) fairly present the financial condition of Holdings and its Subsidiaries as of the date thereof and their results of operations for the period covered thereby, subject, in the case of clauses (i) and (ii), to the absence of footnotes and to normal quarter-end or year-end adjustments.

(c) Since the Petition Date, except as disclosed in Holdings' public filings with the SEC made prior to the Closing Date, there has been no event or circumstance, either individually or in the aggregate, that has had or could reasonably be expected to have a Material Adverse Effect.

(d) To the knowledge of Holdings and the Borrower, no Internal Control Event exists or has occurred since the date of the Audited Financial Statements that has resulted in or could reasonably be expected to result in a misstatement in any material respect, in any financial information delivered or to be delivered to the Administrative Agent or the Lenders, of (i) covenant compliance calculations provided

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 363 of 432

hereunder or (ii) the assets, liabilities, financial condition or results of operations of Holdings and its Subsidiaries on a consolidated basis.

(e) The Approved Budget delivered by or on behalf of the Borrower to the Administrative Agent or the Lenders during the period prior to the Closing Date were prepared in good faith on the basis of the assumptions stated therein, which assumptions were reasonable in light of the conditions existing at the time of delivery of such forecasts. There is no fact known to Holdings, the Borrower or any of its Subsidiaries which could reasonably be expected to have a Material Adverse Effect which has not been disclosed herein.

5.06 <u>Litigation</u>.

Other than the Chapter 11 Cases, or as stayed upon commencement of the Chapter 11 Cases, there are no actions, suits, proceedings, investigations, claims or disputes pending or, to the knowledge of Holdings or the Borrower after due and diligent investigation, threatened or contemplated, at law, in equity, in arbitration or before any Governmental Authority, by or against Holdings, the Borrower or any of Holding's other Subsidiaries or against any of their properties or revenues that (a) purport to affect or pertain to this Agreement, any other Loan Document or any Related Document or (b) either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.07 <u>No Default</u>.

Other than defaults directly arising solely as a result of the commencement of the Chapter 11 Cases, neither any Loan Party nor any Subsidiary thereof is in default under or with respect to any Contractual Obligation in any manner that could, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. No Default or Event of Default has occurred and is continuing or would result from the consummation of the Transactions.

5.08 <u>Ownership of Property; Liens; Investments</u>.

(a) Subject to the rights of counterparties under the applicable provisions of the Bankruptcy Code, including, without limitation, Section 365 thereof, each Loan Party and each of its Subsidiaries has good record and marketable title in fee simple to, or valid leasehold interests in, all real property necessary or used in the ordinary conduct of its business, except for such defects in title as could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(b) <u>Schedule 5.08(b)</u> sets forth a complete and accurate list as of the Closing Date of all Liens (except Liens securing the Obligations) on the property or assets of each Loan Party and each of its Subsidiaries, showing as of the Closing Date the property or assets of such Loan Party or such Subsidiary subject thereto. The property of each Loan Party and each of its Subsidiaries is subject to no Liens, other than Liens set forth on <u>Schedule 5.08(b)</u>, and as otherwise permitted by <u>Section 7.01</u>.

(c) <u>Schedule 5.08(c)</u> sets forth a complete and accurate list as of the Closing Date of all real property owned by each Loan Party and each of its Subsidiaries, showing as of the Closing Date the street address, county or other relevant jurisdiction, state, record owner and fair market value thereof (as reasonably determined by the Borrower). Each Loan Party and each of its Subsidiaries has good, marketable and insurable fee simple title to the real property owned by such Loan Party or such Subsidiary, free and clear of all Liens, other than Liens created or permitted by the Loan Documents or the Orders.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 364 of 432

(d) <u>Schedules 5.08(d)(i)</u> and <u>(ii)</u> set forth as of the Closing Date a complete and accurate list of all leases of real property under which any Loan Party or any Subsidiary of a Loan Party is the lessee or the lessor, respectively, showing as of the Closing Date the street address, county or other relevant jurisdiction, state, lessor, lessee and annual rental cost thereof. Subject to the provisions of Section 365(e)(1) of the Bankruptcy Code, each such lease is the legal, valid and binding obligation of such Loan Party or Subsidiary party thereto, enforceable against such Loan Party or Subsidiary in accordance with its terms, except as enforceability may be limited by applicable Debtor Relief Laws and by equitable principles regardless of whether considered in a proceeding in equity or at law.

(e) <u>Schedule 5.08(e)</u> sets forth a complete and accurate list of all Investments held by any Loan Party or any Subsidiary of a Loan Party on the Closing Date (other than Investments in Subsidiaries), showing as of the Closing Date the amount, obligor or issuer and maturity, if any, thereof.

5.09 Environmental Compliance.

Except as specifically disclosed in <u>Schedule 5.09</u> or for any matters, individually or in the aggregate, that could not reasonably be expected to have a Material Adverse Effect:

(a) the Loan Parties and their respective Subsidiaries conduct in the ordinary course of business a review of the effect of existing Environmental Laws and claims alleging potential liability or responsibility for violation of any Environmental Law on their respective businesses, operations and properties, and as a result thereof the Borrower has reasonably concluded that such Environmental Laws and claims could not reasonably be expected to result in an Environmental Liability;

(b) (i) none of the properties currently or, to the best knowledge of Holdings and the Borrower, formerly owned or operated by any Loan Party or any of its Subsidiaries is listed or formally proposed for listing on the NPL or any analogous foreign, state, provincial, territorial or local list; (ii) there are no and never have been any underground or above-ground storage tanks or any surface impoundments, septic tanks, pits, sumps or lagoons in which Hazardous Materials are being or have been treated, stored or disposed on any property currently owned or operated by any Loan Party or, to the best of the knowledge of the Loan Parties, on any property formerly owned or operated by any Loan Party or any of its Subsidiaries, in each case except in compliance with all applicable Environmental Laws; (iii) there is no asbestos or asbestos-containing material on any property currently owned or operated by any Loan Party or any of its Subsidiaries, in each case except in compliance with all applicable Environmental Laws; (iii) there is no asbestos or asbestos-containing material on any property currently owned or operated by any Loan Party or any of its Subsidiaries, in each case of Hazardous Materials on, at, under or from any property currently or formerly owned or operated by any Loan Party or any of its Subsidiaries that could reasonably be expected to result in an Environmental Liability; and

(c) neither any Loan Party nor any of its Subsidiaries is undertaking, and has not completed, either individually or together with other potentially responsible parties, any investigation or assessment or remedial or response action relating to any actual or threatened Release of Hazardous Materials at any site, location or operation, either voluntarily or pursuant to the order of any Governmental Authority or the requirements of any Environmental Law; and all Hazardous Materials generated, used, treated, handled or stored at, or transported to or from, any property currently or, to the knowledge of Holdings and the Borrower, formerly owned or operated by any Loan Party or any of its Subsidiaries have been disposed of in a manner not reasonably expected to result in an Environmental Liability.

5.10 <u>Insurance</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 365 of 432

The properties of the Borrower, Holdings and its other Subsidiaries are insured with financially sound and reputable insurance companies having an A.M. Best Financial Strength Rating of at least A which are not Affiliates of the Borrower, in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where the Borrower or the applicable Subsidiary operates.

5.11 <u>Taxes</u>.

The Borrower, Holdings and each of their respective Subsidiaries have filed all federal, state, territorial and other material tax returns and reports required to be filed, and have paid all federal, state, territorial and other material Taxes, assessments, fees and other governmental charges levied or imposed upon them or their properties, income or assets otherwise due and payable, except those which are being contested in good faith by appropriate proceedings diligently conducted and for which adequate reserves have been established in accordance with GAAP. To the best knowledge of Holdings and the Borrower, there is no proposed Tax assessment against the Borrower, Holdings or any other Subsidiary that would, if made, reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect. Neither any Loan Party nor any Subsidiary thereof is party to any tax sharing agreement.

5.12 ERISA Compliance.

(a) <u>Compliance Generally</u>. Each Benefit Plan is in compliance in all material respects with the applicable provisions of ERISA, the Code and other Federal or state Laws. Each Benefit Plan that is intended to qualify under Section 401(a) of the Code has received a favorable determination letter from the IRS (or is a prototype plan that is the subject of a favorable opinion letter from the IRS) or an application for such a letter is currently being processed by the IRS with respect thereto and, to the knowledge of the Borrower, nothing has occurred which would prevent, or cause the loss of, such qualification. Each Foreign Plan has been maintained in compliance in all material respects with its terms and applicable law. As of the Closing Date, neither Borrower, Holdings or any Subsidiary has any material liability for any Foreign Plan that is required to be registered under the Income Tax Act (Canada).

(b) <u>Pending Claims</u>. There are no pending or, to the best knowledge of the Borrower, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Benefit Plan that could reasonably be expected to have a Material Adverse Effect. There has been no prohibited transaction or violation of the fiduciary responsibility rules with respect to any Benefit Plan that has resulted or could reasonably be expected to result in a Material Adverse Effect.

(c) <u>No ERISA Event</u>. As of the Closing Date, no ERISA Event has occurred or is reasonably expected to occur that could reasonably be expected to have a Material Adverse Effect and there are no unfunded liabilities with respect to any Foreign Plans that could reasonably be expected to have a Material Adverse Effect.

5.13 <u>Use of Proceeds.</u>

The proceeds of the Loans are being and shall at all times be used by the Borrower (and, to the extent provided to them by the Borrower, each other Loan Party and their Subsidiaries) solely as follows: to the extent set forth in the Approved Budget and in accordance with the terms of the Orders, (a) for working capital and general corporate purposes of the Loan Parties and their Subsidiaries during the Chapter 11 Cases (excluding the repayment of any Indebtedness), (b) to enable the Loan Parties and their Subsidiaries to pay (i) interest, fees, costs and expenses related to the Facility, (ii) the fees, costs and expenses of professionals retained in the Chapter 11 Cases and approved by the Bankruptcy Court and (ii)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 366 of 432

the fees, costs and expenses of the Lenders (or any of their Affiliates or related funds), including the fees, costs and expenses of the Lender Advisors, (c) to enable the Loan Parties and their Subsidiaries to make the Adequate Protection Payments, (d) to enable the Loan Parties and their Subsidiaries to make all permitted payments of costs of administration of the Chapter 11 Cases, including prepetition amounts as authorized by the Bankruptcy Court, (e) reduction of the ABL DIP Facility and (f) payment of the Carve-Out; provided, however, that, in no event shall any cash or proceeds of the Loans be used by the Borrower or any other Loan Party or their Subsidiaries (x) in any manner inconsistent with any provision in this Agreement, any other Loan Document, the Approved Budget or the Orders or (y) in connection with investigating (including discovery proceedings), initiating, asserting, joining, commencing, supporting or prosecuting any claims, causes of action, adversary proceedings or other litigation against the Lenders, the Administrative Agent, the holders of the 6% Notes, the trustee under the 6% Notes Indenture and each of their respective Related Parties with respect to any transaction, occurrence, omission, action or other matter (including formal discovery proceedings in anticipation thereof), including, without limitation, (1) any claims or causes of action arising under chapter 5 of the Bankruptcy Code; (2) any so-called "lender liability" claims and causes of action; (3) any action with respect to the validity, enforceability, priority and extent of the 6% Notes or the Loans; (4) any action seeking to invalidate, set aside, avoid or subordinate, in whole or in part, the 6% Notes or the Loans; or (5) any action seeking to modify any of the rights, remedies, priorities, privileges, protections and benefits granted to any or all of the Secured Parties hereunder or under the Loan Documents, or for any purpose that is prohibited under the Bankruptcy Code, in each case, other than an aggregate amount not to exceed \$50,000 that may be used by, or to reimburse, the fees, costs or expenses of, the Committee in connection with the investigation of the prepetition claims of the holders of the 6% Notes or their Affiliates.

5.14 <u>Subsidiaries; Equity Interests; Loan Parties</u>.

As of the Closing Date, no Loan Party has any Subsidiaries other than those specifically disclosed in Part (a) of Schedule 5.14, and all of the outstanding Equity Interests in such Subsidiaries have been validly issued, are fully paid and non-assessable and are owned directly or indirectly by a Loan Party in the amounts specified in Part (a) of Schedule 5.14 free and clear of all Liens except those created under the Collateral Documents or the Orders. As of the Closing Date, no Loan Party has any equity investments in any other corporation or entity other than such Subsidiaries and those specifically disclosed in Part (b) of Schedule 5.14. All of the outstanding Equity Interests in the Borrower have been validly issued, are fully paid and non-assessable and are owned by Holdings free and clear of all Liens except those created under the Collateral Documents or the Orders. Set forth in Part (c) of Schedule 5.14 is a complete and accurate list of Holdings and all Loan Parties, showing as of the Closing Date, the jurisdiction of its incorporation, the address of its principal place of business and its U.S. taxpayer identification number or, in the case of any non-U.S. Loan Party that does not have a U.S. taxpayer identification number, its unique identification number issued to it by the jurisdiction of its incorporation. The copy of the charter of each Loan Party and each amendment thereto provided pursuant to Section 4.01(a) is a true and correct copy of each such document, each of which is valid and in full force and effect as of the Closing Date.

5.15 <u>Margin Regulations; Investment Company Act</u>.

(a) <u>Margin Stock</u>. The Borrower is not engaged and will not engage, principally or as one of its important activities, in the business of "buying" or "carrying" any "margin stock" (within the respective meanings of Regulation U issued by the FRB), or extending credit for the purpose of "buying" or "carrying" any "margin stock". No part of the proceeds of any Loans, and no other extensions of credit hereunder, will be used (a) for purposes of "buying" or "carrying" any "margin stock" or (b) for any purpose that violates the provisions of the Regulations of the FRB.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 367 of 432

(b) <u>Investment Company Act</u>. None of the Borrower, any Person Controlling the Borrower, or any Subsidiary is or is required to be, or after giving effect to the Transaction will be required to be, registered as an "investment company" under the Investment Company Act of 1940.

5.16 <u>Disclosure</u>.

The Borrower has disclosed to the Administrative Agent and the Lenders all agreements, instruments and corporate or other restrictions to which it or any of its Subsidiaries or any other Loan Party is subject, and all other matters known to it, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect. No report, financial statement, certificate or other information furnished (whether in writing or orally) by or on behalf of any Loan Party to the Administrative Agent or any Lender in connection with the transactions contemplated hereby and the negotiation of this Agreement or delivered hereunder or under any other Loan Document, including any confidential information memorandum (in each case as modified or supplemented by other information so furnished), taken as a whole, contains any material misstatement of fact or omits to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not materially misleading; provided that, with respect to projected financial information, each of Holdings and the Borrower represents only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.

5.17 <u>Compliance with Laws</u>.

Subject to the entry of the Orders, as applicable, each Loan Party and each Subsidiary thereof is in compliance in all material respects with the requirements of all Laws and all orders, writs, injunctions and decrees applicable to it or to its properties, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted or (b) the failure to comply therewith, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

5.18 Intellectual Property; Licenses, Etc.

(a) Except as could not reasonably be expected to have a Material Adverse Effect, each Loan Party and each of its Subsidiaries own, or possess adequate rights to use on a worldwide basis, all trademarks, service marks, trade names, trade dress, internet domain names, copyrights published and unpublished works of authorship (including software), patents, patent rights, inventions and discoveries (whether patentable or not), franchises, licenses and all registrations, recordations and applications of the foregoing know-how (including trade secrets and other unpatented and/or unpatentable proprietary or confidential information, systems or procedures) and other intellectual property rights (collectively, "Intellectual Property Rights") owned by such Loan Party ("IP Rights") or as such Intellectual Property is used in or is reasonably necessary for the operation of their respective businesses, free and clear of all Liens, encumbrances, or security interests in favor of any Person, that would conflict with the Liens in favor of the Administrative Agent or that are otherwise permitted under the Orders and without conflict with the rights of any other Person, and Schedule 5.18 sets forth a complete and accurate list of all such registered IP Rights owned by each Loan Party and each of its Subsidiaries as of the Closing Date. To the knowledge of Holdings or the Borrower, the use by the Loan Parties or their Subsidiaries of the IP Rights and the operation of their respective businesses does not infringe, misappropriate or otherwise violate the rights of any Person in any material respect, and no Loan Party has received any notice of any claim of infringement or misappropriation of the Intellectual Property rights of any Person in the past six (6) years. No judgment, holding, decision, claim or litigation regarding any of the IP Rights is pending or, to the best knowledge of the Borrower and Holdings, threatened, or rendered, by any Governmental Authority or Person which, either individually or in the aggregate, could limit or cancel any Loan Party's rights in,

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 368 of 432

any Intellectual Property in any respect that could reasonably be expected to have a Material Adverse Effect. To the best knowledge of the Borrower and Holdings no Person has infringed or misappropriated any IP Rights owned by a Loan Party in any material respect.

(b) Each Loan Party has taken such actions to protect and preserve the confidentiality of all trade secrets and other material confidential information included in the IP Rights as are commercially reasonable in accordance with industry standards.

- (c) [Reserved].
- (d) [Reserved].

(e) Each Loan Party has complied at all times in all material respects with applicable Laws regarding the collection, retention, use and protection of personal information except where the failure to comply therewith, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect. No Person (including any Governmental Body) has made any claim or commenced any action relating to the Loan Party's privacy or data security practices, including with respect to the access, disclosure or use of personal information maintained by or on behalf of any of such business or to the best knowledge of the Borrower and Holdings threatened any such claim or action or conducted any investigation or inquiry thereof. To the best knowledge of the Borrower and Holdings, the execution, delivery, or performance of this Agreement and the consummation of the transactions contemplated hereby, will not violate any applicable Laws or the privacy policy of any Loan Party as it currently exists or as it existed at any time during which any personal information was collected or obtained by or on behalf of such Loan Party.

5.19 <u>Initial Budget.</u> The Borrower has prepared and delivered to the Administrative Agent and the Lenders an initial budget (attached hereto as <u>Exhibit J</u>), which reflects the Loan Parties' anticipated cash receipts and anticipated disbursements for each calendar week during the period from the Petition Date through and including the end of the thirteenth (13th) calendar week following the Petition Date (the "<u>Initial Budget</u>"). The Initial Budget (and each subsequent Approved Budget hereunder) shall be deemed the "Approved Budget" for all purposes hereof until superseded by another Approved Budget pursuant to the provisions set forth herein.

5.20 Casualty, Etc.

Neither the businesses nor the properties of any Loan Party or any of its Subsidiaries are affected by any fire, explosion, accident, strike, lockout or other labor dispute, drought, storm, hail, earthquake, embargo, act of God or of the public enemy or other casualty (whether or not covered by insurance) that, either individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.21 Labor Matters.

Except as described on <u>Schedule 5.21</u> hereto, there are no collective bargaining agreements or Multiemployer Plans covering the employees of the Borrower or any of its Subsidiaries as of the Closing Date and to the knowledge of any Responsible Officer of the Borrower and Holdings, neither the Borrower nor any Subsidiary has suffered any strikes, walkouts, work stoppages or other material labor difficulty within the last five years.

5.22 <u>Collateral Documents</u>.

This Agreement and the Collateral Documents, taken together with the Interim Order and/or the Final Order, as applicable, are effective to create in favor of the Administrative Agent, for the benefit of the Secured Parties valid, binding, enforceable perfected Liens on, and security interests in, the Collateral pledged under the Loan Documents and/or the Orders (subject to Liens permitted by Section 7.01), in each case, having the priorities set forth in the Interim Order and/or the Final Order and subject only to the Carve-Out. Pursuant to the terms of the Interim Order and/or Final Order, no filing or other action will be necessary to perfect or protect such Liens and security interests. Pursuant to and to the extent provided in the Interim Order and the Final Order, as applicable, the Obligations of the Loan Parties under this Agreement will constitute allowed superpriority administrative expense claims in the Chapter 11 Cases under section 364(c) of the Bankruptcy Code, having priority over all administrative expense claims and unsecured claims against such Loan Parties now existing or hereafter arising, of any kind whatsoever, including, without limitation, all administrative expense claims of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code and all superpriority administrative expense claims granted to any other Person (including (for the avoidance of doubt, subject to the entry of the Final Order) avoidance actions and the proceeds thereof), subject only to the Carve-Out, which claims shall have recourse to all of the Loan Parties' assets (including avoidance actions and the proceeds thereof).

5.23 Anti-Terrorism Laws.

(a) No Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or agents of such Loan Party or such Subsidiary (i) has violated or is in violation of Anti-Terrorism Laws or (ii) has engaged or engages in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of offenses designated in the "Forty Recommendations" and "Nine Special Recommendations" published by the Organisation for Economic Co-operation and Development's Financial Action Task Force on Money Laundering.

(b) No Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or agents of such Loan Party or such Subsidiary that is acting or benefiting in any capacity in connection with the Loans is an Embargoed Person.

(c) Except as otherwise authorized by OFAC, no Loan Party, none of its Subsidiaries and, to the actual knowledge of the Responsible Officers of each Loan Party, none of the respective officers, directors, brokers or agents of such Loan Party or such Subsidiary acting or benefiting in any capacity in connection with the Loans (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Embargoed Person, (ii) deals in, or otherwise engages in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law or (iii) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.

5.24 Orders. Each of the Interim Order (to the extent necessary, with respect to the period prior to the entry of the Final Order) or the Final Order (from after the date the Final Order is entered) is in full force and effect and has not been vacated, reversed or rescinded or, without the prior written consent of the Administrative Agent and Required Lenders, in their sole discretion, amended or modified and no appeal of such order has been timely filed or, if timely filed, no stay pending such appeal is currently effective.

5.25 Bankruptcy Matters.

(a) The Chapter 11 Cases were commenced on the Petition Date, in accordance with applicable requirements of law and proper notice thereof under the circumstances, and proper notice of (x) the motion seeking approval of the Loan Documents and entry of the Orders, as applicable and (y) the hearings for the approval of the Interim Order have been held by the Bankruptcy Court.

(b) After the entry of the Orders, the Obligations will constitute DIP Superpriority Claims (as defined in the Orders) having the priority set forth in the Orders.

(c) After the entry of the Orders, as applicable, and pursuant to and to the extent provided in the Orders, as applicable, the Obligations will be secured by valid, binding, enforceable, non-avoidable, and automatically and fully and properly perfected Liens on, and security interests in, the Collateral, in each case, having the priorities set forth in the Interim Order and/or the Final Order and subject only to the Carve-Out.

5.26 <u>Material Contracts</u>. Subject to <u>Schedule 6.21</u>, <u>Schedule 1.02</u> sets forth all Material Contracts to which any Loan Party is a party or is bound as of the Closing Date. To the extent requested by the Administrative Agent, the Loan Parties have delivered true, correct and complete copies of such Material Contracts to the Administrative Agent.

ARTICLE VI

AFFIRMATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder and until all Loans and other Obligations are paid in full in cash in accordance with this Agreement, each of Holdings (except in the case of Section 6.11) and the Borrower shall, and shall (except in the case of the covenants set forth in Sections 6.01, 6.02, 6.03 and 6.11) cause each Subsidiary to:

6.01 <u>Financial Statements</u>.

Deliver to the Administrative Agent and each Lender, in form and detail reasonably satisfactory to the Administrative Agent and the Required Lenders (<u>provided</u> that the Required Lenders shall be deemed to find the following items satisfactory unless the Administrative Agent shall have received notice from the Required Lenders specifying their objections thereto within five (5) Business Days of the Lenders' receipt of such items):

(a) <u>Annual Financials</u>. As soon as available, but in any event on or prior to April 30, 2018, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal year, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year, setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail and prepared in accordance with GAAP, audited and

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 371 of 432

accompanied by a report and opinion of a Registered Public Accounting Firm of nationally recognized standing reasonably acceptable to the Required Lenders, which report and opinion shall be prepared in accordance with generally accepted auditing standards and applicable Securities Laws and shall not be subject to any exception or any qualification with respect to the absence of any material misstatement; and

(b) <u>Quarterly Financials</u>. As soon as available, but in any event within 45 days after the end of each of the first three fiscal quarters of each fiscal year of Holdings, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal quarter, and the related consolidated statements of income or operations and cash flows for such fiscal quarter and for the portion of Holdings' fiscal year then ended, setting forth in each case in comparative form the figures for the corresponding fiscal quarter of the previous fiscal year and the corresponding portion of the previous fiscal year, all in reasonable detail, such consolidated statements to be certified by the chief executive officer, chief financial officer, treasurer or controller of Holdings as fairly presenting the financial condition, results of operations and cash flows of Holdings and its Subsidiaries in accordance with GAAP, subject only to normal quarter-end or year-end adjustments and the absence of footnotes.

(c) <u>Monthly Financials</u>. As soon as available, but in any event within 30 days after the end of each fiscal month, a consolidated balance sheet of Holdings and its Subsidiaries as at the end of such fiscal month, and the related consolidated statements of income or operations and cash flows for such fiscal month and for the portion of Holdings' fiscal year then ended, setting forth in each case in comparative form the figures for the corresponding month of the previous fiscal year and the corresponding portion of the previous fiscal year, all in reasonable detail.

As to any information contained in materials furnished pursuant to <u>Section 6.02(d)</u>, neither Holdings nor the Borrower shall be separately required to furnish such information under <u>Section 6.01(a)</u> or (b) above, but the foregoing shall not be in derogation of the obligation of Holdings and the Borrower to furnish the information and materials described in Sections 6.01(a) and (b) above at the times specified therein.

6.02 <u>Certificates; Other Information</u>.

Deliver to the Administrative Agent and each Lender, in form and detail reasonably satisfactory to the Administrative Agent and the Required Lenders (<u>provided</u> that the Required Lenders shall be deemed to find the following items satisfactory unless the Administrative Agent shall have received notice from the Required Lenders specifying their objections thereto within five Business Days of the Lenders' receipt of such items):

(a) [Intentionally Omitted];

(b) <u>Compliance Certificate</u>. Concurrently with the delivery of the financial statements referred to in Sections 6.01(a), (b) and (c), a duly completed Compliance Certificate signed by the chief executive officer, chief financial officer, treasurer or controller of the Borrower;

(c) [Reserved];

(d) <u>SEC Filings</u>. Promptly after the same are available, copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of the Borrower, and copies of all annual, regular, periodic and special reports and registration statements which the Borrower may file or be required to file with the SEC under Section 13 or

15(d) of the Securities Exchange Act of 1934, or with any national securities exchange, and in any case not otherwise required to be delivered to the Administrative Agent pursuant hereto;

(e) <u>Other Reports</u>. Promptly after the furnishing thereof, copies of any statement or report furnished to any holder of debt securities of any Loan Party or of any of its Subsidiaries pursuant to the terms of any indenture, loan or credit or similar agreement and not otherwise required to be furnished to the Lenders pursuant to <u>Section 6.01</u> or any other clause of this <u>Section 6.02</u>;

(f) <u>Insurance Coverage</u>. As soon as available, but in any event within 30 days after the end of each fiscal year of the Borrower, a report summarizing the insurance coverage (specifying type, amount and carrier) in effect for each Loan Party and its Subsidiaries and containing such additional information as the Administrative Agent, or any Lender through the Administrative Agent, may reasonably specify;

(g) <u>Investigations</u>. Promptly, and in any event within ten Business Days after receipt thereof by any Loan Party or any Subsidiary thereof, copies of each notice or other correspondence received from the SEC (or comparable agency in any applicable non-U.S. jurisdiction) concerning any investigation or possible investigation or other inquiry by such agency regarding financial or other operational results of any Loan Party or any Subsidiary thereof;

(h) <u>Related Document Notices</u>. Not later than five Business Days after receipt thereof by any Loan Party or any Subsidiary thereof, copies of all notices, requests and other documents (including amendments, waivers and other modifications) so received under or pursuant to any Related Document regarding or related to any breach or default by any party thereto or any other event that could materially impair the value of the interests or the rights of any Loan Party or otherwise have a Material Adverse Effect and, from time to time upon request by the Administrative Agent, such information and reports regarding the Related Documents as the Administrative Agent may reasonably request;

(i) <u>Environmental Notices</u>. Promptly after the assertion or occurrence thereof, notice of any action or proceeding alleging an Environmental Liability against any Loan Party or any of its Subsidiaries that could (i) reasonably be expected to have a Material Adverse Effect or (ii) cause any property owned by the Loan Parties with a fair market value in excess of \$3,000,000 to be subject to any restrictions on ownership, occupancy, use or transferability under any Environmental Law;

(j) <u>Schedule Supplements</u>. As soon as available, but in any event within 90 days after the end of each fiscal year of the Borrower, (i) a report supplementing Schedules 5.08(c), 5.08(d)(i) and 5.08(d)(ii), including an identification of all owned and leased real property disposed of by any Loan Party or any Subsidiary thereof during such fiscal year, a list and description (including the street address, county or other relevant jurisdiction, state, record owner, book value thereof and, in the case of leases of property, lessor, lessee, expiration date and annual rental cost thereof) of all real property acquired or leased during such fiscal year and a description of such other changes in the information included in such schedules as may be necessary for such schedules to be accurate and complete and (ii) a report supplementing <u>Schedules 5.08(e)</u> and <u>5.14</u> and each schedule to the Security Agreement containing a description of all changes in the information included in such schedules to be accurate and complete as may be necessary for such schedules to be accurate and schedules as may be necessary for such schedules to be accurate and complete as may be necessary for such schedules to be accurate and complete as may be necessary for such schedules to be accurate and complete as may be necessary for such schedules to be accurate and complete as may be necessary for such schedules to be accurate and complete as may be necessary for such schedules to be accurate and complete, each such report to be signed by a Responsible Officer of the Borrower and to be in a form reasonably satisfactory to the Administrative Agent; and

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 373 of 432

(k) <u>Additional Information</u>. Promptly, such additional information regarding the business, financial, legal or corporate affairs of any Loan Party or any Subsidiary thereof, or compliance with the terms of the Loan Documents, as the Administrative Agent or any Lender may from time to time reasonably request.

Documents required to be delivered pursuant to Section 6.01(a), (b) or (c) or Section 6.02(d) (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date (i) on which the Borrower posts such documents, or provides a link thereto on the Borrower's website on the Internet at the website address listed on Schedule 11.02; or (ii) on which such documents are posted on the Borrower's behalf on an Internet or intranet website, if any, to which each Lender and the Administrative Agent have access (whether a commercial, third-party website or whether sponsored by the Administrative Agent); provided that (i) the Borrower shall deliver paper copies of such documents to the Administrative Agent or any Lender that requests the Borrower to deliver such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender and (ii) the Borrower shall notify the Administrative Agent and each Lender (by facsimile or electronic mail) of the posting of any such documents and provide to the Administrative Agent by electronic mail electronic versions (i.e., soft copies) of such documents. Except for such Compliance Certificates, the Administrative Agent shall have no obligation to request the delivery or to maintain copies of the documents referred to above, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery, and each Lender shall be solely responsible for requesting delivery to it or maintaining its copies of such documents.

The Borrower hereby acknowledges that (a) the Administrative Agent will make available to the Lenders materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform") and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or its securities) (each, a "Public Lender"). The Borrower hereby agrees that it will use commercially reasonable efforts to identify that portion of the Borrower Materials that may be distributed to the Public Lenders and that (w) all such Borrower Materials shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, means that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Administrative Agent and the Lenders to treat such Borrower Materials as not containing any material non-public information (although it may be sensitive and proprietary) with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Information, they shall be treated as set forth in Section 11.07); (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Investor"; and (z) the Administrative Agent shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding the foregoing, the Borrower shall be under no obligation to mark any Borrower Materials "PUBLIC."

6.03 <u>Notices</u>.

Promptly notify the Administrative Agent and each Lender:

(a) of the occurrence of any Default or Event of Default, or any default or event of default or breach of any provision in the Restructuring Support Agreement or the Orders;

(b) of any matter that has resulted or could reasonably be expected to result in a Material Adverse Effect, including (i) breach or non-performance of, or any default under, a Contractual Obligation of any Loan Party or any Subsidiary thereof; (ii) any dispute, litigation, investigation, proceeding or suspension between any Loan Party or any Subsidiary thereof and any Governmental Authority; or (iii) the commencement of, or any material development in, any litigation or proceeding affecting any Loan Party or any Subsidiary thereof, including pursuant to any Environmental Laws, but only if any of the matters described in the foregoing clauses (i) through (iii) has resulted or reasonably could be expected to result in a Material Adverse Effect;

(c) after a Responsible Officer of the Borrower or Holdings knows or has reason to know of the occurrence of any ERISA Event that could reasonably be expected to result in liability of one or more Loan Parties or any Subsidiary thereof in excess of \$2,500,000;

(d) of any material change in accounting policies or financial reporting practices by any Loan Party or any Subsidiary thereof;

(e) of the Borrower's determination at any time of the occurrence or existence of any Internal Control Event;

(f) of any announcement by Moody's or S&P of any change in a Debt Rating; and

(g) after receipt thereof by Borrower or any ERISA Affiliate, of (i) any notice from the PBGC stating its intention to terminate any Benefit Plan or to have a trustee appointed to administer any Benefit Plan or (ii) any notice concerning (A) the imposition of withdrawal liability by any Multiemployer Plan, (B) the insolvency, reorganization or termination, within the meaning of Title IV of ERISA, of any such Multiemployer Plan or (C) the amount of liability incurred, or that may be incurred, by such Loan Party or any Subsidiary thereof or any ERISA Affiliate in connection with any event described in clause (A) or (B). and (iii) after the filing or receipt thereof, of any notice, letter or communication which any Borrower or any ERISA Affiliate receives from, or files with, the PBGC with respect to any Benefit Plan or any Multiemployer Plan, if such notice, letter or communication could result in any material liability to Borrower or any ERISA Affiliate.

Each notice pursuant to <u>Section 6.03</u> (other than <u>Section 6.03(f)</u> and <u>Section 6.03(e)</u>) shall be accompanied by a statement of a Responsible Officer of the Borrower setting forth details of the occurrence referred to therein and stating what action the Borrower has taken and proposes to take with respect thereto. Each notice pursuant to <u>Section 6.03(a)</u> shall describe with particularity any and all provisions of this Agreement and any other Loan Document that have been breached.

6.04 <u>Payment of Obligations</u>.

Pay and discharge as the same shall become due and payable, all its obligations and liabilities, including (a) all Tax liabilities, assessments and governmental charges or levies upon it or its properties or assets, unless the same are being contested in good faith by appropriate proceedings diligently conducted and adequate reserves in accordance with GAAP are being maintained by Holdings, the Borrower or such Subsidiary, (b) all lawful claims which, if unpaid, would by law become a Lien upon its property, unless the same are being contested in good faith by appropriate proceedings diligently conducted, adequate reserves in accordance with GAAP are being maintained by Holdings, the Borrower or such Subsidiary and such contest could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect and (c) as may be required pursuant to the Restructuring Support Agreement; provided that, obligations of the Loan Parties hereunder are subject to the provisions of the

Bankruptcy Code and any required approvals by an applicable order of the Bankruptcy Court; and further provided that the Loan Parties shall be obligated only to pay all of the foregoing to the extent constituting post-petition obligations that constitute administrative expenses under Section 503(b) of the Bankruptcy Code in the Cases and to perform all material obligations arising from Contractual Obligations entered into after the Petition Date or from Contractual Obligations entered into prior to the Petition Date and assumed.

6.05 **Preservation of Existence, Etc.**

(a) Preserve, renew and maintain in full force and effect its legal existence and good standing under the Laws of the jurisdiction of its organization except in a transaction permitted by Section 7.04 or 7.05;

(b) take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent that failure to do so could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and

(c) preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

6.06 <u>Maintenance of Properties</u>.

(a) Maintain, preserve and protect all of its material properties and equipment necessary in the operation of its business in good working order and condition, ordinary wear and tear excepted;

(b) make all necessary repairs thereto and renewals and replacements thereof except where the failure to do so could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect; and

(c) use the standard of care typical in the industry in the operation and maintenance of its facilities.

6.07 <u>Maintenance of Insurance</u>.

(a) Maintain with financially sound and reputable insurance companies having an A.M. Best Financial Strength Rating of at least A which are not Affiliates of the Borrower, insurance with respect to its properties and business against interruption, loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts as are customarily carried under similar circumstances by such other Persons and providing for such insurer to endeavor to give not less than 30 days' prior notice to the Administrative Agent of termination, lapse or cancellation of such insurance. The Loan Parties shall cause the Administrative Agent to be named on each policy evidencing such insurance as secured party, mortgagee, loss payee or additional insured, as applicable, in a manner acceptable to the Administrative Agent.

(b) If any portion of any Mortgaged Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), (x) maintain, or cause to be maintained, with a financially sound and reputable insurer, flood insurance in an amount and otherwise sufficient to

comply with all applicable rules and regulations promulgated pursuant to the Flood Insurance Laws and (y) deliver to the Administrative Agent evidence of such compliance in form and substance reasonably acceptable to the Administrative Agent.

6.08 <u>Compliance with Laws</u>.

Comply in all material respects with the requirements of all Laws and all orders, writs, injunctions and decrees applicable to it or to its business or property, except in such instances in which (a) such requirement of Law or order, writ, injunction or decree is being contested in good faith by appropriate proceedings diligently conducted or (b) the failure to comply therewith could not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect.

6.09 <u>Books and Records</u>.

Maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of Holdings, the Borrower or such Subsidiary, as the case may be.

6.10 <u>Inspection Rights</u>.

(a) Permit the Administrative Agent and any Lender, subject to reasonable advance notice to, and reasonable coordination with, the Borrower and during normal business hours, to visit and inspect the properties of the Loan Parties, at the Borrower's expense as provided in clause (b) below, inspect, audit and make extracts from the Loan Parties' corporate, financial or operating records, and discuss with its officers, employees, agents, advisors and independent accountants (subject to such accountants' customary policies and procedures) the Loan Parties' business, financial condition, assets and results of operations (it being understood that a representative of the Borrower is allowed to be present in any discussions with officers, employees, agent, advisors and independent accountants). No such inspection or visit shall unduly interfere with the business or operations of the Loan Parties, nor result in any damage to the property or other Collateral. Neither the Administrative Agent nor any Lender shall have any duty to the Borrower. The Borrower acknowledges that all inspections, appraisals and reports are prepared by the Administrative Agent and Lenders for their purposes, and the Borrower shall not be entitled to rely upon them.

(b) Reimburse the Administrative Agent for all reasonable and documented out-of-pocket costs and expenses of the Administrative Agent in connection with (i) two (2) examinations per fiscal year of the Loan Parties' books and records or any other financial or Collateral matters as the Administrative Agent deems appropriate and (ii) appraisals of Collateral. Subject to and without limiting the foregoing, the Borrower specifically agrees to pay the Administrative Agent's then standard charges for examination activities, including the standard charges of the Administrative Agent's right to use third parties for such purposes.

6.11 <u>Use of Proceeds</u>.

At all times use the proceeds of the Loans solely for the purposes described in <u>Section 5.13</u> and otherwise in accordance with the terms of this Agreement and the Orders.

6.12 <u>Covenant to Guarantee Obligations and Give Security.</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 377 of 432

(a) Upon the formation or acquisition of any new direct or indirect Domestic Subsidiary (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) by any Loan Party, then the Borrower shall, at the Borrower's expense:

(i) within 30 days after such formation or acquisition, cause such Domestic Subsidiary, and cause each direct and indirect parent of such Domestic Subsidiary (if it has not already done so), to duly execute and deliver to the Administrative Agent a Joinder Agreement, pursuant to which such other Loan Party shall guaranty the other Loan Parties' obligations under the Loan Documents and pledge a security interest in and to all of its assets in support of such guaranty in accordance with the terms and conditions of the Security Agreement (for each Domestic Subsidiary);

(ii) within 15 days after such formation or acquisition, furnish to the Administrative Agent a description of the real and personal properties of such Domestic Subsidiary, in detail satisfactory to the Administrative Agent;

(iii) within 30 days after such formation or acquisition, cause such Domestic Subsidiary, and each direct and indirect parent of such Domestic Subsidiary (if it has not already done so) to duly execute and deliver to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto), as specified by and in form and substance reasonably satisfactory to the Administrative Agent (including delivery of all Pledged Equity in and of such Domestic Subsidiary, and other instruments of the type specified in Section 4.01(a)(iii), securing payment of all the Obligations of such Domestic Subsidiary, or such parent, as the case may be, under the Loan Documents and constituting Liens on all such real and personal properties, and with respect to each Mortgaged Property encumbered by a Mortgage, pursuant to this Section 6.12(a)(iii), deliver the items set forth in Sections 4.01(a)(iv) and 6.21 hereof;

(iv) within 30 days after such formation or acquisition, cause such Domestic Subsidiary, and each direct and indirect parent of such Domestic Subsidiary (if it has not already done so) to take whatever action (including the recording of mortgages, the filing of Uniform Commercial Code financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on the properties purported to be subject to the deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), hypothecs, Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto) delivered pursuant to this <u>Section 6.12</u>, enforceable against all third parties in accordance with their terms;

(v) within 60 days after such formation or acquisition, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the Administrative Agent's discretion, may be given by in-house counsel) as to the matters contained in clauses (i), (iii) and (iv) above, and as to such other matters as the Administrative Agent may reasonably request, and

(vi) as promptly as practicable after such formation or acquisition, deliver, upon the request of the Administrative Agent in its sole discretion, to the Administrative Agent with

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 378 of 432

respect to each parcel of real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of \$3,000,000 or more that is owned or held by the entity that is the subject of such formation or acquisition title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent, <u>provided</u>, <u>however</u>, that to the extent that any Loan Party or any of its Subsidiaries shall have otherwise received any of the foregoing items with respect to such real property, such items shall, promptly after the receipt thereof, be delivered to the Administrative Agent.

(b) Upon the acquisition of any property (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) by any Loan Party, if such property, in the judgment of the Administrative Agent, shall not already be subject to a perfected first priority security interest and Lien in favor of the Administrative Agent for the benefit of the Secured Parties, then the Borrower shall, at the Borrower's expense:

(i) within 15 days after such acquisition, furnish to the Administrative Agent a description of the property so acquired in detail satisfactory to the Administrative Agent,

(ii) within 30 days after such acquisition, cause the applicable Loan Party to duly execute and deliver to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto), as specified by and in form and substance reasonably satisfactory to the Administrative Agent, securing payment of all the Obligations of the applicable Loan Party under the Loan Documents and constituting Liens on all such properties, and with respect to each Mortgaged Property encumbered by a Mortgage pursuant to this Section 6.12(b)(ii), deliver the items set forth in Sections 4.01(a)(iv) and 6.21 hereof,

(iii) within 30 days after such acquisition, cause the applicable Loan Party to take whatever action (including the recording of mortgages, the filing of Uniform Commercial Code financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on such property, enforceable against all third parties,

(iv) within 60 days after such acquisition, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the Administrative Agent's discretion, may be given by in-house counsel) as to the matters contained in clauses (ii) and (iii) above and as to such other matters as the Administrative Agent may reasonably request, and

(v) as promptly as practicable after any acquisition of any real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of \$3,000,000 or more, deliver, upon the request of the Administrative Agent in its sole discretion, to the Administrative Agent with respect to such real property title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent, <u>provided</u>, <u>however</u>, that to the extent that any Loan Party or any of its Subsidiaries shall have otherwise received any of the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 379 of 432

foregoing items with respect to such real property, such items shall, promptly after the receipt thereof, be delivered to the Administrative Agent.

(c) Upon the request of the Administrative Agent following the occurrence and during the continuance of a Default, the Borrower shall, at the Borrower's expense:

(i) within 10 days after such request, furnish to the Administrative Agent a description of the real and personal properties of the Loan Parties and their respective Subsidiaries in detail reasonably satisfactory to the Administrative Agent,

(ii) within 15 days after such request, duly execute and deliver, and cause each Domestic Subsidiary (other than any Subsidiary that is held directly or indirectly by a CFC) of the Borrower (if it has not already done so) to duly execute and deliver, to the Administrative Agent deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto), as specified by and in form and substance reasonably satisfactory to the Administrative Agent (including delivery of all Pledged Equity and Pledged Debt in and of such Subsidiary, and other instruments of the type specified in <u>Section 4.01(a)(iii)</u>), securing payment of all the Obligations of such Domestic Subsidiary under the Loan Documents and constituting Liens on all such properties,

(iii) within 30 days after such request, take, and cause each Domestic Subsidiary (other than any Subsidiary that is held directly or indirectly by a CFC) of the Borrower to take, whatever action (including the recording of mortgages, the filing of Uniform Commercial Code financing statements, the giving of notices and the endorsement of notices on title documents) as may be necessary or advisable in the opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on the properties purported to be subject to the deeds of trust, trust deeds, deeds to secure debt, mortgages (other than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto) delivered pursuant to this <u>Section 6.12</u>, enforceable against all third parties in accordance with their terms,

(iv) within 60 days after such request, deliver to the Administrative Agent, upon the request of the Administrative Agent in its sole discretion, a signed copy of a favorable opinion, addressed to the Administrative Agent and the other Secured Parties, of counsel for the Loan Parties reasonably acceptable to the Administrative Agent (certain of which opinions, in the discretion of the Administrative Agent, may be given by in-house counsel) as to the matters contained in clauses (ii) and (iii) above, and as to such other matters as the Administrative Agent may reasonably request, and

(v) as promptly as practicable after such request, deliver to the Administrative Agent with respect to each parcel of real property owned or held by the Borrower and its Subsidiaries, title reports, surveys and engineering, soils and other reports, and environmental assessment reports, each in scope, form and substance reasonably satisfactory to the Administrative Agent.

(d) At any time upon request of the Administrative Agent, promptly execute and deliver any and all further instruments and documents and take all such other action as the Administrative Agent may deem necessary or desirable in obtaining the full benefits of, or (as applicable) in perfecting and preserving the Liens of, such guaranties, deeds of trust, trust deeds, deeds to secure debt, mortgages (other

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 380 of 432

than leasehold mortgages and leasehold deeds of trust), Security Agreement Supplements and other security and pledge agreements (including amendments and supplements thereto).

Notwithstanding anything in this <u>Section 6.12</u> to the contrary no Loan Party shall be required to deliver a mortgage on any real property or series of related real properties with a fair market value (as reasonably determined by the Borrower) of less than \$3,000,000. Notwithstanding anything in any of the Loan Documents to the contrary, nothing in this Agreement or any other Loan Document shall require any Borrower or Loan Party to make any filings or take any actions to record or to perfect the Administrative Agent's security interest in any IP Rights other than filing of documents effecting the recordation of security interests in the United States Copyright Office or United States Patent and Trademark Office.

6.13 Compliance with Environmental Laws.

Comply, and use commercially reasonable efforts to cause all lessees and other Persons operating or occupying its properties to comply, in all material respects, with all Environmental Laws and Environmental Permits; obtain and renew all Environmental Permits necessary for its operations and properties; and conduct any investigation, study, sampling and testing, and undertake any cleanup, removal, remedial or other action necessary to remove and clean up all Hazardous Materials from any of its properties, in accordance with the requirements of all Environmental Laws; <u>provided</u>, <u>however</u>, that neither Holdings nor any of its Subsidiaries shall be required to undertake any such cleanup, removal, remedial or other action to the extent that its obligation to do so is being contested in good faith and by proper proceedings and appropriate reserves are being maintained with respect to such circumstances in accordance with GAAP.

6.14 <u>Preparation of Environmental Reports.</u>

At the request of the Required Lenders from time to time upon the occurrence of any Release of Hazardous Materials or other event governed by Environmental Law that could reasonably be expected to materially impair the interests of the Secured Parties in any property of the Loan Parties, provide to the Lenders within 90 days after such request, at the expense of the Borrower, an environmental site assessment report for any of such properties, prepared by an environmental consulting firm reasonably acceptable to the Administrative Agent indicating the presence or absence of Hazardous Materials and the estimated cost of any compliance, removal or remedial action in connection with any Hazardous Materials on such properties; without limiting the generality of the foregoing, if the Administrative Agent determines at any time that a material risk exists that any such report will not be provided within the time referred to above, the Administrative Agent may retain an environmental consulting firm to prepare such report at the expense of the Borrower, and the Borrower hereby grants and agrees to cause any Subsidiary that owns any property described in such request to grant at the time of such request to the Administrative Agent, the Lenders, such firm and any agents or representatives thereof an irrevocable non-exclusive license, subject to the rights of tenants, to enter onto their respective properties to undertake such an assessment.

6.15 <u>Further Assurances</u>.

Promptly upon request by the Administrative Agent, or any Lender through the Administrative Agent, (a) correct any material defect or error that may be discovered in any Loan Document or in the execution, acknowledgment, filing or recordation thereof, (b) do, execute, acknowledge, deliver, record, re-record, file, re-file, register and re-register any and all such further acts, deeds, certificates, assurances and other instruments as the Administrative Agent, or any Lender through the Administrative Agent, may reasonably require from time to time in order to (i) carry out more effectively the purposes of the Loan

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 381 of 432

Documents, (ii) to the fullest extent permitted by applicable Law, subject any Loan Party's or any of its Domestic Subsidiaries' (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) properties, assets, rights or interests to the Liens now or hereafter intended to be covered by any of the Collateral Documents, (iii) perfect and maintain the validity, effectiveness and priority of any of the Collateral Documents and any of the Liens intended to be created thereunder and (iv) assure, convey, grant, assign, transfer, preserve, protect and confirm more effectively unto the Secured Parties the rights granted or now or hereafter intended to be granted to the Secured Parties under any Loan Document or under any other instrument executed in connection with any Loan Document to which any Loan Party or any of its Subsidiaries is or is to be a party, and cause each of its Domestic Subsidiaries (other than any CFC and any Subsidiary that is held directly or indirectly by a CFC) to do so, and (c) if an Event of Default has occurred and is continuing, deliver to the Administrative Agent such surveys, appraisals, environmental assessments, title insurance updates or policies and other related documents with respect to each parcel of real estate subject to a Mortgage.

6.16 <u>Compliance with Terms of Leaseholds</u>.

Make all payments and otherwise perform all obligations in respect of all leases of real property to which the Borrower or any of its Subsidiaries is a party, keep such leases in full force and effect and not allow such leases to lapse or be terminated or any rights to renew such leases to be forfeited or cancelled, notify the Administrative Agent of any default by any party with respect to such leases and cooperate with the Administrative Agent in all respects to cure any such default, and cause each of its Subsidiaries to do so, except, in any case, where the failure to do so, either individually or in the aggregate, could not be reasonably likely to have a Material Adverse Effect.

6.17 Additional Chapter 11 Reporting.

(a) Updated Budget. Not later than 5:00 p.m. New York City time on February 23, 2018 and every second Friday thereafter (or as more frequently requested by the Administrative Agent) (each, a "Budget Delivery Date"), the Loan Parties shall deliver (email shall suffice) to the Administrative Agent and the Lenders a supplement to the Approved Budget covering the subsequent 13-week period (and not re-forecasting periods in prior Approved Budgets), consistent with the form and level of detail set forth in the Initial Budget and otherwise in form and substance satisfactory to the Administrative Agent (at the direction of the Required Lenders) in all material respects (each such supplement, solely to the extent approved by the Administrative Agent, an "Updated Budget") reflecting the Loan Parties' anticipated cash receipts and anticipated disbursements for each calendar week during the thirteen (13) calendar week period beginning on the Saturday following the applicable Budget Delivery Date, together with a report setting forth the amount by which any line item has changed since the prior Approved Budget. Upon (and subject to) the approval of any such Updated Budget by the Administrative Agent (at the direction of the Required Lenders), such Updated Budget shall constitute the then-Approved Budget for purposes hereof; provided, however, that in the event the Administrative Agent provides written notice to the Loan Parties (including, by email) that the proposed Updated Budget is not acceptable, and the Administrative Agent (at the direction of the Required Lenders) and the Loan Parties are unable to reach agreement regarding such Updated Budget, then the Approved Budget most recently in effect shall remain the Approved Budget.

(b) <u>Variance Reporting</u>. Beginning on February 16, 2018 and every Friday thereafter (each, a "<u>Variance Reporting Delivery Date</u>"), the Loan Parties shall deliver to the Administrative Agent and the Lenders (email shall suffice) a report, in form and substance satisfactory to the Administrative Agent (at the direction of the Required Lenders) in all material respects (each a "<u>Variance Report</u>"), setting forth (i) actual results for the following items identified in the Approved Budget: (A) Total Cash Receipts, (B) Total Operating Disbursements, (C) "total financing cash receipts/disbursements", (D) "total QP

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 382 of 432

disbursements / adjustments", (E) "total other disbursements / adjustments", and (F) "liquidity" of the Loan Parties for the previous calendar week ended on the immediately preceding Friday, and setting forth all the variances, on a line item and aggregate basis, as compared to the corresponding amounts set forth in the then-current Approved Budget for such week/period, in each case, on a cumulative basis from the beginning of the period covered by the then-current Approved Budget, together with an explanation of (1) in reasonable detail, all material variances from the then current Approved Budget for such period and (2) in the case of a Variance Report immediately following the end of any Testing Period other than a Permitted Variance, the Loan Parties' compliance with the Permitted Variances for such Testing Period, and (ii) the variance, if any, in the amount contained in the Approved Budget with respect to projected Cumulative Net Operating Cash Flow compared to the actual Cumulative Net Operating Cash Flow as of the end of the last day covered by the Variance Report. Variances other than any Permitted Variance, if any, from the applicable Approved Budget, and any proposed changes to the applicable Approved Budget, shall be subject to the prior written consent of the Administrative Agent and the Required Lenders.

(c) [Reserved].

Lender Conference Calls. Arrange for (x) a teleconference (the "Management (d) Conference Call") to take place at least once per calendar month (at such time as is reasonably satisfactory to the Administrative Agent), which Management Conference Call shall (i) require participation by at least one senior member of the Borrower's management team and permit participation by such professional advisors to the Borrower as are required by the Borrower and such Lenders and Lender Advisors as elect to participate therein (provided, that, each Lender and Lender Advisor shall be provided with an invitation to and details of such Management Conference Call at least two (2) Business Days prior to the scheduled date therefor) and (ii) be intended for purposes of discussing the Loan Parties' financials (including any budget proposed to be the Approved Budget, Variance Reports and any projections) and such other information and matters reasonably related thereto or reasonably requested by any Lender and (y) a teleconference with professional advisors to the Borrower (the "Advisors Conference Call") to take place at least once per calendar week (at such time as is reasonably satisfactory to the Administrative Agent), which Advisors Conference Call shall be intended for purposes of discussing any budget proposed to be the Approved Budget, Variance Reports and any projections and such other information and matters reasonably related thereto or reasonably requested by any Lender.

6.18 <u>Material Contracts</u>.

Perform and observe all the terms and provisions of each Material Contract to be performed or observed by it, maintain each such Material Contract in full force and effect, enforce each such Material Contract in accordance with its terms, take all such action to such end as may be from time to time requested by the Administrative Agent and, upon request of the Administrative Agent, make to each other party to each such Material Contract such demands and requests for information and reports or for action as any Loan Party or any of its Subsidiaries is entitled to make under such Material Contract, and cause each of its Subsidiaries to do so, except, in any case, where the failure to do so, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

6.19 <u>Restructuring Obligations</u>.

(a) Maintain at all times the Restructuring Support Agreement in full force and effect with respect to all parties party thereto (including the Loan Parties).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 383 of 432

(b) Keep the Cash Management Order in effect at all times on and after the Closing Date. Maintain a cash management system subject to the Cash Management Order and at all times satisfactory to the Required Lenders.

(c) Comply with all Milestones.

6.20 Collateral Monitoring and Reporting.

(a) Each Loan Party shall keep accurate and complete records in respect of Collateral to the extent and in the manner required by the ABL DIP Facility and shall, promptly upon reasonable request from the Administrative Agent or the Required Lenders, make copies of all documents and other information relating thereto available to the Agent and the Lenders.

- (b) [Reserved].
- (c) [Reserved].
- (d) [Reserved].

(e) <u>Administration of Deposit Accounts</u>. <u>Schedule 6.20</u> sets forth all Deposit Accounts (other than Excluded Deposit Accounts (as defined in the ABL DIP Credit Agreement)) maintained by the Loan Parties, including all Dominion Accounts (as defined in the ABL DIP Credit Agreement), as of the Closing Date. The Borrower shall promptly notify the Administrative Agent of any opening or closing of a Deposit Account by any Loan Party (other than any Excluded Deposit Accounts).

- (f) [Reserved].
- (g) [Reserved].

6.21 <u>Post-Closing Actions</u>.² Take the actions and provide the deliverables, in the manner and form and in substance satisfactory to the Administrative Agent (at the direction of the Required Lenders), described on Schedule 6.21 within the time period after the Closing Date specified on such Schedule 6.21 or such longer period as may be approved by the Required Lenders.

ARTICLE VII

NEGATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder or any Loan or other Obligation hereunder shall remain unpaid or unsatisfied in full in cash, each of Holdings and the Borrower shall not, nor shall it permit any Subsidiary to, directly or indirectly:

7.01 <u>Liens</u>.

Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, or sign or file or suffer to exist under the Uniform Commercial Code of any jurisdiction a financing statement that names the Borrower, Holdings or any of its other Subsidiaries as debtor, or assign any accounts or other right to receive income, other than the following (including any financing statements filed in connection with any of the following):

² NTD: Post-closing actions to be determined.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 384 of 432

(a) Liens pursuant to any Loan Document or the Orders, including Liens securing obligations under the ABL DIP Facility;

(b) Liens existing on the Closing Date, existing immediately prior to the commencement of the Chapter 11 Cases, and, in each case, listed on <u>Schedule 5.08(b)</u>, excluding any renewals or extensions thereof or other modifications thereto;

(c) Liens for Taxes not yet due or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;

(d) carriers', warehousemen's, mechanics', materialmen's, repairmen's, construction or other like Liens arising in the ordinary course of business which are not overdue for a period of more than 30 days or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;

(e) pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA;

(f) deposits to secure the performance of bids, trade contracts and leases (other than Indebtedness), statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business;

(g) (i) easements, (ii) rights-of-way, (iii) servitudes, (iv) restrictions, (v) reservations, limitations, provisos and conditions expressed in an original grant from the Crown and (vi) other similar encumbrances affecting real property which, in the aggregate, are not substantial in amount, and which do not in any case materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of the applicable Person;

(h) Liens securing Indebtedness permitted under <u>Section 7.02(f)</u>; provided that (i) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and (ii) the Indebtedness secured thereby does not exceed the cost or fair market value, whichever is lower, of the property being acquired on the date of acquisition;

(i) Licenses and sublicenses of IP Rights granted to third Persons in the ordinary course of business;

- (j) [reserved];
- (k) [reserved];
- (l) [reserved];
- (m) [reserved]; and

(n) Liens in favor of a depositary institution on cash deposits and other funds maintained with such depository institution in each case arising in the ordinary course of business by virtue of any statutory or common law provision relating to banker's liens, including Section 4-208 of the UCC.

7.02 Indebtedness.

Create, incur, assume or suffer to exist any Indebtedness, except:

(a) obligations (contingent or otherwise) of the Borrower existing or arising under any Swap Contract; <u>provided</u> that (A) such obligations are (or were) entered into by such Person in the ordinary course of business for the purpose of directly mitigating risks associated with liabilities, commitments, investments, assets, or property held or reasonably anticipated by such Person, or changes in the value of securities issued by such Person, and not for purposes of speculation or taking a "market view," and (B) such Swap Contract does not contain any provision exonerating the non-defaulting party from its obligation to make payments on outstanding transactions to the defaulting party;

(b) unsecured Indebtedness of a Subsidiary of the Borrower owed to the Borrower or a wholly-owned Subsidiary of the Borrower, which Indebtedness shall (i) in the case of Indebtedness owed to a Loan Party, constitute "Pledged Debt" under the Security Agreement, (ii) be on terms (including subordination terms) acceptable to the Administrative Agent and (iii) be otherwise permitted under the provisions of <u>Section 7.03</u>;

(c) Indebtedness under the Loan Documents;

(d) Indebtedness outstanding (i) under the Prepetition Debt Documents and (ii) on the Closing Date, as set forth on <u>Schedule 7.02</u>;

(e) Guarantees of Holdings, the Borrower, or any Subsidiary in respect of Indebtedness otherwise permitted hereunder of the Borrower or any of its Subsidiaries; and

(f) Indebtedness in respect of Capitalized Leases, Synthetic Lease Obligations, Synthetic Debt and purchase money obligations for fixed or capital assets within the limitations set forth in <u>Section 7.01(i)</u>; provided, <u>however</u>, that the aggregate amount of all such Indebtedness at any one time outstanding shall not exceed \$5,000,000.

7.03 <u>Investments</u>.

Make or hold any Investments, except:

(a) Investments held by Holdings and its Subsidiaries in the form of Cash Equivalents;

(b) advances to officers, directors and employees of Holdings and its Subsidiaries in an aggregate amount not to exceed \$500,000 at any time outstanding, for travel, entertainment, relocation and analogous ordinary business purposes, in each case, to the extent permitted under the Approved Budget;

(c) (i) Investments by Holdings and its Subsidiaries in Loan Parties (other than Holdings) and (ii) additional Investments by Subsidiaries of the Borrower that are not Loan Parties in other Subsidiaries of the Borrower that are not Loan Parties (but solely in the form of holding Equity Interests therein (but making no contribution thereto after the Closing Date)); provided that each such Investment that consists of intercompany Indebtedness owing to a Loan Party shall be evidenced by an intercompany note which shall be delivered to, and in form

reasonably acceptable to, the Administrative Agent and shall constitute "Pledged Debt" pursuant to the terms of the Security Agreement;

(d) Investments consisting of extensions of credit in the nature of accounts receivable or notes receivable arising from the grant of trade credit in the ordinary course of business consistent with past practice, to the extent such extension is for a period of not longer than 90 days, and Investments received in satisfaction or partial satisfaction thereof from financially troubled account debtors to the extent reasonably necessary in order to prevent or limit loss;

- (e) Guarantees permitted by <u>Section 7.02;</u>
- (f) Investments existing on the Closing Date and set forth on <u>Schedule 5.08(e)</u>;
- (g) Investments by the Borrower in Swap Contracts permitted under <u>Section 7.02(a);</u> and

(h) any other Investments consented to in writing by the Administrative Agent (at the direction of the Required Lenders).

7.04 <u>Fundamental Changes</u>.

Merge, dissolve, liquidate, consolidate or amalgamate with or into another Person, or Dispose of (whether in one transaction or in a series of transactions) all or substantially all of its assets (whether now owned or hereafter acquired) to or in favor of any Person, except that, so long as no Default exists or would result therefrom on a pro forma basis:

(a) any Domestic Subsidiary may merge with (i) the Borrower, <u>provided</u> that the Borrower shall be the continuing or surviving Person, or (ii) any one or more other Domestic Subsidiaries, <u>provided</u> that when any Loan Party is merging with another Subsidiary, such Loan Party shall be the continuing or surviving Person;

(b) [reserved];

(c) any Loan Party may Dispose of all or substantially all of its assets (upon voluntary liquidation or otherwise) to the Borrower or to another Loan Party (other than Holdings); and

(d) any Subsidiary that is not a Loan Party may dispose of all or substantially all its assets (including any Disposition that is in the nature of a liquidation) to any other Subsidiary.

7.05 <u>Dispositions</u>.

Make any Disposition or enter into any agreement to make any Disposition, or issue or sell any shares of Equity Interests to any Person except:

(a) Dispositions of obsolete or worn out property, whether now owned or hereafter acquired, in the ordinary course of business;

(b) Dispositions of inventory in the ordinary course of business consistent with past practice;

(c) Dispositions of equipment to the extent that (i) such property is exchanged for credit against the purchase price of similar replacement property or (ii) the net proceeds of such Disposition are reasonably promptly applied to the purchase price of such replacement property;

(d) Dispositions of property by any Subsidiary to the Borrower or to any other Loan Party;

(e) Dispositions permitted by <u>Section 7.04;</u>

(f) Dispositions of equipment provided that to the extent the aggregate amount of Net Cash Proceeds of all equipment sold pursuant to this clause (f) exceeds \$1,000,000, such excess proceeds are applied to repay the Loans pursuant to <u>Section 2.09(b)</u>; and

(g) non-exclusive licenses of IP Rights in the ordinary course of business and substantially consistent with past practice;

<u>provided</u>, <u>however</u>, that any Disposition pursuant to <u>Section 7.05(a)</u> through <u>Section 7.05(g)</u> shall be for fair market value (as reasonably determined by the Borrower or the applicable Subsidiary).

7.06 <u>Restricted Payments</u>.

Declare or make, directly or indirectly, any Restricted Payment, or incur any obligation (contingent or otherwise) to do so, except that:

(a) each Subsidiary (hereinafter, the "<u>Payor</u>") may make Restricted Payments to (i) the Borrower, (ii) any Subsidiaries of the Borrower that are Guarantors, and (iii) in the event the Payor files a consolidated, combined, unitary or similar type income tax return with Holdings, the Borrower, or any other Subsidiaries that are Guarantors and own a direct or indirect Equity Interest in Payor, to such parties and their Subsidiaries to permit parties to pay federal and state income taxes then due and payable, franchise taxes and other similar licensing expenses incurred in the ordinary course of business; provided, that the amount of such Restricted Payment shall not be greater than the amount of such taxes or expenses that would have been due and payable by such parties and any Subsidiaries of such parties had such parties not filed a consolidated, combined, unitary or similar type return with the Payor;

(b) Holdings and each Subsidiary may declare and make dividend payments or other distributions payable solely in the common stock or other common Equity Interests of such Person;

(c) Holdings and each Subsidiary may purchase, redeem or otherwise acquire its common Equity Interests with the proceeds received from the substantially concurrent issue of new common Equity Interests; and

(d) in addition to the Restricted Payments permitted in clause (a), the Borrower may declare and pay cash dividends to Holdings so long as no Default exists or would be caused thereby, in an aggregate amount not to exceed \$1,000,000 in any fiscal year to permit Holdings to pay reasonable and customary corporate and operating expenses (including reasonable out-of-pocket expenses for legal, administrative and accounting services provided by third parties, and compensation, benefits and other amounts payable to officers and employees in connection with their employment in the ordinary course of business and to board of director observers) but only to the extent such payments are actually made by Holdings.

7.07 <u>Change in Nature of Business</u>.

Engage in any material line of business substantially different from those lines of business conducted by the Borrower and its Subsidiaries on the Closing Date or any business substantially related or incidental thereto.

7.08 <u>Transactions with Affiliates</u>.

Enter into any transaction of any kind with any Affiliate of the Borrower (other than the Loan Parties), whether or not in the ordinary course of business, other than in compliance with the Approved Budget and solely if it is on fair and reasonable terms and conditions at least as favorable to the Borrower or such Subsidiary as would be obtainable by the Borrower or such Subsidiary at the time in a comparable arm's length transaction with a Person other than an Affiliate.

7.09 <u>Burdensome Agreements</u>.

Enter into or permit to exist any Contractual Obligation (other than this Agreement or any other Loan Document) that (a) limits the ability (i) of any Subsidiary to make Restricted Payments to the Borrower or any Guarantor or to otherwise transfer property to or invest in the Borrower or any Guarantor, except for any agreement in effect (A) on the Closing Date and set forth on <u>Schedule 7.09</u> (including the Prepetition Debt Documents) or (B) at the time any Subsidiary becomes a Subsidiary of the Borrower, so long as such agreement was not entered into solely in contemplation of such Person becoming a Subsidiary of the Borrower, (ii) of any Subsidiary to Guarantee the Indebtedness of the Borrower or (iii) of the Borrower or any Subsidiary to create, incur, assume or suffer to exist Liens on property of such Person, except for any agreement in effect on the Closing Date and set forth on <u>Schedule 7.09</u> (including the Prepetition Debt Documents).

7.10 <u>Use of Proceeds</u>.

Use the proceeds of any Borrowing, whether directly or indirectly, and whether immediately, incidentally or ultimately, for purposes of "buying" or "carrying" any "margin stock" (within the respective meanings of Regulation U issued by the FRB), extending credit to others for the purpose of "buying" or "carrying" any "margin stock" or refunding indebtedness originally incurred for such purpose or in an manner other than as asset forth in Section 5.13 and 6.11.

7.11 <u>Financial Covenants</u>.

(a) Permit any Variance to occur for the most recently ended Testing Period (commencing with the first Testing Period after the Petition Date) other than a Permitted Variance.

(b) After the entry of the Final Order, permit Minimum Liquidity at any time to be less than \$35,000,000.

7.12 <u>Capital Expenditures</u>.

Make or become legally obligated to make any Capital Expenditure, except for (i) Capital Expenditures in the ordinary course of business and not exceeding \$15,000,000 in the aggregate for the Borrower and its Subsidiaries during any such fiscal year and (ii) Capital Expenditures in connection with facility and plant closures and consolidations not to exceed \$10,000,000 during any such fiscal year (the "<u>Maximum Cap Ex Amounts</u>"), in each case, in accordance with the Approved Budget.

7.13 <u>Amendments of Organization Documents or Material Contracts.</u>

Amend any of its Organization Documents in any respect.

7.14 <u>Accounting Changes</u>.

Make any change in (a) accounting policies or reporting practices, except as required by GAAP or applicable Laws, or (b) its fiscal year, except in the case of this clause (b) to a fiscal year ending December 31 or otherwise upon 30 days prior written notice to the Administrative Agent.

7.15 <u>Prepayments, Etc. of Indebtedness</u>.

Prepay, redeem, purchase, defease or otherwise satisfy prior to the scheduled maturity thereof in any manner, or make any payment in violation of any subordination terms of, any Indebtedness, except:

(a) the prepayment of the Loans in accordance with the terms of this Agreement and any repayment of the loans under the Prepetition ABL Credit Agreement or the ABL DIP Loans; and

(b) regularly scheduled or required repayments, purchases or redemptions of Indebtedness set forth in <u>Schedule 7.02</u> (except any such repayment, purchase or redemption made under the Prepetition Debt Documents, other than as permitted by the Orders).

Notwithstanding the foregoing clause (a) of this <u>Section 7.15</u>, Holdings and the Borrower shall not, nor shall they permit any Subsidiary to, directly or indirectly, make any payment in violation of the Orders.

7.16 Amendment, Etc. of Related Documents and Indebtedness.

(a) Cancel or terminate any Prepetition Debt Documents or consent to or accept any cancellation or termination thereof, (b) cancel or terminate any Material Contract or consent to or accept any cancellation or termination thereof, unless the cancellation or termination thereof could not reasonably be expected to have a Material Adverse Effect, (c) amend, modify or change in any manner any term or condition of any Prepetition Debt Documents or give any consent, waiver or approval thereunder that, in any such case, could impair the rights and remedies of the Secured Parties under the Loan Documents or otherwise result in a Material Adverse Effect, (d) amend, modify or change in any manner any term or condition of any Material Contract or give any consent, waiver or approval thereunder unless such amendment, modification or change could not reasonably be expected to have a Material Adverse Effect, (e) [intentionally omitted], (f) waive any material default or any breach of any material term or condition of any Material Contract, (g) take any other action in connection with any Related Document that would impair in any material respect the value of the interest or rights of any Loan Party thereunder or that would impair the rights or interests of the Administrative Agent or any Loan Party thereunder or that would impair the rights or condition of any Indebtedness set forth in <u>Schedule 7.02</u>.

7.17 <u>Holding Company</u>.

In the case of Holdings, engage in any business or activity other than (a) the ownership of all outstanding Equity Interests in the Borrower and CNMW Investments, Inc., (b) maintaining its corporate existence, (c) participating in tax, accounting and other administrative activities as the parent of the consolidated group of companies, including the Loan Parties, (d) the execution and delivery of the Loan Documents, the Related Documents to which it is a party and the Senior Notes Documents to which it is a party and the performance of its obligations thereunder, (e) activities incidental to the businesses or

activities described in clauses (a) through (d) of this Section and (f) any other activities substantially similar or related to the lines of business of Holdings in the ordinary course of business as of the Closing Date to the extent otherwise permitted under the Loan Documents.

7.18 <u>Sanctions.</u>

Directly or indirectly, use the proceeds of any Borrowing, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other individual or entity, to fund any activities of or business with any individual or entity, or in any Designated Jurisdiction that, at the time of such funding, is the subject of Sanctions, or in any other manner that will result in a violation by any individual or entity (including any individual or entity participating in the transaction, whether as Lender, Administrative Agent or otherwise) of Sanctions.

7.19 Chapter 11 Cases.

(a) Except for the Carve-Out, incur, create, assume, suffer to exist or permit, or file any motion seeking, any other superpriority claim which is pari passu with, or senior to the claims, security interests and Liens of, the Administrative Agent, Lenders and any other Secured Parties.

(b) Make or permit to be made any change to the Orders, as applicable, without the prior written consent of the Required Lenders.

(c) Commence any adversary proceeding, contested matter or other action asserting any claims or defenses or otherwise against the Administrative Agent, any Lender (or Affiliate thereof) or the holders of the 6% Notes with respect to this Agreement, the other Loan Documents, the transactions contemplated hereby or thereby, the 6% Notes Documents, the other documents or agreements executed or delivered in connection therewith or the transactions contemplated thereby.

(d) Make (i) any prepetition "critical vendor" payments or other payments on account of any creditor's prepetition unsecured claim, (ii) payments on account of claims or expenses arising under section 503(b)(9) of the Bankruptcy Code or (iii) payments under any management incentive plan or on account of claims or expenses arising under section 503(c) of the Bankruptcy Code, except in amounts and on terms and conditions that (a) are approved by order of the Bankruptcy Court after notice and a hearing and (b) are expressly permitted by the terms of the Loan Documents and within the limits, including any allowed variance, of the Approved Budget or otherwise with the consent of the Required Lenders.

(e) File any motion or application with the Bankruptcy Court with regard to actions taken outside the ordinary course of business of the Loan Parties without consulting with the Lenders and providing the Lenders prior (in any case, not less than two (2) Business Days' (or such lesser time as may be acceptable to Required Lenders in their sole discretion)) notice and the opportunity to review and comment on each such motion.

ARTICLE VIII

EVENTS OF DEFAULT AND REMEDIES

8.01 <u>Events of Default</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 391 of 432

Any of the following shall constitute an Event of Default:

(a) <u>Non-Payment</u>. The Borrower or any other Loan Party fails to (i) pay when and as required to be paid herein, any amount of principal of any Loan, (ii) pay within three (3) Business Days after the same becomes due, any interest on any Loan or any fee due hereunder, or (iii) pay within five (5) Business Days after the same becomes due, any other amount payable hereunder or under any other Loan Document; or

(b) <u>Specific Covenants</u>. (i) The Borrower fails to perform or observe any term, covenant or agreement contained in any of <u>Section 6.01</u>, <u>6.02(b)</u>, <u>6.03</u>, <u>6.05</u>, <u>6.10</u>, <u>6.11</u>, <u>6.12</u>, <u>6.14</u>, <u>6.15</u>, <u>6.17</u>, <u>6.19</u> or <u>6.21</u> or <u>Article VII</u>, (ii) any Guarantor fails to perform or observe any term, covenant or agreement set forth in the foregoing subclause (i) that such Guarantor has agreed to perform or observe pursuant to <u>Section 4.01</u> of any Guaranty to which it is a party, or (iii) any Loan Party fails to perform or observe any term, covenant or agreement contained in Section 5.01(c), 5.01(d), Section 5.01(g), or Section 5.02 of the Security Agreement, as applicable; or

(c) <u>Other Defaults</u>. Any Loan Party fails to perform or observe any other covenant or agreement (not specified in <u>Section 8.01(a)</u> or (b) above) contained in any Loan Document on its part to be performed or observed and such failure continues for 30 days from the earlier of (i) the date that any Responsible Officer of a Loan Party has actual knowledge thereof or (ii) the date that the Administrative Agent delivers to Borrower written notice of such failure; or

(d) <u>Representations and Warranties</u>. Any representation, warranty, certification or statement of fact made or deemed made by or on behalf of the Borrower or any other Loan Party herein, in any other Loan Document, or in any document delivered in connection herewith or therewith shall be incorrect or misleading in any material respect when made or deemed made; or

Cross-Default. (i) Any Loan Party or any Subsidiary thereof (A) fails to make (e) any payment when due (whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise) in respect of any Indebtedness or Guarantee (other than Indebtedness hereunder and Indebtedness under Swap Contracts) having an aggregate principal amount (including undrawn committed or available amounts and including amounts owing to all creditors under any combined or syndicated credit arrangement) of more than the Threshold Amount, or (B) fails to observe or perform any other agreement or condition relating to any such Indebtedness or Guarantee or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event occurs, the effect of which default or other event is to cause, or to permit the holder or holders of such Indebtedness or the beneficiary or beneficiaries of such Guarantee (or a trustee or agent on behalf of such holder or holders or beneficiary or beneficiaries) to cause, with the giving of notice if required, such Indebtedness to be demanded or to become due or to be repurchased, prepaid, defeased or redeemed (automatically or otherwise), or an offer to repurchase, prepay, defease or redeem such Indebtedness to be made, prior to its stated maturity, or such Guarantee to become payable or cash collateral in respect thereof to be demanded; or (ii) there occurs under any Swap Contract an Early Termination Date (as defined in such Swap Contract) resulting from any event of default under such Swap Contract as to which a Loan Party or any Subsidiary thereof is the Defaulting Party (as defined in such Swap Contract) and the Swap Termination Value owed by such Loan Party or such Subsidiary as a result thereof is greater than the Threshold Amount; provided, however, if the exercise of any rights or remedies by a holder of such Indebtedness, beneficiary of such Guarantee or counterparty to such Swap Contract with respect to any of the events described in the foregoing clauses (i) and (ii) is staved by the Bankruptcy Code or enjoined by the Bankruptcy Court, then such events shall not

constitute an "Event of Default" so long as such exercise of rights or remedies remain stayed or enjoined; or

- (f) [reserved]; or
- (g) [reserved]; or

(h) <u>Judgments</u>. There is entered against any Loan Party or any Subsidiary thereof (i) one or more final judgments or orders for the payment of money in an aggregate amount (as to all such judgments and orders) exceeding \$750,000 (to the extent not covered by independent third-party insurance as to which the insurer has an A.M. Best Financial Strength Rating of at least "A," been notified of the potential claim and does not dispute coverage), or (ii) any one or more non-monetary final judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect and, in either case, (A) enforcement proceedings are commenced by any creditor upon such judgment or order, or (B) there is a period of 30 consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect; or

(i) <u>ERISA</u>. An ERISA Event occurs, or is reasonably likely to occur, which has resulted or could reasonably be expected to result in a Material Adverse Effect; or

(j) [<u>reserved</u>]; or

(k) <u>Invalidity of Loan Documents</u>. Any material provision of any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect; or any Loan Party or any other Person contests in any manner the validity or enforceability of any provision of any Loan Document; or any Loan Party denies that it has any or further liability or obligation under any provision of any Loan Document, or purports to revoke, terminate or rescind any provision of any Loan Document; or

(l) <u>Change of Control</u>. There occurs any Change of Control; or

(m) <u>Collateral Documents</u>. Any Lien purported to be created pursuant to the Orders or under any Collateral Document shall for any reason (other than pursuant to the terms hereof of thereof) cease to be, or shall be asserted by any Loan Party or any Subsidiary thereof not to create a valid and perfected Lien on the Collateral purported to be covered thereby, with the priority required by the Orders or there occurs any uninsured loss of any portion of the Collateral that could be reasonably likely to result in a Material Adverse Effect; or

(n) <u>Guaranty</u>. The termination or attempted termination of the Guaranty by any Loan Party, any Subsidiary or any Affiliate thereof, except as expressly permitted hereunder or under any other Loan Document

(o) <u>Chapter 11 Cases</u>. There shall have occurred any of the following in the Chapter 11 Cases:

(i) the bringing of a motion or taking of any action, in each case, by any Loan Party in the Chapter 11 Cases, or the entry of any order by the Bankruptcy Court in the Chapter 11 Cases: (w) to obtain additional financing under section 364(c) or (d) of the Bankruptcy Code not otherwise permitted pursuant to this Agreement or that does not provide for the repayment of all

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 393 of 432

Obligations under this Agreement in full in cash on the date of closing of such additional financing; (x) to grant any Lien other than Liens expressly permitted under this Agreement upon or affecting any Collateral; (y) except as provided in the Interim Order or Final Order, as the case may be, to use cash collateral of the Administrative Agent under section 363(c) of the Bankruptcy Code without the prior written consent of the Administrative Agent and the Required Lenders; or (z) that (in the case of any Loan Party) requests or seeks authority for or that (in the case of an order entered by the Bankruptcy Court on account of a request by any Loan Party) approves or provides authority to take any other action or actions adverse to the Administrative Agent and the Lenders or their rights and remedies hereunder or their interest in the Collateral;

(ii) the filing of any plan of reorganization or disclosure statement attendant thereto, or any direct or indirect amendment to such plan or disclosure statement, by any Loan Party which does not provide for the repayment of all Obligations under this Agreement (x) in a manner consistent with the Restructuring Support Agreement or (y) in full in cash, in each case, on the Chapter 11 Effective Date and to which the Administrative Agent and the Required Lenders do not consent or otherwise agree to the treatment of their claims or the termination of any Loan Party's exclusive right to file and solicit acceptances of a plan of reorganization;

(iii) the entry of an order in any of the Chapter 11 Cases confirming a plan or plans of reorganization (or of liquidation) other than the Plan of Reorganization;

(iv) the entry of an order amending, supplementing, staying, vacating or otherwise modifying any Loan Document or the Interim Order or the Final Order in any case without the prior written consent of the Administrative Agent and the Required Lenders;

(v) the failure to comply with any Milestone, except to the extent such Milestone is extended to a later date with the prior written consent of the Required Lenders;

(vi) the payment of, or application by any Loan Party for authority to pay, any prepetition claim without the Administrative Agent's and the Required Lenders' prior written consent other than as provided in any "first day order" in form and substance reasonably acceptable to the Required Lenders and as set forth in the Approved Budget or unless otherwise permitted under this Agreement;

(vii) the entry of an order by the Bankruptcy Court appointing, or the filing of an application by any Loan Party, for an order seeking the appointment of, in either case without the consent of the Required Lenders, an interim or permanent trustee in the Chapter 11 Cases or the appointment of a receiver or an examiner under section 1104 of the Bankruptcy Code in the Chapter 11 Cases with expanded powers (beyond those set forth in sections 1106(a)(3) and 1106(a)(4) of the Bankruptcy Code) to operate or manage the financial affairs, the business, or reorganization of the Loan Parties or with the power to conduct an investigation of (or compel discovery from) the Administrative Agent or the Lenders; or the sale without the Administrative Agent's and the Required Lenders' consent, of all or substantially all of a Loan Party's assets either through a sale under section 363 of the Bankruptcy Code, through a confirmed plan of reorganization in the Chapter 11 Cases, or otherwise, in each case, that does not provide for payment in full in cash of the Obligations on the closing date of such sale;

(viii) the dismissal of the Chapter 11 Cases, or if any Loan Party shall file a motion or other pleading seeking the dismissal of the Chapter 11 Cases;

(ix) the conversion of the Chapter 11 Cases from one under chapter 11 to one under chapter 7 of the Bankruptcy Code or any Loan Party shall file a motion or other pleading seeking the conversion of the Chapter 11 Cases under section 1112 of the Bankruptcy Code or otherwise;

(x) the entry of an order by the Bankruptcy Court granting relief from or modifying the automatic stay of section 362 of the Bankruptcy Code (x) to allow any creditor to execute upon or enforce a Lien on any Collateral, or (y) with respect to any Lien of or the granting of any Lien on any Collateral to any state or local environmental or regulatory agency or authority;

(xi) the entry of an order in the Chapter 11 Cases avoiding or requiring repayment of any portion of the payments made on account of the Obligations owing under this Agreement or the other Loan Documents;

(xii) the failure of any Loan Party to perform any of its obligations under the Interim Order or the Final Order or any violation of any of the terms of the Interim Order or the Final Order;

(xiii) the challenge by any Loan Party to the validity, extent, perfection or priority of any liens granted under, or any obligations arising under, the 6% Notes Documents or the Loan Documents;

(xiv) the remittance, use or application of cash collateral other than in accordance with any cash management procedures and orders entered by the Bankruptcy Court;

(xv) the entry of an order (other than the Orders) in any of the Chapter 11 Cases granting any other super priority administrative claim or Lien equal or superior to that granted to the Agent, on behalf of itself and the Lenders, without the consent in writing of the Administrative Agent and the Required Lenders;

(xvi) the filing of a motion by any Loan Party requesting, or the entry of any order granting, any super-priority claim which is senior or pari passu with the Lenders' claims except to the extent the claim relates to new financing that provides for the repayment of all Obligations under this Agreement irrevocably in full in cash on the closing of such new financing;

(xvii) except to the extent expressly permitted by the terms of the Restructuring Support Agreement, the entry of an order precluding or modifying the Administrative Agent from having the right to or being permitted to "credit bid";

(xviii) any attempt by any Loan Party to reduce, set off or subordinate the Obligations or the Liens securing such Obligations to any other debt;

(xix) the Interim Order or the Final Order or any provision thereof (as applicable) is reversed, vacated or stayed, in each case, without the consent of the Required Lenders;

(xx) the payment of or granting adequate protection with respect to any Prepetition Indebtedness, except as set forth in the Interim order or the Final Order;

(xxi) an application for any of the orders described in this <u>Section 8.01(p)</u> shall be made by a Person other than the Administrative Agent or the Lenders and such application is not contested by the Loan Parties in good faith and the relief requested is granted in an order that is not stayed pending appeal;

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 395 of 432

(xxii) the cessation of Liens or super-priority claims granted with respect to this Agreement to be valid, perfected and enforceable in all respects; or

(xxiii) the Bankruptcy Court shall cease to have exclusive jurisdiction with respect to all matters relating to the exercise of rights and remedies under the Loan Documents, the Orders, the Liens and the Collateral.

(p) <u>Restructuring Support Agreement</u>. The termination or ineffectiveness of, or any breach under, the Restructuring Support Agreement.

8.02 <u>Remedies upon Event of Default.</u>

If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:

(a) declare the Aggregate Commitments to be terminated, whereupon such commitments shall be terminated;

(b) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower;

(c) terminate, reduce or condition any Commitment;

(d) exercise on behalf of itself and the Lenders all rights and remedies available to it and the Lenders under the Loan Documents or applicable Law;

(e) foreclose and sell, or cause to be foreclosed and sold any and all real property of the Loan Parties;

(f) immediately terminate the Loan Parties' use of Cash Collateral (as defined in the Order in effect at such time);

(g) terminate the Facility and any Loan Document as to any future liability or obligation of the Secured Parties, but without affecting any of the Obligations or the Liens securing the Obligations;

(h) freeze monies or balances in each Loan Party's Accounts;

(i) immediately set-off any and all amounts in Accounts maintained by the Loan Parties with the Administrative Agent or the Lenders against the Obligations, or otherwise enforce any and all rights against the Collateral in the possession of the applicable Lenders, including, without limitation, foreclosure on all or any portion of the Collateral, collection of accounts receivable, occupying the applicable Loan Party's premises and sale or disposition of the Collateral;

(i) apply any and all monies owing by any Secured Party to any Loan Party to the payment of the Loans, including interest accrued thereon, or to payment of any or all other Obligations then owing by the Loan Parties; and

(j) subject to the Interim Order and Final Order, as applicable, exercise any other rights and remedies available to the Agent (including, without limitation, hereunder and under any other Loan Documents), the Lenders or any other Secured Party under applicable Law, any Loan Document, the Interim Order, the Final Order or otherwise.

8.03 Code and Other Remedies.

(a) <u>UCC Remedies</u>. Subject to the Orders, during the continuance of an Event of Default, the Administrative Agent may exercise, in addition to all other rights and remedies granted to it in this Agreement, any other Loan Documents and in any other instrument or agreement securing, evidencing or relating to any Obligation, all rights and remedies of a secured party under the UCC or any other applicable law.

(b) Disposition of Collateral. Subject to the Orders, the Administrative Agent may, without demand of performance or other demand, presentment, protest, advertisement or notice of any kind (except any notice required by the Interim Order or Final Order and any notice required by law referred to below) to or upon any Loan Party or any other Person (all and each of which demands, defenses, advertisements and notices are hereby waived (except as required by the Interim Order or the Final Order)), during the continuance of any Event of Default (personally or through its agents or attorneys), (i) enter upon the premises where any Collateral is located, without any obligation to pay rent, through selfhelp, without judicial process, without first obtaining a final judgment or giving any Loan Party or any other Person notice or opportunity for a hearing on the Administrative Agent's claim or action. (ii) collect, receive, appropriate and realize upon any Collateral, (iii) Dispose of, grant option or options to purchase and deliver any Collateral (enter into Contractual Obligations to do any of the foregoing), in one or more parcels at public or private sale or sales, at any exchange, broker's board or office of any Secured Party or elsewhere upon such terms and conditions as it may deem advisable and at such prices as it may deem best, for cash or on credit or for future delivery without assumption of any credit risk, (iv) withdraw all cash and Cash Equivalents in any Deposit Account or Securities Account of a Loan Party and apply such cash and Cash Equivalents and other cash, if any, then held by it as Collateral in satisfaction of the Obligations, and (v) give notice and take sole possession and control of all amounts on deposit in or credited to the DIP Collateral Account or any other Deposit Account or Securities Account pursuant to the related account control agreement. The Administrative Agent shall have the right, upon any such public sale or sales and, to the extent permitted by the UCC and other applicable Laws, upon any such private sale, to purchase the whole or any part of the Collateral so sold (and, in lieu of actual payment of the purchase price, may "credit bid" or otherwise set off the amount of such price against the Obligations), free of any right or equity of redemption of any Loan Party, which right or equity is hereby waived and released.

(c) <u>Management of the Collateral</u>. Subject to the Orders, each Loan Party further agrees, that, during the continuance of any Event of Default, (i) at the Administrative Agent's request, it shall assemble the Collateral and make it available to the Administrative Agent at places that the Administrative Agent shall reasonably select, whether at such Loan Party's premises or elsewhere, (ii) without limiting the foregoing, the Administrative Agent also has the right to require that each Loan Party store and keep any Collateral pending further action by the Administrative Agent and, while any such Collateral is so stored or kept, provide such guards and maintenance services as shall be necessary to protect the same and to preserve and maintain such Collateral in good condition, (iii) until the Administrative Agent is able to Dispose of any Collateral, the Administrative Agent shall have the right to hold or use such Collateral to the extent that it deems appropriate for the purpose of preserving the Collateral or its value or for any other purpose deemed appropriate by the Agent and (iv) the Administrative Agent may, if it so elects, seek the appointment of a receiver or keeper to take possession of any Collateral and to enforce any of the Administrative Agent's remedies (for the benefit of the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 397 of 432

Secured Parties), with respect to such appointment without prior notice or hearing as to such appointment. The Administrative Agent shall not have any obligation to any Loan Party to maintain or preserve the rights of any Loan Party as against third parties with respect to any Collateral while such Collateral is in the possession of the Administrative Agent.

(d) <u>Direct Obligation</u>. Subject to the Orders, neither the Administrative Agent nor any other Secured Party shall be required to make any demand upon, or pursue or exhaust any right or remedy against, any Loan Party, any other Loan Party or any other Person with respect to the payment of the Obligations or to pursue or exhaust any right or remedy with respect to any Collateral therefor or any direct or indirect guaranty thereof. All of the rights and remedies of the Administrative Agent and any other Secured Party under any Loan Document shall be cumulative, may be exercised individually or concurrently and not exclusive of any other rights or remedies provided by any applicable Law. To the extent it may lawfully do so, each Loan Party absolutely and irrevocably waives and relinquishes the benefit and advantage of, and covenants not to assert against the Administrative Agent or any Lender, any valuation, stay, appraisement, extension, redemption or similar laws and any and all rights or defenses it may have as a surety, now or hereafter existing, arising out of the exercise by them of any rights hereunder. If any notice of a proposed sale or other disposition of any Collateral shall be required by law, such notice shall be deemed reasonable and proper if given at least 10 days before such sale or other disposition.

(e) <u>Commercially Reasonable</u>. To the extent that applicable Law imposes duties on the Administrative Agent to exercise remedies in a commercially reasonable manner, each Loan Party acknowledges and agrees that it is not commercially unreasonable for the Administrative Agent to do any of the following:

(i) fail to incur significant costs, expenses or other Liabilities reasonably deemed as such by the Administrative Agent to prepare any Collateral for disposition or otherwise to complete raw material or work in process into finished goods or other finished products for disposition;

(ii) fail to obtain Permits, or other consents, for access to any Collateral to Dispose of or for the collection or Disposition of any Collateral, or, if not required by other applicable Laws fail to obtain Permits or other consents for the collection or disposition of any Collateral;

(iii) fail to exercise remedies against account debtors or other Persons obligated on any Collateral or to remove Liens on any Collateral or to remove any adverse claims against any Collateral;

(iv) advertise dispositions of any Collateral through publications or media of general circulation, whether or not such Collateral is of a specialized nature or to contact other Persons, whether or not in the same business as any Loan Party, for expressions of interest in acquiring any such Collateral;

(v) exercise collection remedies against account debtors and other Persons obligated on any Collateral, directly or through the use of collection agencies or other collection specialists, hire one or more professional auctioneers to assist in the disposition of any Collateral, whether or not such Collateral is of a specialized nature or, to the extent deemed appropriate by the Administrative Agent, obtain the services of other brokers, investment bankers, consultants and other professionals to assist the Administrative Agent in the collection or disposition of any Collateral, or utilize Internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capacity of doing so, or that match buyers and sellers of assets to dispose of any Collateral;

- (vi) dispose of assets in wholesale rather than retail markets;
- (vii) disclaim disposition warranties, such as title, possession or quiet enjoyment; or

(viii) purchase insurance or credit enhancements to insure the Administrative Agent against risks of loss, collection or disposition of any Collateral or to provide to the Administrative Agent a guaranteed return from the collection or disposition of any Collateral.

Each Loan Party acknowledges that the purpose of this Section 8.03 is to provide a nonexhaustive list of actions or omissions that are commercially reasonable when exercising remedies against any Collateral and that other actions or omissions by the Secured Parties shall not be deemed commercially unreasonable solely on account of not being indicated in this Section 8.03. Without limitation upon the foregoing, nothing contained in this Section 8.03 shall be construed to grant any rights to any Loan Party or to impose any duties on the Administrative Agent that would not have been granted or imposed by this Agreement or by applicable Law in the absence of this Section 8.03.

8.04 <u>Application of Funds.</u>

After the exercise of remedies provided for in <u>Section 8.02</u>, <u>Section 8.03</u> or pursuant to the Orders (or after the Loans have automatically become immediately due and payable as set forth above), any amounts received on account of the Obligations (including without limitation, proceeds received by the Administrative Agent in respect of any sale of, collection from, or other realization upon, all or any part of the Collateral (including, without limitation, pursuant to the exercise by the Administrative Agent of its remedies during the continuance of an Event of Default) or otherwise received on account of the Obligations) shall, subject to the provisions of <u>Sections 2.11</u> be applied in the following order:

<u>First</u>, to the payment of all reasonable costs and out-of-pocket expenses, fees, commissions and taxes of any sale, collection, other realization or other enforcement of remedies, including, without limitation, compensation to the Administrative Agent and its agents and counsel payable pursuant to the Loan Documents, and all expenses, liabilities and advances made or incurred by the Administrative Agent in connection therewith;

<u>Second</u>, to the payment of all other reasonable costs and out-of-pocket expenses of any sale, collection, other realization or other enforcement of remedies, including, without limitation, costs and expenses incurred by the other Secured Parties and all expenses, liabilities and advances made or incurred by the other Secured Parties in connection therewith;

<u>Third</u>, to interest and fees then due and payable on Loans and other amounts due pursuant to <u>Sections 3.02</u>, <u>3.03</u>, <u>3.04</u> and <u>3.05</u>;

Sixth, to the principal balance of Loans then outstanding on a pro rata basis;

Seventh, to all other Obligations on a pro rata basis; and

<u>Eighth</u>, the balance, if any, to the Person lawfully entitled thereto (including the applicable Loan Party or its successors or assigns).

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 399 of 432

In the event that any such proceeds are insufficient to pay in full the items described in clauses <u>First through Eighth</u> of this <u>Section 8.03</u>, the Loan Parties shall remain liable for any deficiency.

8.05 <u>Deficiency</u>. Each Loan Party shall remain liable for any deficiency if the proceeds of any sale or other disposition of any Collateral are insufficient to pay the Obligations.

ARTICLE IX

ADMINISTRATIVE AGENT

9.01 Appointment and Authority.

(a) Each of the Lenders hereby irrevocably appoints Wilmington Savings Fund Society, FSB to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent and the Lenders, and none of Holdings, the Borrower nor any other Loan Party shall have rights as a third party beneficiary of any of such provisions.

(b) [Reserved].

(c) Wilmington Savings Fund Society, FSB shall also act as the "collateral agent" under the Loan Documents, and each of the Lenders (in its capacities as a Lender) hereby irrevocably appoints and authorizes Wilmington Savings Fund Society, FSB to act as the agent of such Lender for purposes of acquiring, holding and enforcing any and all Liens on Collateral granted by any of the Loan Parties to secure any of the Obligations, together with such powers and discretion as are reasonably incidental thereto. In this connection, the Administrative Agent, as "collateral agent" and any co-agents, sub-agents and attorneys-in-fact appointed by the Administrative Agent pursuant to <u>Section 9.05</u> for purposes of holding or enforcing any rights and remedies thereunder at the direction of the Administrative Agent, shall be entitled to the benefits of all provisions of this <u>Article IX</u> and <u>Article XI</u> (including <u>Section 11.04(c)</u>, as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto.

(d) [Reserved].

9.02 <u>Rights as a Lender</u>.

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

9.03 <u>Exculpatory Provisions</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 400 of 432

The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:

(a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

(b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), <u>provided</u> that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable Law; and

(c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in <u>Sections 11.01</u> and <u>8.02</u>) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to the Administrative Agent by the Borrower or a Lender.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Collateral Documents, (v) the value or the sufficiency of any Collateral, or (v) the satisfaction of any condition set forth in <u>Article IV</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

9.04 <u>Reliance by Administrative Agent.</u>

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, that by its terms must be fulfilled to the satisfaction of a Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent shall have received notice to the contrary

from such Lender prior to the making of such Loan. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

9.05 <u>Delegation of Duties</u>.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

9.06 <u>Resignation of Administrative Agent.</u>

The Administrative Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower (and, so long as no Event of Default has occurred and is continuing, with the prior written consent of the Borrower, which consent shall not be unreasonably withheld or delayed), to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by the Administrative Agent on behalf of the Lenders under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 11.04 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

9.07 <u>Non-Reliance on Administrative Agent and Other Lenders.</u>

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 402 of 432

Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

9.08 [Reserved].

9.09 Administrative Agent May File Proofs of Claim.

In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to any Loan Party, the Administrative Agent (irrespective of whether the principal of any Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:

(a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and the Administrative Agent under <u>Sections 2.05</u> and <u>11.04</u>) allowed in such judicial proceeding; and

(b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same; and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Lenders, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under <u>Sections 2.05</u> and <u>11.04</u>.

(c) Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender to authorize the Administrative Agent to vote in respect of the claim of any Lender in any such proceeding.

9.10 Collateral and Guaranty Matters.

The Lenders irrevocably authorize the Administrative Agent, at its option and in its reasonable discretion, to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (i) upon termination of the Aggregate Commitments and payment in full of all Obligations (other than contingent indemnification obligations), (ii) that is Disposed of or to be Disposed of as part of or in connection with any Disposition permitted hereunder, or (iii) if approved, authorized or ratified in writing in accordance with <u>Section 11.01</u>.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 403 of 432

Upon request by the Administrative Agent any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guaranty pursuant to this <u>Section 9.10</u>. In each case as specified in this <u>Section 9.10</u>, the Administrative Agent will, at the Borrower's expense, execute and deliver to the applicable Loan Party such documents as such Loan Party may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Collateral Documents or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guaranty, in each case in accordance with the terms of the Loan Documents and this <u>Section 9.10</u>.

Credit Bidding. Subject to any express restrictions set forth in the Restructuring Support 9.11 Agreement, the Loan Parties and the Lenders hereby irrevocably authorize the Administrative Agent, based upon the instruction of the Required Lenders, to (a) consent to, credit bid or purchase (either directly or through one or more acquisition vehicles) all or any portion of the Collateral at any sale thereof conducted under the provisions of the Bankruptcy Code, including under section 363 of the Bankruptcy Code, (b) credit bid or purchase (either directly or through one or more acquisition vehicles) all or any portion of the Collateral at any Disposition thereof conducted under the provisions of the UCC, including pursuant to Sections 9-610 or 9-620 of the UCC, or (c) credit bid or purchase (either directly or through one or more acquisition vehicles) all or any portion of the Collateral at any other sale or foreclosure conducted by the Administrative Agent (whether by judicial action or otherwise) in accordance with applicable law. Subject to any express restrictions set forth in the Restructuring Support Agreement, in connection with any such credit bid or purchase, (i) the Obligations owed to the Lenders shall be entitled to be, and shall be, credit bid on a ratable basis (with Obligations with respect to contingent or unliquidated claims being estimated for such purpose if the fixing or liquidation thereof would not unduly delay the ability of the Administrative Agent to credit bid or purchase at such Disposition of the Collateral and, if such claims cannot be estimated without unduly delaying the ability of the Administrative Agent to credit bid, then such claims shall be disregarded, not credit bid, and not entitled to any interest in the asset or assets purchased by means of such credit bid) and the Lenders whose Obligations are credit bid shall be entitled to receive interests (ratably based upon the proportion of their Obligations credit bid in relation to the aggregate amount of Obligations so credit bid) in the asset or assets so purchased (or in the Equity Interests of the acquisition vehicle or vehicles that are used to consummate such purchase), and (ii) the Administrative Agent, based upon the instruction of the Required Lenders, may accept non-cash consideration, including debt and equity securities issued by such acquisition vehicle or vehicles and in connection therewith the Administrative Agent may reduce the Obligations owed to the Lenders (ratably based upon the proportion of their Obligations credit bid in relation to the aggregate amount of Obligations so credit bid) based upon the value of such non-cash consideration.

9.12 <u>Tax Indemnity</u>.

To the extent required by any applicable Laws (as determined in good faith by the Administrative Agent), the Administrative Agent may withhold from any payment to any Lender under any Loan Document an amount equivalent to any applicable withholding Tax. Without limiting or expanding the provisions of <u>Section 3.01</u>, each Lender shall indemnify and hold harmless the Administrative Agent against, and shall make payable in respect thereof within 10 days after demand therefor, any and all Taxes and any and all related losses, claims, liabilities and expenses (including fees, charges and disbursements of any counsel for the Administrative Agent) incurred by or asserted against the Administrative Agent by the IRS or any other Governmental Authority as a result of the failure of the Administrative Agent to properly withhold Tax from amounts paid to or for the account of such Lender for any reason (including because the appropriate form was not delivered or not properly executed, or because such Lender failed to notify the Administrative Agent of a change in circumstance that rendered the exemption from or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 404 of 432

reduction of withholding Tax ineffective), whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document against any amount due the Administrative Agent under this <u>Section 9.12</u>. The agreements in this <u>Section 9.12</u> shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender and the repayment, satisfaction or discharge of all other Obligations.

ARTICLE X

CONTINUING GUARANTY

10.01 <u>Guaranty</u>.

Holdings hereby absolutely and unconditionally guarantees, as a guaranty of payment and performance and not merely as a guaranty of collection, prompt payment when due, whether at stated maturity, by required prepayment, upon acceleration, demand or otherwise, and at all times thereafter, of any and all of the Obligations, whether for principal, interest, premiums, fees, indemnities, damages, costs, expenses or otherwise, of the Borrower to the Secured Parties, arising hereunder and under the other Loan Documents (including all renewals, extensions, amendments, refinancings and other modifications thereof and all costs, reasonable attorneys' fees and expenses incurred by the Secured Parties in connection with the collection or enforcement thereof), excluding, with respect to any Guarantor at any time, Excluded Swap Obligations with respect to such Guarantor at such time. The Administrative Agent's books and records showing the amount of the Obligations shall be admissible in evidence in any action or proceeding, and shall be binding upon Holdings, and conclusive for the purpose of establishing the amount of the Obligations, absent manifest error. This Guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Obligations or any instrument or agreement evidencing any Obligations, or by the existence, validity, enforceability, perfection, nonperfection or extent of any collateral therefor, or by any fact or circumstance relating to the Obligations which might otherwise constitute a defense to the obligations of Holdings under this Guaranty, and Holdings hereby irrevocably waives any defenses it may now have or hereafter acquire in any way relating to any or all of the foregoing, whether arising as a result of any law or regulation of any jurisdiction or any other event affecting any term of the Obligations.

10.02 <u>Rights of Lenders</u>.

Holdings consents and agrees that the Secured Parties may, at any time and from time to time, without notice or demand, and without affecting the enforceability or continuing effectiveness hereof: (a) amend, extend, renew, compromise, discharge, accelerate or otherwise change the time for payment or the terms of the Obligations or any part thereof; (b) take, hold, exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any security for the payment of this Guaranty or any Obligations; (c) apply such security and direct the order or manner of sale thereof as the Administrative Agent and the Lenders in their sole discretion may determine; and (d) release or substitute one or more of any endorsers or other guarantors of any of the Obligations. Without limiting the generality of the foregoing, Holdings consents to the taking of, or failure to take, any action which might in any manner or to any extent vary the risks of Holdings under this Guaranty or which, but for this provision, might operate as a discharge of Holdings.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 405 of 432

10.03 <u>Certain Waivers</u>.

Holdings waives (a) any defense arising by reason of any disability or other defense of the Borrower or any other guarantor, or the cessation from any cause whatsoever (including any act or omission of any Secured Party) of the liability of the Borrower; (b) any defense based on any claim that Holdings' obligations exceed or are more burdensome than those of the Borrower; (c) the benefit of any statute of limitations affecting Holdings' liability hereunder; (d) subject to Section 10.05, any right to proceed against the Borrower, proceed against or exhaust any security for the Obligations, or pursue any other remedy in the power of any Secured Party whatsoever; (e) subject to Section 10.05, any benefit of and any right to participate in any security now or hereafter held by any Secured Party; and (f) to the fullest extent permitted by law, any and all other defenses or benefits that may be derived from or afforded by applicable Law limiting the liability of or exonerating guarantors or sureties. Holdings expressly waives all setoffs and counterclaims and all presentments, demands for payment or performance, notices of nonpayment or nonperformance, protests, notices of protest, notices of dishonor and all other notices or demands of any kind or nature whatsoever with respect to the Obligations, and all notices of acceptance of this Guaranty or of the existence, creation or incurrence of new or additional Obligations. Holdings waives any rights and defenses that are or may become available to Holdings by reason of §§ 2787 to 2855, inclusive, and §§ 2899 and 3433 of the California Civil Code. As provided below, this Guaranty shall be governed by, and construed in accordance with, the laws of the State of New York. The foregoing waivers and the provisions hereinafter set forth in this Guaranty which pertain to California law are included solely out of an abundance of caution, and shall not be construed to mean that any of the above-referenced provisions of California law are in any way applicable to this Guaranty or the Obligations.

10.04 Obligations Independent.

The obligations of Holdings hereunder are those of primary obligor, and not merely as surety, and are independent of the Obligations and the obligations of any other guarantor, and a separate action may be brought against Holdings to enforce this Guaranty whether or not the Borrower or any other person or entity is joined as a party.

10.05 <u>Subrogation</u>.

Holdings shall not exercise any right of subrogation, contribution, indemnity, reimbursement or similar rights with respect to any payments it makes under this Guaranty until all of the Obligations and any amounts payable under this Guaranty have been paid in full in cash and performed in full and the Aggregate Commitments and this Agreement are terminated. If any amounts are paid to Holdings in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of the Secured Parties and shall forthwith be paid to the Secured Parties to reduce the amount of the Obligations, whether matured or unmatured.

10.06 <u>Termination; Reinstatement</u>.

This Guaranty is a continuing and irrevocable guaranty of all Obligations now or hereafter existing and shall remain in full force and effect until all Obligations and any other amounts payable under this Guaranty are paid in full in cash and the Aggregate Commitments and this Agreement are terminated. Notwithstanding the foregoing, this Guaranty shall continue in full force and effect or be revived, as the case may be, if any payment by or on behalf of the Borrower or Holdings is made, or any of the Secured Parties exercises its right of setoff, in respect of the Obligations and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by any of the Secured

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 406 of 432

Parties in their discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Laws or otherwise, all as if such payment had not been made or such setoff had not occurred and whether or not the Secured Parties are in possession of or have released this Guaranty and regardless of any prior revocation, rescission, termination or reduction. The obligations of Holdings under this paragraph shall survive termination of this Guaranty.

10.07 <u>Subordination</u>.

Holdings hereby subordinates the payment of all obligations and indebtedness of the Borrower owing to Holdings, whether now existing or hereafter arising, including but not limited to any obligation of the Borrower to Holdings as subrogee of the Secured Parties or resulting from Holdings' performance under this Guaranty, to the payment in full in cash of all Obligations. If the Secured Parties so request, any such obligation or indebtedness of the Borrower to Holdings shall be enforced and performance received by Holdings as trustee for the Secured Parties and the proceeds thereof shall be paid over to the Secured Parties on account of the Obligations, but without reducing or affecting in any manner the liability of Holdings under this Guaranty.

10.08 Stay of Acceleration.

If acceleration of the time for payment of any of the Obligations is stayed, in connection with any case commenced by or against Holdings or the Borrower under any Debtor Relief Laws, or otherwise, all such amounts shall nonetheless be payable by Holdings immediately upon demand by the Secured Parties.

10.09 <u>Condition of Borrower</u>.

Holdings acknowledges and agrees that it has the sole responsibility for, and has adequate means of, obtaining from the Borrower and any other guarantor such information concerning the financial condition, business and operations of the Borrower and any such other guarantor as Holdings requires, and that none of the Secured Parties has any duty, and Holdings is not relying on the Secured Parties at any time, to disclose to Holdings any information relating to the business, operations or financial condition of the Borrower or any other guarantor (Holdings waiving any duty on the part of the Secured Parties to disclose such information and any defense relating to the failure to provide the same).

10.10 Additional Guarantor Waivers and Agreements.

(a) Holdings understands and acknowledges that if the Secured Parties foreclose judicially or nonjudicially against any real property security for the Obligations, that foreclosure could impair or destroy any ability that Holdings may have to seek reimbursement, contribution, or indemnification from the Borrower or others based on any right Holdings may have of subrogation, reimbursement, contribution, or indemnification for any amounts paid by Holdings under this Guaranty. Holdings further understands and acknowledges that in the absence of this paragraph, such potential impairment or destruction of Holdings' rights, if any, may entitle Holdings to assert a defense to this Guaranty based on Section 580d of the California Code of Civil Procedure as interpreted in Union Bank v. Gradsky, 265 Cal. App. 2d 40 (1968). By executing this Guaranty, Holdings freely, irrevocably, and unconditionally: (i) waives and relinquishes that defense and agrees that Holdings will be fully liable under this Guaranty even though the Secured Parties may foreclose, either by judicial foreclosure or by exercise of power of sale, any deed of trust securing the Obligations; (ii) agrees that Holdings will not assert that defense in any action or proceeding which the Secured Parties may commence to enforce this Guaranty; (iii) acknowledges and agrees that the rights and defenses waived by Holdings in this Guaranty include any right or defense that Holdings may have or be entitled to assert based upon or arising out of any one or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 407 of 432

more of §§ 580a, 580b, 580d, or 726 of the California Code of Civil Procedure or § 2848 of the California Civil Code; and (iv) acknowledges and agrees that the Secured Parties are relying on this waiver in creating the Obligations, and that this waiver is a material part of the consideration which the Secured Parties are receiving for creating the Obligations.

(b) Holdings waives all rights and defenses that Holdings may have because any of the Obligations is secured by real property. This means, among other things, (i) the Secured Parties may collect from Holdings without first foreclosing on any real or personal property collateral pledged by the other Loan Parties; and (ii) if the Secured Parties foreclose on any real property collateral pledged by the other Loan Parties: (A) the amount of the Obligations may be reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is worth more than the sale price, and (B) the Secured Parties may collect from Holdings even if the Secured Parties, by foreclosing on the real property collateral, have destroyed any right Holdings may have to collect from the Borrower. This is an unconditional and irrevocable waiver of any rights and defenses Holdings may have because any of the Obligations is secured by real property. These rights and defenses include, but are not limited to, any rights or defenses based upon § 580a, 580b, 580d, or 726 of the California Code of Civil Procedure.

(c) Holdings waives any right or defense it may have at law or equity, including California Code of Civil Procedure § 580a, to a fair market value hearing or action to determine a deficiency judgment after a foreclosure.

10.11 <u>Keepwell</u>.

Each Loan Party that is a Qualified ECP Guarantor at the time the Guaranty or the grant of the security interest under the Loan Documents, in each case, by any Specified Loan Party, becomes effective with respect to any Swap Obligation, hereby jointly and severally, absolutely, unconditionally and irrevocably undertakes to provide such funds or other support to each Specified Loan Party with respect to such Swap Obligation as may be needed by such Specified Loan Party from time to time to honor all of its obligations under its Guaranty and the other Loan Documents in respect of such Swap Obligation (but, in each case, only up to the maximum amount of such liability that can be hereby incurred without rendering such Qualified ECP Guarantor's obligations and undertakings under this Article X voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations and undertakings of each Qualified ECP Guarantor under this Section shall remain in full force and effect until the Obligations have been paid and performed in full. Each Qualified ECP Guarantor intends this Section to constitute, and this Section shall be deemed to constitute, a guarantee of the obligations of, and a "keepwell, support, or other agreement" for the benefit of, each Specified Loan Party for all purposes of the Commodity Exchange Act.

ARTICLE XI

MISCELLANEOUS

11.01 <u>Amendments, Etc</u>.

(a) <u>Required Lender and Unanimous Consent</u>. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by the Required Lenders and the Borrower or the applicable Loan Party, as the case may be, and acknowledged by the Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; <u>provided</u>, <u>however</u>, that no such amendment, waiver or consent shall:

- (i) [reserved];
- (ii) [reserved];

(iii) subject to <u>Section 2.19</u>, extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 8.02</u>) without the written consent of such Lender;

(iv) subject to <u>Section 2.19</u>, postpone any date fixed by this Agreement or any other Loan Document for any payment (excluding mandatory prepayments) of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under such other Loan Document or modify the definition of "Maturity Date" (or waive any provision thereof) without the written consent of each Lender directly and adversely affected thereby;

(v) reduce the principal of, or the rate of interest specified herein on, any Loan, or (subject to clause (D) of the second proviso to this <u>Section 11.01(a)</u>) any fees or other amounts payable hereunder or under any other Loan Document, without the written consent of each Lender entitled to such amount; <u>provided</u>, <u>however</u>, that only the consent of the Required Lenders shall be necessary (i) to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest at the Default Rate or (ii) to amend any Financial Covenant hereunder (or any defined term used therein) even if the effect of such amendment would be to reduce the rate of interest on any Loan or to reduce any fee payable hereunder;

(vi) change <u>Section 2.10</u> or <u>Section 8.04</u> in a manner that would expressly alter the pro rata sharing of payments required thereby without the written consent of each Lender;

(vii) change any provision of this <u>Section 11.01</u> or the definition of "Required Lenders," "Supermajority Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;

(viii) subject to <u>Section 9.10</u>, release all or substantially all of the Collateral in any transaction or series of related transactions, without the written consent of each Lender;

(ix) subject to <u>Section 9.10</u>, release all or substantially all of the value of the Guaranty, without the written consent of each Lender;

(x) subordinate the Obligations or the Liens securing the Obligations without the written consent of each Lender; or

(xi) modify the definition of "Interest Period" without the consent of each affected Lender;

and <u>provided</u>, <u>further</u>, that (A) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document and (B) the Payment Letters may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 409 of 432

(b) <u>Defaulting Lenders</u>. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except those described in Section 11.01(a)(iii), (iv) and (v).

(c) <u>Replacement of Non-Consenting Lenders</u>. If any Lender does not consent to a proposed amendment, waiver, consent or release with respect to any Loan Document that requires the consent of each Lender and that has been approved by the Required Lenders, the Borrower may replace such non-consenting Lender in accordance with <u>Section 11.13</u>; <u>provided</u> that such amendment, waiver, consent or release can be effected as a result of the assignment contemplated by such Section (together with all other such assignments required by the Borrower to be made pursuant to this paragraph).

11.02 <u>Notices; Effectiveness; Electronic Communications</u>.

(a) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in subsection (b) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

(i) if to Holdings, the Borrower, any other Loan Party or the Administrative Agent, to the address, facsimile number, electronic mail address or telephone number specified for such Person on <u>Schedule 11.02</u>; and

(ii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications to the extent provided in subsection (b) below shall be effective as provided in subsection (b).

(b) <u>Electronic Communications</u>. Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall not apply to notices to any Lender pursuant to <u>Article II</u> if such Lender has notified the Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), provided that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 410 of 432

intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.

The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." (c) THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to Holdings, the Borrower, any Lender or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to Holdings, the Borrower, any Lender or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

(d) <u>Change of Address, Etc.</u> Each of Holdings, the Borrower and the Administrative Agent may change its address, facsimile or telephone number for notices and other communications hereunder by notice to the other parties hereto. Each Lender may change its address, facsimile or telephone number for notices and other communications hereunder by notice to the Borrower and the Administrative Agent. In addition, each Lender agrees to notify the Administrative Agent from time to time to ensure that the Administrative Agent has on record (i) an effective address, contact name, telephone number, facsimile number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender.

(e) <u>Reliance by Administrative Agent and Lenders</u>. The Administrative Agent and the Lenders shall be entitled to rely and act upon any notices (including any telephonic Notice of Borrowing) purportedly given by or on behalf of the Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify the Administrative Agent, each Lender and the Related Parties of each of them from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower, except to the extent that such losses, costs, expenses or liabilities are determined by a final judgment of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of such Person. All telephonic notices to and other telephonic communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

11.03 <u>No Waiver; Cumulative Remedies</u>.

The Lenders and the Administrative Agents shall have other rights and remedies not inconsistent herewith as provided under the Uniform Commercial Code, by law, or in equity. No failure by any Lender or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder or under any other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 411 of 432

preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided, and provided under each other Loan Document, are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Loan Parties or any of them shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent in accordance with Section 8.02 for the benefit of all the Lenders; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) [reserved], (c) any Lender from exercising setoff rights in accordance with Section 11.08 (subject to the terms of Section 2.10), or (d) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to any Loan Party under any Debtor Relief Law; and provided, further, that if at any time there is no Person acting as Administrative Agent hereunder and under the other Loan Documents, then (i) the Required Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 8.02 and (ii) in addition to the matters set forth in clauses (b), (c) and (d) of the preceding proviso and subject to Section 2.10, any Lender may, with the consent of the Required Lenders, enforce any rights and remedies available to it and as authorized by the Required Lenders.

11.04 Expenses; Indemnity; Damage Waiver.

Costs and Expenses. The Borrower shall pay (i) all reasonable out-of-pocket expenses (a) incurred by the Administrative Agent and its Affiliates (including the reasonable fees, charges and disbursements of counsel for the Administrative Agent, but excluding any costs of maintaining the Platform), in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), in each case, with respect to legal fees and expenses, limited to Stroock & Stroock & Lavan LLP and one local counsel in each jurisdiction where necessary and (ii) all out-of-pocket expenses incurred by the Administrative Agent or any Lender (including the reasonable fees, charges and disbursements of any counsel for the Administrative Agent or any Lender), and shall pay all reasonable, allocated fees and time charges for attorneys (who may be employees of the Administrative Agent or any Lender) in connection with the enforcement or protection of its rights, in each case, with respect to legal fees and expenses, limited to Stroock & Stroock & Lavan LLP and one local counsel in each jurisdiction where necessary (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with Loans made, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans.

(b) <u>Indemnification by the Borrower</u>. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Lender and each Related Party of any of the foregoing Persons (each such Person being called an "<u>Indemnitee</u>") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, liabilities, penalties and related expenses (including the reasonable fees, charges and disbursements of any counsel for any Indemnitee), and shall indemnify and hold harmless each Indemnitee from all reasonable, allocated fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any third party or by the Borrower or any other Loan Party arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 412 of 432

instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or the consummation of the transactions contemplated hereby or thereby, or, in the case of the Administrative Agent (and any sub-agent thereof) and its Related Parties only, the administration of this Agreement and the other Loan Documents, (ii) any Loan or the use or proposed use of the proceeds therefrom or in the Chapter 11 Cases, (iii) any actual or alleged presence or Release of Hazardous Materials on, at, under or from any property owned or operated by the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by the Borrower or any other Loan Party or any of the Borrower's or such Loan Party's directors, shareholders or creditors, and regardless of whether any Indemnitee is a party thereto, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF THE INDEMNITEE; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee.

(c) <u>Reimbursement by Lenders</u>. To the extent that the Borrower for any reason fails to pay any amount required under subsection (a) or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), or any Related Party, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), or such Related Party, as the case may be, such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnify payment is sought) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent) in connection with such capacity.

(d) Waiver of Consequential Damages, Etc. To the fullest extent permitted by applicable Law, neither the Borrower nor any Loan Party shall assert, and each of the Borrower and the Loan Parties hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof. No Indemnite referred to in subsection (b) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final judgment of a court of competent jurisdiction. For the avoidance of doubt, this subsection (d) shall not limit the obligation of the Borrower to indemnify each Indemnitee for any liabilities or damages incurred by such Indemnitee that are asserted against such Indemnitee by a third party and that are payable by the Borrower pursuant to Section 11.04(b).

(e) <u>Payments</u>. All amounts due under this Section shall be payable not later than ten Business Days after demand therefor.

(f) <u>Survival</u>. The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all the other Obligations.

11.05 Payments Set Aside.

To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent or any Lender, or the Administrative Agent or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, <u>plus</u> interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders under clause (b) of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

11.06 Successors and Assigns.

Successors and Assigns Generally. The provisions of this Agreement shall be binding (a) upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that neither the Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder or under the other Loan Documents (except in connection with any transaction permitted by Section 7.04(a), (d) or (e)) without the prior written consent of the Administrative Agent and each Lender, and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of Section 11.06(b), (ii) by way of participation in accordance with the provisions of Section 11.06(d), or (iii) by way of pledge or assignment of a security interest subject to the restrictions of Section 11.06(f) (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of the Administrative Agent and each of the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment(s) and the Loans at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

(i) <u>Minimum Amounts</u>.

(A) in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and

(B) in any case not described in subsection (b)(i)(A) of this Section, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the outstanding principal balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 414 of 432

Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); <u>provided, however</u>, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met;

(ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;

(iii) <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by subsection (b)(i)(B) of this Section and, in addition:

(A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (1) an Event of Default has occurred and is continuing at the time of such assignment or (2) such assignment is to a Lender, a holder of 6% Notes, an Affiliate of a Lender or of a holder of 6% Notes, or an Approved Fund; provided, that if the Borrower fails to respond to a request for a consent to an assignment within one (1) Business Day following the date such request is received by the Borrower, then the Borrower shall be deemed to have consented to such assignment; and

(B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of any Commitment or any Loan.

(iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount, if any, required as set forth in <u>Schedule 11.06</u>; <u>provided</u>, <u>however</u>, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. In addition, each assignee shall be required to (x) be a holder of the 6% Notes and (y) become a party to the Restructuring Support Agreement prior and as a condition to the assignment of any Loans or Commitments to such assignee becoming effective.

(v) <u>No Assignment to the Borrower</u>. No such assignment shall be made to the Borrower or any of the Borrower's Affiliates or Subsidiaries.

(vi) <u>No Assignment to Natural Persons</u>. No such assignment shall be made to a natural person.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (\underline{c}) of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 415 of 432

an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 3.01, 3.04, 3.05 and 11.04 with respect to facts and circumstances occurring prior to the effective date of such assignment). Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 11.06(d).

Notwithstanding anything to the contrary herein, each initial Lender that is a signatory to this Agreement on the Closing Date agrees that (x) after the funding of the Initial Loans on the Closing Date, it shall assign Loans and Commitments held by such Lender (after giving effect to the funding of the Initial Loans), in an amount equal to its Pro Rata Share of an amount to be determined by the Lender Advisors, to one or more Persons in connection with the initial syndication (the "<u>Syndication</u>") of the Loans and Commitments to those parties to the Restructuring Support Agreement that are not Initial Lenders and (y) unless and until such Lender shall have complied with foregoing clause (x) and the Syndication shall have been consummated, it shall not assign, transfer or sell any participations in any Loans or Commitments held by it to any Person (other than in connection with the Syndication).

(c) <u>Register</u>. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal and interest amounts of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and, with respect to itself, any Lender, at any reasonable time and from time to time upon reasonable prior notice. No assignments shall be effective unless and until recorded in the Register.

Participations. Any Lender may at any time, without the consent of, or notice to, the (d) Borrower or the Administrative Agent, sell participations to any Person (other than a natural person or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent and the Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in clauses (iii) through (vi) of the first proviso to Section 11.01(a) that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.01, 3.04 and 3.05 (subject to the requirements and limitations of those Sections, including the documentation requirements in Section 3.01(e), and Section 3.06) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 11.06(b). To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.08 as though it were a Lender, provided such Participant complies with Section 2.10 as though it were a Lender.

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 416 of 432

Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal and interest amounts of each Participant's interest in the Loans or other obligations under this Agreement (the "<u>Participant Register</u>"). The entries in the Participant Register shall be conclusive and such Lender (and the Borrower, to the extent that the Participant requests payment from the Borrower) shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. No Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations.

The parties hereto agree and intend that Section 11.06(c) and Section 11.06(d) shall be construed so that the Obligations are treated as in "registered form" for the purposes of the Code (including Code Sections 163(f), 871(h)(2), and 881(c)(2)), and the Register and Participant Register shall be maintained in accordance with such intention.

(e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Section 3.01</u>, <u>3.04</u> or <u>3.05</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent or except to the extent such entitlement to a greater payment results from a Change in Law occurring after such Participant became a Participant.

(f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank or other central bank having jurisdiction over such Lender; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

(g) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

(h) [Reserved].

11.07 <u>Treatment of Certain Information; Confidentiality.</u>

Each of the Administrative Agent and the Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 417 of 432

required by applicable Laws or regulations or by any subpoena or similar legal process (<u>provided</u>, that in the event of any such disclosure under this clause (c), the Administrative Agent or such Lender, as the case may be, agrees to use commercially reasonable efforts to inform the Borrower of such disclosure to the extent not prohibited by Law), (d) to any other party hereto, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section or (ii) becomes available to the Administrative Agent, any Lender, or any of their respective Affiliates on a nonconfidential basis from a source other than the Borrower (other than through a Person whom the Administrative Agent or such Lender actually knows to be acting in violation of his or its obligations to the Borrower or any other Loan Party).

For purposes of this Section, "<u>Information</u>" means all information received from any Loan Party or any Subsidiary thereof relating to any Loan Party or any Subsidiary thereof or their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by any Loan Party or any Subsidiary thereof, <u>provided</u> that, in the case of information received from a Loan Party or any such Subsidiary after the Closing Date, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

The Administrative Agent and each of the Lenders acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including Federal and state securities Laws.

11.08 Right of Setoff.

Subject to the Orders, if an Event of Default shall have occurred and be continuing, each Lender and each of their respective Affiliates is hereby authorized at any time and from time to time, after obtaining the prior written consent of the Administrative Agent, to the fullest extent permitted by applicable law, to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by such Lender or any such Affiliate to or for the credit or the account of Holdings, the Borrower or any other Loan Party against any and all of the obligations of Holdings, the Borrower or such Loan Party now or hereafter existing under this Agreement or any other Loan Document to such Lender, irrespective of whether or not such Lender shall have made any demand under this Agreement or any other Loan Document and although such obligations of Holdings, the Borrower or such Loan Party may be contingent or unmatured or are owed to a branch or office of such Lender different from the branch or office holding such deposit or obligated on such indebtedness; provided, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.11(b) and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in

reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff. The rights of each Lender and their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff) that such Lender or its Affiliates may have. Each Lender agrees to notify the Borrower and the Administrative Agent promptly after any such setoff and application, <u>provided</u> that the failure to give such notice shall not affect the validity of such setoff and application.

11.09 Interest Rate Limitation.

Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Law (the "<u>Maximum Rate</u>"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

11.10 <u>Counterparts; Effectiveness</u>.

This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Except as provided in <u>Section 4.01</u>, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or electronic transmission shall be effective as delivery of a manually executed counterpart of this Agreement). Except as provided in <u>Section 4.01</u>, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of this Agreement).

11.11 <u>Survival of Representations and Warranties</u>.

All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or on their behalf and notwithstanding that the Administrative Agent or any Lender or knowledge of any Default at the time of any Borrowing, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied.

11.12 <u>Severability</u>.

If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 419 of 432

provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

11.13 <u>Replacement of Lenders</u>.

(a) If any Lender requests compensation under <u>Section 3.04</u>, or if the Borrower is required to pay any Indemnified Taxes or additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, if any Lender is a Defaulting Lender, or if any other circumstance exists hereunder that gives the Borrower the right to replace a Lender as a party hereto, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, <u>Section 11.06</u> (notwithstanding the foregoing, it being understood that such assignment shall not require the consent of such Lender), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignment), <u>provided</u> that:

(i) the Borrower shall have paid to the Administrative Agent the processing and recordation fee specified in <u>Section 11.06(b)</u>;

(ii) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>Sections 3.01, 3.04</u> or <u>3.05</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);

(iii) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.04</u> or payments required to be made pursuant to <u>Section 3.01</u>, such assignment will result in a reduction in such compensation or payments thereafter; and

(iv) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

(b) Notwithstanding anything herein to the contrary, the Borrower may at any time terminate in full the Commitment of any Lender that is a Designated Lender (without any requirement of any reduction in the Commitments of any other Lenders) by giving notice to such Designated Lender and the Administrative Agent (which notice shall specify the effective date of such termination); provided, that (i) at the time of such termination, no Event of Default exists and (ii) concurrently with such termination, the Borrower shall pay to such Designated Lender an amount equal to the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents. Upon satisfaction of the conditions specified in the foregoing clauses (i) and (ii), the Commitment of such Designated Lender shall terminate on the effective date specified in such notice. The termination of the Commitment of a Designated Lender pursuant to this <u>Section 11.13(b)</u> shall not be deemed to be a waiver of any right that the Borrower, the Administrative Agent or any Lender may have against such Designated Lender.

11.14 <u>Governing Law; Jurisdiction; Etc</u>.

(a) <u>GOVERNING LAW</u>. THE VALIDITY, CONSTRUCTION, INTERPRETATION AND ENFORCEMENT OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANY SUCH OTHER LOAN DOCUMENT IN RESPECT THEREOF) AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT AND SUCH OTHER LOAN DOCUMENTS OR RELATED HERETO OR THERETO SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK AND, AS MAY BE APPLICABLE, THE BANKRUPTCY CODE.

SUBMISSION TO JURISDICTION. EACH OF THE PARTIES HERETO AGREES (b) THAT ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL BE BROUGHT IN THE BANKRUPTCY COURTS OF THE STATE OF NEW YORK, AND, BY EXECUTION AND DELIVERY OF THIS AGREEMENT, EACH LOAN PARTY HEREBY IRREVOCABLY ACCEPTS IN RESPECT OF ITS PROPERTY, GENERALLY AND UNCONDITIONALLY, THE JURISDICTION OF THE AFORESAID COURTS; PROVIDED, HOWEVER, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT THE ADMINISTRATIVE AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE THE ADMINISTRATIVE AGENT ELECTS TO BRING SUCH ACTION OR WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. NOTWITHSTANDING ANY OTHER PROVISION OF THIS SECTION 11.14, THE BANKRUPTCY COURT SHALL HAVE EXCLUSIVE JURISDICTION OVER ANY ACTION OR DISPUTE INVOLVING, RELATING TO OR ARISING OUT OF THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS; PROVIDED, THAT NOTHING IN THIS SECTION 11.14 SHALL AFFECT THE RIGHT OF THE ADMINISTRATIVE AGENT OR ANY LENDER TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE REQUIREMENTS OF LAW OR COMMENCE LEGAL PROCEEDINGS OR OTHERWISE PROCEED AGAINST ANY LOAN PARTY OR ANY COLLATERAL IN ANY OTHER JURISDICTION.

(c) <u>WAIVER OF VENUE</u>. THE BORROWER AND EACH OTHER LOAN PARTY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY COURT REFERRED TO IN PARAGRAPH (b) OF THIS SECTION 11.14. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF <u>SUCH ACTION</u> <u>OR PROCEEDING IN ANY SUCH COURT.</u>

(d) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 11.02</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

11.15 <u>Waiver of Jury Trial</u>.

EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 421 of 432

THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

11.16 <u>No Advisory or Fiduciary Responsibility</u>.

In connection with all aspects of each transaction contemplated hereby, the Borrower and Holdings each acknowledge and agree, and acknowledge their respective Affiliates' understanding, that: (i) the credit facility provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between the Borrower, Holdings and their respective Affiliates, on the one hand, and the Administrative Agent, on the other hand, and each of the Borrower and Holdings is capable of evaluating and understanding and understands and accepts the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver or other modification hereof or thereof); (ii) in connection with the process leading to such transaction, the Administrative Agent is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary for the Borrower, Holdings or any of their respective Affiliates, stockholders, creditors or employees or any other Person; (iii) the Administrative Agent has not assumed or will assume an advisory, agency or fiduciary responsibility in favor of the Borrower or Holdings with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any other Loan Document (irrespective of whether the Administrative Agent has advised or is currently advising the Borrower, Holdings or any of their respective Affiliates on other matters) and the Administrative Agent has no obligation to the Borrower, Holdings or any of their respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; (iv) the Administrative Agent and its Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower, Holdings and their respective Affiliates, and the Administrative Agent has no obligation to disclose any of such interests by virtue of any advisory, agency or fiduciary relationship; and (v) the Administrative Agent has not provided and will not provide any legal, accounting, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Loan Document) and each of the Borrower and Holdings has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. Each of the Borrower and Holdings hereby waives and releases, to the fullest extent permitted by law, any claims that it may have against the Administrative Agent with respect to any breach or alleged breach of agency or fiduciary duty.

11.17 Judgment Currency.

If, for the purposes of obtaining judgment in any court in any jurisdiction with respect to this Agreement or any other Loan Document, it becomes necessary to convert into a particular currency (the "<u>Judgment Currency</u>") any amount due under this Agreement or under any other Loan Document in any currency other than the Judgment Currency (the "<u>Currency Due</u>"), then conversion shall be made at the rate of exchange prevailing on the Business Day before the day on which judgment is given. For this purpose "rate of exchange" means the rate at which the Administrative Agent is able, on the relevant date, to purchase the Currency Due with the Judgment Currency in accordance with its normal practice at its

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 422 of 432

head office in New York, New York. In the event that there is a change in the rate of exchange prevailing between the Business Day before the day on which the judgment is given and the date of receipt by the Administrative Agent of the amount due, the Borrower will, on the date of receipt by the Administrative Agent, pay such additional amounts, if any, or be entitled to receive reimbursement of such amount, if any, as may be necessary to ensure that the amount received by the Administrative Agent on such date is the amount in the Judgment Currency that when converted at the rate of exchange prevailing on the date of receipt by the Administrative Agent is the amount then due under this Agreement or such other Loan Document in the Currency Due. If the amount of the Currency Due which the Administrative Agent is so able to purchase is less than the amount of the Currency Due originally due to it, the Borrower shall indemnify and save the Administrative Agent and the Lenders harmless from and against all loss or damage arising as a result of such deficiency. This indemnity shall constitute an obligation separate and independent from the other obligations contained in this Agreement and the other Loan Documents, shall give rise to a separate and independent cause of action, shall apply irrespective of any indulgence granted by the Administrative Agent from time to time and shall continue in full force and effect notwithstanding any judgment or order for a liquidated sum in respect of an amount due under this Agreement or any other Loan Document or under any judgment or order.

11.18 USA PATRIOT Act Notice.

Each Lender that is subject to the USA PATRIOT Act and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies each Loan Party that pursuant to the requirements of the USA PATRIOT Act, it is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify each Loan Party in accordance with the USA PATRIOT Act. The Loan Parties shall, promptly following a request by the Administrative Agent or any Lender, provide all documentation and other information that the Administrative Agent or such Lender requests in order to comply with its ongoing obligations under applicable "know your customer" an anti-money laundering rules and regulations, including the USA PATRIOT Act.

11.19 Acknowledgement and Consent to Bail-In of EEA Financial Institutions.

Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any EEA Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the write-down and conversion powers of an EEA Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Write-Down and Conversion Powers by an EEA Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an EEA Financial Institution; and

- (b) the effects of any Bail-in Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;

(ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEA Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 423 of 432

(iii) the variation of the terms of such liability in connection with the exercise of the write-down and conversion powers of any EEA Resolution Authority.

11.20 ENTIRE AGREEMENT.

THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

11.21 [Reserved].

11.22 Subject to Orders.

Notwithstanding anything herein or in any other Loan Document to the contrary, (a) the provisions of this Agreement and the other Loan Documents (excluding from the definition thereof, solely for purposes of this Section 11.22, the Orders) are subject to the terms, covenants, conditions and provisions of the Orders, which, among other things, set out the priority, with respect to the Collateral, as between, (i) on the one hand, the Prepetition Secured Parties and, on the other hand, the Administrative Agent, the Lenders and any other Secured Parties and (ii) on the one hand, the Administrative Agent, the Lenders and any other Secured Parties pursuant to the Collateral Documents (excluding from the definition thereof, solely for purposes of this Section 11.22, the Orders) are expressly subject to the Orders, (c) the rights and remedies of the Administrative Agent hereunder or under any other Loan Document are in addition to rights and remedies of Administrative Agent hereunder or under any other Loan Document is subject to the limitations and provisions of the Orders. In the event of any conflict between the terms of, on the one hand, this Agreement or any other Loan Document and, on the other hand, the Agreement or any other Corders.

[Remainder of Page Intentionally Left Blank]

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 424 of 432

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

CENVEO CORPORATION, a Delaware corporation

By: _____

Name: Title:

CENVEO, INC., a Colorado corporation

By: _

Name: Title: WILMINGTON SAVINGS FUND SOCIETY, FSB, as Administrative Agent

By: _

Name: Title:

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 426 of 432

[], as a Lender

By: _____

Name: Title: 18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 427 of 432

EXHIBIT 3

Initial Budget

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 428 of 432

Project Chelsea - Weekly Cash Flow Forecast

Dollars US	D		-	_			_	-		_	-	_			
		Week 1 2/3	Week 2 2/10	Week 3 2/17	Week 4 2/24	Week 5 3/3	Week 6 3/10	Week 7 3/17	Week 8 3/24	Week 9 3/31	Week 10 4/7	Week 11 4/14	Week 12 4/21	Week 13 4/28	Total 2/3
		2/9	2/16	2/23	3/2	3/9	3/16	3/23	3/30	4/6	4/13	4/20	4/27	5/4	5/4
	CONSOLIDATED	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast
s	AR Collection	\$ 15,870	\$ 15,679	\$ 15,299	\$ 18,258	\$ 23,237	\$ 31,505	\$ 26,347	\$ 26,347	\$ 26,391	\$ 26,183	\$ 27,404	\$ 28,056	\$ 22,636	\$ 303,213
Cash Receipts	Non-Trade	119	49	31	22	40	18	531	607	733	1,221	733	1,466	1,059	6,627
Rec	Waste	303	303		607	76	607		683	-	326	407	489		3,800
!	Total Cash Receipts	16,293	16,031	15,330	18,886	23,353	32,129	26,878	27,637	27,124	27,730	28,544	30,010	23,695	313,641
Operating Disbursements	Vendor Payments	(11,374)	(11,169)	(14,124)	(14,279)	(16,546)	(15,758)	(11,069)	(7,201)	(18,775)	(15,812)	(15,087)	(16,683)	(13,809)	(181,685)
me	Payroll	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(11,000)	(2,700)	(5,700)	(2,700)	(15,000)	(2,700)	(73,000)
ias.	Taxes	(100)	(400)	(1,000)	(600)	(100)	(400)	(1,000)	(600)	-	(100)	(1,300)	(500)	(200)	(6,300)
Ē	Benefits	(1,708)	(1,108)	(608)	(1,708)	(1,708)	(1,108)	(708)	(1,708)	(1,708)	(1,708)	(1,192)	(1,608)	(1,808)	(18,384)
lsi	Rent	(1,300)	-	-	(1,400)	-	-	-	(100)	(1,300)	-	-	-	(1,400)	(5,500)
I gl	Other Payments	(500)	(500)	(400)	(800)	(500)	(500)	(500)	(500)	(800)	(500)	(500)	(500)	(800)	(7,300)
Ę.	Transfers to Foreign Subsidiaries	-	-	-	(1,100)	-	-	-	-	(1,100)	-	-	-	(1,100)	(3,300)
era	Plant Closures	(250)	(250)	(250)	(250)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(750)	(7,750)
do	Total Operating Cash Disbursements	(17,932)	(19,126)	(19,082)	(31,137)	(22,304)	(24,216)	(16,727)	(21,859)	(27,133)	(24,570)	(21,528)	(35,041)	(22,566)	(303,219)
	OPERATING CASH FLOW	(1,639)	(3,096)	(3,751)	(12,250)	1,049	7,914	10,151	5,778	(9)	3,160	7,016	(5,031)	1,129	10,421
1	Proceeds for Issuance of Debt													-	
Debt Service	Principal														
la ja	Interest and Financing Fees	(5,380)	(10)	(500)	(939)	-		-	(968)	(500)	-	-	-	(944)	(9,241)
- s	Total Financing Cash Receipts / Disbursements		(10)	(500)	(939)				(968)	(500)				(944)	(9,241)
	Total Financing Cash Receipts / Disbursement	(5,380)	(10)	(500)	(939)		-		(908)	(500)	-	-	-	(944)	(9,241)
-	LSC Vendor Reimbursement	1,000	1,000	1,000	1,000	1,000	1,000	1,000	1,000	-	-	-	-	-	8,000
QP LSC / TSA	LSC / QP Related AP	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	(1,000)	-	-	-		(9,000)
\sim	LSC Customer Collection	2,000	2,000	2,000	2,000	2,000	2,000	2,000	2,000	-	-	-	-	-	16,000
ITSC	LSC / QP Related AR	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	(2,000)	-	-	-	-	(18,000)
£,	QP Sale Proceeds	-	-	-	-	-	-	-	-	-	-	-	-	-	-
-	Total QP Disbursements / Adjustments	-	-	-	-	-	-	-	-	(3,000)	-	-	-	-	(3,000)
1	Restructuring Professional Fees			(160)	(160)			(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	(7,820)
	Utility Deposit	(1,400)		-	-	-		-	-	-	-	-	-		(1,400)
Other	DIP Proceeds	50,000		50,000	-				-	-		<u> </u>	-	<u> </u>	100,000
õ	Total Other Disbursements / Adjustments	48,600	-	49,840	(160)	-		(160)	(495)	(425)	(2,415)	(2,425)	(1,085)	(495)	90,780
1	NET CASH FLOW	41,580	(3,105)	45,589	(13,349)	1,049	7,914	9,991	4,315	(3,934)	745	4,591	(6,116)	(310)	88,960
1	Borrowing Base	146,000	146,000	146,000	147,953	147,953	147,953	147,953	159,976	159,976	159,976	159,976	159,976	149,957	149,957
	Professional Fee Reserve	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)	(15,100)
	Net Borrowing Base	130,900	130,900	130,900	132,853	132,853	132,853	132,853	144,876	144,876	144,876	144,876	144,876	134,857	134,857
	Starting ABL Outstanding	120,164	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	120,164
ξ	ABL Draw / (Repayment)	(41,580)	3,105	(45,589)	13,349	(1,049)	(7,914)	(9,991)	(4,315)	3,934	(745)	(4,591)	6,116	310	(88,960)
idi	Ending ABL Outstanding	78,584	81,689	36,100	49,450	48,400	40,486	30,495	26,180	30,114	29,369	24,778	30,894	31,204	31,204
Liquidity	Plus : L/Cs Outstanding	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600	16,600
-	Borrowings + L/Cs	95.184	98,289	52,700	66.050	65.000	57,086	47,095	42,780	46,714	45,969	41,378	47,494	47,804	47.804
	Ending ABL Availability	35,716	32,611	78,200	66,803	67,853	75,767	85,758	102,096	98,162	98,907	103,498	97,382	87,053	87,053
	Less: Outstanding Check Float	(2,000)	(4,000)	(6,000)	(7,000)	(7,000)	(7,000)	(7,000)	(7,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)	(4,000)
	Plus: Unrestricted Cash	-	(4,000)	-	-	-	-	-	-	-	(4,000)	-	(1,000)	(4,000)	(4,000)
	LIQUIDITY	33,716	28,611	72,200	59,803	60,853	68,767	78,758	95,096	94,162	94,907	99,498	93,382	83,053	83,053
	· · · ·	56,715	10,011	,_ 00	2,000	20,000	20,707	. 0,700	10,070	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,,,,,,,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	50,002	50,000	00,000

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 429 of 432

EXHIBIT 4

Weekly Professional Fee Schedule

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 430 of 432

Weekly Professional Fee Schedule \$ in 000s

	Week 0	1. S.	Week 1	Week 2	Week 3	Week 4	Week 5	Week 6	Week 7	Week 8	Week 9	Week 10	Week 11	Week 12	Week 13
	2/1		2/3	2/10	2/17	2/24	3/3	3/10	3/17	3/24	3/31	4/7	4/14	4/21	4/28
	2/2		2/9	2/16	2/23	3/2	3/9	3/16	3/23	3/30	4/6	4/13	4/20	4/27	5/4
	Forecast		Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast
Debtor Professionals		_	_		-	-	-	-	-	-	425	1,650	2,360		-
UCC Professionals		-	-	-	-	-	-	-	-	-	-	-	65	410	-
Debtor + UCC Payments	\$	- \$		ş –	\$ - \$	- \$		\$ -	\$ -	\$ -	\$ 425 \$	\$ 1,650 \$	2,425 \$	410	\$ -
Debtor Professionals		244	2,304	854	854	774	575	575	575	575	610	616	616	616	585
UCC Professionals		42	148	148	148	165	208	208	208	208	124	110	110	110	108
Total Debtor + UCC		286	2,451	1,001	1,001	939	782	782	782	782	734	726	726	726	693
Cumulative Payments	\$	- \$	- 3	\$-	\$ - \$	- \$; -	\$-	\$-	\$ -	\$ 425 \$	\$ 2,075 \$	4,500 \$	\$ 4,910	\$ 4,910
Cumulative Accruals	\$	286 \$	2,737	\$ 3,739	\$ 4,740 \$	5,679 \$	6,461	\$ 7,243	\$ 8,026	\$ 8,808	\$ 9,542 \$	§ 10,268 \$	10,993	5 11,719	\$ 12,412
Total Accrued & Unpaid	\$	(286) \$	(2,737)	\$ (3,739)	\$ (4,740) \$	(5,679) \$	6 (6,461)	\$ (7,243)	\$ (8,026)	\$ (8,808)	\$ (9,117) \$	\$ (8,193) \$	(6,493)	6 (6,809)	\$ (7,502)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 431 of 432

Weekly Professional Fee Schedule \$ in 000s

	We	eek 14	Week 15	Week 16	Week 17	Week 18	Week 19	Week 20	Week 21	Week 22	Week 23	Week 24	Week 25	Week 26	Week 27
		5/5	5/12	5/19	5/26	6/2	6/9	6/16	6/23	6/30	7/7	7/14	7/21	7/28	8/4
	5	5/11	5/18	5/25	6/1	6/8	6/15	6/22	6/29	7/6	7/13	7/20	7/27	8/3	8/10
	Fa	wrecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast
Debtor Professionals		190	2,025	-		335	1,845	130	-	185	1,250	725			210
UCC Professionals		-	-,	740	-	-	-	130	250		-	130	250	-	
Debtor + UCC Payments	\$	190 \$	2,025	\$ 740	\$ -	\$ 335	\$ 1,845	\$ 260	\$ 250	\$ 185	\$ 1,250 \$	\$ 855 \$	250	ş -	\$ 210
Debtor Professionals		562	562	562	566	587	587	587	587	740	765	765	765	680	566
UCC Professionals		106	106	106	107	110	110	110	110	107	106	106	106	125	151
Total Debtor + UCC		668	668	668	672	697	697	697	697	847	872	872	872	805	717
Cumulative Payments	\$	5,100 \$	7,125	\$ 7,865	\$ 7,865	\$ 8,200	\$ 10,045	\$ 10,305	\$ 10,555	\$ 10,740	\$ 11,990 \$	\$ 12,845 \$	13,095	\$ 13,095	\$ 13,305
Cumulative Accruals	\$	13,080 \$	13,749	\$ 14,417	\$ 15,090	\$ 15,786	\$ 16,483	\$ 17,179	\$ 17,876	\$ 18,722	\$ 19,594 \$	\$ 20,465 \$	21,337	\$ 22,142	\$ 22,859
Total Accrued & Unpaid	\$	(7,980) \$	(6,624)	\$ (6,552)	\$ (7,224)	\$ (7,586)	\$ (6,437)	\$ (6,874)	\$ (7,320)	\$ (7,982)	\$ (7,604) \$	\$ (7,620) \$	(8,242)	\$ (9,047)	\$ (9,554)

18-22178-rdd Doc 51-1 Filed 02/06/18 Entered 02/06/18 17:00:35 Exhibit Pg 432 of 432

Weekly Professional Fee Schedule \$ in 000s

	N	Week 28	W	7eek 29	Week 30	We	eek 31	Week 32	Week 33	Week 34	Week 35	Week 36	Week 37	Week 38
		8/11		8/18	8/25	1	9/1	9/8	9/15	9/22	9/29	10/6	10/13	10/20
		8/17		8/24	8/31	9	9/7	9/14	9/21	9/28	10/5	10/12	10/19	10/26
		Forecast	I	Forecast	Forecast	Fa	recast	Forecast						
Debtor Professionals		3,395		-	_		535	2,295	130	2,875	200	1,250	725	4,697
UCC Professionals		475		380	-		-	-,_,~	130	250		-	130	1,910
Debtor + UCC Payments	\$	3,870	\$	380	\$ -	\$	535	\$ 2,295	\$ 260	\$ 3,125	\$ 200	\$ 1,250	\$ 855	\$ 6,607
Debtor Professionals		566		566	566		663	663	663	3,538	784	832	832	5,529
UCC Professionals		151		151	151		215	215	215	215	213	213	213	1,713
Total Debtor + UCC		717		717	717		877	877	877	3,752	997	1,045	1,045	7,241
Cumulative Payments	\$	17,175	\$	17,555	\$ 17,555	\$	18,090	\$ 20,385	\$ 20,645	\$ 23,770	\$ 23,970	\$ 25,220	\$ 26,075	\$ 32,682
Cumulative Accruals	\$	23,576	\$	24,293	\$ 25,010	\$	25,887	\$ 26,765	\$ 27,642	\$ 31,394	\$ 32,391	\$ 33,436	\$ 34,480	\$ 41,722
Total Accrued & Unpaid	\$	(6,401)	\$	(6,738)	\$ (7,455)	\$	(7,797)	\$ (6,380)	\$ (6,997)	\$ (7,624)	\$ (8,421)	\$ (8,216)	\$ (8,405)	\$ (9,040)